As filed with the Securities and Exchange	Registration No. 001-16427	
SECURITIES AND EXCHANG Washington, D.C.		
AMENDMENT NO. 2 TO	2	
FORM 10		
GENERAL FORM FOR REGISTRATI PURSUANT TO SECTION 12(b THE SECURITIES EXCHANGE) OR 12(g) OF	
CERTEGY INC. (Exact Name of Registrant as Spe	ecified in Its Charter)	
Georgia (State or other jurisdiction of incorporation or organization)	58-2606325 (I.R.S. Employer Identification No.)	
1550 Peachtree Street, N.W., Atlanta, Georgi (Address of principal executive offices)	.a 30309 (Zip Code)	
(404) 885-8000 (Registrant's telephone number, including area code)		
Securities to be registered pursuant to Section 12(b) of the Act:		
Title of Each Class to be so Registered:	Name of Each Exchange on Which Each Class is to be Registered:	
Common Stock, par value \$.01 per share	New York Stock Exchange	
Common Stock Purchase Rights	New York Stock Exchange	
Securities to be registered pursuant to Section 12(g) of the Act:		
None.		

CERTEGY INC.

I. INFORMATION INCLUDED IN INFORMATION STATEMENT AND INCORPORATED IN FORM 10 BY REFERENCE

Our Information Statement may be found as Exhibit 99.1 to the Form 10. For your convenience, we have provided below a cross-reference sheet identifying where the items required by Form 10 can be found in the Information Statement.

CROSS REFERENCE SHEET BETWEEN INFORMATION STATEMENT AND ITEMS OF FORM 10

Item No.	Item Caption	Location in Information Statement
1.	Business	"Summary," "The Distribution," "Risk Factors," "Forward- Looking Statements," "Capitalization," "Management's Discussion and Analysis of Financial Condition and Results of Operations," "Business," and "Relationship Between Equifax and Our Company After the Distribution."
2.	Financial Information	"Summary," "Capitalization," "Selected Financial Data," "Management's Discussion and Analysis of Financial Condition and Results of Operations," "Combined Financial Statements of Equifax Payment Services division," and "Pro Forma Combined Financial Statements of Equifax Payment Services division."
3.	Properties	"BusinessProperties."
4.	Security Ownership of Certain Beneficial Owners and Management	"The Distribution," "Management," and "Ownership of Our Common Stock."
5.	Directors and Executive Officers	"Management."
6.	Executive Compensation	"Management," and "Ownership of Our Common Stock."
7.	Our Relationships and Related Transactions	"Summary," and "Relationship Between Equifax and Our Company After the Distribution."
8.	Legal Proceedings	"BusinessLegal Proceedings."
9.	Market Price of and Dividends on the Registrant's Common Equity and Related Shareholder Matters	"The Distribution," "Dividend Policy," and "Description of Capital Stock."
10.	Recent Sales of Unregistered Securities	Not Included (See Part II Below).
11.	Description of Registrant's Securities to be Registered	"The Distribution," "Dividend Policy," and "Description of Capital Stock."
12.	Indemnification of Directors and Officers	"Indemnification of Directors and Officers."

Item
No. Item Caption
Statement

13. Financial Statements and Supplementary Data

"Summary," "Selected Financial Data," and "Index to Financial Statements."

14. Changes In and Disagreements with Accountants on Accounting and Financial Matters

Not Applicable.

15. Financial Statements and Exhibits "Index to Financial Statements."

II. INFORMATION NOT INCLUDED IN INFORMATION STATEMENT

Item 10. Recent Sales of Unregistered Securities.

We were incorporated under the laws of the State of Georgia under the name "Equifax PS, Inc." on March 2, 2001. We issued 1,000 shares of common stock, \$.01 par value per share, to Equifax Inc., a Georgia corporation, in consideration of a capital contribution of \$10.00 by Equifax Inc. This issuance was exempt from registration pursuant to Section 4(2) of the Securities Act of 1933, as amended, because it did not involve any public offering of securities. As of May 25, 2001, our name was changed to "Certegy Inc."

Item 15. Financial Statements and Exhibits.

(a) List of Financial Statements. The following financial statements are included or incorporated by reference in the Information Statement filed as part of this Registration Statement on Form 10:

Description	Page
Equifax Payment Services division (To be reorganized as Certegy Inc.) Historical:	
Report of Independent Public Accountants	F-2
and 1998	F-3
31, 2000 and 1999	F-4
1999, and 1998	F-5
December 31, 2000, 1999, and 1998 Notes to Combined Financial Statements	
Report of Independent Public Accountants as to Schedule	F-20
Introduction to the Pro Forma Combined Financial Statements Pro Forma Combined Statement of Income for the Three Months ended March	F-22
31, 2001 Pro Forma Combined Statement of Income for the Year ended December 31,	F-23
2000 Pro Forma Combined Balance Sheet as of March 31, 2001 Notes to Pro Forma Combined Financial Statements	F-25

(b) Exhibits. The following documents are filed as exhibits to this Form 10:

Exhibit

No. Description

- 2.1 Distribution Agreement, Plan of Reorganization and Distribution**
- 3.1 Amended and Restated Articles of Incorporation of Certegy Inc.+
- 3.2 Amended and Restated Bylaws of Certegy Inc.+
- 4.1 Amended and Restated Articles of Incorporation of Certegy Inc. (filed as Exhibit 3.1)+
- 4.2 Amended and Restated Bylaws of Certegy Inc. (filed as Exhibit 3.2)+
- 4.3 Rights Agreement between Certegy Inc. and SunTrust Bank, as Rights Agreement**
- 4.4 Form of certificate representing Certegy Inc. Common Stock**
- 10.1 Distribution Agreement, Plan of Reorganization and Distribution (filed as Exhibit 2.1).**
- 10.2 Tax Sharing and Indemnification Agreement**
- 10.3 Employee Benefits Agreement**
- 10.4 Intercompany Data Purchase Agreement**
- 10.5 Transition Support Agreement**
- 10.6 Intellectual Property Agreement**
- 10.7 Certegy Inc. 2001 Stock Incentive Plan**
- 10.8 Certegy Inc. 2001 Key Management Long-Term Incentive Plan**
- 10.9 Certegy Inc. 2001 Non-Employee Directors Stock Incentive Plan**
- 10.10 Certegy Inc. Deferred Compensation Plan**
- 10.11 Tier I Change in Control Agreement**
- 10.12 Tier II Change in Control Agreement**
- 10.13 Certegy Inc. Executive Life and Supplemental Retirement Benefit $Plan^{**}$
- 10.14 Grantor Trust Agreement **
- 10.15 Agreement Regarding Leases**
- 21.1 List of Subsidiaries of Certegy Inc.**
- 99.1 Certegy Inc. Information Statement dated June 12, 2001**

^{**} Filed herewith.

⁺Previously filed.

SIGNATURE

Pursuant to the requirements of Section 12 of the Securities Exchange Act of 1934, the registrant has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized.

> CERTEGY INC. (Registrant)

By: /s/ Lee A. Kennedy

Name: Lee A. Kennedy Title: President and Chief Executive Officer

Dated: June 11, 2001

II-3

INDEX TO EXHIBITS

Exhibit No.	Description
2.1	Distribution Agreement, Plan of Reorganization and Distribution**
3.1	Amended and Restated Articles of Incorporation of Certegy Inc.+
3.2	Amended and Restated Bylaws of Certegy Inc.+
4.1	Amended and Restated Articles of Incorporation of Certegy Inc. (filed as Exhibit $3.1)+$
4.2	Amended and Restated Bylaws of Certegy Inc. (filed as Exhibit 3.2)+
4.3	Rights Agreement between Certegy Inc. and SunTrust Bank, as Rights Agreement**
4.4	Form of certificate representing Certegy Inc. Common Stock**
10.1	Distribution Agreement, Plan of Reorganization and Distribution (filed as Exhibit 2.1).**
10.2	Tax Sharing and Indemnification Agreement**
10.3	Employee Benefits Agreement**
10.4	Intercompany Data Purchase Agreement**
10.5	Transition Support Agreement**
10.6	Form of Intellectual Property Agreement**
10.7	Certegy Inc. 2001 Stock Incentive Plan**
10.8	Certegy Inc. 2001 Key Management Long-Term Incentive Plan**
10.9	Certegy Inc. 2001 Non-Employee Directors Stock Incentive Plan**
10.10	Certegy Inc. Deferred Compensation Plan**
10.11	Tier I Change in Control Agreement**
10.12	Tier II Change in Control Agreement**
10.13	Certegy Inc. Executive Life and Supplemental Retirement Benefit $Plan^{**}$
10.14	Grantor Trust Agreement**
10.15	Agreement Regarding Leases**
21.1	List of Subsidiaries of Certegy Inc.**
99.1	Certegy Inc. Information Statement dated June 12, 2001**

** Filed herewith.

 $[\]hbox{+Previously filed}.$

DISTRIBUTION AGREEMENT PLAN OF REORGANIZATION AND DISTRIBUTION

RECITALS

- A. Certegy is a wholly-owned subsidiary of Equifax formed for the purpose of taking title to the stock of certain Equifax subsidiaries, the assets and liabilities of which constitute the businesses of Equifax's Payment Services Group ("PSG").
- B. The Board of Directors of Equifax has determined that it is in the best interests of Equifax and its shareholders to transfer and assign to Certegy effective at and after the Effective Time (as defined herein) and as a contribution to the capital of Certegy, the capital stock of the Equifax subsidiaries that currently operate the PSG businesses and certain related assets and to receive in exchange therefor shares of Certegy Common Stock and the Borrowing Proceeds (each as defined herein).
- C. The Board of Directors of Equifax has further determined that it is in the best interests of Equifax and its shareholders to make a distribution (the "Distribution") to the holders of Equifax Common Stock (as defined herein) of all of the outstanding shares of Certegy Common Stock at the rate of one share of Certegy Common Stock for every two shares of Equifax Common Stock outstanding as of the Record Date (as defined herein).
- D. The parties intend that the Contribution constitute a reorganization described in Section 368(a)(1)(D) of the Code (as defined herein) and that the Distribution not be taxable to Equifax or its shareholders pursuant to Section 355 of the Code.
- E. The parties have determined that it is necessary and desirable to set forth the principal corporate transactions required to effect the Contribution and the Distribution and to set forth other agreements that will govern certain other matters following the Distribution.
- NOW, THEREFORE, in consideration of the foregoing premises and the mutual agreements and covenants contained in this Agreement and other good and valuable consideration the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

ARTICLE I DEFINITIONS

Section 1.01. Definitions. As used herein, the following terms have the

following meaning:

"Action" means any claim, suit, arbitration, inquiry, proceeding, or investigation by or before any court, governmental or other regulatory or administrative agency or commission or any other tribunal.

"Agreement Regarding Leases" means the Agreement Regarding Leases entered into at or prior to the Effective Time among Equifax, Payment Services, and the other entities entering into such agreement, as amended from time to time.

"Ancillary Agreements" means all of the written agreements, instruments, understandings, assignments and other arrangements (other than this Agreement) entered into in connection with the transactions contemplated hereby, including, without limitation, the Employee Benefits Agreement, the Transition Support Agreement, the Intercompany Data Purchase Agreement, the Intellectual Property Agreement, the Tax Sharing and Indemnification Agreement, the Real Estate Agreements, Agreement Regarding IBM UK Mainframe Services, Agreement Regarding IBM Australian Mainframe and Network Services, US and UK Network Services "Authorized User" Agreement, IBM United Kingdom Limited Transition Services Agreement, Agreement Regarding IBM Canadian Data Center Services Agreement and other documents relating to the transfer of assets and liabilities in contemplation of the Contribution and Distribution.

"Applicable Rate" means the Prime Rate plus 2%.

"Assets" means all properties, rights, contracts, leases and claims, of every kind and description, wherever located, whether tangible or intangible, and whether real, personal or mixed.

"Borrowing Proceeds" means proceeds from the Certegy Revolving Credit Agreement in the amount of \$275,000,000.

"Brazil Agreements" means the agreements, assignments and documents relating to the transfer by Equifax do Brasil Holdings, Ltd. of its ownership interests in Unnisa-Solucoes em Meios de Pagamento Ltda., Partech Ltd., and Equifax Cayman Islands Ltd. to Payment Brasil Holdings Ltda.

"Card Solutions France" means Equifax Card Solutions S.A., a wholly-owned subsidiary of Equifax organized under the laws of France.

"Certegy Articles" means the articles of incorporation of Certegy in the form filed as an exhibit to the Form 10 at the time it becomes effective.

"Certegy Assets" means (a) the capital stock of Payment Services, FSB, Telecredit Canada, Card Solutions France, Equifax Asia Pacific and Payment U.K. to be transferred at or prior to the Distribution Date by Equifax to Certegy, (b) the Real Estate Assets, and (c) except as otherwise provided in an Ancillary Agreement, all Assets that are (i) owned of record or held in the name of a member of the Certegy Group on the Distribution Date, (ii) treated for internal financial reporting purposes of Equifax prior to the Distribution Date or on the PSG Balance Sheet as owned by a member of the Certegy Group, (iii) on the Distribution Date used exclusively by one or more members of the Certegy Group, or (iv) transferred to a member of the Certegy Group pursuant to any Ancillary Agreement.

"Certegy Business" means the businesses now or formerly conducted by PSG, including the businesses of providing payment transaction processing and check risk management services to financial institutions and merchants.

"Certegy Bylaws" means the bylaws of Certegy in the form filed as an exhibit to the Form 10 at the time it becomes effective.

"Certegy Common Stock" means the outstanding shares of common stock, \$.01 par value, of Certegy.

"Certegy Group" means Certegy, Payment Services, FBS, Telecredit Canada, Card Solutions France, Equifax Asia Pacific, Payment U.K., any of their respective subsidiaries and any subsidiary or division of any member of the Equifax Group that is included in the assets of the Certegy Business as reflected in the pro forma combined balance sheet of Certegy as of March 31, 2001 contained in the Information Statement.

"Certegy Liabilities" means (a) Liabilities of any member of the Certegy Group under this Agreement or any Ancillary Agreement, (b) except as otherwise expressly provided in this Agreement or any Ancillary Agreement, Liabilities incurred in connection with the conduct or operation of the Certegy Business (including any acquired businesses) or the ownership or use of the Certegy Assets, whether arising before, at or after the Effective Time, (c) Liabilities arising under or in connection with the Form 10, except to the extent such Liabilities arise out of or are based upon information about Equifax included in the sections of the Information Statement attached as Exhibit 99.1 to the Form 10 entitled "Summary - Our Business," "Summary - The Distribution," and "The Distribution - Reasons for the Distribution, "(d) except as otherwise expressly provided in this Agreement or any Ancillary Agreement, Liabilities set forth on the PSG Balance Sheet, (e) except as otherwise provided in this Agreement or any Ancillary Agreement, Liabilities of the Equifax Group or the Certegy Group relating to a Sold PSG Business or arising out of the sale thereof, and (f) any Liabilities relating to or arising out of the acquisition (whether through an acquisition of stock or assets or a merger, share exchange or other form of business combination) of any business prior to the Effective Time by any member of the Certegy Group, except to the extent such Liabilities arise out of or are based upon the issuance of securities of Equifax in any such business combination transaction.

"Certegy Revolving Credit Agreement" means the Revolving Credit Agreement among various lenders and Certegy, which provides for borrowing of the Borrowing Proceeds for the purpose of making the payment provided for in Section 2.01.

"Chile Agreements" means that agreements, assignments and documents relating to the transfer by Equifax de Chile, S.A. of its ownership interests in Procard S.A. to Payment Chile S.A.

"Code" means the United States Internal Revenue Code of 1986, as amended.

"Commission" means Securities and Exchange Commission.

"Contribution" is defined in Section 2.01.

"Distribution" is defined in the recitals to this Agreement.

"Distribution Agent" means SunTrust Bank, Atlanta, in its capacity as agent for Equifax in connection with the Distribution.

"Distribution Date" means the date upon which the Distribution shall be effective, as determined by the Board of Directors of Equifax, or such committee of such Board of Directors as shall be designated by the Board of Directors of Equifax.

"Effective Time" means 11:59 p.m. Atlanta time on June 30, 2001.

"EIS" means Equifax Information Services, LLC, a wholly-owned subsidiary of Equifax organized under the laws of the State of Georgia, formerly known as Equifax Credit Information Services, Inc.

"Employee Benefits Agreement" means the Employee Benefits Agreement entered into at or prior to the Effective Time between Equifax and Certegy, as amended from time to time.

"Equifax Asia Pacific" means Equifax Asia Pacific Holdings Inc., a wholly-owned subsidiary of Equifax organized under the laws of the State of Georgia.

"Equifax Business" means the business now or formerly conducted by Equifax and its present and former subsidiaries, joint ventures and partnerships, other than the Certegy Business.

"Equifax Common Stock" means the outstanding shares of common stock, \$1.25 par value, of Equifax.

"Equifax Group" means Equifax and its subsidiaries, joint ventures and partnerships, excluding any member of the Certegy Group.

"Equifax Liabilities" means (i) Liabilities of any member of the Equifax Group under this Agreement or any Ancillary Agreement, and (ii) Liabilities, other than Certegy Liabilities, incurred in connection with the operation of the Equifax Business, whether arising before, at, or after the Effective Time.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"FBS" means First Bankcard Systems, Inc., a wholly-owned subsidiary of Equifax organized under the laws of the State of Georgia.

"Form 10" means the registration statement on Form 10 filed by Certegy with the Commission to effect the registration of the Certegy Common Stock pursuant to the Exchange Act, as such registration statement may be amended from time to time

"Group" means the Equifax Group or the Certegy Group, as the context so requires.

"Guaranteed Certegy Liabilities" means the Certegy Liabilities on which any member of the Equifax Group is an obligor by reason of any guarantee or contractual commitment, including Liabilities under any contract assumed by any member of the Certegy Group from any member of the Equifax Group with respect to which any member of the Equifax Group remains liable.

"Guaranteed Equifax Liabilities" means the Equifax Liabilities on which any member of the Certegy Group is an obligor by reason of any guarantee or contractual commitment, including Liabilities under any contract assumed by any member of the Equifax Group from any member of the Certegy Group with respect to which any member of the Certegy Group remains liable.

"Indemnifiable Loss" means any and all damage, loss, liability, and expense (including, without limitation, reasonable expenses of investigation and reasonable attorneys' fees and expenses) in connection with any and all Actions or threatened Actions.

"Information Statement" means the information statement required by the Commission to be sent to each holder of Equifax Common Stock in connection with the Distribution, and prepared in accordance with the Exchange Act.

"Intellectual Property Agreement" means the Intellectual Property Agreement entered into at or prior to the Effective Time between Equifax and Certegy, as amended from time to time.

"Intercompany Data Purchase Agreement" means the Intercompany Data Purchase Agreement entered into at or prior to the Effective Time between Equifax and Certegy, as amended from time to time.

"IRS" means the United States Internal Revenue Service.

"Liabilities" means any and all claims, debts, liabilities and obligations, absolute or contingent, matured or not matured, liquidated or unliquidated, accrued or unaccrued, known or unknown, whenever arising, including all costs and expenses relating thereto, and including, without limitation, those debts, liabilities and obligations arising under this Agreement or any Ancillary Agreement, any law, rule, regulation, action, order or consent decree of any governmental entity or any award of any arbitrator of any kind, and those arising under any contract, commitment or undertaking.

"Payment Services" means Equifax Payment Services Inc., a wholly-owned subsidiary of Equifax organized under the laws of the State of Delaware.

"Payment U.K." means Payment U.K. Ltd., a wholly-owned subsidiary of Equifax organized under the laws of England and Wales.

"Prime Rate" means the rate of interest announced by SunTrust Bank from time to time as its "prime rate," "prime lending rate," "base rate" or similar reference rate. In the event the Prime Rate is discontinued as a standard, the parties, by mutual agreement, shall designate a comparable reference rate as a substitute therefor. For purposes hereof, the Prime Rate in effect at the close of business on each business day of SunTrust Bank, Atlanta, Georgia shall be the Prime Rate for that day and any immediately succeeding non-business day or days.

"PSG Balance Sheet" means the consolidated balance sheet of the Certegy Group as of the Effective Time, which balance sheet shall be prepared by Equifax on a basis consistent with Equifax's historical practices for the preparation of monthly divisional balance sheets.

"Rabbi Trusts" means the trusts established by Equifax pursuant to that certain Grantor Trust Agreement dated as of February 4, 2000, as amended, and the trust established by Equifax pursuant to that certain Trust Agreement dated December 29, 1989, as amended.

"Real Estate Agreements" means all deeds, subleases, releases, assignments, consents and agreements (including without limitation the Agreement Regarding Leases) relating to the conveyance to Certegy of the Real Estate Assets and the division of real property and interests therein between members of the Equifax Group and members of the Certegy Group entered into as of or prior to the Distribution Date, in each case as amended from time to time.

"Real Estate Assets" means the real estate owned by the Equifax Group and used by PSG that is to be conveyed to PSG pursuant to the Real Estate Agreements.

"Record Date" means the date designated by or under the authority of Equifax's Board of Directors as the record date for determining the shareholders of Equifax entitled to receive the Distribution.

"Securities Act" means the Securities Act of 1933, as amended.

"Services Agreements" means the Transition Support Agreement, Agreement Regarding IBM UK Mainframe Services, Agreement Regarding IBM Australian Mainframe and Network Services, US and UK Network Services "Authorized User" Agreement, IBM United Kingdom Limited Transition Services Agreement, and Agreement Regarding IBM Canadian Data Center Services Agreement.

"Sold PSG Business" means any of the Assets or businesses related to the Certegy Business formerly owned, directly or indirectly, by Equifax and heretofore sold.

"Stock Benefit Trust" means the Trust Agreement between Equifax and Wachovia Bank of North Carolina, N.A. dated July 7, 1993.

"Tax" shall have the meaning given to such term in the Tax Sharing and Indemnification $\mbox{\sc Agreement}.$

"Tax Sharing and Indemnification Agreement" means the Tax Sharing and Indemnification Agreement entered into at or before the Effective Time between Equifax and Certegy, as amended from time to time.

"Telecredit Canada" means Telecredit Canada, Inc., a wholly-owned subsidiary of Equifax organized under the laws of Canada.

"Transition Support Agreement" means the Transition Support Agreement entered into at or prior to the Effective Time between Equifax and Certegy, as amended from time to time.

ARTICLE II
REORGANIZATION; CONVEYANCE OF CERTAIN ASSETS;
ASSUMPTION OF CERTAIN LIABILITIES;
CERTAIN PAYMENTS; AND TRANSITION ARRANGEMENTS

Section 2.01. Reorganization. On or prior to the Distribution Date and

effective as of the Effective Time, Equifax shall contribute to Certegy all of the Certegy Assets in exchange for (i) the Borrowing Proceeds and (ii) a number of shares of Certegy Common Stock that when combined with the shares of Certegy Common Stock already owned by Equifax shall equal all the shares to be distributed as provided in Section 3.03 below (the "Contribution").

Section 2.02. Conveyance of Assets; Discharge of Liabilities. Except as otherwise expressly provided herein or in any of the Ancillary Agreements:

- (a) Effective as of the Effective Time (i) all Certegy Assets are intended to be and shall become Assets of the Certegy Group, (ii) all Certegy Liabilities are intended to be and shall become the Liabilities of the Certegy Group, and (iii) all other Assets and Liabilities of Equifax and its subsidiaries are intended to be and shall remain exclusively the Assets and Liabilities of the Equifax Group.
- (b) Effective as of the Effective Time, Equifax agrees to transfer or cause to be transferred to Certegy or to such other members of the Certegy Group as Certegy may designate all right, title and interest of the Equifax Group in and to all of the Certegy Assets.
- (c) Certegy agrees that, effective as of the Effective Time, it will transfer or cause to be transferred to Equifax or to such other member of the Equifax Group as Equifax may designate all right, title and interest of the Certegy Group in and to all Assets that are not Certegy Assets.
- (d) Certegy agrees that it will, or will cause another member of the Certegy Group designated by Certegy to, (i) assume any of the Certegy Liabilities for which a member of the Certegy Group is not the obligor, effective as of the Effective Time, and (ii) timely pay and discharge all of the Certegy Liabilities, at and after the Effective Time.
- (e) Equifax agrees that it will, or will cause another member of the Certegy Group designated by Equifax to, (i) assume any of the Equifax Liabilities for which a member of the Equifax Group is not the obligor, effective as of the Effective Time, and (ii) timely pay and discharge all of the Equifax Liabilities, at and after the Effective Time.

- (f) In the event that any conveyance of an Asset required hereby is not effected at or before the Effective Time, the obligation to transfer such Asset shall continue past the Effective Time and shall be accomplished as soon thereafter as practicable.
- (g) If any Asset may not be transferred by reason of the requirement to obtain the consent of any third party and such consent has not been obtained by the Effective Time, then (unless otherwise expressly agreed by Equifax and Certegy) such Asset shall not be transferred until such consent has been obtained, and Equifax and Certegy, as the case may be, shall cause the owner of such Asset to use all reasonable efforts to provide to the appropriate member of the other Group all the rights and benefits under such Asset and cause such owner to enforce such Asset for the benefit of such member, in each case to the extent that such action does not cause a breach or default under such Asset. Both parties shall otherwise cooperate and use all reasonable efforts to provide the economic and operational equivalent of an assignment or transfer of the Asset as of the Effective Time.
- (h) From and after the Effective Time, each party shall promptly transfer or cause the members of its Group promptly to transfer to the other party or the appropriate member of the other party's Group, from time to time, any property received that is an Asset of the other party or a member of its Group. Without limiting the foregoing, funds received by a member of one Group upon the payment of accounts receivable that belong to a member of the other Group shall be transferred to the other Group by wire transfer not more than five business days after receipt of such payment.
- (i) Except as expressly set forth in this Agreement, any Ancillary Agreement, or any instrument or document contemplated by this Agreement or any Ancillary Agreement, neither any member of the Equifax Group nor any member of the Certegy Group has made or shall be deemed to have made any representation or warranty as to (i) the Assets, business or Liabilities retained, transferred or assumed as contemplated hereby or thereby, (ii) any consents or approvals required in connection with the transfer or assumption by such party of any Asset or Liability contemplated by this Agreement, (iii) the value or freedom from any lien, claim, equity or other encumbrance of, or any other matter concerning, any Assets of such party, (iv) the absence of any defenses or right of setoff or freedom from counterclaim with respect to any claim or other Asset of such party, or (v) the legal sufficiency of any assignment, document or instrument delivered to convey title to any Asset transferred. EXCEPT AS MAY BE EXPRESSLY SET FORTH IN THIS AGREEMENT OR ANY ANCILLARY AGREEMENT, ALL ASSETS WERE, OR ARE BEING, TRANSFERRED, OR ARE BEING RETAINED, ON AN "AS IS," "WHERE IS" BASIS AND THE RESPECTIVE TRANSFEREES WILL BEAR THE ECONOMIC AND LEGAL RISKS THAT ANY CONVEYANCE OR OTHER TRANSFER SHALL PROVE TO BE INSUFFICIENT TO VEST IN THE TRANSFEREE A TITLE THAT IS FREE AND CLEAR OF ANY LIEN, CLAIM, EQUITY OR OTHER ENCUMBRANCE.

Section 2.03. Ancillary Agreements. As of the Effective Time, Equifax and Certegy (or their appropriate subsidiaries) will execute and deliver:

(a) A duly executed Employee Benefits Agreement;

- (b) A duly executed Tax Sharing and Indemnification Agreement;
- (c) A duly executed Intercompany Data Purchase Agreement;
- (d) A duly executed copy of each of the Services Agreements;
- (e) A duly executed Intellectual Property Agreement;
- (f) A duly executed copy of each of the Real Estate Agreements; and
- (g) Such other agreements, leases, subleases, documents, or instruments as the parties may agree are necessary or desirable in order to achieve the purposes hereof.

Section 2.04. Issuance of Certegy Common Stock. On or before the

Distribution Date, and in exchange for the transfer by Equifax to Certegy of the stock and assets as provided above, and the surrender for reissue of all certificates representing outstanding Certegy Common Stock, Certegy will issue and deliver to Equifax a certificate representing shares of Certegy Common Stock constituting all the shares to be distributed as provided in Section 3.03 below.

Section 2.05. Resignations.

- (a) On or before the Distribution Date, Certegy will deliver or cause to be delivered to Equifax resignations of each person who will be an employee of Certegy or another member of the Certegy Group from and after the Distribution Date and who is an officer or director of Equifax or any of its subsidiaries or affiliates not constituting a member of the Certegy Group immediately prior to the Distribution Date, except for persons identified in the Information Statement as continuing directors of Equifax or other members of the Equifax Group.
- (b) On or before the Distribution Date, Equifax will deliver or cause to be delivered to Certegy resignations of each person who will be an employee of Equifax or another member of the Equifax Group from and after the Distribution Date and who is an officer or director of Certegy or any of its subsidiaries or affiliates not constituting a member of the Equifax Group immediately prior to the Distribution Date, except for persons identified in the Information Statement as continuing directors of Certegy or other members of the Certegy Group.

Section 2.06. Conduct of Certegy Pending Distribution.

- (a) Prior to the Distribution Date, Certegy shall not, without the prior consent of Equifax, make any press release concerning the Distribution and shall use its best efforts not to take any action which may prejudice or delay the consummation of the Distribution. Prior to the Distribution Date, Certegy further agrees regularly to apprise Equifax of public announcements to or the dissemination of materials for financial analysts or other persons relating to its business and the Distribution.
- (b) Prior to the Distribution Date, the business of Certegy shall be operated for the sole benefit of Equifax as Certegy's sole shareholder; provided however, that upon

consummation of the Distribution, the business of Certegy shall be deemed to have been operated for the sole benefit of Certegy and its new shareholders, as of and after the Effective Time. On the Distribution Date, (i) any amounts advanced, incurred or contributed by any member of the Equifax Group to or for the benefit of any member of the Certegy Group after the Effective Time (other than amounts advanced or contributed to Certegy for costs or expenses to be paid by Equifax in accordance with Section 15.01) shall be repaid by Certegy, in addition to any payments prescribed by Section 8.03 hereof, and (ii) any amounts advanced, incurred or contributed by any member of the Certegy Group to or for the benefit of any member of the Equifax Group after the Effective Time shall be repaid by Equifax, in addition to any payments prescribed by Section 8.03 hereof.

Section 2.07. Certegy Revolving Credit Agreement. Certegy shall use all

reasonable efforts promptly to obtain, and to satisfy all conditions for borrowing under the Certegy Revolving Credit Agreement in an amount sufficient to allow Certegy to generate the Borrowing Proceeds and conduct the business of Certegy after the Distribution Date.

Section 2.08. Guaranteed Certegy and Equifax Liabilities.

- (a) Certegy shall use all reasonable efforts (excluding payment of money or incurrence of Liabilities) to obtain as promptly as practicable after the Distribution Date the release of all members of the Equifax Group from any obligations with respect to Guaranteed Certegy Liabilities. In no event shall any member of the Certegy Group take any action with respect to any Guaranteed Certegy Liabilities which could be reasonably expected to adversely affect the Equifax Group members in any way including, without limitation, extending the term of any Guaranteed Certegy Liabilities or increasing the liability guaranteed thereunder, unless the guarantee or obligation of all Equifax Group members is released as to any extended or modified liability obligations under such Guaranteed Certegy Liabilities or Equifax otherwise consents in writing.
- (b) Equifax shall use all reasonable efforts (excluding payment of money or incurrence of Liabilities) to obtain as promptly as practicable after the Distribution Date the release of all members of the Certegy Group from any obligations with respect to Guaranteed Equifax Liabilities. In no event shall any member of the Equifax Group take any action with respect to any Guaranteed Equifax Liabilities which could be reasonably expected to adversely affect the Certegy Group members in any way including, without limitation, extending the term of any Guaranteed Equifax Liabilities or increasing the liability guaranteed thereunder, unless the guarantee or obligation of all Certegy Group members is released as to any extended or modified liability obligations under such Guaranteed Equifax Liabilities or Certegy otherwise consents in writing.
- (c) In the event that any Equifax Group member is required to pay or otherwise satisfy any Guaranteed Certegy Liabilities, without limiting any of Equifax's rights and remedies against Certegy under this Agreement or otherwise, in order to secure Certegy's indemnity obligations to Equifax hereunder in respect of such Guaranteed Certegy Liabilities, Equifax shall be entitled to all the rights of the payee in any property of any member of the Certegy Group pledged as security for such Guaranteed Certegy Liabilities.

(d) In the event that Certegy Group member is required to pay or otherwise satisfy any Guaranteed Equifax Liabilities, without limiting any of Certegy's rights and remedies against Equifax under this Agreement or otherwise, in order to secure Equifax's indemnity obligations to Certegy hereunder in respect of such Guaranteed Equifax Liabilities, Certegy shall be entitled to all the rights of the payee in any property of any member of the Equifax Group pledged as security for such Guaranteed Equifax Liabilities.

Section 2.09. Insurance.

- (a) Following the Distribution, Certegy will use its best efforts to procure and maintain directors' and officers' liability insurance coverage at least equal to the amount of Equifax's current directors' and officers' insurance coverage for a period of five (5) years from the Distribution Date with respect to directors and officers of Equifax who are or will become directors and officers of Certegy as of the Distribution Date for acts as directors and officers of members of the Certegy Group for periods from and after the Distribution Date.
- (b) Following the Distribution, Equifax will use its best efforts to maintain directors' and officers' liability insurance coverage at least equal to the amount of Equifax's current directors' and officers' liability insurance coverage for a period of five (5) years from the Distribution Date with respect to the directors and officers of Equifax who are or will become directors and officers of members of the Certegy Group as of the Distribution Date for acts as directors and officers of members of the Equifax Group during periods prior to the Distribution Date.

ARTICLE III THE DISTRIBUTION

Section 3.01. Cooperation Prior to the Distribution.

- (a) Equifax and Certegy shall prepare, and Equifax shall mail to the holders of Equifax Common Stock, the Information Statement, which shall set forth appropriate disclosure concerning Certegy, the Distribution and any other appropriate matters. Equifax and Certegy shall also prepare, and Certegy shall file with the Commission, the Form 10, which shall include the Information Statement. Equifax and Certegy shall use all reasonable efforts to cause the Form 10 to become effective under the Exchange Act.
- (b) Equifax shall, as the sole shareholder of Certegy, approve and adopt the Certegy employee benefit plans contemplated by the Employee Benefits Agreement and Equifax and Certegy shall cooperate in preparing, filing with the Commission under the Securities Act and causing to become effective not later than the Distribution Date any registration statements or amendments thereto that are appropriate to reflect the establishment of or amendments to any employee benefit plan of Certegy contemplated by the Employee Benefits Agreement, including without limitation, a Form S-8 with respect thereto.
- (c) Equifax and Certegy shall take all such action as may be necessary or appropriate under the securities or blue sky laws of states or other political subdivisions of the

United States in connection with the transactions contemplated by this Agreement or any Ancillary Agreement.

- (d) Certegy shall prepare, file, and use its best efforts to cause to be approved prior to the Record Date, the application to permit listing of the Certegy Common Stock on the New York Stock Exchange.
- (e) Equifax and Certegy shall take all such actions as may be deemed reasonably necessary to secure a favorable ruling from the IRS that the Distribution is not taxable to Equifax or its shareholders pursuant to Section 355 of the Code.
- Section 3.02. Conditions Precedent to the Distribution. In no event shall the Distribution occur unless the following conditions shall have been satisfied or waived by Equifax:
- (a) Equifax's Board of Directors or a duly appointed committee thereof, shall, in its sole discretion, have established the Record Date and the Distribution Date and any appropriate procedures in connection with the Distribution;
 - (b) all necessary regulatory approvals shall have been received;
- (c) the Information Statement shall have been mailed to the holders of Equifax Common Stock;
- (d) the Form 10 shall have become effective under the Exchange Act, and all registration statements referred to under Section 3.01(b) shall have become effective under the Securities Act;
- (e) the Certegy Board of Directors, as named in the Form 10, shall have been elected by Equifax, as sole shareholder of Certegy, and the Certegy Articles and Certegy Bylaws shall have been adopted and be in effect;
- (f) the Certegy Common Stock shall have been approved for listing on the New York Stock Exchange, subject to official notice of issuance;
- (g) Equifax and Certegy shall have taken all such action as may be necessary or appropriate under the securities or blue sky laws of states or other political subdivisions of the United States in connection with the transactions contemplated by this Agreement or any Ancillary Agreement;
- (h) Equifax shall have received a favorable private letter ruling from the IRS that the Contribution constitutes a reorganization pursuant to Section 368(a)(1)(D) of the Code and that the Distribution will not be taxable to Equifax or its shareholders pursuant to Section 355 of the Code, and such ruling shall continue in effect;
 - (i) Certegy shall have entered into the Certegy Revolving Credit Agreement;
 - (j) the transactions described in Section 2.01 shall have occurred;

- (k) Certegy (or its appropriate subsidiary) shall have performed fully its (or their) obligations under Section 2.02;
- (1) no order, injunction, or decree issued by any court of competent jurisdiction or other legal restraint or prohibition preventing consummation of the Distribution shall be in effect;
- (m) Equifax and Certegy shall each have performed its obligations under this Agreement and each Ancillary Agreement, which are required to be performed prior to or at the time of the Distribution; and
- (n) the parties shall have consummated those other transactions in connection with the Distribution that are contemplated by the Information Statement to be consummated prior to or at the time of the Distribution and are not specifically referred to in this Agreement or the Ancillary Agreements identified in Sections 2.03(a) (f).

Section 3.03. The Distribution. On or before the Distribution Date,

subject to satisfaction or waiver of the conditions set forth in this Agreement, Equifax shall deliver to the Distribution Agent a certificate or certificates representing all of the then outstanding shares of Certegy Common Stock held by the Equifax Group, endorsed in blank, and shall instruct the Distribution Agent, except as otherwise provided in Sections 3.04 and 3.05, to distribute to each holder of record of Equifax Common Stock on the Record Date one share of Certegy Common Stock for each two shares of Equifax Common Stock so held either by crediting the holder's brokerage account or by delivering a certificate or certificates representing such shares. Certegy agrees to provide all certificates for shares of Certegy Common Stock that the Distribution Agent shall require in order to effect the Distribution.

Section 3.04. Fractional Shares. The Distribution Agent shall not

distribute any fractional shares of Certegy Common Stock. The Distribution Agent shall aggregate all such fractional shares and sell them in an orderly manner after the Distribution Date in the open market and, after completion of such sales, distribute a pro rata portion of the proceeds from such sales, based upon the average gross selling price of all such Certegy Common Stock, to each holder of Equifax Common Stock who would otherwise have received a fractional share of Certegy Common Stock.

Section 3.05. Stock Trusts. The Distribution Agent shall not distribute

in the Distribution any shares of Certegy Common Stock to the Stock Benefit Trust, the Rabbi Trusts or any person holding Equifax Common Stock pursuant to any Equifax restricted stock plan, if and to the extent Equifax so instructs the Distribution Agent after receipt of a waiver of the right to receive any Certegy Common Stock from any such Equifax shareholder.

ARTICLE IV INDEMNIFICATION

Section 4.01. Certegy Indemnification of the Equifax Group. On and after the Distribution Date, Certegy shall indemnify, defend and hold harmless each member of the

Equifax Group, and each of their respective directors, officers, employees and agents (the "Equifax Indemnitees") from and against any and all Indemnifiable Losses incurred or suffered by any of the Equifax Indemnitees and arising out of, or due to, (a) the failure of Certegy or any member of the Certegy Group to pay, perform or otherwise discharge, any of the Certegy Liabilities and (b) any untrue statement or alleged untrue statement of any material fact contained in the preliminary or final Form 10, the preliminary or final Information Statement or any amendment or supplement thereto or the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading (other than the information about Equifax included in the sections of the Information Statement attached as Exhibit 99.1 to the Form 10 entitled "Summary - Our Business," "Summary - The Distribution," and "The Distribution - Reasons for the Distribution," or any amendment or supplement thereto).

Section 4.02. Equifax Indemnification of Certegy Group. On and after the $\,$

Distribution Date, Equifax shall indemnify, defend and hold harmless each member of the Certegy Group and each of their respective directors, officers, employees and agents (the "Certegy Indemnitees") from and against any and all Indemnifiable Losses incurred or suffered by any of the Certegy Indemnitees and arising out of, or due to, (a) the failure of Equifax or any member of the Equifax Group to pay, perform or otherwise discharge, any of the Equifax Liabilities and (b) any untrue statement or alleged untrue statement of any material fact regarding Equifax included in the sections of the Information Statement attached as Exhibit 99.1 to the Form 10 entitled "Summary - Our Business," "Summary - The Distribution," and "The Distribution - Reasons for the Distribution," or any amendment or supplement thereto or the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading.

Section 4.03. Contribution. In circumstances in which the indemnity ${\bf r}$

agreements provided for in Sections 4.01(b) and 4.02(b) are unavailable or insufficient, for any reason, to hold harmless an indemnified party in respect of any Indemnifiable Losses, each indemnifying party, in order to provide for just and equitable contribution, shall contribute to the amount paid or payable by such indemnified party as a result of such Indemnifiable Losses, in such proportion as is appropriate to reflect the relative fault of the indemnifying party or parties on the one hand and the indemnified party on the other in connection with the statements or omissions or alleged statements or omissions that resulted in such Indemnifiable Losses, as well as any other relevant equitable considerations. The relative fault of the parties shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by Certegy or Equifax, the parties' relative intents, knowledge, access to information and opportunity to correct or prevent such statement or omission, and any other equitable considerations appropriate in the circumstances.

Section 4.04. Insurance and Third Party Obligations. No insurer or any

other third party shall be, by virtue of the foregoing indemnification provisions, (a) entitled to a benefit it would not be entitled to receive in the absence of such provisions, (b) relieved of the responsibility to pay any claims to which it is obligated, or (c) entitled to any subrogation rights with respect to any obligation hereunder.

ARTICLE V INDEMNIFICATION PROCEDURES

Section 5.01. Notice and Payment of Claims. If any Equifax or Certegy

Indemnitee (the "Indemnified Party") determines that it is or may be entitled to indemnification by a party (the "Indemnifying Party") under Article IV (other than in connection with any Action or claim subject to Section 5.02), the Indemnified Party shall deliver to the Indemnifying Party a written notice specifying, to the extent reasonably practicable, the basis for its claim for indemnification and the amount for which the Indemnified Party reasonably believes it is entitled to be indemnified. After the Indemnifying Party shall have been notified of the amount for which the Indemnified Party seeks indemnification, the Indemnifying Party shall, within 30 days after receipt of such notice, pay the Indemnified Party such amount in cash or other immediately available funds (or reach agreement with the Indemnified Party as to a mutually agreeable alternative payment schedule) unless the Indemnifying Party objects to the claim for indemnification or the amount thereof. If the Indemnifying Party does not give the Indemnified Party written notice objecting to such claim and setting forth the grounds therefor within the same 30 day period, the Indemnifying Party shall be deemed to have acknowledged its liability for such claim and the Indemnified Party may exercise any and all of its rights under applicable law to collect such amount.

Section 5.02. Notice and Defense of Third Party Claims. Promptly $\left(\frac{1}{2} \right)$

following the earlier of (a) receipt of notice of the commencement by a third party of any Action against or otherwise involving any Indemnified Party or (b) receipt of information from a third party alleging the existence of a claim against an Indemnified Party, in either case, with respect to which indemnification may be sought pursuant to this Agreement (a "Third Party Claim"), the Indemnified Party shall give the Indemnifying Party written notice thereof. The failure of the Indemnified Party to give notice as provided in this Section 5.02 shall not relieve the Indemnifying Party of its obligations under this Agreement, except to the extent that the Indemnifying Party is prejudiced by such failure to give notice. Within 30 days after receipt of such notice, the Indemnifying Party shall by giving written notice thereof to the Indemnified Party, (a) acknowledge, as between the parties hereto, liability for, and at its option elect to assume the defense of such Third Party Claim at its sole cost and expense or (b) object to the claim of indemnification set forth in the notice delivered by the Indemnified Party pursuant to the first sentence of this Section 5.02 setting forth the grounds therefor; provided that if the Indemnifying Party does not within the same 30 day period give the Indemnified Party written notice acknowledging liability or objecting to such claim and setting forth the grounds therefor, the Indemnifying Party shall be deemed to have acknowledged, as between the parties hereto, its liability to the Indemnified Party for such Third Party Claim. Any contest of a Third Party Claim as to which the Indemnifying Party has elected to assume the defense shall be conducted by attorneys employed by the Indemnifying Party and reasonably satisfactory to the Indemnified Party; provided that the Indemnified Party shall have the right to participate in such proceedings and to be represented by attorneys of its own choosing at the Indemnified Party's sole cost and expense. If the Indemnifying Party assumes the defense of a Third Party Claim, the Indemnifying Party may settle or compromise the claim without the prior written consent of the Indemnified Party; provided that the Indemnifying Party may not agree to any such settlement pursuant to which any remedy or relief, other than monetary damages for which the Indemnifying Party shall be responsible hereunder, shall be applied to or against the Indemnified

Party, without the prior written consent of the Indemnified Party, which consent shall not be unreasonably withheld. If the Indemnifying Party does not assume the defense of a Third Party Claim for which it has acknowledged liability for indemnification under Article IV, the Indemnified Party may require the Indemnifying Party to reimburse it on a current basis for its reasonable expenses of investigation, reasonable attorneys' fees and reasonable out-ofpocket expenses incurred in defending against such Third Party Claim, and the Indemnifying Party shall be bound by the result obtained with respect thereto by the Indemnified Party; provided that the Indemnifying Party shall not be liable for any settlement effected without its consent, which consent shall not be unreasonably withheld. The Indemnifying Party shall pay to the Indemnified Party in cash the amount for which the Indemnified Party is entitled to be indemnified (if any) within 15 days after the final resolution of such Third Party Claim (whether by the final nonappealable judgment of a court of competent jurisdiction or otherwise), or, in the case of any Third Party Claim as to which the Indemnifying Party has not acknowledged liability, within 15 days after such Indemnifying Party's objection has been resolved by settlement, compromise or the final nonappealable judgment of a court of competent jurisdiction.

ARTICLE VI EMPLOYEE MATTERS

Section 6.01. Employee Benefits Agreement. All matters relating to or

arising out of any employee benefit, compensation or welfare arrangement in respect of any present and former employee of the Equifax Group or the Certegy Group shall be governed by the Employee Benefits Agreement, except as may be expressly stated herein or therein, and except that Article VIII shall apply with respect to such matters to the extent otherwise applicable. In the event of any inconsistency with respect to such matters between the Employee Benefits Agreement and this Agreement or any Ancillary Agreement, the Employee Benefits Agreement shall govern to the extent of the inconsistency.

ARTICLE VII

Section 7.01. Tax Sharing and Indemnification Agreement. All matters

relating to Taxes shall be governed exclusively by the Tax Sharing and Indemnification Agreement, except as may be expressly stated herein or therein, and except that Article VIII shall apply with respect to such matters to the extent otherwise applicable. In the event of any inconsistency with respect to such matters between the Tax Sharing and Indemnification Agreement and this Agreement or any other Ancillary Agreement, the Tax Sharing and Indemnification Agreement shall govern to the extent of the inconsistency.

ARTICLE VIII ACCOUNTING MATTERS

Section 8.01. Allocation of Prepaid Items and Reserves. All prepaid items

and reserves that have been maintained by Equifax on a consolidated basis but that relate in part to assets or liabilities of the Certegy Group shall be allocated between Equifax and Certegy as determined by Equifax in its reasonable discretion.

Equifax shall prepare and deliver to Certegy a preliminary PSG Balance Sheet which shall set forth good faith estimates of all intercompany account balances between members of the Equifax Group and members of the Certegy Group as of the Effective Time, after giving effect to any intercompany dividends made or declared on or before the Distribution Date and any other transfers of assets or liabilities between the Groups before the Distribution Date pursuant to the express terms of this Agreement or any Ancillary Agreement. On or before the Distribution Date, all estimated intercompany account balances owing as set forth on the preliminary PSG Balance Sheet shall be settled and paid in full by Certegy or Equifax, as the case may be. Within 30 business days after the Effective Time, Equifax shall prepare and deliver to Certegy a final PSG Balance Sheet which shall set forth all intercompany account balances between members of the Equifax Group and members of the Certegy Group as of the Effective Time, after giving effect to any intercompany dividends made or declared on or before the Distribution Date and any other transfers of assets or liabilities between the Groups before the Distribution Date pursuant to the express terms of this Agreement or any Ancillary Agreement. Within ten business day after the delivery of the final PSG Balance Sheet, Equifax shall pay to Certegy or Certegy shall pay to Equifax, as the case may be, the difference between the estimated account balances set forth on the preliminary PSG Balance Sheet and the final account balances set forth on the final PSG Balance Sheet. Notwithstanding anything contained herein to the contrary, Equifax may at its option contribute to Certegy amounts owed by any member of the Certegy Group to any member of the Equifax Group as a contribution to the capital of Certegy in lieu of payment of such amounts to the Equifax Group by the Certegy Group. Any disputes arising from the adjustments required by the final PSG Balance Sheet shall be resolved in accordance with Section 15.10 hereof.

ARTICLE IX DATA, PRODUCTS AND SERVICES

Section 9.01. Intercompany Data Purchase Agreement. All matters relating

to the ownership of and the right to use data owned or maintained by any member of the Equifax Group or the Certegy Group shall be governed exclusively by the Intercompany Data Purchase Agreement, except as may be expressly stated herein or therein, and except that Article VIII shall apply with respect to such matters to the extent otherwise applicable. In the event of any inconsistency with respect to such matters between the Intercompany Data Purchase Agreement and this Agreement or any Ancillary Agreement, the Intercompany Data Purchase Agreement shall govern to the extent of the inconsistency.

ARTICLE X INTELLECTUAL PROPERTY

Section 10.01. Intellectual Property Agreement. All matters relating to

the ownership and right to use intellectual property, other than data, shall be governed exclusively by the Intellectual Property Agreement, except as may be expressly stated herein or therein, and except that Article VIII shall apply with respect to such matters to the extent otherwise applicable. In the event of any inconsistency with respect to such matters between the Intellectual Property Agreement and this Agreement or any Ancillary Agreement, the Intellectual Property Agreement shall govern to the extent of the inconsistency.

ARTICLE XI TRANSITION SUPPORT

Section 11.01. Services Agreements. All matters relating to the provision ${\bf r}$

of support and other services by the Equifax Group to the Certegy Group and the Certegy Group to the Equifax Group after the Effective Time, covered by the Services Agreements, shall be governed exclusively by the respective Services Agreements, except as may be expressly stated herein or therein, and except that Article VIII shall apply with respect to such matters to the extent otherwise applicable. In the event of any inconsistency with respect to such matters between a particular Services Agreement and this Agreement or any other Ancillary Agreement, the particular Services Agreement shall govern to the extent of the inconsistency.

ARTICLE XII REAL PROPERTY MATTERS

Section 12.01. Real Estate Agreements. All matters relating to real

property to be owned, leased, subleased, occupied, or shared by the Equifax Group or the Certegy Group after the Effective Time shall be governed by the Real Estate Agreements, except as may be expressly stated herein or therein, and except that Article VIII shall apply with respect to such matters to the extent otherwise applicable. In the event of any inconsistency with respect to such matters between the Real Estate Agreements and this Agreement or any Ancillary Agreement, the Real Estate Agreements shall govern to the extent of the inconsistency.

ARTICLE XIII INFORMATION

Section 13.01. Provision of Corporate Records. As soon as practicable

following the Effective Time, Equifax and Certegy shall each arrange for the provision to the other of existing corporate documents (e.g. minute books, stock registers, stock certificates, documents of title, contracts, etc.) in its possession relating to the other or its business and affairs or to any other entity that is part of such other's respective Group or to the business and affairs of such other entity.

Section 13.02. Access to Information. From and after the Effective Time,

Equifax and Certegy shall each afford the other and its accountants, counsel and other designated representatives reasonable access (including using reasonable efforts to give access to persons or firms possessing information) and duplicating rights during normal business hours to all records, books, contracts, instruments, computer data and other data and information in its possession relating to the business and affairs of the other or a member of its Group (other than data and information subject to an attorney/client or other privilege), insofar as such access is reasonably required by the other including, without limitation, for audit, accounting and litigation purposes.

Section 13.03. Litigation Cooperation. Equifax and Certegy shall each use

reasonable efforts to make available to the other, upon written request, its officers, directors, employees and agents, and the officers, directors, employees and agents of its subsidiaries, as witnesses to the extent that such persons may reasonably be required in connection with any legal, administrative or other proceedings arising out of the business of the other, or of any entity that is part of the

others' respective Group, prior to the Effective Time in which the requesting party or one of its subsidiaries may from time to time be involved. The requesting party shall bear all out-of-pocket costs and expenses in connection therewith. In connection with any matter contemplated by this Section 13.03, at the request of either party, the parties will enter into a mutually acceptable joint defense agreement so as to maintain to the extent practicable any attorney-client privilege or work product immunity of any member of any Group.

Section 13.04. Retention of Records. Except as otherwise required by law

or agreed to in writing, each party shall, and shall cause the members of its Group to, retain all information relating to the other's business in accordance with the past practice of such party. Notwithstanding the foregoing, either party may destroy or otherwise dispose of any information at any time in accordance with the corporate record retention policy maintained by such party with respect to its own records.

Section 13.05. Confidentiality. Each party shall, and shall cause each

member of its Group to, hold and cause its directors, officers, employees, agents, consultants and advisors to hold, in strict confidence, unless compelled to disclose by judicial or administrative process or, in the opinion of its counsel, by other requirements of law, all information of or concerning the other party or its Group (except to the extent that this Agreement or any Ancillary Agreement permits or requires the use or disclosure of such information or to the extent such information can be shown to have been (a) in the public domain through no fault of the persons subject to the restrictions of this Section ("receiving party"), (b) later lawfully acquired after the Effective Time on a non-confidential basis from a third party or (c) independently generated without any reference to any proprietary or confidential information of the other party or its Group), and no receiving party shall (i) use such information, except for the benefit of the other party's Group in connection with the performance of this Agreement or the Ancillary Agreements, or (ii) disclose such information to any other person or entity, except its employees, directors, officers, agents, auditors, attorneys, financial advisors, bankers and other consultants and advisors who need to know such information and who shall be advised of the obligations contained in this Section 13.05 and be bound by them. Each receiving party shall be deemed to have satisfied its obligation to hold confidential any information concerning or owned by the other party or its Group if it exercises the same care as it takes to preserve confidentiality for its own similar information. The covenants in this Section 13.05 shall survive the transactions contemplated by this Agreement and shall continue indefinitely; provided, however, that the covenants in this Section 13.05 shall terminate with respect to any information not constituting a trade secret under applicable law on the third anniversary of the later of the Distribution Date or the date on which the party subject to such covenants with respect to such information receives it (but any such termination shall not terminate or otherwise limit any other covenant or restriction regarding the disclosure or use of such information under any Ancillary Agreement or other agreement, instrument or legal obligation).

Section 13.06. Ownership of Information. Any information owned by one

party or any of its subsidiaries that is provided to a requesting party pursuant to Article V, Article XV, or this Article XIII shall be deemed to remain the property of the providing party. Unless specifically set forth herein, nothing contained in this Agreement shall be construed as granting or conferring rights of license or otherwise in any such information.

ARTICLE XIV INTEREST ON PAYMENTS

Section 14.01. Interest. Except as otherwise expressly provided in this

Agreement or an Ancillary Agreement, all payments by one party to the other under this Agreement or any Ancillary Agreement shall be paid, by company check or wire transfer of immediately available funds to an account in the United States designated by the recipient, within 30 days after receipt of an invoice or other written request for payment setting forth the specific amount due and a description of the basis therefor in reasonable detail. Any amount remaining unpaid beyond its due date, including disputed amounts that are ultimately determined to be payable, shall bear interest at a rate of simple interest per annum equal to the Applicable Rate. Notwithstanding anything to the contrary contained herein or in any Ancillary Agreement, in no event shall the amount or rate of interest due and payable exceed the maximum amount or rate of interest allowed by applicable law (including, without limitation, O.C.G.A. (S) 7-4-18) and, in the event any such excess payment is made or received, such excess sum shall be credited as a payment of principal (or if no principal shall remain outstanding, shall be refunded).

ARTICLE XV MISCELLANEOUS

Section 15.01. Expenses. Except as specifically provided in this

Agreement or any Ancillary Agreement or in any other agreement between the parties, all costs and expenses incurred in connection with the preparation, execution, delivery and implementation of this Agreement and the Ancillary Agreements and with the consummation of the transactions contemplated by this Agreement (including transfer taxes and the fees and expenses of the Distribution Agent and of all counsel, accountants and financial and other advisors), together with any other amount expressly agreed by Equifax in writing, shall be paid by Equifax. Without limiting the foregoing, Equifax shall pay the legal, filing, accounting, printing, and other expenses in connection with the preparation, printing and filing of the Form 10 and the Information Statement.

Section 15.02. Notices. All notices and communications under this

Agreement shall be in writing and shall be deemed to have been given (a) when received, if such notice or communication is delivered by facsimile, hand delivery or overnight courier, and, (b) three (3) business days after mailing if such notice or communication is sent by United States registered or certified mail, return receipt requested, first class postage prepaid. All notices and communications, to be effective, must be properly addressed to the party to whom the same is directed at its address as follows:

If to Equifax, to: Equifax Inc.

1550 Peachtree Street, N.W. Atlanta, Georgia 30309

Attention: Phillip J. Mazzilli

Chief Financial Officer

Fax: (404) 885-8682

with a copy to: Equifax Inc.

1550 Peachtree Street Atlanta, Georgia 30309 Attention: Kent E. Mast General Counsel

Fax: (404) 885-8988

If to Certegy, to: Certegy Inc.

555 North Pointe Centre East Alpharetta, Georgia 30022 Attention: Bruce S. Richards

Corporate Vice President General Counsel and Secretary

Fax:

with a copy to: Certegy Inc. P.O. Box 349

Alpharetta, Georgia 30009 Attention: Michael T. Vollkommer

Corporate Vice President and

Chief Financial Officer

Fax:

Either party may, by written notice delivered to the other party in accordance with this Section 15.02, change the address to which delivery of any notice shall thereafter be made.

Section 15.03. Amendment and Waiver. This Agreement may not be altered or

amended, nor may any rights hereunder be waived, except by an instrument in writing executed by the party or parties to be charged with such amendment or waiver. No waiver of any terms, provision or condition of or failure to exercise or delay in exercising any rights or remedies under this Agreement, in any one or more instances, shall be deemed to be, or construed as, a further or continuing waiver of any such term, provision, condition, right or remedy or as a waiver of any other term, provision or condition of this Agreement.

Section 15.04. Entire Agreement. This Agreement, together with the

Ancillary Agreements, constitutes the entire understanding of the parties hereto with respect to the subject matter hereof, superseding all negotiations, prior discussions and prior agreements and understandings relating to such subject matter. To the extent that the provisions of this Agreement are inconsistent with the provisions of any Ancillary Agreement with respect to the subject matter thereof, the provisions of such Ancillary Agreement shall prevail to the extent of the inconsistency.

Section 15.05. Consolidation, Merger, Etc.; Parties in Interest; Termination.

(a) Without limiting Section 15.05(b), neither party (referred to in this Section 15.05(a) as a "transferring party") shall consolidate with or merge into any other entity or directly or indirectly convey, transfer or lease all or substantially all of its properties and assets to any entity, unless, in each case, the other party to such transaction expressly assumes, by a written agreement, executed and delivered to the other party hereto, in form reasonably

satisfactory to such other party, all of the Liabilities of the transferring party under this Agreement and the Ancillary Agreements and the due and punctual performance or observance of every agreement and covenant of this Agreement and Ancillary Agreements on the part of the transferring party to be performed or observed.

- (b) Neither of the parties hereto may assign its rights or delegate any of its duties under this Agreement without the prior written consent of each other party. This Agreement shall be binding upon, and shall inure to the benefit of, the parties hereto and their respective successors and permitted assigns. Nothing contained in this Agreement, express or implied, is intended to confer any benefits, rights or remedies upon any person or entity other than members of the Equifax Group and the Certegy Group and the Equifax Indemnitees and Certegy Indemnitees under Articles IV and V hereof.
- (c) This Agreement (including Articles IV and V hereof) may be terminated and the Distribution may be amended, modified or abandoned at any time prior to the Distribution by and in the sole discretion of Equifax without the approval of Certegy or the shareholders of Equifax. In the event of such termination, no party shall have any liability of any kind to any other party or any other person. After the Distribution, this Agreement may not be terminated except by an agreement in writing signed by the parties; provided, however, that Articles IV and V shall not be terminated or amended after the Distribution in respect of any Equifax Indemnitee or Certegy Indemnitee without the consent of such person.

Section 15.06. Further Assurances and Consents. In addition to the

actions specifically provided for elsewhere in this Agreement, each of the parties hereto will use its reasonable efforts to (a) execute and deliver such further instruments and documents and take such other actions as any other party may reasonably request in order to effectuate the purposes of this Agreement and to carry out the terms hereof and (b) take, or cause to be taken, all actions, and do, or cause to be done, all things, reasonably necessary, proper or advisable under applicable laws, regulations and agreements or otherwise to consummate and make effective the transactions contemplated by this Agreement, including, without limitation, using its reasonable efforts to obtain any consents and approvals, make any filings and applications and remove any liens, claims, equity or other encumbrance on an Asset of the other party necessary or desirable in order to consummate the transactions contemplated by this Agreement; provided that no party hereto shall be obligated to pay any consideration therefor (except for filing fees and other similar charges) to any third party from whom such consents, approvals and amendments are requested or to take any action or omit to take any action if the taking of or the omission to take such action would be unreasonably burdensome to the party or its Group or the business thereof.

Section 15.07. Severability. The provisions of this Agreement are

severable and should any provision hereof be void, voidable or unenforceable under any applicable law, such provision shall not affect or invalidate any other provision of this Agreement, which shall continue to govern the relative rights and duties of the parties as though such void, voidable or unenforceable provision were not a part hereof.

Section 15.08. Governing Law. This Agreement shall be construed in accordance with, and governed by, the laws of the State of Georgia, without

accordance with, and governed by, the laws of the State of Georgia, withour egard to the conflicts of law rules of such state.

Section 15.09. Counterparts. This Agreement may be executed in one or

more counterparts, each of which shall be deemed an original instrument, but all of which together shall constitute but one and the same agreement.

Section 15.10. Disputes.

- (a) Except as otherwise provided in any Ancillary Agreement, all disputes, controversies or claims between members of the Equifax Group and the Certegy Group, which are parties to this Agreement or any Ancillary Agreement, arising out of or relating to this Agreement or any Ancillary Agreement or the performance by the parties of its or their terms, whether based on contract, tort, statute or otherwise, including, but not limited to, disputes in connection with claims by third parties (collectively, "Disputes"), shall be resolved only in accordance with the provisions of this Section 15.10; provided, however, that nothing contained herein shall preclude any party to a Dispute from seeking or obtaining (i) injunctive relief to prevent an actual or threatened breach of any of the provisions of this Agreement or any Ancillary Agreement, or (ii) equitable or other judicial relief to enforce the provisions of this Section 15.10 hereof or to preserve the status quo pending resolution of Disputes hereunder.
- (b) Any party or parties to a Dispute of either Group may give the parties to the Dispute of the other Group written notice of the Dispute initiating the procedures hereunder. Within 10 days after delivery of the notice of a Dispute, the receiving parties shall submit to the other a written response. The notice and the response shall include a statement of each party's respective position and a summary of arguments supporting that position and the name and title of the executive who will represent the claimants and of any other person who will accompany such executive in resolving the Dispute. Within twenty (20) days after delivery of the first notice, the executives of both Groups shall meet at a mutually acceptable time and place, and thereafter as often as they reasonably deem necessary, and shall negotiate in good faith to attempt to resolve the Dispute. All reasonable requests for information made by one party to the other will be honored.
- (c) If the Dispute has not been resolved by negotiation within sixty (60) days of the notice of Dispute, the Dispute may then be submitted for resolution by binding arbitration by a panel of three arbitrators conducted in accordance with the Rules of Arbitration of the International Chamber of Commerce (the "Rules"), as modified by this Section 15.10. Arbitration shall be initiated by notice from the parties to a Dispute of either Group ("claimants") given to the parties to the Dispute of the other Group ("respondents"). The claimants acting jointly, on the one hand, and the respondents acting jointly, on the other hand, shall each appoint one arbitrator within 14 days after the claimants give an arbitration notice. The two arbitrators so appointed shall designate the third arbitrator by mutual agreement within 30 days after the arbitration notice is given. If the two arbitrators so appointed fail to designate the third arbitrator within such period, then any party may request the International Chamber of Commerce to appoint the third arbitrator within 14 days after such request. The third arbitrator shall be a lawyer licensed to practice in the State of Georgia who shall not be related to, employed by, affiliated with or have had a substantial or ongoing business relationship with any member of either Group.

(d) The arbitration shall be conducted in Atlanta, Georgia (or at any other place agreed upon by the parties and the arbitrators). The parties will facilitate the arbitration by: (i) making available to one another and to the arbitrators for examination, inspection and extraction all documents, books, records and personnel under their control if determined by the arbitrators to be relevant to the dispute; (ii) conducting arbitration hearings to the greatest extent possible on successive days; and (iii) observing strictly the time periods established by this Section 15.10, the Rules or by the Arbitrators for submission of evidence or briefs. All issues in connection with the Dispute, including procedural issues, shall be decided by the concurrence of at least two arbitrators, and all decisions by the arbitrators shall be accompanied by a written opinion setting forth the findings of fact and conclusions of law relied upon in reaching the decision. The panel of arbitrators shall decide the issues submitted to it in accordance with the language and commercial purposes of this Agreement or the relevant Ancillary Agreement (as applicable); provided that all questions of law shall be governed by the internal laws of the State of Georgia, without regard to its conflict of laws rules. The arbitrators' decision shall be final and binding as to all matters at issue in the Dispute; provided, however, if necessary such decision may be enforced by either party in any court having jurisdiction over the parties or the subject matter of the Dispute. Unless the arbitrators shall assess the costs and expenses of the arbitration proceeding and of the parties differently, each party shall pay its costs and expenses incurred in connection with the arbitration proceeding, and the costs and expenses of the arbitrators shall be shared equally by the parties.

Section 15.11. Headings. The Article and Section headings set forth in

this Agreement are included for administrative, organizational and convenience purposes, and are not intended to affect the meaning of the provisions set forth in this Agreement or to be used in the interpretation of this Agreement.

[signatures appear on following page]

EQUIFAX INC.

Ву:	
	Title
CER	TEGY INC.
Ву:	
	Title
	-25-

RIGHTS AGREEMENT

This RIGHTS AGREEMENT, dated as of ______, 2001 (this "Agreement"), is made and entered into by and between Certegy Inc., a Georgia corporation (the "Company"), and SunTrust Bank, a Georgia banking corporation (the "Rights Agent").

RECITALS

WHEREAS, on ________, 2001, the Board of Directors of the Company authorized and declared a dividend distribution of one right ("Right") for each share of Common Stock, par value \$.01 per share, of the Company (a "Common Share") outstanding as of the Close of Business (as hereinafter defined) on ______, 2001 (the "Record Date"), each Right initially representing the right to purchase one Common Share, upon the terms and subject to the conditions herein set forth, and further authorized and directed the issuance of one Right with respect to each Common Share issued or delivered by the Company (whether originally issued or delivered from the Company's treasury) after the Record Date but prior to the earlier of the Distribution Date (as hereinafter defined) and the Expiration Date (as hereinafter defined).

NOW THEREFORE, in consideration of the premises and the mutual agreements herein set forth, the parties hereby agree as follows:

Section 1. Certain Definitions. For purposes of this Agreement, the following terms shall have the meanings indicated:

(a) "Acquiring Person" shall mean any Person (other than the Company or any Subsidiary of the Company or any employee benefit or stock ownership plan of the Company or of any Subsidiary of the Company or any entity holding Common Shares for or pursuant to the terms of any such plan or, prior to the Spinoff Date, Equifax Inc. or any of its Affiliates or Associates) who or which, together with all Affiliates and Associates of such Person, shall be the Beneficial Owner of 15% or more of the Common Shares then outstanding; provided, however, that a Person shall not be deemed to have become an Acquiring Person solely as a result of a reduction in the number of Common Shares outstanding or solely as a result of any acquisition that is approved in advance by the Board of Directors of the Company, in each case unless and until such time as (i) such Person or any Affiliate or Associate of such Person shall thereafter become the Beneficial Owner of any additional Common Shares, other than as a result of a stock dividend, stock split or similar transaction effected by the Company in which all holders of Common Shares are treated equally or in a transaction that is approved in advance by the Board of Directors, or (ii) any other Person who is the Beneficial Owner of any Common Shares shall thereafter become an Affiliate Notwithstanding the foregoing, if the Board of or Associate of such Person. Directors determines in good faith that a Person who otherwise would be an Acquiring Person as defined under the foregoing provisions of this Section 1(a) has become such inadvertently, and such Person has divested or agrees to divest as promptly as practicable (as determined by the Board of Directors of the Company) a sufficient number of Common Shares so that such Person would no longer be an Acquiring Person as defined under the foregoing

provisions, then such Person shall not be deemed an Acquiring Person for any purposes of this Agreement unless and until such Person shall again become an Acquiring Person as herein defined.

- (b) "Affiliate" and "Associate" shall have the respective meanings ascribed to such terms in Rule 12b-2 of the General Rules and Regulations under the Exchange Act, as in effect on the date of this Agreement.
- (c) A Person shall be deemed the Beneficial Owner of, and to "beneficially own," any securities:
 - (i) which such Person or any of such Person's Affiliates or Associates, directly or indirectly, has the right to acquire (whether such right is exercisable immediately or only after the passage of time) pursuant to any agreement, arrangement or understanding (whether or not in writing), or upon the exercise of conversion rights, exchange rights, rights, warrants or options, or otherwise (in each case, other than upon exercise or exchange of the Rights); provided, however, that a Person shall not be deemed the Beneficial Owner of, or to beneficially own, securities tendered pursuant to a tender or exchange offer made by or on behalf of such Person or any of such Person's Affiliates or Associates until such tendered securities are accepted for purchase or exchange; or
 - (ii) which such Person or any of such Person's Affiliates or Associates, directly or indirectly, has the right to vote or dispose of, including pursuant to any agreement, arrangement or understanding (whether or not in writing); or
 - (iii) of which any other Person is the Beneficial Owner, if such Person or any of such Person's Affiliates or Associates has any agreement, arrangement or understanding (whether or not in writing) with such other Person (or any of such other Person's Affiliates or Associates) with respect to acquiring, holding, voting or disposing of any securities of the Company; provided, however, that a Person shall not be deemed the Beneficial Owner of, or to beneficially own, any security (A) if such Person has the right to vote such security pursuant to an agreement, arrangement or understanding (whether or not in writing) which (1) arises solely from a revocable proxy given to such Person in response to a public proxy or consent solicitation made pursuant to, and in accordance with, the applicable rules and regulations of the Exchange Act and (2) is not also then reportable on Schedule 13D under the Exchange Act (or any comparable or successor report), or (B) if such beneficial ownership arises solely as a result of such Person's status as a "clearing agency", as defined in Section 3(a)(23) of the Exchange Act; and provided, further, that nothing in this paragraph (c) shall cause a Person engaged in business as an underwriter of securities to be the Beneficial Owner of, or to beneficially own, any securities acquired through such Person's participation in good faith in an underwriting syndicate until the expiration of 40 calendar days after the date of such acquisition, or such later date as the Board of Directors of the Company may determine in any specific case.

- (d) "Business Day" shall mean any day other than a Saturday, Sunday or a day on which banking institutions in the State of Georgia (or such other state in which the principal office of the Rights Agent is located) are authorized or obligated by law or executive order to close.
- (e) "Close of Business" on any given date shall mean 5:00 P.M., Eastern time, on such date; provided, however, that if such date is not a Business Day it shall mean 5:00 P.M., Eastern time, on the next succeeding Business Day.
- (f) "Common Shares" when used with reference to the Company shall mean the Common Stock, par value \$.01 per share, of the Company; provided, however, that, if the Company is the continuing or surviving corporation in a transaction described in Section 11(a)(ii) or Section 13(a)(ii) hereof, "Common Shares" when used with reference to the Company shall mean the capital stock or equity security with the greatest aggregate voting power of the Company. Common Shares when used with reference to any corporation or other legal entity, other than the Company, including an Issuer, shall mean the capital stock or equity security with the greatest aggregate voting power of such corporation or other legal entity.
 - (g) "Company" shall mean Certegy Inc., a Georgia corporation.
- (h) "Distribution Date" shall mean the earliest of: (i) the Close of Business on the tenth calendar day (or, unless the Distribution Date shall have previously occurred, such later date as may be specified by the Board of Directors of the Company) after the Share Acquisition Date, (ii) the Close of Business on the tenth Business Day (or, unless the Distribution Date shall have previously occurred, such later date as may be specified by the Board of Directors of the Company) after the date of the commencement of a tender or exchange offer by any Person (other than the Company or any Subsidiary of the Company or any employee benefit or stock ownership plan of the Company or of any Subsidiary of the Company or any entity holding Common Shares for or pursuant to the terms of any such plan), if upon the consummation thereof such Person would be the Beneficial Owner of 15% or more of the outstanding Common Shares, and (iii) the Close of Business on the tenth calendar day after the first date of public announcement by the Company or an Acquiring Person (by press release, filing made with the Securities and Exchange Commission or otherwise) of the first occurrence of a Triggering Event; provided, however, that if the earliest of such dates would otherwise occur prior to the Record Date, the Distribution Date shall mean the Close of Business on the Record Date.
- (i) "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.
- (j) "Expiration Date" shall mean the earliest of (i) the Close of Business on the Final Expiration Date, (ii) the time at which the Rights are redeemed as provided in Section 23 hereof, and (iii) the time at which all exercisable Rights are exchanged as provided in Section 27 hereof.
- (k) "Final Expiration Date" shall mean the tenth anniversary of $% \left(1\right) =\left(1\right) +\left(1\right) +\left($

- (1) "Flip-in Event" shall mean any event described in clauses (A) (B) or (C) of Section 11(a)(ii) hereof.
- (m) "Flip-over Event" shall mean any event described in subsections (i),
 (ii), (iii) or (iv) of Section 13(a) hereof.
 - (n) "Issuer" shall have the meaning set forth in Section 13(b) hereof.
- (o) "NASDAQ" shall mean the National Association of Securities Dealers, Inc. Automated Quotation System.
 - (p) "NYSE" shall mean the New York Stock Exchange, Inc.
- (q) "Person" shall mean any individual, firm, corporation, partnership or other legal entity, and shall include any successor (by merger or otherwise) of such entity.
- (r) "Purchase Price" shall mean initially \$125.00 per Common Share, and shall be subject to further adjustment from time to time as provided in this Agreement.
- (s) "Redemption Price" shall mean \$0.01 per Right, subject to adjustment by resolution of the Board of Directors of the Company to reflect any stock split, stock dividend or similar transaction occurring after the date hereof.
- (t) "Right" shall have the meaning set forth in the Recitals to $\,$ this Agreement.
- (u) "Right Certificates" shall mean certificates evidencing the Rights, in substantially the form of Exhibit A attached hereto.
- (v) "Rights Agent" shall mean SunTrust Bank, unless and until a successor Rights Agent shall have become such pursuant to the terms of this Agreement, and thereafter, "Rights Agent" shall mean such successor Rights Agent.
 - (w) "Securities Act" shall mean the Securities Act of 1933, as amended.
- (x) "Share Acquisition Date" shall mean the first date of public announcement by the Company or an Acquiring Person (by press release, filing made with the Securities and Exchange Commission or otherwise) that an Acquiring Person has become such.
- (y) "Spinoff Date" shall mean the time at which Equifax Inc. shall distribute all of the Common Shares of the Company it owns to its shareholders pursuant to the Distribution Agreement, Plan of Reorganization and Distribution dated as of ______, 2001, as amended or supplemented from time to time.
- (z) "Subsidiary" when used with reference to any Person shall mean any corporation or other legal entity of which a majority of the voting power of the voting equity securities or equity interests is owned, directly or indirectly, by such Person; provided, however, that for

purposes of Section 13(b) hereof, Subsidiary when used with reference to any Person shall mean any corporation or other legal entity of which at least 20% of the voting power of the voting equity securities or equity interests is owned, directly or indirectly, by such Person.

- (aa) "Trading Day" shall mean any day on which the principal national securities exchange on which the Common Shares are listed or admitted to trading is open for the transaction of business or, if the Common Shares are not listed or admitted to trading on any national securities exchange, a Business Day.
 - (bb) "Triggering Event" shall mean any Flip-in Event or Flip-over Event.

Section 2. Appointment of Rights Agent. The Company hereby appoints the Rights Agent to act as agent for the Company and the holders of the Rights (who, in accordance with Section 3 hereof, shall also be, prior to the Distribution Date, the holders of the Common Shares) in accordance with the terms and conditions hereof, and the Rights Agent hereby accepts such appointment and hereby certifies that it complies with the requirements of the NYSE governing transfer agents and registrars. The Company may from time to time act as Co-Rights Agent or appoint such Co-Rights Agents as it may deem necessary or desirable. Any actions which may be taken by the Rights Agent pursuant to the terms of this Agreement may be taken by any such Co-Rights Agent. To the extent that any Co-Rights Agent takes any action pursuant to this Agreement, such Co-Rights Agent shall be entitled to all of the rights and protections of and subject to all of the applicable duties and obligations imposed upon, the Rights Agent pursuant to the terms of this Agreement.

Section 3. Issue of Right Certificates.

- (a) Until the Distribution Date, (i) the Rights shall be evidenced by the certificates representing Common Shares registered in the names of the record holders thereof (which certificates representing Common Shares shall also be deemed to be Right Certificates), (ii) the Rights shall be transferable only in connection with the transfer of the underlying Common Shares, and (iii) the surrender for transfer of any certificates evidencing Common Shares in respect of which Rights have been issued shall also constitute the transfer of the Rights associated with the Common Shares evidenced by such certificates.
- (b) Promptly upon request, the Company shall send a copy of this Agreement by first-class, postage prepaid mail, to any record holder of Common Shares requesting the same (or, at the Company's option, shall send a letter summarizing the terms of the Rights), at the address of such holder shown on the records of the Company as of such date.
- (c) Rights shall be issued by the Company in respect of all Common Shares (other than Common Shares issued upon the exercise or exchange of any Right) issued or delivered by the Company (whether originally issued or delivered from the Company's treasury) after the Record Date but prior to the earlier of the Distribution Date and the

Expiration Date. Certificates evidencing such Common Shares shall have stamped on, impressed on, printed on, written on or otherwise affixed to them the following legend or such similar legend as the Company may deem appropriate and as is not inconsistent with the provisions of this Agreement, or as may be required to comply with any applicable law or with any rule or regulation made pursuant thereto or with any rule or regulation of any stock exchange or transaction reporting system on which the Common Shares may from time to time be listed or quoted, or to conform to usage:

This Certificate also evidences and entitles the holder hereof to certain Rights as set forth in a Rights Agreement between Certegy Inc. and SunTrust Bank, dated as of ________, 2001 (the "Rights Agreement"), the terms of which are hereby incorporated herein by reference and a copy of which is on file at the principal executive offices of Certegy Inc. Under certain circumstances, as set forth in the Rights Agreement, such Rights may be redeemed, may expire, may be amended or may be evidenced by separate certificates and no longer be evidenced by this Certificate. Certegy Inc. will mail to the holder of this Certificate a copy of the Rights Agreement without charge promptly after receipt of a written request therefor. Under certain circumstances as set forth in the Rights Agreement, Rights beneficially owned by an Acquiring Person or any Affiliate or Associate of an Acquiring Person (as such terms are defined in the Rights Agreement) may become null and void.

(d) As promptly as practicable after the Distribution Date, the Company shall prepare and execute, the Rights Agent will countersign and the Company shall send or cause to be sent (and the Rights Agent shall, if requested, send), by first-class, insured, postage prepaid mail, to each record holder of Common Shares as of the Close of Business on the Distribution Date, at the address of such holder shown on the records of the Company, a Right Certificate, evidencing one Right for each Common Share so held, subject to adjustment. As of and after the Distribution Date, the Rights shall be evidenced solely by such Right Certificates.

Section 4. Form of Right Certificates. (a) The Right Certificates (and the form of election to purchase and form of assignment to be printed on the reverse thereof) shall be substantially in the form set forth as Exhibit A hereto with such changes, marks of identification or designation and such legends, summaries or endorsements printed thereon as the Company may deem appropriate and as are not inconsistent with the provisions of this Agreement, or as may be required to comply with any applicable law or with any rule or regulation made pursuant thereto or with any rule or regulation of any stock exchange or transaction reporting system on which the Rights may from time to time be listed or quoted, or to conform to usage. Subject to the provisions of Section 22 hereof, the Right Certificates, whenever issued, on their face shall entitle the holders thereof to purchase such number of Common Shares as shall be set forth therein at the Purchase Price set forth therein, but the Purchase Price, the number and kind of securities issuable upon exercise of each Right and the number of Rights outstanding shall be subject to adjustment as provided herein.

6

(b) Any Rights Certificate issued hereunder that represents Rights beneficially owned by (i) an Acquiring Person or any Associate or Affiliate of an Acquiring Person, (ii) a transferee of an Acquiring Person (or an any such Associate or Affiliate) who becomes a transferee after the Acquiring Person becomes such, or (iii) a transferee of an Acquiring Person (or of any such Associate or Affiliate) who becomes a transferee prior to or concurrently with the Acquiring Person becoming such and receives such Rights pursuant to either (A) a transfer (whether or not for consideration) from the Acquiring Person to holders of equity interests in such Acquiring Person or to any Person with whom such Acquiring Person has any continuing agreement, arrangement or understanding regarding the transferred Rights or (b) a transfer that the Board of Directors of the Company has determined is part of a plan, arrangement or understanding that has the primary purpose or effect the avoidance of Section 7(d) hereof, and any Right Certificate issued pursuant to Section 6 or Section 11 hereof upon transfer, exchange, replacement or adjustment of any other Right Certificate referred to in this sentence, shall contain (to the extent feasible) the following legend:

The Rights represented by this Right Certificate are or were beneficially owned by Person who was or became an Acquiring Person or an Affiliate or Associate of an Acquiring Person (as such terms are defined in the Rights Agreement). Accordingly, this Right Certificate and the Rights represented hereby may become null and void in the circumstances specified in the Rights Agreement.

Section 5. Countersignature and Registration.

- (a) The Right Certificates shall be executed on behalf of the Company by its Chairman of the Board, President or any Vice President, either manually or by facsimile signature, and shall have affixed thereto the Company's seal or a facsimile thereof which shall be attested by the Secretary or an Assistant Secretary of the Company, either manually or by facsimile signature. The Right Certificates shall be manually countersigned by the Rights Agent and shall not be valid for any purpose unless so countersigned. In case any officer of the Company who shall have signed any of the Right Certificates shall cease to be such officer of the Company before countersignature by the Rights Agent and issuance and delivery by the Company, such Right Certificates, nevertheless, may be countersigned by the Rights Agent, and issued and delivered by the Company with the same force and effect as though the person who signed such Right Certificates had not ceased to be such officer of the Company; and any Right Certificate may be signed on behalf of the Company by any person who, at the actual date of the execution of such Right Certificate, shall be a proper officer of the Company to sign such Right Certificate, although at the date of the execution of this Rights Agreement any such person was not such an officer.
- (b) Following the Distribution Date, the Rights Agent shall keep or cause to be kept, at the principal office of the Rights Agent designated for such purpose and at such other offices as may be required to comply with any applicable law or with any rule or regulation made pursuant thereto or with any rule or regulation of any stock exchange or any transaction reporting system on which the Rights may from time to time be listed

or quoted, books for registration and transfer of the Right Certificates issued hereunder. Such books shall show the names and addresses of the respective holders of the Right Certificates, the number of Rights evidenced on its face by each of the Right Certificates and the date of each of the Right Certificates.

Section 6. Transfer, Split Up, Combination and Exchange of Right Certificates; Mutilated, Destroyed, Lost or Stolen Right Certificates.

- (a) Subject to the provisions of Sections 7(d) and 14 hereof, at any time after the Close of Business on the Distribution Date and prior to the Expiration Date, any Right Certificate or Right Certificates representing exercisable Rights may be transferred, split up, combined or exchanged for another Right Certificate or Right Certificates, entitling the registered holder to purchase a like number of Common Shares (or other securities, as the case may be) as the Right Certificate or Right Certificates surrendered then entitled such holder (or former holder in the case of a transfer) to purchase. Any registered holder desiring to transfer, split up, combine or exchange Right Certificates shall make such request in writing delivered to the Rights Agent, and shall surrender the Right Certificate or Right Certificates to be transferred, split up, combined or exchanged at the principal office of the Rights Agent designated for such purpose. Thereupon or as promptly as practicable thereafter, subject to the provisions of Sections 7(d) and 14 hereof, the Company shall prepare, execute and deliver to the Rights Agent, and the Rights Agent shall countersign and deliver a Right Certificate or Right Certificates, as the case may be, as so requested. The Company may require payment of a sum sufficient to cover any tax or governmental charge that may be imposed in connection with any transfer, split up, combination or exchange of Right Certificates.
- (b) Upon receipt by the Company and the Rights Agent of evidence reasonably satisfactory to them of the loss, theft, destruction or mutilation of a Right Certificate, and of indemnity or security reasonably satisfactory to them, and reimbursement to the Company and the Rights Agent of all reasonable expenses incidental thereto, and upon surrender to the Rights Agent and cancellation of the Right Certificate if mutilated, the Company shall prepare, execute and deliver a new Right Certificate of like tenor to the Rights Agent and the Rights Agent shall countersign and deliver such new Right Certificate to the registered holder in lieu of the Right Certificate so lost, stolen, destroyed or mutilated.

Section 7. Exercise of Rights; Purchase Price; Expiration Date of Rights.

(a) The registered holder of any Right Certificate may exercise the Rights evidenced thereby (except as otherwise provided herein) in whole or in part at any time after the Distribution Date and prior to the Expiration Date, upon surrender of the Right Certificate, with the form of election to purchase on the reverse side thereof duly executed, to the Rights Agent at the office or offices of the Rights Agent designated for such purpose, together with payment in cash, in lawful money of the United States of America by certified check or bank draft payable to the order of the Company equal to the sum of (i) the aggregate Purchase Price for the total number of securities as to which

such surrendered Rights are exercised and (ii) an amount equal to any applicable transfer tax required to be paid by the holder of such Right Certificate in accordance with the provisions of Section 9 hereof. In lieu of the cash payment referred to in the immediately preceding sentence, following the occurrence of a Triggering Event, the registered holder of a Right Certificate may exercise the Rights evidenced thereby (except as otherwise provided herein) in whole or in part upon surrender of the Right Certificate as described above together with an election to exercise such Rights without payment of cash on the reverse side thereof duly completed. With respect to any Rights as to which such an election is made, the holder shall receive a number of Common Shares or other securities having a value equal to the difference between (x) the value of the Common Shares or other securities that would have been issuable upon payment of the cash amount as described above, and (y) the sum of items (i) and (ii) above. For purposes of this Section 7(a), the value of any Common Share or other security shall be the current per share market price of a Common Share (determined pursuant to Section 11(d) hereof) on the Trading Day immediately preceding the date of the first occurrence of a Triggering Event.

- (b) Upon receipt of a Right Certificate representing exercisable Rights with the form of election to purchase duly executed, accompanied by either payment as described in Section 7(a) above or a duly completed election to exercise without payment of cash, the Rights Agent shall promptly (i) requisition from any transfer agent of the Common Shares (or make available, if the Rights Agent is the transfer agent) certificates representing the number of Common Shares to be purchased (and the Company hereby irrevocably authorizes and directs its transfer agent to comply with all such requests), (ii) after receipt of such certificates, cause the same to be delivered to or upon the order of the registered holder of such Right Certificate, registered in such name or names as may be designated by such holder, (iii) when appropriate, requisition from the Company or any transfer agent therefor (or make available, if the Rights Agent is the transfer agent) certificates representing the number of equivalent common shares to be issued in lieu of the issuance of Common Shares in accordance with the provisions of Section 11(a)(iii) hereof, (iv) when appropriate, after receipt of such certificates, cause the same to be delivered to or upon the order of the registered holder of such Right Certificate, registered in such name or names as may be designated by such holder, (v) when appropriate, requisition from the Company the amount of cash to be paid in lieu of the issuance of fractional shares in accordance with the provisions of Section 14 hereof or in lieu of the issuance of Common Shares in accordance with the provisions of Section 11(a)(iii) hereof, (vi) when appropriate, after receipt, deliver such cash to or upon the order of the registered holder of such Right Certificate, and (vii) when appropriate, deliver any due bill or other instrument provided to the Rights Agent by the Company for delivery to the registered holder of such Right Certificate as provided by Section 11(1) hereof.
- (c) In case the registered holder of any Right Certificate shall exercise less than all the Rights evidenced thereby, the Company shall prepare, execute and deliver a new Right Certificate evidencing Rights equivalent to the Rights remaining unexercised and the Rights Agent shall countersign and deliver such new Right Certificate to the

registered holder of such Right Certificate or to his duly authorized assigns, subject to the provisions of Section 14 hereof.

- (d) Notwithstanding anything in this Agreement to the contrary, from and after the later of the Distribution Date and the first occurrence of a Triggering Event, any Rights beneficially owned by (i) an Acquiring Person or an Associate or Affiliate of an Acquiring Person, (ii) a transferee of an Acquiring Person (or of any such Associate or Affiliate) who becomes a transferee after the Acquiring Person becomes such, or (iii) a transferee of an Acquiring Person (or of any such Associate or Affiliate) who becomes a transferee prior to or concurrently with the Acquiring Person becoming such and receives such rights pursuant to either (A) a transfer (whether or not for consideration) from the Acquiring Person to holders of equity interests in such Acquiring Person or to any Person with whom the Acquiring Person has any continuing agreement, arrangement or understanding regarding the transfer of Rights, or (B) a transfer that the Board of Directors of the Company has determined is part of a plan, arrangement or understanding that has as a primary purpose or effect the avoidance of this Section 7(d), shall become null and void without any further action, no holder of such Rights shall have any rights whatsoever with respect to such Rights, whether under any provision of this Agreement or otherwise, and any Right Certificate delivered to the Rights Agent for transfer to an Acquiring Person or any Affiliate or Associate thereof shall be cancelled. The Company shall use all reasonable efforts to ensure that the provisions of this Section 7(d) and Section 4(b) hereof are complied with, but shall have no liability to any holder of Rights Certificates or other Person as a result of its failure to make any determinations with respect to an Acquiring Person or its Affiliates, Associates or transferees hereunder.
- (e) Notwithstanding anything in this Agreement to the contrary, neither the Rights Agent nor the Company shall be obligated to undertake any action with respect to any purported transfer, split up, combination or exchange of any Right Certificate pursuant to Section 6 hereof or exercise of a Right Certificate as set forth in this Section 7 unless the registered holder of such Right Certificate shall have (i) completed and signed the certificate following the form of assignment or form of election to purchase, as applicable, set forth on the reverse side of the Right Certificate surrendered for such transfer, split up, combination, exchange or exercise, and (ii) provided such additional evidence of the identity of the Beneficial Owner (or former Beneficial Owner) or Affiliates or Associates thereof as the Company shall have reasonably requested.

Section 8. Cancellation and Destruction of Right Certificates. All Right Certificates surrendered for the purpose of exercise, transfer, split up, combination or exchange shall, if surrendered to the Company or to any of its stock transfer agents, be delivered to the Rights Agent for cancellation or in cancelled form, or, if surrendered to the Rights Agent, shall be cancelled by it, and no Right Certificates shall be issued in lieu thereof except as expressly permitted by this Agreement. The Company shall deliver to the Rights Agent for cancellation and retirement, and the Rights Agent shall so cancel and retire, any other Right Certificate purchased or acquired by the Company otherwise than upon the exercise thereof. The Rights Agent shall deliver all cancelled Right Certificates to the Company, or shall, at the written request of the Company, destroy such

cancelled Right Certificates, and in such case shall deliver a certificate of destruction thereof to the Company.

Section 9. Company Covenants Concerning Securities and Rights. The Company covenants and agrees that:

- (a) So long as the Common Shares issuable upon the exercise of the Rights may be listed on a national securities exchange, it shall endeavor to cause, from and after such time as the Rights become exercisable, all securities reserved for issuance upon the exercise of Rights to be listed on such exchange upon official notice of issuance.
- (b) It shall take all such action as may be necessary to ensure that all Common Shares and/or other securities delivered upon exercise of Rights, at the time of delivery of the certificates for such securities shall be (subject to payment of the Purchase Price) duly and validly authorized and issued, fully paid and nonassessable securities.
- (c) It shall pay when due and payable any and all federal and state transfer taxes and charges that may be payable in respect of the issuance or delivery of the Right Certificates and of any certificates representing securities issued upon the exercise of Rights; provided, however, that the Company shall not be required to pay any transfer tax or charge which may be payable in respect of any transfer or delivery of Right Certificates to a person other than, or the issuance or delivery of certificates representing securities issued upon the exercise of Rights in a name other than that of, the registered holder of the Right Certificate evidencing Rights surrendered for exercise, or to issue or deliver any certificates representing securities issued upon the exercise of any Rights until any such tax or charge shall have been paid (any such tax or charge being payable by the holder of such Right Certificate at the time of surrender) or until it has been established to the Company's reasonable satisfaction that no such tax is due.
- (d) It shall use its best efforts (i) to file on an appropriate form, as soon as practicable following the later of the first occurrence of a Triggering Event or the Distribution Date, a registration statement under the Securities Act with respect to the securities issuable upon exercise of the Rights, (ii) to cause such registration statement to become effective as soon as practicable after such filing, and (iii) to cause such registration statement to remain effective (with a prospectus at all times meeting the requirements of the Act) until the earlier of (A) the date as of which the Rights are no longer exercisable for such securities and (B) the Expiration Date. The Company shall also take such action as may be appropriate under, or to ensure compliance with, the securities or "blue sky" laws of the various states in connection with the exercisability of the Rights. The Company may temporarily suspend, for a period of time after the date set forth in clause (i) of the first sentence of this Section 9(d), the exercisability of the Rights in order to prepare and file such registration statement and to permit it to become effective. Upon any such suspension, the Company shall issue a public announcement stating that the exercisability of the Rights has been temporarily suspended, as well as a public announcement at such time as the suspension is no longer in effect. In addition, if the Company shall determine that a registration statement should be filed under the

Securities Act or any state securities laws following the Distribution Date, the Company may temporarily suspend the exercisability of the Rights in each relevant jurisdiction until such time as a registration statement has been declared effective and, upon any such suspension, the Company shall issue a public announcement stating that the exercisability of the Rights has been temporarily suspended, as well as a public announcement at such time as the suspension is no longer in effect. Notwithstanding anything in this Agreement to the contrary, the Rights shall not be exercisable in any jurisdiction if the requisite registration or qualification in such jurisdiction shall not have been effected or the exercise of the Rights shall not be permitted under applicable law.

- (e) Notwithstanding anything in this Agreement to the contrary, after the Distribution Date it shall not, except as permitted by Section 23, 26 or 27 hereof, take (or permit any Subsidiary to take) any action if at the time such action is taken it is reasonably foreseeable that such action will eliminate or otherwise diminish the benefits intended to be afforded by the Rights.
- (f) In the event that the Company is obligated to issue other securities of the Company and/or pay cash pursuant to Section 11, 13 or 14 hereof, it shall make all arrangements necessary so that such other securities and/or cash are available for distribution by the Rights Agent, if and when appropriate.

Section 10. Record Date. Each Person in whose name any certificate representing Common Shares is issued upon the exercise of Rights shall for all purposes be deemed to have become the holder of record of the Common Shares represented thereby on, and such certificate shall be dated, the date upon which the Right Certificate evidencing such Rights was duly surrendered and payment of the Purchase Price (and all applicable transfer taxes) was made; provided, however, that if the date of such surrender and payment is a date upon which the Common Shares transfer books of the Company are closed, such Person shall be deemed to have become the record holder of such securities on, and such certificate shall be dated, the next succeeding Business Day on which the Common Shares transfer books of the Company are open. Prior to the exercise of the Rights evidenced thereby, the holder of a Right Certificate shall not be entitled to any rights of a shareholder of the Company with respect to securities for which the Rights shall be exercisable, including, without limitation, the right to vote, to receive dividends or other distributions or to exercise any preemptive rights, and shall not be entitled to receive any notice of any proceedings of the Company, except as provided herein.

Section 11. Adjustment of Purchase Price, Number and Kind of Securities or Number of Rights. The Purchase Price, the number and kind of securities issuable upon exercise of each Right and the number of Rights outstanding are subject to adjustment from time to time as provided in this Section 11.

(a) (i) In the event that the Company shall at any time after the date of this Agreement (A) effect a dividend on the Common Shares payable in Common Shares, (B) subdivide the outstanding Common Shares, (C) combine the outstanding Common Shares into a smaller number of shares or (D) issue any shares of its capital stock in a

reclassification of the Common Shares (including any such reclassification in connection with a consolidation or merger in which the Company is the continuing or surviving corporation), the Purchase Price in effect at the time of the record date for such dividend or on the effective date of such subdivision, combination or reclassification, and/or the number and/or kind of shares of capital stock issuable on such date upon exercise of a Right, shall be proportionately adjusted so that the holder of any Right exercised after such time shall be entitled to receive upon payment of the Purchase Price then in effect the aggregate number and kind of shares of capital stock which, if such Right had been exercised immediately prior to such date and at a time when the Common Shares transfer books of the Company were open, the holder of such Right would have owned upon such exercise and been entitled to receive by virtue of such dividend, subdivision, combination or reclassification. If an event occurs which would require an adjustment under both this Section 11(a)(i) and Section 11(a)(ii) hereof or Section 13 hereof, the adjustment provided for in this Section 11(a)(i) shall be in addition to, and shall be made prior to, any adjustment required pursuant to Section 11(a)(ii) or Section 13 hereof.

(ii) Subject to the provisions of Section 23, 26 or $\,$ 27 hereof, in the event that:

(A) any Acquiring Person or any Affiliate or Associate of any Acquiring Person, at any time after the date of this Agreement, directly or indirectly, shall (1) merge into the Company or otherwise combine with the Company and the Company shall be the continuing or surviving corporation of such merger or combination (other than in a transaction subject to Section 13 hereof), (2) merge or otherwise combine with any Subsidiary of the Company, (3) in one or more transactions (other than in connection with the exercise or exchange of Rights or the exercise or conversion of securities exercisable for or convertible into shares of any class of capital stock of the Company or any of its Subsidiaries) transfer any assets to the Company or any of its Subsidiaries in exchange (in whole or in part) for shares of any class of capital stock of the Company or any of its Subsidiaries or for securities exercisable for or convertible into shares of any class of capital stock of the Company or any of its Subsidiaries, or otherwise obtain from the Company or any of its Subsidiaries, with or without consideration, any additional shares of any class of capital stock of the Company or any of its Subsidiaries or securities exercisable for or convertible into shares of any class of capital stock of the Company or any of its Subsidiaries (other than as part of a pro rata distribution to all holders of such shares of any class of capital stock of the Company, or any of its Subsidiaries), (4) sell, purchase, lease, exchange, mortgage, pledge, transfer or otherwise dispose (in one or more transactions), to, from, with or of, as the case may be, the Company or any of its Subsidiaries (other than in a transaction subject to Section 13 hereof), assets, including securities, on terms and conditions less favorable to the Company than the Company would be able to obtain in arm's-length negotiation with an unaffiliated third party, (5) receive any compensation from the Company or any of its Subsidiaries other than compensation as a director or for full-time employment as a regular employee, in either case, at rates in accordance with the

Company's (or its Subsidiaries') past practices, or (6) receive the benefit, directly or indirectly (except proportionately as a shareholder), of any loans, advances, guarantees, pledges or other financial assistance or any tax credits or other tax advantage provided by the Company or any of its Subsidiaries; or

- (B) during such time as there is an Acquiring Person, there shall be any reclassification of securities (including any reverse stock split), or recapitalization of the Company, or any merger or consolidation of the Company with any of its Subsidiaries or any other transaction or series of transactions involving the Company or any of its Subsidiaries (whether or not with or into or otherwise involving an Acquiring Person), other than a transaction subject to Section 13 hereof, which has the effect, directly or indirectly, of increasing by more than 1% the proportionate share of the outstanding shares of any class of equity securities or of securities exercisable for or convertible into equity securities of the Company or any of its Subsidiaries of which an Acquiring Person or any Affiliate or Associate of any Acquiring Person, is the Beneficial Owner; or
- (C) any Acquiring Person shall at any time after the date of this Agreement, become the Beneficial Owner of 20% or more of the Common Shares then outstanding (other than pursuant to any transaction set forth in Section 13(a) hereof); provided, however, that a Person shall not be deemed to have become the Beneficial Owner of 20% or more of the Common Shares then outstanding for the purposes of this Section 11(a)(ii)(C) solely as a result of a reduction in the number of Common Shares outstanding or solely as a result of any acquisition that is approved in advance by the Board of Directors of the Company, in each case unless and until such time as (1) such Person or any Affiliate or Associate of such Person shall thereafter become the Beneficial Owner of any additional Common Shares, other than as a result of a stock dividend, stock split or similar transaction effected by the Company in which all holders of Common Shares are treated equally or in a transaction that is approved in advance by the Board of Directors, or (2) any other Person who is the Beneficial Owner of any Common Shares shall thereafter become an Affiliate or Associate of such Person; then, and in each such case, proper provision shall be made so that each holder of a Right, except as provided below, shall thereafter have a right to receive, upon exercise thereof in accordance with the terms of this Agreement at an exercise price per Right equal to the product of the then-current Purchase Price multiplied by the number of Common Shares for which a Right was exercisable immediately prior to the first occurrence of a Triggering Event, such number of Common Shares as shall equal the result obtained by (x) multiplying the thencurrent Purchase Price by the number of Common Shares for which a Right was exercisable immediately prior to the first occurrence of a Triggering Event, and dividing that product by (y) 50% of the current per share market price of the Common Shares (determined pursuant to Section 11(d) hereof) on the date of the first occurrence of a Triggering Event.
- (iii) Upon the occurrence of the Distribution Date or a Flip-in Event, if there shall not be sufficient Common Shares authorized but unissued or issued but not

outstanding to permit the issuance of all the Common Shares issuable in accordance with the provisions hereof upon the exercise of a Right, the Board of Directors of the Company shall use its best efforts promptly to authorize and, subject to the provisions of Section 9(d) hereof, make available for issuance additional Common Shares or other equity securities of the Company having equivalent voting rights and an equivalent value (as determined in good faith by the Board of Directors of the Company) to the Common Shares (for purposes of this Section 11(a)(iii), "equivalent common shares"). In the event that equivalent common shares are so authorized, upon the exercise of a Right in accordance with the provisions of Section 7 hereof, the registered holder shall be entitled to receive (A) Common Shares, to the extent any are available and (B) a number of equivalent common shares, which the Board of Directors of the Company shall have determined in good faith to have a value equivalent to the excess of (x)the aggregate current per share market value of all the Common Shares issuable in accordance with subsection (ii) hereof upon the exercise of a Right (the "Exercise Value") over (y) the aggregate current per share market value of any Common Shares available for issuance upon the exercise of such Right; provided, however, that if at any time after 90 calendar days after the first occurrence of a Flip-in Event, there shall not be sufficient Common Shares and/or equivalent common shares available for issuance upon the exercise of a Right, then the Company shall be obligated to deliver, upon the surrender of such Right and without requiring payment of the Purchase Price, Common Shares (to the extent available), equivalent common shares (to the extent available) and then cash (to the extent permitted by applicable law and any agreements or instruments to which the Company is a party in effect immediately prior to the first occurrence of any Flip-in Event), which securities and cash shall have an aggregate value equal to the excess of (1) the Exercise Value over (2) the product of the then-current Purchase Price multiplied by the number of Common Shares for which a Right was exercisable immediately prior to the first occurrence of a Triggering Event. To the extent that any legal or contractual restrictions prevent the Company from paying the full amount of cash payable in accordance with the foregoing sentence, the Company shall pay to holders of the Rights as to which such payments are being made all amounts which are not then restricted on a pro rata basis and shall continue to make payments on a pro rata basis as funds become available until the full amount due to each such Rights holder has been paid.

(b) In the event that the Company shall fix a record date for the issuance of rights, options or warrants to all holders of Common Shares entitling them (for a period expiring within 45 calendar days after such record date) to subscribe for or purchase Common Shares (or securities having equivalent rights, privileges and preferences as the Common Shares (for purposes of this Section "equivalent common shares")) or securities convertible into Common Shares or equivalent common shares at a price per Common Share or equivalent common share (or having a conversion price per share, if a security convertible into Common Shares or equivalent common shares) less than the current per share market price of the Common Shares (determined pursuant to Section 11(d) hereof) on such record date, the Purchase Price to be in effect after such record date shall be determined by multiplying the Purchase Price in effect immediately prior to such record date by a fraction, the numerator of which shall be the number of Common Shares outstanding on such record date plus the number of Common Shares which the aggregate offering price of the total number of Common Shares and/or equivalent common shares so to be offered (and/or the aggregate initial conversion price of the convertible securities so to be offered) would purchase at such current per share market price and the denominator of which shall be the number of Common Shares outstanding on such record date plus the number of additional Common Shares and/or equivalent common

15

shares to be offered for subscription or purchase (or into which the convertible securities so to be offered are initially convertible). In case such subscription price may be paid in a consideration part or all of which shall be in a form other than cash, the value of such consideration shall be as determined in good faith by the Board of Directors of the Company, whose determination shall be described in a statement filed with the Rights Agent. Common Shares owned by or held for the account of the Company shall not be deemed outstanding for the purpose of any such computation. Such adjustment shall be made successively whenever such a record date is fixed; and in the event that such rights, options or warrants are not so issued, the Purchase Price shall be adjusted to be the Purchase Price which would then be in effect if such record date had not been fixed.

- (c) In the event that the Company shall fix a record date for the making of a distribution to all holders of Common Shares (including any such distribution made in connection with a consolidation or merger in which the Company is the continuing or surviving corporation) of evidences of indebtedness, cash (other than a regular periodic cash dividend), assets, stock (other than a dividend payable in Common Shares) or subscription rights, options or warrants (excluding those referred to in Section 11(b) hereof), the Purchase Price to be in effect after such record date shall be determined by multiplying the Purchase Price in effect immediately prior to such record date by a fraction, the numerator of which shall be the current per share market price of the Common Shares (as determined pursuant to Section 11(d) hereof) on such record date or, if earlier, the date on which Common Shares begin to trade on an ex-dividend or when-issued basis for such distribution, less the fair market value (as determined in good faith by the Board of Directors of the Company, whose determination shall be described in a statement filed with the Rights Agent) of the portion of the evidences of indebtedness, cash, assets or stock so to be distributed or of such subscription rights, options or warrants applicable to one Common Share, and the denominator of which shall be such current per share market price of the Common Shares. Such adjustments shall be made successively whenever such a record date is fixed; and in the event that such distribution is not so made, the Purchase Price shall again be adjusted to be the Purchase Price which would then be in effect if such record date had not been fixed.
- (d) For the purpose of any computation hereunder, the "current per share market price" of Common Shares on any date shall be deemed to be the average of the daily closing prices per share of such Common Shares for the 30 consecutive Trading Days immediately prior to such date; provided, however, that in the event that the current per share market price of the Common Shares is determined during a period following the announcement by the issuer of such Common Shares of (i) a dividend or distribution on such Common Shares payable in such Common Shares or securities convertible into such Common Shares (other than the Rights) or (ii) any subdivision, combination or reclassification of such Common Shares, and prior to the expiration of 30 Trading Days after the ex-dividend date for such dividend or distribution, or the record date for such subdivision, combination or reclassification, then, and in each such case, the current per share market price shall be appropriately adjusted to take into account ex-dividend trading or to reflect the current per share market price per Common Share equivalent. The

closing price for each day shall be the last sale price, regular way, or, in case no such sale takes place on such day, the average of the closing bid and asked prices, regular way, in either case as reported in the principal consolidated transaction reporting system with respect to securities listed or admitted to trading on the NYSE or, if the Common Shares are not listed or admitted to trading on the NYSE, as reported in the principal consolidated transaction reporting system with respect to securities listed on the principal national securities exchange on which the Common Shares are listed or admitted to trading or, if the Common Shares are not listed or admitted to trading on any national securities exchange, the last quoted price or, if not so quoted, the average of the high bid and low asked prices in the over-the-counter market, as reported by NASDAQ or such other system then in use, or, if on any such date the Common Shares are not quoted by any such organization, the average of the closing bid and asked prices as furnished by a professional market maker making a market in the Common Shares selected by the Board of Directors of the Company. If the Common Shares are not publicly held or not so listed or traded, or not the subject of available bid and asked quotes, "current per share market price" shall mean the fair value per share as determined in good faith by the Board of Directors of the Company, whose determination shall be described in a statement filed with the Rights Agent.

- (e) Except as set forth below, no adjustment in the Purchase Price shall be required unless such adjustment would require an increase or decrease of at least 1% in such price; provided, however, that any adjustments which by reason of this Section 11(e) are not required to be made shall be carried forward and taken into account in any subsequent adjustment. All calculations under this Section 11 shall be made to the nearest cent or to the nearest one ten thousandth of a Common Share or other security, as the case may be. Notwithstanding the first sentence of this Section 11(e), any adjustment required by this Section 11 shall be made no later than the earlier of (i) three years from the date of the transaction which requires such adjustment and (ii) the Expiration Date.
- (f) If as a result of an adjustment made pursuant to Section 11(a) hereof, the holder of any Right thereafter exercised shall become entitled to receive any securities of the Company other than Common Shares, thereafter the number of such other securities so receivable upon exercise of any Right shall be subject to adjustment from time to time in a manner and on terms as nearly equivalent as practicable to the provisions with respect to the Common Shares contained in this Section 11, and the provisions of Sections 7, 9, 10 and 13 hereof with respect to the Common Shares shall apply on like terms to any such other securities.
- (g) All Rights originally issued by the Company subsequent to any adjustment made to the Purchase Price hereunder shall evidence the right to purchase, at the adjusted Purchase Price, the number of Common Shares issuable from time to time hereunder upon exercise of the Rights, all subject to further adjustment as provided herein.
- (h) Unless the Company shall have exercised its election as provided in Section 11(i) hereof, upon each adjustment of the Purchase Price as a result of the calculations made in Section 11(b) and Section 11(c) hereof made with respect to a distribution of subscription rights, options or warrants applicable to Common Shares, each Right outstanding immediately prior to the making of such adjustment shall thereafter evidence the right to purchase, at the adjusted Purchase Price, that number of Common Shares (calculated to the nearest one-thousandth of a

Common Share) obtained by (i) multiplying (x) the number of Common Shares issuable upon exercise of a Right immediately prior to this adjustment by (y) the Purchase Price in effect immediately prior to such adjustment of the Purchase Price, and (ii) dividing the product so obtained by the Purchase Price in effect immediately after such adjustment of the Purchase Price.

- (i) The Company may elect, on or after the date of any adjustment of the Purchase Price, to adjust the number of Rights in substitution for any adjustment in the number of Common Shares issuable upon the exercise of a Right. Each of the Rights outstanding after such adjustment of the number of Rights shall be exercisable for the number of Common Shares for which a Right was exercisable immediately prior to such adjustment. Each Right held of record prior to such adjustment of the number of Rights shall become that number of Rights (calculated to the nearest one-thousandth) obtained by dividing the Purchase Price in effect immediately prior to adjustment of the Purchase Price by the Purchase Price in effect immediately after adjustment of the Purchase Price. The Company shall make a public announcement of its election to adjust the number of Rights, indicating the record date for the adjustment, and, if known at the time, the amount of the adjustment to be made. This record date may be the date on which the Purchase Price is adjusted or any day thereafter, but, if the Right Certificates have been issued, shall be at least 10 calendar days later than the date of the public announcement. If Right Certificates have been issued, upon each adjustment of the number of Rights pursuant to this Section 11(i), the Company shall, as promptly as practicable, cause to be distributed to holders of record of Right Certificates on such record date Right Certificates evidencing, subject to the provisions of Section 14 hereof, the additional Rights to which such holders shall be entitled as a result of such adjustment, or, at the option of the Company, shall cause to be distributed to such holders of record in substitution and replacement for the Right Certificates held by such holders prior to the date of adjustment, and upon surrender thereof if required by the Company, new Right Certificates evidencing all the Rights to which such holders shall be entitled after such adjustment. Right Certificates so to be distributed shall be issued, executed and countersigned in the manner provided for herein (and may bear, at the option of the Company, the adjusted Purchase Price) and shall be registered in the names of the holders of record of Right Certificates on the record date specified in the public announcement.
- (j) Irrespective of any adjustment or change in the Purchase Price or the number or kind of securities issuable upon the exercise of the Rights, the Right Certificates theretofore and thereafter issued may continue to express the Purchase Price and the number and kind of securities which were expressed in the initial Right Certificate issued hereunder.
- (k) Before taking any action that would cause an adjustment reducing the Purchase Price below the then par value, if any, of the Common Shares or other securities issuable upon exercise of the Rights, the Company shall take any corporate action which may, in the opinion of its counsel, be necessary in order that the Company may validly and legally issue fully paid and nonassessable Common Shares or such other securities at such adjusted Purchase Price.
- (1) In any case in which this Section 11 shall require that an adjustment in the Purchase Price be made effective as of a record date for a specified event, the Company may elect to defer until the occurrence of such event the issuance to the holder of any Right exercised after such record date the number of Common Shares or other securities of the Company, if any,

issuable upon such exercise over and above the number of Common Shares or other securities of the Company, if any, issuable upon such exercise on the basis of the Purchase Price in effect prior to such adjustment; provided, however, that the Company shall deliver to such holder a due bill or other appropriate instrument evidencing such holder's right to receive such additional Common Shares or other securities upon the occurrence of the event requiring such adjustment.

(m) Notwithstanding anything in this Agreement to the contrary, the Company shall be entitled to make such reductions in the Purchase Price, in addition to those adjustments expressly required by this Section 11, as and to the extent that in its good faith judgment the Board of Directors of the Company shall determine to be advisable in order that any (i) consolidation or subdivision of the Common Shares, (ii) issuance wholly for cash of Common Shares at less than the current per share market price therefor, (iii) issuance wholly for cash of Common Shares or securities which by their terms are convertible into or exchangeable for Common Shares, (iv) stock dividends, or (v) issuance of rights, options or warrants referred to in this Section 11, hereafter made by the Company to holders of its Common Shares shall not be taxable to such shareholders.

Section 12. Certificate of Adjusted Purchase Price or Number of Securities. Whenever an adjustment is made as provided in Section 11 or Section 13 hereof, the Company shall promptly (a) prepare a certificate setting forth such adjustment and a brief statement of the facts accounting for such adjustment, (b) file with the Rights Agent and with each transfer agent for the Common Shares a copy of such certificate, and (c) if such adjustment is made after the Distribution Date, mail a brief summary of such adjustment to each holder of a Right Certificate in accordance with Section 25 hereof.

Section 13. Consolidation, Merger or Sale or Transfer of Assets or Earning Power.

- (a) In the event that, following the Share Acquisition Date, directly or indirectly:
 - (i) the Company shall consolidate with, or merge with or into, any other Person and the Company shall not be the continuing or surviving corporation of such consolidation or merger; or
 - (ii) any Person shall consolidate with the Company, or merge with or into the Company and the Company shall be the continuing or surviving corporation of such merger or consolidation and, in connection with such merger or consolidation, all or part of the Common Shares shall be changed into or exchanged for stock or other securities of any other Person or cash or any other property; or
 - (iii) the Company shall be a party to any statutory share exchange with any other Person; or

(iv) the Company shall sell or otherwise transfer (or one or more of its Subsidiaries shall sell or otherwise transfer), in one or more transactions, assets or earning power (including, without limitation, securities creating any obligation on the part of the Company and/or any of its Subsidiaries) representing in the aggregate more than 50% of the assets or earning power of the Company and its Subsidiaries (taken as a whole) to any Person or Persons; then, and in each such case, proper provision shall be made so that (A) each holder of a Right (except as otherwise provided herein) shall thereafter have the right to receive, upon the exercise thereof in accordance with the terms of this Agreement at an exercise price per Right equal to the product of the then-current Purchase Price multiplied by the number of Common Shares for which a Right was exercisable immediately prior to the first occurrence of a Triggering Event, such number of validly authorized and issued, fully paid, nonassessable and freely tradable Common Shares of the Issuer, free and clear of any liens, encumbrances and other adverse claims and not subject to any rights of call or first refusal, as shall be equal to the result obtained by (x)multiplying the then-current Purchase Price by the number of Common Shares for which a Right is exercisable immediately prior to the first occurrence of a Triggering Event and dividing that product by (y) 50% of the current per share market price of the Common Shares of the Issuer (determined pursuant to Section 11(d) hereof), on the date of consummation of such Flip-over Event; (B) the Issuer shall thereafter be liable for, and shall assume, by virtue of the consummation of such Flip-over Event, all the obligations and duties of the Company pursuant to this Agreement; (C) the term "Company" shall thereafter be deemed to refer to the Issuer; and (D) the Issuer shall take such steps (including, without limitation, the reservation of a sufficient number of its Common Shares to permit the exercise of all outstanding Rights) in connection with such consummation as may be necessary to assure that the provisions hereof shall thereafter be applicable, as nearly as reasonably may be possible, in relation to its Common Shares thereafter deliverable upon the exercise of the Rights.

(b) For purposes of this Section 13, "Issuer" shall mean (i) in the case of any Flip-over Event described in Sections 13(a)(i) or (ii) above, the Person that is the continuing, surviving, resulting or acquiring Person (including the Company as the continuing or surviving corporation of a transaction described in Section 13(a)(ii) above), (ii) in the case of any Flip-over Event described in Section 13(a)(iii) above, the Person acquiring the securities of the shareholders of the Company in such exchange and (iii) in the case of any Flip-over Event described in Section 13(a)(iv) above, the Person that is the party receiving the greatest portion of the assets or earning power (including, without limitation, securities creating any obligation on the part of the Company and/or any of its Subsidiaries) transferred pursuant to such transaction or transactions; provided, however, that, in any such case, (A) if (1) no class of equity security of such Person is, at the time of such merger, consolidation or transaction and has been continuously over the preceding 12-month period, registered pursuant to Section 12 of the Exchange Act, and (2) such Person is a Subsidiary, directly or indirectly, of another Person, a class of equity security of which is and has been so registered, the term "Issuer" shall mean such other Person; and (B) in case such Person is a Subsidiary, directly or indirectly, of more than

one Person, a class of equity security of two or more of which are and have been so registered, the term Issuer shall mean whichever of such Persons is the issuer of the equity security having the greatest aggregate market value. Notwithstanding the foregoing, if the Issuer in any of the Flip-over Events listed above is not a corporation or other legal entity having outstanding equity securities, then, and in each such case, (x) if the Issuer is directly or indirectly wholly owned by a corporation or other legal entity having outstanding equity securities, then all references to Common Shares of the Issuer shall be deemed to be references to the Common Shares of the corporation or other legal entity having outstanding equity securities which ultimately controls the Issuer, and (y) if there is no such corporation or other legal entity having outstanding equity securities, (I) proper provision shall be made so that the Issuer shall create or otherwise make available for purposes of the exercise of the Rights in accordance with the terms of this Agreement, a kind or kinds of security or securities having a fair market value at least equal to the economic value of the Common Shares which each holder of a Right would have been entitled to receive if the Issuer had been a corporation or other legal entity having outstanding equity securities; and (II) all other provisions of this Agreement shall apply to the issuer of such securities as if such securities were Common Shares.

- (c) The Company shall not consummate any Flip-over Event, unless the Issuer shall have a sufficient number of authorized Common Shares (or other securities as contemplated in Section 13(b) above) which have not been issued or reserved for issuance to permit the exercise in full of the Rights in accordance with this Section 13 and unless prior to such consummation the Company and the Issuer shall have executed and delivered to the Rights Agent a supplemental agreement providing for the terms set forth in subsections (a) and (b) of this Section 13 and further providing that as promptly as practicable after the consummation of any Flip-over Event, the Issuer shall:
- (i) prepare and file a registration statement under the Securities Act, with respect to the Rights and the securities issuable upon exercise of the Rights on an appropriate form, and shall use its best efforts to cause such registration statement to (A) become effective as soon as practicable after such filing and (B) remain effective (with a prospectus at all times meeting the requirements of the Securities Act) until the Expiration Date;
- (ii) take all such action as may be appropriate under, or to ensure compliance with, the securities or "blue sky" laws of the various states in connection with the exercisability of the Rights; and
- (iii) deliver to holders of the Rights historical financial statements for the Issuer and each of its Affiliates which comply in all respects with the requirements for registration on Form 10 (or any successor form) under the Exchange Act.
- (d) The provisions of this Section 13 shall similarly apply to successive mergers or consolidations or sales or other transfers. In the event that a Flip-over Event occurs at any time after the occurrence of a Flip-in Event, the Rights which have not

theretofore been exercised shall thereafter become exercisable in the manner described in Section 13(a) hereof.

Section 14. Fractional Rights and Fractional Securities.

- (a) The Company shall not be required to issue fractions of Right or to distribute Right Certificates which evidence fractional Rights. In lieu of such fractional Rights, the Company shall pay as promptly as practicable to the registered holders of the Right Certificates with regard to which such fractional Rights otherwise would be issuable, an amount in cash equal to the same fraction of the current market value of a whole Right. For the purposes of this Section 14(a), the current market value of a whole Right shall be the closing price of the Rights for the Trading Day immediately prior to the date on which such fractional Rights otherwise would have been issuable. The closing price for any day shall be the last sale price, regular way, or, in case no such sale takes place on such day, the average of the closing bid and asked prices, regular way, in either case as reported in the principal consolidated transaction reporting system with respect to securities listed or admitted to trading on the NYSE or, if the Rights are not listed or admitted to trading on the NYSE, as reported in the principal consolidated transaction reporting system with respect to securities listed on the principal national securities exchange on which the Rights are listed or admitted to trading or, if the Rights are not listed or admitted to trading on any national securities exchange, the last quoted price or, if not so quoted, the average of the high bid and low asked prices in the over-the-counter market, as reported by NASDAQ or such other system then in use or, if on any such date the Rights are not quoted by any such organization, the average of the closing bid and asked prices as furnished by a professional market maker making a market in the Rights selected by the Board of Directors of the Company. If on any such date no such market maker is making a market in the Rights, the fair value of the Rights on such date as determined in good faith by the Board of Directors of the Company shall be used.
- (b) The Company shall not be required to issue fractions of Common Shares or other securities issuable upon exercise or exchange of the Rights or to distribute certificates which evidence any such fractional securities. In lieu of issuing any such fractional securities, the Company may pay to any Person to whom or which such fractional securities would otherwise be issuable an amount in cash equal to the same fraction of the current market value of one such security. For purposes of this Section 14(b), the current market value of a Common Share or other security issuable upon the exercise or exchange of Rights shall be the closing price thereof (as determined in the same manner as set forth for Common Shares in the second sentence of Section 11(d) hereof) for the Trading Day immediately prior to the date of such exercise or exchange; provided, however, that if neither the Common Shares nor any such other securities are publicly held or listed or admitted to trading on any national securities exchange, or the subject of available bid and asked quotes, the current market value of one Common Share or such other security shall be determined in good faith by the Board of Directors of the Company.

Section 15. Rights of Action. All rights of action in respect of this Agreement, excepting the rights of action given to the Rights Agent under Section 18 hereof, are vested in the respective registered holders of the Right Certificates (and, prior to the Distribution Date, the registered holders of the Common Shares); and any registered holder of any Right Certificate (or, prior to the Distribution Date, of the Common Shares) without the consent of the Rights Agent or of the holder of any other Right Certificate (or, prior to the Distribution Date, of the holder of any Common Shares), may in his own behalf and for his own benefit enforce, and may institute and maintain any suit, action or proceeding against the Company to enforce, or otherwise act in respect of, his right to exercise Rights evidenced by such Right Certificate or Common Share certificate in the manner provided in such Right Certificate and in this Agreement. Without limiting the foregoing or any remedies available to the holders of Rights, it is specifically acknowledged that the holders of Rights would not have an adequate remedy at law for any breach of this Agreement and shall be entitled to specific performance of the obligations under this Agreement, and injunctive relief against actual or threatened violations of the obligations of any Person subject to this Agreement.

Section 16. Agreement of Rights Holders. Every holder of a Right by accepting the same consents and agrees with the Company and the Rights Agent and with every other holder of a Right that:

- (a) Prior to the Distribution Date, the Rights shall be transferable only in connection with the transfer of the Common Shares;
- (b) After the Distribution Date, the Right Certificates are transferable only on the registry books of the Rights Agent if surrendered at the principal office of the Rights Agent designated for such purpose, duly endorsed or accompanied by a proper instrument of transfer;
- (c) The Company and the Rights Agent may deem and treat the person in whose name the Right Certificate (or, prior to the Distribution Date, the associated Common Share certificate) is registered as the absolute owner thereof and of the Rights evidenced thereby (notwithstanding any notations of ownership or writing on the Right Certificate or the associated Common Share certificate made by anyone other than the Company or the Rights Agent) for all purposes whatsoever, and neither the Company nor the Rights Agent shall be affected by any notice to the contrary;
- (d) Such holder expressly waives any right to receive any fractional Rights and any fractional securities upon exercise or exchange of a Right, except as otherwise provided in Section 14 hereof; and
- (e) Notwithstanding anything in this Agreement to the contrary, neither the Company nor the Rights Agent shall have any liability to any holder of a Right or other Person as a result of its inability to perform any of its obligations under this Agreement by reason of any preliminary or permanent injunction or other order, decree or ruling issued by a court of competent jurisdiction or by a governmental, regulatory or

administrative agency or commission, or any statute, rule, regulation or executive order promulgated or enacted by any governmental authority, prohibiting or otherwise restraining performance of such obligation; provided, however, that the Company shall use its best efforts to have any such order, decree or ruling lifted or otherwise overturned as soon as possible.

Section 17. Right Certificate Holder Not Deemed a Shareholder. No holder, as such, of any Right Certificate shall be entitled to vote, receive dividends or be deemed for any purpose the holder of the Common Shares or any other securities of the Company which may at any time be issuable upon the exercise of the Rights represented thereby, nor shall anything contained herein or in any Right Certificate be construed to confer upon the holder of any Right Certificate, as such, any of the rights of a shareholder of the Company or any right to vote for the election of directors or upon any matter submitted to shareholders at any meeting thereof, or to give or withhold consent to any corporate action, or to receive notice of meetings or other actions affecting shareholders (except as provided in Section 24 hereof), or to receive dividends or subscription rights, or otherwise, until the Right or Rights evidenced by such Right Certificate shall have been exercised in accordance with the provisions of this Agreement or exchanged pursuant to the provisions of Section 27 hereof.

Section 18. Concerning the Rights Agent.

- (a) The Company agrees to pay to the Rights Agent reasonable compensation for all services rendered by it hereunder and, from time to time, on demand of the Rights Agent, its reasonable expenses and counsel fees and other disbursements incurred in the administration and execution of this Agreement and the exercise and performance of its duties hereunder. The Company also agrees to indemnify the Rights Agent for, and to hold it harmless against, any loss, liability, suit, action, proceeding or expense, incurred without gross negligence, bad faith or willful misconduct on the part of the Rights Agent, for anything done or omitted by the Rights Agent in connection with the acceptance and administration of this Agreement, including the costs and expenses of defending against any claim of liability arising therefrom, directly or indirectly.
- (b) The Rights Agent shall be protected and shall incur no liability for or in respect of any action taken, suffered or omitted by it in connection with its administration of this Agreement in reliance upon any Right Certificate or certificate evidencing Common Shares or other securities of the Company, instrument of assignment or transfer, power of attorney, endorsement, affidavit, letter, notice, direction, consent, certificate, statement or other paper or document believed by it to be genuine and to be signed, executed and, where necessary, verified or acknowledged, by the proper Person or Persons.

Section 19. Merger or Consolidation or Change of Name of Rights Agent.

(a) Any corporation into which the Rights Agent or any successor Rights Agent may be merged or with which it may be consolidated, or any corporation resulting

from any merger or consolidation to which the Rights Agent or any successor Rights Agent shall be a party, or any corporation succeeding to the corporate trust business of the Rights Agent or any successor Rights Agent, shall be the successor to the Rights Agent under this Agreement without the execution or filing of any paper or any further act on the part of any of the parties hereto, provided that such corporation would be eligible for appointment as a successor Rights Agent under the provisions of Section 21 hereof. In case at the time such successor Rights Agent shall succeed to the agency created by this Agreement, any of the Right Certificates shall have been countersigned but not delivered, any such successor Rights Agent may adopt the countersignature of the predecessor Rights Agent and deliver such Right Certificates so countersigned; and in case at that time any of the Right Certificates shall not have been countersigned, any successor Rights Agent may countersign such Right Certificates either in the name of the predecessor Rights Agent or in the name of the successor Rights Agent; and in all such cases such Right Certificates shall have the full force provided in the Right Certificates and in this Agreement.

(b) In case at any time the name of the Rights Agent shall be changed and at such time any of the Right Certificates shall have been countersigned but not delivered, the Rights Agent may adopt the countersignature under its prior name and deliver Right Certificates so countersigned; and in case at that time any of the Right Certificates shall not have been countersigned, the Rights Agent may countersign such Right Certificates either in its prior name or in its changed name; and in all such cases such Right Certificates shall have the full force provided in the Right Certificates and in this Agreement.

Section 20. Duties of Rights Agent. The Rights Agent undertakes the duties and obligations imposed by this Agreement upon the following terms and conditions, by all of which the Company and the holders of Right Certificates, by their acceptance thereof, shall be bound:

- (a) The Rights Agent may consult with legal counsel (who may be legal counsel for the Company), and the opinion of such counsel shall be full and complete authorization and protection to the Rights Agent as to any action taken or omitted by it in good faith and in accordance with such opinion.
- (b) Whenever in the performance of its duties under this Agreement the Rights Agent shall deem it necessary or desirable that any fact or matter be proved or established by the Company prior to taking or suffering any action hereunder, such fact or matter (unless other evidence in respect thereof be herein specifically prescribed) may be deemed to be conclusively proved and established by a certificate signed by any one of the Chairman of the Board, the President or any Vice President of the Company and delivered to the Rights Agent; and such certificate shall be full authorization to the Rights Agent for any action taken or suffered in good faith by it under the provisions of this Agreement in reliance upon such certificate.

- (c) The Rights Agent shall be liable hereunder only for its own gross negligence, bad faith or willful misconduct.
- (d) The Rights Agent shall not be liable for or by reason of any of the statements of fact or recitals contained in this Agreement or in the Right Certificates (except its countersignature thereof) or be required to verify the same, but all such statements and recitals are and shall be deemed to have been made by the Company only.
- (e) The Rights Agent shall not be under any responsibility in respect of the validity of this Agreement or the execution and delivery hereof (except the due execution and delivery hereof by the Rights Agent) or in respect of the validity or execution of any Right Certificate (except its countersignature thereof); nor shall it be responsible for any breach by the Company of any covenant or condition contained in this Agreement or in any Right Certificate; nor shall it be responsible for any adjustment required under the provisions of Section 11 or Section 13 hereof (including any adjustment which results in Rights becoming void) or responsible for the manner, method or amount of any such adjustment or the ascertaining of the existence of facts that would require any such adjustment (except with respect to the exercise of Rights evidenced by Right Certificates after actual notice of any such adjustment); nor shall it by any act hereunder be deemed to make any representation or warranty as to the authorization or reservation of any shares of stock or other securities to be issued pursuant to this Agreement or any Right Certificate or as to whether any shares of stock or other securities will, when issued, be validly authorized and issued, fully paid and nonassessable.
- (f) The Company agrees that it will perform, execute, acknowledge and deliver or cause to be performed, executed, acknowledged and delivered all such further and other acts, instruments and assurances as may reasonably be required by the Rights Agent for the carrying out or performing by the Rights Agent of the provisions of this Agreement.
- (g) The Rights Agent is hereby authorized and directed to accept instructions with respect to the performance of its duties hereunder from any one of the Chairman of the Board, the President or any Vice President of the Company, and to apply to such officers for advice or instructions in connection with its duties, and it shall not be liable for any action taken or suffered to be taken by it in good faith in accordance with instructions of any such officer.
- (h) The Rights Agent and any shareholder, director, officer or employee of the Rights Agent may buy, sell or deal in any of the Rights or other securities of the Company or become pecuniarily interested in any transaction in which the Company may be interested, or contract with or lend money to the Company or otherwise act as fully and freely as though it were not Rights Agent under this Agreement. Nothing herein shall preclude the Rights Agent from acting in any other capacity for the Company or for any other legal entity.

- (i) The Rights Agent may execute and exercise any of the rights or powers hereby vested in it or perform any duty hereunder either itself or by or through its attorneys or agents, and the Rights Agent shall not be answerable or accountable for any act, default, neglect or misconduct of any such attorneys or agents or for any loss to the Company resulting from any such act, default, neglect or misconduct, provided reasonable care was exercised in the selection and continued employment thereof. The Rights Agent shall not be under any duty or responsibility to insure compliance with any applicable federal or state securities laws in connection with the issuance, transfer or exchange of Right Certificates.
- (j) If, with respect to any Right Certificate surrendered to the Rights Agent for exercise, transfer, split up, combination or exchange, the certificate attached to the form of assignment or form of election to purchase, as the case may be, has either not been completed or indicates an affirmative response to clause 1 or 2 thereof, the Rights Agent shall not take any further action with respect to such requested exercise, transfer, split up, combination or exchange, without first consulting with the Company.

Section 21. Change of Rights Agent. The Rights Agent or any successor Rights Agent may resign and be discharged from its duties under this Agreement upon 30 calendar days' notice in writing mailed to the Company and to each transfer agent of the Common Shares by registered or certified mail, and to the holders of the Right Certificates by first-class mail. The Company may remove the Rights Agent or any successor Rights Agent upon 30 calendar days' notice in writing, mailed to the Rights Agent or successor Rights Agent, as the case may be, and to each transfer agent of the Common Shares by registered or certified mail, and to the holders of the Right Certificates by first-class mail. If the Rights Agent shall resign or be removed or shall otherwise become incapable of acting, the Company shall appoint a successor to the Rights Agent. If the Company shall fail to make such appointment within a period of 30 calendar days after giving notice of such removal or after it has been notified in writing of such resignation or incapacity by the resigning or incapacitated Rights Agent or by the holder of a Right Certificate (who shall, with such notice, submit his Right Certificate for inspection by the Company), then the registered holder of any Right Certificate may apply to any court of competent jurisdiction for the appointment of a new Rights Agent. Any successor Rights Agent, whether appointed by the Company or by such a court, shall be a corporation organized and doing business under the laws of the United States or of the State of Georgia or the State of New York (or of any other state of the United States so long as such corporation is authorized to do business as a banking institution in the State of Georgia or the State of New York), in good standing, having a principal office in the State of Georgia or the State of New York, which is authorized under such laws to exercise corporate trust powers and is subject to supervision or examination by federal or state authority and which has at the time of its appointment as Rights Agent a combined capital and surplus of at least \$50 million. After appointment, the successor Rights Agent shall be vested with the same powers, rights, duties and responsibilities as if it had been originally named as Rights Agent without further act or deed; but the predecessor Rights Agent shall deliver and transfer to the successor Rights Agent any property at the time held by it hereunder, and execute and deliver any further assurance, conveyance, act or

deed necessary for the purpose. Not later than the effective date of any such appointment, the Company shall file notice thereof in writing with the predecessor Rights Agent and each transfer agent of the Common Shares, and mail a notice thereof in writing to the registered holders of the Right Certificates. Failure to give any notice provided for in this Section 21, however, or any defect therein, shall not affect the legality or validity of the resignation or removal of the Rights Agent or the appointment of the successor Rights Agent, as the case may be.

Section 22. Issuance of New Right Certificates. Notwithstanding any of the provisions of this Agreement or of the Rights to the contrary, the Company may, at its option, issue new Right Certificates evidencing Rights in such form as may be approved by its Board of Directors to reflect any adjustment or change in the Purchase Price per share and the number or kind of securities issuable upon exercise of the Rights made in accordance with the provisions of this Agreement. In addition, in connection with the issuance or sale by the Company of Common Shares following the Distribution Date and prior to the Expiration Date, the Company (a) shall, with respect to Common Shares so issued or sold pursuant to the exercise or conversion of securities issued prior to the Distribution Date which are exercisable for, or convertible into, Common Shares, and (b) may, in any other case, if deemed necessary, appropriate or desirable by the Board of Directors of the Company, issue Right Certificates representing an equivalent number of Rights as would have been issued in respect of such Common Shares if they had been issued or sold prior to the Distribution Date, as appropriately adjusted as provided herein as if they had been so issued or sold; provided, however, that (i) no such Right Certificate shall be issued if, and to the extent that, in its good faith judgment the Board of Directors of the Company shall have determined that the issuance of such Right Certificate could have a material adverse tax consequence to the Company or to the Person to whom or which such Right Certificate otherwise would be issued, and (ii) no such Right Certificate shall be issued if, and to the extent that, appropriate adjustment otherwise shall have been made in lieu of the issuance thereof.

Section 23. Redemption.

- (a) The Board of Directors of the Company may, at its option, redeem all but not less than all of the then-outstanding Rights at the Redemption Price at any time prior to the Close of Business on the later of (i) the Distribution Date and (ii) the Share Acquisition Date, or on such later date as the Board of Directors may designate prior to such time as the Rights are no longer redeemable hereunder.
- (b) Immediately upon the action of the Board of Directors of the Company ordering the redemption of the Rights, and without any further action and without any notice, the right to exercise the Rights shall terminate and the only right thereafter of the holders of Rights shall be to receive the Redemption Price. Promptly after the action of its Board of Directors ordering the redemption of the Rights, the Company shall publicly announce such action, and within 10 calendar days thereafter, the Company shall give notice of such redemption to the holders of the then-outstanding Rights by mailing such notice to all such holders at their last addresses as they appear upon the registry books of

the Company; provided, however, that the failure to give, or any defect in, any such notice shall not affect the validity of the redemption of the Rights. Any notice which is mailed in the manner herein provided shall be deemed given, whether or not the holder receives the notice. The notice of redemption mailed to the holders of Rights shall state the method by which the payment of the Redemption Price will be made. The Company may, at its option, pay the Redemption Price in cash, Common Shares (based upon the current per share market price of the Common Shares (determined pursuant to Section 11(d) hereof) at the time of redemption) or any other form of consideration deemed appropriate by the Board of Directors of the Company (based upon the fair market value of such other consideration, determined by the Board of Directors of the Company in good faith) or any combination thereof.

- (c) At any time following the Share Acquisition Date, the Board of Directors of the Company may relinquish the right to redeem the Rights under this Section 23 by duly adopting a resolution to that effect. Immediately upon adoption of such resolution, the rights of the Board of Directors of the Company to redeem the Rights shall terminate without further action and without any notice. Promptly after adoption of such a resolution, the Company shall publicly announce such action; provided, however, that the failure to give, or any defect in, any such notice shall not affect the validity of the action of the Board of Directors of the Company.
- (d) If, following the occurrence of a Share Acquisition Date and following the expiration of the right of redemption hereunder, (i) a Person who is an Acquiring Person shall have transferred or otherwise disposed of a number of Common Shares in one transaction or series of transactions, not directly or indirectly involving the Company or any of its Subsidiaries, which did not result in the occurrence of a Triggering Event such that such Person is thereafter the Beneficial Owner of 10% or less of the outstanding Common Shares, and (ii) there are no other Persons immediately following the occurrence of the event described in clause (i) who are Acquiring Persons, then the right of redemption hereunder shall be reinstated and shall thereafter be exercisable subject to the provisions of this Section 23.

Section 24. Notice of Certain Events.

(a) In case, after the Distribution Date, the Company shall propose (i) to pay any dividend payable in stock of any class to the holders of Common Shares or to make any other distribution to the holders of Common Shares (other than a regular periodic cash dividend), (ii) to offer to the holders of Common Shares rights, options or warrants to subscribe for or to purchase any additional Common Shares or shares of stock of any class or any other securities, rights or options, (iii) to effect any reclassification of its Common Shares (other than a reclassification involving only the subdivision of outstanding Common Shares), (iv) to effect any consolidation or merger into or with, or to effect any sale or other transfer (or to permit one or more of its Subsidiaries to effect any sale or other transfer), in one or more transactions, of assets or earning power (including, without limitation, securities creating any obligation on the part of the Company and/or any of its Subsidiaries) representing more than 50% of the assets and

29

earning power of the Company and its Subsidiaries, taken as a whole, to any other Person or Persons, or (v) to effect the liquidation, dissolution or winding up of the Company, then, in each such case, the Company shall give to each holder of a Right Certificate, in accordance with Section 25 hereof, a notice of such proposed action, which shall specify the record date for the purposes of such stock dividend, distribution or offering of rights, options or warrants, or the date on which such reclassification, consolidation, merger, sale, transfer, liquidation, dissolution, or winding up is to take place and the date of participation therein by the holders of the Common Shares, if any such date is to be fixed, and such notice shall be so given, in the case of any action covered by clause (i) or (ii) above, at least 10 calendar days prior to the record date for determining holders of the Common Shares for purposes of such action, and, in the case of any such other action, at least 10 calendar days prior to the date of the taking of such proposed action or the date of participation therein by the holders of the Common Shares, whichever shall be the earlier.

(b) In case any Triggering Event shall occur, then, in any such case, the Company shall as soon as practicable thereafter give to the Rights Agent and each holder of a Right Certificate, in accordance with Section 25 hereof, a notice of the occurrence of such event, which shall specify the event and the consequences of the event to holders of Rights.

Section 25. Notices.

(a) Notices or demands authorized by this Agreement to be given or made by the Rights Agent or by the holder of any Right Certificate to or on the Company shall be sufficiently given or made if sent by first-class mail, postage prepaid, addressed (until another address is filed in writing with the Rights Agent) as follows:

Certegy 1	.nc.	
Attention	n: Corporate	Secretary

(b) Subject to the provisions of Section 21 hereof, any notice or demand authorized by this Agreement to be given or made by the Company or by the holder of any Right Certificate to or on the Rights Agent shall be sufficiently given or made if sent by first-class mail, postage prepaid, addressed (until another address is filed in writing with the Company) as follows:

SunTrust Bank Stock Transfer Trust Department P.O. Box 4625 Atlanta, Georgia 30302 Attention: Department Manager (c) Notices or demands authorized by this Agreement to be given or made by the Company or the Rights Agent to the holder of any Right Certificate (or, if prior to the Distribution Date, to the holder of any certificate evidencing Common Shares) shall be sufficiently given or made if sent by first-class mail, postage prepaid, addressed to such holder at the address of such holder as shown on the registry books of the Company.

Section 26. Supplements and Amendments. Prior to the Distribution Date and subject to the last sentence of this Section 26, the Company may in its sole and absolute discretion and the Rights Agent shall, if the Company so directs, supplement or amend any provision of this Agreement without the approval of any holders of Rights or shares of the Company. From and after the Distribution Date and subject to the last sentence of this Section 26, the Company may in its sole and absolute discretion and the Rights Agent shall, if the Company so directs, supplement or amend this Agreement without the approval of any holders of Right Certificates in order (i) to cure any ambiguity, (ii) to correct or supplement any provision contained herein which may be defective or inconsistent with any other provisions herein, (iii) to shorten or lengthen any time period hereunder, or (iv) to supplement or amend the provisions hereunder in any manner which the Company may deem desirable, including, without limitation, the addition of other events requiring adjustment to the Rights under Sections 11 or 13 hereof or procedures relating to the redemption of the Rights, which supplement or amendment shall not, in the good faith determination of the Board of Directors of the Company, adversely affect the interests of the holders of Right Certificates (other than an Acquiring Person or an Affiliate or Associate of an Acquiring Person). Upon the delivery of a certificate from an officer of the Company that states that the proposed supplement or amendment is in compliance with the terms of this Section 26, the Rights Agent shall execute such supplement or amendment; provided, however, that the failure or refusal of the Rights Agent to execute such supplement or amendment shall not affect the validity of any supplement or amendment adopted by the Company, any of which shall be effective in accordance with the terms thereof. Notwithstanding anything in this Agreement to the contrary, no supplement or amendment shall be made at such time as the Rights are not then redeemable which decreases the stated Redemption Price or the period of time remaining until the Final Expiration Date or which modifies a time period relating to when the Rights may be redeemed.

Section 27. Exchange.

(a) The Board of Directors of the Company may, at its option, at any time after the later of the Distribution Date and the first occurrence of a Triggering Event, exchange all or part of the then-outstanding and exercisable Rights (which shall not include Rights that have become void pursuant to the provisions of Section 11(a)(ii) hereof) for Common Shares at an exchange ratio of one Common Share per Right, appropriately adjusted to reflect any stock split, stock dividend or similar transaction occurring after the date hereof (such exchange ratio being hereinafter referred to as the "Exchange Ratio"). Notwithstanding the foregoing, the Board of Directors shall not be empowered to effect such exchange at any time after any Person (other than the Company, any Subsidiary of the Company, any employee benefit plan of the Company or

31

any such Subsidiary, or any entity holding Common Shares for or pursuant to the terms of any such plan), together with all Affiliates and Associates of such Person, becomes the Beneficial Owner of 50% or more of the Common Shares then outstanding.

- (b) Immediately upon the action of the Board of Directors of the Company ordering the exchange of any Rights pursuant to Section 27(a) hereof, and without any further action and without any notice, the right to exercise such Rights shall terminate and the only right with respect to such Rights thereafter of the holder of such Rights shall be to receive that number of Common Shares equal to the number of such Rights held by such holder multiplied by the Exchange Ratio. Promptly after the action of the Board of Directors of the Company ordering the exchange of any Rights pursuant to Section 27(a) hereof, the Company shall publicly announce such action, and within 10 calendar days thereafter shall give notice of any such exchange to all of the holders of such Rights at their last addresses as they appear upon the registry books of the Rights Agent; provided, however, that the failure to give, or any defect in, such notice shall not affect the validity of such exchange. Any notice which is mailed in the manner herein provided shall be deemed given, whether or not the holder receives the notice. Each such notice of exchange shall state the method by which the exchange of the Common Shares for Rights will be effected and, in the event of any partial exchange, the number of Rights which will be exchanged. Any partial exchange shall be effected pro rata based on the number of Rights (other than Rights which have become void pursuant to the provisions of Section 11(a)(ii) hereof) held by each holder of Rights.
- (c) In any exchange pursuant to this Section 27, the Company, at its option, may substitute for any Common Share exchangeable for a Right, (i) equivalent common shares (as such term is used in Section 11(a)(iii) hereof), (ii) cash, (iii) debt securities of the Company, (iv) other assets, or (v) any combination of the foregoing, in any event having an aggregate value which the Board of Directors of the Company shall have determined in good faith to be equal to the current market value of one Common Share (determined pursuant to Section 11(d) hereof) on the Trading Day immediately preceding the date of exchange pursuant to this Section 27.

Section 28. Successors; Certain Covenants. All the covenants and provisions of this Agreement by or for the benefit of the Company or the Rights Agent shall bind and inure to the benefit of their respective successors and assigns hereunder.

Section 29. Benefits of this Agreement. Nothing in this Agreement shall be construed to give to any Person other than the Company, the Rights Agent and the registered holders of the Right Certificates (and, prior to the Distribution Date, the Common Shares) any legal or equitable right, remedy or claim under this Agreement; but this Agreement shall be for the sole and exclusive benefit of the Company, the Rights Agent and the registered holders of the Right Certificates (or prior to the Distribution Date, the Common Shares).

Section 30. Determinations and Actions by the Board of Directors, etc. For all purposes of this Agreement, any calculation of the number of Common Shares or other

securities outstanding at any particular time, including for purposes of determining the particular percentage of such outstanding shares of which any Person is the Beneficial Owner, shall be made in accordance with the last sentence of Rule 13d-3(d)(1)(i) of the General Rules and Regulations under the Exchange Act as in effect on the date of this Agreement. The Board of Directors of the Company shall have the exclusive power and authority to administer this Agreement and to exercise all rights and powers specifically granted to the Board of Directors or to the Company, or as may be necessary or advisable in the administration of this Agreement, including, without limitation, the right and power to (i) interpret the provisions of this Agreement, and (ii) make all determinations deemed necessary or advisable for the administration of this Agreement (including a determination to redeem or exchange or not to redeem or exchange the Rights or to supplement or amend the Agreement). All such actions, calculations, interpretations and determinations (including, for purposes of clause (y) below, all omissions with respect to the foregoing) which are done or made by the Board of Directors in good faith, shall (x) be final, conclusive and binding on the Company, the Rights Agent, the holders of the Rights and all other parties, and (y) not subject any member of the Board to any liability to the holders of the Rights.

Section 31. Severability. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void or unenforceable, the remainder of the terms, provisions, covenants and restrictions of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.

Section 32. Governing Law. This Agreement and each Right Certificate issued hereunder shall be deemed to be a contract made under the internal substantive laws of the State of Georgia and for all purposes shall be governed by and construed in accordance with the internal substantive laws of such State applicable to contracts to be made and performed entirely within such State.

Section 33. Counterparts. This Agreement may be executed in any number of counterparts and each of such counterparts shall for all purposes be deemed to be an original, and all such counterparts shall together constitute but one and the same instrument.

Section 34. Descriptive Headings. Descriptive headings of the several Sections of this Agreement are inserted for convenience only and shall not control or affect the meaning or construction of any of the provisions hereof.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and their respective corporate seals to be hereunto affixed and attested, all as of the day and year first above written.

[SEAL]

Attest: CERTEGY INC.

By:	By:
Secretary	Name: Title:
[SEAL]	
Attest:	SUNTRUST BANK
Ву:	Ву:
Name: Title:	Name: Title:

Exhibit A Form of Right Certificate Rights

Certificate No. R-

NOT EXERCISABLE AFTER ______, 2011 OR EARLIER IF REDEEMED. THE RIGHTS ARE SUBJECT TO REDEMPTION, AT THE OPTION OF THE COMPANY, AT \$0.01 PER RIGHT ON THE TERMS SET FORTH IN THE RIGHTS AGREEMENT. UNDER CERTAIN CIRCUMSTANCES SPECIFIED IN THE RIGHTS AGREEMENT, RIGHTS BENEFICIALLY OWNED BY AN ACQUIRING PERSON OR AN AFFILIATE OR AN ASSOCIATE OF AN ACQUIRING PERSON (AS SUCH TERMS ARE DEFINED IN THE RIGHTS AGREEMENT) MAY BECOME NULL AND VOID.

Right Certificate

Certegy Inc.

This certifies that	, or registered assigns, is the
	Rights set forth above, each of which entitles
	terms, provisions and conditions of the Rights
Agreement, dated as of	_, 2001 (the "Rights Agreement"), between
Certegy Inc., a Georgia corporatio	n (the "Company"), and SunTrust Bank, a
Georgia banking corporation (the "	Rights Agent"), to purchase from the Company
at any time after the Distribution	Date (as such term is defined in the Rights
Agreement) and prior to 5:00 P.M.	(Eastern time) on, 2011 at the
principal office or offices of the	Rights Agent designated for such purpose, one
fully paid nonassessable share of	common stock, par value \$.01 per share (the
"Common Shares"), of the Company,	at a purchase price of [\$] per Common
Share (the "Purchase Price"), upon	presentation and surrender of this Right
Certificate with the Form of Elect	ion to Purchase and related Certificate duly
executed. If this Right Certificat	e shall be exercised in part, the holder shall
be entitled to receive upon surren	der hereof another Right Certificate or Right
	le Rights not exercised. The number of Rights
evidenced by this Right Certificat	e (and the number of Common Shares which may
) set forth above, and the Purchase Price set
forth above, are the number and Pu	rchase Price as of, 2001, based on
the Common Shares as constituted a	t such date.

As provided in the Rights Agreement, the Purchase Price and the number and kind of securities issuable upon the exercise of the Rights evidenced by this Right Certificate are subject to adjustment upon the happening of certain events.

This Right Certificate is subject to all of the terms, provisions and conditions of the Rights Agreement, which terms, provisions and conditions are hereby incorporated herein by reference and made a part hereof and to which Rights Agreement reference is hereby made for a full description of the rights, limitations of rights, obligations, duties and immunities of the Rights Agent, the Company and the holders of the Right Certificates, which limitations of rights

include the temporary suspension of the exercisability of the Rights under the circumstances specified in the Rights Agreement. Copies of the Rights Agreement are on file at the above-mentioned office of the Rights Agent.

Pursuant to the Rights Agreement, from and after the later of the Distribution Date and the first occurrence of a Flip-in Event (as such term is defined in the Rights Agreement), (i) any Rights that are or were acquired or beneficially owned by any Acquiring Person (or any Affiliate or Associate of such Acquiring Person) shall be void and any holder of such Rights shall thereafter have no right to exercise such Rights under any provision of the Rights Agreement, (ii) no Right Certificate shall be issued pursuant to the Rights Agreement that represents Rights beneficially owned by an Acquiring Person or any Affiliate or Associate thereof, (iii) no Right Certificate shall be issued at any time upon the transfer of any Rights to an Acquiring Person or any Affiliate or Associate thereof or to any nominee of such Acquiring Person or Affiliate or Associate thereof, and (iv) any Right Certificate delivered to the Rights Agent for transfer to an Acquiring Person or any Affiliate or Associate thereof, and (iv) any Right Certificate delivered to the Rights Agent for transfer to an Acquiring Person or any Affiliate or Associate thereof shall be cancelled.

This Right Certificate, with or without other Right Certificates, may be transferred, split up, combined or exchanged for another Right Certificate or Right Certificates, entitling the holder to purchase a like number of Common Shares (or other securities, as the case may be) as the Right Certificate or Right Certificates surrendered shall have entitled such holder (or former holder in the case of a transfer) to purchase, upon presentation and surrender hereof at the principal office of the Rights Agent designated for such purpose, with the Form of Assignment (if appropriate) and the related Certificate duly executed.

Subject to the provisions of the Rights Agreement, the Rights evidenced by this Certificate may be redeemed by the Company at its option at a redemption price of \$0.01 per Right. The Rights Agreement may be supplemented and amended by the Company, as provided therein.

The Company is not required to issue fractional Common Shares or other securities issuable upon the exercise of any Right or Rights evidenced hereby. In lieu of issuing such fractional Common Shares or other securities, the Company may make a cash payment, as provided in the Rights Agreement.

No holder of this Right Certificate, as such, shall be entitled to vote or receive dividends or be deemed for any purpose the holder of the Common Shares or of any other securities of the Company which may at any time be issuable upon the exercise of the Right or Rights represented hereby, nor shall anything contained herein or in the Rights Agreement be construed to confer upon the holder hereof, as such, any of the rights of a shareholder of the Company or any right to vote for the election of directors or upon any matter submitted to shareholders at any meeting thereof, or to give or withhold consent to any corporate action, or to receive notice of meetings or other actions affecting shareholders (except as provided in the Rights Agreement), or to receive dividends or subscription rights, or otherwise, until the Right or Rights evidenced by this Right Certificate shall have been exercised in accordance with the provisions of the Rights Agreement.

This Right Certificate shall not be valid or obligatory for any purpose until it shall have been countersigned by the Rights Agent.		
WITNESS the facsimile signature of the proper officers of the Company and its corporate seal. Dated as of, 20		
ATTEST:	Certegy Inc.	
ву:	ву:	
Secretary	Title:	
[SEAL]		
Countersigned:		
SunTrust Bank		
Dvv		

37

Authorized Signature

Form of Reverse Side of Right Certificate

FORM OF ASSIGNMENT

(To be executed by the registered holder if such holder desires to transfer the Right Certificate) $\,$

FOR VALUE RECEIVED,sells, assigns and transfers unto	_ hereby -
(Please print name and address of transferee)	
this Right Certificate, together with all right, title and interest ther does hereby irrevocably constitute and appoint Attotransfer the within Right Certificate on the books of the within-named Cwith full power of substitution.	rein, and orney, to
Dated:, 20	
 Signature	
Signature Guaranteed:	

CERTIFICATE

The undersigned hereby certifies by checking the appropriate boxes that:

- (1) the Rights evidenced by this Right Certificate [] are [] are not being sold, assigned, transferred, split up, combined or exchanged by or on behalf of a Person who is or was an Acquiring Person or an Affiliate or Associate of any such Person (as such terms are defined in the Rights Agreement);
- (2) after due inquiry and to the best knowledge of the undersigned, it [] did [] did not acquire the Rights evidenced by this Right Certificate from any Person who is, was or became an Acquiring Person or an Affiliate or Associate of an Acquiring Person.

Dated: _		20	
 Signatur	e		

FORM OF ELECTION TO PURCHASE

(To be executed if holder desires to exercise the Right Certificate)

To Certegy I	nc.
--------------	-----

The undersigned hereby irrevocably elects to exercise Rights represented by this Right Certificate to purchase the Common Shares or other securities issuable upon the exercise of such Rights and requests that certificates for such securities be issued in the name of:
Please insert social security or other identifying number:
(Please print name and address)
If such number of Rights shall not be all the Rights evidenced by this Right Certificate, a new Right Certificate for the balance remaining of such Rights shall be registered in the name of and delivered to:
Please insert social security or other identifying number:
(Please print name and address)
Optional Election to Exercise without Payment of Cash:
With respect to the exercise of the Rights specified above, the undersigned hereby elects to exercise such Rights without payment of cash and to receive a number of Common Shares or other securities having a value (as determined pursuant to the Rights Agreement) equal to the difference between (i) the value of the Common Shares or other securities that would have been issuable upon the exercise thereof upon payment of the cash amount as provided in the Rights Agreement, and (ii) the amount of such cash payment.
Dated:, 20
Signature
Signature Guaranteed:

40

CERTIFICATE

The undersigned hereby certifies by checking the appropriate boxes that:

	(1)	the	Rights	evid	enced	d by	this	s R	ight	Cer	tific	ate	[] are) (] ;	are
not	being	exe	ercised	by o	r on	beha	alf o	of .	а Ре	rson	who	is	or 1	was a	ın A	Acqu:	iring
Pers	on or	an	Affilia	ate o	r Ass	socia	ate o	of .	any	such	Pers	son	(as	such	ιte	erms	are
defi	ned p	ursi	uant to	the	Right	s Ag	reen	nen	t);								

after due inquiry and to the best knowledge of the undersigned,
it [] did [] did not acquire the Rights evidenced by this Right
Certificate from any Person who is, was or became an Acquiring Person or an
Affiliate or Associate of an Acquiring Person.

Dated:	, 2	20			
					 •
			Signatu	re	

NOTICE

Signatures on the foregoing Form of Assignment and Form of Election to Purchase and in the related Certificates must correspond to the name as written upon the face of this Right Certificate in every particular, without alteration or enlargement or any change whatsoever.

Signatures must be guaranteed by a member firm of a registered national securities exchange, a member of the National Association of Securities Dealers, Inc., or a commercial bank or trust company having an office or correspondent in the United States.

EXHIBIT 4.4 [SHARES]

[NUMBER]
[C]
COMMON STOCK
INCORPORATED UNDER THE LAWS OF THE
STATE OF GEORGIA

THIS CERTIFICATE IS TRANSFERABLE EITHER IN ATLANTA, GEORGIA OR IN NEW YORK, NEW YORK SEE REVERSE FOR CERTAIN DEFINITIONS CUSIP 156880 10 6

CERTEGY INC.

THIS CERTIFIES THAT

IS THE OWNER OF

FULLY PAID AND NON-ASSESSABLE SHARES WITH THE PAR VALUE OF \$.01 EACH OF THE COMMON STOCK OF

Certegy Inc. transferable on the books of the Company in person or by duly authorized attorney, upon the surrender of this certificate properly endorsed. The Articles of Incorporation of the Company authorize the issuance of preferred stock. The shares represented hereby will be subordinate to any outstanding shares of preferred stock issued pursuant to such authority with respect to dividends and amounts payable on liquidation. This certificate is not valid unless countersigned by the Transfer Agent and registered by the Registrar.

Witness the facsimile seal of the Company and the facsimile signatures of its duly authorized officers.

Dated: Certegy Inc. /s/ Bruce S. Richards /s/ Lee A. Kennedy [CERTEGY INC. **SECRETARY** PRESIDENT AND CHIEF CORPORATE **EXECUTIVE OFFICER** SEAL] COUNTERSIGNED AND SUNTRUST BANK **REGISTERED:** TRANSFER AGENT AND REGISTRAR BY -----______ AUTHORIZED OFFICER

CERTEGY INC.

The Company will furnish without charge to each shareholder who so requests, the designations, preferences and relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights. Any such request should be addressed to the Secretary of Certegy Inc. or to the Transfer Agent named on the face of this certificate.

The following abbreviations, when used in the inscription on the face of this certificate, shall be construed as though they were written out in full $\frac{1}{2} \int_{-\infty}^{\infty} \frac{1}{2} \left(\frac{1}{2} \int_{-\infty}^{\infty} \frac{1}{2} \left(\frac{1}$ according to applicable laws or regulations:

UPON THE FACE OF THE CERTIFICATE IN EVERY

PARTICULAR, WITHOUT ALTERATION OR ENLARGEMENT OR ANY CHANGE WHATEVER.

TEN COM - as tenants in common TEN ENT - as tenants by the entireties JT TEN - as joint tenants with right of survivorship and not as tenants in common	UNIF GIFT MIN ACTCustodian (Cust) (Minor) Under Uniform Gifts to Minors Act (State)
Additional abbreviations may also be use	ed though not in the above list.
For value received, PLEASE INSERT SOCIAL SECURITY OR OTHER IDENTITY OF ASSIGNEE //	
(PLEASE PRINT OR TYPEWRITE NAME AND ADDRESS,	, INCLUDING ZIP CODE, OF ASSIGNEE)
	shares
of the capital stock represented by the with irrevocably constitute and appoint	
	Attorney
to transfer the said stock on the books of power of substitution in the premises. Dated	
X	
NOTIO	CE: THE SIGNATURE TO THIS ASSIGNMENT
MUST	CORRESPOND WITH THE NAME AS WRITTEN

SIGNATURE(S) GUARANTEED:

THE SIGNATURE(S) SHOULD BE GUARANTEED BY AN ELIGIBLE GUARANTOR INSTITUTION (BANKS, STOCKBROKERS, SAVINGS AND LOAN ASSOCIATIONS AND CREDIT UNIONS WITH MEMBERSHIP IN AN APPROVED SIGNATURE GUARANTEE MEDALLION PROGRAM), PURSUANT TO S.E.C. RULE 17Ad-15.

LEGEND: This Certificate also evidences and entitles the holder hereof to certain Rights as set forth in a Rights Agreement between Certegy Inc. and SunTrust Bank, dated as of _____, 2001 (the "Rights Agreement"), the terms of which are hereby incorporated herein by reference and a copy of which is on file at the principal executive offices of Certegy Inc. Under certain circumstances, as set forth in the Rights Agreement, such Rights may be redeemed, may expire, may be amended or may be evidenced by separate certificates and no longer be evidenced by this Certificate. Certegy Inc. will mail to the holder of this Certificate a copy of the Rights Agreement without charge promptly after receipt of a written request therefor. Under certain circumstances as set forth in the Rights Agreement, Rights beneficially owned by an Acquiring Person or any Affiliate or Associate of an Acquiring Person (as such terms are defined in the Rights Agreement) may become null and void.

TAX SHARING AND INDEMNIFICATION AGREEMENT

THIS TAX SHARING AND INDEMNIFICATION AGREEMENT ("Agreement") is entered into as of _______, 2001, by and between Equifax Inc., a Georgia corporation ("Equifax"), and Certegy Inc., a Georgia corporation ("Certegy").

WHEREAS, Equifax is the common parent and Certegy is currently a member of an "affiliated group," as that term is defined in section 1504 of the Code (such term and certain capitalized terms being defined in Section 1.1), that currently files consolidated federal income tax returns; and

WHEREAS, Certegy is a holding company and a wholly-owned subsidiary of Equifax; and

WHEREAS, pursuant to that certain Distribution Agreement Plan of Reorganization and Distribution dated as of even date with this Agreement (the "Distribution Agreement") and subject to the terms and conditions thereof, Equifax will transfer and assign or cause members of the Equifax Group to transfer and assign to Certegy certain assets and businesses associated with the Payment Services Business and the stock of certain corporations that conduct the Payment Services Business; and

WHEREAS, prior to the Contribution, Equifax and certain Equifax Affiliates will undertake the Foreign Restructuring to separate the Payment Services Business from Equifax in foreign jurisdictions; and

WHEREAS, pursuant to the Distribution Agreement and subject to the terms and conditions thereof, after the Contribution, Equifax will distribute to its shareholders on a pro rata basis all of the issued and outstanding stock of Certegy; and

WHEREAS, the parties intend that the Contribution and the Distribution qualify as a tax-free reorganization and distribution under section 368(a)(1)(D) and section 355 of the Code;

WHEREAS, in contemplation of the Distribution, the Equifax Group and the Certegy Group desire to set forth their rights and obligations with respect to foreign, federal, state and local taxes due for periods both before and after the Distribution and with respect to certain tax and other liabilities that might be arise in connection with the Distribution;

NOW THEREFORE, in consideration of the foregoing premises and the mutual agreements and covenants contained in this Agreement and other good and valuable consideration the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

ARTICLE I DEFINITIONS

1.1 For purposes of this Agreement, the following definitions shall apply:

- (a) "2000 Certegy Tax Liabilities" has the meaning ascribed to such term in Section 5.5.
- (b) "2001 Certegy Estimated Tax Liabilities" has the meaning ascribed to such term in Section 5.6.
- (c) "Additional Restructuring Tax" means any Restructuring Tax other than any Contemplated Restructuring Taxes.
- (d) "Affiliated Group" means an affiliated group of corporations within the meaning of section 1504(a) (determined without regard to the exceptions contained in section 1504(b)) of the Code for the taxable period in question.
- (e) "Certegy Group" means (i) with respect to any period prior to the Distribution Date, Certegy, Payment Services, any other entity directly or indirectly conducting the Payment Services Business, and each of such entities' wholly-owned subsidiaries; provided that the Certegy Group shall not include Light Signatures, Inc., High Integrity Systems, Inc., or any other entity owned by Equifax after giving effect to the Separation and the Distribution, and (ii) with respect to any period on or after the Distribution Date, (A) the Affiliated Group of which Certegy or any successor of Certegy is the common parent; and (B) any entity in which any member of such Affiliated Group owns some or all of the equity.
- (f) "Certegy Issue" means any issue raised by any Tax Authority, which issue results in (i) proposed Restructuring Taxes for which Certegy could be liable pursuant to Section 2.3 hereof, or (ii) proposed Taxes for which Certegy could be liable pursuant to Section 2.2 hereof.
- (g) "Certegy Tainting Act" means (i) any breach by any member of the Certegy Group of any written representation or covenant relating to the Certegy Group made in any Ruling Document, or (ii) any action or actions of or involving any member of the Certegy Group, or any omission or omissions of any such Person (whether or not such action or omission is permitted pursuant to this Agreement), of an action or actions available to it, after the Distribution Date, if such breach, action or omission described in (i) or (ii) contributes to a Final Determination imposing any Additional Restructuring Tax on any member of the Equifax Group or the Certegy Group.
- (h) "Code" means the Internal Revenue Code of 1986, as amended from time to time.
- (i) "Consolidated Returns" means the consolidated United States federal income tax returns of the Affiliated Group of which Equifax is the common parent for

consolidated return years beginning before the Distribution Date and any consolidated, combined or similar state income tax returns of any members of the Equifax Group for taxable years beginning before the Distribution Date (including, in each case, any amendments thereto).

- (j) "Contemplated Restructuring Tax" means any Restructuring Tax set forth on Exhibit B to this Agreement.

- (m) "Distribution Agreement" has the meaning ascribed to such term in the recitals to this Agreement.
- (n) "Distribution Date" has the meaning ascribed to such term in the Distribution Agreement.
- - (p) "Equifax Affiliate" means any member of the Equifax Affiliated Group.
- (q) "Equifax Group" means, for each taxable period, (i) the Affiliated Group of which Equifax or any successor of Equifax is the common parent; and (ii) any entity in which any member of the Affiliated Group described in clause (i) above owns some or all of the equity, provided, however, that the Equifax Group shall not include any member of the Certegy Group.
- (r) "Equifax Issue" means any issue raised by any Tax Authority, which issue results in (i) proposed Restructuring Taxes (whether or not it is alleged that a member of the Equifax Group is at fault or is partially at fault), or (ii) proposed Taxes for which Equifax could be liable pursuant to Section 2.1 hereof.
- (s) "Equifax Tainting Act" means (i) any breach by any member of the Equifax Group of any written representation or covenant relating to the Equifax Group made in any Ruling Document, or (ii) any action or actions of or involving any member of the Equifax Group or any omission or omissions of any such Person of an action or actions available to it (whether or not such action or omission is permitted pursuant to this Agreement), after the

Distribution Date, if such breach, action or omission described in (i) or (ii) contributes to a Final Determination imposing any Additional Restructuring Tax on any member of the Equifax Group or the Certegy Group.

- (t) "Expenses" means out-of-pocket expenses and shall not include any overhead or indirect costs.
- (u) "Final Determination" means the final resolution of liability for any Tax for a taxable period (i) by IRS Form 870 or 870-AD (or any successor forms thereto), on the date of acceptance by or on behalf of the IRS, or by a comparable form under the laws of other jurisdictions, except that a Form 870 or 870-AD, successor form, or comparable form that reserves the right of the taxpayer to file a claim for refund and/or the right of the Tax Authority to assert a further deficiency shall not constitute a Final Determination; (ii) by a decision, judgment, decree, or other order by a court of competent jurisdiction which has become final and unappealable; (iii) by a closing agreement or offer in compromise under section 7121 or 7122 of the Code or any subsequently enacted corresponding provisions of the Code, or comparable agreements under the laws of other jurisdictions; (iv) by an allowance of a refund or credit in respect of an overpayment of Tax, but only after the expiration of all periods during which such refund may be recovered (including by way of offset) by the Tax imposing jurisdiction; or (v) by any other final disposition by reason of the expiration of the applicable statute of limitations.
- (v) "Foreign Restructuring" means the transactions undertaken prior to the Contribution to separate the Payment Services Business from the other Equifax businesses in foreign jurisdictions, as described on Exhibit A.
- (w) "Group" means the Equifax Group or the Certegy Group, as the context so requires.
- (x) "Initial IRS Ruling" means the private letter ruling issued by the IRS regarding certain Federal income Tax consequences of the Separation and the Distribution.
 - (y) "IRS" means the United States Internal Revenue Service.
- (z) "Payment Services" means Equifax Payment Services, Inc., a wholly-owned subsidiary of Equifax organized under the laws of the State of Delaware.
- (aa) "Payment Services Business" means the businesses conducted by Equifax and its affiliates of providing payment transaction processing and check risk management services to financial institutions and merchants.

- (bb) "Post-Distribution Period" means (i) any Tax Period beginning after the Effective Time and (ii) the portion of any Straddle Period beginning after the Effective Time.
- (cc) "Pre-Distribution Period" means (i) any Tax Period ending on or before the Effective Time and (ii) the portion of any Straddle Period ending on or before the Effective Time.
- (dd) "Regulations" means Income Tax Regulations issued by the United States Department of Treasury pursuant to the Code.
- (ee) "Restructuring Tax" means any Tax imposed on any member of the Equifax Group or the Certegy Group resulting from the Separation or the Distribution that would not have been imposed had such transactions not occurred.
- (ff) "Ruling Documents" means (i) the request for the Initial IRS Ruling, filed with the IRS in connection with the Separation and Distribution, together with any supplemental filings or requests for Supplemental Ruling or other materials subsequently submitted on behalf of Equifax, its subsidiaries and shareholders to the IRS, the appendices and exhibits thereto, and any rulings issued by the IRS to Equifax in connection with the Separation and Distribution or (ii) any similar filings submitted to, or rulings issued by, any other Tax Authority in connection with the Separation or Distribution.
- (gg) "Separation" means the transactions necessary to transfer the Payment Services Business to Certegy, including without limitation, the transactions necessary to transfer the equity of High Integrity Systems, LLC to Equifax, the Foreign Restructuring, and the Contribution.
- (hh) "Special Refund Allocation Ratio" means with respect to either Group, the aggregate outstanding amount of Special Refund Claims filed by a member of such Group pending as of the Effective Date divided by the aggregate outstanding amount of all Special Refund Claims.
- (ii) "Special Refund Claims" means the claims for refund of state income Taxes filed by members of the Equifax Group and the Certegy Group prior to the Effective Date and still outstanding as of the Effective Date, as set forth on Exhibit C to this Agreement.
- (jj) "Straddle Period" means any Tax Period that begins before the Effective Time and ends after the Effective Time.

- (kk) "Supplemental Ruling" means any ruling issued after the issuance of the Initial IRS Ruling (a) by the IRS in connection with the Distribution or any transactions undertaken in connection with the Distribution or (b) by any other Tax Authority, addressing the application of a provision of the laws of another jurisdiction to any transaction undertaken in connection with the Distribution.
- (11) "Supplemental Ruling Documents" has the meaning ascribed to such term in Section 2.5(d)(i).
- (mm) "Tax" or "Taxes" means all forms of taxation, whenever created or imposed, whether domestic or foreign, imposed by any Tax Authority, and without limiting the generality of the foregoing shall include net income, alternative or add-on minimum tax, gross income, sales, use, franchise, gross receipts, value added, ad valorem, profits, license, payroll, withholding, social security, unemployment insurance, employment, property, transfer, recording, excise, severance, stamp, occupation, premium, windfall profit, custom duty, or other tax, governmental fee or other like assessment or charge of any kind whatsoever, together with any related interest, penalties or other additions to tax, or additional amounts imposed by any such Tax Authority. For purposes of computing the Taxes of a party for any purpose of this Agreement, interest shall be computed without regard to any Tax Items attributable to any other party (as determined pursuant to Section 2.4) and without regard to any netting of interest on any refund of Tax owed by the Tax Authority to the other party.
- (nn) "Tax Authority" means any nation, locality, municipality, government, authority, state, federation, or other governmental body or agency.
- (00) "Tax Benefit" means any Tax Item which decreases Taxes paid or payable.
- (pp) "Tax Controversy" means any audit, examination, dispute, suit, action, litigation, or other judicial or administrative proceeding by or against the IRS or any other Tax Authority. Notwithstanding the foregoing, any proceeding pursuant to the Special Refund Claims shall not constitute a Tax Controversy for purposes of this Agreement.
- (qq) "Tax Item" means any item of income, gain, loss, deduction, credit, recapture of credit or any other item, including, but not limited to, an adjustment under Code section 481 resulting from a change in accounting method, which increases or decreases Taxes paid or payable.
- (rr) "Tax Period" means with respect to any Tax, the period for which the Tax is reported. $\label{eq:taylor}$

(ss) "Tax Returns" means all reports, estimates, declarations of estimated tax, information statements, returns or other documents required or permitted to be filed with a Tax Authority in connection with any Taxes, including but not limited to requests for extensions of time, information statements and reports, claims for refund, and amended returns.

ARTICLE II ALLOCATION OF TAX LIABILITIES

2.1 Equifax Group.

- (a) Current and Prior Periods. Except as otherwise provided in this Agreement, Equifax shall be responsible and liable for all Taxes attributable to the Equifax Group that are imposed for any Pre-Distribution Period. Equifax hereby assumes all such liability and shall indemnify and hold harmless Certegy and any member of the Certegy Group from and against any share or amount of all Taxes attributable to the Equifax Group that are imposed for any Pre-Distribution Period.
- (b) Future Periods. Except as otherwise provided in this Agreement, Equifax shall be responsible and liable for all Taxes attributable to the Equifax Group that are imposed for any Post-Distribution Period and shall indemnify and hold harmless Certegy and any member of the Certegy Group from and against all such Taxes.

2.2 Certegy Group.

- (a) Current and Prior Periods. Except as otherwise provided in this Agreement, Certegy shall be responsible and liable for all Taxes attributable to the Certegy Group that are imposed for all Pre-Distribution Periods. Certegy hereby assumes all such liability and shall indemnify and hold harmless Equifax and any member of the Equifax Group from and against any share or amount of Taxes attributable to the Certegy Group that are imposed for any Pre-Distribution Period.
- (b) Future Periods. Except as otherwise provided in this Agreement, Certegy shall be liable for all Taxes attributable to the Certegy Group that are imposed for any Post-Distribution Period, and shall indemnify and hold harmless Equifax and any member of the Equifax Group from and against all such Taxes.

2.3 Restructuring Taxes.

(a) Generally. Notwithstanding Section 2.1 or Section 2.2 to the contrary, Equifax shall be responsible and liable for and shall indemnify and hold harmless each member of the Certegy Group from and against any and all Contemplated Restructuring Taxes. Notwithstanding anything in this Article II (other than Section 2.3(b) or Section 2.3(c)) to the contrary, Equifax shall be responsible and liable for, and shall indemnify and hold harmless

Certegy and any member of the Certegy Group from and against, any and all Additional Restructuring Taxes.

- (b) Liability and Indemnification for Certegy Tainting Acts. Notwithstanding anything in this Article II (other than Section 2.3(c)) to the contrary, Certegy shall be responsible and liable for and shall indemnify and hold harmless each member of the Equifax Group from and against, (i) any Additional Restructuring Taxes, and (ii) any liability resulting from a decision that Equifax is liable to Equifax's or Certegy's shareholders because of a Final Determination that the Distribution is taxable, but in any case only to the extent such Additional Restructuring Taxes or liability to shareholders is due to a Certegy Tainting Act (and not an Equifax Tainting Act).
- (c) Liability and Indemnification for Combined Tainting Acts. Notwithstanding anything in this Article II to the contrary, in the event of a Final Determination that Additional Restructuring Taxes are due to a Tax Authority and such Additional Restructuring Taxes are caused by both an Equifax Tainting Act and a Certegy Tainting Act, then the liability of Equifax and Certegy for any Restructuring Taxes arising from such Final Determination and any liability to shareholders arising from such Final Determination shall be borne fifty percent (50%) by Equifax and fifty percent (50%) by Certegy. Each party, jointly and severally with its Affiliated Group, agrees to pay and to indemnify and hold the other party harmless from and against the amount of Additional Restructuring Taxes and liability to shareholders allocated to such first party under this subsection 2.3(c).
- 2.4 Allocation of Tax Liabilities. For purposes of determining whether any Taxes or Tax Items are attributable to the Certegy Group or the Equifax Group and whether any Tax Item is attributable to a Pre-Distribution Period or a Post-Distribution Period, the following principles shall apply:
- (a) For purposes of this Agreement, the determination of whether Taxes or Tax Items are "attributable" to a member of the Certegy Group or the Equifax Group shall be made using the methods that Equifax has used in Tax Periods prior to the Distribution Date to allocate its Tax liabilities among the various members of its affiliated group; provided that in any state that imposes income Tax on a unitary basis, such determination for any Straddle Period shall be based on apportionment factors applicable for the businesses and operations of each Group as of the Effective Time notwithstanding any change in apportionment factors resulting from a change in such businesses and operations occurring after the Effective Time. The fact that a member of the Certegy Group or the Equifax Group prepared or filed a return with respect to any Taxes is not relevant in determining whether such Taxes are "attributable" to such member.
- (b) The principles of Treasury Regulation Section 1.1502-76(b) as reasonably interpreted and applied by Equifax shall apply in determining whether a Tax Item is attributable to a Tax Period provided that (i) no election shall be made under Treasury Regulation Section 1.1502-76(b)(2)(ii) (relating to ratable allocation of a year's item); and (ii) if the Distribution

Date does not coincide with the end of a calendar month, the provisions of Treasury Regulation Section 1.1502-76(b)(2)(iii) will be applied to ratably allocate the items (other than extraordinary items) for the month that includes the Distribution Date (or the parties shall consistently use such accounting conventions as the parties shall hereafter agree in writing).

- (c) In determining the apportionment of Tax Items between Pre-Distribution Periods and Post-Distribution Periods, any Tax Items relating to the Separation and the Distribution shall be treated as an extraordinary item described in Treasury Regulation Section 1.1502-76(b)(2)(ii)(C) and shall be allocated to Pre-Distribution Periods, and any Taxes related to such items shall be treated under Treasury Regulation Section 1.1502-76(b)(2)(iv) as relating to such extraordinary item and shall be allocated to Pre-Distribution Periods.
 - 2.5 Certain Representations and Covenants.
- $\mbox{\ \ (a)\ \ }$ The parties hereby make the following representations and covenants.
 - (i) Certegy for itself and on behalf of each member of the Certegy Group represents to Equifax that the information and representations furnished in any Ruling Document (as modified, qualified or elaborated in any subsequent Ruling Documents) are accurate and complete as of the date hereof, to the extent that such information and representations relate to the Certegy Group or the business or activities of such entity.
 - (ii) Equifax for itself and on behalf of each member of the Equifax Group represents to Certegy that, as of the date hereof, there is no plan or intention to take any action inconsistent with the information and representations furnished in any Ruling Documents (as modified, qualified or elaborated in any subsequent Ruling Documents).
 - (iii) Each of Certegy and Equifax respectively represents to the other party that, as of the date hereof, it is not aware of any plan or intention by the current shareholders of Equifax to sell, exchange, transfer by gift, or otherwise dispose of any of their stock in, or securities of, Equifax or Certegy subsequent to the Distribution, except as described in any Ruling Document (as modified, qualified or elaborated in any subsequent Ruling Documents).
 - (iv) Each of Certegy and Equifax respectively covenants to the other party (A) that it will use its best efforts to verify that the foregoing representations made by it in this Section 2.5(a) are accurate and complete as of the Distribution Date and (B) if, after the date hereof, it obtains information indicating, or otherwise becomes aware, that any such representations are or may be inaccurate or incomplete, that it will promptly inform the other party, as the case may be.
 - (v) Each of Certegy and Equifax respectively covenants to the other party that it shall treat for income Tax purposes all transactions that are the subject of the Initial IRS Ruling consistently therewith.

- (b) Each of Certegy and Equifax covenants to the other party that, except as permitted in Section 2.5(c):
 - (i) During the two-year period following the Distribution Date, it will not merge or consolidate with any other person, or enter into any transaction that constitutes a liquidation of such entity for federal income tax purposes.
 - (ii) During the two-year period following the Distribution Date, it will not sell, exchange, distribute or otherwise dispose of assets used in the active conduct of the historic businesses relied upon to satisfy Code section 355(b), if such sale, exchange, distribution, or other disposition would cause Code section 355(b) not to be satisfied if such test were applied immediately after such transaction.
 - (iii) Following the Distribution, it will, for a minimum of two years, continue the active conduct of the historic businesses relied upon to satisfy Code section 355(b).
 - (iv) It will not redeem, reacquire or otherwise repurchase stock in a manner contrary to the requirements of Section 4.05 of Revenue Procedure 96-30.
 - (v) During the two-year period following the Distribution Date, it will not engage in any transaction resulting in the direct or indirect acquisition of such party's stock representing a 50% or greater interest in such party within the meaning of Code section 355(d)(4).

Equifax further covenants that it will prevent any member of the Equifax Group which is a party to the Canadian Restructuring or the U.K. Restructuring (each as defined in Exhibit A) from undertaking any transaction described in Section 2.5(b)(i) through Section 2.5(b)(v) above except as permitted in Section 2.5(c) below. Certegy further covenants that it will prevent any member of the Certegy Group which is a party to the Canadian Restructuring or the U.K. Restructuring from undertaking any transaction described in Section 2.5(b)(i) through Section 2.5(b)(v) above except as permitted in Section 2.5(c) below.

- (c) A party may take actions inconsistent with the covenants contained in Section 2.5(b), if prior to taking such action:
 - (i) Such party obtains an opinion of counsel, which counsel and which opinion are acceptable to the other party in its reasonable discretion, to the effect that such actions should not affect the Federal income Tax treatment of the Separation and the Distribution to the parties and the shareholders of Equifax as set forth in any prior private letter ruling issued by the IRS, it being understood that each party agrees to cooperate with the other party and use its reasonable best efforts to assist the other party in attempting to obtain, as expeditiously as possible, any opinion requested by the other party described in this Section 2.5(c)(i);

- (ii) Equifax obtains a Supplemental Ruling to the effect that such actions will not affect the Federal income Tax treatment of the Separation and the Distribution to the parties and the shareholders of Equifax as set forth in any prior private letter ruling issued by the IRS; or
 - (iii) Such party obtains the written consent of the other party.

(d) Supplemental Rulings.

- Equifax agrees that at the reasonable request of Certegy, Equifax shall cooperate with Certegy and use its reasonable best efforts to seek to obtain, as expeditiously as possible, a Supplemental Ruling or other guidance from a Tax Authority for the purpose of confirming (A) the continuing validity of any ruling (including another Supplemental Ruling) previously issued by the IRS or any other Tax Authority, or (B) compliance on the part of a member of the Certegy Group with its obligations under this Section 2.5. Equifax shall be obligated to seek a Supplemental Ruling requested by Certegy unless it reasonably believes that the relevant Tax Authority would not issue such a ruling. Notwithstanding the foregoing sentence, in no event shall Equifax be obligated to file a request for a Supplemental Ruling unless Certegy represents that (a) it has read the request for the Supplemental Ruling and any materials, appendices and exhibits to be submitted or filed therewith (`Supplemental Ruling Documents") and (B) all information (other than information provided by an external expert) and representations, if any, relating to any member of the Certegy Group contained in the Supplemental Ruling Documents are true, correct and complete in all material respects. Certegy shall reimburse Equifax for all reasonable costs and expenses incurred by Equifax in seeking or obtaining a Supplemental Ruling requested by Certegy. Certegy hereby agrees that Equifax shall have sole and exclusive control over the process of obtaining a Supplemental Ruling, and that only Equifax shall apply for a Supplemental Ruling. Certegy further agrees that it shall not seek any guidance from the IRS or any other Tax Authority concerning the Separation and the Distribution except as set forth in this Section 2.5(d).
- (ii) If Equifax determines that it will seek a Supplemental Ruling or other guidance from a Tax Authority after the date of this Agreement: (A) Equifax shall keep Certegy informed in a timely manner of all material actions taken or proposed to be taken in connection therewith; (B) Equifax shall (1) reasonably in advance of the submission of any such Supplemental Ruling Documents, provide Certegy with a draft copy thereof, (2) reasonably consider Certegy's comments on such draft copy, and (3) provide Certegy with a final copy of the Supplemental Ruling Documents; and (C) Equifax shall provide Certegy with notice reasonably in advance of, and Certegy shall have the right to attend, any formally scheduled meetings with the Tax Authority (subject to the approval of the Tax Authority) that relate to such Supplemental Ruling.

ARTICLE III REFUNDS OF TAXES

3.1 General. Except as provided in Section 3.2, each party shall be entitled to retain or be paid all refunds of Tax received, whether in the form of payment, credit or otherwise, from any Tax Authority with respect to any Tax Returns filed or to be filed by such party or a member of such party's Group in accordance with Article V of this Agreement, provided, however, that Certegy shall be entitled to retain or be paid all such refunds with respect to any Taxes to the extent such Taxes exclusively pertain to property or operations of the Certegy Group. Notwithstanding anything contained in this Section 3.1 to the contrary, Equifax shall be entitled to be paid and to retain, and Certegy shall not be entitled to retain and shall be required to pay over to Equifax, any refunds of Tax received to the extent (i) Equifax indemnified Certegy for the Taxes attributable to such refunds, or (ii) Equifax paid to the Tax Authority the Taxes attributable to such refunds and Equifax has not been indemnified by Certegy.

3.2 Special Refund Claims.

- (a) The parties acknowledge that various members of the Equifax Group and the Certegy Group have filed the Special Refund Claims prior to the Effective Date. Notwithstanding any other provision of this Agreement, the parties shall be paid or shall retain any refunds received as a result of the Special Refund Claims as follows:
 - (i) Until the aggregate amount of Tax refunds received in respect of Special Refund Claims is equal to the product of (A) the balance of general ledger account "Company 25 220-02-06" as of the Effective Time multiplied by (B) 2.05, Equifax shall be entitled to be paid or to retain all proceeds from Tax refunds received by any party in respect of the Special Refund Claims. Equifax shall be responsible and liable for paying all professional fees and other costs directly related to the pursuit and collection and all Taxes attributable to the receipt of such Tax refunds and shall indemnify and hold harmless Certegy and any member of the Certegy Group against any such fees or Taxes.
 - (ii) The proceeds of any Tax refunds received by any party in respect of the Special Refund Claims in excess of the amount set forth in Section 3.2(a)(i) shall be shared by the parties in proportion to their respective Special Refund Allocation Ratios. The parties shall be responsible and liable for the cost of all professional fees and other costs and expenses directly related to the pursuit and collection of Special Refund Claims and the collection of Tax refunds (including, without limitation, all Taxes attributable thereto) described in the preceding sentence in proportion to their respective Special Refund Allocation Ratios. Each party shall indemnify the other party for any amounts described in the preceding sentence to the extent that such party (or any member of such party's Group) incurs any such amounts in excess of such party's Special Refund Allocation Ratio.

- (b) Each party shall notify the other party promptly of any communication with any Tax Authority relating in whole or in part to any Special Refund Claim. Equifax and Certegy shall cooperate with each other to allow each party to pursue the Special Refund Claims, and each party agrees to take all actions reasonably necessary or advisable to pursue the collection of all Tax refunds pursuant to the Special Refund Claims; provided that neither party shall have any obligation to take any action that would cause such party to incur additional Taxes (other than Taxes attributable to the receipt of the proceeds of the refund).
- (c) In the event that either Equifax or Certegy elects to withdraw from the pursuit of any Special Refund Claim by giving notice of such withdrawal to the other party, the Withdrawing Party (or any member of the Withdrawing Party's Group) shall thereafter have no right to retain or to be paid any Tax refund pursuant to such Special Refund Claim and shall have no obligation to pay any professional fees or other costs or expenses incurred after the date of such notice related to the pursuit of such Special Refund Claim. The Withdrawing Party shall continue to have the obligations described in Section 3.2(b). The remaining party may continue to pursue such Special Refund Claim, and shall be entitled to retain or to be paid all proceeds from any Tax refund received pursuant to such Special Refund Claim.

ARTICLE IV CARRYBACKS FROM SEPARATE RETURN YEARS

- 4.1 General. Notwithstanding anything herein to the contrary, with the prior written consent of Equifax, which consent may not be unreasonably withheld, the Certegy Group may elect to carry back to any Taxable Period beginning before the Distribution Date any Tax Item arising in any Taxable Period beginning after the Distribution Date that the Certegy Group may properly elect to carry back for Federal income Tax purposes or combined state Tax purposes to a Consolidated Return. With respect to any such carryback, Equifax agrees to file such claims for refund and other returns as may be required to claim the Tax refunds attributable to such carryback items and to pay promptly after receipt to Certegy the cash amount of any refunds of Taxes, including the cash amount of any interest resulting from the utilization of such Tax Items, after taking into consideration any resulting increase or decrease in the Tax liability of any member of the Equifax Group. To the extent authorized by law, Equifax shall act as collection agent for the Certegy Group with respect to any such refund.
- 4.2 Review and Expenses. The amount of any carryback by the Certegy Group shall be reviewed and approved (on the basis of Tax information contained in Certegy's Tax Return) by Equifax's certified public accountants as to the amount and validity of such carryback. Certegy agrees to reimburse Equifax for its reasonable Expenses incurred in reviewing, filing and securing any Certegy refund claims hereunder.
- 4.3 Subsequent Disallowance. In the event that any Tax attribute for which Equifax has made a payment pursuant to Section 4.1 is subsequently reduced or disallowed, Certegy shall indemnify Equifax and hold it harmless from any Tax liability, including interest and penalties, incurred by reason of such reduction or disallowance.

ARTICLE V TAX RETURN PREPARATION

5.1 Consolidated Returns.

- (a) Equifax shall prepare and timely file all Consolidated Returns. Certegy shall have a reasonable period under the circumstances to review the 2000 and 2001 Consolidated Returns. The Consolidated Returns shall be prepared and filed by Equifax in compliance with applicable Tax laws and on a basis that is consistent with any Ruling Documents or legal opinion obtained by Equifax in connection with the Distribution or Separation and, subject to the foregoing, consistent with Equifax's prior Consolidated Returns.
- (b) Certegy shall be responsible for preparing all information relating to the Certegy Group necessary for Equifax to prepare and file the Consolidated Returns. Such information shall include the annual federal and state, if any, Tax work preparation package, necessary to enable Equifax to prepare the Consolidated Returns, completed and delivered to Equifax on or before the same deadline imposed upon other Equifax business units. Such information shall be used as the basis for Equifax's preparation of the Consolidated Returns.
- (c) Equifax shall not make any election or consent in connection with such Consolidated Returns without the consent of Certegy, which consent shall not be unreasonably withheld, unless (i) such election or consent is not binding on any member of the Certegy Group for any Tax Period; or (ii) Certegy may revoke such elections without the consent of any Tax Authority. Certegy and the Certegy Group agree not to elect to be excluded from any such Consolidated Return.
- (d) Certegy and the Certegy Group agree to cooperate with Equifax, at Equifax's expense, in the preparation of any valuation studies or other reports which are appropriate or necessary for the preparation of the Consolidated Returns.

5.2 Other Pre-Distribution Returns.

(a) Equifax shall prepare and timely file all other Tax Returns of any member of the Equifax Group or the Certegy Group for all Tax Periods that are exclusively Pre-Distribution Periods and all Straddle Periods.

Notwithstanding the foregoing sentence, Certegy shall prepare and timely file Tax Returns for any Pre-Distribution Period and any Straddle Period if such Tax Returns pertain exclusively to property or operations of the Certegy Group; provided that Equifax, at the reasonable request of Certegy, shall file sales and use tax returns pertaining to the operations of the Certegy Group that are required to be filed on or before September 30, 2001. Notwithstanding anything contained in the previous sentence, Equifax shall continue to file all Tax Returns of the members of the Certegy Group for any Pre-Distribution Period if the Tax Returns for such period are required to be filed on or prior to the Distribution Date. Certegy shall have a reasonable period under the circumstances to review each such Tax Return.

- (b) Certegy shall reimburse Equifax the aggregate amount of \$63,000.00 for professional fees paid to Tarpley & Underwood, P.C. attributable to the preparation of 2000 and 2001 Tax returns for the Certegy Group, which amount (without interest) shall be payable in six equal monthly installments beginning on July 31, 2001.
- 5.3 Post-Distribution Returns. Certegy shall prepare and timely file all Tax Returns of any member of the Certegy Group for Tax Periods that are exclusively Post-Distribution Periods, and Equifax shall prepare and timely file all Tax Returns of any member of the Equifax Group for Tax Periods that are exclusively Post-Distribution Periods.
- 5.4 Cooperation; Exchange of Information. Each party shall be responsible for the timely submission to the other party of information of which it has knowledge regarding any Tax Item which may properly be included in any Tax Return to be filed by the other party or any member of the other party's Affiliated Group, and shall provide any and all other information and documentation (including, but not by way of limitation, working papers and schedules) reasonably requested by the other party for use in connection with the preparation and filing of any Tax Returns.
 - 5.5 Payment of 2000 Certegy Tax Liabilities; Certegy Tax Benefits.
- (a) Certegy shall pay to Equifax an amount equal to the aggregate amount of Taxes that would be owed by the Certegy Group for the Tax Period consisting of the 2000 calendar year to the extent that such Taxes must be paid by Equifax as part of a Consolidated Return (the "2000 Certegy Tax Liabilities"). Such amount shall be computed in a manner consistent with Section 2.4, and shall be reduced by any Tax Benefit of Equifax attributable to a Certegy Tax Item. Notwithstanding the above, such amount shall not include amounts related to separate returns filed on a combined or unitary basis, except to the extent consistent with the manner in which the Tax liabilities on such returns were allocated in prior years. Payment of the 2000 Certegy Tax Liabilities shall be due on or before the tenth (10th) day after receipt by Certegy of a notice including the applicable computations prepared by Equifax in connection with the filing of Equifax's Consolidated Return for the 2000 calendar year.
- (b) Equifax shall pay to Certegy any Tax Benefit of Equifax attributable to a Certegy Tax Item, to the extent that such Tax Benefit exceeds the amount of the 2000 Certegy Tax Liabilities. Payment of such amount shall be due on or before the tenth (10th) day after receipt by Certegy of a notice including the applicable computations prepared by Equifax in connection with the filing of Equifax's Consolidated Return for the 2000 calendar year.
 - 5.6 Payment of 2001 Certegy Estimated Tax Liabilities; Adjustment.
- (a) Certegy shall pay to Equifax an amount equal to the estimated aggregate amount of Taxes that would be owed by the Certegy Group for the Tax Period consisting of the 2001 calendar year to the extent that such Taxes must be paid by Equifax as part of a Consolidated Return (the "2001 Certegy Estimated Tax Liabilities"). Such estimated amount shall be computed in a manner consistent with Section 2.4, and shall be reduced by any Tax

Benefit of Equifax attributable to a Certegy Tax Item. Notwithstanding the above, such estimated amount shall not include amounts related to separate returns filed on a combined or unitary basis, except to the extent consistent with the manner in which the Tax liabilities on such returns were allocated in prior years. Payment of the 2001 Certegy Estimated Tax Liabilities shall be due on or before the tenth (10th) day after receipt by Certegy of a notice including the applicable computations prepared by Equifax.

- (b) Upon the filing of the Consolidated Return for the Equifax Group's Tax Period ending December 31, 2001, the 2001 Certegy Estimated Tax Liabilities previously computed shall be restated and adjusted by Equifax based upon information then available. An adjusting payment shall be made by Equifax or Certegy as shall be required by any restatement or adjustment of the 2001 Certegy Estimated Tax Liabilities. Such payment shall be due on or before the tenth (10th) day after receipt by Certegy of a notice including the applicable computations prepared by Equifax.
- (c) Equifax shall pay to Certegy any Tax Benefit of Equifax attributable to a Certegy Tax Item, to the extent that such Tax Benefit exceeds the amount of the 2001 Certegy Tax Liabilities. Payment of such amount shall be due on or before the tenth (10th) day after receipt by Certegy of a notice including the applicable computations prepared by Equifax in connection with the filing of Equifax's Consolidated Return for the 2001 calendar year.

ARTICLE VI TAX CONTROVERSIES AND RECORDS

6.1 Tax Controversies.

- (a) Each of Certegy and Equifax shall have full responsibility and discretion in handling, settling, or contesting any Tax Controversy involving a Tax for which such party is liable pursuant to Article II of this Agreement. If a Tax Controversy proceeding involves both (i) one or more issues that would result in Taxes for which Equifax is liable under this Agreement and (ii) one or more issues that would result in Taxes for which Certegy is liable under this Agreement, then Equifax and Certegy shall cooperate with each other to allow each party to conduct the Tax Controversy with respect to those issues that would result in Taxes for which such party is liable. Furthermore, Equifax may participate in any Tax Controversy with respect to Restructuring Taxes regardless of whether it has liability or indemnification obligations with respect to such Taxes under this Agreement.
- (b) The party responsible for any Tax Controversy shall use all reasonable efforts (taking into consideration all relevant facts and circumstances known to the party) to resist any deficiency assertions by any Tax Authority regardless of which party is ultimately responsible for any such Tax under this Agreement.

- (c) Equifax shall notify Certegy promptly of any communication with the IRS or other Tax Authority relating in whole or in part to any Certegy Issue.
 - (i) Certegy shall have 30 days after receipt of such notice from Equifax within which to object to the proposed adjustment relating to a Certegy Issue (that is not an Equifax Issue). If Certegy does not notify Equifax within such 30 day period that it objects to the proposed adjustment, then subsections 6.1(c)(ii) through 6.1(c)(v) below shall not apply, and Equifax shall have exclusive control over all stages of the Tax Controversy, including full authority to determine whether and in what manner to contest or compromise the issue, unless and until Certegy so notifies Equifax.
 - (ii) If Certegy notifies Equifax that it objects to the proposed adjustment relating to a Certegy Issue (that is not an Equifax Issue), then Equifax shall not thereafter consent to the adjustment or compromise of such Certegy Issue without the consent of Certegy, but shall cooperate with Certegy to resolve the Certegy Issue on a basis acceptable to Certegy. Prior to the issuance of a notice of proposed adjustment or similar stage in the proceedings, however, Equifax shall be responsible for the conduct of the audit, including matters pertaining to such Certegy Issue. Equifax shall notify Certegy in advance of any conferences, meetings, and proceedings pertaining to the audit and, at its own expense, Certegy shall have the right to attend all such proceedings with any Tax Authority, the subject matter of which is or includes such Certegy Issue.
 - (iii) Upon the issuance of a notice of proposed adjustment or similar stage in the proceedings, Certegy shall assume the conduct of all further proceedings, with counsel selected by it, at Certegy's sole expense, insofar as the proceedings relate to a Certegy Issue (that is not an Equifax Issue), and thereafter Certegy and Equifax shall jointly be responsible for the conduct of proceedings to contest such Certegy Issue.
 - (iv) In the event that Equifax receives a notice of deficiency from the IRS, or a similar notice from any other Tax Authority, and such notice relates exclusively to one or more Certegy Issues (none of which are Equifax Issues) and does not relate to an Equifax Issue then:
 - (A) upon receiving a written request from Certegy, given no later than a date reasonably necessary to permit preparation and timely filing of a petition in the United States Tax Court for redetermination of the deficiency, or a court of similar jurisdiction with respect to Taxes imposed by any other Tax Authority, Equifax shall timely file such petition (at Certegy's sole expense); or
 - (B) If (1) Certegy does not request Equifax to file a petition for redetermination of the deficiency pursuant to subsection 6.1(c)(iv)(A) hereof, (2) Certegy requests that Equifax file a claim for refund, and (3) Certegy provides Equifax with sufficient funds to pay the deficiency relating to the Certegy Issue, then Equifax (at Certegy's sole expense) shall file a claim for refund thereof and,

if the claim is denied, bring an action in a court of competent jurisdiction seeking such refund.

- (C) In the event that a judgment of the United States Tax Court or other court of competent jurisdiction results in an adverse determination with respect to the Certegy Issue, then Certegy shall have the right to cause Equifax to appeal from such adverse determination at Certegy's sole expense.
- (D) Certegy and its representatives, at Certegy's sole expense, shall be entitled to the extent permitted by law to participate in (1) all conferences, meetings, or proceedings with any Tax Authority, the subject matter of which is a Certegy Issue (that is not an Equifax Issue), and (2) all appearances before any court, the subject matter of which is a Certegy Issue (that is not an Equifax Issue). The right to participate referred to in this subsection 6.1(c)(iv)(D) hereof shall include the submission and content of documentation, memoranda of fact and law and briefs, the conduct of oral arguments or presentations, the selection of witnesses, and the negotiation of stipulations of fact with respect to a Certegy Issue (that is not an Equifax Issue).
- (v) If the proposed adjustment relating to a Certegy Issue is also an Equifax Issue (or if the proposed adjustment relates solely to an Equifax Issue that is not a Certegy Issue), then Equifax shall be fully responsible for the conduct of the Tax Controversy, including matters pertaining to any Certegy Issue, but Equifax shall use reasonable efforts to involve Certegy in the conduct of the Tax Controversy insofar as it relates to any Certegy Issue. Equifax shall notify Certegy in advance of any such proceedings and, at its own expense, Certegy may attend all conferences, meetings, or proceedings with any Tax Authority, the subject matter of which is or includes any Certegy Issue. Certegy shall use all reasonable efforts to assist Equifax in resisting any deficiency assertions by any Tax Authority relating to any such Certegy Issue.
- (d) Certegy shall notify Equifax promptly of any communication with the IRS or other Tax Authority relating in whole or in part to any Equifax Issue.
 - (i) Equifax shall have 30 days after receipt of such notice from Certegy within which to object to the proposed adjustment relating to an Equifax Issue (that is not a Certegy Issue). If Equifax does not notify Certegy within such 30 day period that it objects to the proposed adjustment, then subsections 6.1(d)(ii) through 6.1(d)(v) below shall not apply, and Certegy shall have exclusive control over all stages of the Tax Controversy, including full authority to determine whether and in what manner to contest or compromise the issue, unless and until Equifax so notifies Certegy.
 - (ii) If Equifax notifies Certegy that it objects to the proposed adjustment relating to an Equifax Issue (that is not a Certegy Issue), then Certegy shall not thereafter consent to the adjustment or compromise of such Equifax

Issue without the consent of Equifax, but shall cooperate with Equifax to resolve the Equifax Issue on a basis acceptable to Equifax. Prior to the issuance of a notice of proposed adjustment or similar stage in the proceedings, however, Certegy shall be responsible for the conduct of the audit, including matters pertaining to such Equifax Issue. Certegy shall notify Equifax in advance of any conferences, meetings, and proceedings pertaining to the audit and, at its own expense, Equifax shall have the right to attend all such proceedings with any Tax Authority, the subject matter of which is or includes such Equifax Issue.

- (iii) Upon the issuance of a notice of proposed adjustment or similar stage in the proceedings, Equifax shall assume the conduct of all further proceedings, with counsel selected by it, at Equifax's sole expense, insofar as the proceedings relate to an Equifax Issue (that is not a Certegy Issue), and thereafter Certegy and Equifax shall jointly be responsible for the conduct of proceedings to contest such Equifax Issue.
- (iv) In the event that Certegy receives a notice of deficiency from the IRS, or a similar notice from any other Tax Authority, and such notice relates exclusively to one or more Equifax Issues (none of which are Certegy Issues) and does not relate to a Certegy Issue then:
- (A) upon receiving a written request from Equifax, given no later than a date reasonably necessary to permit preparation and timely filing of a petition in the United States Tax Court for redetermination of the deficiency, or a court of similar jurisdiction with respect to Taxes imposed by any other Tax Authority, Certegy shall timely file such petition (at Equifax's sole expense); or
- (B) If (1) Equifax does not request Certegy to file a petition for redetermination of the deficiency pursuant to subsection 6.1(d)(iv)(A) hereof, (2) Equifax requests that Certegy file a claim for refund, and (3) Equifax provides Certegy with sufficient funds to pay the deficiency relating to the Equifax Issue, then Certegy (at Equifax's sole expense) shall file a claim for refund thereof and, if the claim is denied, bring an action in a court of competent jurisdiction seeking such refund.
- (C) In the event that a judgment of the United States Tax Court or other court of competent jurisdiction results in an adverse determination with respect to the Equifax Issue, then Equifax shall have the right to cause Certegy to appeal from such adverse determination at Equifax's sole expense.
- (D) Equifax and its representatives, at Equifax's sole expense, shall be entitled to the extent permitted by law to participate in (1) all conferences, meetings, or proceedings with any Tax Authority, the subject matter of which is an Equifax Issue (that is not a Certegy Issue), and (2) all appearances before any court, the subject matter of which is an Equifax Issue (that is not a Certegy Issue). The right to participate referred to in this subsection 6.1(d)(iv)(D) shall include the submission and content of documentation, memoranda of fact

and law and briefs, the conduct of oral arguments or presentations, the selection of witnesses, and the negotiation of stipulations of fact with respect to an Equifax Issue (that is not a Certegy Issue).

(v) If the proposed adjustment relating to an Equifax Issue is also a Certegy Issue (or if the proposed adjustment relates solely to a Certegy Issue that is not an Equifax Issue), then Certegy shall be fully responsible for the conduct of the Tax Controversy, including matters pertaining to any Equifax Issue, but Certegy shall use reasonable efforts to involve Equifax in the conduct of the Tax Controversy insofar as it relates to any Equifax Issue. Certegy shall notify Equifax in advance of any such proceedings and, at its own expense, Equifax may attend all conferences, meetings, or proceedings with any Tax Authority, the subject matter of which is or includes any Equifax Issue. Certegy shall use all reasonable efforts to assist Equifax in resisting any deficiency assertions by any Tax Authority relating to any such Equifax Issue.

6.2 Cooperation.

- (a) Equifax and Certegy agree to afford full cooperation to one another and to their respective representatives, if any, in any Tax Controversy involving:
 - (i) any Tax Return filed or required to be filed by or for any member of the Equifax Group or the Certegy Group for any Pre-Distribution Period, or
 - (ii) any item or issue affecting Equifax's or Certegy's potential liability hereunder.
 - (b) Such cooperation shall include, but not by way of limitation:
 - (i) preparing responses to information requests by any Tax Authority;
 - (ii) making available books, records and other documentation (including, but not by way of limitation, working papers and schedules) relevant to such proceeding, and systems support for documentation furnished in electronic form;
 - (iii) making directors, officers, or employees available to appear in person for interview or for testimony;
 - (iv) making employees available on a mutually convenient basis to provide additional information and explanation of materials provided hereunder;
 - (v) executing powers of attorney, tax information authorizations and any other necessary or appropriate authorizations;

- (vi) executing agreements with the Tax Authority or other documents reasonably necessary or appropriate for the settlement or pursuit of the contest of such issue; and
- (vii) doing whatever is reasonable under the circumstances to assist the other party in proving that a transaction on or after the Distribution Date does not give rise to Additional Restructuring Taxes.
- 6.3 Record Retention. The parties, on behalf of themselves and the members of their respective Affiliated Groups, agree to retain all books, records, returns, schedules, documents and all material papers or relevant items of information for periods prior to the Distribution Date for the later of (a) seven (7) years or (b) the full period of the applicable statute of limitations, including any extensions thereof. If, under legislation enacted after the date of this Agreement, the statute of limitations with respect to a transaction does not begin to run until the IRS or other Tax Authority is notified of the transaction, then the statute of limitations for purposes of clause (b) of this Section 6.3 shall also not begin to run until such notification is given.

ARTICLE VII PAYMENTS

- 7.1 Payments in General. Any amount required to be paid by one party to the other pursuant to this Agreement (other than the payments described in Sections 5.5 and 5.6 and subsection 6.1(c)(iv)(B) and 6.1(d)(iv)(B) shall be paid in immediately available funds within thirty (30) days after written demand therefor from the other party given after a Final Determination of the amount thereof
- 7.2 Treatment of Payments. In absence of any change in Tax treatment under the Code or other applicable Tax law, (a) any payments made pursuant to this Agreement by Certegy to Equifax shall be reported for Tax purposes by the parties as a distribution from Certegy to Equifax occurring on the Distribution Date immediately before the Distribution; and (b) any payments made pursuant to this Agreement by Equifax to Certegy shall be reported for Tax purposes by the parties as a contribution to the capital of Certegy by Equifax occurring on the Distribution Date immediately before the Distribution.
- 7.3 Interest on Late Payments. Any amount payable under this Agreement by one party to another party shall, if not paid within ten (10) business days after the due date specified in this Agreement, bear interest from such due date until the date paid at the applicable Federal "short term rate" as defined in section 6621 of the Code in effect on the due date.
- $7.4\,$ Notice. Equifax and Certegy shall give each other prompt notice of any payment that may be due under this Agreement.
- 7.5 Tax Items. Except to the extent already provided for in this Agreement, the amount of any indemnification payment required hereunder shall be reduced to take into account the Tax Benefit, if any, allowable to the indemnified party resulting from the event giving rise to such indemnification payment and shall be increased to take into account additional Taxes, if any, incurred by the indemnified party resulting from the receipt of such indemnification

payment and any additional payment required by this section. The parties will cooperate with each other in good faith to determine the amounts described in this section.

ARTICLE VIII ADMINISTRATIVE PROVISIONS

- 8.1 Interest. Except as expressly provided herein, no obligation to pay or right to collect interest or other amounts shall arise by virtue of this Agreement.
- 8.2 Agency. It is understood and acknowledged that in accordance with Regulations section 1.1502-77, Equifax, as the common parent, is the agent for the members of the Affiliated Group of which Equifax is the common parent (including all members of the Certegy Group with respect to taxable years beginning before the Distribution Date) with respect to all matters referred to therein.
- 8.3 Expenses. Except as otherwise expressly provided herein, each party to this Agreement hereby agrees to be responsible for all of the costs and expenses which it may incur in carrying out its duties hereunder.

ARTICLE IX DISPUTE RESOLUTION

Any controversy or claim between the parties arising out of or relating to this Agreement, or the breach hereof not resolved in the normal course of business shall be resolved pursuant to Section 15.10 of the Distribution Agreement, which provisions are hereby incorporated herein by this reference thereto.

ARTICLE X MISCELLANEOUS

- 10.1 Entire Agreement. This Agreement constitutes the entire understanding of the parties hereto with respect to the subject matter hereof, superseding all negotiations, prior discussions and prior agreements and understandings relating to such subject matter.
- 10.2 Severability. The provisions of this Agreement are severable and should any provision hereof be void, voidable or unenforceable under any applicable law, such provision shall not affect or invalidate any other provision of this Agreement, which shall continue to govern the relative rights and duties of the parties as though such void, voidable or unenforceable provision were not a part hereof.
- 10.3 Amendment and Waiver. This Agreement may not be altered or amended, nor may any rights hereunder be waived, except by an instrument in writing executed by the party or parties to be charged with such amendment or waiver. No waiver of any terms, provision or condition of or failure to exercise or delay in exercising any rights or remedies under this Agreement, in any one or more instances, shall be deemed to be, or construed as, a further or continuing waiver of any such term, provision, condition, right or remedy or as a waiver of any other term, provision or condition of this Agreement.

- 10.4 Successors and Assigns. Except to the extent provided by operation of law or as provided herein, neither of the parties hereto may assign its rights or delegate any of its duties under this Agreement without the prior written consent of each other party. This Agreement shall be binding upon, and shall inure to the benefit of, the successors and assigns of each party. If one or more persons acquires all or substantially all of the assets of Equifax or Certegy, Equifax and Certegy each agree that, as a condition to the closing of such acquisition, such person or persons must agree to indemnify the nonacquired party for any Restructuring Taxes incurred by that party as a result of such acquisition.
- 10.5 Term. This Agreement shall commence on the date of execution indicated above and shall continue in effect until otherwise agreed to in writing by the parties or their successors and assigns.
 - 10.6 Guarantee of Performance; Rights Confined to Parties.
- (a) Each party hereby guarantees the performance of all actions, covenants, agreements, and obligations provided under this Agreement of each of its subsidiaries. Each party shall, upon the written request of the other party, cause any of its subsidiaries to formally execute this Agreement.
- (b) Nothing contained in this Agreement, express or implied, is intended to confer any benefits, rights or remedies upon any person or entity other than members of the Equifax Group and the Certegy Group.
- 10.7 Notices. All notices and communications under this Agreement shall be in writing and shall be deemed to have been given (a) when received, if such notice or communication is delivered by facsimile, hand delivery or overnight courier, and, (b) three (3) business days after mailing if such notice or communication is sent by United States registered or certified mail, return receipt requested, first class postage prepaid. All notices and communications, to be effective, must be properly addressed to the party to whom the same is directed at its address as follows:

If to Equifax, to: Equifax Inc.

1550 Peachtree Street, N.W. Atlanta, Georgia 30309

Attention: Phillip J. Mazzilli, Chief Financial Officer

with a copy to: Equifax Inc.

1550 Peachtree Street, N.W. Atlanta, Georgia 30309

Attn: Kent E. Mast, General Counsel

If to Certegy, to: Certegy Inc.

555 NorthPointe Centre East Alpharetta, Georgia 30022

Attn: Bruce S. Richards, Corporate

Vice President, General Counsel and

Secretary

with a copy to: Co

Certegy Inc. P.O. Box 349

Alpharetta, Georgia 30009 Attn: Michael T. Volkommer,

Corporate Vice President and Chief

Financial Officer

Either party may, by written notice delivered to the other party in accordance with this Section 10.7, change the address to which delivery of any notice shall thereafter be made.

- 10.8 Further Assurances and Consents. In addition to the actions specifically provided for elsewhere in this Agreement, each of the parties hereto will use its reasonable efforts to (a) execute and deliver such further instruments and documents and take such other actions as any other party may reasonably request in order to effectuate the purposes of this Agreement and to carry out the terms hereof and (b) take, or cause to be taken, all actions, and do, or cause to be done, all things, reasonably necessary, proper or advisable under applicable laws, regulations and agreements or otherwise to consummate and make effective the transactions contemplated by this Agreement; provided that no party hereto shall be obligated to pay any consideration therefor (except for filing fees and other similar charges) to any third party from whom such consents, approvals and amendments are requested or to take any action or omit to take any action if the taking of or the omission to take such action would be unreasonably burdensome to the party or its Group or the business thereof.
- 10.9 Headings. The Article and Section headings set forth in this Agreement are included for administrative, organizational and convenience purposes, and are not intended to affect the meaning of the provisions set forth in this Agreement or to be used in the interpretation of this Agreement.
- 10.10 Governing Law. This Agreement shall be construed in accordance with, and governed by, the laws of the State of Georgia, without regard to the conflicts of law rules of such state.
- 10.11 Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original instrument, but all of which together shall constitute but one and the same agreement.

10.12 Prior Tax Sharing Agreements. This Agreement shall supersede any and all tax sharing and indemnification (or similar) agreements between any of the members of the Equifax Group, on the one hand, and any of the members of the Certegy Group, on the other hand.

[signatures appear on following page]

IN WITNESS	WHEREOF, the	e parties here	o have executed	l and delivered thi	ĹS
Agreement as of	the day and	year first abo	ve written.		

EQUIFAX INC.
By:
CERTEGY INC.
By:

EXHIBIT A Foreign Restructuring

- Transfer of shares of Acrofax, Inc. from Equifax Credit Information Services, Inc. to Equifax and transfer of shares of Telecredit Canada Inc. from Equifax Canada Inc. to Equifax, including all interim transactions related to such transfer (the "Canadian Restructuring").
- 2. Transfer of shares of Procard S.A. from Equifax de Chile, S.A. to Payment Chile, S.A. ("Payment Chile") and transfer of shares of Payment Chile to Payment South America Holdings, Inc. ("Payment South America"), including all interim transactions related to such transfers (the "Chilean Restructuring").
- 3. Transfer of shares of Unnisa-Solucoes em Meios de Pagamento Ltda. ("Unnisa"), Equifax Cayman Islands, Ltd. and Partech Ltda. ("Partech") from Equifax do Brasil Ltda to Payment Brasil Holdings Ltda. ("Payment Brazil") and transfer of shares of Payment Brazil to Payment South America, including all interim transactions related to such transfers (the "Brazilian Restructuring").
- 4. Transfer of certain assets of Equifax plc related to U.K. Payment Services Business and stock of Transax plc and Equifax Card Solutions Ltd. to Payment U.K. Ltd. ("Payment U.K."), and transfer of shares of Payment U.K. to Equifax, including all interim transactions related to such transfers (the "U.K. Restructuring").

$\begin{array}{c} {\sf EXHIBIT} \ {\sf B} \\ {\sf Contemplated} \ {\sf Restructuring} \ {\sf Taxes} \end{array}$

- ${\bf 1.} \quad {\bf Canadian \ income/withholding \ Tax \ on \ Canadian \ Restructuring \ transactions.}$
- 2. Income/capital gain Tax on Brazilian Restructuring transactions.
- 3. Income/capital gain Tax on Chilean Restructuring transactions.

EXHIBIT 10.3

EMPLOYEE BENEFITS AGREEMENT	
-----------------------------	--

between

EQUIFAX INC.

and

CERTEGY INC.

Dated as of the _____ day of ______, 2001

TABLE OF CONTENTS

ARTICLE I	DEFINITIONS AND REFERENCES	2
1.01 Defini	itions	2
(a)	Action	2
(b)	Agreement	2
(c)	ASO Contract	2
(d)	Award	2
(e)	Bulk Asset Transfer	2
(f)	Certegy Business	2
(g)	Certegy Common Stock	2
(h)	Certegy Group	2
(i)	Close of the Distribution Date	2
(j)	Code	3
(b)	Compensation Committee	3
(1)	Conversion Formula	3
(m)	Deferral Programs	3
(m)	Distribution	3
(0)	Distribution Agreement	3
(p)	Distribution Date	3
(p)	ERISA	3
(q) (r)	Equifax Executive	3
(s)	Equifax Group	4
(t)	Executive Life Plan	4
` ,	Executive Life Fight	4
(u)		4
(v)	Executive Programs	4
(w)	Foreign Plan	-
(x)	Governmental Authority	4
(y)	Group Insurance Policy	5
(z)	Health and Welfare Plans	5
(aa)	HMO	5
(bb)	HMO Agreements	5
(cc)	Immediately after the Distribution Date	5
(qq)	Individual Agreement	5
(ee)	Initial Asset Transfer	5
(ff)	Liabilities	5
(gg)	Long-Term Incentive Plan	6
(hh)	Material Feature	6
(ii)	Participating Company	6
(jj)	Pension Plan	6
(kk)	Pension Plan Spinoff Date	6
(11)	Pension Trust	6
(mm)	Person	7
(nn)	Plan	7
(00)	Plan Termination Liability	7
(pp)	Reimbursement Plans	7
(qq)	Savings Plan	7
(rr)	Short-Term Incentive Plan	7
(ss)	Stock Incentive Plan	7
(tt)	Subsequent Asset Transfer	8
(uu)	Subsidiary	8
(vv)	Transferred Individual	8
(ww)	Transition Support Agreement	10

	(xx) VEBA	10
1.02	References	10
ARTICLE	E II GENERAL PRINCIPLES	11
2.01	Assumption of Liabilities	11
	Certegy Group Participation in Equifax Plans	11
2.02	(a) Participation in Equifax Plans	11
	(b) Equifax's General Obligations as Plan Sponsor	11
	(a) Equitax 5 General Obligations as Flant Sponsor.	
	(c) Certegy's General Obligations as Participating Company	12
	(d) Termination of Participating Company Status	12
2.03	Establishment of The Certegy Plans	12
2.04	Terms of Participation by Transferred Individuals	12
ARTICLE	E III DEFINED BENEFIT PLANS	14
3.01	Establishment of Mirror Pension Trusts	14
3 02	Assumption of Pension Plan and Liabilities and Allocation of Interests in the Equifax Pension	
3.02	Trust	14
	(a) Accomption of Liebilities by Contour Paging Plan	
	(a) Assumption of Liabilities by Certegy Pension Plan	14
	(b) Asset Allocations and Transfers	14
	(c) Valuation of Venture Capital Assets	16
	(d) Pension Plan Transition Period	16
	(e) Negotiation if There Is No Pension Plan Spinoff Date	17
	(f) Allocation of Code Section 401(h) Opportunity	17
3 03	Governmental Compliance	17
0.00	GOVERNMENTAL COMPTIANCE	
ARTICLE	E IV DEFINED CONTRIBUTION PLANS	18
4.01	Savings Plan	18
	(a) Savings Plan Trust	18
	(b) Savings Plan Participation	18
	(c) Assumption of Liabilities and Transfer of Assets	18
	(d) Specific Stock Funds in the Equifax and Certegy Savings Plan	19
	(e) Miscellaneous Funds	19
	(f) Transfer of Assets Concerning Matching Contribution Account	19
4 00		
4.02	Non-Employer Stock Fund Procedures	20
	(a) Application of Procedures	20
	(b) Right of First Refusal and Notice Procedures	20
	(c) Procedures After Notice is Given	20
ARTICLE	E V HEALTH AND WELFARE PLANS	22
5.01	Assumption of Health and Welfare Plan Liabilities	22
0.01	(a) Assumption by Certegy	22
	(b) Certain Audit Procedures with Respect to Health and Welfare Plans	22
F 00		
	Establishment of Certegy VEBA	23
5.03	VEBA Asset Transfers	23
	(a) Aggregate Amount to be Transferred	23
	(b) Timing of Asset Transfers / Interim Period	23
5.04	Contributions to, Investments of, and Distributions from VEBAs	24
	Vendor Contracts	24
2.00	(a) ASO Contracts, Group Insurance Policies, HMO Agreements, and Letters of Understanding	24
	(b) Payment and Effect of Change in Rates	25
- 00		
	Equifax Short-Term Disability Plan	25
5.07	Retiree Health and Life Insurance Benefits	26
	(a) Establishment of Certegy Plan	26
	(b) Equifax Retiree Health and Life Benefits	26

5.08	COBRA and HIPAA	26
5.09	Leave of Absence Programs	26
5.10	Post-Distribution Transitional Arrangements	27
	(a) Continuance of Elections, Co-Payments, and Maximum Benefits	27
	(b) Administration	27 28
5 11	Severance Pay Plan	28
3.11	(a) Transferred Individuals	28
	(b) Additional Liability Assumed by Certegy	29
5.12	Application of Article V to the Certegy Group	29
ARTICLE	VI EXECUTIVE PROGRAMS	30
6.01	Assumption of Obligations	30
6.02	Short-Term Incentive Awards	30
6.03	Long-Term Incentive Plan	30
	(a) Plan 19 (1999-2001)	30
0.04	(b) Plan 20 (2000-2002) and 2001 Awards	31
6.04		31 31
	(a) Stock Options	32
6.05	Deferred Compensation Plan	32
6.06	Executive Life and Supplemental Retirement Benefit Plan	32
0.00	(a) Plan Agreements	32
	(b) Executive Life Trust	32
6.07	Global Stock Purchase Plan Loan Program	33
6.08	Change In Control Agreements	33
6.09	Restricted Stock	33
6.10	Waiver of Dividend Relating to Certain Trusts	33
6.11	Automobile Program	33
ARTICLE	VII GENERAL	34
7.01	Payment of and Accounting Treatment for Expenses and Balance Sheet Amounts	34
	(a) Expenses	34
	(b) Balance Sheet Amounts	34
	Sharing of Participant Information	34
7.03	Restrictions on Extension of Option Exercise Periods, Amendment or Modification of Option Terms	
	and Conditions	34
7.04	Non-Solicitation of Employees/Others	35
7.05 7.06	Plan Audits	35 35
7.00	(a) Audit Rights with Respect to the Allocation or Transfer of Plan Assets	35
	(b) Audit Rights With Respect to Information Provided	36
	(c) Audits Regarding Vendor Contracts	36
	(d) Audit Assistance	37
7.07	Beneficiary Designations/Release of Information/Right to Reimbursement	37
7.08	Requests for Internal Revenue Service Rulings and United States Department of Labor Opinions and	
	Satisfaction of Obligations Arising From Voluntary Compliance Programs	37
	(a) Cooperation	37
	(b) Applications	37
7.09	Fiduciary and Related Matters	38
7.10	No Third-Party Beneficiaries; Non-Termination of Employment	38
7.11 7.12	Collective Bargaining	38 39
7.12	Foreign Plans	39
1.13	Lai etali etalis.	39

7.14	Effect If Distribution Does Not Occur	36
7.15	Relationship of Parties	39
7.16	Affiliates	39
7.17	Dispute Resolution	38
7.18	Indemnification	39
7.19	W-2 Matters	41
7.20	Confidentiality	41
7.21	Notices	42
7.22	Interpretation Severability Governing Law/Execution	42
7.23	Severability	42
7.24	Governing Law/Execution	42
Appendi	X A CERTEGY EXECUTIVE PROGRAMS	44
Appendi	x B CERTEGY HEALTH AND WELFARE PLANS	45
Appendi	x C FOREIGN PLANS	46
Annandiy	V D CHANGE IN CONTROL AGREEMENTS	10

EMPLOYEE BENEFITS AGREEMENT

THIS EMPLOYEE BENEFITS AGREEMENT, dated as of the _____ day of ______, 2001, is by and between Equifax Inc., a Georgia corporation ("Equifax" or the "Corporation"), and Certegy Inc., a Georgia corporation ("Certegy").

WHEREAS, Equifax's Board of Directors has determined that separation from the Corporation of substantially all of the Corporation's payment services assets and businesses and public ownership of such assets and businesses is in the best interests of Equifax and its shareholders; and

WHEREAS, Equifax is consolidating the assets and operations of substantially all of the payment services businesses owned by it and its Subsidiaries into Certegy and its Subsidiaries; and

WHEREAS, Equifax intends to accomplish the separation of Certegy through a distribution of the stock of Certegy to the shareholders of Equifax that is intended to be tax free pursuant to section 355 of the Internal Revenue Code of 1986, as amended (the "Distribution"); and

WHEREAS, Equifax and Certegy have entered into a Distribution Agreement and Plan of Reorganization and Distribution, dated as of ________, 2001 (the "Distribution Agreement"), and several other agreements that will govern certain matters relating to the Distribution and the relationship of Equifax and Certegy and their respective Subsidiaries following the Distribution; and

WHEREAS, pursuant to the Distribution Agreement, Equifax and Certegy have agreed to enter into this Agreement for the purpose of allocating assets, liabilities, and responsibilities with respect to certain employee compensation and benefit plans and programs between them.

NOW, THEREFORE, in consideration of the mutual promises contained herein and in the Distribution Agreement, the parties agree as follows:

Article I Definitions and References

1.01 Definitions

For purposes of this Agreement, capitalized terms used (other than the formal names of Equifax Plans (as defined below)) and not otherwise defined shall have the respective meanings assigned to them below or as assigned to them in the Distribution Agreement (as defined above):

(a) Action

"Action" means any demand, action, cause of action, suit, countersuit, arbitration, inquiry, proceeding, or investigation by or before any Governmental Authority or any arbitration or mediation tribunal, pending or threatened, known or unknown.

(b) Agreement

"Agreement" means this Employee Benefits Agreement, including all the attached Appendices.

(c) ASO Contract

"ASO Contract" means an administrative services contract, related prior practice, or related understanding with a third-party administrator that pertains to any Equifax Health and Welfare Plan or any Certegy Health and Welfare Plan.

(d) Award

"Award" means a long-term or short-term award under a Long-Term Incentive Plan or a Short-Term Incentive Plan or, as the context or facts may require, any other award under another incentive or special bonus, incentive, or award program or arrangement.

(e) Bulk Asset Transfer

"Bulk Asset Transfer" is defined in Section 3.02(b)(2).

(f) Certegy Business

"Certegy Business" has the meaning given that term under the Distribution Agreement.

(g) Certegy Common Stock

(h) Certegy Group

"Certegy Group" has the meaning given that term under the Distribution Agreement. $% \label{eq:continuous}%$

(i) Close of the Distribution Date

"Close of the Distribution Date" means $11:59:59\ P.M.$, Eastern Time, on the Distribution Date.

(j) Code

"Code" means the Internal Revenue Code of 1986, as amended, or any successor federal income tax law. Reference to a specific Code provision also includes any proposed, temporary, or final regulation in force under that provision.

(k) Compensation Committee

(1) Conversion Formula

"Conversion Formula" means the appropriate formula which shall be applied in adjusting the exercise price and award size of Equifax options under the Equifax Stock Incentive Plans and in determining the exercise price and number of Certegy options under the Certegy Stock Incentive Plans. The Conversion Formula shall be based on the closing per share prices of Equifax common stock (with a due bill) and Certegy Common Stock (on a when-issued basis) as traded on the New York Stock Exchange on the last trading day immediately preceding the Distribution or such other prices as jointly agreed upon by Certegy and Equifax, and shall be determined and applied in such a manner as to maintain (1) the aggregate spread of such options, which is the difference between the exercise price per share of Equifax Common Stock covered by the option and the price per share of Equifax Common Stock (with a due bill) immediately preceding the Distribution, multiplied by the total number of shares covered by the option to the price per share of Equifax Common Stock (with a due bill) immediately preceding the Distribution.

(m) Deferral Programs

"Deferral Programs," when immediately preceded by "Equifax" means the Equifax Deferred Compensation Plan. When immediately preceded by "Certegy," "Deferral Programs" means the executive deferred compensation plan to be established or maintained by Certegy pursuant to Section 2.03.

(n) Distribution

"Distribution" has the meaning given that term under the Distribution $\ensuremath{\mathsf{Agreement}}.$

(o) Distribution Agreement

"Distribution Agreement" is defined in the preamble of this Agreement.

(p) Distribution Date

"Distribution Date" has the meaning given that term under the Distribution $\mbox{\sc Agreement.}$

(q) ERISA

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended. Reference to a specific provision of ERISA also includes any proposed, temporary, or final regulation in force under that provision.

(r) Equifax Executive

"Equifax Executive" means an employee or former employee of a member of the Equifax Group or a member of the Certegy Group who, as of the Close of the Distribution Date, is or was eligible to participate in or receive a benefit under any Equifax Executive Program.

(s) Equifax Group

"Equifax Group" has the meaning given that term under the Distribution Agreement.

(t) Executive Life Plan

"Executive Life Plan," when immediately preceded by "Equifax," means the Executive Life and Supplemental Retirement Benefit Plan maintained by the Equifax Group or when immediately preceded by "Certegy," means a plan to be established or which is maintained pursuant to Section 2.03 by the Certegy Group that corresponds to the Equifax Executive Life Plan, including, in each case, any and all agreements, documents, and policies relating to such Executive Life Plan.

(u) Executive Life Trust

"Executive Life Trust," when immediately preceded by "Equifax," means the trust evidenced by the Equifax Grantor Trust Agreement, dated as of February 4, 2000, and currently associated with the Equifax Executive Life Plan. When immediately preceded by "Certegy," "Executive Life Trust" means the grantor trust to be established or maintained by Certegy pursuant to Section 6.05 that corresponds to the Equifax Executive Life Trust.

(v) Executive Programs

"Executive Programs," when immediately preceded by "Equifax" means the executive benefit and nonqualified plans, programs, and arrangements established, maintained, agreed upon, or assumed by a member of the Equifax Group for the benefit of employees and former employees of members of the Equifax Group before the Close of the Distribution Date. When immediately preceded by "Certegy," "Executive Programs" means the executive benefit plans and programs to be established or which are maintained by Certegy pursuant to Section 2.03 that correspond to the respective Equifax Executive Programs, including those plans and programs listed in Appendix A.

(w) Foreign Plan

"Foreign Plan," when immediately preceded by "Equifax," means a Plan maintained by the Equifax Group or when immediately preceded by "Certegy," a plan to be established or which is maintained by the Certegy Group, in either case for the benefit of employees who are compensated under a payroll which is administered outside the 50 United States, its territories and possessions, and the District of Columbia, including those Plans described in Appendix C.

(x) Governmental Authority

"Governmental Authority" means any federal, state, local, foreign, or international court, government, department, commission, board, bureau, agency, official, or other regulatory, administrative, or governmental authority, including the Department of Labor, the Securities and Exchange Commission, the Internal Revenue Service, and the Pension Benefit Guaranty Corporation.

(y) Group Insurance Policy

"Group Insurance Policy" means a group insurance policy issued in connection with any Equifax Health and Welfare Plan, or any Certegy Health and Welfare Plan, as applicable.

(z) Health and Welfare Plans

"Health and Welfare Plans," when immediately preceded by "Equifax" means the health and welfare benefit plans, programs, and policies (including the Reimbursement Plans) which are sponsored by Equifax. When immediately preceded by "Certegy," "Health and Welfare Plans" means the benefit plans, programs, and policies (including the Reimbursement Plans) corresponding to those plans, programs, and policies sponsored by Equifax as of the Distribution Date, including those plans, programs, and policies listed in Appendix B to this Agreement which will be sponsored by a member of the Certegy Group Immediately after the Distribution Date.

(aa) HMO

"HMO" means a health maintenance organization that provides benefits under the Equifax Health and Welfare Plans or the Certegy Health and Welfare Plans, as applicable.

(bb) HMO Agreements

"HMO Agreements" means contracts, letter agreements, practices, and understandings with HMOs that provide medical, dental, or vision services under the Equifax Health and Welfare Plans and the Certegy Health and Welfare Plans, as applicable.

(cc) Immediately after the Distribution Date

"Immediately after the Distribution Date" means 12:00 A.M., Eastern Time, on the day after the Distribution Date.

(dd) Individual Agreement

"Individual Agreement" means an individual contract or agreement (whether written or unwritten) entered into between a member of the Equifax Group or a member of the Certegy Group and any employee or individual who will be an employee of, or otherwise assigned to, the Certegy Group Immediately after the Distribution Date that establishes the right of such individual to special compensation or benefits, special bonuses, supplemental pension benefits, hiring bonuses, loans, guaranteed payments, special allowances, tax equalization payments, special expatriate compensation payments, disability benefits, or other forms of compensation and benefits, or that provides benefits similar to those identified in Appendix A.

(ee) Initial Asset Transfer

"Initial Asset Transfer" is defined in Section 3.02(b)(2).

(ff) Liabilities

"Liabilities" means any and all losses, claims, charges, debts, premiums, demands, actions, costs, and expenses (including any current or future benefit payments or other

entitlements, and administrative and related costs and expenses of any Plan, program, service or consulting agreement, or arrangement), of any nature whatsoever, whether absolute or contingent, matured or unmatured, liquidated or unliquidated, accrued or unaccrued, known or unknown, whether or not imposed or determined by a court, whenever arising.

(gg) Long-Term Incentive Plan

"Long-Term Incentive Plan," when immediately preceded by "Equifax" means the Equifax Inc. Key Management Long-Term Incentive Plan, the Equifax Inc. 1988 Performance Share Plan for Officers, and any other long-term incentive plans established or maintained by a member of the Equifax Group. When immediately preceded by "Certegy," "Long-Term Incentive Plan" means the long-term incentive plan to be established or assumed by Certegy pursuant to Section 2.03.

(hh) Material Feature

"Material Feature" means any feature of a Plan that could reasonably be expected to be of material importance to the sponsoring employer or the participants and beneficiaries of the Plan, which could include, depending on the type and purpose of the particular Plan, the class or classes of employees eligible to participate in such Plan, the nature, type, form, source, and level of benefits provided by the employer under such Plan and the amount or level of contributions, if any, required or permitted to be made by participants (or their dependents or beneficiaries) to such Plan.

(ii) Participating Company

"Participating Company" means any Person (other than an individual) that is participating in a Plan sponsored by a member of the Equifax Group or a member of the Certegy Group, as the context requires.

(jj) Pension Plan

"Pension Plan," when immediately preceded by "Equifax," means the Equifax Inc. United States Retirement Income Plan. When immediately preceded by "Certegy," "Pension Plan" means the plan to be established or maintained by Certegy pursuant to Section 2.03 that corresponds to the Equifax Pension Plan.

(kk) Pension Plan Spinoff Date

"Pension Plan Spinoff Date" shall have the meaning given that term under Section 3.02(b)(1).

(ll) Pension Trust

"Pension Trust," when immediately preceded by "Equifax," means the trust evidenced by the Equifax Trust Agreement, as amended from time to time, and currently associated with the Equifax Pension Plan. When immediately preceded by "Certegy," "Pension Trust" means the trust to be established or maintained by Certegy pursuant to Section 3.01 that corresponds to the Equifax Pension Trust.

(mm) Person

"Person" means an individual, a general or limited partnership, a corporation, a trust, a joint venture, an unincorporated organization, a limited liability entity, any other entity, or any Governmental Authority.

(nn) Plan

"Plan," when immediately preceded by "Equifax" or "Certegy," means any plan, policy, program, payroll practice, on-going arrangement, contract, trust, insurance policy, or other agreement or funding vehicle, whether written or unwritten, providing benefits to employees or former employees of the Equifax Group or the Certegy Group, as applicable.

(00) Plan Termination Liability

"Plan Termination Liability" shall be calculated in accordance with Section 414(1) of the Code based on the assumptions used by the Pension Benefit Guaranty Corporation as of the Pension Plan Spinoff Date.

(pp) Reimbursement Plans

"Reimbursement Plans," when immediately preceded by "Equifax," means the Equifax Health Care Account Plan and the Equifax Dependent Care Account Plan, as applicable. When immediately preceded by "Certegy," "Reimbursement Plans" means the health care flexible spending account plan and the dependent care flexible spending account plan to be established or maintained by Certegy pursuant to Section 2.03 that corresponds to the corresponding Equifax Reimbursement Plan.

(qq) Savings Plan

"Savings Plan," when immediately preceded by "Equifax," means the Equifax Inc. 401(k) Retirement and Savings Plan. When immediately preceded by "Certegy," "Savings Plan" means the Certegy 401(k) plan to be established by Certegy pursuant to Section 2.03.

(rr) Short-Term Incentive Plan

"Short-Term Incentive Plan," when immediately preceded by "Equifax" means any short-term compensation, bonus, or incentive compensation programs established or maintained by the Equifax Group. When immediately preceded by "Certegy," "Short-Term Incentive Plan" means any short-term compensation, bonus, or incentive compensation programs to be established or maintained by Certegy pursuant to Section 2.03.

(ss) Stock Incentive Plan

"Stock Incentive Plan," when immediately preceded by "Equifax," means the Equifax Inc. 2000 Stock Incentive Plan, the Equifax Inc. Omnibus Stock Incentive Plan, the 1995 Employee Stock Incentive Plan and the 1993 Employee Stock Incentive Plan and any other stock-based incentive plan established or maintained by a member of the Equifax Group. When immediately preceded by "Certegy," "Stock Incentive Plan" means the stock incentive plans to be established or assumed by Certegy pursuant to Section 2.03.

(tt) Subsequent Asset Transfer

"Subsequent Asset Transfer" is defined in Section 3.02(b)(2).

(uu) Subsidiary

"Subsidiary" of any Person means any corporation or other organization, whether incorporated or unincorporated, of which at least a majority of the securities or interests having by the terms thereof ordinary voting power to elect at least a majority of the board of directors or others performing similar functions with respect to such corporation or other organization is, directly or indirectly, owned or controlled by such Person or by any one or more of its Subsidiaries, or by such Person and one or more of its Subsidiaries; provided,

however, that no Person that is not directly or indirectly wholly owned by any

other Person shall be a Subsidiary of such other Person unless such other Person controls, or has the right, power, or ability to control, that Person.

(vv) Transferred Individual

"Transferred Individual" means any individual who, as of the Close of the Distribution Date:

- (1) is actively employed by, or on a leave of absence (including those individuals receiving short-term disability benefits and those taking leave pursuant to the Family and Medical Leave Act of 1993, as amended) from, a member of the Certegy Group or the Certegy Business (or is deemed to be so employed or on leave in accordance with an agreement between Equifax and Certegy); or
- (2) is not actively employed by, nor on a leave of absence from, Equifax or a member of the Certegy Group or the Certegy Business, and:
 - (A) whose most recent (through the Close of the Distribution Date) active employment with Equifax or a past or present affiliate of Equifax was with an entity or a corporate division of the Certegy Business, and the predecessors of any such entities; or
 - (B) who otherwise is identified pursuant to a methodology approved by Equifax and Certegy, which methodology shall be consistent with the intent of the parties that former employees of Equifax or a past or present affiliate of Equifax and such other individuals who performed services for Equifax (including non-employees) will be aligned with the entity for which they most recently (through the Close of the Distribution Date) were employed or otherwise provided services.

Notwithstanding the prior sentence, the term "Transferred Individual" shall not include any individual who is otherwise designated in (2) above and who as of the Close of the Distribution Date (i) is receiving payments from the Equifax Pension Plan, (ii) has retiree health coverage from Equifax in effect or (iii) is receiving benefits under the Equifax Long-Term Disability Plan.

For purposes of this Agreement with respect to all Health and Welfare Plans (other than Plans providing retiree life insurance and health benefits), all Foreign Plans providing health and welfare benefits, and all Executive Programs, the term "Transferred Individual" shall only mean those individuals described in (1) above, together with those individuals that are related thereto as described below

An alternate payee under a qualified domestic relations order (within the meaning of Code (S) 414(p) and ERISA (S) 206(d)), alternate recipient under a qualified medical child support order (within the meaning of ERISA (S) 609(a)), beneficiary, or covered dependent, in each case, of an employee or former employee described in (1) or (2) above shall also be a Transferred Individual with respect to the interest of such alternate payee, alternate recipient, beneficiary, or covered dependent in that employee's or former employee's benefit under the applicable Plans. Such an alternate payee, alternate recipient, beneficiary, or covered dependent shall not otherwise be considered a Transferred Individual with respect to his or her own benefits under any applicable Plans, unless he or she is a Transferred Individual by virtue of either of subparagraphs (1) or (2) above. In addition, Equifax and Certegy may agree to designate any other individuals, or group of individuals, as Transferred Individuals.

Subject to the other provisions of this definition, an individual may be a Transferred Individual pursuant to this definition regardless of whether such individual is, as of the Distribution Date, alive, actively employed, on a temporary leave of absence from active employment, on layoff, terminated from employment, retired or on any other type of employment, post-employment, or independent contract status relative to Equifax or Certegy or to an Equifax or Certegy Plan. Solely for purposes of assigning any Liabilities from Equifax to Certegy under this Agreement, an individual may be a Transferred Individual regardless of whether the individual is, or was, a common law employee, independent contractor, temporary employee, temporary service worker, consultant, freelancer, agency employee, leased employee, on-call worker, incidental worker, or nonpayroll worker of a member of the Equifax Group or of the Certegy Group or in any other employment, non-employment, or retainer arrangement or other relationship with any member of the Equifax Group or the Certegy Group. Transferred Individual includes any individual who is on an international assignment whether paid on a U.S. payroll or a payroll outside the U.S. if such individual otherwise falls within any of the above categories.

Notwithstanding anything to the contrary in this definition, the term Transferred Individual under this Agreement shall not include any individual who was otherwise directly transferred from former employment with a member of the Equifax Group to employment with any other Person (under an agreement whereby such Person assumed full liability with respect to all claims and rights related to such individual's pre-transfer employment), unless such individual has been re-employed as an active employee of a member of the Equifax Group prior to the Close of the Distribution Date.

Nothing contained in this Agreement shall permit, or be construed or interpreted to permit, any non-employee of Equifax or Certegy to participate, at any time, in any Plan of Equifax or Certegy.

(ww) Transition Support Agreement

"Transition Support Agreement" means the Transition Support Agreement entered into by Equifax and Certegy governing certain matters related to the relationship of the parties after the Distribution.

(xx) VEBA

"VEBA," when immediately preceded by "Equifax," means the Equifax Inc. Voluntary Employees Beneficiary Association. When immediately preceded by "Certegy," "VEBA" means the welfare benefit fund to be established by Certegy pursuant to Section 5.02 that corresponds to the Equifax VEBA.

1.02 References

Unless the context clearly indicates otherwise, reference to a particular Article, Section, subsection or paragraph means the Article, Section, subsection or paragraph so delineated in this Agreement.

2.01 Assumption of Liabilities

Except for each Liability that is expressly retained in writing by Equifax or excluded in writing by Equifax from those being assumed by Certegy and unless otherwise provided for elsewhere in this Agreement, Certegy hereby assumes and agrees to pay, perform, fulfill, and discharge, in accordance with their respective terms and conditions, all of the following (regardless of when or where such Liabilities arose or arise or were or are incurred): (i) all Liabilities to or relating to Transferred Individuals arising out of or resulting from employment by, the performance of services for, or any other type of financial relationship with, a member of the Equifax Group before becoming Transferred Individuals and thereafter (including Liabilities under Equifax Plans and Certegy Plans), (ii) all other Liabilities to or relating to Transferred Individuals and other employees or former employees of a member of the Certegy Group, and their dependents and beneficiaries, to the extent relating to, arising out of or resulting from future, present, or former employment with, or the provision of services for, a member of the Certegy Group or the Certegy Business (including Liabilities under Equifax Plans and Certegy Plans), (iii) all Liabilities relating to, arising out of, or resulting from any other actual or alleged employment, performance of services for, or any other type of financial relationship with the Certegy Group or the Certegy Business; (iv) all Liabilities under any Individual Agreements relating to Transferred Individuals, and (v) all other Liabilities relating to, arising out of, or resulting from obligations, liabilities, and responsibilities expressly assumed or retained by a member of the Certegy Group or a Certegy Plan pursuant to this Agreement; provided, however, that, consistent with Section 7.18, Certegy shall

not assume any Liability with respect to any claim which is made by any Person (other than a Person who is a member of the Certegy Group) and which is related solely to Equifax's exercise of its fiduciary responsibility for the investment of the assets of the Equifax Plans prior to the Distribution Date (or prior to the Pension Plan Spinoff Date for purposes of the Equifax Pension Plan and prior to the Savings Plan Transfer Date for purposes of the Equifax Savings Plan).

2.02 Certegy Group Participation in Equifax Plans

(a) Participation in Equifax Plans

Subject to the terms and conditions of this Agreement, each member of the Certegy Group that is, as of the date of this Agreement, a Participating Company in any of the Equifax Plans shall continue as such through the Close of the Distribution Date unless, for periods before the Distribution Date, the parties mutually agree otherwise. Effective as of any date before the Distribution Date, a member of the Certegy Group not described in the preceding sentence may, at its request and with the consent of Equifax (which consent shall not be unreasonably withheld), become a Participating Company in any or all of the Equifax Plans in which Transferred Individuals participate.

(b) Equifax's General Obligations as Plan Sponsor

Equifax shall continue through the Close of the Distribution Date to administer, or cause to be administered, in accordance with their terms and applicable law, the Equifax Plans and

Equifax (or its designee) shall have the sole discretion and authority to interpret the Equifax Plans through such date and during any subsequent period.

(c) Certegy's General Obligations as Participating Company

Certegy shall perform with respect to its participation in the Equifax Plans, and shall cause each other member of the Certegy Group that is a Participating Company in any Equifax Plan to perform the duties of a Participating Company as set forth in such Plans, and any written or oral procedures adopted pursuant thereto, including: (i) assisting in the administration of claims, to the extent requested by the claims administrator or plan administrator of the applicable Equifax Plan, (ii) cooperating fully with Equifax Plan auditors, benefit personnel and benefit vendors, (iii) preserving the confidentiality of all financial and business arrangements Equifax has or may have with any vendors, claims administrators, trustees or any other entity or individual with whom Equifax has entered into an agreement relating to the Equifax Plans, and (iv) preserving the confidentiality of participant health information (including health information in relation to leaves under the Family and Medical Leave Act of 1993, as amended).

(d) Termination of Participating Company Status

Unless otherwise provided in this Agreement, effective as of the Close of the Distribution Date, Certegy and each member of the Certegy Group shall cease to be a Participating Company in the Equifax Plans.

2.03 Establishment of The Certegy Plans

Unless otherwise provided in this Agreement, either before the Close of the Distribution Date or effective Immediately after the Distribution Date, Certegy shall adopt, or shall cause to be adopted, the Certegy Pension Plan, the Certegy Savings Plan, the Certegy Health and Welfare Plans listed in Appendix B, and the Certegy Executive Programs listed in Appendix A for the benefit of Transferred Individuals and other current, future, and former employees of the Certegy Group. The foregoing Certegy Plans as in effect Immediately after the Distribution Date shall be substantially identical in all Material Features to the corresponding Equifax Plans as in effect as of the Close of the Distribution Date. Except for the Certegy Savings Plan and VEBA, any Certegy Plans that are adopted by Certegy before the Close of the Distribution Date shall be adopted to become effective Immediately after the Distribution Date. Notwithstanding the above, Certegy shall not be required to adopt any stock benefits trust or rabbi trust related to any Equifax Executive Program or Health and Welfare Plan (including any program or plan maintained by Equifax that is not listed in Appendix A) other than the Executive Life Trust.

2.04 Terms of Participation by Transferred Individuals

The Certegy Plans shall be, with respect to Transferred Individuals, in all respects the successors in interest to, shall recognize all rights and entitlements as of the Close of the Distribution Date under, and shall not provide benefits that duplicate benefits provided by, the corresponding Equifax Plans for such Transferred Individuals. Equifax and Certegy shall agree on methods and procedures, including amending the respective Plan documents, to prevent Transferred Individuals from receiving duplicative benefits from the Equifax Plans and the Certegy Plans. Certegy shall not permit any Certegy Plan to commence benefit payments to

Transferred Individuals until it receives notice from Equifax regarding the date on which payments under the corresponding Equifax Plan shall cease. With respect to Transferred Individuals, each Certegy Plan shall provide that all service, all compensation, and all other benefit-affecting determinations that, as of the Close of the Distribution Date, were recognized under the corresponding Equifax Plan (for periods immediately before the Close of the Distribution Date) shall, as of Immediately after the Distribution Date, receive full recognition, credit, and validity and be taken into account under such Certegy Plan to the same extent as if such items originally occurred under such Certegy Plan, except to the extent that duplication of benefits would result. The provisions of this Agreement for the transfer of assets, if any, from certain trusts relating to Equifax Plans (including Foreign Plans) to the corresponding trusts relating to Certegy Plans (including Foreign Plans) are based upon the understanding and agreement of the parties that each such Certegy Plan will assume all Liabilities of the Transferred Individuals and corresponding Equifax Plan to or relating to Transferred Individuals, as provided for herein. If there are any legal or other authoritative reasons that any such Liabilities are not effectively assumed by the appropriate Certegy Plan, then the amount of assets transferred to the trust relating to such Certegy Plan from the trust relating to the corresponding Equifax Plan shall be recomputed, ab initio, as set forth in this

Agreement but taking into account the retention of such Liabilities by such Equifax Plan, and assets shall be transferred by the trust relating to such Certegy Plan to the trust relating to such Equifax Plan so as to place each such trust in the position it would have been in, had the initial asset transfer been made in accordance with such recomputed amount of assets.

3.01 Establishment of Mirror Pension Trusts

Effective Immediately after the Distribution Date, Certegy shall establish, or cause to be established, the Certegy Pension Trust which shall be qualified under Code (S) 401(a), be exempt from taxation under Code (S) 501(a)(1), and form part of the Certegy Pension Plan. Certegy shall, prior to the end of the remedial amendment period for the Certegy Pension Plan, apply for a determination letter from the Internal Revenue Service that shall provide that the Certegy Pension Plan and the Certegy Pension Trust satisfy the requirements for qualification under Code sections 401(a) and 501(a), and Certegy shall take all actions necessary or appropriate to obtain such letter.

- 3.02 Assumption of Pension Plan and Liabilities and Allocation of Interests in the Equifax Pension Trust
 - (a) Assumption of Liabilities by Certegy Pension Plan

Immediately after the Pension Plan Spinoff Date all Liabilities to or relating to Transferred Individuals under the Equifax Pension Plan shall cease to be Liabilities of the Equifax Pension Plan, and shall be assumed in full and in all respects by the Certegy Pension Plan. If the Pension Plan Spinoff Date does not occur Immediately after the Distribution Date, accruals for Transferred Individuals for the period beginning Immediately after the Distribution Date shall occur under the Certegy Pension Plan, and the benefits paid to Transferred Individuals under the Certegy Pension Plan for periods prior to the Pension Plan Spinoff Date shall be reduced by benefits paid to them under the Equifax Pension Plan. No pension benefits with respect to such Transferred Individuals shall commence from the Equifax Pension Plan while a Transferred Individual is employed by the Certegy Group.

- (b) Asset Allocations and Transfers
 - (1) Determination of Asset Transfer Date and Allocation
- (A) It is expected that assets will be transferred from the Equifax Pension Trust to the Certegy Pension Trust effective as of the date described in this subparagraph (A) (the "Pension Plan Spinoff Date"). The asset transfer is scheduled to occur Immediately after the Distribution Date (or on such rescheduled date as may be separately agreed to), provided that the assets of the Equifax Pension Plan on such date are at least 106% of its Plan Termination Liability on such date. If the assets of the Equifax Pension Plan are not at least 106% of its Plan Termination Liability on such date, either Equifax or Certegy may elect not to proceed with the transfer as of the scheduled date. If the transfer of assets does not occur on the scheduled date, Equifax and Certegy will mutually reschedule the transfer date; provided, however, such rescheduled

date must occur (i) not later than December 31, 2001, and (ii) on a date when the assets of the Equifax Pension Plan are at least 106% of its Plan Termination Liability (or another percentage of such liability that is agreed to by Equifax and Certegy). If the assets of the Equifax Pension Plan are not at least equal to the applicable minimum percentage of its Plan Termination Liability on such date, either Equifax or Certegy may elect not to proceed with the transfer as of the rescheduled date.

(B) If a Pension Plan Spinoff Date occurs in accordance with the conditions described in subparagraph (A) above, then effective as of the Pension Plan Spinoff Date and pursuant to the procedures described in this Section 3.02(b), Equifax shall transfer from the Equifax Pension Trust an amount of assets equal to the Plan Termination Liability with respect to Transferred Individuals under the Equifax Pension Plan to the Certegy Pension Trust. In addition, Equifax shall also transfer an amount of assets determined in the sole discretion of Equifax that is in addition to the amount of assets determined in the prior sentence; provided, however, Equifax shall not transfer any such

additional amount of assets to the extent, determined as of immediately after the Pension Plan Spinoff Date, such transfer would cause (i) the Equifax Pension Plan to be funded at less than 101% of its Plan Termination Liability or (ii) the Certegy Pension Plan to be funded at more than 250% of its Plan Termination Liability.

- (2) Transfer of Assets to Certegy Pension Trusts
- (A) Effective as soon as practicable after the Pension Plan Spinoff Date, Equifax shall cause to be transferred from the Equifax Pension Trust to the Certegy Pension Trust an initial amount of assets in cash (the "Initial Asset Transfer"). The amount of the Initial Asset Transfer shall not be less than Equifax's estimate, determined by Equifax in its sole discretion, of the cash required by the Certegy Pension Plan to make payment of benefits and appropriate expenses from the Certegy Pension Trust in accordance with the Plan from the time of the Initial Asset Transfer to the time of the Bulk Asset Transfer, described below. In the event Certegy notifies Equifax that the Initial Asset Transfer provides insufficient cash for this purpose, or if Equifax decides to provide additional funds, Equifax will cause to be transferred other amounts of cash or marketable assets (a "Subsequent Asset Transfer").
- (B) As soon as practicable after the calculation of each Plan's interest in the Equifax Pension Trust, but in no event before Equifax (or its authorized representative) determines that the calculation and the data on which it is based are acceptably complete, accurate, and consistent, Equifax will cause the appropriate amount of assets to be transferred from the Equifax Pension Trust to the Certegy Pension Trust (the "Bulk Asset Transfer"). The amount of assets to be transferred in the Bulk Asset Transfer shall be equal to the interests of the Certegy Pension Plan determined pursuant to Section 3.02(b)(1), adjusted by Equifax as of the date of the Bulk Asset Transfer to the extent necessary to reflect (i) any additional pension contributions that are specifically designated by Equifax for the Certegy Pension Plan, (ii) a proportionate share of actual investment gains and losses experienced in the Equifax Pension Trust, (iii) benefit payments and expenses allocable to the Certegy Plan, and (iv) the Initial Asset Transfer, Subsequent Asset Transfers, data corrections, enhancements, and computational refinements from immediately after the Pension Plan Spinoff Date through the date of the actual transfer of such assets. If circumstances permit and if agreed to by Equifax and Certegy, the parties may perform the Bulk Asset Transfer at the same time as the Initial Asset Transfer or any Subsequent Asset Transfers.
 - (3) Transfer of Specific Assets in Bulk Asset Transfer

The specific assets to be transferred from the Equifax Pension Trust to the Certegy Pension Trust in the Bulk Asset Transfer shall be made up of cash and marketable assets selected by Equifax consistent with the objective of enabling Certegy to implement prospectively an

investment program for the Certegy Pension Trust, but in no event shall Equifax or the Equifax Pension Trust be required to incur unreasonable transaction costs in the process of transferring assets and subsequently re-balancing the investment portfolio held by the Equifax Pension Trust. Furthermore, Equifax shall not be required to transfer any specific asset or any portion of any specific fund or investment manager account, and Equifax shall not transfer the Venture Capital Assets, as defined in subsection (c) below, and any amount of Certegy Common Stock that would place the Certegy Pension Plan in violation of the employer stock acquisition limitations of ERISA section 407; provided,

however, that Equifax shall transfer interests in group annuity contracts held

by the Equifax Pension Trust to the extent such group annuity contracts (in whole or in part, as the case may be) specifically cover the accrued pension benefits of Transferred Individuals. In transferring specific assets, Equifax makes no representation as to the appropriateness of the resulting asset allocation or investment performance resulting from the specific assets transferred. By accepting the assets transferred, Certegy acknowledges that it and not Equifax is serving as the fiduciary for the Certegy Pension Trust with respect to the determination and actual transfer of assets from the Equifax Pension Trust and that, acting as fiduciary for the Certegy Pension Plan, Certegy further acknowledges that it is able to change the asset allocation as it deems appropriate at any time. Once the assets have been transferred to and received by the Certegy Pension Trust, such event shall fully and finally foreclose any issue or matter of any nature whatsoever by Certegy, the Certegy Pension Trust, the Certegy Pension Plan, or any other trust(s) related to such Plan against Equifax, the Equifax Pension Trust, the Equifax Pension Plan, or any other trust(s) related to such Plan with respect to the condition, identity, or value of such assets and Certegy shall fully indemnify Equifax, its employees, officers, directors, and the Equifax Pension Plan, the Equifax Pension Trust, and any trustees or fiduciaries thereof regarding any Liability or legal or regulatory issue of any nature with respect thereto.

(c) Valuation of Venture Capital Assets

For all purposes under this Article III, the assets of the Equifax Pension Trust that are invested in a portfolio of venture capital funds (collectively, the "Venture Capital Assets") shall be valued based upon the most recent quarter-end value reported to the Equifax Pension Plan by the venture capital funds, provided that Equifax determines that such valuation remains sufficiently current to be a permissible valuation for purposes of Code section 414(1).

(d) Pension Plan Transition Period

If the Pension Plan Spinoff Date does not occur Immediately after the Distribution Date, there shall be a transition period (the "Pension Plan Transition Period") during which Transferred Individuals shall be participants in both the Equifax Pension Plan and the Certegy Pension Plan. The Pension Plan Transition Period shall begin Immediately after the Distribution Date and shall end effective as of the Pension Plan Spinoff Date. During the Pension Plan Transition Period, Equifax shall retain all assets and Liabilities with respect to Transferred Individuals under the Equifax Pension Plan and the Certegy Pension Plan shall provide equivalent pension benefits (taking into account service and compensation with Equifax) with an offset for any benefits provided under the Equifax Pension Plan.

(e) Negotiation if There Is No Pension Plan Spinoff Date

In the event that Equifax is unable to select a Pension Plan Spinoff Date under the conditions described in Section 3.02(b)(1), Equifax and Certegy shall enter negotiations as soon as practicable after January 1, 2002 on how to structure the Equifax Pension Plan and the Certegy Pension Plan with respect to Transferred Individuals. During the period that negotiations are ongoing and during any period that Equifax and Certegy are unable to reach an agreement that is mutually satisfactory to Equifax and Certegy respecting such matters, the Pension Plan Transition Period shall be continued as provided in Section 3.02(d).

(f) Allocation of Code Section 401(h) Opportunity

For purposes of allocating between the Equifax and Certegy Pension Plans the Code Section 401(h) contribution limitation that is derived from (i) contributions made to the Equifax Pension Plan for periods before the Close of the Distribution Date, and (ii) any contribution described in Section 3.02(b)(2)(B)(i), the Certegy Pension Plan shall be entitled to 8.3% of the aggregate dollar amount of the contribution limitation derived from such contributions. Such percentage reflects the expected postretirement benefit obligations that Certegy is assuming under this Agreement as compared to the aggregate expected postretirement benefit obligations of both Equifax and Certegy as of the Close of the Distribution Date.

3.03 Governmental Compliance

Notwithstanding any provision of this Agreement to the contrary, in the event that at any time any Governmental Authority challenges or seeks to prevent the transfer of assets and Liabilities provided for in Section 3.02, (i) Equifax may, in its sole discretion, retain all assets and Liabilities with respect to Transferred Individuals under the Equifax Pension Plan, and require Certegy to provide equivalent benefits under plans maintained by it with an offset for any benefits to be provided under the Equifax Pension Plan, or (ii) reach such other agreement as may be mutually satisfactory to Equifax and Certegy respecting the matters covered by this Article III. If Equifax, in its sole discretion, retains any Liability of any Transferred Individual under the Equifax Pension Plan, Certegy shall fully reimburse Equifax for the full cash costs of, including any administrative expenses relating to, any Liabilities that it has not otherwise agreed to assume that result from the intervention of the Governmental Authority.

4.01 Savings Plan

(a) Savings Plan Trust

Prior to the Distribution Date, Certegy shall establish, or cause to be established, a trust qualified under Code (S) 401(a), which shall be exempt from taxation under Code (S) 501(a)(1), and form part of the Certegy Savings Plan. Certegy shall, prior to the end of the remedial amendment period for the Certegy Savings Plan, apply for a determination letter from the Internal Revenue Service that shall provide that the Certegy Savings Plan and its related trust satisfy the requirements for qualification under Code sections 401(a) and 501(a), and Certegy shall take all actions necessary or appropriate to obtain such letter.

(b) Savings Plan Participation

Effective Immediately after the Distribution Date, the Certegy Savings Plan shall assume and be solely responsible for all ongoing rights of Transferred Individuals for future participation (including the right to make contributions through payroll deductions) in the Certegy Savings Plan. Certegy shall cause the Certegy Savings Plan to recognize and maintain all contribution and investment elections made by Transferred Individuals under the Equifax Savings Plan as such elections were last in effect during the period immediately prior to the Distribution Date and shall apply such elections under the Certegy Savings Plan for the remainder of the period or periods for which such elections are by their terms applicable (subject in all cases to applicable election change rights of the Transferred Individuals).

(c) Assumption of Liabilities and Transfer of Assets

Effective on a date selected by Equifax (the "Savings Plan Transfer Date") and except as provided in the last sentence of this subsection: (i) the Certegy Savings Plan shall assume and be solely responsible for all Liabilities to or relating to Transferred Individuals under the Equifax Savings Plan; and (ii) Equifax shall cause the accounts of the Transferred Individuals under the Equifax Savings Plan which are held by its related trust as of the Savings Plan Transfer Date to be transferred to the Certegy Savings Plan and its related trust, and Certegy shall cause such transferred accounts to be accepted by such plan and trust. Effective no later than the Savings Plan Transfer Date, Certegy shall use its reasonable best efforts to enter into such agreements to accomplish such assumptions and transfers and the maintenance of the necessary participant records. As soon as practicable after the Savings Plan Transfer Date, assets related to the accounts of all Transferred Individuals shall be transferred from the Equifax Savings Plan to the Certegy Savings Plan in cash or in kind, at Equifax's discretion, and, to the extent practicable, shall be invested in investment options in the Certegy Savings Plan which are comparable to the investment options in which such accounts were invested immediately before the Savings Plan Transfer Date. Until such transfer (but subject to reasonable blackout requirements and subject to subsection (d) below), Transferred Individuals shall be able to exercise customary investment discretion over their accounts in the Equifax Savings Plan. No benefits with respect to a Transferred Individual from the Equifax Savings Plan shall be paid while he or she is employed by the Certegy Group after the Savings Plan Transfer

(d) Specific Stock Funds in the Equifax and Certegy Savings Plan

Effective Immediately after the Distribution Date, a Certegy Common Stock fund shall be added as an investment option to the Equifax Savings Plan, and the Certegy Savings Plan shall provide for both an Equifax Common Stock fund and a Certegy Common Stock fund as investment options. The Certegy Common Stock fund in the Equifax Savings Plan and the Equifax Common Stock fund in the Certegy Savings Plan are each referred to as a "Non-Employer Stock Fund" with respect to the applicable Plan. Unless Equifax and Certegy agree otherwise, each Non-Employer Stock Fund shall be maintained under the respective Plan through December 31, 2002. After such date, Equifax and Certegy agree to cooperate with each other with respect to the disposition of the stock when either party decides to liquidate or otherwise terminate the Non-Employer Stock Fund in its Savings Plan. In the absence of any agreement regarding such liquidation, the liquidation of the Non-Employer Stock Fund shall be made in a ratable manner over a period of six months. Furthermore, at any time that Equifax or Certegy decides to sell or otherwise dispose of any amount of shares in their Non-Employer Stock Fund, the selling party shall follow the procedures of Section 4.02 (except to the extent waived in writing by an authorized representative of the other party). The Equifax Savings Plan and the Certegy Savings Plan shall each provide that, after the Distribution Date, no new contributions may be invested in, and no amounts may be transferred from other investment options to the Non-Employer Stock Fund under the respective Plan. To the extent the parties determine it is necessary to preserve the tax status of the Distribution under the Code, the Equifax Savings Plan shall provide that no earnings or dividends under its Non-Employer Stock Fund may be reinvested in Certegy Common Stock and the Certegy Savings Plan shall provide that no earnings or dividends under its Non-Employer Stock Fund may be reinvested in Equifax Common Stock; provided, however, this requirement shall not prohibit such earnings and

dividends from remaining in the applicable Non-Employer Stock Fund as cash or as an amount that is invested in any non-stock investment in such fund.

(e) Miscellaneous Funds

In the event that Equifax determines that it is not feasible or appropriate to transfer in-kind the assets of a particular investment fund from the Equifax Savings Plan to the Certegy Savings Plan, then the value of the assets in the affected investment, as of the close of business on the Savings Plan Transfer Date (plus actual earnings or losses attributable to such amount from the Savings Plan Transfer Date to the date the assets are actually transferred) shall be transferred in cash to the Certegy Savings Plan and Certegy shall invest such cash in its Savings Plan and trust in the same manner and proportion as it was invested in the Equifax Savings Plan or otherwise at the direction of the affected participant.

(f) Transfer of Assets Concerning Matching Contribution Account

The transfer of assets to the Certegy Savings Plan shall include the transfer of a portion of an account, which holds shares of Equifax common stock that were intended to be used to fund the Equifax Savings Plan's matching contributions for the 2001 plan year (the "Special Account"). The apportionment of the Special Account between the Equifax Savings Plan and the Certegy Savings Plan shall be determined as follows as of the Savings Plan Transfer Date. The portion of the Special Account transferred to the Certegy Savings Plan shall be determined by multiplying three components of the Special Account (determined immediately prior to the

Savings Plan Transfer Date) by a fraction, the numerator of which shall be the total elective deferrals made by Transferred Individuals during the portion of the 2001 calendar year ending immediately prior to the Savings Plan Transfer Date and the denominator of which shall be the total elective deferrals made by all participants in the Equifax Savings Plan during this portion of the 2001 calendar year. The three components are (1) shares of Equifax common stock in the Special Account, (2) shares of Certegy Common Stock in the Special Account (if any), and (3) cash. The product of each such component, after it has been multiplied by the above fraction, shall be the portion transferred to the Certegy Savings Plan. With respect to components (1) and (2), cash shall be transferred in lieu of a fractional share.

4.02 Non-Employer Stock Fund Procedures

(a) Application of Procedures

Whenever Equifax or Certegy (the "Selling Party") desires to sell or otherwise dispose of any or all of the shares of stock of the other party (the "Notice Party") in their respective Non-Employer Stock Funds, the Selling Party shall follow the procedures for selling or otherwise disposing of such stock, and the Notice Party shall be entitled to advance notice and a right of first refusal to purchase such stock all as provided in this section.

(b) Right of First Refusal and Notice Procedures

When the Selling Party desires to sell or otherwise dispose of any or all of the shares of stock in their Non-Employer Stock Fund, the Selling Party shall first offer such stock for sale to the Notice Party by delivering to the Notice Party's Chief Financial Officer, or his or her designee, a written offer of such shares of stock for sale to the Notice Party. The offer for sale to the Notice Party shall designate the total number of shares of stock desired to be sold (the "Offered Shares") and any special terms of sale, if the Offered Shares will not be sold on the market.

(c) Procedures After Notice is Given

The Notice Party shall within 30 days after receipt of notice advise the Selling Party in writing whether it desires to purchase the Offered Shares, and the terms upon which such purchase would be consummated. Failure to so notify the Selling Party shall be deemed a rejection of the offer, and the Selling Party after the close of the 30-day period may sell or otherwise dispose of the Offered Shares in its sole discretion. In the event that the Notice Party advises the Selling Party of its acceptance of the offer for sale, such acceptance shall specify a purchase date (the "Purchase Date") that is not less than ten nor more than 30 days after the date of such acceptance, and on which the New York Stock Exchange is scheduled to be open for trading. The purchase price paid for the Offered Shares and the terms and conditions of the payment shall be negotiated and agreed upon between the Selling Party and the Notice Party before the Purchase Date. If the Selling Party and the Notice Party have not agreed upon the price and the terms and conditions for payment by the Purchase Date, the Notice Party shall have the right to purchase all of the Offered Shares for cash immediately following the close of the New York Stock Exchange on the Purchase Date, at a price per Offered Share equal to its closing price on the New York Stock Exchange (but with any adjustment for the size of the block that is necessary, in the judgment of the valuation expert designated by Equifax and Certegy for this

purpose, to accomplish the sale at fair market value). If the Notice Party does not purchase and pay for all of the Offered Shares on the Purchase Date, it shall be deemed to have completely rejected the offer and all of the Offered Shares may be sold or otherwise disposed of by the Selling Party in its sole discretion.

5.01 Assumption of Health and Welfare Plan Liabilities

(a) Assumption by Certegy

Immediately after the Distribution Date, all Liabilities for or relating to Transferred Individuals under the Equifax Health and Welfare Plans and all Liabilities relating to the Equifax Severance Pay Plan specified in Section 5.11(b), shall cease to be Liabilities of Equifax or the Equifax Health and Welfare Plans and shall be assumed by Certegy and the corresponding Certegy Health and Welfare Plans. Thus, Certegy and the Certegy Health and Welfare Plans shall be responsible for all Liabilities that pertain to Transferred Individuals regardless of when incurred, including all reported claims that are unpaid, all incurred but not reported claims as of the Close of the Distribution Date, and all claims incurred after the Close of the Distribution Date that pertain to Transferred Individuals under the Equifax Health and Welfare Plans and the Certegy Health and Welfare Plans. Certegy shall be required to make all payments due or payable to Transferred Individuals under the appropriate Certegy Health and Welfare Plans for the period beginning Immediately after the Distribution Date, including all reported claims that are unpaid and all incurred but not reported claims as of the Close of the Distribution Date. All treatments which have been pre-certified for or are being provided on an on-going basis to a Transferred Individual under the Equifax Health and Welfare Plans as of the Close of the Distribution Date shall continue to be provided without interruption under the appropriate Certegy Health and Welfare Plan until such treatment is concluded or discontinued pursuant to applicable plan rules and limitations, and Certegy and the Certegy Health and Welfare Plans shall be responsible for all Liabilities relating to, arising out of, or resulting from such pre-certified or on-going treatments as of the Close of the Distribution Date. Notwithstanding any of the foregoing in this Section 5.01, neither Certegy nor a Certegy Health and Welfare Plan shall assume any Liability with respect to: (i) a claim incurred on or prior to the Close of the Distribution Date for which Equifax or an Equifax Health and Welfare Plan has insurance coverage, and (ii) benefit claims for which Equifax is responsible under Section 5.07. Furthermore, except to the extent related to claims referenced in the prior sentence, Equifax shall pay to Certegy any prescription drug rebates received by Equifax under the Equifax Major Medical Plan relating to claims incurred for Transferred Individuals in respect of periods beginning Immediately after the Distribution Date.

(b) Certain Audit Procedures with Respect to Health and Welfare Plans

At periodic intervals beginning Immediately after the Distribution Date, Equifax and Certegy shall examine their respective payments and receipts for health and welfare coverages to ascertain whether Equifax has mistakenly made or received payments for coverages with respect to Transferred Individuals and whether Certegy has mistakenly made or received payments for coverages with respect to participants and beneficiaries in the Equifax Health and Welfare Plans (other than Transferred Individuals). If any such mistaken payments have been made or received by Equifax or Certegy, such mistaken payments and receipts shall first be netted against each other by Equifax and Certegy and thereafter such net payments or net receipts shall be further netted against the other party's net payments or net receipts. The party with the remaining amount of mistaken payments shall transfer such amount in cash to the other party at such time

or times as agreed upon by Equifax and Certegy, but not less than semi-annually. Furthermore, at periodic intervals beginning Immediately after the Distribution Date, Equifax and Certegy shall examine the payment of benefits and claims, and reimbursements for expenses, by their respective Health and Welfare Plans to ascertain whether any Equifax Health and Welfare Plan has mistakenly paid claims or benefits, or reimbursed expenses, with respect to Transferred Individuals and whether any Certegy Health and Welfare Plan has mistakenly paid claims or benefits, or reimbursed expenses, with respect to participants and beneficiaries in the Equifax Health and Welfare Plans (other than Transferred Individuals). If any such mistaken payments or reimbursements have been made by any Equifax or Certegy Health and Welfare Plan, such mistaken payments and reimbursements shall be netted against the other corresponding Health and Welfare Plan's mistaken payments and reimbursements. The Health and Welfare Plan with the remaining amount of mistaken payments and reimbursements shall transfer such amount in cash to the other party's corresponding Health and Welfare Plan at such time or times as agreed upon by Equifax and Certegy, but not less than semi-annually.

5.02 Establishment of Certegy VEBA

On or before the Distribution Date, Certegy shall establish, or cause to be established, the Certegy VEBA, which shall constitute a voluntary employees' beneficiary association under Code (S) 501(c)(9) that is exempt from the imposition of federal income tax under Code (S) 501(a). The Certegy VEBA shall be established to provide funding for benefits that are permissible under Code (S) 501(c)(9), and in particular those benefits which Equifax funded through the VEBA as of the Distribution Date. Nothing in this Section 5.02 shall restrict the right of Certegy to discontinue providing any benefit or to discontinue funding any benefit through the VEBA after the Distribution Date.

5.03 VEBA Asset Transfers

(a) Aggregate Amount to be Transferred

Equifax shall transfer at the times enumerated in Section 5.03(b) from the Equifax VEBA to the Certegy VEBA an aggregate amount (the "VEBA Transfer Amount") equal to the sum of (i) the amount of the incurred but unpaid claims for which the Certegy VEBA is liable as of the Close of the Distribution Date, plus (ii) if the Equifax VEBA contains assets in excess of the amount necessary both to make such transfer (as described in the preceding clause (i)) and to establish a reserve for the incurred but unpaid claims for which the Equifax VEBA is liable Immediately after the Distribution Date (the "Excess Assets"), a portion of the Excess Assets, determined in the discretion of Equifax. For the above purpose, the Equifax VEBA's Liabilities will be determined by Equifax in accordance with the procedure it uses, as of the Close of Distribution Date, to predict outstanding VEBA Liabilities on an ongoing basis.

(b) Timing of Asset Transfers / Interim Period

Not later than as soon as practicable after the Distribution Date, Equifax will transfer from the Equifax VEBA to the Certegy VEBA a nominal amount of assets, provided that such transfer is necessary to initially fund the Certegy

VEBA (the "Initial VEBA Transfer"). After December 31, 2001 and on a date mutually acceptable to both Equifax and Certegy, Equifax shall transfer the VEBA Transfer Amount from the Equifax VEBA to the Certegy VEBA, (i) reduced

by the amount of the Initial VEBA Transfer and claims that were paid on behalf of Transferred Individuals by the Equifax VEBA during the period beginning Immediately after the Distribution Date and that is expected to end on December 31, 2001, and (ii) adjusted by Equifax for actual investment gains and losses experienced in the Equifax VEBA, allocable expenses, data corrections, enhancements and computational refinements from Immediately after the Distribution Date to the date of the transfer. Before performing the above transfer, Equifax and Certegy shall examine the payment of claims by the Equifax VEBA so that retrospective identification can be made of the particular claims that were paid by the Equifax VEBA on behalf of Transferred Individuals. If this identification reflects that the Equifax VEBA has already made payments on behalf of the Certegy VEBA in excess of the amount to be transferred, then as soon as practicable the Certegy VEBA shall return such excess to the Equifax VEBA, adjusted by Certegy for actual investment gains and losses experienced by the Certegy VEBA, allocable expenses, data corrections, enhancements and computational refinements.

5.04 Contributions to, Investments of, and Distributions from VEBAs

Equifax shall have sole authority to direct the trustee of the Equifax VEBA, as to the timing and manner of any contributions to the Equifax VEBA, the investment of any trust assets, and the distributions and/or transfers of trust assets, in accordance with applicable law to Equifax, Certegy, any Participating Company in the trust, any paying agent, any successor trustee, or any other Person. Certegy shall have sole authority to direct the trustee of the Certegy VEBA, as to the timing and manner of any contributions to the Certegy VEBA, the investment of any trust assets, and the distributions and/or transfers of trust assets, in accordance with applicable law to Certegy, Equifax, any Participating Company in the trust, any paying agent, any successor trustee, or any other Person.

5.05 Vendor Contracts

- (a) ASO Contracts, Group Insurance Policies, HMO Agreements, and Letters of Understanding
- (1) Before the Distribution Date, Equifax shall, in its sole discretion, take such steps as are necessary under each ASO Contract, Group Insurance Policy, HMO Agreement, letter of understanding, and arrangement in existence as of the date of this Agreement to permit Certegy to participate in the terms and conditions of such ASO Contract, Group Insurance Policy, HMO Agreement, letter of understanding, or arrangement from Immediately after the Distribution Date through December 31, 2002. Alternatively, Equifax shall take such steps as are necessary to arrange for an ASO Contract, Group Insurance Policy, HMO Agreement, letter of understanding, or arrangement covering Certegy that mirrors substantively that covering Equifax. This mirror arrangement shall apply for all or a portion of such period, as necessary under the circumstances. Equifax, in its sole discretion, may cause one or more of its ASO Contracts, Group Insurance Policies, HMO Agreements, letters of understanding, and arrangements into which Equifax enters after the date of this Agreement to allow Certegy to participate in the terms and conditions thereof. Nothing contained in this Section 5.05(a) shall preclude Equifax from choosing to enter into ASO Contracts, Group Insurance Policies, HMO Agreements, letters of understandings, or other arrangements with new or different vendors. Furthermore, nothing contained in this paragraph (1) shall require Equifax to use more than its reasonable best efforts in complying with the provisions of the first and second sentence of this paragraph (1).
- (2) Equifax shall have the right to determine, and shall promptly notify Certegy of, the manner in which Certegy's participation in the terms and conditions of ASO Contracts, Group Insurance Policies, HMO Agreements, letters of understanding and

arrangements as set forth above shall be effectuated; provided, however, Equifax

shall use its best efforts to accommodate any reasonable needs communicated to Equifax by Certegy that relate thereto. Such terms and conditions shall include the financial and termination provisions, performance standards, methodologies, auditing policies, quality measures, reporting requirements, and target claims. Certegy hereby authorizes Equifax to act on its behalf to extend to Certegy the terms and conditions of the ASO Contracts, Group Insurance Policies, HMO Agreements, and letters of understanding and arrangements. Certegy shall fully cooperate with Equifax in such efforts, and, for periods through December 31, 2002, Certegy shall not perform any act or fail to take any action that would prejudice Equifax's efforts and financial arrangements under the Health and Welfare Plans (other than taking reasonable steps to enter into replacement ASO Contracts, Group Insurance Policies, HMO Agreements, and letters of understanding and arrangements for periods after December 31, 2002, which steps shall not commence publicly prior to January 1, 2002).

(b) Payment and Effect of Change in Rates

Equifax and Certegy shall use their reasonable best efforts to cause each of the insurance companies, HMOs, paid provider organizations and third-party administrators providing services and benefits under the Equifax Health and Welfare Plans and the Certegy Health and Welfare Plans to maintain the premium and/or administrative rates based on the aggregate number of participants in the Equifax Health and Welfare Plans and the Certegy Health and Welfare Plans, from the Close of the Distribution Date through December 31, 2001, separately rated or adjusted for the demographics, experience or other relevant factors related to the covered participants of Equifax and Certegy, respectively. To the extent they are not successful in such efforts, Equifax and Certegy shall each bear the revised premium or administrative rates for health and welfare benefits attributable to the individuals covered by their respective Health and Welfare Plans. The assistance that Equifax shall provide Certegy pursuant to this subsection (b) shall be provided subject to the terms and conditions of the Transition Support Agreement.

5.06 Equifax Short-Term Disability Plan

Any final determination made or settlements entered into by Equifax with respect to claims incurred under the Equifax Short Term Disability Plan by Transferred Individuals prior to the Distribution Date shall be final and binding. Equifax shall transfer to Certegy, effective Immediately after the Distribution Date, and Certegy shall assume responsibility for (i) administering all claims incurred by Transferred Individuals before the Close of the Distribution Date that are administered by Equifax as of the Close of the Distribution Date, and (ii) all Liabilities under the Equifax Short Term Disability Plan to Transferred Individuals as of the Close of the Distribution Date, in the same manner, and using the same methods and procedures, as Equifax used in determining and paying such claims. Effective Immediately after the Distribution Date, Certegy shall have sole discretionary authority to make any necessary determinations with respect to such claims, including entering into settlements with respect to such claims, and shall be solely responsible for any costs, Liabilities or related expenses of any nature whatsoever related to such claims, payments or obligations.

5.07 Retiree Health and Life Insurance Benefits

(a) Establishment of Certegy Plan

Effective Immediately after the Distribution Date, Certegy shall adopt or cause to be adopted a Certegy Plan or Plans that provide medical and life benefits to retirees. As of Immediately after the Distribution Date, the retiree health and life benefits under such plan or plans shall be substantially identical to the retiree health and life benefits that would have been provided under the Equifax Plan or Plans to similarly situated retirees immediately prior to the Close of the Distribution Date. Thereafter, nothing in this Agreement shall restrict the right of Certegy to amend or terminate benefits for retirees. In addition, nothing in this Agreement shall restrict the right of Equifax to amend or terminate, at any time, benefits for retirees.

(b) Equifax Retiree Health and Life Benefits

In the case of any former employee of the Equifax Group described in clauses (i), (ii) or (iii) of the second sentence of Section 1.01(vv)(2), Equifax will retain the Liability related to such individual's retiree health and life benefits. In addition, any Transferred Individual who (i) is eligible for retirement under the Equifax Pension Plan as of the Close of Distribution Date or (ii) becomes eligible for retirement from Certegy by December 31, 2002 (under the terms of the Certegy Pension Plan in effect Immediately after the Distribution Date) shall be eligible for retiree health and life benefits from Equifax.

5.08 COBRA and HIPAA

Effectively Immediately after the Distribution Date, Certegy or a member of the Certegy Group shall be responsible for administering compliance and providing coverage in accordance with the health care continuation coverage requirements for "group health plans" under Title X of the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended ("COBRA"), and the portability requirements (including the requirements for issuance of certificates of creditable coverage) under the Health Insurance Portability and Accountability Act of 1996 with respect to all Transferred Individuals and other employees and former employees of Certegy or a member of the Certegy Group and any beneficiaries and dependents thereof who experience a COBRA qualifying event or loss of coverage under the Certegy Health and Welfare Plans after the Close of the Distribution Date. Effective Immediately after the Distribution Date, Certegy or a member of the Certegy Group shall be responsible for filing all necessary employee change notices with respect to these persons identified in the previous sentence in accordance with applicable law.

5.09 Leave of Absence Programs

Effective Immediately after the Distribution Date, Certegy or a member of the Certegy Group shall assume sole responsibility for the administration and compliance of all leaves of absences and related programs (including compliance with the Family and Medical Leave Act of 1993, as amended) affecting Transferred Individuals.

(a) Continuance of Elections, Co-Payments, and Maximum Benefits

- (1) Certegy shall cause the Certegy Health and Welfare Plans to recognize and maintain all coverage and contribution elections made by Transferred Individuals under the Equifax Health and Welfare Plans, as such elections were last in effect during the period immediately prior to the Distribution Date, and shall apply such elections under the Certegy Health and Welfare Plans for the remainder of the period or periods for which such elections are by their terms applicable (subject to applicable election change rights). Equifax shall cause the claims administrator for the Equifax Health and Welfare Plans to transfer to the claims administrator for the Certegy Health and Welfare Plan all data necessary to maintain such coverage and elections.
- (2) Certegy shall cause the Certegy Health and Welfare Plans to recognize and give credit for (A) all amounts applied to deductibles, out-of-pocket maximums, and other applicable benefit coverage limits with respect to such expenses which have been incurred by Transferred Individuals under the Equifax Health and Welfare Plans for the remainder of the benefit limit year in which the Distribution occurs, and (B) all benefits paid to Transferred Individuals under the Equifax Health and Welfare Plans, during and prior to the benefit limit year in which the Distribution occurs, for purposes of determining when such persons have reached their maximum benefits under the Certegy Health and Welfare Plans.
- (3) Certegy shall recognize and cover under the Certegy Health and Welfare Plans all eligible employee populations covered by the Equifax Health and Welfare Plans (pertaining to Transferred Individuals) as of the Close of the Distribution Date (determined under the applicable Plan documents), including all categories of part-time employees (which are fully or partially eligible for employer contributions).
- (4) Certegy shall (A) provide coverage to Transferred Individuals under the Certegy Health and Welfare Plans without the need to undergo a physical examination or otherwise provide evidence of insurability, and (B) recognize and maintain all irrevocable assignments and elections made by Transferred Individuals in connection with their life insurance coverage under the Equifax Health and Welfare Plans and any predecessor plans.

(b) Administration

(1) Coordination of Benefits for Spouses and Dependents

Effective as of the earlier of the first January 1 or the first applicable change in status (as defined under the Certegy Health and Welfare Plans) that occurs after the Distribution Date, Certegy shall cause the Certegy Health and Welfare Plans to permit eligible Transferred Individuals to cover their lawful spouses as dependents if such lawful spouses are active or retired Equifax employees (but were not otherwise covered as a dependent under the Equifax Health and Welfare Plans or other Equifax Plans due to their previous status as both employee and dependent of an Equifax employee). As of the earlier of the first January 1 or the first applicable status change (as defined under the Certegy Health and Welfare Plans) that occurs Immediately after the Distribution Date, Equifax shall cause the Equifax Health and Welfare

Plans to permit eligible Equifax employees to cover their lawful spouses as dependents if such lawful spouses are active or retired Certegy employees. All benefits provided under any such Health and Welfare Plans to a lawful spouse or dependent of the other company's employees shall be coordinated pursuant to the terms and conditions of the applicable Equifax and Certegy Plans.

(2) Health Care Financing Administration Data Match

Effective Immediately after the Distribution Date, Certegy shall assume all Liabilities relating to, arising out of or resulting from claims verified by Equifax or Certegy under the Health Care Financing Administration data match reports that relate to Transferred Individuals. Certegy and Equifax shall share all information necessary to verify Health Care Financing Administration data match reports regarding Transferred Individuals. Certegy shall not change any employee identification numbers assigned by Equifax without notifying Equifax of the change and the new Employee Identification Number.

(c) Equifax Reimbursement Plans

To the extent any Transferred Individual contributed to an account under the Equifax Reimbursement Plans during the calendar year that includes the Distribution Date, effective Immediately after the Distribution Date, Certegy shall recognize any such Transferred Individual's account balance, determined as of the Close of the Distribution Date, and Certegy shall thereafter be solely responsible for making any and all payments relative to such account balance of the Transferred Individual for all claims during such calendar year under the applicable Certegy Reimbursement Plan. As soon as practicable after the Distribution Date, Equifax shall calculate as of the Close of the Distribution Date the aggregate net balance in the accounts of Transferred Individuals under the Equifax Reimbursement Plans, expressed relative to the contributions received from such Transferred Individuals. If the contributions received from a Transferred Individual exceed the reimbursements made to or on behalf of such Transferred Individual, the Transferred Individual shall be deemed to have a positive account balance. In turn, if the contributions received from a Transferred Individual are less than the reimbursements made to or on behalf of such Transferred Individual, the Transferred Individual shall be deemed to have a negative account balance. If the aggregate net balance in the accounts of all such Transferred Participants is a positive number, then Equifax shall pay this amount in cash to Certegy as soon as practicable after the Distribution Date, and if the aggregate net balance in the accounts of all such Transferred Participants is a negative number, then Certegy shall pay this amount in cash to Equifax as soon as practicable after the Distribution Date.

5.11 Severance Pay Plan

(a) Transferred Individuals

Effective Immediately after the Distribution Date, Certegy shall be responsible for all Liabilities relating to the Equifax Severance Pay Plan and all other severance programs and obligations, both written and unwritten, that may pertain to Transferred Individuals. Furthermore, Certegy shall be solely responsible for all Liabilities for Transferred Individuals in connection with each Individual Agreement. As of the date of this Agreement, Equifax has, to its best knowledge, provided Certegy with a copy of each Individual Agreement.

(b) Additional Liability Assumed by Certegy

Effective Immediately after the Distribution Date, Certegy shall also be responsible for all Liabilities relating to the Equifax Severance Pay Plan and all other severance programs and obligations (including Individual Agreements), both written and unwritten, that may pertain to any individual who, as of the Close of the Distribution Date, is receiving benefits under the Equifax Severance Pay Plan or any other severance program, obligation or agreement (including Individual Agreements) with Equifax and whose most recent active employment was with a member of the Certegy Group or an entity or affiliate related to the Certegy Business.

Any final determination made or settlements entered into by Equifax with respect to the Equifax Severance Pay Plan and all other severance programs and obligations relating to individuals described in this subsection (b) prior to the Distribution Date shall be final and binding. Effective Immediately after the Distribution Date, Certegy shall have sole discretionary authority to make any necessary determinations with respect to the payment of severance benefits, including entering into settlements with respect to such benefits, and shall be solely responsible for any costs, Liabilities or related expenses of any nature whatsoever related thereto.

5.12 Application of Article V to the Certegy Group

Any reference in this Article V to "Certegy" shall include a reference to another member of the Certegy Group when and to the extent Certegy has caused the other member of the Certegy Group to (a) become a party to an ASO Contract, Group Insurance Policy, HMO Agreement, letter of understanding or arrangement associated with a Certegy Health and Welfare Plan, (b) become a self-insured entity for the purposes of one or more Certegy Health and Welfare Plans, (c) assume all or a portion of the Liabilities or the administrative responsibilities with respect to benefits which arose before the Close of the Distribution Date under a Equifax Health and Welfare Plan and which were expressly assumed by Certegy pursuant to this Agreement, or (d) take any other action, extend any coverage, assume any other Liability or fulfill any other responsibility that Certegy would otherwise be required to take under the terms of this Article V, unless it is clear from the context that the particular reference is not intended to include another member of the Certegy Group. In all such instances in which a reference in this Article V to "Certegy" includes a reference to another member of the Certegy Group, Certegy shall be responsible to Equifax for ensuring that the other member of the Certegy Group complies with the applicable terms of this Agreement and that the Transferred Individuals employed by such member of the Certegy Group shall have the same rights and entitlements to benefits under the applicable Certegy Health and Welfare Plans that the Transferred Individual would have had, if he or she had instead been employed by Certegy.

Article VI Executive Programs

6.01 Assumption of Obligations

Consistent with the principles set forth in Article II and except as otherwise provided herein, effective Immediately after the Distribution Date, the Certegy Group shall assume and be solely responsible for all Liabilities to or relating to Transferred Individuals under all Equifax Executive Programs. Certegy shall be solely responsible for all such Liabilities, notwithstanding any failure by Certegy to complete its obligations under this Article.

6.02 Short-Term Incentive Awards

With respect to short-term incentive Awards that would otherwise be payable under an Equifax Plan to Transferred Individuals and that are based in whole or in part on the performance of Equifax for the 2001 performance year ("Equifax-Based Awards"), effective immediately prior to the Distribution a determination shall be made by Equifax of the extent to which the Equifax-Based Awards have been earned for the 2001 performance year as of such date and no further amounts shall be earned by Transferred Individuals after such date. The amounts earned (if any) under the Equifax-Based Awards as of the Close of the Distribution Date shall be payable to Transferred Individuals after December 31, 2001 in accordance with the usual terms of the Equifax Plan; provided, that the Transferred Individual must be employed by Certegy on December 31, 2001. With respect to short-term incentive Awards that would otherwise be payable under an Equifax Plan to Transferred Individuals whose Award is based solely on the performance of the Certegy Business for the 2001 performance year, such Awards shall be paid by Certegy in accordance with the terms of the Equifax Plan under which they were granted. Certegy shall be solely responsible for paying all Awards pursuant to this section and shall pay such Awards in cash or options to acquire Certegy Common Stock as elected by the Transferred Individual. Effective Immediately after the Distribution Date, Certegy shall make new short-term incentive Awards under a Plan established pursuant to Section 2.03 for Transferred Individuals who have Equifax-Based Awards for the period beginning Immediately after the Distribution Date through December 31, 2001. Certegy shall establish Award levels, performance criteria and payment terms for such Awards which shall be consistent with the past practices of Equifax with respect to such Awards. Certegy shall be solely responsible for paying all such Awards.

6.03 Long-Term Incentive Plan

(a) Plan 19 (1999-2001)

Equifax will take the actions required, including any necessary plan amendment, to provide that all Transferred Individuals who are participants in Plan 19 (1999-2001) of the Equifax Inc. Performance Share Plan for Officers on the Distribution Date shall be treated as if their employment with Certegy counted as employment by Equifax pursuant to the provisions of such Plan. The effect of this treatment shall be that each such Transferred Individual shall receive the payment he or she would have been ultimately entitled to receive under "Plan 19" of said Plan at the expiration of the performance period of the outstanding Awards, provided the Transferred Individual remains employed by Certegy through December 31, 2001 and the other

terms and conditions of the Performance Share Plan are satisfied. The payments shall be made in cash, in accordance with the terms of the Performance Share Plan at the conclusion of the Plan 19 performance period. Equifax shall be solely responsible for the payments of all amounts pursuant to this subsection (a). The performance goals for Plan 19, for awards previously granted but not yet earned, will be adjusted as appropriate to reflect the Distribution.

(b) Plan 20 (2000-2002) and 2001 Awards

The outstanding Awards to Transferred Individuals under Plan 20 (2000-2002) of the Plan shall be replaced by Awards to such Transferred Individuals under the Certegy Key Management Long-Term Incentive Plan for a performance period ending on December 31, 2002. The Award levels and performance criteria for the Awards shall be designed to allow the Transferred Individuals to earn amounts similar to what they could have received under Plan 20. No Awards shall be made to Transferred Individuals under the Equifax Inc Key Management Long-Term Incentive Plan for the performance period starting in 2001.

6.04 Stock Incentive Plan

The treatment of outstanding Awards described in this Section 6.04 shall apply to Transferred Individuals, including Transferred Individuals who are compensated under a payroll which is administered outside the 50 United States, its territories and possessions, and the District of Columbia; provided,

however, if such treatment is not legally permitted, or results in adverse

consequences for Equifax, any of its affiliates or the Transferred Individual, as determined by Equifax in its sole discretion, Equifax may determine, in its sole discretion, a different treatment.

(a) Stock Options

Effective Immediately after the Distribution Date, each Award or grant consisting of an option, regardless of the date of the grant, under an Equifax Stock Incentive Plan that is outstanding as of the Close of the Distribution Date for all Transferred Individuals shall be converted to options for Certegy Common Stock with the same material terms and conditions under the Certegy Stock Incentive Plan, and shall be transferred to the recordkeeper of the Certegy Stock Incentive Plan. As soon as practicable after the Distribution Date, the number of options and the exercise price for such options converted to options for Certegy Common Stock shall be determined in accordance with the Conversion Formula. Such converted Certegy stock option grants shall continue to vest and become exercisable under the Certegy Stock Incentive Plan in accordance with the terms of the original grant under the Equifax Stock Incentive Plan. Certegy shall be the obligor with respect to such options and shall be solely responsible for all stock option grants and payments under the Certegy Stock Incentive Plan, with respect to, but not limited to, recordkeeping, administrative costs and fees, payroll taxes, plan maintenance, option exercise and related tax filings. Certegy shall, as soon as practicable after the Distribution Date provide Transferred Individuals with an agreement or notice relating to the Transferred Individual's options under the Certegy Stock Incentive Plan.

(b) Transferred Individuals Who are Not Active Employees of Certegy

Each outstanding Award or grant under the Equifax Stock Incentive Plan that is held by a Transferred Individual who, as of the Close of the Distribution Date, is not an active employee of, or on leave of absence from and expected to return to employment with, the Certegy Group shall not be converted to a Certegy option and shall remain outstanding Immediately after the Distribution Date in accordance with its terms as applicable as of the Close of the Distribution Date, subject to such adjustments as may be applicable to outstanding Awards held by individuals who remain active employees of, or on leave of absence from, the Equifax Group after the Distribution Date.

6.05 Deferred Compensation Plan

Effective Immediately after the Distribution Date, Certegy shall establish the Certegy Deferred Compensation Plan and the amounts credited to the accounts of Transferred Individuals under the Equifax Deferred Compensation Plan shall be transferred to the Certegy Plan. The Transferred Individuals' termination of employment with the Equifax Group as a result of the Distribution shall not constitute a termination of employment for purposes of the Certegy Plan. Nothing herein shall restrict Equifax's or Certegy's rights to amend or terminate their respective Plans in the future.

6.06 Executive Life and Supplemental Retirement Benefit Plan

(a) Plan Agreements

Equifax hereby assigns to Certegy, Equifax's right, title and interest in, to and from any and all policies, agreements or documents that relate to Transferred Individuals under the Equifax Executive Life Plan, including the split-dollar life insurance agreements and the collateral assignments, effective Immediately after the Distribution Date. To the extent necessary to perfect the above assignment, Equifax shall enter into such agreements and execute such documents as may be necessary (i) for Certegy to fulfill any obligations of Equifax relating to the Equifax Executive Life Plan to any insurance company, agent or broker and the Transferred Individuals, and (ii) for Certegy to succeed to Equifax's above-referenced rights. To the extent necessary to perfect the assignment of the collateral assignments and the split-dollar life insurance agreements, Equifax may execute any agreements or documents reflecting the assignment to Certegy of its right, title and interest in, to and from the policies, collateral assignments and split-dollar life insurance agreements of Transferred Individuals. Effective Immediately after the Distribution Date, Certegy shall assume and be solely responsible for all Liabilities and shall be entitled to all benefits of Equifax under the Equifax Executive Life Plan relating to Transferred Individuals.

(b) Executive Life Trust

Equifax shall amend the Equifax Executive Life Trust to require the trustee to waive its rights to the distribution of Certegy Common Stock otherwise payable with respect to the Equifax stock held in the Equifax Executive Life Trust. Effective Immediately after the Distribution Date, Certegy shall establish the Certegy Executive Life Trust which shall be substantially identical in all Material Features to the Equifax Executive Life Trust, provided that Certegy is not required to fund the Certegy Executive Life Trust.

6.07 Global Stock Purchase Plan Loan Program

Effective Immediately after the Distribution Date, Equifax shall assign and transfer to Certegy the outstanding loans, and any related security arrangements, relating to Transferred Individuals under the Equifax Global Stock Purchase Plan Loan Program. Certegy agrees to assume the Liability and be solely responsible for the ongoing administration of such loans.

6.08 Change In Control Agreements

Effective Immediately after the Distribution Date, Certegy shall enter into change in control agreements with the Transferred Individuals listed in Appendix D which are substantially identical in all Material Features to the corresponding Equifax change in control agreements. Certegy shall be solely responsibly for all Liabilities related to the Certegy change in control agreements with the Transferred Individuals.

6.09 Restricted Stock

Effective Immediately after the Distribution Date, Transferred Individuals who hold shares of Equifax restricted stock will receive the dividend of Certegy Common Stock and such shares will be subject to the same restrictions as the Equifax common stock (counting employment with Certegy as employment with Equifax). In addition, Transferred Individuals may elect to convert their Equifax restricted stock into shares of Certegy Common Stock in accordance with the Conversion Formula.

6.10 Waiver of Dividend Relating to Certain Trusts

Prior to the Close of the Distribution Date, Equifax shall cause the appropriate parties responsible for the investment of the assets of the Executive Life Trust, the Equifax Stock Benefits Trust and all other rabbi trusts established by Equifax to waive their receipt of the dividend of Certegy Common Stock to be received as a result of the Distribution.

6.11 Automobile Program

As of the Close of the Distribution Date, Certegy shall assume all of Equifax's Liabilities and obligations with respect to the motor vehicles leased by Equifax for Transferred Individuals pursuant to lease agreements under any Equifax Executive automobile program.

Article VII General

7.01 Payment of and Accounting Treatment for Expenses and Balance Sheet Amounts

(a) Expenses

All expenses (and the accounting treatment related thereto) through the Close of the Distribution Date regarding matters addressed herein shall be handled and administered by Equifax and Certegy in accordance with past Equifax accounting and financial practices and procedures pertaining to such matters. To the extent expenses that pertain to Transferred Individuals are unpaid as of the Close of the Distribution Date, Certegy or any member of the Certegy Group shall be solely responsible for such payment, without regard to any accounting treatment to be accorded such expense by Equifax or Certegy on their respective books and records. The accounting treatment to be accorded all expenses incurred prior to the Distribution Date, whether such expenses are paid by Equifax or Certegy, shall be determined by Equifax in its sole discretion.

(b) Balance Sheet Amounts

Certegy shall assume any and all balance sheet liability that relates to any Liability assumed by it under this Agreement as of the Close of the Distribution Date or thereafter. The balance sheet liabilities to be assumed pursuant to this Section shall be determined by Equifax in its sole discretion consistent with past accounting practices, consistently applied.

7.02 Sharing of Participant Information

Equifax and Certegy shall share, Equifax shall cause each applicable member of the Equifax Group to share, and Certegy shall cause each applicable member of the Certegy Group to share, with each other and their respective agents and vendors (without obtaining releases) all participant information necessary for the efficient and accurate administration of each of the Equifax Plans and the Certegy Plans during the period Equifax and Certegy are coordinating with respect to vendor contracts under Section 5.05. Equifax and Certegy and their respective authorized agents shall, subject to applicable laws on confidentiality, be given reasonable and timely access to, and may make copies of, all information relating to the subjects of this Agreement in the custody of the other party, to the extent necessary for such administration. Until the Close of the Distribution Date, all participant information shall be provided in the manner and medium applicable to Participating Companies in the Equifax Plans generally, and thereafter until December 31, 2002, all participant information shall be provided in a manner and medium that is compatible with the data processing systems of Equifax as in effect on the Close of the Distribution Date, unless otherwise agreed to by Equifax and Certegy.

7.03 Restrictions on Extension of Option Exercise Periods, Amendment or Modification of Option Terms and Conditions

Certegy agrees that, without the prior written consent of Equifax, neither Certegy nor any of its affiliates or Subsidiaries shall take any action to extend the exercise period of or to provide for additional vesting with respect to any Certegy options, which were received pursuant to

Section 6.04 in replacement of Equifax options, for Transferred Individuals, including providing such Transferred Individuals with leaves of absences or special termination or severance arrangements.

7.04 Non-Solicitation of Employees/Others

For the period through December 31, 2002, Certegy and its affiliates and Subsidiaries will not, without the prior written consent of Equifax, and Equifax and its affiliates and Subsidiaries will not, without the prior written consent of Certegy, whether directly or indirectly, solicit (in writing or orally) for employment or other services, whether as an employee, officer, director, agent, consultant, or independent contractor, any person who is or was at the time of such solicitation an employee, agent, consultant, independent contractor, representative, officer, or director of the other party; provided, however, that

this covenant shall not apply in cases where such solicitation occurs more than 30 days after the individual to be solicited has had his employment or other service relationship with the other party terminated by the employer.

7.05 Reporting and Disclosure and Communications to Participants

While Certegy is a Participating Company in the Equifax Plans, Certegy shall take, and shall cause each other applicable member of the Certegy Group to take, all actions necessary or appropriate to facilitate the distribution of all Equifax Plan-related communications and materials to employees, participants and beneficiaries, including summary plan descriptions and related summaries of material modification, summary annual reports, investment information, prospectuses, notices and enrollment materials for the Certegy Plans. Certegy shall assist, and Certegy shall cause each other applicable member of the Certegy Group to assist, Equifax in complying with all reporting and disclosure requirements of ERISA for plan years ending on or before December 31, 2002, including the preparation of Form 5500 annual reports for the Equifax Plans, where applicable.

7.06 Plan Audits

(a) Audit Rights with Respect to the Allocation or Transfer of Plan Assets $\,$

The allocation of Pension Plan assets and liabilities pursuant to Section 3.02 and the transfer of assets from Equifax's VEBA pursuant to Section 5.03, shall, at the election of Certegy, be audited on behalf of both Equifax and Certegy by an actuarial and benefit consulting firm mutually selected by the parties; provided, however, that no audit shall be permitted after the date of

the Bulk Asset Transfer, in the case of the Pension Plans, or the actual transfer of assets, in the case of the Equifax VEBA. The scope of such audit shall be limited to the accuracy of the data and the accuracy of the computation and adherence to the methodology specified in this Agreement and, except as set forth in the penultimate sentence of this Section 7.06(a), such audit shall not be binding on the parties. The actuarial and benefit consulting firm shall provide its report to both Equifax and Certegy. No other audit shall be conducted with respect to the allocation of Plan assets and no issue of any nature whatsoever may be raised by Certegy once the transfer of assets has been completed. Subject to the following two sentences, no transfer of assets shall occur unless and until Certegy agrees to the allocation of assets. To the extent such audit recommends a change to the value of assets allocated to a Certegy Plan of less than 5%, the original determination shall be binding on the parties and shall not be subject to the dispute

resolution process provided in Section 7.17. To the extent such audit recommends such a change of 5% or more (a "Significant Allocation Change"), any unresolved dispute between the parties as to whether and how to make any change in response to such recommendation shall be subject to the dispute resolution process provided in Section 7.17. Certegy shall pay or shall be responsible for the payment of the full costs of such audit; provided, however, that in the event

such audit recommends a Significant Allocation Change and such recommendation is attributable to variances in actuarial assumptions or simplification or modification of the allocation calculated by Equifax, Equifax shall be responsible for the full costs of such audit.

(b) Audit Rights With Respect to Information Provided

- (1) Subject to Section 7.06(b)(2), each of Equifax and Certegy, and their duly authorized representatives, shall have the right to conduct audits at any time upon reasonable prior notice, at their own expense, with respect to all information provided to it or to any Plan recordkeeper or third-party administrator by the other party. Subject to Sections 7.06(a) and 7.06(b)(2), the party conducting the audit shall have the sole discretion to determine the procedures and guidelines for conducting audits and the selection of audit representatives. The auditing party shall have the right to make copies of any records at its expense, subject to the confidentiality provisions set forth in the Distribution Agreement, which are incorporated by reference herein. The party being audited shall provide the auditing party's representatives with reasonable access during normal business hours to its operations, computer systems and paper and electronic files, and provide workspace to its representatives. After any audit is completed, the party being audited shall have the right to review a draft of the audit findings and to comment on those findings in writing within five business days after receiving such draft.
- (2) The auditing party's audit rights under this Section 7.06(b) shall include the right to audit, or participate in an audit facilitated by the party being audited, of any Subsidiaries and affiliates of the party being audited and of any benefit providers and third parties with whom the party being audited has a relationship, or agents of such party, to the extent any such persons are affected by or addressed in this Agreement (collectively, the "Non-parties"). The party being audited shall, upon written request from the auditing party, provide an individual (at the auditing party's expense) to supervise any audit of any Non-party. The auditing party shall be responsible for supplying, at its expense, additional personnel sufficient to complete the audit in a reasonably timely manner.

(c) Audits Regarding Vendor Contracts

From Immediately after the Distribution Date through December 31, 2002, Equifax and Certegy and their duly authorized representatives shall have the right to conduct joint audits with respect to any vendor contracts that relate to both the Equifax Health and Welfare Plans and the Certegy Health and Welfare Plans. The scope of such audits shall encompass the review of all correspondence, account records, claim forms, canceled drafts (unless retained by the bank), provider bills, medical records submitted with claims, billing corrections, vendor's internal corrections of previous errors and any other documents or instruments relating to the services performed by the vendor under the applicable vendor contracts. Equifax and Certegy shall agree on the performance standards, audit methodology, auditing policy and quality measures and

reporting requirements relating to the audits described in this Section 7.06(c) and the manner in which costs incurred in connection with such audits will be shared.

(d) Audit Assistance

To the extent that either Equifax or Certegy is required to respond to any Governmental Authority, vendor or recordkeeper audit, or otherwise conducts an audit with respect to any provision or obligation of the other party under this Agreement, Equifax or Certegy, whichever is applicable, shall be required to fully cooperate with the audit, including providing such records and data as may be necessary to respond to any document or data request that may arise by reason of such audit. The party being audited shall provide the auditing party's representatives with reasonable access during normal business hours to its operations, computer systems and paper and electronic files, and provide workspace to its representatives. To the extent the results of an audit result in any correction to the Liabilities involving any Transferred Individuals, Certegy shall be solely responsible for all such costs and expenses associated with such Liabilities and any related corrections.

7.07 Beneficiary Designations/Release of Information/Right to Reimbursement

All beneficiary designations, authorizations for the release of information and rights to reimbursement made by or relating to Transferred Individuals under Equifax Plans shall be transferred to and be in full force and effect under the corresponding Certegy Plans until such beneficiary designations, authorizations or rights are replaced or revoked by, or no longer apply, to the relevant Transferred Individual.

7.08 Requests for Internal Revenue Service Rulings and United States Department of Labor Opinions and Satisfaction of Obligations Arising From Voluntary Compliance Programs

(a) Cooperation

Certegy shall cooperate fully with Equifax on any issue relating to the transactions contemplated by this Agreement for which Equifax elects to seek a determination letter or private letter ruling from the Internal Revenue Service or an advisory opinion from the United States Department of Labor. Equifax shall cooperate fully with Certegy with respect to any request for a determination letter or private letter ruling from the Internal Revenue Service or advisory opinion from the United States Department of Labor with respect to any of the Certegy Plans relating to the transactions contemplated by this Agreement.

(b) Applications

Equifax and Certegy shall make such applications to regulatory agencies, including the Internal Revenue Service and the United States Department of Labor, as may be necessary to ensure that any transfers of assets from the Equifax VEBA to the Certegy VEBA will neither (i) result in any adverse tax, legal or fiduciary consequences to Equifax and Certegy, the Equifax VEBA, the Certegy VEBA, any participant therein or beneficiaries thereof, any successor welfare benefit funds established by or on behalf of Certegy, or the trustees of such trusts, nor (ii) contravene any statute, regulation or technical pronouncement issued by any regulatory agency. Before the Close of the Distribution Date, Certegy shall prepare all forms required to obtain

favorable determination letters from the Internal Revenue Service with respect to the tax-exempt status of the Certegy VEBA. Certegy and Equifax agree to cooperate with each other to fulfill any filing and/or regulatory reporting obligations with respect to such transfers.

7.09 Fiduciary and Related Matters

The parties acknowledge that Equifax will not be a fiduciary with respect to the Certegy Plans and that Certegy will not be a fiduciary with respect to the Equifax Plans. The parties also acknowledge that neither party shall be deemed to be in violation of this Agreement if it fails to comply with any provisions hereof based upon its good faith determination that to do so would violate any applicable fiduciary duties or standards of conduct under ERISA or other applicable law. Notwithstanding any other provision in this Agreement, the parties may take such actions as necessary or appropriate to effectuate the terms and provisions of this Agreement.

7.10 No Third-Party Beneficiaries; Non-Termination of Employment

This Agreement is not intended and shall not be construed as to confer upon any Person other than the parties hereto any rights or remedies hereunder. No provision of this Agreement or the Distribution Agreement shall be construed to create any right, or accelerate entitlement, to any compensation or benefit whatsoever on the part of any Transferred Individual or other future, present, or former employee of the Equifax Group or the Certegy Group under any Equifax Plan or Certegy Plan or otherwise. Without limiting the generality of the foregoing, except as expressly provided in this Agreement: (i) neither the Distribution nor the termination of the Participating Company status of a member of the Certegy Group shall cause any employee to be deemed to have incurred a termination of employment which entitles such individual to the commencement of benefits under any of the Equifax Plans, any of the Certegy Plans, or any of the Individual Agreements; and (ii) nothing in this Agreement other than those provisions specifically set forth herein to the contrary shall preclude Certegy, at any time after the Close of the Distribution Date, from amending, merging, modifying, terminating, eliminating, reducing, or otherwise altering in any respect any Certegy Plan, any benefit under any Plan or any trust, insurance policy or funding vehicle related to any Certegy Plan.

7.11 Collective Bargaining

To the extent any provision of this Agreement is contrary to the provisions of any applicable collective bargaining agreement to which Equifax or any affiliate of Equifax is a party, the terms of such collective bargaining agreement shall prevail. Should any provision of this Agreement be deemed to relate to a topic determined by an appropriate authority to be a mandatory subject of collective bargaining, Equifax or Certegy may be obligated to bargain with the union representing affected employees concerning those subjects. In the event a force surplus affecting members of a bargaining unit in both the Equifax Group (on the one hand) and the Certegy Group (on the other hand) directly results, due to the provisions of such a collective bargaining agreement, in an employee involuntarily leaving the payroll of the party not declaring the surplus, then the party declaring the surplus shall bear the cost of any severance payable to such employee.

7.12 Consent of Third Parties

If any provision of this Agreement is dependent on the consent of any third party (such as a vendor) and such consent is withheld, Equifax and Certegy shall use their reasonable best efforts to implement the applicable provisions of this Agreement to the full extent practicable. If any provision of this Agreement cannot be implemented due to the failure of such third party to consent, Equifax and Certegy shall negotiate in good faith to implement the provision in a mutually satisfactory manner. The phrase "reasonable best efforts" as used in this Agreement shall not be construed to require the incurrence of any non-routine or unreasonable expense or liability or the waiver of any right.

7.13 Foreign Plans

The treatment of Foreign Plans by Equifax and Certegy shall be set forth in Appendix C.

7.14 Effect If Distribution Does Not Occur

If the Distribution does not occur, then all actions and events that are, under this Agreement, to be taken or occur before or effective as of the Close of the Distribution Date, Immediately after the Distribution Date, or otherwise in connection with the Distribution, shall not be taken or occur except to the extent specifically agreed by Certegy and Equifax.

7.15 Relationship of Parties

Nothing in this Agreement shall be deemed or construed by the parties or any third party as creating the relationship of principal and agent, partnership or joint venture between the parties, it being understood and agreed that no provision contained herein, and no act of the parties, shall be deemed to create any relationship between the parties other than the relationship set forth herein.

7.16 Affiliates

Each of Equifax and Certegy shall cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth in this Agreement to be performed by members of the Equifax Group or members of the Certegy Group, respectively, where relevant.

7.17 Dispute Resolution

Any controversy or claim arising out of or relating to this Agreement, or the breach hereof, shall be settled pursuant to the dispute resolution provisions described in the Distribution Agreement.

7.18 Indemnification

Effective on the Distribution Date, Certegy and each member of the Certegy Group agree to jointly and severally indemnify and hold harmless Equifax and each member of the Equifax Group and each of their respective officers, directors, employees and agents and the Equifax Plans and any related trusts, including the Equifax Pension Trust and the trustees thereof, from and against any and all losses, Liabilities, claims, suits, damages, costs and expenses (including

without limitation, reasonable attorneys' fees and any and all expenses reasonably incurred in investigating, preparing or defending against any pending or seriously threatened litigation or claim) (collectively, "Indemnifiable Expenses") arising out of or related in any manner to Transferred Individuals, except for Indemnifiable Expenses arising solely out of a claim which is made by any Person (other than a Person who is a member of the Certegy or Equifax Group) and which is related solely to Equifax's exercise of its fiduciary responsibility for the investment of the assets of the Equifax Plans prior to the Distribution Date (or prior to the Pension Plan Spinoff Date for purposes of the Equifax Pension Plan and prior to the Savings Plan Transfer Date for purposes of the Equifax Savings Plan). In addition, effective on the Distribution Date, Equifax agrees to indemnify and hold harmless each member of the Certegy Group and each of their respective officers, directors, employees and agents and the Certegy Plans and any related trusts, including the Certegy Pension Trust and the trustees thereof, from and against any and all Indemnifiable Expenses arising solely out of a claim which is made by any Person (other than a Person who is a member of the Certegy or Equifax Groups) and which is related solely to Equifax's exercise of its fiduciary responsibility for the investment of the assets of the Equifax Plans prior to the Distribution Date (or prior to the Pension Plan Spinoff Date for purposes of the Equifax Pension Plan and prior to the Savings Plan Transfer Date for purposes of the Equifax Savings Plan).

If any action is brought or any claim is made against a party or person in respect of which indemnity may be sought pursuant to this Section 7.18 (the "Indemnitee"), the Indemnitee shall, within twenty (20) days after the receipt of information indicating that an action or claim is likely, notify in writing the party from whom indemnification is sought (the "Indemnitor") of the institution of the action or the making of the claim, and the Indemnitor shall have the right, and at the request of the Indemnitee, shall have the obligation, to assume the defense of the action or claim, including the employment of counsel. If the Indemnitor assumes the defense of the action or claim, the Indemnitor shall be entitled to settle the action or claim on behalf of the Indemnitee without the prior written consent of the Indemnitee, unless such settlement would, in addition to the payment of money, materially affect the ongoing business or employment of the Indemnitee.

The Indemnitee shall have the right to interpret the provisions of its own Plans and to employ its own counsel, but the fees and expenses of that counsel shall be the responsibility of the Indemnitee unless: (i) the employment of that counsel shall have been authorized in writing by the Indemnitor in connection with the defense of the action or claim; (ii) the Indemnitor shall not have employed counsel to have charge of the defense of such action or claim; or (iii) such Indemnitee shall have reasonably concluded that there may be defenses available to it which are different from or additional to those available to the Indemnitor (in which case the Indemnitor shall not have the right to direct any different defense of the action or claim on behalf of the Indemnitee). The Indemnitee shall, in any event, be kept fully informed of the defense of any such action or claim. Except as expressly provided above, in the event that the Indemnitor shall not previously have assumed the defense of an action or claim, at such time as the Indemnitor does assume the defense of the action or claim, the Indemnitor shall not thereafter be liable to any Indemnitee for legal or other expenses subsequently incurred by the Indemnitee in investigating, preparing or defending against such action or claim.

Anything in this Section 7.18 to the contrary notwithstanding, the Indemnitor shall not be liable for any settlement of any claim or action effected without its written consent; provided, however, that if after due

notice the Indemnitor refuses to defend a claim or action, the Indemnitee shall have the right to defend and/or settle such action, and the Indemnitee shall not be precluded from making a claim against the Indemnitor for reasonable expenses and liabilities resulting from such defense and/or settlement in accordance with this Section 7.18.

Notwithstanding the foregoing provisions of this Section 7.18, there may be particular actions or claims which reasonably could result in both parties being liable to the other under the indemnification provisions of this Agreement. In such events, the Parties shall endeavor, acting reasonably and in good faith, to agree upon a manner of conducting the defense and settlement of the action or claim with a view to minimizing the legal expenses and associated costs that might otherwise be incurred by the parties, such as, by way of illustration only, agreeing to use the same legal counsel.

The indemnification provisions of this Section 7.18 shall not inure to the benefit of any third party. By way of illustration only, an insurer who would otherwise be obligated to pay any claim shall not be relieved of the responsibility with respect thereto, or, solely by virtue of the indemnification provisions, hereof, have any subrogation rights with respect thereto, it being expressly understood and agreed that no insurer or any other third party shall be entitled to a "windfall" (i.e., a benefit they would not be entitled to receive in the absence of the indemnification provisions) by virtue of these indemnification provisions.

7.19 W-2 Matters

Pursuant to the alternative procedure set forth in Internal Revenue Service Rev. Proc. 96-60, Certegy will assume Equifax's obligations to furnish Forms W-2 to all Transferred Individuals for the year in which the Distribution Date occurs. Equifax will provide to Certegy the information not available to Certegy relating to periods ending on the Distribution Date necessary for Certegy to prepare and distribute Forms W-2 to Transferred Individuals for the year in which the Distribution Date occurs, which will include all remuneration earned by Transferred Individuals before the Distribution Date and Forms W-4 provided to Equifax by Transferred Individuals to the extent that Certegy is not already in possession of such information. Certegy shall prepare and distribute such forms. To the extent permitted by applicable law, in particular Code (S)(S) 3121(a)(1) and 3306(b)(1), Certegy shall be deemed a successor employer to Equifax with respect to Transferred Individuals for purposes of calculating the annual wage limitation to which state and federal payroll taxes apply.

7.20 Confidentiality

Except as required by applicable law, for the purpose of satisfying any obligation under this Agreement or with the consent of the other party, neither Equifax nor Certegy shall disclose to any Person (other than members of the Equifax Group or the Certegy Group) any information (including, but not limited to, information regarding fees, expenses, assets, Liabilities and Plan terms) relating to the Equifax Plans, Certegy Plans or Transferred Individuals. Each of Equifax and Certegy shall be permitted to disclose such information within the Equifax Group and Certegy Group only to the extent reasonably necessary in the ordinary course of business.

7.21 Notices

Any notice, demand, claim, or other communication under this Agreement shall be in writing and shall be given in accordance with the provisions for giving notice under the Distribution Agreement.

7.22 Interpretation

Words in the singular shall be held to include the plural and vice versa and words of one gender shall be held to include the other genders as the context requires. The terms "hereof," "herein," and "herewith" and words of similar import shall, unless otherwise stated, be construed to refer to this Agreement as a whole (including all Appendices hereto) and not to any particular provision of this Agreement. The word "including" and words of similar import when used in this Agreement shall mean "including, without limitation," unless the context otherwise requires or unless otherwise specified. The word "or" shall not be exclusive.

7.23 Severability

The provisions of this Agreement are severable and should any provision hereof be void, voidable or unenforceable under any applicable law, such provision shall not affect or invalidate any other provision of this Agreement, which shall continue to govern the relative rights and duties of the parties as though such void, voidable or unenforceable provision were not a part hereof.

7.24 Governing Law/Execution

This Agreement shall be construed in accordance with, and governed by, the laws of the State of Georgia without regard to the conflicts of law rules of such state, may not be assigned by either party without the prior written consent of the other, and shall bind and inure to the benefit of the parties hereto and their respective successors and permitted assignees. This Agreement may not be amended or supplemented except by an agreement in writing signed by Equifax and Certegy. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original instrument, but all of which together shall constitute one and the same Agreement.

[signature page follows]

EQUIFAX INC.	
By: Name: Title:	
CERTEGY INC.	
By: Name: Title:	

IN WITNESS WHEREOF, the parties have caused this Employee Benefits Agreement to be duly executed as of the day and year first above written.

Appendix A Certegy Executive Programs

Certegy Inc. 2001 Stock Incentive Plan
Certegy Inc. Key Management Long-Term Incentive Plan
Certegy Inc. Deferred Compensation Plan
Certegy Inc. Executive Life and Supplemental Retirement Benefit Plan
Global Stock Purchase Plan Loan Program
Individual Agreements

Including employment, separation, change in control and consulting agreements.

Health Plans (ERISA):

Certegy Major Medical Plan (which includes medical, pre-65 post-retirement medical, dental, prescription drug, various HMOs and OSCs, vision, and employee assistance benefits).

Certegy Retiree Medical Plan (post-65 medical)

Group Insurance Plans (ERISA):

Certegy Group Basic Lifé and Supplemental Life Insurance Plan Certegy Accidental Death and Dismemberment Insurance Plan Certegy Group 24-Hour Business Travel Accident Plan

Disability Plans (ERISA):

Certegy Long Term Disability Plan Certegy Short Term Disability Plan

Severance Plans (ERISA):

Certegy Severance Pay Plan

Cafeteria Plan (ERISA):

Certegy Before Tax Premium Plan Certegy Health Care Account Plan Certegy Dependent Care Account Plan (non-ERISA)

Miscellaneous Plans (non-ERISA):

Certegy Educational Assistance Program Certegy Vacation Program

* Certegy shall have the right to combine or disaggregate any of the above Plans for any purpose, including the satisfaction of any disclosure or reporting requirements under ERISA.

Appendix C Foreign Plans

This Appendix C describes the principles under which Foreign Plans shall be treated. For purposes of this Appendix, outside the U.S. means outside the 50 United States, its territories and possessions, and the District of Columbia, and employed outside the U.S. means compensated under a payroll which is administered outside the United States.

C.1 Plans Covering only Employees of Equifax or Certegy

Effective as of the Close of the Distribution Date or such later date as may be required by applicable law, union, or works council agreement, any Foreign Plan that covers only individuals employed outside the U.S. by the Equifax Group shall be the sole responsibility of the Equifax Group and no member of the Certegy Group shall have any Liability with respect to such a Plan; and any Foreign Plan that covers only individuals employed outside the U.S. by the Certegy Group shall be the sole responsibility of the Certegy Group and no member of the Equifax Group shall have any Liability with respect to such a Plan.

C.2 Plans Covering Employees of Both Equifax and Certegy

(a) Termination of Participation

To the extent legally permitted and except as otherwise provided herein, effective as of the Close of the Distribution Date, or as soon as practicable thereafter, Certegy and each other applicable member of the Certegy Group shall cease to be a Participating Company in each Foreign Plan maintained by Equifax or the Equifax Group and each other applicable member of the Equifax Group shall cease to be a Participating Company in each Foreign Plan maintained by the Certegy Group. Equifax and Certegy recognize that because of the requirements of local law or administrative considerations, a transition period extending beyond the Close of the Distribution Date may be required during which Equifax or Certegy may continue to participate in each other's Foreign Plans. During such transition period, Equifax and Certegy agree to cooperate in handling any and all matters with respect to the Foreign Plans.

(b) Mirror Plans

(1) Effective Immediately after the Distribution Date, Certegy shall adopt, or cause to be adopted, Foreign Plans for the benefit of employees of the Certegy Group employed outside the United States who are eligible to participate in Equifax Foreign Plans and shall cause such Certegy Foreign Plans to be substantially identical in all Material Features to the corresponding Equifax Foreign Plans as in effect on the Distribution Date. Notwithstanding the preceding sentence - (i) Certegy may satisfy this requirement by extending coverage to such individuals under a Foreign Plan of the Certegy Group which was in effect before the Distribution Date and which is, with respect to all Material Features, at least equal to the corresponding Equifax Foreign Plan, and (ii) Certegy is not required to adopt a defined benefit pension plan for the benefit of its Canadian employees (but instead shall make an equitable adjustment to the defined contribution plan covering these employees).

(2) Effective Immediately after the Distribution Date, Equifax shall adopt, or cause to be adopted, Plans for the benefit of employees of the Equifax Group employed outside the United States who are eligible to participate in Certegy Plans and shall cause such Plans to be substantially identical in all Material Features to the corresponding Certegy Foreign Plans as in effect on the Distribution Date; provided, however, that Equifax may satisfy this requirement

by extending or continuing coverage to such individuals under an Equifax Foreign Plan of the Equifax Group Which was in effect before the Distribution Date.

(3) The continuation by Equifax or Certegy of separate employment terms and conditions for employees previously covered by the other entity's Plans shall not continue beyond the time legally required.

(c) Transfer of Assets

As of the Close of the Distribution Date, Equifax and Certegy will use their reasonable best efforts to ensure that, to the extent legally permitted: (i) Liabilities of the Foreign Plans of Equifax relating to Transferred Individuals shall be assumed by the appropriate Foreign Plans of Certegy; and (ii) an appropriate portion of any assets of the Foreign Plans of Equifax shall be transferred to the appropriate Foreign Plans of Certegy, and vice versa.

C.3 Severance Issues

If under applicable law, any Transferred Individual employed outside the U.S. is deemed to have incurred a termination of employment as a result of the Distribution or any other transaction contemplated by the Distribution Agreement or this Agreement, which entitles such individual to receive any payment or benefit under any Foreign Plan, governmental plan or arrangement or pursuant to any law or regulation, including severance benefits, notwithstanding such individual's continued employment by the Certegy Group, then Certegy shall be liable for any such payment or benefit and, notwithstanding any other provision hereof, to the extent legally permitted, appropriate adjustments shall be made to the treatment of such individual during such continued employment, including not giving such individual credit for prior service and/or treating such individual as having been newly hired immediately after such deemed termination, for purposes of all applicable Foreign Plans. Liability with respect to such payments shall be the responsibility of Certegy.

C.4 Legally Permitted

For purposes of this Appendix C, "legally permitted" means permitted under the laws of the country, the labor union, works council, or collective agreement without adverse consequences to Equifax, Certegy or Transferred Individuals, as determined by Equifax, in its sole discretion, including mandated waiting periods before which working conditions (including benefits) cannot be changed, and upon receiving required agreement from individual employees and/or Plan trustees, foundation boards and members, and any other organizations having a recognized right to determine or affect benefits and/or funding of the Plan.

C.5 Multinational Pooling

Equifax and Certegy shall keep their existing multinational pooling arrangements intact so long as the parties mutually agree. If there is any dividend payable from the consolidated pooling arrangements with respect to the 2001 pool accounting year, that dividend will be allocated between Equifax and Certegy proportionately, based on the contribution to the overall surplus of the pooling arrangements by the Equifax Group and the Certegy Group, respectively. Alternatively, any net deficits incurred under any one (or all) consolidated pooling arrangement(s) will be apportioned back to the entity which incurred the deficit proportionately based on each entities' contribution to the net deficit.

Appendix D Change in Control Agreements

D.1 Tier I Agreements

Lee Kennedy Bruce Richards Larry Towe Mike Vollkommer

D.2 Tier II Agreements

Joseph G. Ballard Jeff Carbiener Dick Gapen Gerald A. Hines Mary Waggoner INTERCOMPANY DATA PURCHASE AGREEMENT
BETWEEN EQUIFAX INC.
AND
CERTEGY INC.
DATED ______, 2001

TABLE OF CONTENTS

	Page
ARTICLE I DEFINITIONS	1 1
ARTICLE II DATA PURCHASE Section 2.1. General Section 2.2. Supplemental Terms Section 2.3. FCRA Certification Section 2.4. Effect of Data Purchase Section 2.5. Effect of Change in Law Section 2.6. Obligations of Providers and Recipients	5 5 6 7 7 7 8
ARTICLE III FEES AND PAYMENT. Section 3.1. Fees and Charges. Section 3.2. Payment Terms. Section 3.3. Disputed Amounts. Section 3.4. Interest. Section 3.5. Taxes.	8 8 8 8 9
ARTICLE IV REPRESENTATIONS AND WARRANTIES	9
Section 5.1. Covenant Not to Sue	10 10 10 10
Section 6.1. Certegy Indemnification of the Equifax Group Section 6.2. Equifax Indemnification of the Certegy Group	11 11 12 12
Section 7.1. Notice and Payment of Claims	12 12 13
Section 8.1. Exclusions Section 8.2. Confidentiality Section 8.3. Employee Confidentiality Agreements Section 8.4. Rights and Remedies Section 8.5. Competitive Activities	14 14 15 16 16 16
ARTICLE IX CONTINUED ASSISTANCE	17 17

TABLE OF CONTENTS (continued)

Section 9.2. I	Litigation Cooperation	17
ARTICLE X TERM Section 10.1. Section 10.2.	AND TERMINATION Term; Termination Effect of Termination and Expiration	17
ARTICLE XI MISO	CELLANEOUS	19
Section 11.1.	Expenses	19
Section 11.2.	Notices	
Section 11.3.	Amendment and Waiver	20
Section 11.4.	Entire Agreement	20
Section 11.5.	Parties in Interest	20
Section 11.6.	Further Assurances and Consents	
Section 11.7.	Severability	21
Section 11.8.	Governing Law	21
Section 11.9.	Counterparts	21
Section 11.10.	Disputes	21
Section 11.11.	Force Majeure	21
Section 11.12.	Headings	22

TABLE OF CONTENTS (continued)

EXHIBITS

EXHIBIT A	Α	Certegy	Group	Entities			
EXHIBIT	В	Certegy	Group	Products	&	Pricing	
EXHIBIT	С	Equifax	Group	Products	&	Pricing	
EXHIBIT	D	Equifax	Group	Products	-	Supplemental	Terms
EXHIBIT	E	Certegy	Group	Products	-	Supplemental	Terms

THIS INTERCOMPANY DATA PURCHASE AGREEMENT dated as of ______, 2001, is entered into by EQUIFAX INC., a Georgia corporation ("Equifax"), and CERTEGY INC., a Georgia corporation ("Certegy").

BACKGROUND

- A. Certegy is a wholly-owned subsidiary of Equifax formed among other reasons for the purposes of receiving and conducting the business of the Certegy Group (as defined below) and taking title to the intellectual property assets and assuming the associated liabilities related to the business operations of the Certegy Group.
- B. The transfer of such business, assets and liabilities, as well as other assets, will be effected pursuant to the Distribution Agreement (as defined below).
- C. The parties intend that the Distribution (as defined in the Distribution Agreement) not be taxable to Equifax or its shareholders pursuant to Section 355 of the Code (as defined below).
- D. Equifax and its Affiliates (as defined below) provide various Data (as defined below) to each other and to their respective customers. Prior to the Distribution, Equifax and its Affiliates utilized certain common resources and discrete, commercial products purchased from one another in the conduct of their respective businesses. At and after the Effective Time, Equifax and Certegy desire to purchase from each other and their Group members certain Data for their own use and for resale to their Group customers.
- E. The parties have determined that it is necessary and desirable to set forth the arrangements required to effect the purchase and sale of their respective Data to the members of each Group and to set forth their other agreements that will govern certain other matters regarding the parties' respective purchase and sale of Data following the Distribution.
- NOW, THEREFORE, in consideration of the foregoing premises and the mutual agreements and covenants contained in this Agreement and other valuable consideration the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

ARTICLE I

DEFINITIONS

Section 1.1. Definitions.

As used herein, the following terms have the following meanings:

(a) "Action" means any claim, suit, arbitration, inquiry, proceeding or investigation by or before any court, governmental or other regulatory or administrative agency or commission or any other tribunal.

- (b) "Affiliate" means, with respect to a party, any corporation, partnership (general or limited), limited liability company, joint venture or other form of business entity that controls, is controlled by or is under common control with that party; and "control" means (i) either directly or indirectly owning or having the right to vote ownership interests possessing a majority of the aggregate voting power of all ownership interests of that entity, or (ii) having the power to control and direct, either directly or indirectly, the business and affairs of that entity or to elect or appoint the person (or if more than one, a majority of the persons) who is responsible for the management and control of the business and affairs of that entity.
- (c) "Agreement" means the terms and conditions of this Intercompany Data Purchase Agreement and the Exhibits attached hereto.
- (d) "Ancillary Agreements" means all of the written agreements, instruments, understandings, assignments or other arrangements entered into in connection with the transactions contemplated hereby and in the Distribution Agreement, including without limitation, the Distribution Agreement, the Employee Benefits Agreement, the Transition Support Agreement, the Intellectual Property Agreement, the Tax Sharing and Indemnification Agreement, and the Real Estate Agreements.
- (e) "Certegy Group" means the entities set forth on Exhibit A, and, at any point in time following the Distribution Date, a Person that is then an Affiliate of any such entity.
 - (f) "Certegy Indemnitees" has the meaning given in Section 6.2.
- (g) "Certegy Products" means those products, services and deliverables listed on Exhibit B. $\,$
 - (h) "Code" means the Internal Revenue Code of 1986, as amended.
- (i) "Company Information" means collectively the Proprietary Information and the Confidential Information of the disclosing party. Company Information also includes information that has been disclosed to Equifax or any of its Affiliates prior to the Effective Time, or to any member of either Group after the Effective Time, by a third party subject to an obligation to treat such information as confidential or secret.
- (j) "Confidential Information" means any and all confidential company business information of the disclosing party that does not constitute Proprietary Information and that is the subject of efforts by the disclosing party that are reasonable under the circumstances to maintain its secrecy and confidentiality, including without limitation, the Data furnished hereunder, the existence and nature of the relationship between the parties including, without limitation, the prices set forth in the Exhibits to this Agreement, employees of the disclosing party, and any and all additional information disclosed by the disclosing party to the receiving party as a result of the receiving party's access to and presence at the disclosing party's facilities.
 - (k) "Consumer Report" has the meaning given in the FCRA.
 - (1) "Consumer Reporting Agency" has the meaning given in the FCRA.

- (m) "DAT" means a personal computer or direct access terminal.
- (n) "Data" means information provided by either party or the members of its respective Group pursuant to this Agreement.
- (o) "Direct Damages" means actual, direct damages incurred by the claiming party which include, by way of example but without limitation, the costs to correct any deficiencies in the Data rendered by Provider and similar damages, but "Direct Damages" shall not include (A) loss of interest, profit or revenue of the claiming party or (B) incidental, consequential, special, indirect, punitive or exemplary damages suffered by the claiming party arising from or related to this Agreement, even if such party has been advised of the possibility of such losses or damages described in (A) and (B) above.
- (p) "Distribution" has the meaning given in the Distribution $\mbox{\sc Agreement.}$
- (q) "Distribution Agreement" means that certain Distribution Agreement entered into on or prior to the Distribution Date between Equifax and Certegy, as amended from time to time.
- (r) "Distribution Date" means the date upon which the Distribution shall be effective, as determined by the Board of Directors of Equifax, or such committee of such Board of Directors as shall be designated by the Board of Directors of Equifax.
 - (s) "Effective Time" means 11:59 p.m. Atlanta time on June 30, 2001.
- (t) "End User" means a Recipient or its Subscriber, which Person will use a given Report, or on whose behalf such Report will be used, in making a business decision concerning the subject of such Report.
- (u) "Equifax Group" means Equifax and its Affiliates existing at the Effective Time and as modified from time to time thereafter, excluding all members of the Certegy Group.
 - (v) "Equifax Indemnitees" has the meaning given in Section 6.1.
- (w) "Equifax Products" means those products, services and deliverables listed on Exhibit C. $\,$
- (x) "FCRA" means the federal Fair Credit Reporting Act, 15 U.S.C. Section 1681, et. seq.
 - (y) "GLBA" has the meaning given in Section 2.5.
 - (z) "Group" means the Certegy Group or the Equifax Group.
- (aa) "Indemnifiable Loss" means any and all Liability and expense (including, without limitation, reasonable expenses of investigation and reasonable attorneys' fees and expenses) incurred by an Indemnified Party in connection with any and all Actions or threatened Actions.

- (bb) "Indemnified Party" has the meaning given in Section 7.1.
- (cc) "Indemnifying Party" has the meaning given in Section 7.1.
- (dd) "Liabilities" means any and all claims, debts, liabilities and obligations, absolute or contingent, matured or not matured, liquidated or unliquidated, accrued or unaccrued, known or unknown, whenever arising, with respect to a specified object, matter, contract, commitment or undertaking, including without limitation, all debts, liabilities and obligations arising under any law, rule, regulation, action, order or consent decree of any governmental entity or any award of any arbitrator of any kind, related thereto and those arising under any contract, commitment or undertaking relating to such specified object, matter, contract, commitment or undertaking.
- (ee) "Permissible Purposes" means one of the permitted purposes for which a Consumer Report may be provided pursuant to the FCRA.
- (ff) "Person" means any individual, entity (whether or not formed as a corporation, sole proprietorship, partnership, limited liability company or other form of entity and whether or not conducting its affairs for profit), and any governmental entity or agency.
- (gg) "Prime Rate" shall have the meaning set forth in the Distribution Agreement.
- (hh) "Proprietary Information" means all non-public information, whether tangible or intangible, related to the services or business of the $\frac{1}{2}$ disclosing party that (i) derives economic value, actual or potential, from not being generally known to or readily ascertainable by another Person who can obtain economic value from its disclosure or use; and (ii) is the subject of efforts by the disclosing party that are reasonable under the circumstances to maintain its secrecy, including without limitation, (A) marking any information reduced to tangible form clearly and conspicuously with a legend identifying its confidential or proprietary nature; (B) identifying any oral communication as confidential immediately before, during, or after such oral communication; or (C) otherwise treating such information as confidential or secret. Assuming the criteria in clauses (i) and (ii) above are met, Proprietary Information includes information, without regard to form, including but not limited to, technical and nontechnical data, databases, formulas, patterns, designs, compilations, computer programs and software, devices, inventions, methods, techniques, drawings, processes, financial data, financial plans, product plans, lists of actual or potential customers and suppliers (which are not commonly known by or available to the public), research, development, and existing and future products.
- (ii) "Provider" means any member of either Group when providing Data to the members of the other Group pursuant to this Agreement.
- (jj) "Recipient" means any member of either Group when acquiring Data from the other Group pursuant to this Agreement.

- (kk) "Report" means Data regarding any Person, at the individual or aggregated levels, contained in any Certegy Product or Equifax Product, respectively, offered to any member of the other party's Group.
- (11) "Representatives" means, individually and collectively, officers, directors, employees, agents, and/or independent contractors of each member of the Group.
- (mm) "Subscriber" means any customer of a Recipient which obtains Data furnished hereunder from that Recipient; provided, however, that a Subscriber will be deemed to be a "Qualified Subscriber" only if the Subscriber meets the criteria therefor set forth in the Provider's Supplemental Terms in effect from time to time, as amended and/or modified by the Provider.
- (nn) "Supplemental Terms" means the Equifax Group's supplemental terms and conditions set forth in Exhibit D and the Certegy Group's supplemental terms and conditions set forth in Exhibit E, as amended from time to time during the term of this Agreement.
- (oo) "System Affiliate" means a Person that is not an Affiliate of Equifax that has entered into a written agreement with Equifax Information Services, LLC regarding the provision of consumer reporting services through the automated credit reporting service operated by Equifax Information Services, LLC.
 - (pp) "Third Party Claim" has the meaning given in Section 7.2.

ARTICLE II

DATA PURCHASE

Section 2.1. General.

- (a) During the term hereof, each provision of Data by any member of one Group to any member of the other Group will be subject to the terms and conditions of this Agreement and the Supplemental Terms applicable to the elements comprising the Data as amended and/or modified from time to time during the term of this Agreement by the Provider. The Data which are provided by members of one Group to the members of the other Group as of the Distribution Date, and which will continue to be provided during the term of this Agreement, are set forth on Exhibits B and C.
- (b) The parties acknowledge that the Data to be provided by the Equifax Group may contain Data owned by the Equifax Group as well as by System Affiliates, and that the Equifax Group can only provide Data containing information owned by its System Affiliates to the extent that the System Affiliates authorize the provision of such information and subject to the applicable charges to be paid by, and payment terms for, the Certegy Group as established by the applicable System Affiliate.
- (c) The parties have each exerted their best efforts to identify each type of Data presently provided by the members of one Group to the members of the other Group and to address such Data in this Agreement or in an appropriate Ancillary Agreement. The parties

acknowledge that there may be Data provided by the members of one Group to the members of the other Group which are not addressed in this Agreement, but agree that all such items are intended to be governed by this Agreement except to the extent that such items are governed by an Ancillary Agreement. Moreover, the parties agree: that the Data provided may require adjustments during the term of this Agreement to reflect the evolving business and operations of each Group and applicable law; that the relationship memorialized by this Agreement is dynamic in nature and will evolve as the operating and business environment of each Group changes and evolves; and that the scope of the Data that will be provided during the term and the corresponding fees charged and payment terms extended by the parties may need to be modified to reflect the foregoing. The parties agree to cooperate and negotiate with each other in good faith in order to modify this Agreement as appropriate to give effect to the intent of the parties and to satisfy the requirements of each party for additional data; provided, however, that no member of either group shall be obligated to provide Data that is not identified on Exhibit B or C, as applicable, or to provide Data that is identified on Exhibit B or C, as applicable, for a purpose that is not identified on such Exhibit, except as provided in Section 2.1(d).

- (d) For the one hundred-eighty (180) days following the Effective Time, each party shall have the right to inventory, validate and update any information and the services, products and deliverables that comprise the Data that are reflected in or omitted from this Agreement and attached Exhibits. If the parties determine that this Agreement or the Exhibits hereto contain discrepancies from the intent of this Agreement, this Agreement and/or Exhibits shall be promptly changed, modified, updated and adjusted to correct such discrepancies, so that this Agreement and/or Exhibits will be correct and accurately reflect the Data, and attendant charges and payment terms, provided by one Group to the other Group at the Effective Time. In the event that either party discovers such an omission from this Agreement or a discrepancy from the intent hereof: (a) that party will promptly notify the other party and the parties will promptly negotiate in good faith to establish the specific terms and conditions applicable to such Data; (b) this Agreement will govern the general terms and conditions applicable to the provision of such Data; and (c) the Provider will not cease to provide such Data, unless the parties are unable to agree upon the specific terms and conditions applicable to such Data. If either party disputes the existence of a discrepancy identified by the other party, the parties will submit the matter for dispute resolution as specified in Section 11.10.
- (e) Subject to Article IV, each Provider will perform its obligations in a commercially reasonable manner, and in accordance with any service levels specified in a particular Exhibit.
- (f) This Agreement is nonexclusive. Each Provider shall furnish Data on an as-ordered basis only. Each Recipient shall have the right to satisfy any and all of its requirements for information of any type from any third party in addition to, or in lieu of, any Data it is permitted to purchase hereunder.

Section 2.2. Supplemental Terms.

The terms and conditions applicable to any given provision of Data pursuant to this Agreement will vary depending on (a) whether such Data are regulated pursuant to the FCRA or other law or regulation, (b) the type of Data purchased, (c) the role in which the

Recipient purchases such Data, including without limitation, as a Non-Consumer Reporting Agency, Consumer Reporting Agency or End User, (d) the purpose for which the Recipient's Subscriber ordered the products, services and deliverables based upon such Data, and (e) the third party limitations and restrictions applicable to the provision of the Data by the Provider. Accordingly, the terms and conditions applicable to any given acquisition of Data by any Recipient pursuant to this Agreement are supplemented by the terms and conditions of the Supplemental Terms with respect to the type of Data purchased, the role in which the Recipient purchases such Data and the purpose for which the Recipient's Subscriber ordered the products, services and deliverables that are based upon such Data, as such Supplemental Terms are modified and/or amended from time to time during the term of this Agreement by the Provider. In the event of a conflict between the terms and conditions set forth in the body of this Agreement and the specific terms and conditions set forth in the Provider's Supplemental Terms, the terms and conditions set forth in the Supplemental Terms will be controlling; provided, however if any provision contained in Articles III, IV, V, VI, VIII, VIII, X and/or XI conflicts with and/or overlaps in subject matter with any provision contained in the Supplemental Terms, the terms and conditions set forth in such Articles will prevail and be controlling.

Section 2.3. FCRA Certification.

Certegy and Equifax each certify, and shall ensure that each member of the Certegy Group and Equifax Group shall certify, that it will order Reports that are Consumer Reports only for use for a Permissible Purpose in accordance with the FCRA, as amended from time to time.

Section 2.4. Effect of Data Purchase.

Notwithstanding anything in this Agreement to the contrary, each Recipient of Data hereunder acquires only the right to use such Data for the use or uses specified on Exhibit B or C, as applicable. Except as specifically set forth in Exhibit B or C, as applicable, the Recipient is not authorized to resell Data purchased hereunder. Except for the rights expressly granted in this Agreement and the applicable Exhibits, each Provider reserves all right, title and interest in and to the Data it provides hereunder.

Section 2.5. Effect of Change in Law.

Notwithstanding anything to the contrary in this Agreement, if the continued provision of all or any portion of the Data becomes impossible, impractical, or undesirable due to (i) a change in applicable national, state or local laws or regulations, (ii) a change in the regulatory enforcement of such laws or regulations, (iii) a change in the Provider's good faith interpretation of laws or regulations, presently in force or subsequently enacted, or (iv) pending or threatened litigation, the Provider may cease to provide the affected Data within, or pertaining to persons residing within, the affected jurisdiction. The Provider will attempt to provide written notice of its actions as far in advance of the effective date as is reasonably possible under the circumstances. In particular, but not in limitation of the foregoing, upon and with the effective date of the Gramm-Leach-Bliley Act, 15 U.S.C. Section 6801, et seq. ("GLBA") and its implementing regulations, the performance of any covenant or term hereof shall be required only insofar as the use, disclosure or provision of any Data in the United States that does not comprise

a Consumer Report will comply with Title V of the GLBA and the implementing regulations issued thereunder and shall be pursuant to Section 6802(c) or one of the general exceptions of Section 6802(e) of the GLBA and the corresponding provisions of the applicable regulations, and no Provider shall have any liability hereunder for any failure to perform which is excused in accordance with the foregoing terms.

Section 2.6. Obligations of Providers and Recipients.

- (a) Certegy agrees that it will cause each member of the Certegy Group that is designated as a Provider or a Recipient of Data in Exhibit B or C to assume the obligations applicable to such Provider or Recipient, effective as of the Effective Time, and to timely perform, pay or discharge all such obligations.
- (b) Equifax agrees that it will cause each member of the Equifax Group that is designated as a Provider or a Recipient of Data in Exhibit B or C to assume the obligations applicable to such Provider or Recipient, effective as of the Effective Time, and to timely perform, pay or discharge all such obligations.

ARTICLE III

FEES AND PAYMENT

Section 3.1. Fees and Charges.

Each Recipient will pay the Provider for all Data supplied to such Recipient according to the rates set forth on Exhibits B and C during the two-year term of this Agreement and according to such rates as are subsequently established by the Provider from time to time thereafter.

Section 3.2. Payment Terms.

Each Recipient shall pay all invoices for Data requested by such Recipient within thirty (30) days after the date of each invoice.

Section 3.3. Disputed Amounts.

In the event that a member of either Group disputes the accuracy or applicability of a charge or credit by a member of the other Group, such member will notify Equifax or Certegy, respectively, of such dispute prior to the date on which the charge is to be paid or the credit issued or as soon as practicable after the discrepancy has been discovered after the applicable invoice is paid or credit is issued. The parties will investigate and resolve such disputes using the dispute resolution process provided in Section 11.10 of this Agreement. Any undisputed amounts contained on the invoice will be paid pursuant to Section 3.2.

Section 3.4. Interest

Any and all amounts not paid when due shall bear interest on a day to day basis from the due date to the date of payment (both dates inclusive) at the lower rate of one and one-half percent (1 1/2%) per month or the highest rate allowable under applicable law.

Section 3.5. Taxes.

All fees and charges payable under this Agreement are exclusive of all federal, state and local sales, excise, use, value added and similar taxes not measured by the income of Provider, which taxes shall be the sole responsibility of Recipient and shall be paid by Recipient at the rate and in the manner prescribed by applicable law and Section 3.2 hereof; provided, however, if required by applicable law, Provider shall separately itemize such taxes on Provider's invoice to Recipient, Recipient shall pay the amount of such taxes to Provider and Provider shall remit such amount to the applicable taxing authority. In lieu thereof, Recipient shall, where applicable, furnish Provider with a properly executed tax exemption certificate acceptable to the applicable taxing authority.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES

The parties and each Recipient and Provider recognize that every business decision represents an assumption of risk and that neither party, nor the members of either party's Group, in furnishing the Data to the other Group underwrites or assumes the other's risk in any manner. NEITHER PARTY, NOR ANY MEMBER OF EITHER PARTY'S GROUP NOR ANY OF ITS INFORMATION SOURCES, GUARANTEES OR WARRANTS THE CORRECTNESS, COMPLETENESS, CURRENTNESS, MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE OF THE DATA PROVIDED TO ANY MEMBER OF THE OTHER PARTY'S GROUP. NEITHER PARTY, NOR ANY OF ITS REPRESENTATIVES, INFORMATION SOURCES, LICENSORS, SYSTEM AFFILIATES OR MEMBERS OF ITS RESPECTIVE GROUP WILL BE LIABLE TO ANY OTHER PARTY OR ANY OF ITS REPRESENTATIVES, INFORMATION SOURCES, LICENSORS, SYSTEM AFFILIATES OR MEMBERS OF THE OTHER PARTY'S GROUP FOR ANY LOSS OR INJURY ARISING OUT OF, OR CAUSED IN WHOLE OR IN PART BY, ITS ACTS OR OMISSIONS IN PROCURING, COMPILING, COLLECTING, INTERPRETING, PROCESSING, REPORTING OR TRANSMITTING ANY INFORMATION OR THE DATA, EVEN IF DUE TO NEGLIGENCE. THE DATA ARE PROVIDED ON AN "AS IS" BASIS.

No Provider makes any warranties whatsoever in connection with the performance of the DAT, or any other means of communication or provision of the Data from one party to the other, and no Provider will be responsible for transmission distortion, interruptions or failures of the DAT, or any other means of communication or provision of the Data, or of any Data. Each Recipient will use a secure means to deliver Data received hereunder to its Subscribers, unless otherwise agreed by the Provider, and will not deliver such information by means of any publicly accessible network (including without limitation, the Internet) without Provider's express written permission.

ARTICLE V

LIMITATION ON ACTIONS AND LIABILITIES

Section 5.1. Covenant Not to Sue.

Each member of each Group covenants not to sue or maintain any claim, cause of action, demand, cross-action, counterclaim, third-party action or other form of pleading against a Provider or a Provider's Representatives, sources of information, affiliated Credit Reporting Agencies, and/or System Affiliates (if applicable) arising out of or relating in any way to the accuracy or inaccuracy and/or validity or invalidity of any of the Data received hereunder, even if caused by the negligence of any such Person; provided, however, this Section shall not be applicable if the injury is caused by the willful misconduct of such Person.

Section 5.2. Release.

Each member of each Group recognizes that the accuracy of any Data received hereunder furnished is not warranted or guaranteed by the Provider thereof for any purpose, and releases the Provider and the Provider's Representatives, sources of information, affiliated Credit Reporting Agencies and System Affiliates (if applicable) from all liability in connection with the accuracy or inaccuracy and/or validity or invalidity of the Provider Information and from any and all charge, damage, loss and/or expense suffered by Recipient resulting directly or indirectly from such matters, even if caused by the negligence of any such Person; provided, however, this sentence shall not be applicable if the injury is caused by the willful misconduct of such Person. Each member of each Group shall include in its agreements with its Subscribers a release concerning the information received from any Provider substantially equivalent to the first sentence of this paragraph; provided that if a Group member currently has Subscriber contracts in place, or standard contract forms, that omit such a release, it will not be in breach of this section, but that member will use its best efforts to incorporate such a release in its future contracts as soon as reasonably practicable.

Section 5.3. Limitation of Liability.

- (a) Unless specifically provided to the contrary in this Agreement, neither party shall, nor shall any member of the party's respective Group, have any liability to the other party for any damages other than Direct Damages whether based on contract, tort (including without limitation, that caused by negligence, but excluding willful misconduct), warranty, guarantee or any other legal or equitable grounds. The liability of any member of either Group to the other Group arising out of or resulting from all breaches by such member and/or its Representatives of its obligations under this Agreement during any Contract Year shall not exceed the fees received by such member from the other Group during such Contract Year. "Contract Year" means a twelve month period beginning at the Effective Time or the first anniversary of the Effective Time.
 - (b) The limitations set forth in Section 5.3(a) shall not apply to:

- (i) Any failure by any member of either Group to pay fees and expenses for the Data that are due and payable hereunder up to the effective date of the termination of the Agreement;
- (ii) Claims arising under Section 5.1 (Covenant Not to Sue) and/or Section 5.2 (Release);
 - (iii) Indemnification obligations under Article VI; and
- (iv) Any incident or event resulting in damages, charges and/or losses caused by the action or inaction of a member of either Group constituting willful misconduct.
- (c) Neither party nor the members of its Group shall be liable for any damages to the other party or the members of its Group if and to the extent caused by the failure of such other party or the members of its Group to perform its responsibilities under this Agreement.
- (d) NOTWITHSTANDING ANYTHING TO THE CONTRARY IN ANY SECTION OF THIS AGREEMENT EXCEPT THIS SECTION 5.3, INCLUDING ANY AND ALL FUTURE AMENDMENTS AND ADDENDA, NEITHER PARTY, NOR ANY OF ITS REPRESENTATIVES, INFORMATION SOURCES, LICENSORS, SYSTEM AFFILIATES OR MEMBERS OF ITS RESPECTIVE GROUP WILL HAVE ANY LIABILITY, WHETHER BASED ON CONTRACT, TORT (INCLUDING, WITHOUT LIMITATION, NEGLIGENCE OR STRICT LIABILITY), WARRANTY OR ANY OTHER LEGAL OR EQUITABLE GROUNDS FOR ANY PUNITIVE, CONSEQUENTIAL, INCIDENTAL, INDIRECT, EXEMPLARY OR SPECIAL DAMAGES, INCLUDING, WITHOUT LIMITATION, LOSS OF DATA, PROFITS, INTEREST OR REVENUE, OR INTERRUPTION OF BUSINESS, WHETHER OR NOT SUCH PARTY HAS BEEN INFORMED OF OR MIGHT OTHERWISE HAVE ANTICIPATED OR FORESEEN THE POSSIBILITY OF SUCH DAMAGES.

ARTICLE VI

INDEMNIFICATION

Section 6.1. Certegy Indemnification of the Equifax Group.

If the Distribution occurs, on and after the Distribution Date, Certegy shall indemnify, defend and hold harmless each member of the Equifax Group and each of their respective Representatives (the "Equifax Indemnitees") from and against any and all Indemnifiable Losses incurred or suffered by any of the Equifax Indemnitees and arising out of, or due to, (a) any breach of Sections 5.1 (Covenant Not to Sue), 5.2 (Release) or 8.2 (Confidentiality) by a member of the Certegy Group; (b) any Third Party Claim arising out of or in connection with any breach of Section 2.3 or any Supplemental Terms relating to permitted uses of Data or compliance with laws applicable to the Data by a member of the Certegy Group, or a breach of any certification made by a member of the Certegy Group in the Supplemental Terms; or (c) any Third Party Claim alleging that Data supplied by a Provider that is a member of the Certegy Group, or the act of providing such Data, infringes upon a copyright or other

intellectual property right under the law of the country in which such Data is intended to be used; except such damages, loss, liability and expense resulting from the willful misconduct of a member of the Equifax Group.

Section 6.2. Equifax Indemnification of the Certegy Group.

If the Distribution occurs, on and after the Distribution Date, Equifax shall indemnify, defend and hold harmless each member of the Certegy Group and each of their respective Representatives (the "Certegy Indemnitees") from and against any all Indemnifiable Losses incurred or suffered by any of the Certegy Indemnitees and arising out of, or due to, (a) any breach of Sections 5.1 (Covenant Not to Sue), 5.2 (Release) or 8.2 (Confidentiality) by a member of the Equifax Group; (b) any Third Party Claim arising out of or in connection with any breach of Section 2.3 or any Supplemental Terms relating to permitted uses of Data or compliance with laws applicable to the Data by a member of the Equifax Group, or a breach of any certification made by a member of the Equifax Group in the Supplemental Terms; or (c) any Third Party Claim alleging that Data supplied by a Provider that is a member of the Equifax Group, or the act of providing such Data, infringes upon a copyright or other intellectual property right under the law of the country in which such Data is intended to be used; except such damages, loss, liability and expense resulting from the willful misconduct of a member of the Certegy Group.

Section 6.3. Insurance and Third Party Obligations.

No insurer or any other third party shall be, by virtue of the foregoing indemnification provisions, (a) entitled to a benefit it would not be entitled to receive in the absence of such provisions, (b) relieved of the responsibility to pay any claims to which it is obligated, or (c) entitled to any subrogation rights with respect to any obligation hereunder.

ARTICLE VII

INDEMNIFICATION PROCEDURES

Section 7.1. Notice and Payment of Claims.

If any Equifax or Certegy Indemnitee (the "Indemnified Party") determines that it is or may be entitled to indemnification by a party (the "Indemnifying Party") under Article VI (other than in connection with any Action or claim subject to Section 7.2), the Indemnified Party shall deliver to the Indemnifying Party a written notice specifying, to the extent reasonably practicable, the basis for its claim for indemnification and the amount for which the Indemnified Party reasonably believes it is entitled to be indemnified. After the Indemnifying Party shall have been notified of the amount for which the Indemnified Party seeks indemnification, the Indemnifying Party shall, within thirty (30) days after receipt of such notice, pay the Indemnified Party such amount in cash or other immediately available funds (or reach agreement with the Indemnified Party as to a mutually agreeable alternative payment schedule) unless the Indemnifying Party objects to the claim for indemnification or the amount thereof. If the Indemnifying Party does not give the Indemnified Party written notice objecting to such claim and setting forth the grounds therefor within the same thirty (30) day period, the Indemnifying Party shall be deemed to have acknowledged its liability for such claim and the Indemnified

Party may exercise any and all of its rights under applicable law to collect such amount. Any amount owed under this Section 7.1 that is past due shall bear interest at a simple rate of interest per annum equal to the Prime Rate plus 2%.

Section 7.2. Notice and Defense of Third Party Claims.

- (a) Promptly following the earlier of (i) receipt of notice of the commencement by a third party of any Action against or otherwise involving any Indemnified Party or (ii) receipt of information from a third party alleging the existence of a claim against an Indemnified Party, in either case, with respect to which indemnification may be sought pursuant to Article VI of this Agreement (a "Third Party Claim"), the Indemnified Party shall give the Indemnifying Party written notice thereof. The failure of the Indemnified Party to give notice as provided in this Section 7.2 shall not relieve the Indemnifying Party of its obligations under this Agreement, except to the extent that the Indemnifying Party is prejudiced by such failure to give notice. Within thirty (30) days after receipt of such notice, the Indemnifying Party shall by giving written notice thereof to the Indemnified Party, (i) acknowledge, as between the parties hereto, liability for, and at its option, assumption of the defense of such Third Party Claim at its sole cost and expense or (ii) object to the claim of indemnification set forth in the notice delivered by the Indemnified Party pursuant to the first sentence of this Section 7.2 setting forth the grounds therefor; provided that if the Indemnifying Party does not within the same thirty (30) day period give the Indemnified Party written notice acknowledging liability and electing to assume the defense or objecting to such claim and setting forth the grounds therefor, the Indemnifying Party shall be deemed to have acknowledged, as between the parties hereto, its liability to the Indemnified Party for such Third Party Claim.
- (b) Any contest of a Third Party Claim as to which the Indemnifying Party has elected to assume the defense shall be conducted by attorneys employed by the Indemnifying Party and reasonably satisfactory to the Indemnified Party; provided that the Indemnified Party shall have the right to participate in such proceedings and to be represented by attorneys of its own choosing at the Indemnified Party's sole cost and expense. If the Indemnifying Party assumes the defense of a Third Party Claim, the Indemnifying Party may settle or compromise the claim without the prior written consent of the Indemnified Party; provided that the Indemnifying Party may not agree to any such settlement pursuant to which any remedy or relief, other than monetary damages for which the Indemnifying Party shall be responsible hereunder, shall be applied to or against the Indemnified Party, without the prior written consent of the Indemnified Party, which consent shall not be unreasonably withheld.
- (c) If the Indemnifying Party does not assume the defense of a Third Party Claim for which it has acknowledged liability for indemnification that arises under Article VI, the Indemnified Party may require the Indemnifying Party to reimburse it on a current basis for its reasonable expenses of investigation, reasonable attorneys' fees and reasonable out-of-pocket expenses incurred in defending against such Third Party Claim, and the Indemnifying Party shall be bound by the result obtained with respect thereto by the Indemnified Party; provided that the Indemnifying Party shall not be liable for any settlement effected without its consent, which consent shall not be unreasonably withheld. The Indemnifying Party shall pay to the Indemnified Party in cash the amount for which the Indemnified Party is entitled to be indemnified (if any) within fifteen (15) days after the final resolution of such Third Party Claim

(whether by settlement, compromise, or the final nonappealable judgment of a court of competent jurisdiction or otherwise), or, in the case of any Third Party Claim as to which the Indemnifying Party has not acknowledged liability, within fifteen (15) days after such Indemnifying Party's objection has been resolved by settlement, compromise or the final nonappealable judgment of a court of competent jurisdiction.

ARTICLE VIII

CONFIDENTIALITY

Section 8.1. Exclusions.

Notwithstanding anything to the contrary contained in this Agreement, "Company Information" does not include any information that before being divulged by the receiving party (a) has become generally known to the public through no wrongful act of the receiving party; (b) has been rightfully received by the receiving party from a third party without restriction on disclosure and without, to the knowledge of the receiving party, a breach of an obligation of confidentiality running directly or indirectly to the disclosing party; (c) has been approved for release to the general public by a written authorization of the disclosing party; (d) has been independently developed by the receiving party without use, directly or indirectly, of Company Information received from the disclosing party; or (e) has been furnished to a third party by the disclosing party without restrictions on the third party's rights to disclose the information.

Section 8.2. Confidentiality.

- (a) Each party acknowledges, and shall cause each member of its Group to acknowledge, that it is in possession of significant confidential or proprietary information concerning the business, operations and tangible and assets of the members of the other Group.
- (b) Each party shall, and shall ensure that each member of its Group shall, (i) protect the Company Information of the other Group from disclosure and in no event take any action causing, or fail to take the action necessary in order to prevent, any such Company Information to lose its character as Company Information; (ii) exercise at a minimum the same care it would exercise to protect its own highly confidential information (which shall in no event be less than a reasonable standard of care); and (iii) not use, reproduce, distribute, disclose, or otherwise disseminate the Company Information of the other party (x) except as authorized pursuant to this Agreement and in the Supplemental Terms attached hereto, as amended from time to time, or any Ancillary Agreement, or (y) except pursuant to a requirement of a governmental agency or of law, without similar restrictions or other protections against public disclosure; provided, however, with respect to disclosures pursuant to (y) above the receiving party must first give written notice of such required disclosure to the disclosing party, take reasonable steps to allow the disclosing party to seek to protect the confidentiality of the Company Information required to be disclosed, make a reasonable effort to obtain a protective order requiring that the Company Information so disclosed be used only for the purposes for which disclosure is required, and shall disclose only that part of the Company Information which, in the written opinion of its legal counsel, it is required to disclose. Any and all reproductions of such Company Information must prominently contain a confidential legend.

- (c) The receiving party may make disclosures of the Company Information of the disclosing party only to Representatives of the receiving party (i) who have a specific need to know such information; and (ii) who the receiving party has obligated under a written agreement to hold such Company Information in trust and in strictest confidence and otherwise to comply with the terms and provisions of this Agreement or terms and conditions substantially similar to and implementing the same restrictions and covenants as those set forth in this Agreement. Certegy and Equifax agree, and shall ensure that each member of their respective Groups agrees, to diligently monitor each such Representative, diligently enforce such agreements with its Representatives, and, upon request by the other party, promptly to furnish to the other party a list of the receiving party's Representatives having had access to such Company Information.
- (d) Within thirty (30) days following the receipt of a written request from the disclosing party, the receiving party must deliver to the disclosing party all tangible materials containing or embodying the disclosing party's Company Information applicable to a triggering event, together with a certificate executed by the president or any vice president of the receiving party certifying that all such materials in the receiving party's possession have been delivered to the disclosing party or, at the disclosing party's option, certify that all such materials in the receiving party's possession have been destroyed. For the purposes of this Article VIII, a "triggering event" means the cessation of the provision of Data by the disclosing party to the receiving party under this Agreement or the termination of the Agreement.
- (e) The covenants of confidentiality set forth in this Agreement (i) will apply after the Effective Time to all Company Information disclosed to the receiving party before and after the Distribution Date and (ii) will continue and must be maintained from the Effective Time through the termination of the relationship under this Agreement between Equifax and Certegy and (A) with respect to Proprietary Information, at any and all times after the termination of the relationship under this Agreement between Equifax and Certegy during which such Proprietary Information retains its status as a "trade secret" under applicable law; and (B) with respect to Confidential Information, for three (3) years after termination of the relationship under this Agreement between Equifax and Certegy.

Section 8.3. Employee Confidentiality Agreements.

The members of each Group have entered into confidentiality and non-disclosure agreements with their respective employees. To the extent that any employee during or after employment violates such agreement and such violation is or may in the future be to the detriment of the other Group, at the written request of the affected party, the other party shall promptly take remedial measures with such employee if and to the extent reasonable under the circumstances to preserve the value of the Data and to enforce its obligations hereunder. The party employing the affected employee shall have the unilateral right to determine the forum for, the manner of proceeding in, and legal counsel for such action and shall be entitled to any damages or other relief against such employee awarded in such action to the extent related to such party's Data or business. Such enforcement against and recovery by a party from its breaching employee shall not constitute a release or sole remedy for the members of the other Group injured by such breaching employee's actions, and such members of the other Group may bring a claim against the party employing the breaching employee for a breach of

this Agreement. Each party shall bear all of its own out-of-pocket costs of pursuing such action and the other party shall cooperate in connection therewith.

Section 8.4. Rights and Remedies.

- (a) If either party, or any member of the Group, should breach or threaten to breach any of the provisions of this Article VIII, the non-breaching party, in addition to any other remedies it may have at law or in equity, will be entitled to a restraining order, injunction, or other similar remedy in order to specifically enforce the provisions of this Article VIII. Each party specifically acknowledges, and shall cause each member of its respective Group to acknowledge, that money damages alone would be an inadequate remedy for the injuries and damage that would be suffered and incurred by the non-breaching party as a result of a breach of any of the provisions of this Article VIII. In the event that either party, or a member of such party's Group, should seek an injunction hereunder, the other party hereby waives, and shall cause each member of its Group to waive, any requirement for the submission of proof of the economic value of any Company Information or the posting of a bond or any other security. In the event of a dispute between the parties under this Article VIII, the non-prevailing party shall pay all costs and expenses associated with resolving the dispute, including, but not limited to, reasonable attorneys' fees.
- (b) The receiving party shall notify the disclosing party upon discovery of any unauthorized use or disclosure of Company Information, or any other breach of Article VIII of this Agreement by the receiving party, or any Representative of the receiving party's Group, and will cooperate with the disclosing party in every reasonable way to help the disclosing party regain possession of its Company Information and prevent its further unauthorized use or disclosure. The receiving party shall be responsible for the acts of any Representative of its Group that are in violation of this Article VIII.

Section 8.5. Competitive Activities.

Subject to the rights and obligations set forth in Article VIII, each party understands and acknowledges that the other party's Group may now market or have under development products that are competitive with products or services now offered or that may be offered by it and/or the members of its Group, and the parties' communications hereunder will not serve to impair the right of either party or any member of its respective Group to independently develop, make, use, procure, or market products or services now or in the future that may be competitive with those offered by the other party's Group, nor require either party and/or the members of its Group to disclose any planning or other information to the other party.

Section 8.6. No Implied Rights.

All Company Information is and shall remain the property of the disclosing party and/or the member of its respective Group. By disclosing Company Information to the receiving party's Group, the disclosing party and/or the members of its respective Group do(es) not grant any express or implied rights or license to the receiving party's Group to or under any patents, patent applications, inventions, copyrights, trademarks, trade secret information, or other

intellectual property rights heretofore or hereafter possessed by the disclosing party and/or the members of its respective Group.

ARTICLE IX

CONTINUED ASSISTANCE

Section 9.1. Continued Assistance and Transition.

(a) Following the Effective Time, Equifax and Certegy shall, and shall cause each member of their respective Groups to, cooperate in the orderly purchase and sale of Data hereunder. From time to time, at Equifax's or Certegy's request and without further consideration, the other party shall, and shall cause each member of the other party's Group, as applicable, to execute, acknowledge and deliver such documents, instruments or assurances and take such other action as the requesting party may reasonably request to more effectively evidence compliance with the Supplemental Terms applicable to the purchase and sale of Data as reflected in the Supplemental Terms for the applicable Data in effect from time to time, as amended and/or modified by the Provider.

Section 9.2. Litigation Cooperation.

- (a) Upon written request, Equifax and Certegy shall, and shall cause each member of its Group to, use reasonable efforts to cooperate in the evaluation and defense of third party Actions regarding (i) Data provided by the other party or a member of its Group hereunder or (ii) arising out of the business of the other party or of any member of the other party's Group prior to the Distribution Date in which the requesting party or any member of its Group may from time to time be involved, at the cost and expense of the requesting party. Such cooperation shall include, without limitation, making its Representatives available as witnesses or consultants to the extent that such persons may reasonably be required in connection with such third party Actions.
- (b) Equifax and Certegy shall, and shall cause each member of its respective Group, to use reasonable efforts to notify the other if it learns of a potential or actual third party claim related to the Data provided pursuant to this Agreement to be brought against any member of the other's Group.

ARTICLE X

TERM AND TERMINATION

Section 10.1. Term; Termination

- (a) The term of this Agreement shall commence at the Effective Time and shall continue for two (2) years.
- (b) Notwithstanding Section 10.1(a) above, either Equifax or Certegy, as the Provider, may terminate provision of Data to any given Recipient if such Recipient materially breaches any of the terms hereof, including without limitation the applicable Supplemental

Terms, and the breach is not cured within thirty (30) days after written notice of breach is delivered to the breaching Recipient; provided, however, that if the breach is not capable of being cured within such thirty (30) day period and the breaching party is proceeding to cure the breach with reasonable diligence, the cure period shall be extended to sixty (60) days.

- (c) Notwithstanding Section 10.1(a) above, this Agreement may be terminated in its entirety in accordance with any of the following:
 - (i) Upon written agreement of the parties;
- (ii) By either Certegy or Equifax for material breach by the other of any of the terms hereof if the breach is not cured within thirty (30) calendar days after written notice of breach is delivered to the breaching party; provided, however, that if the breach is not capable of being cured within such thirty (30) day period and the breaching party is proceeding to cure the breach with reasonable diligence, the cure period shall be extended to sixty (60) days;
- (iii) By either Certegy or Equifax, upon written notice to the other if the other becomes insolvent or makes an assignment of substantially all of its assets for the benefit of creditors, or is placed in receivership, reorganization, liquidation or bankruptcy;
- (iv) By Equifax, upon written notice to Certegy, if, for any reason, the ownership or control of Certegy or any of Certegy's operations becomes vested in, or is made subject to the control or direction of, any direct competitor of Equifax or one of its subsidiaries, but such termination shall be applicable only with respect to Data provided by Equifax Group members to Recipients that have been affected by the change in control; or
- (v) By Certegy, upon written notice to Equifax, if, for any reason, the ownership or control of Equifax or any of Equifax's operations becomes vested in, or is made subject to the control or direction of, any direct competitor of Certegy or one of its subsidiaries, but such termination shall be applicable only with respect to Data provided by Certegy Group members to Recipients that have been affected by the change in control.
- (d) Upon any termination pursuant to Sections 10.1(b) or 10.1(c) above, the involved Providers shall be compensated for all Data provided up to the date of termination in accordance with the provisions of this Agreement.
 - Section 10.2. Effect of Termination and Expiration.

Upon termination or expiration of the term of this Agreement, all rights and obligations of the parties under this Agreement will immediately cease and terminate (except for the rights and obligations pursuant to Articles IV (Representations and Warranties), V (Limitations on Actions and Liabilities), VI (Indemnification), VII (Indemnification Procedures), and VIII (Confidentiality) and XI (Miscellaneous) and Sections 9.2 and 10.2, and the definitions required thereby which will survive such termination or expiration), and neither party will have

any further obligation to the other party with respect to this Agreement, except for (a) fees and reimbursable expenses payable to the other party accrued but unpaid at the date of termination or expiration, and (b) the provisions of this Agreement which are specifically designated herein as surviving such termination or expiration.

ARTICLE XI

MISCELLANEOUS

Section 11.1. Expenses.

Except as specifically provided in this Agreement or any Ancillary Agreement, all costs and expenses incurred in connection with the preparation, execution, delivery and implementation of this Agreement and the Ancillary Agreements shall be paid by Equifax.

Section 11.2. Notices.

All notices and communications under this Agreement shall be deemed to have been given (a) when received, if such notice or communication is delivered by facsimile, hand delivery or overnight courier, or (b) three (3) business days after mailing if such notice or communication is sent by United States registered or certified mail, return receipt requested, first class postage prepaid. All notices and communications, to be effective, must be properly addressed to the party to whom the same is directed at its address as follows:

If to Equifax, to:

Equifax Inc. 1550 Peachtree Street, N.W. Atlanta, Georgia 30309 Attention: Phillip J. Mazzilli, Chief Financial Officer Fax: (404) 885-8988

with a copy to:

Equifax Inc. 1550 Peachtree Street, N.W. Atlanta, Georgia 30309 Attention: Kent E. Mast, Vice President & General Counsel Fax: (404) 885-8988

If to Certegy, to:

Certegy Inc. 555 North Point Centre East Alpharetta, Georgia 30022 Attention: Bruce Richards, Corporate Vice President, General Counsel and Secretary

Fax:

with a copy to:

Certegy Inc. P.O. Box 349 Alpharetta, G

Alpharetta, Georgia 30009

Attention: Michael T. Vollkommer, Corporate Vice President

and Chief Financial

Officer
Fax: _____

Either party may, by written notice delivered to the other party in accordance with this Section 11.2, change the address to which delivery of any notice shall thereafter be made.

Section 11.3. Amendment and Waiver.

This Agreement may not be altered or amended, nor may any rights hereunder be waived, except by an instrument in writing executed by the party or parties to be charged with such amendment or waiver. No waiver of any terms, provision or condition of or failure to exercise or delay in exercising any rights or remedies under this Agreement, in any one or more instances, shall be deemed to be, or construed as, a further continuing waiver of any such term, provision, condition, right or remedy or as a waiver of any other term, provision or condition of this Agreement.

Section 11.4. Entire Agreement.

This Agreement and Exhibits constitute the entire understanding of the parties hereto with respect to the subject matter hereof, superseding all negotiations, prior discussions and prior agreements and understandings relating to such subject matter.

Section 11.5. Parties in Interest.

Neither of the parties hereto may assign its rights or delegate any of its duties under this Agreement without the prior written consent of each other party. This Agreement shall be binding upon, and shall inure to the benefit of, the parties hereto and their respective successors and permitted assigns. Without limiting the generality of the foregoing statement, the parties acknowledge that certain of the legal entities designated as Providers and Recipients in Exhibits B and C are intended to be renamed, formed or reorganized prior to or concurrently with the Distribution, and this Agreement shall be binding upon, and shall inure to the benefit of, the successors of such designated Providers and Recipients. Nothing contained in this Agreement, express or implied, is intended to confer any benefits, rights or remedies upon any person or entity other than members of the Equifax Group and the Certegy Group and the Equifax Indemnitees and Certegy Indemnitees under Articles VI and VII hereof.

Section 11.6. Further Assurances and Consents.

In addition to the actions specifically provided for elsewhere in this Agreement, each of the parties hereto will use its reasonable efforts to (a) execute and deliver such further instruments and documents and take such other actions as any other party may reasonably

request in order to effectuate the purposes of this Agreement and to carry out the terms hereof and (b) take, or cause to be taken, all actions, and do, or cause to be done, all things reasonably necessary, proper or advisable under applicable laws, regulations and agreements or otherwise to consummate and make effective the transactions contemplated by this Agreement, including without limitation, using its reasonable efforts to obtain any consents and approvals and to make any filings and applications necessary or desirable in order to consummate the transactions contemplated by this Agreement; provided that no party hereto shall be obligated to pay any consideration therefor (except for filing fees and other similar charges) to any third party from whom such consents, approvals and amendments are requested or to take any action or omit to take any action if the taking of or the omission to take such action would be unreasonably burdensome to the party or its Group or the business thereof.

Section 11.7. Severability.

The provisions of this Agreement are severable and should any provision hereof be void, voidable or unenforceable under any applicable law, such provision shall not affect or invalidate any other provision of this Agreement, which shall continue to govern the relative rights and duties of the parties as though such void, voidable or unenforceable provision were not a part hereof.

Section 11.8. Governing Law.

This Agreement shall be construed in accordance with, and governed by, the laws of the State of Georgia, without regard to the conflicts of law rules of such state. Notwithstanding the foregoing, if both the Provider and Recipient of certain Data are residents of the same county (other than the United States), this Agreement shall be construed in accordance with, and governed by, the laws of such country with respect to such Data only.

Section 11.9. Counterparts.

This Agreement may be executed in one or more counterparts, each of which shall be deemed an original instrument, but all of which together shall constitute but one and the same agreement.

Section 11.10. Disputes.

All disputes arising from or in connection with this Agreement, whether based on contract, tort, statute or otherwise, including, but not limited to, disputes in connection with claims by third parties, shall be resolved only in accordance with the provisions of Section 15.10 of the Distribution Agreement.

Section 11.11. Force Majeure.

Neither party will be liable for any loss or damage due to causes beyond its control, including, but not limited to, fire, accident, labor difficulty, war, power or transmission failures, riot, Acts of God or changes in laws and regulations, provided that the affected party must (a) promptly notify the other party in writing and furnish all relevant information concerning the event of force majeure; (b) use reasonable efforts to avoid or remove the cause of

Section 11.12. Headings.

The Article and Section headings set forth in this Agreement are included for administrative, organizational and convenience purposes, and are not intended to affect the meaning of the provisions set forth in this Agreement or to be used in the interpretation of this Agreement.

IN WITNESS WHEREOF, the parties hereto have executed and delivered this Agreement as of the day and year first above written.

EQUIFAX INC.
Ву:
Title:
Date:
CERTEGY INC.
Ву:
Title:
Date:

TRANSTITON SUPPORT AGREEMENT

THIS AGREEMENT ("Agreement") for the performance of corporate services is executed and made effective as of _______, 2001, between Equifax Inc., a Georgia corporation ("Equifax"), and Certegy Inc., a Georgia corporation ("Certegy").

WHEREAS, Equifax, through the operation of its Payment Services Group, is engaged in the business of providing payment transaction processing solutions and services to financial institutions and merchants;

WHEREAS, the Board of Directors of Equifax has determined that it would be advisable and in the best interests of Equifax and its shareholders for Equifax to contribute the businesses, operations, assets and liabilities of its Payment Services Group (collectively, the "Business") to Certegy in exchange for Certegy common stock and thereafter to distribute all of the outstanding shares of Certegy's common stock on a pro rata basis to the holders of Equifax's common stock (the "Distribution") pursuant to a Distribution Agreement, dated as of the date hereof, between Equifax and Certegy (the "Distribution Agreement");

WHEREAS, the parties intend that the transactions described herein will be effective at the Effective Time (as defined in the Distribution Agreement); and

WHEREAS, the parties hereto deem it to be appropriate and in the best interests of the parties that they provide certain interim services to each other, including financial, accounting, tax, computer systems support and other services on the terms and conditions set forth herein.

NOW, THEREFORE, in consideration of the mutual promises contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

- 1. Definitions. Capitalized terms not defined herein shall have the meanings set forth in the Distribution Agreement.
 - 2. Description of Services.
- (a) Equifax shall, subject to the terms and provisions of this Agreement, provide Certegy with (i) services, with respect to the Business, as set forth on the Exhibits hereto, including, without limitation, computer system support services, and (ii) such other specific services as Certegy may from time to time reasonably request, subject to Equifax's sole discretion and its being in a position to supply such additional services at the time of such request (collectively, the "Equifax Services").
- (b) Certegy shall, subject to the terms and provisions of this Agreement, provide Equifax with (i) services as set forth on the Exhibits hereto, including, without limitation, computer system support services, and (ii) such other services as Equifax may from time to time reasonably request, subject to Certegy's sole discretion and its being in a position to supply such additional services at the time of the request (collectively, the "Certegy Services").

- (c) Each Exhibit hereto (i) may identify which specific member(s) of the Equifax Group or the Certegy Group that the parties intend will provide the services and which member(s) of the Equifax Group or the Certegy Group that the parties intend will receive the services; (ii) will provide a description of the services; and (iii) may also include applicable warranties, service levels, service credits and any other special terms and conditions with respect to the services described therein.
- (d) The parties have each exerted their best efforts to identify each material service to be provided by the members of one Group to the members of the other Group during a transition period after the Effective Time and to address such services in this Agreement and the Exhibits hereto. However, the parties acknowledge that there may be material services that the parties would have intended to be included which have inadvertently been omitted from this Agreement and the Exhibits. The parties agree to cooperate and negotiate with each other in good faith in order to come to an agreement regarding the continued provision of such material services that have inadvertently been omitted from this Agreement and the Exhibits, on terms that are acceptable to both parties. Moreover, the parties agree that the services to be provided may require adjustments during the term of this Agreement to reflect (i) the evolving business and operations of each Group and applicable law; (ii) that the relationship memorialized by this Agreement is dynamic in nature and will evolve as the operating and business environment of each Group changes and evolves; and (iii) that the scope of the services that will be provided during the term of this Agreement and the corresponding fees charged and payment terms extended by the parties may need to be modified to reflect the foregoing. The parties agree to cooperate and negotiate with each other in good faith in order to modify this Agreement as appropriate to give effect to the intent of the parties and the enduring and dynamic nature of each of the parties' respective businesses and the relationship between the parties.
- (e) Each of Equifax and Certegy, as the case may be, shall use commercially reasonable efforts to transition from using the services provided by the other under this Agreement during the first year of the term of this Agreement. Equifax and Certegy agree that they shall use commercially reasonable efforts to assist each other with the development of transition plans to assure a smooth and orderly transition.

3. Consideration for Services.

- (a) Certegy shall pay Equifax for all the Equifax Services as described on the applicable Exhibits, and Equifax shall pay Certegy for all the Certegy Services as described on the applicable Exhibits, at the rates specified on each such Exhibit. Unless otherwise stated in an applicable Exhibit, the costs and fees payable under such Exhibit shall be exclusive of any and all sales, use, ad valorem or similar taxes.
- (b) The costs and fees for services set forth on each Exhibit are based on certain assumptions acknowledged and agreed to by the parties as of the Effective Time regarding the underlying costs of providing such services. If events occur that cause a material change in the underlying costs of providing the services set forth on a particular Exhibit, Equifax and Certegy hereby agree to negotiate with each other in good faith in order to (i) provide for a proper refund to the recipient of such services by the provider, in cases where the recipient has already paid the provider and costs and fees have been overestimated, (ii) provide for a proper supplemental payment to the provider of such services by the recipient, in cases where the recipient has

already paid the provider and costs and fees have been underestimated and (iii) adjust the costs and fees that the recipient of such services will be required to pay the provider going forward for the continued provision of such services.

- 4. Terms of Payment. Within thirty (30) business days after the end of each month during the term of this Agreement, each member of the Equifax Group or the Certegy Group providing services hereunder will submit a written invoice to the member of the Equifax Group or the Certegy Group (as applicable) receiving such services for fees for the services provided during the immediately preceding month together with an accounting of the charges for the immediately preceding month's services. Within twenty (20) business days after the receipt of such invoices, each member of the Equifax Group or the Certegy Group will remit payment of the full amount of such invoices to the member of the Equifax Group or the Certegy Group that provided the applicable service(s) in the manner provided in Section 5 below. Interest shall accrue on a day to day basis at the lower of one and one-half percent (1 1/2%) per month or the highest rate allowable under applicable law on any amounts not received by the party providing the service hereunder within thirty (30) business days after receipt by the other of the invoice. The amount of any monthly service fee shall be prorated to correspond with the portion of a given month for which services were actually rendered.
- 5. Method of Payment. Unless otherwise stated in the applicable Exhibit or agreed to by the parties, all amounts payable by the parties for the services rendered by the other pursuant to this Agreement shall be remitted in United States dollars in the form of a wire transfer.
- 6. No Service Levels or Service Credits. All services will be provided in a commercially reasonable manner in accordance with any service levels specified on a particular Exhibit. Except as set forth in the immediately preceding sentence or pursuant to Section 20 hereof, unless set forth on the applicable Exhibit, there shall be no service levels or service credits with respect to any of the services to be performed hereunder by any member of the Equifax Group or the Certegy Group.
- 7. WARRANTIES. THIS IS A SERVICE AGREEMENT. EXCEPT AS EXPRESSLY STATED IN THIS AGREEMENT OR ON AN APPLICABLE EXHIBIT, (A) NEITHER PARTY, NOR ANY MEMBER OF EITHER PARTY'S GROUP, GUARANTEES OR WARRANTS THE SERVICES TO BE PROVIDED HEREUNDER, (B) THE SERVICES WILL BE PROVIDED ON AN "AS IS" AND "WITH ALL FAULTS" BASIS AND (C) THERE ARE NO EXPRESS OR IMPLIED WARRANTIES OR GUARANTIES, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY, TITLE, NONFRINGEMENT OR FITNESS FOR A PARTICULAR PURPOSE, AND ALL SUCH WARRANTIES ARE EXPRESSLY DISCLAIMED. NEITHER PARTY, NOR ANY MEMBER OF EITHER PARTY'S GROUP NOR ANY OF ITS INFORMATION SOURCES, GUARANTEES OR WARRANTS THE CORRECTNESS, COMPLETENESS, CURRENTNESS, MERCHANTABILITY OR FITNESS FOR A

PARTICULAR PURPOSE OF ANY DATA PROVIDED TO ANY MEMBER OF THE OTHER PARTY'S GROUP.

8. Liability; Indemnification.

- (a) In no event shall either Equifax or Certegy, or any member of their respective Groups, have any liability, whether based on contract, tort (including, without limitation, negligence or strict liability), warranty or any other legal or equitable grounds, for any punitive, consequential, indirect, exemplary, special or incidental loss or damage suffered by the other arising from or related to this Agreement, including without limitation, loss of data, profits, interest or revenue, or interruption of business, even if such party has been informed of or might otherwise have anticipated or foreseen the possibility of such losses or damages. Notwithstanding the foregoing, any damages awarded or obtained (whether by settlement, compromise or judgment) as a result of Third Party Claims shall be considered direct damages for purposes of this Agreement.
- (b) The limitations set forth in Section 8(a) above shall not apply to liabilities which may arise as the result of the willful misconduct of a party.
- (c) Certegy shall indemnify, defend and hold harmless Equifax and its affiliates and their respective directors, officers, employees and agents (the "Equifax Indemnitees") from and against any and all damage, loss, liability and expense (including, without limitation, reasonable expenses of investigation and reasonable attorneys' fees and expenses) in connection with any and all actions or threatened actions) ("Indemnifiable Losses") incurred or suffered by any of the Equifax Indemnitees arising from, related to or associated with (i) Equifax's furnishing or failure to furnish the services provided for in this Agreement, other than liabilities arising out of the willful misconduct of the Equifax Indemnitees and (ii) the willful misconduct of Certegy in furnishing or failing to furnish the services to be provided by Certegy in this Agreement.
- (d) Equifax shall indemnify, defend and hold harmless Certegy and its affiliates and their respective directors, officers, employees and agents (the "Certegy Indemnitees") from and against any and all Indemnifiable Losses incurred or suffered by any of the Certegy Indemnitees arising from, related to or associated with (i) Certegy's furnishing or failure to furnish the services provided for in this Agreement, other than liabilities arising out of the willful misconduct of the Certegy Indemnitees, and (ii) the willful misconduct of Equifax in furnishing or failing to furnish the services to be provided by Equifax to Certegy in this Agreement.
- (e) No insurer or any other third party shall be, by virtue of the foregoing indemnification provisions, (i) entitled to a benefit it would not be entitled to receive in the absence of such provisions, (ii) relieved of the responsibility to pay any claims to which it is obligated, or (iii) entitled to any subrogation rights with respect to any obligation hereunder.
- (f) The procedures for indemnity claims pursuant to this Section 8 shall be those set forth in Article V of the Distribution Agreement.

- (g) Notwithstanding anything contained herein to the contrary, in no event shall either Equifax or Certegy, or any member of either of their respective Groups, in the aggregate, (each a "Liable Party"), have any liability including, without limitation, the obligation to indemnify the recipient of a particular service, to any member of the other Group during any Contract Year (as defined below), for Indemnifiable Losses arising under a particular Exhibit in an amount in excess of the fees received by such Liable Party under such Exhibit from the other Group during such Contract Year. "Contract Year" means a twelve (12) month period beginning at the Effective Time or the first anniversary of the Effective Time (or the first date upon which the provider of services under an Exhibit is obligated to begin providing such services (if later than the Effective Time), and the first anniversary of such date); provided, however, that the limitation on liability set forth in this Section 8(g) shall not apply to losses caused by willful misconduct.
- (h) Notwithstanding anything contained herein to the contrary, in no event shall either Equifax or Certegy, or any member of either of their respective Groups, in the aggregate, have any liability to any member of the other Group during any Contract Year for Indemnifiable Losses that arise from a breach of the main body of this Agreement (in contrast to an Exhibit) in an amount in excess of the aggregate fees paid by both parties for all services rendered hereunder during such Contract Year; provided, however, that the limitation on liability set forth in this Section 8(h) shall not apply to losses caused by willful misconduct.

9. Term; Termination; Survival.

- (a) The term of this Agreement will begin as of the Effective Time and will expire two (2) years from the Effective Time, unless sooner terminated as provided below.
- (b) Notwithstanding Section 9(a) above, either Equifax or Certegy, as the recipient of a particular service, may, at its option, upon no less than sixty (60) days prior written notice to the other (or such other period as the parties may mutually agree in writing), direct the other to no longer provide such service. Notwithstanding anything to the contrary contained in an Exhibit, the sixty (60) days minimum notice requirement contained in this Section 9(b) may not be shortened, unless an Exhibit states explicitly that the minimum notice requirement contained in this Section 9(b) shall not apply. In the event of any termination with respect to one or more, but less than all, services to be provided hereunder, this Agreement will continue in full force and effect with respect to any services not terminated.
- (c) Notwithstanding Section 9(a) above, either Equifax or Certegy, as the provider or recipient of a particular service, may terminate a particular service being provided under an applicable Exhibit if the other party materially breaches any of the terms hereof or of such applicable Exhibit if the breach is not cured within thirty (30) days after written notice of breach is delivered to the breaching party; provided, however, that if the breach is not capable of being cured within such thirty (30) day period and the breaching party is proceeding to cure the breach with reasonable diligence, the cure period shall be extended to sixty (60) days.
- (d) Notwithstanding Section 9(a) above, this Agreement may be terminated in its entirety in accordance with any of the following:

- (i) Upon written agreement of the parties;
- (ii) By either Certegy or Equifax for material breach by the other of any of the terms hereof if the breach is not cured within thirty (30) days after written notice of breach is delivered to the breaching party; provided, however, that if the breach is not capable of being cured within such thirty (30) day period and the breaching party is proceeding to cure the breach with reasonable diligence, the cure period shall be extended to sixty (60) days;
- (iii) By either Certegy or Equifax, upon written notice to the other if the other becomes insolvent or makes an assignment of substantially all of its assets for the benefit of creditors, or is placed in receivership, reorganization, liquidation or bankruptcy;
- (iv) By Equifax, upon written notice to Certegy, if, for any reason, the ownership or control of Certegy or any of Certegy's operations, becomes vested in, or is made subject to the control or direction of, any direct competitor of Equifax or one of its subsidiaries, but such termination shall be applicable only with respect to services provided by Equifax to the portion of Certegy's businesses that has been affected by the change in control; or
- (v) By Certegy, upon written notice to Equifax, if for any reason, the ownership or control of Equifax or any of Equifax's operations becomes vested in, or is made subject to the control or direction of, any direct competitor of Certegy or one of its subsidiaries, but such termination shall be applicable only with respect to services provided by Certegy to the portion of Equifax's businesses that has been affected by the change in control.
- (e) Upon any termination or expiration pursuant to this Section 9, Equifax and Certegy shall be compensated for all services performed to the date of termination or expiration in accordance with the provisions of this Agreement, and Equifax and Certegy, as the case may be, will consider hiring (but shall not be obligated to hire) certain employees of the other identified by the other prior to the termination or expiration to the extent that Equifax or Certegy, as the case may be, does not contract with third parties to provide the services rendered by Equifax or Certegy pursuant to this Agreement.
- (f) Upon termination or expiration of this Agreement (or an Exhibit, as the case may be), all rights and obligations of the parties under this Agreement (or such Exhibit, as the case may be) will immediately cease and terminate (except for the rights and obligations pursuant to Sections 6 through 9, 14 and 16 through 25, and the definitions required thereby, which will survive such termination or expiration), and neither party will have any further obligation to the other party with respect to this Agreement (or such Exhibit, as the case may be), except (i) for fees and reimbursable expenses payable to the other party accrued but unpaid at the date of termination or expiration, and (ii) as set forth in the provisions of this Agreement which are specifically designated herein as surviving such termination or expiration.

- 10. Access. During the term of this Agreement, each member of the Equifax members of the other Group access to its premises if reasonably necessary to receive any of the services provided hereunder. While on the premises of the other Group, the employees and agents of the visiting party shall abide by the rules and regulations of the hosting party. The visiting party shall indemnify, defend and hold the hosting party harmless from and against any and all damages, losses, costs and expenses suffered or incurred by reason of damage to person or property caused by the gross negligence or willful misconduct of its employees or agents while on the premises of the hosting party.
- 11. Amendment and Waiver. This Agreement may not be altered or amended, nor may any rights hereunder be waived, except by an instrument in writing executed by the party or parties to be charged with such amendment or waiver. No waiver of any terms, provision or condition of or failure to exercise or delay in exercising any rights or remedies under this Agreement, in any one or more instances, shall be deemed to be, or construed as, a further or continuing waiver of any such term, provision, condition, right or remedy or as a waiver of any other term, provision or condition of this Agreement.
- 12. Force Majeure. Any delays in or failure of performance by Equifax or Certegy shall not constitute a default hereunder if and to the extent such delay or failure of performance is caused by occurrences beyond the reasonable control of Equifax or Certegy, as the case may be, including, but not limited to: acts of God or the public enemy; compliance with any order or request of any governmental authority; acts of war; riots or strikes or other concerted acts of personnel; network failures or failures in communications; the wrongful termination of any Third Party Agreement (as defined in Section 20 below) by such third party; or any other causes beyond the reasonable control of Equifax or Certegy, whether or not of the same class or kind as those specifically named above; provided that the affected party must (a) promptly notify the other party in writing and furnish all relevant information concerning the event of force majeure; (b) use reasonable efforts to avoid or remove the cause of its non-performance; and (c) proceed to perform its obligations with dispatch when such cause is removed.
- 13. Assignment; Parties in Interest. Neither of the parties hereto may assign its rights or delegate any of its duties under this Agreement without the prior written consent of the other party. This Agreement shall be binding upon, and shall inure to the benefit of, the parties hereto and their respective successors and permitted assigns. Nothing contained in this Agreement, express or implied, is intended to confer any benefits, rights or remedies upon any person or entity other than members of the Equifax Group and the Certegy Group and the Equifax Indemnitees and Certegy Indemnitees under Section 8 hereof.
 - 14. Confidentiality; Ownership of Information.
- (a) Each party shall, and shall cause each member of its Group to hold, and cause its directors, officers, employees, agents, consultants and advisors to hold, in strict confidence, unless compelled to disclose by judicial or administrative process or, in the opinion of its counsel, by other requirements of law, all information of or concerning the other party or its Group obtained or created pursuant to this Agreement (except to the extent that the Distribution

Agreement, this Agreement or any other Ancillary Agreement permits or requires the use or disclosure of such information or to the extent such information can be shown to have been (i) in the public domain through no fault of the persons or entities subject to the restrictions of this Section 14(a) (each a "receiving party"), (ii) later lawfully acquired after the Effective Time on a nonconfidential basis from a third party or (iii) independently generated without any reference to any proprietary or confidential information of the other party or its Group), and no receiving party shall (x) use such information, except for the benefit of the other party's Group in connection with the performance of this Agreement, the Distribution Agreement or the Ancillary Agreements or (y) disclose such information to any other person or entity, except its employees, directors, officers, agents, auditors, attorneys, financial advisors, bankers and other consultants and advisors who need to know such information and who shall be advised of the obligations contained in this Section 14(a) and be bound by them. Each receiving party shall be deemed to have satisfied its obligation to hold confidential any information concerning or owned by the other party or its Group if it exercises the same care as it takes to preserve confidentiality for its own similar information. The covenants in this Section 14(a) shall survive any termination of this Agreement or of any obligation to provide services hereunder pursuant to Section 9 and shall continue indefinitely; provided, however, that the covenants in this Section 14(a) shall terminate with respect to any information not constituting a trade secret under applicable law on the third anniversary of the first date on which all obligations to provide services hereunder have terminated (but any such termination shall not terminate or otherwise limit any other covenant or restriction regarding the disclosure or use of such information under the Distribution Agreement or any other Ancillary Agreement or other agreement, instrument or legal obligation).

- (b) Any information owned by one party or any of its subsidiaries that is provided to a requesting party pursuant to this Agreement shall be deemed to remain the property of the providing party. Unless specifically set forth herein, nothing contained in this Agreement shall be construed as granting or conferring rights of license or otherwise in any such information.
- 15. Conflicts Between this Agreement and an Exhibit. As long as any Exhibit remains in effect, the terms of this Agreement shall govern such Exhibit. Except for the minimum notice requirement contained in Section 9(b), if any provision of an applicable Exhibit conflicts with a provision of this Agreement, the provision of such Exhibit will control; provided, however, that in no event shall the term for the provision of any service under this Agreement or an Exhibit exceed two (2) years from the Effective Time.
- 16. Applicability to Group Members. Equifax and Certegy shall each cause the members of their respective Groups to (a) comply with this Agreement and the Exhibits hereto and (b) perform the services described on the Exhibits hereto. From time to time after the date hereof, Equifax and Certegy may change which member(s) of their respective Groups provide or receive services hereunder, provided that such changes do not materially change the nature of the services being provided.
- 17. Notices. All notices and communications under this Agreement shall be in writing and shall be deemed to have been given (a) when received, if such notice or communication is delivered by facsimile, hand delivery or overnight courier, and (b) three (3)

8

business days after mailing if such notice or communication is sent by United States registered or certified mail, return receipt requested, first class postage prepaid. All notices and communications, to be effective, must be properly addressed to the party to whom the same is directed at its address as follows:

If to Equifax, to: Equifax Inc.

1550 Peachtree Street, N.W. Atlanta, Georgia 30309

Attention: Phillip J. Mazzilli Chief Financial Officer

Fax: (404) 885-8682

with a copy to: Equifax Inc.

1550 Peachtree Street Atlanta, Georgia 30309 Attention: Kent E. Mast General Counsel

Fax: (404) 885-8988

If to Certegy, to: Certegy Inc.

555 North Pointe Centre East Alpharetta, Georgia 30022 Attention: Bruce S. Richards

Corporate Vice President, General Counsel and Secretary

Fax:

with a copy to: Certegy Inc.

P.O. Box 349

Alpharetta, Georgia 30009 Attention: Michael T. Vollkommer

Corporate Vice President and

Chief Financial Officer

Fax:

Either party may, by written notice delivered to the other party in accordance with this Section 17, change the address to which delivery of any notice shall thereafter be made.

18. Entire Agreement. Except as set forth in the Intercompany Data Purchase Agreement (the "Data Purchase Agreement") and the Intellectual Property Agreement (the "Intellectual Property Agreement"), each executed as of the date hereof by the parties hereto, this Agreement, including all Exhibits hereto, constitutes the entire understanding of the parties hereto with respect to the subject matter hereof, superseding all negotiations, prior discussions

and prior agreements and understandings relating to such subject matter. All Exhibits attached hereto are by this reference made a part of this Agreement and are incorporated herein.

19. Severability. The provisions of this Agreement are severable and should any provision hereof be void, voidable or unenforceable under any applicable law, such provision shall not affect or invalidate any other provision of this Agreement, which shall continue to govern the relative rights and duties of the parties as though such void, voidable or unenforceable provision were not a part hereof.

20. Third Party Agreements.

- (a) Equifax and Certegy recognize that certain support services described on the Exhibits hereto and certain related software and hardware licenses ("Licenses") are provided by third parties under specific third party agreements ("Third Party Agreements"). Equifax and Certegy further recognize that the Third Party Agreements may have been entered into by either Equifax or Certegy and that the other receives support services and Licenses as a result of the Third Party Agreements. Equifax and Certegy shall (i) use their respective commercially reasonable efforts to cause the third party providers to continue to provide the support and Licenses to the other under the terms of the Third Party Agreements as in effect as of the Effective Time and (ii) if successful in causing third party providers to provide support and Licenses to the other party, pass through to the other party any service levels to which Equifax or Certegy is entitled under such Third Party Agreements. In the event that a Third Party Agreement terminates prior to the expiration of the term for the related services described on the applicable Exhibit hereto, the parties shall use their reasonable commercial efforts to secure an alternative method of making the services or Licenses available to the recipient of such services or Licenses, which alternative has terms and results in economic benefits and burdens to the parties which are substantially similar to those that exist as of the date hereof.
- (b) In the event that either Equifax or Certegy allows a member of the other party's Group to use software or hardware that is licensed from a third party under a Third Party Agreement, the other party shall abide by the applicable terms and conditions of such Third Party Agreement.
- 21. Further Assurances and Consents. In addition to the actions specifically provided for elsewhere in this Agreement, each of the parties hereto will use its reasonable efforts to (a) execute and deliver such further instruments and documents and take such other actions as any other party may reasonably request in order to effectuate the purposes of this Agreement and to carry out the terms hereof and (b) take, or cause to be taken, all actions, and do, or cause to be done, all things, reasonably necessary, proper or advisable under applicable laws, regulations and agreements or otherwise to consummate and make effective the transactions contemplated by this Agreement, including, without limitation, using its reasonable efforts to obtain any consents and approvals, make any filings and applications necessary or desirable in order to consummate the transactions contemplated by this Agreement; provided that no party hereto shall be obligated to pay any consideration therefor (except for filing fees and other similar charges) to any third party from whom such consents, approvals and amendments are requested or to take any action or omit to take any action if the taking of or the omission to

take such action would be unreasonably burdensome to the party or its Group or the business thereof.

- 22. Governing Law. This Agreement shall be construed in accordance with, and governed by, the laws of the State of Georgia, without regard to the conflicts of law rules of such state. Notwithstanding the foregoing, if all of the providers and recipients of services under a particular Exhibit are residents of the same country (other than the United States), this Agreement shall be construed in accordance with, and governed by, the laws of such country with respect to such Exhibit only.
- 23. Disputes. Any disputes arising under this Agreement shall be resolved in accordance with Section 15.10 of the Distribution Agreement; provided, however, that if all of the providers and recipients of services under a particular Exhibit are residents of the same country (other than the United States), the following terms will apply in place of those described in Sections 15.10(c) and 15.10(d) of the Distribution Agreement, with respect to disputes arising under such Exhibit only: (a) the third arbitrator as described in Section 15.10(c) of the Distribution Agreement shall be a lawyer licensed to practice in such country, (b) all questions of law shall be governed by the laws of such country and (c) the arbitration described in Section 15.10(d) of the Distribution Agreement shall be conducted in such country (or such other place as agreed upon by the parties and the arbitrators).
- 24. Headings. The Section headings set forth in this Agreement are included for administrative, organizational and convenience purposes, and are not intended to affect the meaning of the provisions set forth in this Agreement or to be used in the interpretation of this Agreement.
- 25. Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original instrument, but all of which together shall constitute but one and the same agreement.

IN WITNESS WHEREOF, the parties hereto have executed and delivered this Agreement as of the day and year first above written.

EOUIFAX INC.

By:
Name:
Title:
CERTEGY INC.
By:
Name:
Title:
By:
Title:

	INTEL	LECTUAL	PR0PERTY	AGREEMEN1
--	-------	---------	----------	-----------

BETWEEN

EQUIFAX INC.

AND

CERTEGY INC.

______, 2001

TABLE OF CONTENTS

ARTICLE I DEFINITION Section 1.1.	TIONS Definitions	4
ARTICLE II CONVE Section 2.1. Section 2.2. Section 2.3. Section 2.4.	YANCE OF CERTAIN ASSETS; ASSUMPTION OF Transferred Equifax Assets Transferred Certegy Assets Assumption of Liabilities Completion of Transactions	9 9 10 10
ARTICLE III THIR Section 3.1. Section 3.2. Section 3.3.	D PARTY AGREEMENTS Third Party Agreements Required Consents Discharge of Liabilities	12 12 13
ARTICLE IV LICENT Section 4.1. Section 4.2. Section 4.3. Section 4.4. Section 4.5. Section 4.6. Section 4.7.	SED MATERIALS Grant of Licenses by Equifax Ownership of Enhancements by Certegy License to Marks Grant of License by Certegy Ownership of Enhancements by Equifax Data Mutual Obligations	14 14 17 18 18 21 21 22
ARTICLE V THE CL Section 5.1. Section 5.2.	OSINGEquifax DeliverablesCertegy Deliverables	22 23 23
ARTICLE VI REPRE	SENTATIONS AND WARRANTIES	23
ARTICLE VII INDE Section 7.1. Section 7.2. Section 7.3.	MNIFICATIONCertegy Indemnification of the Equifax Group. Equifax Indemnification of the Certegy Group. Insurance and Third Party Obligations	24 24 24 24
ARTICLE VIII IND Section 8.1. Section 8.2.	EMNIFICATION PROCEDURES	25 25 25
ARTICLE IX CONFI	DENTIALITYExclusions	27 27

Section 9.2. Section 9.3. Section 9.4. Section 9.5. Section 9.6.	Confidentiality Employee Confidentiality Agreements Rights and Remedies Competitive Activities No Implied Rights	27 28 28 29 29
ARTICLE X CON	TINUED ASSISTANCE	30
Section 10.1		30
Section 10.2	. Records and Documents	30
Section 10.3		31
ARTICLE XT MT	SCELLANEOUS	31
Section 11.1		31
Section 11.2	•	31
Section 11.3	. Amendment and Waiver	32
Section 11.4	. Entire Agreement	32
Section 11.5		33
Section 11.6		33
Section 11.7	. Severability	33
Section 11.8		33
Section 11.9	. Counterparts	34
Section 11.1	0. Disputes	34
Section 11.1	1. Force Majeure	34
Section 11.1	2. Documentation	34
Section 11.1	3. Headings	34
EXHIBIT A -	CERTEGY GROUP	
EXHIBIT B -	TRANSFERRED EQUIFAX ASSETS	
EXHIBIT C -	EQUIFAX THIRD PARTY AGREEMENTS - TRANSFERS	
EXHIBIT D -	EQUIFAX THIRD PARTY AGREEMENTS - RIGHTS GRANTED	
EXHIBIT E -	TRANSFERRED CERTEGY ASSETS	
EXHIBIT F -	CERTEGY THIRD PARTY AGREEMENTS - TRANSFERS	
EXHIBIT G -	CERTEGY THIRD PARTY AGREEMENTS - RIGHTS GRANTED	
EXHIBIT H -	SPECIFIED EQUIFAX LIABILITIES	
EXHIBIT I -	SPECIFIED CERTEGY LIABILITIES	
EXHIBIT J -	LICENSED EQUIFAX MATERIALS	
EXHIBIT K -	LICENSED CERTEGY MATERIALS	
EXHIBIT L -	UTILITY SOFTWARE PROGRAMS	
EXHIBIT M -	JOINTLY OWNED IP ASSETS	

INTELLECTUAL PROPERTY AGREEMENT

THIS INTELLECTUAL PROPERTY AGREEMENT ("Agreement"), dated as of _______, 2001, is entered into by Equifax Inc., a Georgia corporation ("Equifax"), and Certegy Inc., a Georgia corporation ("Certegy").

BACKGROUND

- A. Certegy is a wholly owned subsidiary of Equifax formed among other reasons for the purpose of taking title to the intellectual property assets and assuming the associated liabilities related to the business operations of the Certegy Group (as defined below).
- B. The Board of Directors of Equifax has determined that it is in the best interests of Equifax and its shareholders to transfer, assign and/or license to Certegy and Designated Certegy Members (defined below), as part of the contribution to the capital of Certegy, certain intellectual property assets used in the business operations of the Certegy Group as described herein and currently utilized to operate the Certegy Business (as defined below), and to receive in exchange therefor the consideration described in the Distribution Agreement (as defined below).
- C. The Board of Directors of Certegy has determined that it is in the best interests of Certegy and its shareholders to transfer, assign and/or license to Equifax and Designated Equifax Members (defined below) certain intellectual property assets.
- D. The parties intend that the Distribution (as defined in the Distribution Agreement) not be taxable to Equifax or its shareholders pursuant to Section 355 of the Code (as defined below).
- E. Equifax and its Affiliates (defined below) own certain intellectual property that is used in, or may be useful in, the conduct of the business operations of the Equifax Group (defined below) and/or the Certegy Group. Equifax and Certegy have determined that subject to the terms herein: (1) ownership of certain of such intellectual property shall be transferred to the entity specified in this Agreement on or before the Distribution Date (defined below); (2) certain intellectual property owned by Equifax and/or its Affiliates shall be licensed to the entity(ies) specified in this Agreement on or before the Distribution Date; and (3) the respective rights and obligations of Equifax and/or its Affiliates under certain Third Party Agreements (defined below) shall be acquired, assumed or otherwise transferred to the entity(ies) specified in this Agreement, subject to the consent of the applicable Third Party Provider (defined below).
- F. The parties have determined that it is necessary and desirable to describe the principal transactions required to effect the allocation of their respective intellectual property rights in conjunction with the Distribution and to set forth other agreements that will govern

certain other matters regarding the parties' respective intellectual property rights following the Distribution.

NOW, THEREFORE, in consideration of the foregoing premises, the mutual agreements and covenants contained in this Agreement, and other valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

ARTICLE I

DEFINITIONS

Section 1.1. Definitions

As used herein, the following terms have the following meanings:

- (a) "Action" means any claim, suit, arbitration, inquiry, proceeding or investigation by or before any court, governmental or other regulatory or administrative agency or commission or any other tribunal.
- (b) "Affiliate" means, with respect to Equifax, any Person, which, whether directly or indirectly, is Controlled by or is under common Control with Equifax prior to the Distribution Date.
- (c) "Certegy Business" means the businesses conducted by the members of the Certegy Group as of the Distribution Date.
- (d) "Certegy Continued Use Materials" means any and all IP Assets (other than Transferred Assets) owned and/or held by a member of the Equifax Group that satisfy each of the following criteria: (i) such IP Assets were used in the Certegy Business during the twelve (12) calendar months prior to the Distribution Date and for which a continuing business requirement exists on the Distribution Date, and (ii) such IP Assets or the services, information or deliverables produced with such IP Assets (A) are not made commercially available by the Equifax Group to third parties on the Distribution Date, and (B) are not made available to the Certegy Group after the Distribution Date pursuant to the Intercompany Data Purchase Agreement or the Transition Support Agreement.
- (e) "Certegy Enhancements" means software and/or associated documentation created by or for any member of the Certegy Group on or after the Closing Date, that provides processing capabilities, functionality or efficiencies, maintenance, bug fixes or updates not contained in the Transferred Equifax Assets as of the Closing Date and which is intended for use with and requires a portion of the Transferred Equifax Assets in order to function properly.
- (f) "Certegy Group" means the entities set forth on Exhibit A and any of their respective subsidiaries.

- (g) "Certegy Indemnitees" has the meaning given in Section 7.2.
- (h) "Certegy Liabilities" means all unsatisfied Liabilities, whether arising before, on or after the Distribution Date, based upon or arising out of the ownership, use or possession by the Certegy Group of the Transferred Equifax Assets, the Licensed Equifax Materials or the Equifax Marks.
- (i) "Certegy Third Party Use Rights" means the rights granted to or secured for Equifax or one or more Designated Equifax Members pursuant to Section 3.1(b)(ii).
- (j) "Closing Date" means the Effective Time, as defined in the Distribution Agreement. $\,$
 - (k) "Code" means the Internal Revenue Code of 1986, as amended.
- (1) "Company Information" means collectively the Proprietary Information and the Confidential Information of the disclosing party. Company Information also includes information that has been disclosed to Equifax or any of its Affiliates prior to the Distribution Date, or to any member of either Group after the Distribution Date, by a third party subject to an obligation to treat such information as confidential or secret.
- (m) "Confidential Information" means any and all confidential business information of the disclosing party that does not constitute Proprietary Information and that is the subject of efforts by the disclosing party that are reasonable under the circumstances to maintain its secrecy and confidentiality, including without limitation, the existence and nature of the relationship between the parties, employees of the disclosing party, and any and all additional information disclosed by the disclosing party to the receiving party as a result of the receiving party's access to and presence at the disclosing party's facilities.
- (n) "Control" means the ownership, directly or indirectly, of more than fifty percent (50%) of the voting shares of an entity, or other possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of an entity, or the power to veto major policy decisions of any such entity, whether through the ownership of voting securities by contract, or otherwise.
- (o) "Derivative Work" means a work based on one or more pre-existing works, including without limitation, a condensation, transformation, expansion or adaptation, that would constitute a copyright infringement if prepared without authorization of the owner of the copyright of such pre-existing work.
- (p) "Designated Certegy Member" means a member of the Certegy Group, as designated by Certegy in its sole discretion.
- (q) "Designated Equifax Member" means a member of the Equifax Group, as designated by Equifax in its sole discretion.

- (r) "Disputes" has the meaning given in the Distribution Agreement.
- (s) "Distribution Agreement" means that certain Distribution Agreement entered into on or prior to the Distribution Date between Equifax and Certegy, as amended from time to time.
- (t) "Distribution Date" means the day as of which the Distribution shall be effective, as determined by the Board of Directors of Equifax, or such committee of such Board of Directors as shall be designated by the Board of Directors of Equifax.
- (u) "Divested Business" means the sale or other transfer of a member of either Group, or a portion of the business operations of any such member, to an unrelated third party after the Distribution Date.
- (v) "Equifax Business" means the businesses now or formerly conducted by Equifax and its present and former Affiliates, other than the Certegy Business.
- (w) "Equifax Continued Use Materials" means any and all IP Assets (other than Transferred Assets) owned and/or held by a member of the Certegy Group that satisfy each of the following criteria: (i) such IP Assets were used in the Equifax Business during the twelve (12) calendar months prior to the Distribution Date and for which a continuing business requirement exists on the Distribution Date, and (ii) such IP Assets or the services, information or deliverables produced with such IP Assets (A) are not made commercially available by the Certegy Group to third parties on the Distribution Date, and (B) are not made available to the Equifax Group after the Distribution Date pursuant to the Intercompany Data Purchase Agreement or the Transition Support Agreement.
- (x) "Equifax Enhancements" means software and/or associated documentation created by or for any member of the Equifax Group on or after the Closing Date, that provides processing capabilities, functionality or efficiencies, maintenance, bug fixes or updates not contained in the Transferred Certegy Assets on the Closing Date and which is intended for use with and requires a portion of the Transferred Certegy Assets in order to function properly.
- (y) "Equifax Group" means Equifax and its Affiliates existing on the Distribution Date and as modified from time to time thereafter, excluding all members of the Certegy Group.
 - (z) "Equifax Indemnitees" has the meaning given in Section 7.1.
- (aa) "Equifax Liabilities" means all unsatisfied Liabilities, whether arising before, on or after the Distribution Date, based upon or arising out of the ownership, use or possession by the Equifax Group of the Transferred Certegy Assets or the Licensed Certegy Materials.
 - (bb) "Equifax Marks" mean the Marks owned by Equifax or its Affiliates.
- (cc) "Equifax Third Party Use Rights" means the rights granted to or secured for Certegy or one or more Designated Certegy Members pursuant to Section 3.1(a)(ii).

- (dd) "Group" means the Certegy Group and/or the Equifax Group.
- (ee) "Indemnifiable Losses" has the meaning given in Section 7.1.
- (ff) "Indemnified Party" has the meaning given in Section 8.1.
- (gg) "Indemnifying Party" has the meaning given in Section 8.1.
- (hh) "Intercompany Data Purchase Agreement" means that certain Intercompany Data Purchase Agreement entered into on or prior to the Distribution Date between Equifax and Certegy, as amended from time to time.
- (ii) "IP Assets" means all intellectual property rights in and to any ideas, trade secrets, specifications, designs, masks, mask works, copyrights, patents, Marks and other proprietary rights, of every kind and description, wherever located, including without limitation, all electronic circuit designs, works of authorship, databases, compositions of matter, computer software (whether such computer software constitutes custom software, firmware or systems created by, or for the exclusive use of either party, or otherwise), algorithms, and works of authorship expressing such algorithms.
- (jj) "Jointly Owned IP Assets" shall mean those assets that are jointly owned by one or more members of the Equifax Group and one or more members of the Certegy Group, as set forth on Exhibit M.
- (kk) "Liabilities" means any and all claims, debts, liabilities and obligations, absolute or contingent, matured or not matured, liquidated or unliquidated, accrued or unaccrued, known or unknown, whenever arising, with respect to a specified object, matter, contract, commitment or undertaking, including without limitation, all claims, debts, liabilities and obligations arising under any law, rule, regulation, action, order or consent decree of any governmental entity or any award of any arbitrator of any kind, related thereto or arising under any contract, commitment or undertaking relating to such specified object, matter, contract, commitment or undertaking.
- (11) "Licensed Certegy Materials" means those IP Assets identified on Exhibit K and the Equifax Continued Use Materials.
- (mm) "Licensed Equifax Materials" means those IP Assets identified on Exhibit J and the Certegy Continued Use Materials.
- (nn) "Licensed Materials" means the Licensed Certegy Materials and/or Licensed Equifax Materials.
- (oo) "Marks" means trademarks, service marks, domain names, tradenames, and other slogans, designs and distinctive advertising, whether or not registered or filed with any governmental agency.

- (pp) "Person" means an individual, partnership, joint venture, association, corporation, limited liability company, trust or any other legal entity.
- (qq) "Proprietary Information" means all non-public information whether tangible or intangible related to the services or business of the disclosing party that (i) derives economic value, actual or potential, from not being generally known to or readily ascertainable by another Person who can obtain economic value from its disclosure or use; and (ii) is the subject of efforts by the disclosing party that are reasonable under the circumstances to maintain its secrecy, including without limitation, (A) marking any information reduced to tangible form clearly and conspicuously with a legend identifying its confidential or proprietary nature; (B) identifying any oral communication as confidential immediately before, during, or after such oral communication; or (C) otherwise treating such information as confidential or secret. Assuming the criteria in clauses (i) and (ii) above are met, Proprietary Information includes information, without regard to form, including, but not limited to, technical and nontechnical data, databases, formulas, patterns, designs, compilations, computer programs and software, devices, inventions, methods, techniques, drawings, processes, financial data, financial plans, product plans, lists of actual or potential customers and suppliers (which are not commonly known by or available to the public), research, development, and existing and future
- (rr) "Related Agreements" means all of the written agreements, instruments, understandings, assignments or other arrangements entered into in connection with the transactions contemplated hereby (other than this Agreement), including without limitation, the Distribution Agreement, Intercompany Data Purchase Agreement and Transition Support Agreement.
- (ss) "Representatives" means, individually and collectively, officers, directors, employees, agents, and/or independent contractors of each member of the Group.
- (tt) "Required Consents" means any consents or approvals required to be obtained (i) to allow the transfer of any assets to and the assumption of the obligations attendant therewith by a party and release of the transferring party from such obligations; (ii) to allow a party to assume financial, support, operational, management and/or administrative responsibility for the Third Party Rights utilized in the operation of the Equifax Business or Certegy Business, respectively; (iii) for the licensing, transfer and/or grant of the rights to the Equifax Group or Certegy Group, respectively, to use the Third Party Rights as contemplated by this Agreement; and/or (iv) for a party to have access to and use of the space, equipment, software and/or third party services provided under the Third Party Agreements entered into by the other party as contemplated by this Agreement.
- (uu) "Third Party Agreements" means agreements, contracts or arrangements between Equifax and/or its Affiliates, on the one hand, and a Third Party Provider, on the other.
 - (vv) "Third Party Claim" has the meaning given in Section 8.2.

- (ww) "Third Party Provider" means a Person other than a member of either Group that provides products, software, services, maintenance and/or support to Equifax or one or more of its Affiliates.
- (xx) "Third Party Rights" means rights granted to Equifax and/or its Affiliates pursuant to a Third Party Agreement, including (i) all service, support and maintenance rights related thereto or attendant therewith, and (ii) all contractual rights, commitments, undertakings and obligations (including service, data processing, support and maintenance rights and obligations) attendant therewith or directly related thereto.
- (yy) "Third Party Use Rights" means the respective Equifax Third Party Use Rights and Certegy Third Party Use Rights.
- (zz) "Transferred Assets" means the Transferred Equifax Assets and Transferred Certegy Assets.
- (aaa) "Transferred Equifax Assets" means the assets to be transferred to Certegy or one or more Designated Certegy Members as described in Section 2.1.
- (bbb) "Transferred Equifax Third Party Agreements" means the Third Party Agreements, the rights and obligations of which are to be transferred to Certegy or one or more Designated Certegy Members pursuant to Section 3.1(a)(i).
- (ccc) "Transferred Certegy Assets" means the assets to be transferred to Equifax or the one or more Designated Equifax Members as described in Section 2.2.
- (ddd) "Transferred Certegy Third Party Agreements" means the Third Party Agreements, the rights and obligations of which are to be transferred to Equifax or one or more Designated Equifax Members pursuant to Section 3.1(b)(i).
- (eee) "Transferred Third Party Agreements" means the respective Transferred Equifax Third Party Agreements and/or the Certegy Transferred Third Party Agreements.
- (fff) "Transition Support Agreement" means that certain Transition Support Agreement entered into on or prior to the Distribution Date between Equifax and Certegy, as amended from time to time.
- (ggg) "Utility Software Programs" means the software programs set forth on Exhibit L.

ARTICLE II

CONVEYANCE OF CERTAIN ASSETS; ASSUMPTION OF CERTAIN LIABILITIES

Section 2.1. Transferred Equifax Assets.

Effective as of the Closing Date, and subject to Sections 2.3 and 2.4 and Article III hereof, Equifax agrees, at its expense, to transfer, or cause to be transferred, to Certegy or to a Designated Certegy Member all right, title and interest held by Equifax and/or its Affiliates as of the Closing Date in and to each of the assets identified on Exhibit B hereto, subject to the retained rights described therein. Except as set forth on Exhibit B, no other assets (other than Transferred Equifax Third Party Agreements) are being transferred by Equifax (or a member of the Equifax Group) pursuant to this Agreement.

Section 2.2. Transferred Certegy Assets.

Effective as of the Closing Date, and subject to Sections 2.3 and 2.4 and Article III hereof, Certegy agrees to transfer, or cause to be transferred, to Equifax or to a Designated Equifax Member all right, title and interest held by the members of the Certegy Group as of the Closing Date in and to each of the assets identified on Exhibit E, subject to the retained rights described therein. Except as set forth on Exhibit E, no other assets (other than Transferred Certegy Third Party Agreements) are being transferred by Certegy (or a member of the Certegy Group) pursuant to this Agreement. The expenses payable to third parties that are not members of either Group to effect such transfers shall be the financial responsibility of Equifax.

Section 2.3. Assumption of Liabilities.

- (a) As of the Closing Date, Equifax shall, or shall cause the respective Designated Equifax Member to, assume all payment and performance obligations attendant with the Transferred Certegy Assets and the Equifax Liabilities, including, without limitation, the Liabilities identified on Exhibit H.
- (b) As of the Closing Date, Certegy shall, or shall cause the respective Designated Certegy Member to, assume all payment and performance obligations attendant with the Transferred Equifax Assets and the Certegy Liabilities, including, without limitation, the Liabilities identified on Exhibit I.

Section 2.4. Completion of Transactions.

(a) In the event that any conveyance of a Transferred Asset, Transferred Third Party Agreement, or the provision of a Third Party Right or Third Party Use Right, or assumption of any Liability, required by this Agreement is not effected on or before the Closing Date, the obligation to transfer such Transferred Asset or Transferred Third Party Agreement, provide such Third Party Right or Third Party Use Right, and assume such Liability shall continue past the Closing Date and shall be effected by the parties as soon thereafter as practicable; provided, however, that neither party shall be obligated under this paragraph to transfer any Transferred Third Party Agreement and or provide Third Party Use Rights that either (i) did not exist as of the Closing Date or (ii) are no longer required by the party who is the intended transferee of the respective Third Party Agreement or is entitled to receive the Third Party Use Rights for the continued operation of such party's business.

- (b) If any Transferred Asset or Transferred Third Party Agreement may not be transferred or acquired by reason of a requirement to obtain a Required Consent or any other approval of any third party and such Required Consent or other approval has not been obtained by the Closing Date, then such Transferred Asset or Transferred Third Party Agreement shall not be transferred or acquired until such Required Consent or other approval has been obtained. Equifax and Certegy shall, and as the case may be, shall cause the member of its respective Group which is the holder of such Transferred Asset or Transferred Third Party Agreement prior to transfer, to use all reasonable efforts to provide to the applicable member of the other Group all the rights and benefits under such Transferred Asset or Transferred Third Party Agreement and to cause such holder to enforce such Transferred Asset or Transferred Third Party Agreement for the benefit of such member of the other Group; provided, however, that the foregoing obligation shall not, in any way, require Equifax, Certegy or any member of a respective Group to breach any Transferred Third Party Agreement or incur or suffer any liability with respect to any Transferred Third Party Agreement. Moreover, if any transfer of a Transferred Asset or Transferred Third Party Agreement or provision of a Third Party Right or Third Party Use Right, is not completed by the Closing Date in accordance with this Agreement for any reason, each of Equifax and Certegy shall, and shall cause the members of its Group to, cooperate in achieving a reasonable alternative arrangement for the affected members of the Groups to obtain the economic and operational equivalent of the intended transfer of such Transferred Asset or Third Party Agreement and/or provision of such Third Party Right or Third Party Use Right, and assumption of the attendant Liabilities, with minimum interference to such members' business operations until such transfer of such Transferred Asset or Third Party Agreement, and/or provision of such Third Party Right or Third Party Use Right, is completed. The costs payable to third parties that are not members of either Group to achieve any such reasonable alternative arrangement shall be the financial responsibility of Equifax.
- (c) From time to time on and after the Closing Date, each party shall promptly transfer, and cause the appropriate members of its Group promptly to transfer, to the other party, or the designated member of the other party's Group, any property and other benefits received by such party, or the members of its Group, that are intended to be or are a Transferred Asset or Transferred Third Party Agreement of the other party under this Agreement. Without limiting the foregoing, funds received by a member of either Group that belong to a member of the other Group (whether by payment of accounts receivable, credits, rebates or other amounts, however described) shall be delivered to the other Group by wire transfer not more than five (5) business days after receipt of such payment.

ARTICLE III

THIRD PARTY AGREEMENTS

Section 3.1. Third Party Agreements.

- (a) Effective as of the Closing Date, Equifax shall (i) transfer, or cause to be transferred, to Certegy or a Designated Certegy Member, the rights and obligations of Equifax and its Affiliates in and to the Third Party Agreements identified on Exhibit C (including all Third Party Rights related thereto) and (ii) grant rights to or secure rights (including rights as an "authorized user") for Certegy or a Designated Certegy Member under the Third Party Agreements identified on Exhibit D, subject to the payment obligations or other terms set forth on Exhibit D.
- (b) Effective as of the Closing Date, Certegy shall (i) transfer, or cause to be transferred, to Equifax or a Designated Equifax Member, the rights and obligations of the members of the Certegy Group in and to the Third Party Agreements identified on Exhibit F (including all Third Party Rights related thereto) and (ii) grants rights to or secure rights (including rights as an "authorized user") for Equifax or a Designated Equifax Member under the Third Party Agreements identified on Exhibit G, subject to the payment obligations or other terms set forth on Exhibit G.
- (c) Notwithstanding anything herein to the contrary, for a period of twelve (12) months following the Closing Date, upon Certegy's written request, Equifax shall, at its expense, acquire, purchase or secure for Certegy, rights substantially similar to all Third Party Rights and Third Party Agreements evidencing such Third Party Rights, held by Equifax and/or any of its Affiliates on or prior to the Closing Date, to any generally commercially available software (including, but not limited to, applications, systems and/or operating systems software) that was used in the Certegy Business during the twelve (12) calendar months prior to the Closing Date and for which a continuing business requirement exists on the Closing Date; provided, however, that Equifax shall not be required to acquire, purchase or secure for Certegy any such Third Party Rights or Third Party Agreements (i) to the extent that it would entitle Certegy to receive any greater rights relating to usage (including, without limitation, usage rights relating to the number of users, the number of concurrent users or the number of MIPS) of the applicable software than Equifax possessed on the Closing Date, (ii) that Certegy previously had requested and received, at Equifax's expense, rights substitutable for such Third Party Rights or such Third Party Agreements, or (iii) that Certegy had determined previously that such rights were not necessary, unless (x) Certegy's determination was based on a belief that it had no legal obligation to acquire usage rights with respect to the software for its operations after the Closing Date, and (y) both Equifax and Certegy determine in good faith that the software is, in fact, legally required by Certegy. Notwithstanding the foregoing, and unless otherwise provided in Exhibit C, (i) Equifax shall not have any obligation to pay any maintenance fees with respect to such software licenses unless such maintenance fee obligations are bundled as part of the license fee for the first year of the license and (ii) Equifax's obligations to pay ongoing license fees with respect to any such software shall expire after the ____ anniversary of the Closing Date. Accordingly, if the use of

any such software obtained from or through Equifax prior to the Closing Date (but otherwise in accordance with the terms and conditions of such Third Party Agreement) by any member(s) of the Certegy Group on or after the Closing Date infringes or otherwise violates the scope of use, or permitted users, under the applicable Third Party Agreement(s), then (i) the Certegy Group, or any affected member thereof, may request Equifax to provide, at Equifax's expense in accordance with the first sentence of this paragraph, an ongoing license to use such software (including, but not limited to, any Third Party Rights related thereto) in the name of the Certegy Group on substantially the same terms as set forth in the applicable Third Party Agreement(s); and (ii) Equifax shall promptly reimburse the Certegy Group, or any affected member thereof, for all charges, penalties or other costs actually incurred by any such member(s) related to its or their use of the software after the Distribution Date through the date of discovery by the Third Party Provider of such infringement or violation.

(d) Without limiting each party's specific obligations pursuant hereto (or in any separate agreement) with respect to Third Party Use Rights, each of Certegy and Equifax agrees to, in connection with its use of, exploitation of and performance pursuant to any Third Party Use Rights, including, without limitation, such party's rights to use, copy, exploit, distribute, display, copy and sublicense any software secured for or granted to such party pursuant to such Third Party Use Rights, comply with the terms, scope, restrictions and provisions (including, without limitation, usage limitations) of any Third Party Agreements that govern such Third Party Use Rights. A failure to comply with this paragraph shall constitute a breach of this Agreement.

Section 3.2. Required Consents.

- (a) Equifax with respect to Transferred Equifax Third Party Agreements and Equifax Third Party Use Rights, and Certegy with respect to Transferred Certegy Third Party Agreements and Certegy Third Party Use Rights, shall, or shall cause the appropriate member of its respective Group to, use its reasonable commercial efforts to obtain the grant to the applicable member of the other Group, the Required Consents from the Third Party Providers under such respective Third Party Agreements as necessary to effect the provisions of this Agreement. Each party will provide the other party with advice on its experience and agreements with the Third Party Providers with regard to obtaining any Required Consent under such Third Party Agreements. Equifax and Certegy will each have management and administrative responsibilities for obtaining all Required Consents required as of the Closing Date to which a member of its respective Group is a party. Equifax shall have the right of prior approval of the terms upon which all Required Consents are obtained.
- (b) Equifax shall bear the costs payable to third parties that are not members of either Group, if any, of obtaining all Required Consents, including without limitation, all charges and fees related to obtaining the Required Consents for the Transferred Third Party Agreements and Third Party Use Rights.
- (c) Equifax and Certegy shall use reasonable commercial efforts to obtain all Required Consents with regard to Transferred Third Party Agreements and Third Party Use Rights within

one hundred eighty (180) days after the Closing Date, unless otherwise agreed by the parties in writing. Until all Required Consents are obtained, Equifax and Certegy shall each periodically publish a list setting forth the status of each Required Consent for which a member of its respective Group is the contracting party immediately prior to the Closing Date. Equifax and Certegy shall timely cooperate with each other in order to facilitate the proper and timely publication of such periodic Required Consents list. If any Required Consent is not obtained with respect to any of the Third Party Agreements or Third Party Use Rights, the parties shall cooperate with each other in achieving a reasonable alternative arrangement for the affected Group to continue to process its work with minimum interference to its business operations until such Required Consents are obtained, including without limitation, implementing the provisions of Section 2.4(b). The cost payable to third parties that are not members of either Group of achieving such reasonable alternative arrangements with respect to Third Party Rights that are a part of the Transferred Assets or Transferred Third Party Agreements shall be borne by Equifax.

- (d) The financial obligations of Equifax under Sections 3.2(b) and (c) for Required Consents and alternative arrangements, shall terminate with respect to all such Required Consents and alternative arrangements not identified by the parties to each other in a writing within twelve (12) months after the Closing Date, and for all Required Consents and alternative arrangements identified thereafter, all such financial obligations shall be borne by the party needing the Required Consent or alternative arrangement to operate under or take assignment of the Third Party Agreement or to obtain such Third Party Right for which such Required Consent or alternative arrangement is required.
- (e) For all periods after the Closing Date, except as set forth in Sections 3.2(b) and 3.2(c) for Required Consents and alternative arrangements, Equifax and Certegy shall each bear financial responsibility and pay the Third Party Providers, under all Transferred Third Party Agreements transferred to its respective Group pursuant to Sections 3.1(a)(i) and 3.1(b)(i) above.

Section 3.3. Discharge of Liabilities.

- (a) Certegy agrees that on and after the Closing Date it will timely pay, perform and discharge, or cause to be timely paid, performed and discharged, all of the Certegy Liabilities.
- (b) Equifax agrees that on and after the Closing Date it will timely pay, perform and discharge, or cause to be timely paid, performed and discharged, all of the Equifax Liabilities.

ARTICLE IV

LICENSED MATERIALS

Section 4.1. Grant of Licenses by Equifax.

(a) Equifax hereby grants, and will cause the other members of the Equifax Group to grant, to Certegy a fully paid, non-exclusive, perpetual, worldwide, non-transferable license to

use, modify, copy, improve, create Derivative Works and Certegy Enhancements from, and sublicense the Licensed Equifax Materials (excluding the Utility Software Programs) solely for use in the Certegy Business and as that business may evolve and change in the future, subject to the following:

- (i) Certegy shall not sublicense, or otherwise disclose or distribute, or permit any Person to use, the Licensed Equifax Materials (excluding the Utility Software Programs), except in accordance with Section 4.1(b);
- (ii) Certegy shall hold the Licensed Equifax Materials (excluding the Utility Software Programs) in strict confidence; will not remove or destroy any proprietary markings of the Equifax Group on or contained in the Licensed Equifax Materials (excluding the Utility Software Programs); and will include the copyright and patent notices of the licensor as specified from time to time by the licensor for the Licensed Equifax Materials (excluding the Utility Software Programs) on and in all copies of the Licensed Equifax Materials (excluding the Utility Software Programs);
- (iii) Certegy shall not export or re-export the Licensed Equifax Materials (excluding the Utility Software Programs) without the appropriate United States or foreign government licenses; and
- (iv) all sublicenses from Certegy to members of the Certegy Group (A) shall contain the rights and restrictions set forth in this Section 4.1(a) with respect to the license granted to Certegy and comply with Sections 4.1(b) through (d) hereof and (B) shall be diligently enforced by Certegy.
- (b) The sublicense rights granted to Certegy pursuant to Section 4.1(a) include the right for Certegy to grant sublicenses to the Licensed Equifax Materials (excluding the Utility Software Programs) to the members of the Certegy Group, which sublicenses may include the right to further sublicense such Licensed Equifax Materials (excluding the Utility Software Programs) to such Group member's customers solely for each such customer's internal business purposes to the extent related to the Certegy Business. All sublicensing by Certegy and other members of the Certegy Group to any one of their customers shall be pursuant to written agreements with such customer, executed before or at the time of furnishing each copy of the Licensed Equifax Materials (excluding the Utility Software Programs) to such customer, and which provide at a minimum that such customer:
 - (i) receives only a personal, non-transferable and nonexclusive right to use such copy of the Licensed Equifax Materials (excluding the Utility Software Programs);

- (ii) receives no title in the intellectual property contained in the Licensed Equifax Materials (excluding the Utility Software Programs);
- (iii) will not copy the Licensed Equifax Materials (excluding the Utility Software Programs), except as necessary to use such Licensed Equifax Materials (excluding the Utility Software Programs) in accordance with the license grant and to make one archival copy;
- (iv) will not export or re-export the Licensed Equifax Materials (excluding the Utility Software Programs) without the appropriate United States or foreign government licenses;
- (v) will hold the Licensed Equifax Materials (excluding the Utility Software Programs) in confidence; will not reverse compile or disassemble the Licensed Equifax Materials (excluding the Utility Software Programs); will not remove or destroy any proprietary markings of the licensor on or contained in the Licensed Equifax Materials (excluding the Utility Software Programs), and will include the copyright and patent notices of the licensor as specified from time to time by the licensor for the Licensed Equifax Materials (excluding the Utility Software Programs) on and in all copies of the Licensed Equifax Materials (excluding the Utility Software Programs); and
- (vi) will not sublicense, assign or otherwise transfer the Licensed Equifax Materials (excluding the Utility Software Programs) to any other Person.
- (c) In the event any member of the Certegy Group sublicenses any portion of the Licensed Equifax Materials (excluding the Utility Software Programs) to any third party pursuant to Section 4.1(a) and (b) above, Certegy agrees to ensure that such member shall diligently enforce the terms and conditions of all sublicenses granted pursuant to this Section 4.1.
- (d) In the event that Certegy, or another member of the Certegy Group, shall enter into a Divested Business transaction with respect to the Certegy Group, and the scope of permitted use or other terms applicable to the Licensed Equifax Materials (excluding the Utility Software Programs) under the license or sublicenses granted in this Section 4.1 are required to be modified to effect such transaction, Equifax will, or will cause the sublicensor under the applicable sublicense to, agree to such modifications to the extent (i) required for the transaction to be effected and (ii) not materially detrimental to the interests of the Equifax Group. Such modifications shall not be effective until the Divested Business or the acquiror thereof, as required by Equifax, has entered into a license agreement with the appropriate member of the Equifax Group incorporating the terms of Section 4.1 and Section 4.2 and such other terms as

Equifax reasonably deems appropriate for the protection of its interests in the Licensed Equifax Materials.

- (e) Without limiting the foregoing, Equifax hereby grants, and will cause the other members of the Equifax Group to grant, to Certegy a fully paid, non-exclusive, perpetual, worldwide, transferable license to use, modify, improve, create Derivative Works from, and sublicense, the Utility Software Programs (in both object and source code format) identified on Exhibit L as being owned by Equifax or a member of the Equifax Group for any and all fields of use and to any and all Persons.
- (f) The Licensed Equifax Materials may be marketed under such name and in such manner as Certegy chooses, consistent with the terms and conditions of this Agreement.
- (g) Except for the Certegy Group's rights described in Section 4.1(a), (b) and (e) above, the Equifax Group's rights in and to the Licensed Equifax Materials shall be and remain the exclusive property of Equifax or the members of the Equifax Group, and their respective successors and assigns.

Section 4.2. Ownership of Enhancements by Certegy.

- (a) Unless Exhibit J provides otherwise, Certegy, or the respective Designated Certegy Member, shall own all the modifications and improvements to, and the Certegy Enhancements and/or Derivative Works made from, the Licensed Equifax Materials developed by any member of the Certegy Group, or by any party other than a member of the Equifax Group at the expense of the Certegy Group. Equifax hereby assigns, and shall cause each member of the Equifax Group to assign, to Certegy, or the respective Designated Certegy Member, all right, title and interest it may hold in and to such modifications, improvements, Certegy Enhancements and Derivative Works. Certegy shall, or shall cause the respective Designated Certegy Member to, have the right to make and file all applications and other documents required to register the copyright(s) and file for patents for such modifications, improvements, Certegy Enhancements and Derivative Works in its discretion and at its sole cost and expense.
- (b) Should Certegy elect to file any application for the registration, perfection or protection of any modifications, improvements, Certegy Enhancements or Derivative Works described in Section 4.2(a), under any copyright, patent or other law of any country or jurisdiction, Equifax will, at the request and expense of Certegy, do all things and sign all documents or instruments reasonably necessary in the opinion of Certegy to assist in the registration of such claims, file such applications, and obtain, defend and enforce such copyright, patent, mask work and other rights.
- (c) Subject to the license rights granted in Section 4.1, as between the parties. the Licensed Equifax Materials shall be and shall remain the sole and exclusive property of the Equifax Group and the members of the Equifax Group may make any internal use and may commercially exploit any enhancements to the Licensed Materials made or caused to be made by

members of the Equifax Group, as they shall deem appropriate without any obligation to any member of the Certegy Group or other restriction. The Equifax Group may in particular distribute and manufacture, or cause to be manufactured or distributed by any third party, any such enhancements and/or the Licensed Equifax Materials.

Section 4.3. License to Marks.

- (a) Equifax hereby grants, and will cause each member of the Equifax Group to grant, to Certegy and each member of the Certegy Group a fully paid, non-exclusive, worldwide, non-transferable right to continue to use the Equifax Marks employed in the Certegy Business, but only to the extent such Equifax Marks were displayed by the Certegy Group prior to the Distribution Date (i) on the Transferred Equifax Assets, (ii) on premises jointly occupied with Equifax, and (iii) on letterhead, product and services documentation, invoices, software programs, packaging and similar materials used by the members of the Certegy Group, and such Equifax Marks are used in accordance with the guidelines for usage of the Equifax Marks published and amended by Equifax from time to time. Certegy will terminate the use of such Equifax Marks as soon as commercially practical but in any event within twelve (12) months after the Closing Date.
- (b) Certegy hereby grants, and will cause each member of the Certegy Group to grant, to Equifax and each member of the Equifax Group a fully paid, non-exclusive, worldwide, non-transferable right to continue to use the Marks that were owned immediately prior to the Distribution Date by a member of the Certegy Group and employed in the Equifax Business, but only to the extent such Marks were displayed by the Equifax Group prior to the Distribution Date (i) on assets owned by Equifax or any member of the Equifax Group (other than the Transferred Assets), (ii) on premises jointly occupied with one or more members of the Certegy Group, and (iii) on letterhead, product and services documentation, invoices, software programs, packaging and similar materials used by the members of the Equifax Group, and such Marks are used in accordance with the same guidelines for usage as the Equifax Marks as described in subsection (a) above. Equifax will terminate the use of such Marks as soon as commercially practical but in any event within twelve (12) months after the Distribution Date.

Section 4.4. Grant of License by Certegy.

- (a) Certegy hereby grants, and will cause the other members of the Certegy Group to grant, to Equifax a fully paid, non-exclusive, perpetual, worldwide, non-transferable license to use, modify, copy, improve, create Derivative Works and Equifax Enhancements from, and sublicense the Licensed Certegy Materials (excluding the Utility Software Programs) solely for use in the Equifax Business and as that business may evolve and change in the future, subject to the following:
 - (i) Equifax shall not sublicense, or otherwise disclose or distribute, or permit any Person to use, the Licensed Certegy Materials

(excluding the Utility Software Programs), except in accordance with Section 4.4(b);

- (ii) Equifax shall hold the Licensed Certegy Materials (excluding the Utility Software Programs) in strict confidence; will not remove or destroy any proprietary markings of the Certegy Group on or contained in the Licensed Certegy Materials (excluding the Utility Software Programs); and will include the copyright and patent notices of the licensor as specified from time to time by the licensor for the Licensed Certegy Materials (excluding the Utility Software Programs) on and in all copies of the Licensed Certegy Materials (excluding the Utility Software Programs);
- (iii) Equifax shall not export or re-export the Licensed Certegy Materials (excluding the Utility Software Programs) without the appropriate United States or foreign government license; and
- (iv) all sublicenses from Equifax to members of the Equifax Group (A) shall contain the rights and restrictions set forth in this Section 4.4(a) with respect to the license granted to Equifax and comply with Sections 4.4(b) through (d) hereof and (B) shall be diligently enforced by Equifax.
- (b) The sublicense rights granted to Equifax pursuant to Section 4.4(a) include the right for Equifax to grant sublicenses to the Licensed Certegy Materials (excluding the Utility Software Programs) to the members of the Equifax Group, which sublicenses may include the right to further sublicense such Licensed Certegy Materials (excluding the Utility Software Programs) to such Group member's customers solely for each such customer's internal business purposes to the extent related to the Equifax Business. All sublicensing by Equifax and other members of the Equifax Group to any one of their customers shall be pursuant to written agreements with such customer, executed before or at the time of furnishing each copy of the Licensed Certegy Materials (excluding the Utility Software Programs) to such customer, and which provide at a minimum that such customer:
 - (i) receives only a personal, non-transferable and nonexclusive right to use such copy of the Licensed Certegy Materials (excluding the Utility Software Programs);
 - (ii) receives no title in the intellectual property contained in the Licensed Certegy Materials (excluding the Utility Software Programs);
 - (iii) will not copy the Licensed Certegy Materials (excluding the Utility Software Programs), except as necessary to use such Licensed

Certegy Materials (excluding the Utility Software Programs) in accordance with the license grant and to make one archival copy;

- (iv) will not export or re-export the Licensed Certegy Materials (excluding the Utility Software Programs) without the appropriate United States or foreign government licenses;
- (v) will hold the Licensed Certegy Materials (excluding the Utility Software Programs) in confidence; will not reverse compile or disassemble the Licensed Certegy Materials (excluding the Utility Software Programs); will not remove or destroy any proprietary markings of the licensor on or contained in the Licensed Certegy Materials (excluding the Utility Software Programs); and will include the copyright and patent notices of the licensor as specified from time to time by the licensor for the Licensed Certegy Materials (excluding the Utility Software Programs) on and in all copies of the Licensed Certegy Materials (excluding the Utility Software Programs); and
- (vi) will not sublicense, assign or otherwise transfer the Licensed Certegy Materials (excluding the Utility Software Programs) to any other Person.
- (c) In the event any member of the Equifax Group sublicenses any portion of the Licensed Certegy Materials (excluding the Utility Software Programs) to any third party pursuant to Section 4.4(a) and (b) above, Equifax agrees to ensure that such member shall diligently enforce the terms and conditions of all sublicenses granted pursuant to this Section 4.4.
- (d) In the event that Equifax, or another member of the Equifax Group, shall enter into a Divested Business transaction with respect to the Equifax Group, and the scope of permitted use or other terms applicable to the Licensed Certegy Materials (excluding the Utility Software Programs) under the license or sublicenses granted in this Section 4.4 are required to be modified to effect such transaction, Certegy will, or will cause the sublicensor under the applicable sublicense to, agree to such modifications to the extent (i) required for the transaction to be effected and (ii) not materially detrimental to the interests of the Certegy Group. Such modifications shall not be effective until the Divested Business or the acquiror thereof, as required by Certegy, has entered into a license agreement with the appropriate member of the Certegy Group incorporating the terms of Section 4.4 and Section 4.5 and such other terms as Certegy reasonably deems appropriate for the protection of its interests in the Licensed Certegy Materials.
- (e) Without limiting the foregoing, Certegy hereby grants, and will cause the other members of the Certegy Group to grant, to Equifax a fully paid, non-exclusive, perpetual, worldwide, transferable license to use, modify, improve, create Derivative Works from, and

sublicense, the Utility Software Programs (in both object and source code format) identified on Exhibit L as being owned by Certegy or a member of the Certegy Group for any and all fields of use and to any and all Persons.

- (f) The Licensed Certegy Materials may be marketed under such name and in such manner as Equifax chooses, consistent with the terms and conditions of this Agreement.
- (g) Except for the Equifax Group's rights described in Section 4.4(a), (b) and (e) above, the Certegy Group's rights in and to the Licensed Certegy Materials shall be and remain the exclusive property of Certegy or the respective Designated Certegy Member.

Section 4.5. Ownership of Enhancements by Equifax.

- (a) Unless Exhibit K provides otherwise, Equifax, or the respective Designated Equifax Member, shall own all the modifications and improvements to, and the Equifax Enhancements and/or Derivative Works made from, the Licensed Certegy Materials developed by any member of the Equifax Group, or by any party other than a member of the Certegy Group at the expense of the Equifax Group. Certegy hereby assigns, and shall cause each member of the Certegy Group to assign, to Equifax, or the respective Designated Equifax Member, all right, title and interest it may hold in and to such modifications, improvements, Equifax Enhancements and Derivative Works. Equifax shall, or shall cause the respective Designated Equifax Member to, have the right to make and file all applications and other documents required to register the copyright(s) and file for patents for such modifications, improvements, Equifax Enhancements and Derivative Works in its discretion and at its sole cost and expense.
- (b) Should Equifax elect to file any application for the registration, perfection or protection of any modifications, improvements, Equifax Enhancements or Derivative Works described in Section 4.5(a), under any copyright, patent or other law of any country or jurisdiction, Certegy will, at the request and expense of Equifax, do all things and sign all documents or instruments reasonably necessary in the opinion of Equifax to assist in the registration of such claims, file such applications, and obtain, defend and enforce such copyright, patent, mask work and other rights.
- (c) Subject to the license rights granted in Section 4.4, as between the parties, the Licensed Certegy Materials shall be and shall remain the sole and exclusive property of the Certegy Group and the members of the Certegy Group may make any internal use and may commercially exploit any enhancements to the Licensed Materials made or caused to be made by members of the Equifax Group, as they shall deem appropriate without any obligation to any member of the Equifax Group or other restriction. The Certegy Group may in particular distribute and manufacture, or cause to be manufactured or distributed by any third party, any such enhancements and/or Licensed Certegy Materials.

Section 4.6. Data.

In no event shall any member of either Group be deemed to have been granted any rights under this Agreement in or to any data owned or maintained by any other member of the other Group, except as specifically provided in Sections 2.1 or 2.2. The respective rights of the members of each Group in and to such data shall be governed exclusively by Sections 2.1 or 2.2 and the Intercompany Data Purchase Agreement.

Section 4.7. Mutual Obligations.

- (a) The parties agree and acknowledge that all right, title and interest in and to any Jointly Owned IP Assets shall be jointly owned by the respective owners thereof, as identified on Exhibit M. Subject to any applicable provisions set forth on Exhibit M or a separate agreement between the joint owners thereof with respect to specific Jointly Owned IP Assets, each respective owner shall have the right to make, use, sell, copy, distribute, import, develop Derivative Works or enhancements of, and publicly perform and display, for any purpose, any Jointly Owned IP Assets without any obligation (including any obligation to account or pay royalties) or liability to the other owner(s). Except as otherwise set forth on Exhibit M, any modifications, enhancements or Derivative Works made by a joint owner to any Jointly Owned IP Asset shall be owned by the entity that modified or enhanced such Jointly Owned IP Asset or created such Derivative Work from such Jointly Owned IP Asset. Any patent rights to any invention that (i) has been incorporated into a Jointly Owned IP Asset and (ii) was created or developed (in any form or manner) prior to the Distribution Date, shall by jointly owned by the respective parties. Each owner of a Jointly Owned IP Asset shall (i) have the right to enforce, in any country, all rights embodied in such Jointly Owned IP Asset, and each other owner thereof agrees (at its expense) to cooperate in such enforcement action as reasonably requested by the other owner thereof, and (ii) have the right to file (in the names of all joint owners, except in the case filings with respect to modifications, enhancements or Derivative Works) appropriate patent, trademark, copyright or other applications, in any country, with respect to such Jointly Owned IP Assets.
- (b) The parties acknowledge that the Licensed Materials are "intellectual property" within the meaning of Section 101 of the Federal Bankruptcy Act and shall be subject to Section 365(n) thereof, all as set forth in the Intellectual Property Bankruptcy Protection Act, Public Law 100-506, 102 Stat. 2538.
- (c) In full and complete payment of the licenses granted in this Agreement, the parties have made the payment described in the Distribution Agreement as set forth in the Distribution Agreement.
- (d) Each party shall notify the other party of any involuntary attachment or other judicial process affecting the Licensed Materials.

ARTICLE V

THE CLOSING

Section 5.1. Equifax Deliverables.

On or before the Distribution Date, and effective as of the Closing Date, Equifax will, and/or will cause each member of the Equifax Group to, deliver to Certegy each of the following:

- (a) Duly executed assignment and assumption agreements necessary for the assignment and transfer to, and the assumption by Certegy of, the Transferred Equifax Assets and Transferred Equifax Third Party Agreements;
- (b) Duly executed assignment and assumption agreements necessary for the assignment and transfer to, and the assumption by Equifax of, the Equifax Liabilities; and
- (c) Such other agreements, leases, documents or instruments as the parties may agree are necessary or desirable in order to achieve the purposes of this Agreement.

Section 5.2. Certegy Deliverables.

On the Closing Date, Certegy will, and/or will cause each member of the Certegy Group to, deliver to Equifax each of the following:

- (a) Duly executed assignment and assumption agreements necessary for the assignment and transfer to, and the assumption by Equifax of, the Transferred Certegy Assets and the Transferred Certegy Third Party Agreements;
- (b) Duly executed assignment and assumption agreements necessary for the assignment and transfer to, and the assumption by Certegy of, the Certegy Liabilities; and
- (c) Such other agreements, documents or instruments as the parties may agree are necessary or desirable in order to achieve the purposes of this Agreement.

ARTICLE VI

REPRESENTATIONS AND WARRANTIES

Except as expressly set forth in this Agreement or any Related Agreement, neither any member of the Equifax Group, nor any member of the Certegy Group, has given or is giving any representation or warranty regarding the assets or Liabilities retained, transferred, assumed or licensed as contemplated hereby or thereby, including without limitation, (i) title to the assets, (ii) validity of the Liabilities, (iii) any lien, claim or other encumbrance affecting the assets or Liabilities, or (iv) the value of the assets and the amount of the Liabilities. Except as may be expressly set forth in this Agreement or any Related Agreement, all assets and Liabilities were, or are being, transferred, assigned, licensed, assumed, or are being retained, on an "AS IS",

"WHERE IS" basis and the respective transferees, licensees and assignees will bear the economic and legal risks that any such conveyance (x) shall prove to be insufficient to vest in the transferee a title that is free and clear of any lien, claim or other encumbrance, or (y) shall not constitute an infringement of a third party's rights.

ARTICLE VII

INDEMNIFICATION

Section 7.1. Certegy Indemnification of the Equifax Group.

If the Distribution occurs, on and after the Distribution Date, Certegy shall indemnify, defend and hold harmless each member of the Equifax Group, and each of their respective directors, officers, employees and agents (collectively the "Equifax Indemnitees") from and against any and all damage, loss, liability and expense, (including without limitation, reasonable expenses of investigation and reasonable attorneys' fees and expenses) in connection with any and all Actions or threatened Actions) (collectively, "Indemnifiable Losses") incurred or suffered by any of the Equifax Indemnitees and arising out of, or due to, the failure of Certegy, or any other member of the Certegy Group, to timely pay, perform or otherwise discharge, any of the Certegy Liabilities or its obligations under this Agreement.

Section 7.2. Equifax Indemnification of the Certegy Group.

If the Distribution occurs, on and after the Distribution Date, Equifax shall indemnify, defend and hold harmless each member of the Certegy Group and each of their respective directors, officers, employees and agents (collectively the "Certegy Indemnitees") from and against any and all Indemnifiable Losses incurred or suffered by any of the Certegy Indemnitees and arising out of, or due to, the failure of Equifax, or any other member of the Equifax Group, to timely pay, perform or otherwise discharge, any of the Equifax Liabilities or its obligations under this Agreement or any Related Agreement.

Section 7.3. Insurance and Third Party Obligations.

No insurer or any other third party shall be, by virtue of the foregoing indemnification provisions, (a) entitled to a benefit it would not be entitled to receive in the absence of such provisions, (b) relieved of the responsibility to pay any claims to which it is obligated, or (c) entitled to any subrogation rights with respect to any obligation hereunder.

ARTICLE VIII

INDEMNIFICATION PROCEDURES

Section 8.1. Notice and Payment of Claims.

If any Equifax Indemnitee or Certegy Indemnitee (the "Indemnified Party") determines that it is or may be entitled to indemnification by a party (the "Indemnifying Party") under Article VII (other than in connection with any Action or claim subject to Section 8.2), the Indemnified Party shall deliver to the Indemnifying Party a written notice specifying, to the extent reasonably practicable, the basis for its claim for indemnification and the amount for which the Indemnified Party reasonably believes it is entitled to be indemnified. After the Indemnifying Party shall have been notified of the amount for which the Indemnified Party seeks indemnification, the Indemnifying Party shall, within thirty (30) days after receipt of such notice, pay the Indemnified Party such amount in cash or other immediately available funds (or reach agreement with the Indemnified Party as to a mutually agreeable alternative payment schedule) unless the Indemnifying Party objects to the claim for indemnification or the amount thereof. If the Indemnifying Party does not give the Indemnified Party written notice objecting to such claim and setting forth the grounds therefor within the same thirty (30) day period, the Indemnifying Party shall be deemed to have acknowledged its liability for such claim and the Indemnified Party may exercise any and all of its rights under applicable law to collect such amount. Any amount owed under this Section 8.1 that is past due shall bear interest at a simple rate of interest per annum equal to the lesser of 1% per month or the maximum amount permitted by law.

Section 8.2. Notice and Defense of Third Party Claims.

(a) Promptly following the earlier of (i) receipt of notice of the commencement by a third party of any Action against or otherwise involving any Indemnified Party, or (ii) receipt of information from a third party alleging the existence of a claim against an Indemnified Party, with respect to which indemnification may be sought pursuant to this Agreement (a "Third Party Claim"), the Indemnified Party shall give the Indemnifying Party written notice thereof. The failure of the Indemnified Party to give notice as provided in this Section 8.2(a) shall not relieve the Indemnifying Party of its obligations under this Agreement, except to the extent that the Indemnifying Party is prejudiced by such failure to give notice. Within thirty (30) days after receipt of such notice, the Indemnifying Party shall by giving written notice thereof to the Indemnified Party, (i) acknowledge, as between the parties hereto, liability for and, at its option, elect to assume the defense of such Third Party Claim at its sole cost and expense, or (ii) object to the claim of indemnification set forth in the notice delivered by the Indemnified Party pursuant to the first sentence of this Section 8.2(a) setting forth the grounds therefor; provided that if the Indemnifying Party does not within the same thirty (30) day period give the Indemnified Party written notice acknowledging liability or objecting to such claim and setting forth the grounds therefor, the Indemnifying Party shall be deemed to have acknowledged, as between the parties hereto, its liability for such Third Party Claim.

- (b) Any contest of a Third Party Claim as to which the Indemnifying Party has elected to assume the defense shall be conducted by attorneys employed by the Indemnifying Party and reasonably satisfactory to the Indemnified Party; provided that the Indemnified Party shall have the right to participate in such proceedings and to be represented by attorneys of its own choosing at the Indemnified Party's sole cost and expense. If the Indemnifying Party assumes the defense of a Third Party Claim, the Indemnifying Party may settle or compromise the claim without the prior written consent of the Indemnified Party; provided that the Indemnifying Party may not agree to any such settlement pursuant to which any such remedy or relief, other than monetary damages for which the Indemnifying Party shall be responsible hereunder, shall be applied to or against the Indemnified Party without the prior written consent of the Indemnified Party, which consent shall not be unreasonably withheld.
- (c) If the Indemnifying Party does not assume the defense of a Third Party Claim for which it has acknowledged liability for indemnification under Article VII, the Indemnified Party may require the Indemnifying Party to reimburse it on a current basis for its reasonable expenses of investigation, reasonable attorneys' fees and reasonable out-of-pocket expenses incurred in defending against such Third Party Claim and the Indemnifying Party shall be bound by the result obtained with respect thereto by the Indemnified Party; provided that the Indemnifying Party shall not be liable for any settlement effected without its consent, which consent shall not be unreasonably withheld. The Indemnifying Party shall pay to the Indemnified Party in cash the amount for which the Indemnified Party is entitled to be indemnified (if any) within fifteen (15) days after the final resolution of such Third Party Claim (whether by settlement, compromise, or by the final nonappealable judgment of a court of competent jurisdiction or otherwise), or, in the case of any Third Party Claim as to which the Indemnifying Party has not acknowledged liability, within fifteen (15) days after the Indemnifying Party's objection has been resolved by settlement, compromise, or the final nonappealable judgment of a court of competent jurisdiction.
- (d) Notwithstanding the foregoing, in no event shall either Equifax or Certegy, or any Member of their respective Group, have any liability, whether based on contract, tort (including, without limitation, negligence or strict liability), warranty or any other legal or equitable grounds, for any punitive, consequential, indirect, exemplary, special or incidental loss or damage suffered by the other arising from or related to this Agreement, including without limitation, loss of data, profits, interest or revenue, or interruption of business, even if such party has been informed of or might otherwise anticipated or foreseen the possibility of such losses or damages; provided, however, that such limitations shall not apply to liabilities which may arise as the result of willful misconduct of a party. Notwithstanding the foregoing, any damages awarded or obtained (whether by settlement, compromise or judgment) as a result of Third Party Claims shall be considered direct damages for purposes of this Agreement.

ARTICLE IX

CONFIDENTIALITY

Section 9.1. Exclusions.

Notwithstanding anything to the contrary contained in this Agreement, "Company Information" does not include any information that before being divulged by the receiving party (a) has become generally known to the public through no wrongful act of the receiving party; (b) has been rightfully received by the receiving party from a third party without restriction on disclosure and without, to the knowledge of the receiving party, a breach of an obligation of confidentiality running directly or indirectly to the disclosing party; (b) has been approved for release to the general public by a written authorization of the disclosing party; (d) has been independently developed by the receiving party without use, directly or indirectly, of Company Information received from the disclosing party; or (e) has been furnished to a third party by the disclosing party without restrictions on the third party's rights to disclose the information.

Section 9.2. Confidentiality.

- (a) Each party acknowledges, and shall cause each member of the Group to acknowledge, that it is in possession of significant confidential or proprietary information concerning the business, operations and assets of the members of the other Group.
- (b) Each party shall, and shall ensure that each member of its Group shall, (i) receive and hold the Company Information of the other Group in trust and in strictest confidence; (ii) protect such Company Information from disclosure and in no event take any action causing, or fail to take the action necessary in order to prevent, any such Company Information to lose its character as Company Information; (iii) exercise at a minimum the same care it would exercise to protect its own highly confidential information; and (iv) not use, reproduce, distribute, disclose, or otherwise disseminate the Company Information of the other Group, (A) except as authorized pursuant to this Agreement or any Related Agreement, or (B) except pursuant to a requirement of a governmental agency or of law without similar restrictions or other protections against public disclosure; provided, however, with respect to disclosures pursuant to (B) above, the receiving party must first give written notice of such required disclosure to the disclosing party, take reasonable steps to allow the disclosing party to seek to protect the confidentiality of the Company Information required to be disclosed, make a reasonable effort to obtain a protective order requiring that the Company Information so disclosed be used only for the purposes for which disclosure is required, and shall disclose only that part of the Company Information which, in the written opinion of its legal counsel, it is required to disclose. In no event shall the receiving party exercise less than a reasonable standard of care to keep confidential the Company Information. Any and all reproductions of such Company Information must prominently contain a confidential legend.

- (c) The receiving party may make disclosures of the Company Information of the disclosing party only to Representatives of the receiving party's Group (i) who have a specific need to know such information; and (ii) who the receiving party has obligated under a written agreement to hold such Company Information in trust and in strictest confidence and otherwise to comply with the terms and provisions of this Agreement or terms and conditions substantially similar to and implementing the same restrictions and covenants as those set forth in this Agreement. Certegy and Equifax agree, and shall ensure that each member of their respective Group agrees, to diligently monitor each such Representative, diligently enforce such agreements with its Representatives, and, upon request by the other party, promptly to furnish to the other party a certified list of the receiving party's Representatives having had access to such Company Information.
- (d) The covenants of confidentiality set forth in this Agreement (i) will apply after the Closing Date to all Company Information disclosed to the receiving party before, on and after the Closing Date and (ii) will continue and must be maintained from the Closing Date through the termination of the relationship under this Agreement between Equifax and Certegy (A) with respect to Proprietary Information, the period during which the Proprietary Information constituting a part of the Company Information retains its status as a "trade secret" under applicable law; and (B) with respect to Confidential Information constituting a part of the Company Information, for the shorter of a period equal to three (3) years after the Closing Date, or until such Confidential Information no longer qualifies as confidential under applicable law.

Section 9.3. Employee Confidentiality Agreements.

The members of each Group have entered into confidentiality and nondisclosure agreements with their respective employees. To the extent that any employee during or after employment violates any such agreement and such violation is or may in the future be to the detriment of the other Group, at the written request of the affected party, the other party shall, or shall cause the appropriate members of its Group to, promptly bring and diligently pursue an action against such employee if and to the extent reasonable under the circumstances to preserve the value of the assets and Licensed Materials. The Group member employing the employee violating his/her confidentiality and nondisclosure agreement shall have the unilateral right to determine the forum for, the manner of proceeding in, and legal counsel for such action and shall be entitled to any damages or other relief against such employee awarded in such action to the extent related to such Group's assets or business or to the Licensed Materials. Such enforcement against and recovery by a Group member from its breaching employee shall not constitute a release or sole remedy for the members of the other Group injured by such breaching employee's actions, and such members of the other Group may bring a claim against the Group members employing the breaching employee for a breach of this Agreement. Each party shall bear all out-of-pocket costs of pursuing such action and the other party shall cooperate in connection therewith.

Section 9.4. Rights and Remedies.

- (a) If either party, or any member of the Group, should breach or threaten to breach any of the provisions of this Agreement, the non-breaching party, in addition to any other remedies it may have at law or in equity, will be entitled to a restraining order, injunction, or other similar remedy in order to specifically enforce the provisions of this Agreement. Each party specifically acknowledges, and shall cause each member of its respective Group to acknowledge, that money damages alone would be an inadequate remedy for the injuries and damage that would be suffered and incurred by the non-breaching party as a result of a breach of any of the provisions of this Agreement. In the event that either party, or a member of such party's Group, should seek an injunction hereunder, the other party hereby waives, and shall cause each member of its Group to waive, any requirement for the submission of proof of the economic value of any Company Information or the posting of a bond or any other security. In the event of a dispute between the parties, the non-prevailing party shall pay all costs and expenses associated with resolving the dispute, including, but not limited to, reasonable attorneys' fees.
- (b) The receiving party shall notify the disclosing party immediately upon discovery of any unauthorized use or disclosure of Company Information, or any other breach of this Agreement by the receiving party or any Representative of the receiving party's Group, and will cooperate with the disclosing party in every reasonable way to help the disclosing party regain possession of its Company Information and prevent its further unauthorized use or disclosure. The receiving party shall be responsible for the acts of any Representative of its Group that are in violation of this Agreement.

Section 9.5. Competitive Activities.

- (a) Subject to the rights and obligations set forth in this Article IX, each party understands and acknowledges that the other party's Group may now market or have under development products that are competitive with products or services now offered or that may be offered by it and/or members of its Group, and the parties' communications hereunder will not serve to impair the right of either party, or any member of its respective Group, to independently develop, make, use, procure, or market products or services now or in the future that may be competitive with those offered by the other party's Group, nor require either party, and/or the members of its Group, to disclose any planning or other information to the other party.
- (b) Neither party will be restricted in using, in the development, manufacturing and marketing of its products and services and its operations, any data processing or network management or operation ideas, concepts, know-how and techniques which are retained in the minds of employees who have had access to the other party's Company Information subject to the restrictions set forth in this Agreement.

Section 9.6. No Implied Rights.

Except as provided herein or in any Related Agreement, all Company Information is and shall remain the property of the disclosing party and/or the respective member of its Group. By disclosing Company Information to the receiving party's Group, the disclosing party and/or the

members of its Group do(es) not grant any express or implied rights or license to the receiving party's Group to or under any patents, patent applications, inventions, copyrights, trademarks, trade secret information, or other intellectual property rights heretofore or hereafter possessed by the disclosing party and/or the members of its Group.

ARTICLE X

CONTINUED ASSISTANCE

Section 10.1. Continued Assistance and Transition.

- (a) Following the Closing Date, Equifax shall, and shall cause each member of the Equifax Group to, cooperate in an orderly transfer of the Transferred Equifax Assets and the Transferred Equifax Third Party Agreements to Certegy or the respective Designated Certegy Member. From time to time, at Certegy's request and without further consideration, Equifax shall, and shall cause each member of the Equifax Group, as applicable, to execute, acknowledge and deliver such documents, instruments or assurances and take such other action as Certegy may reasonably request to more effectively assign, convey and transfer any of the Transferred Equifax Assets and the Transferred Equifax Third Party Agreements. Equifax will assist Certegy in the vesting, collection or reduction to possession of such Transferred Equifax Assets and Transferred Equifax Third Party Agreements.
- (b) Following the Closing Date, Certegy shall, and shall cause each member of the Certegy Group to, cooperate in an orderly transfer of the Transferred Certegy Assets and Transferred Certegy Third Party Agreements to Equifax or the respective Designated Equifax Member. From time to time, at Equifax's request and without further consideration, Certegy shall, and shall cause each member of the Certegy Group, as applicable, to execute, acknowledge and deliver such documents, instruments or assurances and take such other action as Equifax may reasonably request to more effectively assign, convey and transfer any of the Transferred Certegy Assets and Transferred Certegy Third Party Agreements. Certegy will assist Equifax in the vesting, collection or reduction to possession of such Transferred Certegy Assets and Transferred Certegy Third Party Agreements.

Section 10.2. Records and Documents.

- (a) As soon as practicable following the Closing Date, Equifax and Certegy shall each arrange for the delivery to the other of existing corporate and other documents (e.g. documents of title, source code, contracts, etc.) in its possession relating to the Transferred Assets, Transferred Third Party Agreements and assumed Liabilities.
- (b) From and after the Closing Date, Equifax and Certegy shall each, and shall cause each member of its Group to, afford the other and its accountants, counsel and other designated Representatives reasonable access (including using reasonable efforts to give access to person or firms possessing such information) and duplicating rights during normal business hours to all

records, books, contracts, instruments, computer data and other data and information in its possession relating to the assets, Liabilities, Licensed Materials, business and affairs of the other (other than data and information subject to any attorney/client or other privilege), insofar as such access is reasonably required by the other, including without limitation, for audit, accounting and litigation purposes.

(c) Notwithstanding the foregoing, either party may destroy or otherwise dispose of any information at any time in accordance with the corporate record retention policy maintained by such party with respect to its own records.

Section 10.3. Litigation Cooperation.

Upon written request, Equifax and Certegy shall, and shall cause each member of its Group to, use reasonable efforts to cooperate in the evaluation and defense of third party Actions arising out of the business of the other party or of any member of the other party's Group prior to the Distribution Date in which the requesting party or any member of its Group may from time to time be involved, at the cost and expense of the requesting party. Such cooperation shall include, without limitation, making its Representatives available as witnesses or consultants to the extent that such persons may reasonably be required in connection with such third party Actions.

ARTICLE XI

MISCELLANEOUS

Section 11.1. Expenses.

Except as specifically provided in this Agreement or any Related Agreement, all costs and expenses incurred in connection with the preparation, execution, delivery and implementation of this Agreement and with the consummation of the transactions contemplated by this Agreement (including transfer taxes and the fees and expenses of all counsel, accountants and financial and other advisors) shall be paid by Equifax.

Section 11.2. Notices.

All notices and communications under this Agreement shall be deemed to have been given (a) when received, if such notice or communication is delivered by facsimile, hand delivery or overnight courier, or (b) three (3) business days after mailing if such notice or communication is sent by United States registered or certified mail, return receipt requested, first class postage prepaid. All notices and communications, to be effective, must be properly addressed to the party to whom the same is directed at its address as follows:

If to Equifax, to: Equifax Inc. 1550 Peachtree Street Atlanta, Georgia 30309

Attn: Phillip J. Mazzilli, Chief Financial Officer

Fax: (404) 885-8682

with a copy to:

Equifax Inc. 1550 Peachtree Street Atlanta, Georgia 30309

Attn: Kent E. Mast, General Counsel

Fax: (404) 885-8988

If to Certegy, to:

Certegy Inc. P.O. Box 349

Alpharetta, Georgia 30009

Attn: Bruce S. Richards, General Counsel

with a copy to:

Certegy Inc. 555 North Point Centre East Alpharetta, Georgia 30022

Attn: Michael T. Vollkommer, Corporate Vice President and Chief Financial

Officer

Either party may, by written notice so delivered to the other party in accordance with this Section 11.2, change the address to which delivery of any notice shall thereafter be made.

Section 11.3. Amendment and Waiver.

This Agreement may not be altered or amended, nor may any rights hereunder be waived, except by an instrument in writing executed by the party or parties to be charged with such amendment or waiver. No waiver of any terms, provision or condition of or failure to exercise or delay in exercising any rights or remedies under this Agreement, in any one or more instances, shall be deemed to be, or construed as, a further continuing waiver of any such term, provision, condition, right or remedy or as a waiver of any other term, provision or condition of this Agreement.

Section 11.4. Entire Agreement.

This Agreement, together with the Related Agreements, constitutes the entire understanding of the parties hereto with respect to the subject matter hereof, superseding all negotiations, prior discussions and prior agreements and understandings relating to such subject matter. To the extent that the provisions of this Agreement are inconsistent with the provisions of any other Related Agreement, the provisions of this Agreement shall prevail with respect to the subject matter hereof.

Section 11.5. Parties in Interest.

Neither of the parties hereto may assign its rights or delegate any of its duties under this Agreement without the prior written consent of each other party. This Agreement shall be binding upon, and shall inure to the benefit of, the parties hereto and their respective successors and permitted assigns. Nothing contained in this Agreement, express or implied, is intended to confer any benefits, rights or remedies upon any person or entity other than members of the Equifax Group and the Certegy Group and the Equifax Indemnitees and Certegy Indemnitees under Articles VII and VIII hereof.

Section 11.6. Further Assurances and Consents.

In addition to the actions specifically provided for elsewhere in this Agreement, each of the parties hereto will use its reasonable efforts to (a) execute and deliver such further instruments and documents and take such other actions as any other party may reasonably request in order to effectuate the purposes of this Agreement and to carry out the terms hereof and (b) take, or cause to be taken, all actions, and to do, or cause to be done, all things, reasonably necessary, proper or advisable under applicable laws, regulations and agreements or otherwise to consummate and make effective the transactions contemplated by this Agreement, including without limitation, using its reasonable efforts to obtain any consents and approvals and to make any filings and applications necessary or desirable in order to consummate the transactions contemplated by this Agreement; provided that no party hereto shall be obligated to pay any consideration therefor (except for filing fees and other similar charges) to any third party from whom such consents, approvals and amendments are requested or to take any action or omit to take any action if the taking of or the omission to take such action would be unreasonably burdensome to the party or its Group or the business thereof.

Section 11.7. Severability.

The provisions of this Agreement are severable and should any provision hereof be void, voidable or unenforceable under any applicable law, such provision shall not affect or invalidate any other provision of this Agreement, which shall continue to govern the relative rights and duties of the parties as though such void, voidable or unenforceable provision were not a part hereof.

Section 11.8. Governing Law.

This Agreement shall be construed in accordance with, and governed by, the laws of the State of Georgia, without regard to the conflicts of law rules of such state.

Section 11.9. Counterparts.

This Agreement may be executed in one or more counterparts, each of which shall be deemed an original instrument, but all of which together shall constitute but one and the same agreement.

Section 11.10. Disputes.

Any Disputes arising under this Agreement, whether based on contract, tort, statute or otherwise, including but not limited to, disputes in connection with claims by third parties, shall be resolved in accordance with Section 15.10 of the Distribution Agreement; provided that the parties shall retain the rights and remedies specified in Section 9.4 hereof.

Section 11.11. Force Majeure.

Neither party will be liable for any loss or damage due to causes beyond its control, including, but not limited to, fire, accident, labor difficulty, war, power or transmission failures, riot, Acts of God or changes in laws and regulations, provided that the affected party must (a) promptly notify the other party in writing and furnish all relevant information concerning the event of force majeure; (b) use reasonable efforts to avoid or remove the cause of its nonperformance; and (c) proceed to perform its obligations with dispatch when such cause is removed.

Section 11.12. Documentation.

Prior to the Distribution Date and from time to time thereafter, the parties will prepare, maintain and update schedules of the Transferred Equifax Assets, the Transferred Certegy Assets, the Licensed Equifax Materials, the Licensed Certegy Materials, and the Third Party Agreements, the Third Party Use Rights and the Third Party Rights transferred and/or provided by each Group to the other Group, in such detail as shall be appropriate for the management and administration of these items as described in this Agreement.

Section 11.13. Headings.

The Article and Section headings set forth in this Agreement are included for administrative, organizational and convenience purposes, and are not intended to affect the meaning of the provisions set forth in this Agreement or to be used in the interpretation of this Agreement.

IN WITNESS WHEREOF, the parties hereto have executed and delivered this Agreement as of the day and year first above written.

EQUIFAX	INC.
By:	
Title:	
Date:	
CERTEGY	INC.
By:	
Title:	
Date:	

CERTEGY INC. 2001 STOCK INCENTIVE PLAN

1. Purpose and Effective Date. The purpose of the 2001 Stock Incentive

Plan is to attract and retain directors, officers and other key employees for Certegy Inc., a Georgia corporation (the "Company"), and its Subsidiaries and to provide those persons with incentives and rewards for superior performance. The Plan is effective as of June ____, 2001, the date the Plan was approved by the Company's Board of Directors. The Plan was approved by Equifax Inc., as the sole shareholder of the Company, on June ___, 2001.

2. Definitions. As used in this Plan:

"Appreciation Right" means a right granted pursuant to Section 5 of this Plan, and shall include both Tandem Appreciation Rights and Free-Standing Appreciation Rights.

"Base Price" means the price to be used as the basis for determining the Spread upon the exercise of a Free-Standing Appreciation Right and a Tandem Appreciation Right.

"Board" means the Board of Directors of the Company.

"Change in Control" shall have the meaning provided in Section 11 of this $\operatorname{Plan}\nolimits.$

"Code" means the Internal Revenue Code of 1986, as amended from time to time.

"Committee" means the Compensation and Human Resources Committee of the Board, or any successor committee to which the responsibilities of that Committee are assigned.

"Common Share" means shares of common stock, par value \$.01 per share, of the Company or any security into which such Common Shares may be changed by reason of any transaction or event of the type referred to in Section 10 of this Plan.

"Company" means Certegy Inc., a Georgia corporation.

"Covered Employee" means a Participant who is, or is determined by the Board to be likely to become, a "covered employee" within the meaning of Section 162(m) of the Code (or any successor provision).

"Date of Grant" means the date specified by the Board on which a grant of Option Rights or Appreciation Rights, or a grant or sale of Restricted Shares or Deferred Shares shall become effective (which date shall not be earlier than the date on which the Board or its designee takes action with respect thereto).

"Deferral Period" means the period of time during which Deferred Shares are subject to deferral limitations under Section 7 of this Plan.

"Deferred Shares" means an award made pursuant to Section 7 of this Plan of the right to receive Common Shares at the end of a specified Deferral Period.

"Director" means a member of the Board of Directors of the Company.

"Employee Benefits Agreement" means the Employee Benefits Agreement between Equifax Inc. and the Company dated as of June ____, 2001 which provides for the treatment of the employee plans in connection with the spin-off of the Company from Equifax, Inc.

"Equifax Stock Incentive Plans" means the stock incentive plans sponsored by Equifax Inc., including the Equifax Inc. 2000 Stock Incentive Plan, the Equifax Inc. Omnibus Stock Incentive Plan, the 1995 Employee Stock Incentive Plan or the 1993 Employee Stock Incentive Plan.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder, as such law, rules and regulations may be amended from time to time, including any successor statutes of similar intent.

"Free-Standing Appreciation Right" means an Appreciation Right granted pursuant to Section 5 of this Plan that is not granted in tandem with an Option Right.

"Immediate Family" has the meaning ascribed thereto in Rule 16a-1(e) under the Exchange Act (or any successor rule to the same effect).

"Incentive Stock Options" means Option Rights that are intended to qualify as "incentive stock options" under Section 422 of the Code or any successor provision.

"Management Objectives" means the measurable performance objective or objectives established pursuant to this Plan for Participants who have received grants of Option Rights, Appreciation Rights, Restricted Shares and dividend credits pursuant to this Plan, which are subject to the achievement of Management Objectives. Management Objectives may be described in terms of Company-wide objectives or objectives that are related to the performance of the individual Participant or of the Subsidiary, division, department, region or function within the Company or Subsidiary in which the Participant is employed. The Management Objectives may be made relative to the performance of other corporations. The Management Objectives applicable to any award to a Covered Employee shall be based on specified levels of, or growth in, one or more of the following criteria, as determined for a single year, or cumulatively for a stated number of years, or as an average over a stated number of years, or otherwise as determined by the Committee at the time the Management Objective is established:

- earnings
- 2. earnings per share;
- 3. economic value added;
- 4. revenue;
- operating profit;
- 6. net income;
- total return to shareholders;
- 8. market share;
- 9. profit margins;
- cash flow/net assets ratio;
- debt/capital ratio;

- 12. return on total capital;
- 13. return on equity;
- 14. return on assets; and
- 15. common stock price.

If the Committee determines that a change in the business, operations, corporate structure or capital structure of the Company, or the manner in which it conducts its business, or other events or circumstances render the Management Objectives unsuitable, the Committee may in its discretion modify such Management Objectives or the related minimum acceptable level of achievement, in whole or in part, as the Committee deems appropriate and equitable, except in the case of a Covered Employee where such action would result in the loss of the otherwise available exemption of the award under Section 162(m) of the Code. In the case of a Covered Employee, in determining financial results, items whose exclusion from consideration will increase the Award shall only have their effects excluded if they constitute "extraordinary" or "unusual" events or items under generally accepted accounting principles and all such events and items shall be excluded. The Committee shall also adjust the performance calculations to exclude the unanticipated effect on financial results of changes in the Code, or other tax laws, and the regulations thereunder. The Committee may decrease the amount of an Award otherwise payable if, in the Committee's view, the financial performance during the performance cycle justifies such adjustment, regardless of the extent to which the Performance Measure was achieved.

"Market Value per Share" means, (i) the closing sale price per Common Share as reported on the principal exchange on which Common shares are then trading, if any, or, if applicable, the NASDAQ National Market System, on the Date of Grant, or if there are no sales on such day, on the next preceding trading day during which a sale occurred, or (ii) if clause (i) does not apply, the fair market value of the Common Shares as determined by the Board.

"Non-Employee Director" means a Director who is not an employee of the Company or any Subsidiary.

"Optionee" means the optionee named in an agreement evidencing an outstanding Option Right.

"Option Price" means the purchase price payable on exercise of an Option Right.

"Option Right" means the right to purchase Common Shares upon exercise of an option granted pursuant to Section 4 or Section 8 of this Plan.

"Participant" means a person who is selected by the Committee to receive benefits under this Plan and who is at the time an officer, or other key employee of the Company or any one or more of its Subsidiaries, or who has agreed to commence serving in any of such capacities within 60 days of the Date of Grant, and shall also include each Non-Employee Director who receives an award of Option Rights or Restricted Shares, or any other person, whether or not an employee, Non-Employee Director or officer, who renders significant services as a consultant or otherwise, in the discretion of the Committee.

"Plan" means this Certegy Inc. 2001 Stock Incentive Plan, as it may be amended from time to time.

"Reload Option Rights" means additional Option Rights granted automatically to an Optionee upon the exercise of Option Rights pursuant to Section 4(f) of this Plan.

"Replacement Awards" means Option Rights or Restricted Shares that are issued in substitution of awards of option rights or restricted shares that were granted under the Equifax Stock Incentive Plans to former employees of Equifax Inc. or subsidiaries of Equifax Inc. who are employees of the Company or its Subsidiaries as of the date of the spin-off of the Company to the shareholders of Equifax Inc. or who become employees of the Company after such date pursuant to the Employee Benefits Agreement. As provided in Section 4(o), the Replacement Awards shall have the same material terms and conditions under the Plan as such awards had under the respective Equifax Stock Incentive Plans.

"Restricted Shares" means Common Shares granted or sold pursuant to Section 6 or Section 8 of this Plan as to which neither the substantial risk of forfeiture nor the prohibition on transfers referred to in Section 6 has expired.

"Rule 16b-3" means Rule 16b-3 under the Exchange Act (or any successor rule to the same effect) as in effect from time to time.

"Securities Act" means the Securities Act of 1933, as amended.

"Spread" means the excess of the Market Value per Share on the date when an Appreciation Right is exercised, or on the date when Option Rights are surrendered in payment of the Option Price of other Option Rights, over the Option Price or Base Price provided for in the related Option Right or Free-Standing Appreciation Right, respectively.

"Subsidiary" means a corporation, company or other entity (i) more than 50 percent of whose outstanding shares or securities (representing the right to vote for the election of directors or other managing authority) are, or (ii) which does not have outstanding shares or securities (as may be the case in a partnership, joint venture or unincorporated association), but more than 50 percent of whose ownership interest representing the right generally to make decisions for such other entity is, now or hereafter, owned or controlled, directly or indirectly, by the Company, except that for purposes of determining whether any person may be a Participant for purposes of any grant of Incentive Stock Options, "Subsidiary" means any corporation in which, at the time of the grant, the Company owns or controls, directly or indirectly, more than 50 percent of the total combined voting power represented by all classes of stock issued by such corporation.

"Tandem Appreciation Right" means an Appreciation Right granted pursuant to Section 5 of this Plan that is granted in tandem with an Option Right.

"Voting Power" means at any time, the total votes relating to the thenoutstanding securities entitled to vote generally in the election of Directors.

- 3. Shares Available Under the Plan.
- (a) Subject to the adjustments provided for in Section 3(b) and Section 10 of this Plan, the number of Common Shares that may be issued or transferred (i) upon the exercise of Option Rights or Appreciation Rights, (ii) as Restricted Shares and released from substantial risks of forfeiture thereof, (iii) as Deferred Shares, (iv) as awards to Non-Employee Directors or (v) in payment of dividend equivalents paid with respect to Awards made under the Plan shall not exceed in the aggregate 6,600,000 Common Shares. Such shares may be shares of original issuance or treasury shares or a combination of the foregoing.
- (b) The number of Common Shares available in Section 3(a) above shall be adjusted to account for shares relating to awards that expire, are forfeited or are transferred, surrendered or relinquished upon the payment of any Option Price by the transfer to the Company of Common Shares or upon satisfaction of any withholding amount. Upon payment in cash of the benefit provided by any award granted under this Plan, any shares that were covered by that award shall again be available for issue or transfer hereunder. In addition to these adjustments, commencing on January 1, 2002, and on each January 1 thereafter ending on January 1, 2008, an additional number of Common Shares shall be added to the total available under Section 3(a), equal to one and a half percent (1 1/2%) of the number of Common Shares issued and outstanding on that January 1st date.
- (c) Notwithstanding anything in this Section 3, or elsewhere in this Plan, to the contrary and subject to adjustment as provided in Section 10 of this Plan, the aggregate number of Common Shares actually issued or transferred by the Company under this Plan upon the exercise of Incentive Stock Options shall not exceed 10,000,000 Common Shares. Subject to adjustments as provided in Section 10, no Participant shall be granted Option Rights and Appreciation Rights, in the aggregate, for more than 750,000 Common Shares during any one calendar year; the number of shares issued as Restricted Shares to any Participant shall not exceed 500,000 Common Shares in any one calendar year; and no Non-Employee Director shall be granted Option Rights, Appreciation Rights and Restricted Shares, in the aggregate, for more than 100,000 Common Shares in any one calendar year.
- 4. Option Rights. The Committee may, from time to time and upon such terms and conditions as it may determine, authorize the granting to Participants of options to purchase Common Shares. Such grants may be original awards or Replacement Awards. Each such grant may utilize any or all of the authorizations, and shall be subject to all of the requirements contained in the following provisions:
- (a) Each grant shall specify the number of Common Shares to which it pertains subject to the limitations set forth in Section 3 of this plan.
- (b) Each grant shall specify an Option Price per share, which may not be less than the Market Value per Share on the Date of Grant, provided that this restriction shall not apply to Replacement Awards .
- (c) Each grant shall specify whether the Option Price shall be payable (i) in cash or by check acceptable to the Company, (ii) by the actual or constructive transfer to the Company

of Common Shares owned by the Optionee for at least 6 months (or other consideration authorized pursuant to Section 4(d)) having a value at the time of exercise equal to the total Option Price, or (iii) by a combination of such methods of payment.

- (d) The Committee may determine, at or after the Date of Grant, that payment of the Option Price of any Option Right (other than an Incentive Stock Option) may also be made in whole or in part in the form of Restricted Shares or other Common Shares that are forfeitable or subject to restrictions on transfer, Deferred Shares, (based, in each case, on the Market Value per Share on the date of exercise), or other Option Rights (based on the Spread on the date of exercise). Unless otherwise determined by the Committee at or after the Date of Grant, whenever any Option Price is paid in whole or in part by means of any of the forms of consideration specified in this Section 4(d), the Common Shares received upon the exercise of the Option Rights shall be subject to such risks of forfeiture or restrictions on transfer as may correspond to any that apply to the consideration surrendered, but only to the extent, determined with respect to the consideration surrendered, of (i) the number of shares, or (ii) the Spread of any unexercisable portion of Option Rights.
- (e) Any grant may provide for deferred payment of the Option Price from the proceeds of sale through a bank or broker on a date satisfactory to the Company of some or all of the shares to which such exercise relates.
- (f) Any grant may, at or after the Date of Grant, provide for the automatic grant of Reload Option Rights to an Optionee upon the exercise of Option Rights (including Reload Option Rights) using Common Shares or other consideration specified in Section 4(d). Reload Option Rights shall cover up to the number of Common Shares, Deferred Shares, or Option Rights surrendered to the Company upon any such exercise in payment of the Option Price or to meet any withholding obligations. Reload Options may not have an Option Price that is less than the applicable Market Value per Share at the time of exercise and shall be on such other terms as may be specified by the Committee, which may be the same as or different from those of the original Option Rights.
- (g) Successive grants may be made to the same Participant whether or not any Option Rights previously granted to such Participant remain unexercised.
- (h) Each grant shall specify the period or periods of continuous service by the Optionee with the Company or any Subsidiary that is necessary before the Option Rights or installments thereof will become exercisable and may provide for continued vesting of the Option Rights after a termination of employment by reason of the Optionee's retirement, death, disability or other events as specified by the Committee. Each grant may also provide for the earlier exercise of such Option Rights in the event of a Change in Control, retirement, death or disability of the Optionee or other similar transaction or event.
- (i) Any grant of Option Rights may specify Management Objectives that must be achieved as a condition to the exercise of such rights.

- (j) Option Rights granted under this Plan may be (i) options, including, without limitation, Incentive Stock Options that are intended to qualify under particular provisions of the Code, (ii) options that are not intended so to qualify, or (iii) combinations of the foregoing.
- (k) The Committee may, at or after the Date of Grant of any Option Rights (other than Incentive Stock Options), provide for the payment of dividend equivalents to the Optionee on either a current or deferred or contingent basis or may provide that such equivalents shall be credited against the Option Price.
- (1) The exercise of an Option Right shall result in the cancellation on a share-for-share basis of any Tandem Appreciation Right authorized under Section 5 of this Plan.
- (m) No Option Right shall be exercisable more than 10 years from the Date of Grant.
- (n) Each grant of Option Rights shall be evidenced by an agreement or other written notice from the Company by an officer and delivered to the Optionee and containing such terms and provisions, consistent with this Plan, as the Committee may approve.
- (o) Each Replacement Award shall reflect the adjustments provided for in the Employee Benefits Agreements and shall have the same material terms and conditions as the award it replaces under the Equifax Stock Incentive Plans, as determined by the Committee. Notwithstanding any other provision in this Plan to the contrary, no Replacement Award in substitution of an award that qualified as an Incentive Stock Option immediately before the grant of the Replacement Award shall contain any term that is more favorable than the terms of the substituted award.

5. Appreciation Rights.

- (a) The Committee may authorize the granting (i) to any Optionee, of Tandem Appreciation Rights in respect of Option Rights granted hereunder, and (ii) to any Participant, of Free-Standing Appreciation Rights. A Tandem Appreciation Right shall be a right of the Optionee, exercisable by surrender of the related Option Right, to receive from the Company an amount determined by the Board, which shall be expressed as a percentage of the Spread (not exceeding 100 percent) at the time of exercise. Tandem Appreciation Rights may be granted at any time prior to the exercise or termination of the related Option Rights; provided, however, that a Tandem Appreciation Right awarded in relation to an Incentive Stock Option must be granted concurrently with such Incentive Stock Option. A Free-Standing Appreciation Right shall be a right of the Participant to receive from the Company an amount determined by the Committee, which shall be expressed as a percentage of the Spread (not exceeding 100 percent) at the time of exercise.
- (b) Each grant of Appreciation Rights may utilize any or all of the authorizations, and shall be subject to all of the requirements, contained in the following provisions:
 - (i) Any grant may specify that the amount payable on exercise of an Appreciation Right may be paid by the Company in cash, in Common Shares or in any combination thereof and may either grant to the Participant or retain in the Committee the right to elect among those alternatives .

- (ii) Any grant may specify that the amount payable on exercise of an Appreciation Right may not exceed a maximum specified by the Committee at the Date of Grant.
- (iii) Any grant may specify waiting periods before exercise and permissible exercise dates or periods.
- (iv) Any grant may specify that such Appreciation Right may be exercised only in the event of, or earlier in the event of, a Change in Control or other transaction or event.
- (v) Any grant may provide for the payment to the Participant of dividend equivalents thereon in cash or Common Shares on a current, deferred or contingent basis.
- (vi) Any grant of Appreciation Rights may specify Management Objectives that must be achieved as a condition of the exercise of such Rights.
- (vii) Each grant of Appreciation Rights shall be evidenced by an agreement executed on behalf of the Company by an officer and delivered to and accepted by the Participant, which agreement shall describe such Appreciation Rights, identify the related Option Rights (if applicable), state that such Appreciation Rights are subject to all the terms and conditions of this Plan, and contain such other terms and provisions, consistent with this Plan, as the Committee may approve.
- (c) Any grant of Tandem Appreciation Rights shall provide that such Rights may be exercised only at a time when the related Option Right is also exercisable and at a time when the Spread is positive, and by surrender of the related Option Right for cancellation.
 - (d) Regarding Free-standing Appreciation Rights only:
 - (i) Each grant shall specify in respect of each Free-standing Appreciation Right a Base Price, which shall be equal to or greater than the Market Value per Share on the Date of Grant;
 - (ii) Successive grants may be made to the same Participant regardless of whether any Free-standing Appreciation Rights previously granted to the Participant remain unexercised; and
- (iii) No Free-standing Appreciation Right granted under this Plan may be exercised more than 10 years from the Date of Grant.
- (a) Each such grant or sale shall constitute an immediate transfer of the ownership of Common Shares to the Participant in consideration of the performance of services, entitling such

Participant to voting, dividend and other ownership rights, but subject to the substantial risk of forfeiture and restrictions on transfer hereinafter referred to.

- (b) Each such grant or sale may be made without additional consideration or in consideration of a payment by such Participant that is less than Market Value per Share at the Date of Grant.
- (c) Each such grant or sale shall provide that the Restricted Shares covered by such grant or sale shall be subject to a "substantial risk of forfeiture" within the meaning of Section 83 of the Code for a period to be determined by the Committee at the Date of Grant and may provide for the earlier lapse of such substantial risk of forfeiture in the event of a Change in Control or other transaction or event; provided, however, that the Restricted Shares covered by any Replacement Award shall be subject to a "substantial risk of forfeiture" for the period provided in the award it replaced, as determined by the Committee.
- (d) Each such grant or sale shall provide that during the period for which such substantial risk of forfeiture is to continue, the transferability of the Restricted Shares shall be prohibited or restricted in the manner and to the extent prescribed by the Board at the Date of Grant (which restrictions may include, without limitation, rights of repurchase or first refusal in the Company or provisions subjecting the Restricted Shares to a continuing substantial risk of forfeiture in the hands of any transferee).
- (e) Any grant of Restricted Shares may specify Management Objectives that, if achieved, will result in termination or early termination of the restrictions applicable to such shares. Each grant may specify in respect of such Management Objectives a minimum acceptable level of achievement and may set forth a formula for determining the number of Restricted Shares on which restrictions will terminate if performance is at or above the minimum level, but falls short of full achievement of the specified Management Objectives.
- (f) Any grant or sale of Restricted Shares may require that any or all dividends or other distributions paid thereon during the period of such restrictions be automatically deferred and reinvested in additional Restricted Shares, which may be subject to the same restrictions as the underlying award.
- (g) Each grant or sale of Restricted Shares shall be evidenced by an agreement executed on behalf of the Company by any officer and delivered to and accepted by the Participant and shall contain such terms and provisions, consistent with this Plan, as the Committee may approve. Unless otherwise directed by the Committee, all certificates representing Restricted Shares shall be held in custody by the Company until all restrictions thereon shall have lapsed, together with a stock power or powers executed by the Participant in whose name such certificates are registered, endorsed in blank and covering such Shares.
- 7. Deferred Shares. The Committee may also authorize the granting or sale of Deferred Shares to Participants. Each such grant or sale may utilize any or all of the authorizations, and shall be subject to all of the requirements contained in the following provisions:

- (a) Each such grant or sale shall constitute the agreement by the Company to deliver Common Shares to the Participant in the future in consideration of the performance of services, but subject to the fulfillment of such conditions during the Deferral Period as the Committee may specify.
- (b) Each such grant or sale may be made without additional consideration or in consideration of a payment by such Participant that is less than the Market Value per Share at the Date of Grant.
- (c) Each such grant or sale shall be subject to a Deferral Period of not less than one year, as determined by the Committee at the Date of Grant, and may provide for the earlier lapse or other modification of such Deferral Period in the event of a Change in Control, or other transaction or event.
- (d) During the Deferral Period, the Participant shall have no right to transfer any rights under his or her award and shall have no rights of ownership in the Deferred Shares and shall have no right to vote them, but the Committee may, at or after the Date of Grant, authorize the payment of dividend equivalents on such Shares on either a current or deferred or contingent basis, either in cash or in additional Common Shares.
- (e) Each grant or sale of Deferred Shares shall be evidenced by an agreement executed on behalf of the Company by any officer and delivered to and accepted by the Participant and shall contain such terms and provisions, consistent with this Plan, as the Committee may approve.
- 8. Awards to Non-Employee Directors. The Committee may, from time to time and upon such terms and conditions as it may determine, authorize the granting to Non-Employee Directors of Option Rights and may also authorize the grant or sale of Restricted Shares to Non-Employee Directors.
- (a) Each grant of Option Rights awarded pursuant to this Section 8 shall be upon terms and conditions consistent with Section 4 of this Plan and shall be evidenced by an agreement in such form as shall be approved by the Committee. Each grant shall specify an Option Price per share, which shall not be less than the Market Value per Share on the Date of Grant. Each such Option Right granted under the Plan shall expire not more than 10 years from the Date of Grant and shall be subject to earlier termination as hereinafter provided. Unless otherwise determined by the Committee, such Option Rights shall be subject to the following additional terms and conditions:
 - (i) Each grant shall specify the number of Common Shares to which it pertains subject to the limitations set forth in Section 3 of this plan.
 - (ii) In the event of the death or disability of the holder of any such Option Rights, each of the then outstanding vested Option Rights of such holder may be exercised at any time within a stated period after such death or disability, as provided by the Committee in the grant, but in no event after the expiration date of the term of such Option Rights.

- (iii) If a Non-Employee Director subsequently becomes an employee of the Company or a Subsidiary while remaining a member of the Board, any Option Rights held under the Plan by such individual at the time of such commencement of employment shall not be affected thereby.
- (iv) Option Rights may be exercised by a Non-Employee Director only upon payment to the Company in full of the Option Price of the Common Shares to be delivered. Such payment shall be made in cash or in Common Shares then owned by the optionee for at least six months, or in a combination of cash and such Common Shares.
- (v) Any grant may provide for deferred payment of the Option Price from the proceeds of sale through a bank or broker on a date satisfactory to the Company of some or all of the shares to which such exercise relates.
- (b) Each grant or sale of Restricted Shares pursuant to this Section 8 shall be upon terms and conditions consistent with Section 6 of this Plan.

9. Transferability.

- -----
- (a) Except as otherwise determined by the Committee, no Option Right, Appreciation Right or other derivative security granted under the Plan shall be transferable by a Participant other than by will or the laws of descent and distribution. Except as otherwise determined by the Committee, Option Rights and Appreciation Rights shall be exercisable during the Optionee's lifetime only by him or her or by his or her guardian or legal representative.
- (b) The Committee may specify at the Date of Grant that part or all of the Common Shares that are (i) to be issued or transferred by the Company upon the exercise of Option Rights or Appreciation Rights, or upon the termination of the Deferral Period applicable to Deferred Shares or (ii) no longer subject to the substantial risk of forfeiture and restrictions on transfer referred to in Section 6 of this Plan, shall be subject to further restrictions on transfer.
- (c) Notwithstanding the provisions of Section 9(a), the Committee may provide that any grant of Option Rights (other than Incentive Stock Options), Appreciation Rights, Restricted Shares, and Deferred Shares shall be transferable by a Participant, without payment of consideration therefor by the transferee, to any one or more members of the Participant's Immediate Family (or to one or more trusts established solely for the benefit of one or more members of the Participant's Immediate Family or to one or more partnerships in which the only partners are members of the Participant's Immediate Family); provided, however, that (i) no such transfer shall be effective unless reasonable prior notice thereof is delivered to the Company and the Committee and such transfer is thereafter effected in accordance with any terms and conditions that shall have been made applicable thereto by the Company or the Committee and (ii) any such transferee shall be subject to the same terms and conditions hereunder as the Participant.
- 10. Adjustments. The Committee may make or provide for such adjustments

in the numbers of Common Shares covered by outstanding Option Rights, Appreciation Rights and Deferred Shares granted hereunder, and in the Option Price and Base Price, and in the kind of shares covered thereby, as the Committee, in its sole discretion, exercised in good faith, may

determine is equitably required to prevent dilution or enlargement of the rights of Participants or Optionees that otherwise would result from (a) any stock dividend, stock split, combination of shares, recapitalization or other change in the capital structure of the Company, or (b) any merger, consolidation, spin-off, split-off, spin-out, split-up, reorganization, partial or complete liquidation or other distribution of assets, issuance of rights or warrants to purchase securities, or (c) any other corporate transaction or event having an effect similar to any of the foregoing. Moreover, in the event of any such transaction or event, the Committee, in its discretion, may provide in substitution for any or all outstanding awards under this Plan such alternative consideration as it, in good faith, may determine to be equitable in the circumstances and may require in connection therewith the surrender of all awards so replaced. The Committee may also make or provide for such adjustments in the numbers of shares specified in Section 3 of this Plan as the Committee in its sole discretion, exercised in good faith, may determine is appropriate to reflect any transaction or event described in this Section 10; provided, however, that any such adjustment to the number specified in Section 3(c)(i) shall be made only if and to the extent that such adjustment would not cause any Option intended to qualify as an Incentive Stock Option to fail so to qualify, and the Committee may take into consideration, as to any award subject to a proposed adjustment, the potential adverse effect thereof under applicable tax or other laws, and may adjust such awards inconsistently as a consequence of those effects.

- 11. Change in Control. For purposes of this Plan, except as may be otherwise prescribed by the Committee in an agreement evidencing a grant or award made under the Plan, a "Change in Control" shall mean if at any time any of the following events shall have occurred:
- (a) Voting Stock Accumulations. The accumulation by any Person of Beneficial Ownership of twenty percent (20%) or more of the combined voting power of the Company's Voting Stock; provided that for purposes of this Section 11(a), a Change in Control will not be deemed to have occurred if the accumulation of twenty percent (20%) or more of the voting power of the Company's Voting Stock results from any acquisition of Voting Stock (i) directly from the Company that is approved by the Incumbent Board, (ii) by the Company, (iii) by any employee benefit plan (or related trust) sponsored or maintained by the Company or any Subsidiary, or (d) by any Person pursuant to a Business Combination that complies with all of the provisions of clauses (i), (ii) and (iii) of Section 11(b); or
- (b) Business Combinations. Consummation of a Business Combination, unless, immediately following that Business Combination, (i) all or substantially all of the Persons who were the beneficial owners of Voting Stock of the Company immediately prior to that Business Combination beneficially own, directly or indirectly, more than sixty-six and two-thirds percent (662/3%) of the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of Directors of the entity resulting from that Business Combination (including, without limitation, an entity that as a result of that transaction owns the Company or all or substantially all of the Company's assets either directly or through one or more subsidiaries) in substantially the same proportions relative to each other as their ownership, immediately prior to that Business Combination, of the Voting Stock of the Company, (ii) no Person (other than the Company, that entity resulting from that Business Combination, or any employee benefit plan (or related trust) sponsored or maintained by the Company, any Eighty Percent (80%) Subsidiary or that entity resulting from that Business Combination) beneficially owns, directly or indirectly, twenty percent (20%) or more of the then

outstanding shares of common stock of the entity resulting from that Business Combination or the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors of that entity, and (iii) at least a majority of the members of the Board of Directors of the entity resulting from that Business Combination were members of the Incumbent Board at the time of the action of the Board of Directors providing for that Business Combination; or

- (c) Sale of Assets. A sale or other disposition of all or substantially all of the assets of the Company; or
- (d) Liquidations or Dissolutions. Approval by the shareholders of the Company of a complete liquidation or dissolution of the Company, except pursuant to a Business Combination that complies with all of the provisions of clauses (i), (ii) and (iii) of Section 11(b).

For purposes of this Section 11, the following definitions will apply:

- (i) "Beneficial Ownership" means beneficial ownership as that term is used in Rule 13d-3 promulgated under the Exchange Act.
- (ii) "Business Combination" means a reorganization, merger or consolidation of the Company.
- (iii)"Eighty Percent (80%) Subsidiary" means an entity in which the Company directly or indirectly beneficially owns eighty percent (80%) or more of the outstanding Voting Stock.
- (iv) "Exchange Act" means the Securities Exchange Act of 1934, including amendments, or successor statutes of similar intent.
- (v) "Incumbent Board" means a Board of Directors at least a majority of whom consist of individuals who either are (a) members of the Company's Board of Directors as of June ___, 2001, or (b) members who become members of the Company's Board of Directors subsequent to June ___, 2001, whose election, or nomination for election by the Company's shareholders, was approved by a vote of at least two-thirds (2/3) of the directors then comprising the Incumbent Board (either by a specific vote or by approval of the proxy statement of the Company in which that person is named as a nominee for director, without objection to that nomination), but excluding, for that purpose, any individual whose initial assumption of office occurs as a result of an actual or threatened election contest (within the meaning of Rule 14a-11 of the Exchange Act) with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board of Directors.
- (vi) "Person" means any individual, entity or group (within the meaning of Section 13(d)(3) or 14 (d)(2) of the Exchange Act).
- (vii) "Voting Stock" means the then outstanding securities of an entity entitled to vote generally in the election of members of that entity's Board.

- 12. Fractional Shares. The Company shall not be required to issue any fractional Common Shares pursuant to this Plan. The Committee may provide for the elimination of fractions or for the settlement of fractions in cash.
 - 13. Withholding Taxes. To the extent that the Company is required to

withhold federal, state, local or foreign taxes in connection with any payment made or benefit realized by a Participant or other person under this Plan, and the amounts available to the Company for such withholding are insufficient, it shall be a condition to the receipt of such payment or the realization of such benefit that the Participant or such other person make arrangements satisfactory to the Company and the Committee for payment of the balance of such taxes required to be withheld, which arrangements (in the discretion of the Committee) may include relinquishment of a portion of such benefit. With the consent of the Committee, a Participant or such other person may also make similar arrangements with respect to the payment of any taxes with respect to which withholding is not required.

14. Foreign Employees. In order to facilitate the making of any grant or

combination of grants under this Plan, the Committee may provide for such special terms for awards to Participants who are foreign nationals or who are employed by the Company or any Subsidiary outside of the United States of America as the Committee may consider necessary or appropriate to accommodate differences in local law, tax policy or custom, which special terms may be contained in an Appendix attached hereto. Moreover, the Committee may approve such supplements to or amendments, restatements or alternative versions of this Plan as it may consider necessary or appropriate for such purposes, without thereby affecting the terms of this Plan as in effect for any other purpose, and the Secretary or other appropriate officer of the Company may certify any such document as having been approved and adopted in the same manner as this Plan. No such special terms, supplements, amendments or restatements, however, shall include any provisions that are inconsistent with the terms of this Plan as then in effect unless this Plan could have been amended to eliminate such inconsistency without further approval by the shareholders of the Company.

- 15. Administration of the Plan.
- (a) This Plan shall be administered by the Committee. A majority of the Committee shall constitute a quorum, and the action of the members of the Committee present at any meeting at which a quorum is present, or acts unanimously approved in writing, shall be the acts of the Committee. Except as the Board may otherwise determine, so long as the Company's outstanding Common Shares are owned by Equifax Inc., all matters relating to Awards under the Plan shall be, and hereby are, delegated to the Compensation and Human Resources Committee of Equifax Inc., provided that all actions taken shall be subject to the approval of the Board.
- (b) The Committee, in its discretion, may delegate to one or more officers of the Company, all or part of the Committee's authority and duties with respect to Participants who are not subject to the reporting and other provisions of Section 16 of the Exchange Act or any successor rule to the same effect. In the event of such delegation, and as to matters encompassed by the delegation, references in the Plan to the Committee shall be interpreted as a reference to the Committee's delegate or delegates. The Committee may revoke or amend the terms of a

delegation at any time but such action shall not invalidate any prior actions of the Committee's delegate or delegates that were consistent with the terms of the Plan

(c) The interpretation and construction by the Committee of any provision of this Plan or of any agreement, notification or document evidencing the grant of Option Rights, Appreciation Rights, Restricted Shares, or Deferred Shares, and any determination by the Committee pursuant to any provision of this Plan or of any such agreement, notification or document shall be final and conclusive. No member of the Committee shall be liable for any such action or determination made in good faith.

16. Amendments, Etc.

- (a) The Committee may at any time and from time to time amend the Plan in whole or in part; provided, however, that any amendment which must be approved by the shareholders of the Company in order to comply with applicable law or the rules of the New York Stock Exchange or, if the Common Shares are not traded on the New York Stock Exchange, the principal national securities exchange upon which the Common Shares are traded or quoted, shall not be effective unless and until such approval has been obtained. Presentation of this Plan or any amendment hereof for shareholder approval shall not be construed to limit the Company's authority to offer similar or dissimilar benefits under other plans without shareholder approval. No amendment shall, without a Participant's consent, adversely affect any rights of any Participant with respect to any award outstanding at the time such amendment is made. No amendment to this Plan shall become effective until shareholder approval is obtained if (i) the amendment increases the aggregate number of Common Shares that may be issued under the Plan, (ii) the amendment changes the class of individuals eligible to become Participants, or (iii) the amendment extends the duration of the Plan.
- (b) The Committee shall not, without the further approval of the shareholders of the Company, authorize the amendment of any outstanding Option Right to reduce the Option Price. Furthermore, no Option Right shall be canceled and replaced with awards having a lower Option Price without further approval of the shareholders of the Company. This Section 16(b) is intended to prohibit the repricing of "underwater" Option Rights and shall not be construed to prohibit the adjustments provided for in Section 10 of this Plan.
- (c) The Committee also may permit Participants to elect to defer the issuance of Common Shares or the settlement of awards in cash under the Plan pursuant to such rules, procedures or programs as it may establish for purposes of this Plan. The Committee also may provide that deferred issuances and settlements include the payment or crediting of dividend equivalents or interest on the deferral amounts.
- (d) The Committee may condition the grant of any award or combination of awards authorized under this Plan on the surrender or deferral by the Participant of his or her right to receive a cash bonus or other compensation otherwise payable by the Company or a Subsidiary to the Participant.
- (e) In case of termination of employment by reason of death, disability or normal or early retirement, or in the case of hardship or other special circumstances, of a Participant who

holds an Option Right or Appreciation Right not immediately exercisable in full, or any Restricted Shares as to which the substantial risk of forfeiture or the prohibition or restriction on transfer has not lapsed, or any Deferred Shares as to which the Deferral Period has not been completed, or who holds Common Shares subject to any transfer restriction imposed pursuant to Section 9(b) of this Plan, the Committee may, in its sole discretion, accelerate the time at which such Option Right or Appreciation Right may be exercised or the time at which such substantial risk of forfeiture or prohibition or restriction on transfer will lapse or the time when such Deferral Period will end or the time when such transfer restriction will terminate or may waive any other limitation or requirement under any such award.

- (f) This Plan shall not confer upon any Participant any right with respect to continuance of employment or other service with the Company or any Subsidiary, nor shall it interfere in any way with the right of the Company or any Subsidiary to terminate such Participant's employment or other service at any time.
- (g) To the extent that any provision of this Plan would prevent any Option Right that was intended to qualify as an Incentive Stock Option from qualifying as such, that provision shall be null and void with respect to such Option Right. Such provision, however, shall remain in effect for other Option Rights and there shall be no further effect on any provision of this Plan.
- 17. Termination. No grant shall be made under this Plan more than 10 _______ years after the date on which this Plan is first approved by the Board of Directors of the Company, but all grants made on or prior to such date shall continue in effect thereafter subject to the terms thereof and of this Plan. The Committee may terminate the Plan at any time provided that such termination shall not adversely affect the rights of any Participant or beneficiary under any Award granted prior to the date of such termination.

CERTEGY INC.

2001 KEY MANAGEMENT LONG-TERM INCENTIVE PLAN

Effective June ___, 2001

ARTICLE I Purpose

The purpose of the plan is to provide long-term incentive compensation to Eligible Executives of Certegy Inc. and/or its subsidiaries who make substantial contributions to the success of their employers, to provide a means for such Eligible Executives to participate in such success, and to assist in attracting and retaining the highest quality individuals in key executive positions. This plan is effective June ___, 2001.

ARTICLE II Definitions

The following words and phrases shall have the respective meanings set forth below (unless the context indicates otherwise).

- 2.1 "Approval of Shareholders" shall mean the affirmative vote of the holders of at least a majority of the shares of common stock of the Company then outstanding.
- 2.2 "Award" shall mean the stated cash amount(s) to which Participants will be entitled upon achievement of goals based on Management Objectives established at the time the Award is granted.
- 2.3 "Committee" shall mean the Compensation and Human Resources Committee of the Company's Board of Directors, as the same from time to time may be constituted, or such other committee as may be appointed by the Board of Directors.
- 2.4 "Common Stock" means the Common Stock, \$.01 par value per share, of the Company.
 - 2.5 "Company" shall mean Certegy Inc.
- 2.6 "Eligible Executive" shall mean elected officers and any other key management personnel of the Company or a subsidiary or division of the Company as determined by the Committee, from time to time, including, without limitation, any officer who is a Director. An Eligible Executive shall not include an officer who is not a full-time employee, even though said officer is a Director, except that a person who was an Eligible Executive and a Director immediately prior to his retirement as an employee of the Company shall continue to be an Eligible Executive so long as he retains his position as an officer and Director.

- 2.7 "Employer" shall mean the Company or the subsidiary or affiliate of the Company by whom the Participant is employed at the time in question.
- 2.8 "Management Objective" shall mean specified levels of, or growth in, one or more of the following criteria:
 - (a) earnings;
 - (b) earnings per share;
 - (c) economic value added;
 - (d) revenue;
 - (e) sales;
 - (f) operating profit;
 - (g) net income;
 - (h) total return to shareholders;
 - (i) market share;
 - (j) profit margin;
 - (k) cash flow/net assets ratio;
 - (1) debt/capital ratio;
 - (m) return on total capital;
 - (n) return on equity;
 - (o) return on assets; and
 - (p) common stock price.

If the Committee makes an Award subject to a particular Management Objective, the Committee shall adopt or confirm a written definition of that Management Objective at the time of the Award. Management Objectives may be described in terms of Company-wide objectives or objectives that are related to a specific division, subsidiary, Employer, department, region, or function in which the Participant is employed. Management Objectives may be made relative to the performance of other corporations.

If the Committee determines that a change in the business, operations, corporate structure or capital structure of the Company, or the manner in which it conducts its business, or other events or circumstances render the Management Objectives unsuitable, the Committee may in its discretion modify such Management Objectives or the related minimum acceptable level of achievement, in whole or in part, as the Committee deems appropriate and equitable, except in the case of a Covered Employee where such action would result in the loss of the otherwise available exemption of the award under Section 162(m) of the Code. In the case of a Covered Employee, in determining financial results, items whose exclusion from consideration will increase the Award shall only have their effects excluded if they constitute "extraordinary" or "unusual" events or items under generally accepted accounting principles and all such events and items shall be excluded. The Committee shall also adjust the performance calculations to exclude the unanticipated effect on financial results of changes in the Code, or other tax laws, and the regulations thereunder. The Committee may decrease the amount of an Award otherwise payable if, in the Committee's view, the financial performance during the performance cycle justifies such adjustment, regardless of the extent to which the Performance Measure was achieved.

- 2.9 "Measurement Period:" Management Objectives may be calculated on the basis of a portion of a year, a single year, cumulatively for a stated number of years, as an average over a stated number of years, or otherwise, as determined by the Committee at the time the Management Objective is established, which shall be the "Measurement Period."
- 2.10 "Participant" means any Eligible Executive to whom an Award has been granted but not yet paid pursuant to this Plan.
- 2.11 "Plan" means this Certegy Inc. 2001 Key Management Long-Term Incentive Plan.

ARTICLE III Eligibility

All Eligible Executives, as determined by the Committee, from time to time, shall be eligible for participation in this Plan.

ARTICLE IV

Selection of Participants, Grant of Awards and Administration of Plan

The Committee shall determine, from time to time, those officers who are to be granted Awards pursuant to Article V below. This Plan shall be administered by the Committee, and the Committee shall (1) construe and interpret the Plan, and (2) make such reasonable rules and regulations for the administration of the Plan as it deems advisable. Any determination by the Committee in administering, interpreting or construing the Plan in accordance with this Article shall be final, binding and conclusive for all purposes and upon all interested persons.

ARTICLE V Grants of Awards, Effective Date and Termination

Subject to the provisions below, the maximum Award granted to any Participant in any fiscal year of the Company shall not exceed \$5,000,000. Subject to the approval of Equifax Inc. as the shareholder of the Company, this Plan shall become effective June ___, 2001. No Awards may be granted under this Plan after the 10th anniversary of the date the Plan is approved by the Board of Directors.

6.1 Subject to the provisions of Article V, the Participant shall be entitled to receive the cash to which his Award entitles him as soon as practical after the end of the Measurement Period with respect to that Award; provided, however, that:

- (a) Each Award granted under the Plan shall be forfeited and canceled in all respects, and no cash shall be delivered or paid to the Participant thereof, in the event that:
 - (i) The employment of the Participant by the Employer is terminated, either voluntarily or involuntarily, by the Employer or the Participant, for any reason whatsoever (subject to the provisions of Article VII hereof) prior to the end of the Measurement Period for that Award;
 - (ii) The employment status of the Participant has changed prior to the end of the Measurement Period for that Award so that the Participant is no longer an Eligible Executive; or
- (iii) The Management Objective for the Measurement Period for such Award is less than the minimum stated in the Award.
- (b) A portion, or all, of each Award shall be forfeited and canceled in all respects, and no cash shall be delivered or paid with respect to the portion of such Award so forfeited and canceled, in the event that the aggregate Management Objective for the Measurement Period with respect to the Award is not at least equal to a minimum stated in the Award.
- (c) The Committee shall establish, for each Measurement Period, the goals based on one or more Management Objectives. These goals will be established on or before the date any Award relating to said Measurement Period is granted. The goals will be established with consideration given to the economic conditions existing at the time said goals are established. A portion, or all, of each Award shall be forfeited and canceled in all respects, and no cash shall be delivered or paid with respect to the portion of such Award so forfeited and canceled, in the event that the goals established for the Measurement Period are not achieved, all as prescribed by the Committee. The Committee shall deliver to each Participant written notice of the goals established for the Measurement Period to which said Award relates, along with the forfeiture provisions relating to said Award. Even though performance goals established for each Measurement Period are met or exceeded, the Committee shall have the discretion, as to each Participant, to reduce the amount of an Award that would otherwise be paid or to determine that no portion should be paid. The Committee may not increase the amount of an Award that would otherwise be paid.
- (d) Nothing contained in this Article VI or elsewhere in this Plan shall eliminate, impair or otherwise affect the right of the Employer to terminate or change the employment of any Eligible Executive at any time, and the grant of an Award to any such Eligible Executive shall not be deemed to, and shall not, result in any agreement, expressed or implied, by the Employer to retain such person in any specific position or in its employ for the duration of the Measurement Period with respect to such Award or for any other period.

(e) Subject to the provisions of this paragraph, the terms of an Award may provide, if the Committee so directs in each instance, that each Participant may elect, by delivering written notice of such election to the Secretary of the Company during the period defined below, to surrender his or her right to receive up to the full value of the Award that would otherwise be paid to the Participant at the end of the Measurement Period, in exchange for the right to receive an equity interest as described below. In order to be effective, such written notice of election must be delivered to the Secretary of the Company during a period beginning on the third business day following release for publication (in the manner hereinafter set forth) of the Company's quarterly statements of sales and earnings for the final fiscal quarter ending within the Measurement Period and ending on the twelfth business day following said release for publication. Any such election shall be subject to the right of the Committee to disapprove the same, in whole or in part, at any time after such election but prior to the issuance of cash with respect to the particular Award in accordance with the provisions of this Plan. In the event of the death, disability or retirement of a Participant, at any time during the Measurement Period to which an Award relates, the Award shall be distributed as provided in Article VII hereof regardless of any election made by such Participant. The release for publication of the Company's quarterly statements as referred to in the second sentence of this paragraph shall be deemed to have been made at the time such data appears (i) on a wire service, (ii) in a financial news service, (iii) in a newspaper of general circulation or (iv) is otherwise made publicly available. For purposes of this paragraph, the determination of the appropriate equity interest into which the cash award is converted shall be made based on rules adopted by the Committee and uniformly applied, and said rules shall be adopted prior to or at the time of the grant of the Award in question, and the aggregate value of the cash portion and the value of the equity interest for any individual, determined at the date of grant, shall not exceed the maximum referred to in Article V. The equity interest may be an option for purchase of Common Stock, restricted shares of Common Stock, or any other equity interest determined by the Committee. The equity interest may be issued by the Committee on its own action or pursuant to the Company's 2001 Stock Incentive Plan or any such other stock incentive plan as the Committee deems appropriate.

ARTICLE VII Death, Disability or Retirement of Eligible Executive or Change in Control of the Company

7.1 In the event of the termination of employment with the Employer during any Measurement Period of any Participant by reason of the death or disability or retirement of such Participant, the Committee may, but shall not be obligated to, waive the continuation of the employment requirement set forth in section 6.1(a)(i) above. In the event that such requirement is waived, such Participant or his estate, as the case may be, will be entitled to receive an Award in cash equivalent to a pro rata portion of the amount which said Participant would have received, if the employment of such Participant had continued through the Measurement Period for such Award. For purposes of Article VI and this Article VII, a Participant shall not be deemed to have terminated his employment although he retires from said employment, if he continues to serve as

an elected officer of the Company or a subsidiary of the Company and to serve as a Director of the Company; said Eligible Executive shall be deemed to have terminated his employment when his term of office expires and he is not reelected thereto, or when he is removed or resigns from office, if earlier.

- 7.2 This pro rata portion shall be computed as follows: The cash Award which would have been earned based on the level of actual achievement of the Management Objective at the end of the Measurement Period will be multiplied by a fraction, the numerator of which shall be the number of full calendar months during the Measurement Period prior to the Participant's death, disability or retirement, and the denominator of which shall be the number of full calendar months contained in the complete Measurement Period.
- 7.3 In the event of the termination of employment with the Employer of any Participant after completing a Measurement Period, but before distribution of his Award is made, such Participant or his estate, as the case may be, will be entitled to receive the Award to the same extent, in the same manner and at the same time as if the employment of such Eligible Executive had not terminated, except that if the Participant has directly or indirectly engaged in any activity that is harmful to the Company or the Employer, as determined by the Committee in its sole discretion (including without limitation the disclosure or misuse of any confidential information or trade secrets of the Company or the Employer), then Participant shall forfeit any entitlement to such Award.
- 7.4 If there is a "change in control of the Company," as hereinafter defined, during any Measurement Period, then, notwithstanding any other provision of this Plan to the contrary, any Participant holding any Award shall be irrevocably entitled to receive an amount in cash which is equal to (i) the target award if the change in control occurs during the first measurement year, or (ii) 150% of the target award if the change in control occurs after said first year (but no less than the projected payout determined on the effective date of the change in control if the change in control occurs during the last three months of the Measurement Period). Such payment will be made within sixty (60) days following the change in control of the Company.
- 7.5 For purposes of this Article VII, a "change in control of the Company" shall be deemed to have occurred upon the occurrence of any of the following events:
 - (a) Voting Stock Accumulations. The accumulation by any Person of Beneficial Ownership of twenty percent (20%) or more of the combined voting power of the Company's Voting Stock; provided that for purposes of this Section 7.5, a Change in Control will not be deemed to have occurred if the accumulation of twenty percent (20%) or more of the voting power of the Company's Voting Stock results from any acquisition of Voting Stock (i) directly from the Company that is approved by the Incumbent Board, (ii) by the Company, (iii) by any employee benefit plan (or related trust) sponsored or maintained by the Company or any Subsidiary, or (iv) by any Person pursuant to a Business Combination that complies with all of the provisions of clauses (i), (ii) and (iii) of subparagraph 7.5(b); or
 - (b) Business Combinations. Consummation of a Business Combination, unless, immediately following that Business Combination, (i) all or substantially all of

the Persons who were the beneficial owners of Voting Stock of the Company immediately prior to that Business Combination beneficially own, directly or indirectly, more than sixty-six and two-thirds percent (66 2/3%) of the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of Directors of the entity resulting from that Business Combination (including, without limitation, an entity that as a result of that transaction owns the Company or all or substantially all of the Company's assets either directly or through one or more subsidiaries) in substantially the same proportions relative to each other as their ownership, immediately prior to that Business Combination, of the Voting Stock of the Company, (ii) no Person (other than the Company, that entity resulting from that Business Combination, or any employee benefit plan (or related trust) sponsored or maintained by the Company, any Eighty Percent (80%) Subsidiary or that entity resulting from that Business Combination) beneficially owns, directly or indirectly, twenty percent (20%) or more of the then outstanding shares of common stock of the entity resulting from that Business Combination or the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors of that entity, and (iii) at least a majority of the members of the Board of Directors of the entity resulting from that Business Combination were members of the Incumbent Board at the time of the of the action of the Board of Directors providing for that Business Combination;

- (c) Sale of Assets. A sale or other disposition of all or substantially all of the assets of the Company; or
- (d) Liquidations or Dissolutions. Approval by the shareholders of the Company of a complete liquidation or dissolution of the Company, except pursuant to a Business Combination that complies with all of the provisions of clauses (i), (ii) and (iii) of section 7.5(b)
- 7.6 For purposes of this Article VII, the following definitions will apply:
- (a) "Beneficial Ownership" means beneficial ownership as that term is used in Rule 13d-3 promulgated under the Exchange Act.
- (c) "Disability" means permanently and totally disabled as defined in Code Section 22(e)(3).
- (d) "Eighty Percent (80%) Subsidiary" means an entity in which the Company directly or indirectly beneficially owns eighty percent (80%) or more of the outstanding Voting Stock.
- (e) "Exchange Act" means the Securities Exchange Act of 1934, including amendments, or successor statutes of similar intent.

- (f) "Incumbent Board" means a Board of Directors at least a majority of whom consist of individuals who either are (a) members of the Company's Board of Directors as of the effective date of this Plan, or (b) members who become members of the Company's Board of Directors subsequent to such date, whose election, or nomination for election by the Company's shareholders, was approved by a vote of at least two-thirds (2/3) of the directors then comprising the Incumbent Board (either by a specific vote or by approval of the proxy statement of the Company in which that person is named as a nominee for director, without objection to that nomination), but excluding, for that purpose, any individual whose initial assumption of office occurs as a result of an actual or threatened election contest (within the meaning of Rule 14a-11 of the Exchange Act) with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board of Directors.
- (g) "Person" means any individual, entity or group (within the meaning of Section 13(d)(3) or 14 (d)(2) of the Exchange Act).
- (h) "Voting Stock" means the then outstanding securities of an entity entitled to vote generally in the election of members of that entity's Board.

ARTICLE VIII Nonalienation of Benefits

Neither the Award nor any other right or benefit under this Plan shall be subject to anticipation, alienation, sale, assignment, pledge, encumbrance or charge, and any attempt to anticipate, alienate, sell, assign, pledge, encumber or charge the same shall be void and shall not be recognized or given effect by the Company.

ARTICLE IX Certificates of Award

The Company shall execute and deliver to each Participant to whom an Award is granted a certificate, in the form prescribed by the Committee, evidencing such Award and stating the date thereof and cash amount that is the subject of the Award.

The Board of Directors of the Company may amend, suspend or terminate this Plan in whole or in part at any time; provided that, except as expressly provided in this Plan, no such amendment, suspension or termination shall adversely affect the rights of the holders of any Award then outstanding.

CERTEGY INC.

NON-EMPLOYEE DIRECTOR STOCK OPTION PLAN

EFFECTIVE JUNE ____, 2001

ARTICLE I. PURPOSE

The purpose of this Plan is to promote the interest of Certegy Inc. and its Subsidiaries by granting Options to Non-Employee Directors in order:

- (1) to attract and retain Non-Employee Directors,
- (2) to provide Non-Employee Directors with long-term financial incentives to increase the value of Equifax, and
- (3) to provide each Non-Employee Director with a stake in the future of Certegy Inc. which corresponds to the stake of each of the Company's shareowners.

ARTICLE II. DEFINITIONS AND GENDER AND NUMBER

- 2.1 Definitions. Each term set forth in this Article II shall have the respective meaning Set forth opposite such term for purposes of this Plan, and when the defined meaning is intended the term is capitalized.
 - (a) "Agreement" means a written agreement, substantially in the form attached hereto as Exhibit A, which sets forth the Option Price and other terms and conditions with respect to an Option granted to a Non-Employee Director under this Plan.
 - (b) "Board" means the Board of Directors of Certegy Inc.
 - (c) "Code" means the Internal Revenue Code of 1986, as amended from time to time.
 - (d) "Company" means Certegy Inc., a Georgia corporation.
 - (e) "Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time.
 - (f) "Fair Market Value" means, on any given date, the closing price of a share of Stock as reported on the New York Stock Exchange composite tape on such day or, if the Stock was not traded on the New York Stock Exchange on such day, then on the next preceding day that the share of Stock was traded on such exchange.

- (g) "Non-Employee Director" means a member of the Board who is not an officer or employee of the Company or its affiliates.
- (h) "Option" means an option granted under this Plan to purchase Stock, which shall constitute a nonqualified or nonstatutory stock option and not an incentive stock option under Code Section 422.
- (i) "Option Price" means the price (determined in accordance with Section 6.2) which shall be paid to purchase one share of Stock upon the exercise of an Option granted under this Plan.
- (j) "Plan" means this Certegy Inc. Non-Employee Director Stock Option Plan as effective June ____, 2001, and as thereafter amended from time to time.
 - (k) "Stock" means the \$.01 par value common stock of the Company.
- 2.2 Gender and Number. Unless the context clearly requires otherwise, the masculine pronoun whenever used shall include the feminine and neuter pronouns, the singular shall include the plural and the plural shall include the singular.

ARTICLE III. SHARES SUBJECT TO OPTIONS

The aggregate number of shares of Stock with respect to which the grant of Options (collectively referred to as "Grants" in this Article III) may be made shall not exceed 200,000 shares of Stock (as adjusted in accordance with Article X whenever such an adjustment is called for). Any shares of Stock subject to a Grant after the exchange, cancellation, forfeiture or expiration of such Grant thereafter shall again become available for use under this Article III as if such shares of Stock had never been subject to a Grant.

ARTICLE IV. EFFECTIVE DATE

The effective date of this Plan shall be the date it was approved by the Board, June ____, 2001. The Plan was approved by Equifax Inc., as the Company's sole shareholder, on June ____, 2001.

ARTICLE V. ELIGIBILITY

Only Non-Employee Directors shall be eligible for the grant of Options under this Plan.

ARTICLE VI. OPTIONS

capacity on the day following the date of the distribution of the Company's common stock to the shareholders of Equifax Inc. shall be granted an option to purchase 2,000 shares of Stock. Each

6.1 Initial Grant of Options. Each Non-Employee Director serving in such

person who first becomes a Non-Employee Director after such date shall be granted an option to purchase 2,000 shares of Stock on the date that he or she became a Non-Employee Director.

- 6.2 Grant of Options. Subject to the terms and conditions of this Plan, on
- the day following each annual meeting of the shareholders of the Company occurring after the effective date of this Plan, each Non-Employee Director serving on such date shall receive an Option to purchase 2,000 shares of Stock. Each grant of an Option shall be evidenced by and subject to an Agreement.
- 6.3 Option Price; Form of Payment. The Option Price for each share of

 Stock subject to an Option shall be the greater of (i) the par value of a share of Stock, or (ii) the Fair Market Value of a share of Stock on the date the Option is granted. Payment of the Option Price may be made in any one or more of the following ways: (a) in cash, or (b) in cash by a broker-dealer acceptable to the Company to whom the Non-Employee Director has submitted an irrevocable notice of exercise, or (c) by delivery to the Company of previously-owned shares of Stock which the Non-Employee Director has held for at least six months prior to the date of exercise, and which have a Fair Market Value as of the date of
- 6.4 Option Period. Each Option granted under this Plan shall be

 exercisable at such time or times as set forth in the Agreement, and each Option shall expire automatically on the earliest of (i) the date such Option is exercised in full, or (ii) the date such Option expires in accordance with the Agreement.

ARTICLE VII. NONTRANSFERABILITY

exercise.

No Option granted under this Plan shall be transferable by a Non-Employee Director other than by will, by the applicable laws of descent and distribution (including such beneficiary designations as may be made in accordance with the Agreement) or pursuant to a qualified domestic relations order as defined by the Code, and such Option shall be exercisable during a Non-Employee Director's lifetime only by the Non-Employee Director or such qualified transferee. Any such qualified transferee shall be treated as the Non-Employee Director only to the extent that the Non-Employee Director's rights under such Option are properly transferred to such person as provided above.

ARTICLE VIII. STOCK RESTRICTIONS

The Company shall have the right under this Plan to restrict or otherwise delay the issuance of any shares of Stock purchased or paid under this Plan until the requirements of any applicable laws or regulations and any stock exchange requirements have been in the Company's judgment satisfied in full. Furthermore, any shares of Stock which are issued as a result of purchases or payments made under this Plan shall be issued subject to such restrictions and conditions on resale and any other disposition as the Company shall deem necessary or desirable under any applicable laws or regulations or in light of any stock exchange requirements.

ARTICLE IX. LIFE OF PLAN

This Plan shall terminate automatically on the second day following our 2010 annual meeting of shareholders. At any time prior to such automatic termination date, the Board may terminate the Plan. No Option shall be granted under this Plan after the date this Plan terminates. However, with respect to any Options which are outstanding on the Plan's termination date, the applicable terms of the Plan and the Agreement shall survive the termination until such Options have been exercised in full, forfeited in full or otherwise completely expired.

ARTICLE X. ADJUSTMENT

The number of shares of Stock subject to Options granted under this Plan (and the related Option Prices) shall be administratively adjusted (in a manner which does not constitute a "modification," "extension" or "renewal" as those terms are used under Code Section 424(h)) to reflect any change in the capitalization of the Company, including, but not limited to, such changes as stock dividends or stock splits. Furthermore, the number of shares of Stock under Article III of this Plan and the number of shares subject to Options granted under this Plan (and the related Option Prices) shall be administratively adjusted (in a manner which satisfies the requirements of Code Section 424(a)) in the event of any corporate transaction described in Code Section 424(a) which provides for the substitution or assumption of such Options. If any adjustment under this Article X would create a fractional share of stock or a right to acquire a fractional share of Stock, such fractional share shall be disregarded and the number of shares of Stock reserved under this Plan and the number of shares of stock subject to any Options granted under this Plan shall be the next lower whole number of shares of Stock, rounding all fractions downward. Any adjustment made under this Article X shall be conclusive and binding on all affected persons.

ARTICLE XI. SALE OR MERGER OF EQUIFAX

If the Company agrees to sell substantially all of its assets for cash or property or for a combination of cash and property or agrees to any merger, consolidation, reorganization, division or other corporate transaction in which Stock is converted into another security or into the right to receive securities or property and such agreement does not provide for the assumption or substitution of the Options granted under this Plan, each outstanding Option shall be cancelled in exchange for the same consideration each Non-Employee Director otherwise would receive as a shareholder of the Company in connection with such sale or other corporate transaction if he had the right to exercise his Option in full under this Plan for shares of Stock immediately before such sale or other transaction and he exercised that right. The number of shares of Stock subject to an Option which each Non-Employee Director shall be deemed to have a right to receive upon such exercise shall be determined by dividing the excess of the Fair Market Value of the shares of Stock subject to his Option immediately before such sale or other corporate transaction over the Option Price for such shares by the Fair Market Value of a share of Stock immediately before the consummation of such sale or other corporate transaction. If any calculation under this Article XI results in a fractional share of Stock, such fractional share shall be paid in cash.

12.1 General. Amendments with respect to this Plan shall be accomplished

pursuant to authority and procedures established and in effect from time to time through resolutions adopted by the Board; provided, however, that this Plan may not be amended more than once every six months, other than to comport with changes in the Code, the Employee Retirement Income Security Act of 1974, or the rules thereunder; provided further, this Plan may not be amended with respect to the number of shares subject to an Option granted to a Non-Employee Director, Option Price or method for determining Fair Market Value of shares of Stock, and the timing of awards. Amendments to this Plan may be made by the Board without the approval of the shareholders of the Company.

12.2 Acceleration. Notwithstanding any other provision of this Plan, the

Board may, in its discretion, accelerate the time at which any Option may be exercised, but only in the event of the death, retirement or disability of a Non-Employee Director or a change in control of the Company. For purposes of this Plan, "retirement" means termination of service as a director on the Board after age 55 and completion of five years of service as a director. "Disability" means permanently and totally disabled as defined in Code Section 22(e)(3). A "change in control of the Company" means the occurrence of any of the following events:

(a) Voting Stock Accumulations. The accumulation by any Person of

Beneficial Ownership of twenty percent (20%) or more of the combined voting power of the Company's Voting Stock; provided that for purposes of this Section, a Change in Control will not be deemed to have occurred if the accumulation of twenty percent (20%) or more of the voting power of the Company's Voting Stock results from any acquisition of Voting Stock (1) directly from the Company that is approved by the Incumbent Board, (2) by the Company, (3) by any employee benefit plan (or related trust) sponsored or maintained by the Company or any Subsidiary, or (4) by any Person pursuant to a Business Combination that complies with all of the provisions of the clauses (1) (2) and (3) of subparagraph (b) below; or

(b) Business Combinations. Consummation of a Business Combination,

unless, immediately following that Business Combination, (1) all or substantially all of the Persons who were the beneficial owners of Voting Stock of the Company immediately prior to that Business Combination beneficially own, directly or indirectly, more than sixty-six and twothirds percent (66 2/3%) of the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of Directors of the entity resulting from that Business Combination (including, without limitation, an entity that as a result of that transaction owns the Company or all or substantially all of the Company's assets either directly or through one or more subsidiaries) in substantially the same proportions relative to each other as their ownership, immediately prior to that Business Combination, of the Voting Stock of the Company, (2) no Person (other than the Company, that entity resulting from that Business Combination, or any employee benefit plan (or related trust) sponsored or maintained by the Company, any Eighty Percent (80%) Subsidiary or that

entity resulting from that Business Combination) beneficially owns, directly or indirectly, twenty percent (20%) or more of the then outstanding shares of common stock of the entity resulting from that Business Combination or the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors of that entity, and (3) at least a majority of the members of the Board of Directors of the entity resulting from that Business Combination were members of the Incumbent Board at the time of the execution of the initial agreement or of the action of the Board providing for that Business Combination; or

- (c) Sale of Assets. A sale or other disposition of all or substantially all of the assets of the Company; or
 - (d) Liquidations or Dissolutions. Approval by the shareholders of the

Company of a complete liquidation or dissolution of the Company, except pursuant to a Business Combination that complies with all of the clauses (1) (2) and (3) of subparagraph (b) above.

For purposes of this Section, the following definitions will apply:

- (i) "Beneficial Ownership" means beneficial ownership as that term is used in Rule 13d-3 promulgated under the Exchange Act.
- (ii) "Business Combination" means a reorganization, merger or consolidation of the Company.
- (iii) "Eighty Percent (80%) Subsidiary" means an entity in which the Company directly or indirectly beneficially owns eighty percent (80%) or more of the outstanding Voting Stock.
- (iv) "Exchange Act" means the Securities Exchange Act of 1934, including amendments, or successor statutes of similar intent.
- (v) "Incumbent Board" means a Board of Directors at least a majority of whom consist of individuals who either are (a) members of the Company's Board of Directors as of the date of Grant, or (b) members who become members of the Company's Board of Directors subsequent to the date of Grant whose election, or nomination for election by the Company's shareholders, was approved by a vote of at least two-thirds (2/3) of the directors then comprising the Incumbent Board (either by a specific vote or by approval of the proxy statement of the Company in which that person is named as a nominee for director, without objection to that nomination), but excluding, for that purpose, any individual whose initial assumption of office occurs as a result of an actual or threatened election contest (within the meaning of Rule 14a-11 of the Exchange Act) with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board of Directors.
- (vi) "Person" means any individual, entity or group (within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act).

(vii) "Voting Stock" means the then outstanding securities of an entity entitled to vote generally in the election of members of that entity's Board of Directors.

ARTICLE XIII. MISCELLANEOUS

- 13.3 Income Tax Withholding. To satisfy any tax withholding payments that become necessary as the result of an exercise of an option under this Plan, the Agreement may provide that the Non-Employee Director will be entitled to satisfy such withholding payments in any one or more of the following ways: (a) by cash payment to the Company, or (b) by cash payment by a broker-dealer acceptable to the Company to whom the Non-Employee Director has submitted an irrevocable notice of exercise, or (c) by delivery to the Company of previously-owned shares of Stock which the Non-Employee Director has held for at least six months prior to the date of delivery, and which have a Fair Market Value as of the date on which the withholding obligation arose that is not less than the amount of the withholding obligation, or (d) by authorizing the Company to withhold shares of Stock which would otherwise be delivered to the Non-Employee Director upon exercise of the option, and which have a Fair Market Value as of the date on which the withholding obligation arose that is not less than the amount of the withholding obligation.
- 13.4 No Shareholder Rights. No Non-Employee Director will have any rights as a shareholder of the Company as a result of the grant of an Option to him under this Plan or his exercise of such Option pending the actual delivery of Stock to him as a result of such exercise, and his rights upon such delivery shall be prospective only.

CERTEGY INC.

NON-EMPLOYEE DIRECTOR STOCK OPTION AGREEMENT

Number	of Shares:	
Option	Price:	\$
Option	Grant Date:	
Expirat	tion Date:	

CERTEGY INC. (the "Company"), a Georgia corporation, in accordance with the Certegy Inc. Non-Employee Director Stock Option Plan, as may be amended from time to time (the "Plan"), hereby grants a Nonqualified Stock Option ("Option") to the Optionee named above to purchase from the Company the above stated number of shares of the Company common stock, \$.01 par value ("Stock"), at an option price per share ("Option Price") as stated above. This Option is subject to the terms and conditions contained in this Agreement and to the further terms and conditions set forth in the Plan. This Option is granted effective as of the Option Grant Date stated above and shall expire on the Expiration Date stated above, subject to any earlier exchange, cancellation, forfeiture, expiration or extension under Section 3 of this Agreement or the provisions of the Plan.

1. General. The terms and conditions of this Agreement constitute a part of -----the Certegy Inc. Non-Employee Director Stock Option Plan and apply to each Option granted under the Plan.

2. Date Exercisable.

- (a) An Option shall first become exercisable as to one hundred percent (100%) of the shares of Stock subject to this Option on the first anniversary of the Grant Date; provided, however, that an Option shall become immediately exercisable as to all of the shares of Stock subject to the Option upon the occurrence of one of the events described in Section 2(d) or clauses (i), (ii) or (iii) of Section 3(a).
- (b) Once it has become exercisable, an Option will continue to be exercisable at any time in whole or in part (but if in part, in an amount equal to at least 100 shares or, if less, the number of shares remaining to be exercised under the Option) on any business day of the Company before the date such Option expires under Section 3 of this Agreement.
- (c) If Optionee is subject to minimum stock ownership guidelines as may be in effect form time to time, and if upon Optionee's election to exercise his/her stock option(s) Optionee has not satisfied the stock ownership guidelines then in effect, then Optionee will be precluded from using the Company's cashless exercise program. In that

event, any exercisable Options may be exercised through the payment of cash or cash equivalent or by tendering previously owned shares of Company stock.

- (d) If a Change in Control of the Company occurs while Optionee is serving on the Board, then the Option will become immediately exercisable with respect to that portion of the Number of Shares with respect to which the Option has not yet been exercised or is not yet exercisable (the "Unexercised Portion").
- 3. Expiration. An Option shall expire and, except as otherwise noted,
 ----Optionee shall have no further rights under this Agreement, upon the earlier of:
 - (a) subject to extension under Section 3(c), the last day of the 36-month period which begins on the date when Optionee terminates his service on the Board by reason of (i) death, (i) disability, or (iii) retirement, which shall mean termination of service on the Board after the Optionee has reached age 55 and completed at least five years of service as a director on the Board;
 - (b) the date on which Optionee (i) resigns from or is not re-elected to the Board prior to being eligible for retirement under clause (iii) of Section 3(a); (ii) resigns for the purpose of accepting, or retires and subsequently accepts, a directorship or employment, or becomes associated with, employed by or renders service to, or owns an interest in (other than as a shareholder with a less than 5% interest in a publicly traded company) any business that is competitive with the Company of any of its affiliates or with any other business in which the Company or any of its affiliates have a substantial direct or indirect interest; or (iii) resigns as a result of an interest or affiliation that would prohibit continued service as a director;
 - (c) if Optionee terminates service on the Board under the conditions described under Section 3(a) and at his death the Option is exercisable as to any number of shares of Stock, the last day of the 6-month period which begins on the date of Optionee's death, notwithstanding any earlier expiration of the Option that may otherwise be provided under this Agreement; or
 - (d) if Optionee's service on the Board terminates after the Date on which a Change in Control occurs, other than as a result of any events described in clauses (ii) or (iii) of Section 3(b) above, then Optionee may exercise the Unexercised Portion until the last day of the sixty (60) month period following the termination of Optionee's service on the Board, notwithstanding any earlier expiration of the Option that may otherwise be provided under this Agreement;
 - (e) the date the Option has been exercised in full; or
 - (f) subject to extension under Sections 3(c) and 3(d), one day after the expiration of the 5-year period which begins on the Option Grant Date.
- 4. Method of Exercise. An Option may be exercised by properly completing and actually delivering the applicable Notice of Exercise form to the Company, together with payment in full of the Option Price for the shares of Stock the Optionee desires to purchase

through such exercise. Payment of the Option Price may be made in any one or more of the following ways: (a) in cash, or (b) in cash by a broker-dealer acceptable to the Company to whom the Optionee has submitted an irrevocable notice of exercise, or (c) by delivery to the Company of previously-owned shares of Stock which the Optionee has held for at least six months prior to the date of exercise, and which have a Fair Market Value as of the date of exercise that is not less than the Option Price.

- 5. Effective Date of Exercise. An exercise shall be effective on the date a properly completed Notice of Exercise form, together with payment of the Option Price, actually is delivered to and accepted by the Executive Compensation Department at the Company headquarters.
- 6. Nontransferable. No rights granted under this Agreement shall be transferable by Optionee during Optionee's lifetime, and such rights shall be exercisable during Optionee's lifetime only by Optionee, except that Optionee's rights under this Agreement may be transferred by will or by the laws of descent and distribution or pursuant to a qualified domestic relations order as defined by the Code.
- 7. Stockholder Status. Optionee shall have no rights as a stockholder with respect to any shares of Stock under an Option before the date such shares have been duly issued to Optionee and no adjustment shall be made for dividends of any kind or description whatsoever or for distributions of other rights of any kind or description whatsoever respecting such Stock except as expressly set forth in the Plan.
- 8. Other Laws. The Company shall have the right to refuse to issue or transfer any shares of Stock under an Option if the Company, acting in its absolute discretion, determines that the issuance of transfer of such shares might violate any applicable law or regulation or cause any violation under Section 16(b) of the Securities Exchange Act of 1934, and any payment tendered in such event to exercise this Option shall be promptly refunded to Optionee.
- 9. Exercise Restrictions. The Company shall have the right to restrict or otherwise delay the issuance of any shares of Stock purchased or paid for under this Agreement until the requirements of any applicable laws or regulations and any stock exchange requirements have been in the Company's judgment satisfied in full. Furthermore, any shares of Stock which are issued as result of purchases or payments made under this Agreement shall be issued subject to such restrictions and conditions on resale and on any other transfer or disposition as the Company shall deem necessary or desirable under any applicable laws or regulations or in light of any stock exchange requirements.
- 10. Taxes. If any tax withholding payments become necessary as the result of an exercise of an option under this Agreement, Optionee may satisfy these withholding payments by doing any one or more of the following: (a) by making cash payment to the Company, or (b) by making cash payment by a broker-dealer acceptable to the Company to whom the Optionee has submitted an irrevocable notice of exercise, or (c) by delivering to the Company previously-owned shares of Stock which the Optionee has held for at least six months prior to the date of delivery, and which have a Fair Market Value as of the date on which the withholding obligation arose that is not less than the amount of the withholding obligation, or (d) by authorizing the

Company to withhold shares of Stock which would otherwise be delivered to the Optionee upon exercise of the option, and which have a Fair Market Value as of the date on which the withholding obligation arose that is not less than the amount of the withholding obligation.

11. Amendment. Optionee's rights under this Agreement can be modified,

suspended or canceled in accordance with the terms of the Plan. However, the provisions of the Plan and the Agreement may not be amended more than once every six months, other than to comport with changes in the Code, the Employee Retirement Income Security Act of 1974, or the rules thereunder. This Agreement may not be amended with respect to the amount, Option Price or method for determining Fair Market Value of shares of Stock, and the timing of awards.

12. Miscellaneous.

- (a) This Agreement shall be subject to the provisions, definitions, terms and conditions set forth in the Plan, all of which are incorporated by this reference, and unless defined in this Agreement, any capitalized terms shall have the same meaning assigned to those terms under the Plan.
- (b) The Plan and this Agreement shall be governed by and construed under the laws of the State of Georgia.
- (c) The exercise of this Option shall not be effected by the exercise or non-exercise of any other option that may be granted under any other plan or arrangement.

IN WITNESS WHEREOF, the Company has caused this Agreement to be signed by a duly authorized officer, and the Optionee has affixed his signature below.

CERTEGY INC.
Ву:
Name: Title:
OPTIONEE:

CERTEGY INC.
DEFERRED COMPENSATION PLAN
Effective as of June ____, 2001

CERTEGY INC. DEFERRED COMPENSATION PLAN Effective as of June ____, 2001

TABLE OF CONTENTS

Article I.	Establishment and Purpose	1
1.1 1.2 1.3	Establishment of Plan	1 1 1
Article II.	Definitions	1
2.1 2.2 2.3 2.4 2.5 2.6 2.7 2.8 2.9 2.10 2.11 2.12 2.13 2.14 2.15 2.16 2.17 2.18 2.19 2.20 2.21 2.22	"Account". "Affiliate". "Beneficiary". "Board". "Code". "Committee". "Company". "Director Fees". "Distribution Date". "Employee". "Employee". "Employer". "Entry Date". "Equifax Plan". "Financial Hardship". "Investment Fund". "Non-Employee Director". "Participant". "Plan". "Plan Year". "Termination of Service". "Transferred Benefits". "Valuation Date".	1 1 2 2 2 2 2 2 2 2 2 2 2 3 3 3 3 3 3 3
Article III.	Eligibility and Participation	3
3.1 3.2	EligibilityParticipation	3 4
Article IV.	Contributions	4
4.1	Deferrals	4
Article V.	Participants' Accounts	5

CERTEGY INC. DEFERRED COMPENSATION PLAN Effective as of June ____, 2001

TABLE OF CONTENTS

(Continued)

5.1 5.2 5.3 5.4 5.5	Investment of Accounts. Valuation of Accounts. Financing. Unsecured Interest. Nontransferability.	6
Article VI.	Payment of Accounts	7
6.1 6.2 6.3	Payments to Participant	7
Article VII.	Administration	ç
7.1 7.2 7.3 7.4	Administration. Appeals from Denial of Claims. Tax Withholding.	10 11
Article VIII.	Adoption of the Plan by Affiliate	11
8.1 8.2	Adoption of the Plan by Affiliate	
Article IX.	Miscellaneous Provisions	11
9.1 9.2 9.3	No Contract of Employment.	11

CERTEGY INC. DEFERRED COMPENSATION PLAN

Effective as of June ___, 2001

Article I. Establishment and Purpose

1.1 Establishment of Plan. Certegy Inc. (the "Company") does hereby adopt
----the Certegy Inc. Deferred Compensation Plan (the "Plan") effective as of the
date it was approved by the Company's Board of Directors, June ___, 2001.

The Plan is an unfunded plan of deferred compensation for a select group of management or highly compensated employees. The Plan, therefore, is intended to be exempt from the participation, vesting, funding, and fiduciary requirements of Title I of the Employee Retirement Income Security Act of 1974.

1.2 Purpose of Plan. The purpose of the Plan is to provide Non-Employee
------Directors with an effective means of deferring all or a portion of retainer fees and meeting fees they are entitled to receive, and to provide certain Employees a means of continued deferral of certain Transferred Benefits from the Equifax Inc. Deferred Compensation Plan.

1.3 Applicability of Plan. The provisions of this Plan are applicable only

(a) Employees who have Transferred Benefits; and

tο

(b) individuals who are serving as Non-Employee Directors of the Company.

Article II. Definitions

Whenever used in this Plan, the following terms shall have the meanings set forth below unless otherwise expressly provided. When the defined meaning is intended, the term is capitalized. The definition of any term in the singular shall also include the plural, whichever is appropriate in the context.

 ${\tt 2.1}$ "Account" means the bookkeeping account maintained for each Participant

that represents the Participant's total interest under the Plan as of any Valuation Date. An Account shall consist of the sum of the Participant's Transferred Benefits or the deferrals of Director Fees credited pursuant to section 4.1, plus any gains and losses credited on these amounts. A Participant shall have a fully vested and nonforfeitable interest at all times in his or her Account.

2.2 "Affiliate" means any corporation, association, joint venture,
-----proprietorship, or partnership while it is connected with the Company through

ownership, common control, membership in an affiliated service group, or otherwise within the meaning of Code section 414(b), (c), (m), or (o).

2.3 "Beneficiary" means the person or persons designated by the $\,$

Participant to receive any benefits payable on behalf of the Participant after his or her death. Each Participant shall designate his or her Beneficiary (or change this designation) at a time and in a manner specified by the Committee. If no person is designated as a Beneficiary, if a designation is revoked, or if no designated Beneficiary survives the Participant, the Beneficiary shall be the Participant's estate.

- 2.4 "Board" means the Board of Directors of the Company.
- 2.5 "Code" means the Internal Revenue Code of 1986, as amended, or as it $\hfill {}_{---}$ may be amended from time to time.
- 2.6 "Committee" means the committee appointed by the Board to administer the Plan.
 - 2.7 "Company" means Certegy Inc. or any successor thereto.
- 2.8 "Director Fees" mean the annual retainer and any meeting fees paid by
 the Company to a Non-Employee Director for duties performed as a member of the
 Board.
- - 2.10 "Employee" means any person who is employed by an Employer.
- 2.11 "Employer" means the Company and any Affiliate that elects to become a party to the Plan with the approval of the Company.
 - 2.12 "Entry Date" means January 1, April 1, July 1, or October 1.
- 2.13 "Equifax Plan" means the Equifax Inc. Deferred Compensation Plan,
 from which certain Participants received Transferred Benefits.
- 2.14 "Financial Hardship" means a severe financial hardship resulting from a sudden and unexpected illness or accident of the Participant or one of his or her dependents, loss of the Participant's property due to casualty, or other similar unforeseeable circumstance arising from events that are beyond the control of the Participant. The existence of a Financial Hardship shall be determined by the Committee in a manner consistent with Treasury regulations and rulings of the Internal Revenue Service. The Committee's decision with respect to the existence of a Financial Hardship shall be final and binding.

- 2.15 "Investment Fund" means any fund designated by the Committee as an investment medium for the deemed investment of a Participant's Account. There shall be--
 - (a) a Prime Rate Fund, which shall have earnings based on the prime lending rate (determined as of the first day of each month) as reported in the Wall Street Journal; and
- 2.16 "Non-Employee Director" means a member of the Board who is not an Employee of the Company.
- 2.17 "Participant" means an individual who has met and continues to meet
 ----the eligibility requirements described in section 3.1.
- 2.18 "Plan" means this Certegy Inc. Deferred Compensation Plan, as it may be amended from time to time.
- 2.19 "Plan Year" means the calendar year; provided, however, there shall
 be a short first plan year corresponding to the short fiscal year of the Company ending December 31, 2001.
 - 2.20 "Termination of Service" means--
 - (a) for an Employee, a separation from employment with the Company and its Affiliates; and $\,$
 - (b) for a Non-Employee Director, the date on which such individual ceases to be a member of the Board.
- 2.21 "Transferred Benefits" means the amount transferred to this Plan from the Equifax Plan as provided in Section 3.1.
- 2.22 "Valuation Date" means the last business day of each Plan Year and any other date that the Committee selects in its sole discretion for the revaluation and adjustment of Accounts.

Article III. Eligibility and Participation

- 3.1 Eligibility. An individual shall be eligible to participate in this Plan if he or she--
 - (a) is a Non-Employee Director; or

(b) is an Employee who is a member of a select group of management or highly compensated Employees, who as of the Distribution Date, has an account balance in the Equifax Plan and consents in writing to the substitution of Certegy Inc. for Equifax Inc. as the obligor with respect to the liabilities under the Equifax Plan. The amount credited to the Participant's account under the Equifax Plan immediately prior to the Distribution Date shall be transferred to and constitute the beginning balance of the Participant's account in this Plan (the "Transferred Benefits").

3.2 Participation.

- (a) Commencement of Participation.
 - (1) Non-Employee Directors. A Non-Employee Director shall be eligible to become a Participant as of the earlier of--
 - (A) the day after the Distribution Date; or
 - (B) the Entry Date next following the date on which he or she first becomes a Non-Employee Director.
- (b) Duration of Participation. A Participant shall continue to be an active Participant until he or she ceases to meet the eligibility requirements under section 3.1 or revokes a deferral election under section 4.1(b). Thereafter, he or she shall be an inactive Participant and shall retain all the rights described under this Plan.

Article IV. Contributions

- 4.1 Deferrals.
- (a) Election of Deferral.
 - (1) General Rule. Prior to the first day of each Plan Year, a Non-

Employee Director may elect, by executing a Deferral Election Form in accordance with the procedures prescribed by the Committee, to defer up to 100 percent (in 1 percent increments or a specified dollar amount) of the Director Fees that would otherwise be payable to the Non-Employee Director for the Plan Year. In accordance with procedures established by the Committee, a Non-Employee Director may make separate deferral elections under with respect to retainer fees and meeting fees.

(2) New Participants. In the case of a deferral election which becomes effective on a date other than the first day of a Plan Year, the election

shall relate only to Director Fees which have not yet been earned as of the Participant's Entry Date; or

(3) Allocation to Accounts. Each deferral of Director Fees under

this section 4.1 shall be credited to the Participant's Account as of the date when the amount deferred would have been paid to the Participant.

(c) Default Elections. If an individual participates under the Plan for a

given Plan Year, but does not file a timely election form for the next Plan Year, such individual shall be deemed to elect for the next Plan Year the same deferrals of Director Fees elected for the prior Plan Year.

Article V. Participants' Accounts

5.1 Investment of Accounts.

Fees deferred for that Plan Year.

(a) Investment of Deferrals. For each Plan Year, each Participant shall

elect in writing to deem to have the deferrals made on his or her behalf invested in any one or more of the Investment Funds in 10 percent increments. A Participant may change his or her deemed investment elections with respect to future deferrals as of any January 1. The Participant shall make or change an election of Investment Funds by giving notice to the Committee at a time and in a manner specified by the Committee.

- (b) Transferred Benefits. Any portion of the Transferred Benefits that
 ----was invested in the Equifax Plan's prime rate fund immediately prior to the
 Distribution Date shall be initially invested in the Prime Rate Fund. The
 remaining value of the Transferred Benefits shall be initially invested in
 the Certegy Inc. Common Stock Fund.
- (c) Investment Transfers.
 - (1) General Rule. Except as otherwise provided in paragraph (2)

below, each Participant may elect as of any January 1 to have the amounts that are deemed invested in any one or more of the Investment Funds transferred to any one or more of the other Investment Funds in increments of 10 percent.

(2) Section 16 Participants. Each Participant who is subject to the

reporting and short-swing profit recovery rules of section 16 of the Securities Exchange Act of 1934 may not transfer previously deferred amounts into or out of the Certegy Inc. Common Stock Fund.

- (3) Election Procedures. A Participant shall make an election to transfer among Investment Funds under this subsection (b) by giving notice to the Committee at a time and manner specified by the Committee.
- (d) Committee Discretion. Notwithstanding any provision in this section
- 5.1 to the contrary, the Committee, in its sole and absolute discretion, may disregard the Participant's investment elections and deem the Participant's Account to be invested in any manner it chooses. If the Committee deems the Participant's Account to be invested in a manner other than that elected by the Participant under subsections (a) and (b), it shall notify the Participant in advance of its deemed investment selection. The Company shall incur no liability on account of its selection of deemed investments or on account of the performance of those investments.
- 5.2 Valuation of Accounts.
- (a) Allocation of Earnings and Losses. A Participant's Account shall

be adjusted as of each Valuation Date to reflect any gains or losses that would have been credited or debited to the Account if it had actually been invested in the manner described in section 5.1. Amounts paid from Accounts between these dates will be credited or charged for any investment gains or losses since the last Valuation Date.

- (b) Charges Against Account. Any payments made to a Participant or Beneficiary under Article VI shall be charged against the Participant's Account.
- 5.3 Financing. The benefits under this Plan shall be paid out of the general assets of the Employer, except to the extent they are paid from the assets of a grantor trust established by an Employer to pay these benefits. Whether to establish such a trust is a matter that is within the sole and absolute discretion of the Employer.
- 5.4 Unsecured Interest. No Participant shall have any interest whatsoever in any specific asset of the Employer. To the extent that any person acquires a right to receive payments under this Plan, this right shall be no greater than the right of any unsecured general creditor of the Employer.
- 5.5 Nontransferability. In no event shall an Employer make any payment under this Plan to any assignee or creditor of a Participant or Beneficiary. Prior to the time of payment hereunder, no Participant or Beneficiary shall have any right by way of anticipation or otherwise to assign or otherwise dispose of any interest under this Plan, nor shall rights be assigned or transferred by operation of law.

Article VI. Payment of Accounts

6.1 Payments to Participant.

- (a) Commencement of Payments. Payment of a Participant's Account shall begin within 90 days after the date determined under section 6.2(a).
- (b) Form of Payments. All amounts payable to a Participant shall be distributed in cash in a single sum or in a series of installments, as provided under section 6.2(b).
- 6.2 Distribution Elections.
- (a) Time of Payment.
 - (1) General Rule. Upon making the initial deferral election under section 4.1, the Participant shall also designate the date on which payments from his or her Account shall begin. A Participant may elect initially to have payments begin as of:
 - (A) a date specified by the Participant which must be at least one year after the end of the Plan Year for which the initial deferral is made; or
 - (B) the date of the Participant's Termination of Service.

For Participants with Transferred Benefits, the initial election under this paragraph (1)(A) shall be the last election that was in effect under the Equifax Plan.

- (2) Second Election. If a Participant makes an initial election under paragraph (1)(A), he or she may then elect, with respect to deferrals made after that date, to have payments begin as of:
 - (A) a date specified by the Participant which must be at least one year after the end of the Plan Year containing the date specified under paragraph (1)(A); or
 - (B) the date of the Participant's Termination of Service.
- (3) Administrative Rules. If a Participant specifies a date other

than Termination of Service for the distribution of his or her Account, but incurs a Termination of Service or dies before such date, payments shall begin as soon as practicable following such earlier Termination of Service or death.

If a Participant specifies a distribution date which precedes his or her Termination of Service, the amount distributable shall equal the Account as of such specified date. Deferrals made under section 4.1 after such specified date or dates shall be payable upon the Participant's Termination of Service.

(4) Financial Hardship. A Participant may withdraw all or part of

his or her Account before the distribution date specified in paragraph (1) or (2) above in the event of a Financial Hardship. A withdrawal under this paragraph (4) shall not exceed the amount necessary to satisfy the Financial Hardship. A Participant may request a hardship withdrawal in accordance with procedures established by the Committee.

A Participant who is subject to the reporting and short-swing profit rules of section 16 of the Securities Exchange Act of 1934 may not make a withdrawal under this paragraph (4).

(b) Form of Payment.

(1) General Rule. At the time a Participant makes his or her initial

deferral election under section 4.1, the Participant shall separately elect the manner in which his or her Account shall be paid--

- (A) to the Participant, upon the date determined under subsection (a)(1); and
- (B) to his or her Beneficiary, upon the Participant's death prior to the complete distribution of his or her Account.

Additionally, a Participant who makes a second payment election under subsection (a)(2) shall be permitted to make a payment form election at the same time he or she makes the second election under subsection (a)(2).

The Participant may choose to have the Account paid either in a lump sum (within 90 days of the distribution date determined under subsection (a)) or in a series of annual installments over a fixed number of years (not to exceed ten years).

(2) Limitation on Elections. A Participant shall be permitted to

elect a different payment form for amounts that are distributable as of different payment dates under subsection (a).

(3) Transferred Benefits. A Participant who has Transferred Benefits

will be deemed to have elected the form of payment that was elected by such Participant under the Equifax Plan, provided that such election will only be given effect if payments have not yet commenced and if the election was made in writing at least 12 months prior to the commencement of payment of the Transferred Benefits. In the event that the Participant has made more than one such election, the most recent election filed with the plan administrator but no more than 12 months prior to the date payments are to commence will control.

(c) Discretion of Committee. Notwithstanding a Participant's election of
the time or form of payment for his or her Account, the Committee may

direct, in its sole and absolute discretion, that the Account shall be distributed in any time, and/or in any form, permitted under subsection (a) or (b).

- 6.3 Payments to Beneficiary.
- (a) Commencement of Payments. If a Participant dies before his or her

Account has been completely distributed, the remaining balance shall be paid to the Participant's Beneficiary beginning within 90 days after the Participant's death.

- (b) Form of Payments. Payments to the Beneficiary shall be made in a single sum or in a series of installments, as provided under section 6.2.
- (c) Death of Participant and Beneficiary. If the Participant and
 Beneficiary both die before the Participant's Account has been completely distributed, these remaining benefits shall be paid as follows.
 - (1) If the Beneficiary dies before the Participant, the balance of the Participant's Account shall be paid to the Participant's estate in a single sum.
 - (2) If the Beneficiary dies after the Participant, the balance of the Account shall be paid to the Beneficiary's estate in a single ${\sf sum.}$

Article VII. Administration

7.1 Administration. The Plan shall be administered by the Committee. A

majority of the members of the Committee at the time in office shall constitute a quorum for the transaction of business. All resolutions and other actions taken by the Committee at any meeting shall be by a majority vote of those present at the meeting. Upon the unanimous concurrence in writing of all Committee members, action of the Committee may be taken other than at a meeting.

The Committee shall have all powers necessary or appropriate to carry out the provisions of the Plan. It may, from time to time, establish rules for the administration of the Plan and the transaction of the Plan's business.

The Committee shall have the exclusive right to make any finding of fact necessary or appropriate for any purpose under the Plan including, but not limited to, the determination of eligibility for and amount of any benefit.

The Committee shall have the exclusive right to interpret the terms and provisions of the Plan and to determine any and all questions arising under the Plan or in connection with its administration, including, without limitation, the right to remedy or resolve possible ambiguities, inconsistencies, or omissions by general rule or particular decision, all in its sole and absolute discretion.

All findings of fact, determinations, interpretations, and decisions of the Committee shall be conclusive and binding upon all persons having or claiming to have any interest or right under the Plan and shall be given the maximum possible deference allowed by law.

7.2 Appeals from Denial of Claims. If any claim for benefits under the

Plan is wholly or partially denied, the claimant shall be given notice in writing of the denial. This notice shall be in writing, within a reasonable period of time after receipt of the claim by the Committee. This period shall not exceed 90 days after receipt of the claim, except that if special circumstances require an extension of time, written notice of the extension shall be furnished to the claimant, and an additional 90 days will be considered reasonable.

This notice shall be written in a manner calculated to be understood by the claimant and shall set forth the following information:

- (a) the specific reasons for the denial;
- (b) specific reference to the Plan provisions on which the denial is based;
- (c) a description of any additional material or information necessary for the claimant to perfect the claim and an explanation of why this material or information is necessary;
- (d) an explanation that a full and fair review by the Committee of the decision denying the claim may be requested by the claimant or an authorized representative by filing with the Committee, within 60 days after the notice has been received, a written request for the review; and (e) if this request is so filed, an explanation that the claimant or an authorized representative may review pertinent documents and submit issues and comments in writing within the same 60-day period specified in subsection (d).

The decision of the Committee upon review shall be made promptly, and not later than 60 days after the Committee's receipt of the request for review, unless special circumstances require extension of time for processing. In this case the claimant shall be so notified, and a decision shall be rendered as soon as possible, but not later than 120 days after receipt of the request for review. If the claim is denied, wholly or in part, the claimant shall be given a copy of the decision promptly. The decision shall be in writing, shall include specific reasons for the denial, shall include specific references to the pertinent Plan provisions on which the denial is based, and shall be written in a manner calculated to be understood by the claimant.

- 7.3 Tax Withholding. The Employer may withhold from any payment under this Plan any federal, state, or local taxes required by law to be withheld with respect to the payment and any sum the Employer may reasonably estimate as necessary to cover any taxes for which they may be liable and that may be assessed with regard to the payment.
- 7.4 Expenses. All expenses incurred in the administration of the Plan shall be paid by the Employer.

Article VIII. Adoption of the Plan by Affiliate;

- 8.1 Adoption of the Plan by Affiliate. An Affiliate may adopt the Plan by appropriate action of its board of directors or authorized officers or representatives, subject to the approval of the Board.
- 8.2 Amendment and Termination. The Company hereby reserves the right to amend, modify, or terminate the Plan at any time, and for any reason, by action of the Board. However, no amendment or termination shall adversely affect benefits accrued prior to the date of the amendment or termination.

Article IX. Miscellaneous Provisions

- 9.1 No Contract of Employment. Nothing contained in the Plan shall be construed to give any Participant the right to be retained in the service of an Employer or to interfere with the right of an Employer to discharge a Participant at any time.
- 9.2 Severability. If any provision of this Plan shall be held illegal or invalid, the illegality or invalidity shall not affect its remaining parts. The Plan shall be construed and enforced as if it did not contain the illegal or invalid provision.
- 9.3 Applicable Law. Except to the extent preempted by applicable federal law, this Plan shall be governed by and construed in accordance with the laws of the state of Georgia.

* * * * * * * * * * *

IN WITNESS WHEREOF, CERTEGY INC. has caused this instrument to be executed by its duly authorized officer, effective as of the date specified above.	
CERTEGY INC.	
Ву:	
Title:	
ATTEST:	
By:	

Title: _____

[Date]	
[Name]	
[Title]	
[Address]	

Dear

Certegy Inc. (the "Company") considers the establishment and maintenance of a sound and vital management to be essential to protecting and enhancing the best interests of the Company and its shareholders. In this connection, the Company recognizes that, as is the case with many publicly held corporations, the possibility of a change in control exists and that possibility, and the uncertainty and questions that it may raise among management, may result in the departure or distraction of management personnel to the detriment of the Company and its shareholders. Accordingly, the Board of Directors of the Company has determined that appropriate steps should be taken to reinforce and encourage the continued attention and dedication of members of the Company's management, including yourself, to their assigned duties without distraction in the face of potentially disturbing circumstances arising from the possibility of a change in control of the Company.

In order to induce you to remain in its employ, the Company agrees to provide you the payments and benefits described in this Letter (in lieu of any severance payments and benefits you would otherwise receive in accordance with the Company's severance pay practices) if your employment with the Company is terminated subsequent to a "Change in Control" of the Company (as defined in paragraph 3) under the circumstances described in paragraph 4.

- 1. No Right to Continued Employment. This Letter does not give you any right
- to continued employment by the Company or a Subsidiary, and it will not interfere in any way with the right the Company or a Subsidiary otherwise may have to terminate your employment at any time.
- 2. Term of This Letter. The terms of this Letter will be effective as of

______, 2001, and, except as otherwise provided in this Letter, will continue in effect until ______, 2006; provided that commencing on January 1, 2002 and each subsequent January 1, the terms of this Letter will be extended automatically so as to remain in effect for five (5) years from that January 1 unless

[Name] [Date] Page 2

at least sixty (60) days prior to January 1 of a given year, the Company notifies you that it does not wish to continue this Letter in effect beyond its then current expiration date; and provided further that if a Change in Control occurs prior to the expiration of this Letter, this Letter will continue in effect for three (3) years from the Change in Control.

3. Change In Control. No benefits will be payable under this Letter unless

there is a Change in Control and your employment by the Company is terminated under the circumstances described in paragraph 4 entitling you to benefits. For purposes of this Letter, a Change in Control of the Company means the occurrence of any of the following events during the period in which this Letter remains in effect:

3.1 Voting Stock Accumulations. The accumulation by any Person of

Beneficial Ownership of twenty percent (20%) or more of the combined voting power of the Company's Voting Stock; provided that for purposes of this subparagraph 3.1, a Change in Control will not be deemed to have occurred if the accumulation of twenty percent (20%) or more of the voting power of the Company's Voting Stock results from any acquisition of Voting Stock (a) directly from the Company that is approved by the Incumbent Board, (b) by the Company, (c) by any employee benefit plan (or related trust) sponsored or maintained by the Company or any Subsidiary, or (d) by any Person pursuant to a Business Combination that complies with all of the provisions of clauses (a), (b) and (c) of subparagraph 3.2; or

 $3.2\,$ Business Combinations. Consummation of a Business Combination,

unless, immediately following that Business Combination, (a) all or substantially all of the Persons who were the beneficial owners of Voting Stock of the Company immediately prior to that Business Combination beneficially own, directly or indirectly, more than sixty-six and twothirds percent (66-2/3%) of the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of Directors of the entity resulting from that Business Combination (including, without limitation, an entity that as a result of that transaction owns the Company or all or substantially all of the Company's assets either directly or through one or more subsidiaries) in substantially the same proportions relative to each other as their ownership, immediately prior to that Business Combination, of the Voting Stock of the Company, (b) no Person (other than the Company, that entity resulting from that Business Combination, or any employee benefit plan (or related trust) sponsored or maintained by the Company, any Eighty Percent (80%)

Subsidiary or that entity resulting from that Business Combination) beneficially owns, directly or indirectly, twenty percent (20%) or more of the then outstanding shares of common stock of the entity resulting from that Business Combination or the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors of that entity, and (c) at least a majority of the members of the Board of Directors of the entity resulting from that Business Combination were members of the Incumbent Board at the time of the execution of the initial agreement or of the action of the Board providing for that Business Combination; or

- 3.3 Sale of Assets. A sale or other disposition of all or substantially
- all of the assets of the Company; or
- 3.4 Liquidations or Dissolutions. Approval by the shareholders of the

Company of a complete liquidation or dissolution of the Company, except pursuant to a Business Combination that complies with all of the provisions of clauses (a), (b) and (c) of subparagraph 3.2.

For purposes of this paragraph 3, the following definitions will apply:

"Beneficial Ownership" means beneficial ownership as that term is used in Rule 13d-3 promulgated under the Exchange Act.

"Business Combination" means a reorganization, merger or consolidation of the Company.

"Eighty Percent (80%) Subsidiary" means an entity in which the Company directly or indirectly beneficially owns eighty percent (80%) or more of the outstanding Voting Stock.

"Exchange Act" means the Securities Exchange Act of 1934, including amendments, or successor statutes of similar intent.

"Incumbent Board" means a Board of Directors at least a majority of whom consist of individuals who either are (a) members of the Company's Board of Directors as of the date of this Letter or (b) members who become members of the Company's Board of Directors subsequent to the date of this Letter whose election, or nomination for election by the Company's shareholders, was approved by a vote of at least two-thirds (2/3) of the directors then comprising the Incumbent Board (either by a specific vote or by approval of

the proxy statement of the Company in which that person is named as a nominee for director, without objection to that nomination), but excluding, for that purpose, any individual whose initial assumption of office occurs as a result of an actual or threatened election contest (within the meaning of Rule 14a-11 of the Exchange Act) with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board of Directors.

"Person" means any individual, entity or group (within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act).

"Voting Stock" means the then outstanding securities of an entity entitled to vote generally in the election of members of that entity's Board of Directors

4. Termination Following Change in Control. If any of the events described in

paragraph 3 constituting a Change in Control occurs, you will be entitled to the payments and benefits provided for in paragraph 5 if your employment is terminated within six (6) months prior to the Change in Control in connection with the Change in Control or your employment is terminated within three (3) years from the date of the Change in Control, unless your termination is (a) because of your death, (b) by the Company for Cause or Disability, or (c) by you other than for Good Reason. The payments and benefits provided for in paragraph 5 will be in lieu of any severance payments you would otherwise receive in accordance with the Company's severance pay practices, but will have no effect on any of the Company's other employee benefit plans or practices, as amended from time to time.

4.1 Cause. Termination by the Company of your employment for "Cause"

means termination by the Company of your employment upon (a) your willful and continued failure to substantially perform your duties with the Company (other than any failure resulting from your incapacity due to physical or mental illness), after a written demand for substantial performance is delivered to you by the Chief Executive Officer of the Company (or if you are the Chief Executive Officer, the Chairman of the Compensation Committee of the Board of Directors) that specifically identifies the manner in which the Chief Executive Officer believes that you have not substantially performed your duties, or (b) your willfully engaging in misconduct that is materially injurious to the Company, monetarily or otherwise. For purposes of this subparagraph 4.1, no act, or failure to act, on your part will be considered "willful" unless done, or omitted to be done, by you not in good faith and without reasonable belief that your action or

omission was in the best interest of the Company. Notwithstanding the above, you will not be deemed to have been terminated for Cause unless and until you have been given a copy of a Notice of Termination from the Chief Executive Officer of the Company (or if you are the Chief Executive Officer, the Chairman of the Compensation Committee of the Board of Directors), after reasonable notice to you and an opportunity for you, together with your counsel, to be heard before (i) the Chief Executive Officer, or (ii) if you are an elected officer of the Company, the Board of Directors of the Company, finding that in the good faith opinion of the Chief Executive Officer, or, in the case of an elected officer, finding that in the good faith opinion of two-thirds of the Board of Directors, you committed the conduct set forth above in clauses (a) or (b) of this subparagraph 4.1, and specifying the particulars of that finding in detail.

4.2 Disability. Termination by the Company of your employment for

"Disability" means termination by the Company of your employment following and because of your failure to perform your duties as an employee for a period of at least one hundred eighty (180) consecutive calendar days as a result of total and permanent incapacity due to physical or mental illness or injury. Your incapacity must be certified by a licensed medical doctor selected by you. You will continue to receive your full base salary at the rate in effect and any bonus payments under the Plan payable during the one hundred eighty (180) day qualification period until termination of your employment for Disability. After that termination, your benefits will be determined in accordance with the Company's long-term disability plan then in effect and any of the Company's other benefit plans and practices then in effect that apply to you. The Company will have no further obligation to you under this Letter and all supplemental benefits will be terminated. If the Company disagrees with the certification of your incapacity, it may appoint another medical doctor to certify his opinion as to your incapacity, and if that doctor does not certify as to your incapacity, then the two doctors will appoint a third medical doctor to certify their opinion as to your incapacity, and the decision of a majority of the three doctors will prevail. (The Company will bear the costs of the doctors opinions.)

- 4.3 Good Reason. Termination by you of your employment for "Good Reason" means termination by you of your employment based on:
 - (a) The assignment to you of duties inconsistent with your position and status with the Company as they existed immediately prior to the Change in Control Date (as defined below), or a substantial change in

your title, offices or authority, or in the nature of your responsibilities, as they existed immediately prior to the Change in Control Date (or if you receive a promotion or an increase in responsibilities or authority after the Change in Control Date, then a change with respect to your enhanced position, status, responsibilities or authority), except in connection with the termination of your employment for Cause or Disability or as a result of your death or by you other than for Good Reason;

- (b) A reduction by the Company in your base salary as in effect on the date of this Letter or as your salary may be increased from time to time;
- (c) A failure by the Company to continue the Company's incentive compensation plan(s) ("Incentive Plan"), as it may be modified from time to time, substantially in the form in effect immediately prior to the Change in Control Date, or a failure by the Company to continue you as a participant in the Incentive Plan on at least the basis of your participation immediately prior to the Change in Control Date or to pay you the amounts that you would be entitled to receive in accordance with the Incentive Plan;
- (d) The Company's requiring you to be based more than thirty-five (35) miles from the location where you are based immediately prior to the Change in Control Date, except for required travel on the Company's business to an extent substantially consistent with your business travel obligations prior to the Change in Control Date, or if you consent to that relocation, the failure by the Company to pay (or reimburse you for) all reasonable moving expenses incurred by you or to indemnify you against any loss realized in the sale of your principal residence in connection with that relocation;
- (e) The failure by the Company to continue in effect any retirement or compensation plan, supplemental retirement plan, performance share plan, stock option plan, life insurance plan, health and accident plan, disability plan or any other benefit plan in which you are participating immediately prior to the Change in Control Date (or provide plans providing you with substantially similar benefits), the taking of any action by the Company that would adversely affect your participation or materially reduce your benefits under any of those plans or deprive you of any material fringe benefit enjoyed by you

immediately prior to the Change in Control Date, or the failure by the Company to provide you with the number of paid vacation days to which you are then entitled in accordance with the Company's normal vacation practices in effect immediately prior to the Change in Control Date;

- (f) The failure by the Company to obtain the assumption of the agreement to perform this Letter by any successor, as contemplated in paragraph 6; or
- (g) Any purported termination of your employment that is not effected pursuant to a Notice of Termination satisfying the requirements of subparagraph 4.4 (and, if applicable, subparagraph 4.1).

For purposes of this subparagraph 4.3, "Change in Control Date" means the date six months prior to the date of the Change in Control.

4.4 Notice of Termination. Any purported termination by the Company

pursuant to subparagraphs 4.1 or 4.2 or by you pursuant to subparagraph 4.3 will be communicated by written Notice of Termination to the other party. For purposes of this Letter, a "Notice of Termination" means a notice that indicates the specific termination provision in this Letter relied upon and setting forth in reasonable detail the facts and circumstances claimed to provide a basis for termination of your employment under the provision so indicated. Any purported termination not effected pursuant to a Notice of Termination meeting the requirements set forth in this Letter will not be effective.

4.5 Date of Termination. For purposes of this Letter, the date of the $\,$

termination of your employment ("Date of Termination") will be (a) if your employment is terminated by your death, the end of the month in which your death occurs, (b) if your employment is terminated for Disability, thirty (30) days after Notice of Termination is given, or (c) if your employment is terminated by you or the Company for any other reason, the date specified in the Notice of Termination, which will not be later than thirty (30) days after the date on which the Notice of Termination is given.

5. Benefits upon Certain Terminations following a Change in Control. If any of the events described in paragraph 3 constituting a Change in Control occurs and your employment is terminated under the circumstances described in paragraph 4

which entitle you to payments and benefits under this paragraph 5, then the provisions of subparagraphs 5.1 through 5.6 will apply. For purposes of determining your compensation and benefits under this paragraph 5, amounts accrued, earned or received under plans or programs maintained by Equifax Inc. prior to the date the Company was spun-off from Equifax Inc. will be treated as if they were accrued, earned or received under plans or programs maintained by the Company.

- ${\tt 5.1}$ Compensation through Date of Termination. The Company will pay you
- (a) any unpaid amount of your base salary through the Date of Termination, (b) with respect to any year then completed, any unpaid amount accrued to you pursuant to the Incentive Plan, and (c) with respect to any year then partially completed, a pro rata portion through the Date of Termination of your target annual bonus under the Incentive Plan. For purposes of item (c) above, your "target annual bonus under the Incentive Plan" will be your annual base salary as of the Date of Termination multiplied by the target percentage of your bonus under the Incentive Plan.
- 5.2 Additional Severance. In lieu of any further salary payments to you

for periods subsequent to the Date of Termination, the Company will pay as severance pay to you on the fifth (5th) business day following the Date of Termination a lump sum equal to three (3) times the sum of (a) your annual base salary at the highest rate in effect during the twelve (12) months immediately preceding the Date of Termination plus (b) the higher of (i) the highest annual bonus paid to you or paid but deferred on your behalf under the Incentive Plan, (ii) any earned, but unpaid, bonus accrued for your benefit under the Incentive Plan, or (iii) your highest target annual bonus under the Incentive Plan, whether or not earned, in each case with respect to the three (3) calendar years immediately preceding the year in which the Date of Termination occurs and the partial calendar year ending on the Date of Termination. For purposes of item (iii) above and subparagraph 5.3, the "highest target annual bonus under the Incentive Plan" for the partial calendar year ending on the Date of Termination will be your annual base salary as of the Date of Termination multiplied by the target percentage of your bonus under the Incentive Plan.

5.3 Additional Retirement Benefit. If you are a participant in the

Company's U.S. Retirement Income Plan (the "Retirement Plan"), the Company will pay you on the fifth (5th) business day following the Date of Termination a lump sum retirement benefit, in addition to the benefits to which you are or would be entitled under the Retirement Plan. That benefit

will be a lump sum amount that is the actuarial equivalent of your benefits calculated pursuant to the terms of the Retirement Plan with the following adjustments: (a) regardless of your Years of Vesting Service under the Retirement Plan, you will be treated as if you were 100% vested under the Retirement Plan, (b) the number of Years of Benefit Service used will be the actual number of Years of Benefit Service accumulated as of the Date of Termination plus an additional number of Years of Benefit Service (up to a maximum of five (5) additional years) equal to the number of additional Years of Benefit Service that you would have earned if you had remained an employee of the Company until attainment of age sixty-two (62), (c) the Final Average Earnings (for purposes of applying the benefit formula under the Retirement Plan) will be determined using (I) the highest monthly rate of Base Salary in effect during the twelve (12) months immediately preceding the Date of Termination, plus (II) the higher of (A) the highest annual bonus paid to you or paid but deferred on your behalf under the Plan, (B) any earned, but unpaid, bonus accrued for your benefit under the Plan, or (C) your highest target annual bonus under the Plan, whether or not earned, in each case with respect to the three (3) calendar years immediately preceding the Date of Termination and the partial calendar year ending on the Date of Termination, divided by twelve (12) (regardless of the earnings limitations under the Retirement Plan or governmental regulations applicable to those plans), and (d) the monthly retirement benefit so calculated will be reduced by an amount equal to the monthly retirement benefit payable to you under the Retirement Plan. All capitalized terms used in this subparagraph, unless otherwise defined, will have the same meanings as those terms are defined in the Retirement Plan. The actuarial equivalent will be calculated based on the assumptions contained in the Retirement Plan on the Date of Termination; provided that the assumptions on which the actuarial equivalent will be calculated will be no less favorable to you than those assumptions contained in the Retirement Plan on the date of the Change in Control.

5.4 Benefit Plans.

(a) Unless your employment is terminated for Cause, the Company will maintain in full force and effect, for your continued benefit for three (3) years after your Date of Termination, the group health, dental, vision, life insurance. disability and similar coverages in which you are entitled to participate immediately prior to the Date of Termination at the same level as for active employees and in the same manner as if your employment had not terminated. Any additional coverages you had at termination, including dependent coverage, will also be

continued for that period on the same terms, to the extent permitted by the applicable policies or contracts. You will be responsible for paying any costs you were paying for those coverages at the time of termination by separate check payable to the Company each month in advance. If the terms of any benefit plan referred to in this subparagraph 5.4(a), or the laws applicable to that plan do not permit your continued participation, then the Company will arrange for other coverages satisfactory to you at the Company's expense that provide substantially similar benefits, or the Company will pay you a lump sum amount equal to the costs you would have to pay to obtain those coverage(s) for the three-year period.

- (b) If you have satisfied the requirements for receiving the Company's retiree medical coverage on your Date of Termination or will satisfy those requirements prior to the last day of the three-year benefit continuation period provided in item (a) above, you (and your dependents) will be covered by, and receive benefits under, the Company's retiree medical coverage program for employees at your level. Your retiree medical coverage will commence on the date your health care coverage terminates under item (a) above, and will continue for your life (i.e., the coverage will be vested and may not be terminated), subject only to those changes in the level of coverage that apply to employees at your level generally.
- (c) You will be entitled to continue to participate in the Company's 401(k) Retirement and Savings Plan for the three-year period after your Date of Termination. For purposes of the 401(k) Plan, you will receive an amount equal to the Company's contributions to the 401(k) Plan, assuming you had made contributions to the 401(k) Plan at the maximum permissible level. If the Company cannot contribute those additional amounts to the 401(k) Plan on your behalf because of the terms of the 401(k) Plan or applicable law, the Company will pay to you within five (5) days of the Date of Termination a lump sum amount equal to the additional amounts the Company would have been required to contribute (based upon the terms of the 401(k) Plan as in effect on the Date of Termination).
- 5.5 No Mitigation Required. You will not be required to mitigate the

amount of any payment or benefits provided for in this paragraph 5 by seeking other employment or otherwise, nor will the amount of any payment or benefits provided for in this paragraph 5 be reduced by any compensation

earned by you, or benefits provided to you, as the result of employment by another employer after the Date of Termination, or otherwise.

5.6 Tax Gross-Up Payment. If any payments or benefits provided pursuant

to this Letter or any other payments or benefits provided to you by the Company are subject to an excise tax on an "excess parachute payment" under Section 4999 of the Internal Revenue Code of 1986 (the "Code"), or any successor provision of the Code, or are subject to an excise or penalty tax under any similar provision of any other revenue system to which you may be subject, the Company will provide a gross-up payment to you in order to place you in the same after-tax position you would have been in had no excise or penalty tax become due and payable under Code Section 4999 (or any successor provision) or any similar provision of that other revenue system. That gross-up payment will not apply to any excise or penalty tax attributable to any incentive stock option granted to you by the Company or Equifax Inc. prior to April 1, 1998. Any gross-up payment to which you are entitled as a result of the applicability of an excise tax under Code Section 4999 or any successor provision of the Code, or as a result of any excise or penalty tax under any similar provision of any other revenue system to which you may be subject, will be determined in accordance with a "Policy with Respect to Tax Gross-up Payments" adopted, or which will be adopted, by the Board of Directors (or a Committee of the Board), and once that policy is adopted, no amendment of that policy that adversely affects you will be effective with respect to your rights under this Letter without your written consent.

6. Successors: Binding Agreement.

6.1 Assumption by Company's Successor. The Company will require any

successor (whether direct or indirect, by purchase, merger, consolidation or otherwise) to all or substantially all of the business and/or assets of the Company, by agreement in form and substance reasonably satisfactory to you, to expressly assume and agree to perform this Letter. Failure of the Company to obtain that agreement prior to the effectiveness of any succession will be a breach of this Letter and will entitle you to compensation from the Company in the same amount and on the same terms as you would be entitled under this Letter if you terminated your employment for Good Reason within three (3) years following a Change in Control, except that for purposes of implementing the foregoing, the date on which that succession becomes effective will be deemed the Date of Termination. As used in this Letter, "Company" means Certegy Inc. and any successor to its business

[Name] [Date] Page 12

and/or assets that executes and delivers the agreement provided for in this subparagraph 6.1 or that otherwise becomes bound by all the terms and provisions of this Letter by operation of law.

6.2 Enforcement by Your Successor. This Letter will inure to the benefit

of and be enforceable by your personal or legal representatives, executors, administrators, successors, heirs, distributees, devisees and legatees. If you die subsequent to the termination of your employment while any amount would still be payable to you pursuant to this Letter if you had continued to live, all those amounts, unless otherwise provided in this Letter, will be paid in accordance with the terms of this Letter to your devisee, legatee or other designee or, if there be no designee, to your estate; that payment to be made in a lump sum within sixty (60) days from the date of your death.

7. Notice. For purposes of this Letter, notices and all other communications

provided for in this Letter will be in writing and will be deemed to have been duly given when delivered or mailed by United States registered mail, return receipt requested, postage pre-paid, addressed to the respective addresses set forth on the first page of this Letter, provided that all notices to the Company will be directed to the attention of the Chief Executive Officer of the Company (or if the notice is from the Chief Executive Officer, to the General Counsel of the Company), or to that other address as either party may have furnished to the other in writing in accordance with this paragraph 7, except that notice of change of address will be effective only upon receipt.

8. Modification and Waiver. No provision of this Letter may be modified,

waived or discharged unless that waiver, modification or discharge is agreed to in writing by you and that officer as may be specifically designated by the Board of Directors of the Company. No waiver by either party at any time of any breach by the other party of, or compliance with, any condition or provision of this Letter to be performed by that other party will be deemed a waiver of similar or dissimilar provisions or conditions at the time or at any prior or subsequent time.

9. Construction. This Letter supersedes (a) any oral agreement between you and

the Company and any oral representation by the Company to you with respect to the subject matter of this Letter, and (b) that letter agreement dated _____, ___ between you and Equifax Inc. pertaining to change in control. The validity, interpretation, construction and performance of this Letter will be governed by the laws of the State of Georgia.

- 10. Severability. If any one or more of the provisions of this Letter or any
- word, phrase, clause, sentence or other portion of a provision is deemed illegal or unenforceable for any reason, that provision or portion will be modified or deleted in such a manner as to make this Letter as modified legal and enforceable to the fullest extent permitted under applicable laws. The validity and enforceability of the remaining provisions or portions of this Letter will remain in full force and effect.
- 11. Counterparts. This Letter may be executed in two or more counterparts,
- each of which will take effect as an original and all of which will evidence one and the same agreement.
- 12. Legal Fees. If the Company breaches this Letter or if, within three (3)

years following a Change in Control, your employment is terminated under circumstances described in paragraph 4 that entitle you to payments and benefits under paragraph 5, the Company will reimburse you for all legal fees and expenses reasonably incurred by you as a result of that termination (including all those fees and expenses, if any, incurred in contesting or disputing the termination or in seeking to obtain or enforce any right or benefit provided by this Letter).

Upon presentation to the Company of the invoice for those legal fees and expenses, the Company will reimburse you monthly for those legal fees and expenses.

- 13. Indemnification. After your termination, the Company will indemnify you
- and hold you harmless from and against any claim relating to your performance as an officer, director or employee of the Company or any of its subsidiaries or other affiliates or in any other capacity, including any fiduciary capacity, in which you served at the Company's request, in each case to the maximum extent permitted by law and under the Company's Articles of Incorporation and Bylaws (the "Governing Documents"), provided that under no circumstances will the protection afforded to you under this paragraph be less than that afforded under the Governing Documents as in effect on the date of this Agreement except for changes mandated by law. You will continue to receive the benefits of, and be covered by, any policy of directors and officers liability insurance maintained by the Company for the benefit of its directors, officers and employees.
- 14. Employment by a Subsidiary. Either the Company or a Subsidiary may be your

legal employer. For purposes of this Letter, any reference to your termination of employment with the Company means termination of employment with the Company and all Subsidiaries, and does not include a transfer of employment between any of them. The actions referred to under the definition of "Good Reason" in subparagraph 4.3 include the actions of the Company or your

[Name]
[Date]
Page 14

employing Subsidiary, as applicable. The obligations created under this Letter are obligations of the Company. A change in control of a Subsidiary will not constitute a Change in Control for purposes of this Letter unless there is also a contemporaneous Change in Control of the Company. For purposes of paragraph 1 and this paragraph, a "Subsidiary" means an entity more than fifty percent (50%) of whose equity interests are owned directly or indirectly by the Company.

If you accept the above terms, please sign and return to me the enclosed copy of this Letter.

Sincer	ely,	
Agreed	to as of	
[Name]		

[Date]	
[Name]	
[Title]	
[Address]	

Dear

Certegy Inc. (the "Company") considers the establishment and maintenance of a sound and vital management to be essential to protecting and enhancing the best interests of the Company and its shareholders. In this connection, the Company recognizes that, as is the case with many publicly held corporations, the possibility of a change in control exists and that possibility, and the uncertainty and questions that it may raise among management, may result in the departure or distraction of management personnel to the detriment of the Company and its shareholders. Accordingly, the Board of Directors of the Company has determined that appropriate steps should be taken to reinforce and encourage the continued attention and dedication of members of the Company's management, including yourself, to their assigned duties without distraction in the face of potentially disturbing circumstances arising from the possibility of a change in control of the Company.

In order to induce you to remain in its employ, the Company agrees to provide you the payments and benefits described in this Letter (in lieu of any severance payments and benefits you would otherwise receive in accordance with the Company's severance pay practices) if your employment with the Company is terminated subsequent to a "Change in Control" of the Company (as defined in paragraph 3) under the circumstances described in paragraph 4.

- 1. No Right to Continued Employment. This Letter does not give you any right
- to continued employment by the Company or a Subsidiary, and it will not interfere in any way with the right the Company or a Subsidiary otherwise may have to terminate your employment at any time.
- 2. Term of This Letter. The terms of this Letter will be effective as of ________, 2001, and, except as otherwise provided in this Letter, will

continue in effect until ______, 2006; provided that commencing on January 1, 2002 and each subsequent January 1, the terms of this Letter will be extended

Page 2

automatically so as to remain in effect for five (5) years from that January 1 unless at least sixty (60) days prior to January 1 of a given year, the Company notifies you that it does not wish to continue this Letter in effect beyond its then current expiration date; and provided further that if a Change in Control occurs prior to the expiration of this Letter, this Letter will continue in effect for two (2) years from the Change in Control.

3. Change In Control. No benefits will be payable under this Letter unless

there is a Change in Control and your employment by the Company is terminated under the circumstances described in paragraph 4 entitling you to benefits. For purposes of this Letter, a Change in Control of the Company means the occurrence of any of the following events during the period in which this Letter remains in effect:

3.1 Voting Stock Accumulations. The accumulation by any Person of

Beneficial Ownership of twenty percent (20%) or more of the combined voting power of the Company's Voting Stock; provided that for purposes of this subparagraph 3.1, a Change in Control will not be deemed to have occurred if the accumulation of twenty percent (20%) or more of the voting power of the Company's Voting Stock results from any acquisition of Voting Stock (a) directly from the Company that is approved by the Incumbent Board, (b) by the Company, (c) by any employee benefit plan (or related trust) sponsored or maintained by the Company or any Subsidiary, or (d) by any Person pursuant to a Business Combination that complies with all of the provisions of clauses (a), (b) and (c) of subparagraph 3.2; or

3.2 Business Combinations. Consummation of a Business Combination,

unless, immediately following that Business Combination, (a) all or substantially all of the Persons who were the beneficial owners of Voting Stock of the Company immediately prior to that Business Combination beneficially own, directly or indirectly, more than sixty-six and two-thirds percent (66-2/3%) of the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of Directors of the entity resulting from that Business Combination (including, without limitation, an entity that as a result of that transaction owns the Company or all or substantially all of the Company's assets either directly or through one or more subsidiaries) in substantially the same proportions relative to each other as their ownership, immediately prior to that Business Combination, of the Voting Stock of the Company, (b) no Person (other than the Company, that entity resulting from

that Business Combination, or any employee benefit plan (or related trust) sponsored or maintained by the Company, any Eighty Percent (80%) Subsidiary or that entity resulting from that Business Combination) beneficially owns, directly or indirectly, twenty percent (20%) or more of the then outstanding shares of common stock of the entity resulting from that Business Combination or the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors of that entity, and (c) at least a majority of the members of the Board of Directors of the entity resulting from that Business Combination were members of the Incumbent Board at the time of the execution of the initial agreement or of the action of the Board providing for that Business Combination; or

- 3.3 Sale of Assets. A sale or other disposition of all or substantially
- all of the assets of the Company; or

3.4 Liquidations or Dissolutions. Approval by the shareholders of the

Company of a complete liquidation or dissolution of the Company, except pursuant to a Business Combination that complies with all of the provisions of clauses (a), (b) and (c) of subparagraph 3.2.

For purposes of this paragraph 3, the following definitions will apply:

"Beneficial Ownership" means beneficial ownership as that term is used in Rule 13d-3 promulgated under the Exchange Act.

"Business Combination" means a reorganization, merger or consolidation of the Company.

"Eighty Percent (80%) Subsidiary" means an entity in which the Company directly or indirectly beneficially owns eighty percent (80%) or more of the outstanding Voting Stock.

"Exchange Act" means the Securities Exchange Act of 1934, including amendments, or successor statutes of similar intent.

"Incumbent Board" means a Board of Directors at least a majority of whom consist of individuals who either are (a) members of the Company's Board of Directors as of the date of this Letter or (b) members who become members of the Company's Board of Directors subsequent to the date of this Letter

whose election, or nomination for election by the Company's shareholders, was approved by a vote of at least two-thirds (2/3) of the directors then comprising the Incumbent Board (either by a specific vote or by approval of the proxy statement of the Company in which that person is named as a nominee for director, without objection to that nomination), but excluding, for that purpose, any individual whose initial assumption of office occurs as a result of an actual or threatened election contest (within the meaning of Rule 14a-11 of the Exchange Act) with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board of Directors.

"Person" means any individual, entity or group (within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act).

"Voting Stock" means the then outstanding securities of an entity entitled to vote generally in the election of members of that entity's Board of Directors.

- 4. Termination Following Change in Control. If any of the events described in paragraph 3 constituting a Change in Control occurs, you will be entitled to the payments and benefits provided for in paragraph 5 if your employment is terminated within six (6) months prior to the Change in Control in connection with the Change in Control or your employment is terminated within two (2) years from the date of the Change in Control, unless your termination is (a) because of your death, (b) by the Company for Cause or Disability, or (c) by you other than for Good Reason. The payments and benefits provided for in paragraph 5 will be in lieu of any severance payments you would otherwise receive in accordance with the Company's severance pay practices, but will have no effect on any of the Company's other employee benefit plans or practices, as amended from time to time
 - 4.1 Cause. Termination by the Company of your employment for "Cause"

means termination by the Company of your employment upon (a) your willful and continued failure to substantially perform your duties with the Company (other than any failure resulting from your incapacity due to physical or mental illness), after a written demand for substantial performance is delivered to you by the Chief Executive Officer of the Company (or if you are the Chief Executive Officer, the Chairman of the Compensation Committee of the Board of Directors) that specifically identifies the manner in which the Chief Executive Officer believes that you have not substantially performed your duties, or (b) your willfully engaging in

misconduct that is materially injurious to the Company, monetarily or otherwise. For purposes of this subparagraph 4.1, no act, or failure to act, on your part will be considered "willful" unless done, or omitted to be done, by you not in good faith and without reasonable belief that your action or omission was in the best interest of the Company. Notwithstanding the above, you will not be deemed to have been terminated for Cause unless and until you have been given a copy of a Notice of Termination from the Chief Executive Officer of the Company (or if you are the Chief Executive Officer, the Chairman of the Compensation Committee of the Board of Directors), after reasonable notice to you and an opportunity for you, together with your counsel, to be heard before (i) the Chief Executive Officer, or (ii) if you are an elected officer of the Company, the Board of Directors of the Company, finding that in the good faith opinion of the Chief Executive Officer, or, in the case of an elected officer, finding that in the good faith opinion of two-thirds of the Board of Directors, you committed the conduct set forth above in clauses (a) or (b) of this subparagraph 4.1, and specifying the particulars of that finding in detail.

4.2 Disability. Termination by the Company of your employment for

"Disability" means termination by the Company of your employment following and because of your failure to perform your duties as an employee for a period of at least one hundred eighty (180) consecutive calendar days as a result of total and permanent incapacity due to physical or mental illness or injury. Your incapacity must be certified by a licensed medical doctor selected by you. You will continue to receive your full base salary at the rate in effect and any bonus payments under the Plan payable during the one hundred eighty (180) day qualification period until termination of your employment for Disability. After that termination, your benefits will be determined in accordance with the Company's long-term disability plan then in effect and any of the Company's other benefit plans and practices then in effect that apply to you. The Company will have no further obligation to you under this Letter and all supplemental benefits will be terminated. If the Company disagrees with the certification of your incapacity, it may appoint another medical doctor to certify his opinion as to your incapacity, and if that doctor does not certify as to your incapacity, then the two doctors will appoint a third medical doctor to certify their opinion as to your incapacity, and the decision of a majority of the three doctors will prevail. (The Company will bear the costs of the doctors opinions.)

Page 6

4.3 Good Reason. Termination by you of your employment for "Good Reason"

means termination by you of your employment based on:

- (a) The assignment to you of duties inconsistent with your position and status with the Company as they existed immediately prior to the Change in Control Date (as defined below), or a substantial change in your title, offices or authority, or in the nature of your responsibilities, as they existed immediately prior to the Change in Control Date (or if you receive a promotion or an increase in responsibilities or authority after the Change in Control Date, then a change with respect to your enhanced position, status, responsibilities or authority), except in connection with the termination of your employment for Cause or Disability or as a result of your death or by you other than for Good Reason;
- (b) A reduction by the Company in your base salary as in effect on the date of this Letter or as your salary may be increased from time to time;
- (c) A failure by the Company to continue the Company's incentive compensation plan(s) ("Incentive Plan"), as it may be modified from time to time, substantially in the form in effect immediately prior to the Change in Control Date, or a failure by the Company to continue you as a participant in the Incentive Plan on at least the basis of your participation immediately prior to the Change in Control Date or to pay you the amounts that you would be entitled to receive in accordance with the Incentive Plan;
- (d) The Company's requiring you to be based more than thirty-five (35) miles from the location where you are based immediately prior to the Change in Control Date, except for required travel on the Company's business to an extent substantially consistent with your business travel obligations prior to the Change in Control Date, or if you consent to that relocation, the failure by the Company to pay (or reimburse you for) all reasonable moving expenses incurred by you or to indemnify you against any loss realized in the sale of your principal residence in connection with that relocation;
- (e) The failure by the Company to continue in effect any retirement or compensation plan, supplemental retirement plan,

performance share plan, stock option plan, life insurance plan, health and accident plan, disability plan or any other benefit plan in which you are participating immediately prior to the Change in Control Date (or provide plans providing you with substantially similar benefits), the taking of any action by the Company that would adversely affect your participation or materially reduce your benefits under any of those plans or deprive you of any material fringe benefit enjoyed by you immediately prior to the Change in Control Date, or the failure by the Company to provide you with the number of paid vacation days to which you are then entitled in accordance with the Company's normal vacation practices in effect immediately prior to the Change in Control Date;

- (f) The failure by the Company to obtain the assumption of the agreement to perform this Letter by any successor, as contemplated in paragraph 6; or
- (g) Any purported termination of your employment that is not effected pursuant to a Notice of Termination satisfying the requirements of subparagraph 4.4 (and, if applicable, subparagraph 4.1).

For purposes of this subparagraph 4.3, "Change in Control Date" means the date six months prior to the date of the Change in Control.

4.4 Notice of Termination. Any purported termination by the Company

pursuant to subparagraphs 4.1 or 4.2 or by you pursuant to subparagraph 4.3 will be communicated by written Notice of Termination to the other party. For purposes of this Letter, a "Notice of Termination" means a notice that indicates the specific termination provision in this Letter relied upon and setting forth in reasonable detail the facts and circumstances claimed to provide a basis for termination of your employment under the provision so indicated. Any purported termination not effected pursuant to a Notice of Termination meeting the requirements set forth in this Letter will not be effective.

4.5 Date of Termination. For purposes of this Letter, the date of the

termination of your employment ("Date of Termination") will be (a) if your employment is terminated by your death, the end of the month in which

your death occurs, (b) if your employment is terminated for Disability, thirty (30) days after Notice of Termination is given, or (c) if your employment is terminated by you or the Company for any other reason, the date specified in the Notice of Termination, which will not be later than thirty (30) days after the date on which the Notice of Termination is given.

5. Benefits upon Certain Terminations following a Change in Control. If any

of the events described in paragraph 3 constituting a Change in Control occurs and your employment is terminated under the circumstances described in paragraph 4 which entitle you to payments and benefits under this paragraph 5, then the provisions of subparagraphs 5.1 through 5.6 will apply. For purposes of determining your compensation and benefits under this paragraph 5, amounts accrued, earned or received under plans or programs maintained by Equifax Inc., prior to the date the Company was spun-off from Equifax Inc., will be treated as if they were accrued, earned or received under plans or programs maintained by the Company.

- 5.1 Compensation through Date of Termination. The Company will pay you
- (a) any unpaid amount of your base salary through the Date of Termination, (b) with respect to any year then completed, any unpaid amount accrued to you pursuant to the Incentive Plan, and (c) with respect to any year then partially completed, a pro rata portion through the Date of Termination of your target annual bonus under the Incentive Plan. For purposes of item (c) above, your "target annual bonus under the Incentive Plan" will be your annual base salary as of the Date of Termination multiplied by the target percentage of your bonus under the Incentive Plan.
- 5.2 Additional Severance. In lieu of any further salary payments to you

for periods subsequent to the Date of Termination, the Company will pay as severance pay to you on the fifth (5th) business day following the Date of Termination a lump sum equal to two (2) times the sum of (a) your annual base salary at the highest rate in effect during the twelve (12) months immediately preceding the Date of Termination plus (b) the higher of (i) the highest annual bonus paid to you or paid but deferred on your behalf under the Incentive Plan, (ii) any earned, but unpaid, bonus accrued for your benefit under the Incentive Plan, or (iii) your highest target annual bonus under the Incentive Plan, whether or not earned, in each case with respect to the three (3) calendar years immediately preceding the year in which the Date of Termination occurs and the partial calendar year ending on the Date

of Termination. For purposes of item (iii) above and subparagraph 5.3, the "highest target annual bonus under the Incentive Plan" for the partial calendar year ending on the Date of Termination will be your annual base salary as of the Date of Termination multiplied by the target percentage of your bonus under the Incentive Plan.

5.3 Additional Retirement Benefit. If you are a participant in the

Company's U.S. Retirement Income Plan (the "Retirement Plan"), the Company will pay you on the fifth (5th) business day following the Date of Termination a lump sum retirement benefit, in addition to the benefits to which you are or would be entitled under the Retirement Plan. That benefit will be a lump sum amount that is the actuarial equivalent of your benefits calculated pursuant to the terms of the Retirement Plan with the following adjustments: (a) regardless of your Years of Vesting Service under the Retirement Plan, you will be treated as if you were 100% vested under the Retirement Plan, (b) the number of Years of Benefit Service used will be the actual number of Years of Benefit Service accumulated as of the Date of Termination plus an additional number of Years of Benefit Service (up to a maximum of five (5) additional years) equal to the number of additional Years of Benefit Service that you would have earned if you had remained an employee of the Company until attainment of age sixty-two (62), (c) the Final Average Earnings (for purposes of applying the benefit formula under the Retirement Plan) will be determined using (I) the highest monthly rate of Base Salary in effect during the twelve (12) months immediately preceding the Date of Termination, plus (II) the higher of (A) the highest annual bonus paid to you or paid but deferred on your behalf under the Plan, (B) any earned, but unpaid, bonus accrued for your benefit under the Plan, or (C) your highest target annual bonus under the Plan, whether or not earned, in each case with respect to the three (3) calendar years immediately preceding the Date of Termination and the partial calendar year ending on the Date of Termination, divided by twelve (12) (regardless of the earnings limitations under the Retirement Plan or governmental regulations applicable to those plans), and (d) the monthly retirement benefit so calculated will be reduced by an amount equal to the monthly retirement benefit payable to you under the Retirement Plan. All capitalized terms used in this subparagraph, unless otherwise defined, will have the same meanings as those terms are defined in the Retirement Plan. The actuarial equivalent will be calculated based on the assumptions contained in the Retirement Plan on the Date of Termination; provided that the assumptions on which the actuarial equivalent

will be calculated will be no less favorable to you than those assumptions contained in the Retirement Plan on the date of the Change in Control.

5.4 Benefit Plans.

- (a) Unless your employment is terminated for Cause, the Company will maintain in full force and effect, for your continued benefit for two (2) years after your Date of Termination, the group health, dental, vision, life insurance. disability and similar coverages in which you are entitled to participate immediately prior to the Date of Termination at the same level as for active employees and in the same manner as if your employment had not terminated. Any additional coverages you had at termination, including dependent coverage, will also be continued for that period on the same terms, to the extent permitted by the applicable policies or contracts. You will be responsible for paying any costs you were paying for those coverages at the time of termination by separate check payable to the Company each month in advance. If the terms of any benefit plan referred to in this subparagraph 5.4(a), or the laws applicable to that plan do not permit your continued participation, then the Company will arrange for other coverages satisfactory to you at the Company's expense that provide substantially similar benefits, or the Company will pay you a lump sum amount equal to the costs you would have to pay to obtain those coverage(s) for the three-year period.
- (b) If you have satisfied the requirements for receiving the Company's retiree medical coverage on your Date of Termination or will satisfy those requirements prior to the last day of the two-year benefit continuation period provided in item (a) above, you (and your dependents) will be covered by, and receive benefits under, the Company's retiree medical coverage program for employees at your level. Your retiree medical coverage will commence on the date your health care coverage terminates under item (a) above, and will continue for your life (i.e., the coverage will be vested and may not be terminated), subject only to those changes in the level of coverage that apply to employees at your level generally.
- (c) You will be entitled to continue to participate in the Company's 401(k) Retirement and Savings Plan for the two-year period after your Date of Termination. For purposes of the 401(k) Plan, you will receive

an amount equal to the Company's contributions to the 401(k) Plan, assuming you had made contributions to the 401(k) Plan at the maximum permissible level. If the Company cannot contribute those additional amounts to the 401(k) Plan on your behalf because of the terms of the 401(k) Plan or applicable law, the Company will pay to you within five (5) days of the Date of Termination a lump sum amount equal to the additional amounts the Company would have been required to contribute (based upon the terms of the 401(k) Plan as in effect on the Date of Termination).

5.5 No Mitigation Required. You will not be required to mitigate the

amount of any payment or benefits provided for in this paragraph 5 by seeking other employment or otherwise, nor will the amount of any payment or benefits provided for in this paragraph 5 be reduced by any compensation earned by you, or benefits provided to you, as the result of employment by another employer after the Date of Termination, or otherwise.

5.6 Tax Gross-Up Payment. If any payments or benefits provided pursuant

to this Letter or any other payments or benefits provided to you by the Company are subject to an excise tax on an "excess parachute payment" under Section 4999 of the Internal Revenue Code of 1986 (the "Code"), or any successor provision of the Code, or are subject to an excise or penalty tax under any similar provision of any other revenue system to which you may be subject, the Company will provide a gross-up payment to you in order to place you in the same after-tax position you would have been in had no excise or penalty tax become due and payable under Code Section 4999 (or any successor provision) or any similar provision of that other revenue system. That gross-up payment will not apply to any excise or penalty tax attributable to any incentive stock option granted to you by the Company or Equifax Inc. prior to April 1, 1998. Any gross-up payment to which you are entitled as a result of the applicability of an excise tax under Code Section 4999 or any successor provision of the Code, or as a result of any excise or penalty tax under any similar provision of any other revenue system to which you may be subject, will be determined in accordance with a "Policy with Respect to Tax Gross-up Payments" adopted, or which will be adopted, by the Board of Directors (or a Committee of the Board), and once that policy is adopted, no amendment of that policy that adversely affects you will be effective with respect to your rights under this Letter without your written consent.

6. Successors: Binding Agreement.

6.1 Assumption by Company's Successor. The Company will require any

successor (whether direct or indirect, by purchase, merger, consolidation or otherwise) to all or substantially all of the business and/or assets of the Company, by agreement in form and substance reasonably satisfactory to you, to expressly assume and agree to perform this Letter. Failure of the Company to obtain that agreement prior to the effectiveness of any succession will be a breach of this Letter and will entitle you to compensation from the Company in the same amount and on the same terms as you would be entitled under this Letter if you terminated your employment for Good Reason within two (2) years following a Change in Control, except that for purposes of implementing the foregoing, the date on which that succession becomes effective will be deemed the Date of Termination. As used in this Letter, "Company" means Certegy Inc. and any successor to its business and/or assets that executes and delivers the agreement provided for in this subparagraph 6.1 or that otherwise becomes bound by all the terms and provisions of this Letter by operation of law.

 ${\tt 6.2}$ $\,$ Enforcement by Your Successor. This Letter will inure to the benefit

of and be enforceable by your personal or legal representatives, executors, administrators, successors, heirs, distributees, devisees and legatees. If you die subsequent to the termination of your employment while any amount would still be payable to you pursuant to this Letter if you had continued to live, all those amounts, unless otherwise provided in this Letter, will be paid in accordance with the terms of this Letter to your devisee, legatee or other designee or, if there be no designee, to your estate; that payment to be made in a lump sum within sixty (60) days from the date of your death.

7. Notice. For purposes of this Letter, notices and all other communications

provided for in this Letter will be in writing and will be deemed to have been duly given when delivered or mailed by United States registered mail, return receipt requested, postage pre-paid, addressed to the respective addresses set forth on the first page of this Letter, provided that all notices to the Company will be directed to the attention of the Chief Executive Officer of the Company (or if the notice is from the Chief Executive Officer, to the General Counsel of the Company), or to that other address as either party may have furnished to the other in writing in accordance with this paragraph 7, except that notice of change of address will be effective only upon receipt.

Page 13

8. Modification and Waiver. No provision of this Letter may be modified,

waived or discharged unless that waiver, modification or discharge is agreed to in writing by you and that officer as may be specifically designated by the Board of Directors of the Company. No waiver by either party at any time of any breach by the other party of, or compliance with, any condition or provision of this Letter to be performed by that other party will be deemed a waiver of similar or dissimilar provisions or conditions at the time or at any prior or subsequent time.

- 9. Construction. This Letter supersedes (a) any oral agreement between you
- and the Company and any oral representation by the Company to you with respect to the subject matter of this Letter, and (b) that letter agreement dated ______, ____ between you and Equifax Inc. pertaining to change in control. The validity, interpretation, construction and performance of this Letter will be governed by the laws of the State of Georgia.
- 10. Severability. If any one or more of the provisions of this Letter or any word, phrase, clause, sentence or other portion of a provision is deemed illegal or unpuforceable for any reason, that provision or portion will be modified or

or unenforceable for any reason, that provision or portion will be modified or deleted in such a manner as to make this Letter as modified legal and enforceable to the fullest extent permitted under applicable laws. The validity and enforceability of the remaining provisions or portions of this Letter will remain in full force and effect.

11. Counterparts. This Letter may be executed in two or more counterparts,

each of which will take effect as an original and all of which will evidence one and the same agreement.

12. Legal Fees. If the Company breaches this Letter or if, within two (2)

years following a Change in Control, your employment is terminated under circumstances described in paragraph 4 that entitle you to payments and benefits under paragraph 5, the Company will reimburse you for all legal fees and expenses reasonably incurred by you as a result of that termination (including all those fees and expenses, if any, incurred in contesting or disputing the termination or in seeking to obtain or enforce any right or benefit provided by this Letter).

Upon presentation to the Company of the invoice for those legal fees and expenses, the Company will reimburse you monthly for those legal fees and expenses.

13. Indemnification. After your termination, the Company will indemnify you

and hold you harmless from and against any claim relating to your performance as an officer, director or employee of the Company or any of its subsidiaries or other

Page 14
affiliates or in any other capacity, including any fiduciary capacity, in which you served at the Company's request, in each case to the maximum extent permitted by law and under the Company's Articles of Incorporation and Bylaws (the "Governing Documents"), provided that under no circumstances will the protection afforded to you under this paragraph be less than that afforded under the Governing Documents as in effect on the date of this Agreement except for changes mandated by law. You will continue to receive the benefits of, and be covered by, any policy of directors and officers liability insurance maintained by the Company for the benefit of its directors, officers and employees.
14. Employment by a Subsidiary. Either the Company or a Subsidiary may be
your legal employer. For purposes of this Letter, any reference to your termination of employment with the Company means termination of employment with the Company and all Subsidiaries, and does not include a transfer of employment between any of them. The actions referred to under the definition of "Good Reason" in subparagraph 4.3 include the actions of the Company or your employing Subsidiary, as applicable. The obligations created under this Letter are obligations of the Company. A change in control of a Subsidiary will not constitute a Change in Control for purposes of this Letter unless there is also a contemporaneous Change in Control of the Company. For purposes of paragraph 1 and this paragraph, a "Subsidiary" means an entity more than fifty percent (50%) of whose equity interests are owned directly or indirectly by the Company.
If you accept the above terms, please sign and return to me the enclosed copy of this Letter.
Sincerely,
Agreed to as of,
[Name]

CERTEGY INC.

Executive Life and Supplemental Retirement Benefit Plan

THIS EXECUTIVE LIFE AND SUPPLEMENTAL RETIREMENT BENEFIT PLAN (the "Plan"), effective immediately following the close of the Distribution Date, is hereby adopted and established by Certegy Inc., a Georgia corporation, (the "Company") and will be maintained by the Company for selected executives as provided herein.

Article I - Purpose

The purpose of the Plan is to reward certain specified executives of the Company for their service to the Company and to provide an incentive to the Participants, including newly hired executives, for future service and loyalty to the Company. This Plan provides benefits through life insurance policies (each a "Policy") on the lives of Participants. Plan benefits and a Participant's interest in his or her Policy shall be as set forth in the Participant Agreements that each Participant is required to execute with the Company before becoming a participant herein. In all cases, a Participant's interest in this Plan and the benefits provided hereunder shall be governed by this Plan and the terms of the Participant Agreements, which shall be considered to be a part of this Plan.

The Plan is established by the Company in connection with the spinoff of the Company from Equifax Inc., and shall be considered a successor to the Equifax Inc. Executive Life and Supplemental Retirement Benefit Plan (U.S.) (the "Prior Plan").

Article II - Eligibility and Participation

2.1 Eligibility and Participation. Participants shall be designated by

the Plan Administrator (but not by any designee thereof) or the Company's Chief Executive Officer and shall be informed in writing of the effective date of their participation in the Plan (the "Commencement Date") and their level of life insurance benefits to which they may be entitled. In order to participate, a Participant must complete certain enrollment documents and must execute (i) a Split-Dollar Life Insurance Agreement which specifies, among other matters, the respective interests of the Participant and the Company in the Policy issued by the Insurance Company, and (ii) a Collateral Assignment of certain rights in the Policy in favor of the Company (collectively, the "Participant Agreements"). Individuals who were participants in the Prior Plan on the close of the Distribution Date and who become employed by the Company, or remain so employed, immediately after the close of the Distribution Date (the "Transferred") Individuals") shall automatically become Participants in this Plan immediately after the close of the Distribution Date, which shall automatically be their Commencement Date, and the Split-Dollar Life Insurance Agreements and Collateral Assignments that were in effect under the Prior Plan (collectively, the "Prior Agreements") shall be applicable under this Plan, subject to the following provisions of this Article II.

2.2 Succession. With respect to Participants who are Transferred

Individuals, the Company shall succeed to all rights and obligations of Equifax Inc. under the Prior Agreements,

including the right to recover from any cash values under each Transferred Individual's Policy by means of enforcing the collateral assignment relating to such Policy. All rights of Transferred Individuals under the Prior Agreements shall be determined consistently with this succession to the rights and obligations of Equifax Inc., and subject thereto and Section 2.3 below, the Prior Agreements shall continue in effect hereunder (and to this extent shall be considered Participant Agreements). Except as provided in this Section 2.2, the Transferred Individuals shall cease to have any rights under the Prior Agreements immediately after the close of the Distribution Date.

2.3 Supersession. The Company and a Transferred Individual may enter into

one or more amendments to the Prior Agreements, and they may enter into a new Split-Dollar Life Insurance Agreement, a new Collateral Assignment or both, which shall supersede in part or in whole, as appropriate, the Prior Agreements. Further, as a condition of continuing as a Participant, the Company may require a Transferred Individual to execute any such document (and to perform such acts) as may be necessary or desirable to effectuate the assumption by the Company of all of the right, title, interests, and obligations of Equifax Inc. in and to the Transferred Individual's Prior Agreements.

Article III - Definitions

The following terms shall have the meanings ascribed to them below for purposes of the Plan, the Participant Agreements and the Questions and Answers for the Executive Life and Supplemental Retirement Benefit Plan, as initially in effect following the Distribution Date, and any amendments, supplements or successors thereto. Other capitalized terms used herein and not defined herein shall have the meanings ascribed to them in the Participant Agreements.

- 3.1 Cause. "Cause" shall mean termination by the Company of the
- Participant's employment upon any one of the following circumstances:
 - (a) the Participant's willful and continued failure to substantially perform the Participant's duties with the Company (other than any failure resulting from the Participant's incapacity due to physical or mental illness, including being Permanently Disabled), after a written demand for substantial performance is delivered to the Participant by the Chief Executive Officer of the Company (or if the Participant is the Chief Executive Officer, the Chairman of the Compensation and Human Resources Committee of the Board of Directors) that specifically identifies the manner in which the Chief Executive Officer (or the Chairman) believes that the Participant has not substantially performed the Participant's duties, or
 - (b) the Participant willfully engaging in conduct that is materially injurious to the Company, monetarily or otherwise.

For purposes of this Section 3.1, no act, or failure to act, on the Participant's part will be considered "willful" unless done, or omitted to be done, by the Participant not in good faith and without reasonable belief that the Participant's action or omission was in the best interest of the Company. Notwithstanding the above, the Participant will not be deemed to have been terminated for Cause unless and until the Participant has been given a copy of a Notice of Termination from the Chief Executive Officer of the Company (or if the Participant is the Chief Executive Officer, the Chairman of the Compensation and Human Resources Committee of the Board of Directors), after reasonable notice to the Participant and an opportunity for the

Participant, together with the Participant's counsel, to be heard before (i) the Chief Executive Officer, or (ii) if the Participant is an elected officer of the Company, the Board of Directors of the Company, finding that in the good faith opinion of the Chief Executive Officer, or, in the case of an elected officer, finding that in the good faith opinion of two-thirds of the Board of Directors, the Participant committed the conduct set forth above in clauses (a) or (b) of this Section 3.1, and specifying the particulars of that finding in detail.

- 3.2 Change in Control. "Change in Control" shall mean the occurrence of any one of the following events during the period in which the Plan remains in effect:
 - (a) Voting Stock Accumulations. The accumulation by any Person of
 Beneficial Ownership of twenty percent (20%) or more of the combined voting
 power of the Company's Voting Stock; provided that for purposes of this

paragraph (a), a Change in Control will not be deemed to have occurred if the accumulation of twenty percent (20%) or more of the voting power of the Company's Voting Stock results from any acquisition of Voting Stock (i) directly from the Company that is approved by the Incumbent Board, (ii) by the Company, (iii) by any employee benefit plan (or related trust) sponsored or maintained by the Company or any Subsidiary, or (iv) by any Person pursuant to a Business Combination that complies with all of the provisions of clauses (i), (ii) and (iii) of paragraph (b) below,

(b) Business Combinations. The consummation of a Business $\,$

Combination, unless, immediately following that Business Combination, (i)all or substantially all of the Persons who were the beneficial owners of Voting Stock of the Company immediately prior to that Business Combination beneficially own, directly or indirectly, more than sixty-six and twothirds percent (66 2/3%) of the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors of the entity resulting from that Business Combination (including an entity that as a result of that transaction owns the Company or all or substantially all of the Company's assets either directly or through one or more subsidiaries) in substantially the same proportions relative to each other as their ownership, immediately prior to that Business Combination, of the Voting Stock of the Company, (ii) no Person (other than the Company, that entity resulting from that Business Combination, or any employee benefit plan (or related trust) sponsored or maintained by the Company, any Eighty Percent (80%) Subsidiary or that entity resulting from that Business Combination) beneficially owns, directly or indirectly, twenty percent (20%) or more of the then outstanding shares of common stock of the entity resulting from that Business Combination or the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors of that entity, and (iii) at least a majority of the members of the Board of Directors of the entity resulting from that Business Combination were members of the Incumbent Board at the time of the action of the board providing for that Business Combination;

- (c) Sale of Assets. A sale or other disposition of all or substantially all of the assets of the Company; or

Business Combination that complies with all of the provisions of clauses (i), (ii) and (iii) of paragraph (b) above.

For purposes of this Section 3.2, the following definitions will apply:

"Beneficial Ownership" means beneficial ownership as that term is used in Rule 13d-3 promulgated under the Exchange Act.

"Business Combination" means a reorganization, merger or consolidation of the Company.

"Eighty Percent (80%) Subsidiary" means an entity in which the Company directly or indirectly beneficially owns eighty percent (80%) or more of the outstanding Voting Stock.

"Exchange Act" means the Securities Exchange Act of 1934, including amendments, or successor statutes of similar intent.

"Incumbent Board" means a Board of Directors at least a majority of whom consist of individuals who either are (a) members of the Company's Board of Directors as of the day after the spinoff of the Company from Equifax Inc. becomes effective, or (b) members who become members of the Company's Board of Directors subsequent to such date whose election, or nomination for election by the Company's shareholders, was approved by a vote of at least two-thirds (2/3) of the directors then comprising the Incumbent Board (either by a specific vote or by approval of the proxy statement of the Company in which that person is named as a nominee for director, without objection to that nomination), but excluding, for that purpose, any individual whose initial assumption of office occurs as a result of an actual or threatened election contest (within the meaning of Rule 14a-11 of the Exchange Act) with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board of Directors.

"Person" means any individual, entity or group (within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act).

"Voting Stock" means the then outstanding securities of an entity entitled to vote generally in the election of members of that entity's Board of Directors.

- 3.3 Claimant. "Claimant" shall have the meaning given to it in Section
- 3.4 Commencement Date. "Commencement Date" shall have the meaning given to it in Section 2.1.

4.1.

3.5 Company. "Company" shall mean Certegy Inc., a Georgia corporation, and its successor or successors.

- 3.6 Competitive Activity. A Participant or former Participant shall be deemed to engage in "Competitive Activity" if he or she:
 - (a) directly or indirectly owns, operates, controls, participates in, performs services for, or otherwise carries on, a business substantially similar to or competitive with the business conducted by the Company or any Subsidiary (without limit to any particular region, because Participant acknowledges that such business may be engaged in effectively from any

location in the United States or Canada); provided that nothing set forth

in this paragraph (a) will prohibit a Participant from owning not in excess of 5% of any class of capital stock of any corporation if such stock is publicly traded and listed on any national or regional stock exchange or on the Nasdaq Stock Market;

- (b) directly or indirectly attempts to persuade any employee or customer of the Company or any Subsidiary to terminate such employment or business relationship in order to enter into any such relationship on behalf of the Participant or any third party in competition with the business conducted by the Company or any Subsidiary; or
- (c) directly or indirectly engages in any activity that is harmful to the interests of the Company or any Subsidiary, as determined by the Compensation and Human Resources Committee in its sole discretion, including the disclosure or misuse of any confidential information or trade secrets of the Company or a Subsidiary.
- 3.7 Distribution Date. "Distribution Date" shall have the meaning given to it in the Employee Benefits Agreement between Equifax Inc. and Certegy Inc.
- - (a) The assignment to the Participant of duties inconsistent with the Participant's position and status with the Company as they existed immediately prior to the Change in Control, or a substantial change in the Participant's title, offices or authority, or in the nature of the Participant's responsibilities, as they existed immediately prior to the Change in Control, except in connection with the termination of the Participant's employment by the Company for Cause, by the Participant other than for Good Reason or as a result of death;
 - (b) A reduction by the Company in the Participant's base salary as in effect on the Commencement Date or as the Participant's salary may be increased from time to time;
 - (c) A failure by the Company to continue the Company's incentive compensation plan(s), as it may be modified from time to time, substantially in the form in effect immediately prior to a Change in Control (the "Incentive Plan"), or a failure by the Company to continue the Participant as a participant in the Incentive Plan on at least the basis of the Participant's participation immediately prior to a Change in Control, or to pay the Participant the amounts that the Participant would be entitled to receive in accordance with the terms of the Incentive Plan (as in effect immediately prior to the Change in Control);

- (d) The Company requiring the Participant to be based more than thirty-five (35) miles from the location where the Participant is based prior to the Change in Control, except for required travel on Company business to an extent substantially consistent with the Participant's business travel obligations immediately prior to the Change in Control; or if the Participant consents to the relocation, the failure by the Company to pay (or reimburse the Participant for) all reasonable moving expenses incurred by the Participant or to indemnify the Participant against any loss realized on the sale of the Participant's principal residence in connection with the relocation;
- (e) The failure by the Company to continue in effect any retirement plan, compensation plan, performance share plan, stock option plan, life insurance plan, health and accident plan, disability plan or another benefit plan in which the Participant is participating immediately prior to a Change in Control (except that the Company may cancel any such plans without triggering this paragraph (e), if it provides the Participant with substantially similar benefits under another plan), the taking of any action by the Company that would adversely affect the Participant's participation or materially reduce the Participant's benefits under any such plans or deprive the Participant of any material fringe benefit enjoyed by the Participant immediately prior to a Change in Control, or the failure by the Company to provide the Participant with the number of paid vacation days to which the Participant is then entitled in accordance with the Company's normal vacation practices in effect immediately prior to a Change in Control; or
- (f) Any purported termination not effected pursuant to a Notice of Termination shall not be valid for purposes of this Plan.
- 3.9 Insurance Company. "Insurance Company" shall mean Pacific Life

Insurance Company, or such successor or successors that have issued life insurance policies on the lives of the Participants with respect to the Plan.

- 3.10 Notice of Termination. A "Notice of Termination" shall mean a written notice that indicates the specific provision in the definition of Cause relied upon as the basis for the Participant's termination of employment and setting forth in reasonable detail the facts and circumstances claimed to provide a basis for the termination of Participant's employment under the provision so indicated.
- 3.11 Participant. "Participant" shall mean those specified executives of the Company or any affiliate who have been designated by the Plan Administrator or Company's Chief Executive Officer as eligible to participate herein (or who are automatically eligible to participate under Section 2.1), who have completed all enrollment documents as specified in Section 2.1 and who remain so qualified.
- 3.12 Participant Agreements. "Participant Agreements" shall have the meaning given to it in Section 2.1.
 - 3.13 Permanently Disabled. "Permanently Disabled" shall mean the

Participant suffering a sickness, accident or injury, which in the determination of the Plan Administrator would entitle the Participant to disability benefits under either social security or the Company's

long-term disability plan. The Company reserves the right to require the Participant to first qualify for disability benefits under either social security or the Company's long-term disability plan before determining whether such Participant is Permanently Disabled for purposes of this Plan.

- 3.14 Plan. "Plan" shall mean this Executive Life and Supplemental
 ---Retirement Benefit Plan, as amended and restated from time to time.
- - 3.16 Policy. "Policy" shall have the meaning given to it in Article I.
- 3.17 Prior Agreements. "Prior Agreements" shall have the meaning given to it in Section 2.1.
- 3.18 Prior Plan. "Prior Plan" shall have the meaning given to it in $$\tt Article\ I.$
- 3.19 Retirement. "Retirement" shall mean a Participant's termination of employment with the Company and all affiliates after (a) attaining age 65, (b) attaining age 55 and five "Years of Vesting Service," or (c) attaining age 50 and the Participant's age plus his or her "Years of Benefit Service" equals at least 75. "Years of Vesting Service" and "Years of Benefit Service" shall have the meanings given to them in the Certegy Inc. U.S. Retirement Income Plan.
- 3.21 Transferred Individual. "Transferred Individual" shall have the meaning given to it in Section 2.1.

Article IV - Claims Procedures

4.1 Claims and Review Procedures. The claims procedures contained in this

Article IV shall apply for all purposes of this Plan and the benefits provided herein and through the Participant Agreements. The claims procedure in Section 4.2 below shall be followed with respect to benefits provided by the Insurance Company under the terms of the Policies. The claims procedures in Section 4.3 below shall be followed with respect to benefits provided directly by the Company. The Participant and his or her heirs, successors, beneficiaries and personal representatives (individually or collectively, a "Claimant") must follow both procedures, if necessary.

4.2 Filing a Claim for Insurance Benefits. A Claimant shall make a claim

for death benefits provided by the Insurance Company by submitting a written claim and proof of claim to the Insurance Company in accordance with procedures and guidelines established from time to time by the Insurance Company. On written request, the Plan Administrator shall provide copies of any claim forms or instructions, or advise the Claimant how to obtain such forms or

instructions. The Insurance Company shall decide whether the claim for death benefits shall be allowed. If a claim is denied in whole or in part, the Insurance Company shall notify the Claimant and explain the procedure for reviewing a denied claim.

- 4.3 Filing a Claim with the Company.
 - (a) Initial Procedures. If a Claimant does not receive the Company

benefits under this Plan to which the Claimant believes he or she is entitled, the Claimant must file a written claim for benefits in accordance with the terms of this Article. Not later than ninety (90) days after receipt of such a claim, the Plan Administrator shall render a written decision on the claim to the Claimant, unless special circumstances require the extension of such ninety (90) day period. If such extension is necessary, the Plan Administrator shall provide the Claimant with written notification of such extension before the expiration of the initial ninety (90) day period. Such notice shall specify the reason or reasons for such extension and the date by which a final decision can be expected. In no event shall such extension exceed a period of sixty (60) days from the end of the initial ninety (90) day period.

- (b) Claim Denial. In the event the Plan Administrator denies the claim of a Claimant in whole or in part, the Plan Administrator's written notification shall specify, in a manner calculated to be understood by the Claimant, (a) the reason for the denial, (b) a reference to the Plan or other document or form that is the basis for the denial, (c) a description of any additional material or information necessary for the Claimant to perfect the claim, (d) an explanation as to why such information or material is necessary, and (e) an explanation of the applicable claims procedure.
- (c) Subsequent Claim Review. If the claim is denied in whole or in part and should the Claimant be dissatisfied with the Plan Administrator's disposition of the Claimant's claim, the Claimant may have a full and fair review of the claim by the Compensation and Human Resources Committee of the Board of Directors of the Company (the "Committee") upon written request. Such request for additional review of the claim must be submitted by the Claimant or the Claimant's duly authorized representative and received by the Committee within sixty (60) days after the Claimant receives written notification that the Claimant's claim has been denied by the Plan Administrator. In connection with such review, the Claimant or the Claimant's duly authorized representative shall be entitled to review pertinent documents and submit the Claimant's views as to the issues, in writing. The Committee shall act to deny or accept the claim within sixty (60) days after receipt of the Claimant's written request for review. The action of the Company shall be in the form of a written notice to the Claimant and its contents shall include all of the requirements for action

on the original claim. In no event may a Claimant commence legal action for benefits the Claimant believes are due until the Claimant has exhausted all of

the remedies and procedures afforded the Claimant by this Article.

Article V - Effect of a Change in Control

In the event of a Change in Control, the trustee of the grantor trust that has been established by the Company with respect to the Plan shall, as provided in such grantor trust, ensure that appropriate Company contributions to the grantor trust and payments of Policy premiums from the grantor trust are made with respect to the Participants.

Article VI - Amendment and Termination

6.1 Amendment. The Company reserves the right to amend this Plan at any

time by action of the Company's Board of Directors. The Company, however, may not make any amendment that changes the definition of "Change in Control" or "Good Reason" after a Change in Control has occurred with respect to such Change in Control without the written consent of all Participants as of the date of such change.

 $6.2\,$ Termination. The Company reserves the right to terminate this Plan,

by action of the Company's Board of Directors, at any time it deems appropriate. Upon termination of the Plan, no further Policy premium payments shall be made by the Company and the rights of Participants with respect to their Policies shall be as set forth in their respective Split-dollar Life Insurance Agreements. Except as expressly provided in this Plan, the Company shall not have any further financial obligations to any Participant after the termination of this Plan. The Company shall provide written notice to all Participants if the Plan is terminated. Notwithstanding the preceding provisions of this Section 6.2, in the event of a Change in Control, the Company shall not be able to reduce a Participant's rights pursuant to this Section 6.2 to an extent that exceeds its ability to reduce the Participant's rights under Section 6.1.

Article VII -- Administration

- 7.2 Powers and Duties of the Plan Administrator. Subject to the express terms and conditions set forth herein, the Plan Administrator shall have the power to perform any and all actions, determinations and interpretations related to the administration of the Plan, including the power from time to time:
 - (a) to carry out the general administration of the Plan;
 - (b) to cause to be prepared all forms necessary or appropriate for the administration of the Plan;
 - (c) to keep appropriate books and records;
 - (d) to determine, consistent with the provisions of this instrument all questions of eligibility, rights, and status of Participants under the Plan:

- (e) to issue, amend, and rescind rules relating to the administration of the Plan, to the extent those rules are consistent with the provisions of this instrument;
- (f) to establish uniform rules that shall govern when a Participant shall be continuously employed by the Company and/or an affiliate despite such Participant's leave of absence;
- (g) to exercise all other powers and duties specifically conferred upon the Committee elsewhere in this instrument; and
- (h) to interpret, with discretionary authority, the provisions of this Plan and to resolve, with discretionary authority, all disputed questions of Plan interpretation and benefit eligibility, provided that such discretionary authority shall not apply following a Change in Control with respect to a benefit claim by (or with respect to) an individual who was a Participant at the time of the Change in Control, unless the Participant consents in writing.
 - 7.3 Plan Expenses. Expenses of Plan administration shall be paid by

the Company, and if not paid within a reasonable time period, such expenses shall be paid by the grantor trust that has been established by the Company with respect to the Plan.

- 7.4 Administration. The Committee shall be entitled to rely on all tables, valuations, certificates, opinions, data and reports furnished by any actuary, accountant, controller, counsel or other person employed or retained by the Company with respect to the Plan. The Plan Administrator shall serve without bond and without compensation for services hereunder.
- 7.5 Payment of Benefits. With respect to the discretion of the Plan Administrator and the standard of review applicable to benefit determinations, benefits under this Plan will be paid only if the Plan Administrator or the Company decides in its discretion that the Participant or Claimant is entitled to them, provided that such discretion shall not apply following a Change in Control with respect to a benefit claim by (or with respect to) an individual who was a Participant at the time of the Change in Control, unless the Participant consents in writing.
- (or the cash surrender value thereof) becomes subject to any estate, inheritance, income, employment or other tax which the Company may be required to pay or withhold for or on behalf of the Participant, the Company shall have the full power and authority to withhold and pay such tax out of any moneys or other property in its hand for the benefit of the Participant. To the extent practicable, the Company shall provide the Participant notice of such withholding. Prior to making any payment, the Company may require such releases or other documents from any lawful taxing authority as it shall deem necessary.

7.6 Plan Taxes. If the whole or any part of a Participant's Policy

7.7 Creditor Status. Any funds invested in any trust established with respect to the Plan shall continue for all purposes to be part of the general assets of the Company and available to its general creditors in the event of bankruptcy or insolvency. A Participant's benefits which may be payable pursuant to this Plan are not subject in any manner to anticipation, sale,

alienation, transfer, assignment, pledge, encumbrance, charge, attachment, or garnishment by a Participant, a Participant's beneficiary, or the creditors of either. The Plan constitutes a mere promise by the Company to make benefit payments in the future. No interest or right to receive a benefit may be taken, either voluntarily or involuntarily, for the satisfaction of the debts of, or other obligations or claims against, a Participant, a Participant's beneficiary, or any other person or entity, including claims for alimony, support, separate maintenance and claims in bankruptcy proceedings.

Article VIII - Miscellaneous

- 8.1 Employment Effects. Nothing contained in this Plan or any action taken under the Plan shall be construed as a contract of employment or as giving any Participant any right to be retained in employment with the Company or its affiliates. The Company and its affiliates specifically reserve the right to terminate any Participant's employment at any time with or without cause, and with or without notice or assigning a reason, subject to the terms of any written employment agreement between the Participant and the Company or any affiliate.
- 8.2 Liability and Indemnification. The Company shall indemnify, to the fullest extent permitted by law, the Plan Administrator and directors, officers and employees of the Company and its affiliates, both past and present, to whom are or were delegated duties, responsibilities or authority with respect to the Plan, against any and all claims, losses, liabilities, fines, penalties and expenses (including, but not limited to, all legal fees relating thereto), reasonably incurred by or imposed upon such persons, arising out of any act or omission in connection with the operation and administration of the Plan, other than willful misconduct.
- 8.3 Waiver of Breach. The Company's or the Plan Administrator's waiver of any Plan or Participant Agreement provision shall not operate or be construed as a waiver of any subsequent breach by a Participant.
- 8.4 Gender, Number and Examples. Except where otherwise indicated by the context, in this Plan, the singular or plural number and the masculine, feminine or neuter gender shall be deemed to include the other. Whenever an example is provided or the text uses the term "including" followed by a specific item or items, or there is a passage having a similar effect, such passage of the Plan shall be construed as if the phrase "without limitation" followed such example or term (or otherwise applied to such passage in a manner that avoids limitation on its breadth of application).
- 8.5 Severability. In the event any provision of the Plan shall be held ------illegal or invalid for any reason, the illegality or invalidity shall not affect the remaining parts of the Plan, and the Plan shall be construed and enforced as if the illegal or invalid provision had not been included.
- 8.6 Successors. All obligations and rights of the Company under the Plan shall be binding on and inure to the benefit of any successor to the Company, whether the existence of such successor is the result of a direct or indirect purchase, merger, acquisition, consolidation, affiliation or other corporate restructuring.

- 8.7 Tax Effects. The Company makes no promise, guarantee or warranty,
 express or implied, concerning the federal, state or local income or employment
 tax treatment of any amount of benefits (including the cash surrender value of
 any Participant's Policy) that may be paid to or accrued for the benefit of a
 Participant.
- 8.8 Benefits Provided Through Insurance. Although the Company may assist
 Participants in obtaining life insurance coverage on the life of the
 Participant, the Company is not responsible for paying any life insurance
 benefits which are not paid by the Insurance Company, including whether such
 nonpayment is caused by refusal of the Insurance Company to pay by virtue of a
 legal reason for nonpayment, inability of the Insurance Company to pay, or any
 other reason.
- 8.9 Applicable Law. All questions pertaining to the construction, validity and effect of the Plan shall be determined in accordance with the laws of the United States and to the extent not preempted by such laws, by the laws of the State of Georgia.
- 8.10 Effect on Other Company Benefits. The benefits provided by this Plan shall replace the Participant's benefits provided by the Company to its employees under the basic life insurance, basic accidental death and dismemberment insurance and its retiree life insurance plans, and after becoming a Plan Participant, the Participant shall no longer be eligible to participate in such plans. However, Plan Participants shall remain eligible to participate in any other benefit plan of the Company as provided in such plans, including supplemental life and supplemental accidental death and dismemberment insurance.
- 8.11 Transferred Individuals. Notwithstanding anything in the Plan or the Participant Agreements to the contrary, and as provided by the Employee Benefits Agreement between Equifax Inc. and Certegy Inc., all service and other benefit-affecting determinations for Transferred Individuals that, as of the close of the Distribution Date, were recognized under the Prior Plan for this purpose for periods before the close of the Distribution Date, shall, effective immediately after the close of the Distribution Date, receive full recognition, credit and validity and shall be taken into account under this Plan to the same extent as if such items occurred under the Prior Plan, except to the extent that duplication of benefits would result. Specifically, even though Transferred Individuals obtain a new Commencement Date as of the close of the Distribution Date by becoming Participants in this Plan, (a) the Policy issued as a result of their participation in the Prior Plan shall continue in effect under this Plan and (b) their "Years of Benefit Service" and "Years of Vesting Service" (as defined in Section 3.19) shall include prior service earned as an employee of Equifax Inc.

[signature page follows]

CERTEGY INC.

By:

Name: Title:

GRANTOR TRUST AGREEMENT

THIS GRANTOR TRUST AGREEMENT (the "Trust Agreement") is made this _____ day of July, 2001, by and between Certegy Inc., a Georgia corporation, (the "Company") and Wachovia Bank, N.A. (the "Trustee").

Recitals

- (a) WHEREAS, the Company has adopted the Executive Life and Supplemental Retirement Benefit Plan (the "Plan");
- (b) WHEREAS, the Company has incurred or expects to incur liability under the terms of the Plan with respect to the individuals participating in the Plan and their designated beneficiaries (the "Participants" and "Beneficiaries");
- (c) WHEREAS, the Company hereby establishes this Trust (the "Trust") and shall contribute to the Trust assets that shall be held therein, subject to the claims of the Company's creditors in the event of the Company's Insolvency, as herein defined, until distributed in such manner and at such times as specified in the Plan and in this Trust Agreement;
- (d) WHEREAS, it is the intention of the parties that this Trust shall constitute an unfunded arrangement and shall not affect the status of the Plan as an unfunded plan maintained for the purpose of providing deferred compensation for a select group of management or highly compensated employees for purposes of Title I of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"); and
- (e) WHEREAS, it is the intention of the Company to make contributions to the Trust to provide itself with a source of funds (the "Fund") to assist it in satisfying its liabilities under the Plan in the circumstances described herein.

NOW, THEREFORE, the parties do hereby establish the Trust and agree that the Trust shall be comprised, held and disposed of as follows:

Section 1. Establishment Of The Trust

- (a) The Trust is intended to be a grantor trust, of which the Company is the grantor, within the meaning of subpart E, part I, subchapter J, chapter 1, subtitle A of the Internal Revenue Code of 1986, as amended, and shall be construed accordingly.
- (b) The Company shall be considered the grantor for the purposes of the Trust.
- (c) The Trust hereby established is revocable by the Company; and it shall become irrevocable upon a Change of Control, as defined in Section 15.

- (d) The Company hereby deposits with the Trustee in the Trust

 Dollars (\$______) which shall become the initial principal of the

 Trust to be held, administered and disposed of by the Trustee as provided in this Trust Agreement.
- (e) The principal of the Trust, and any earnings thereon shall be held in the Trust separate and apart from other funds of the Company and shall be used exclusively for the uses and purposes of Participants and general creditors as herein set forth. Participants and their Beneficiaries shall have no preferred claim on, or any beneficial ownership interest in, any assets of the Trust. Any rights created under the Plan and this Trust Agreement shall be unsecured contractual rights of Participants and their Beneficiaries against the Company. Any assets held by the Trust will be subject to the claims of the general creditors of the Company under federal and state law in the event the Company is Insolvent, as defined in Section 3(a) herein.
- (f) The Company, in its sole discretion, may at any time, and from time to time, make additional deposits of cash or other property, including Company stock, acceptable to the Trustee to augment the principal to be held, administered and disposed of by the Trustee as provided in this Trust Agreement. Prior to a Change of Control, neither the Trustee nor any Participant or Beneficiary shall have any right to compel additional deposits.
- (g) Upon a Potential Change of Control (as defined in Section 15), the Company shall, as soon as possible, but in no event longer than thirty (30) days following the occurrence of a Potential Change of Control, make an additional contribution to the Trust, if required, in an amount that is sufficient, when aggregated with the other assets of the Trust, to fund the Trust in an amount equal to no less than 100% but no more than 120% of the amount necessary to pay the insurance premiums required on Policies, as defined herein, purchased pursuant to the Plan, until such Policies have been fully paid, in accordance with Section 2(c) below.
- (h) In the event a Change of Control does not occur within one year of a Potential Change of Control, the Company shall have the right to recover any amounts contributed to and remaining on hand in the Trust.
- (i) Upon a Change of Control, the Company shall, as soon as possible, but in no event longer than thirty (30) days following the occurrence of a Change of Control make an irrevocable contribution to the Trust in any additional amount which is necessary to be sufficient to fund the Trust in an amount equal to no less than 100% but no more than 120% of the amount necessary to pay the insurance premiums required on Policies purchased pursuant to the Plan, until such Policies have been fully paid, in accordance with Section 2(c) below.

Section 2. Payments From The Trust

- (a) Prior to a Change of Control, distributions from the Trust shall be made by the Trustee to the insurance company identified in or pursuant to Section 2(e) below (the "Insurance Company") at the direction of the Company.
- (b) As insurance premiums become due with respect to the life insurance policies (each a "Policy") purchased pursuant to the Plan on the lives of the Participants, the Company shall (i) pay such insurance premiums directly to the Insurance Company, (ii) transfer to the Trustee within thirty (30) days prior to the premium due date funds sufficient to allow the Trustee to pay to the Insurance Company such insurance premiums, or (iii) direct the Trustee to pay directly to the Insurance Company such insurance premiums from the Fund.
- (c) (1) After a Potential Change of Control and before a Change of Control, the Company shall deliver to the Trustee a schedule of insurance premiums due under the Plan. Subsequent to a Change of Control, the Trustee shall pay insurance premiums due in accordance with such schedule. If the Company has not transferred the required amounts at least thirty (30) days prior to each due date, the Trustee shall make such payments from the assets of the Fund. If the principal of the Trust, and any earnings thereon, are not sufficient to make payments of insurance premiums in accordance with the terms of such schedule, the Company shall make the balance of each such payment as it falls due in accordance with such Schedule. The Trustee shall notify the Company in the event that principal and earnings are not sufficient to make any premium payment. Nothing in this Trust Agreement shall relieve the Company of its liabilities to pay benefits due under the Plan except to the extent such liabilities are met by application of assets of the Trust.
 - (2) Subsequent to a Change of Control, if the Company borrows any portion of the cash surrender value of any Policy, the Trustee shall immediately repay to the Insurance Company any amount that has been so borrowed, as certified to it by the Participant whose Policy is the subject of the loan. The Trustee may request any further reasonable evidence of such a loan.
 - (3) Subsequent to a Change of Control, if the Trustee becomes aware that the Company withdraws any portion of the cash surrender value of any Policy, the Trustee shall consult with the Insurance Company or the broker of record, as it deems appropriate, to determine the maximum premiums which may be paid on an annual basis to restore any such withdrawal and to retain the life insurance nature of the Policy, and shall make said payments.
- (d) The Trustee may institute an action to collect a contribution due the Trust following a Change of Control or in the event that the Trust should ever experience a short-fall in the amount of assets necessary to make current payments pursuant to the terms of the Plan.

(e) The primary purposes of this Trust are to insure (i) that, following a Change of Control, premiums will continue to be paid to Pacific Life Insurance Company, or such successor company as the Company may identify to the Trustee in writing, as required pursuant to the Plan and all splitdollar life insurance agreements with employees of the Company or its subsidiaries which have been entered into by the Company and Plan Participants pursuant to the Plan, and (ii) that any successor to the Company, or its successor management, does not withdraw cash values from the Policies prior to the respective distribution dates of said Policies. Prior to a Change of Control, the payment of Policy insurance premiums will be made pursuant to the provisions of Section 2(b). Subsequent to a Change of Control, the Trustee shall make such payments unless the Company has previously certified to having made them, according to the provisions hereof. In order to make such payments, the Trustee may be required to sell all or a portion of any assets held in the Fund. In the event that the Fund includes Company stock, the Company hereby agrees to promptly, and in any event within sixty (60) days of a request for registration by the Trustee, take any and all actions necessary to register the Company stock held in the fund for sale and to maintain on a continuous basis any registrations required to permit said sales pursuant to applicable federal and state laws, until all Company stock has been sold. In connection with any such securities registrations, the Company shall take any and all actions necessary in connection therewith, including without limitation: (i) causing any special audits to be performed, if required and (ii) if requested by the Trustee, entering into an underwriting agreement with underwriters selected by the Trustee in customary form including providing indemnification for the underwriters and the Trustee. Any and all costs arising in connection with the filing of any securities registrations, including the fees and disbursements of counsel for the Trustee, shall be borne entirely by the Company other than underwriting discounts and commissions or commissions of broker dealers which shall be payable by the Trustee from the assets of the Trust. The Company consents that an action $% \left(1\right) =\left(1\right) \left(1\right)$ may be brought in equity or in law by the Trustee or by any Participant in the Plan, to compel its compliance with the provisions of this Trust, including but not limited to the foregoing sentence and the provisions of Section 2(d) above.

Section 3. Trustee Responsibility Regarding Payments When The Company Is Insolvent

- (a) The Trustee shall cease payment of insurance premiums to the Insurance Company if the Company is Insolvent. The Company shall be considered "Insolvent" for purposes of this Trust Agreement if (i) the Company is unable to pay its debts as they become due, (ii) the Company is subject to a pending proceeding as a debtor under the United States Bankruptcy Code or (iii) the Company is determined to be insolvent by the Federal Deposit Insurance Corporation, the Federal Reserve, or the Office of the Comptroller of Currency.
- (b) At all times during the continuance of this Trust, the principal and income of the Trust shall be subject to claims of general creditors of the Company under federal and state law as set forth below.

- (1) The Board of Directors and the Chief Executive Officer of the Company shall have the duty to inform the Trustee in writing that the Company is Insolvent. If a person claiming to be a creditor of the Company alleges in writing to the Trustee that the Company has become Insolvent, the Trustee shall determine whether the Company is Insolvent and, pending such determination, the Trustee shall discontinue payment of insurance premiums to the Insurance Company.
- (2) Unless the Trustee has actual knowledge that the Company is Insolvent, or has received notice from the Company or a person claiming to be a creditor alleging that the Company is Insolvent, the Trustee shall have no duty to inquire whether the Company is Insolvent. The Trustee may in all events rely on such evidence concerning the Company's solvency as may be furnished to the Trustee and that provides the Trustee with a reasonable basis for making a determination concerning the Company's solvency.
- (3) If at any time the Trustee has determined that the Company is Insolvent, the Trustee shall discontinue paying insurance premiums to the Insurance Company and shall hold the assets of the Trust for the benefit of the Company's general creditors. Nothing in this Trust Agreement shall in any way diminish any rights of Participants or their Beneficiaries to pursue their rights as general creditors of the Company with respect to payments due under the Plan or otherwise.
- (4) The Trustee shall resume the payment of insurance premiums to the Insurance Company in accordance with Section 2 of this Trust Agreement only after the Trustee has determined that the Company is not Insolvent (or is no longer Insolvent).
- (c) Provided that there are sufficient assets, if the Trustee discontinues the payment of insurance premiums from the Trust pursuant to Section 3(b) hereof and subsequently resumes such payments, the first payment following such discontinuance shall include the aggregate amount of all payments due to the Insurance Company under the terms of the Plan for the period of such discontinuance, less the aggregate amount of any payments made to the Insurance Company by the Company in lieu of the payments provided for hereunder during any such period of discontinuance.

Section 4. Payments When A Shortfall Of The Trust Assets Occurs

(a) If there are not sufficient assets for the payment of insurance premiums pursuant to Section 2 or Section 3(c) hereof and the Company does not otherwise make such payments within a reasonable time after demand from the Trustee, the Trustee shall make payment of insurance premiums from the Trust to the Insurance Company for the benefit of Participants and their Beneficiaries in the following order of priority:

- (1) All Policies should be funded based on original expected performance, with premiums adequate to keep the Policies in force until the insured attains age 100; and
- (2) Any remaining funding should be made pro-rata based upon remaining scheduled premium payments.

It is understood that it is not possible to anticipate precisely future financial status of the Policies, and the contingencies that could occur both before and after a Change of Control. Therefore, the Trustee will have discretion to implement any reasonable method of allocating Trust assets that are, in its sole discretion, determined to ensure complete funding of the Policies pursuant to the premium schedule provided. The Trustee may rely solely on the services of the broker of record as well as any other sources in making this determination.

(b) Upon receipt of a contribution from the Company necessary to make up for a shortfall in the payments due, the Trustee shall resume payments to the Insurance Company under the Plan. Following a Change of Control, the Trustee shall have the right to compel a contribution to the Trust from the Company to make up for any shortfall.

Section 5. Payments To The Company

Except as provided in Section 3 hereof in the event the Company is Insolvent, after the Trust has become irrevocable (as provided in Section 1) the Company shall have no right or power to direct the Trustee to return to the Company or to divert to others any of the Trust assets before all payment of insurance premiums have been made to the Insurance Company pursuant to the terms of the Plan.

Section 6. Investment Authority

- (a) Consistent with the provisions of Section 10(a) below, the Trustee shall not be liable in discharging its duties hereunder, including, without limitation, its duty to invest and reinvest the Fund, if it acts for the exclusive benefit of the Participants and their Beneficiaries, in good faith and as a prudent person would act in accomplishing a similar task and in accordance with the terms of this Trust Agreement and any applicable federal or state laws, rules or regulations.
- b) Subsequent to a Change of Control, the Trustee shall have the following powers, in investing and reinvesting the Fund, in its sole discretion:
 - (1) To invest and reinvest in any readily marketable common and preferred stocks, bonds, notes, debentures (including convertible stocks and securities but not including any stock or security of the Trustee other than a de minimis amount held in a collective or mutual fund), certificates of deposit or demand or time deposits (including any such deposits with the Trustee) and shares of investment companies and mutual funds, without being limited to the classes or property in

which the Trustee is authorized to invest by any law or any rule of court of any state and without regard to the proportion any such property may bear to the entire amount of the Fund. Without limitation, the Trustee may invest the Trust in any investment company (including any investment company or companies for which the Trustee or an affiliated company acts as the investment advisor ("Special Investment Companies") or, any insurance contract or contracts issued by an insurance company or companies in each case as the Trustee may determine provided that the Trustee may in its sole discretion keep such portion of the Trust in cash or cash balances for such reasonable periods as may from time to time be deemed advisable pending investment or in order to meet contemplated payments of insurance premiums;

- (2) To commingle for investment purposes all or any portion of the Fund with assets of any other similar trust or trusts established by the Company with the Trustee for the purpose of safeguarding deferred compensation or retirement income benefits of its employees and/or directors;
- (3) To retain any property at any time received by the Trustee;
- (4) To sell or exchange any property held by it at public or private sale, for cash or on credit, to grant and exercise options for the purchase or exchange thereof, to exercise all conversion or subscription rights pertaining to any such property and to enter into any covenant or agreement to purchase any property in the future;
- (5) To participate in any plan of reorganization, consolidation, merger, combination, liquidation or other similar plan relating to property held by it and to consent to or oppose any such plan or any action thereunder or any contract, lease, mortgage, purchase, sale or other action by any person;
- (6) To deposit any property held by it with any protective, reorganization or similar committee, to delegate discretionary power thereto, and to pay part of the expenses and compensation thereof any assessments levied with respect to any such property to be deposited;
- (7) To extend the time of payment of any obligation held by it;
- (8) To hold uninvested any monies received by it, without liability for interest thereon, but only in anticipation of payments due for investments, reinvestments, expenses or disbursements;
- (9) To exercise all voting or other rights with respect to any property held by it and to grant proxies, discretionary or otherwise;
- (10) For the purposes of the Trust, to borrow money from others, to issue its promissory note or notes therefor, and to secure the repayment thereof by pledging any property held by it;

- (11) To employ suitable contractors and counsel, who may be counsel to the Company or to the Trustee, and to pay their reasonable expenses and compensation from the Fund to the extent not paid by the Company;
- (12) To register investments in its own name or in the name of a nominee; to hold any investment in bearer form; and to combine certificates representing securities with certificates of the same issue held by it in other fiduciary capacities or to deposit or to arrange for the deposit of such securities with any depository, even though, when so deposited, such securities may be held in the name of the nominee of such depository with other securities deposited therewith by other persons, or to deposit or to arrange for the deposit of any securities issued or guaranteed by the United States government, or any agency or instrumentality thereof, including securities evidenced by book entries rather than by certificates, with the United States Department of the Treasury or a Federal Reserve Bank, even though, when so deposited, such securities may not be held separate from securities deposited therein by other persons; provided, however, that no securities held in the Fund shall be deposited with the United States Department of the Treasury or a Federal Reserve Bank or other depository in the same account as any individual property of the Trustee, and provided, further, that the books and records of the Trustee shall at all times show that all such securities are part of the Trust Fund;
- (13) To settle, compromise or submit to arbitration any claims, debts or damages due or owing to or from the Trust, respectively, to commence or defend suits or legal proceedings to protect any interest of the Trust, and to represent the Trust in all suits or legal proceedings in any court or before any other body or tribunal; provided, however, that the Trustee shall not be required to take any such action unless it shall have been indemnified by the Company to its reasonable satisfaction against liability or expenses it might incur therefrom;
- (14) To hold and retain policies of life insurance, annuity contracts, and other property of any kind which policies are contributed to the Trust by the Company or any subsidiary of the Company or are purchased by the Trustee;
- (15) To hold any other class of assets which may be contributed by the Company and that is deemed reasonable by the Trustee, unless expressly prohibited herein;
- (16) To loan any securities at any time held by it to brokers or dealers upon such security as may be deemed advisable, and during the terms of any such loan to permit the loaned securities to be transferred into the name of and voted by the borrower or others; and
- (17) Generally, to do all acts, whether or not expressly authorized, that the Trustee may deem necessary or desirable for the protection of the Fund.

- (c) Prior to a Change of Control, the Company shall have the right, subject to this Section, to direct the Trustee with respect to investments. Absent any such direction, the Trustee shall continue the investment of the Fund as provided in this section.
 - (1) The Company may at any time direct the Trustee to segregate all or a portion of the Fund in a separate investment account or accounts and may appoint one or more investment managers and/or an investment committee established by the Company to direct the investment and reinvestment of each such investment account or accounts. In such event, the Company shall notify the Trustee of the appointment of each such investment manager and/or investment committee. No such investment manager shall be related, directly or indirectly, to the Company, but members of the investment committee may be employees of the Company.
 - (2) Thereafter, the Trustee shall make every sale or investment with respect to such investment account as directed in writing by the investment manager or investment committee. It shall be the duty of the Trustee to act strictly in accordance with each direction. The Trustee shall be under no duty to question any such direction of the investment manager or investment committee, to review any securities or other property held in such investment account or accounts acquired by it pursuant to such directions or to make any recommendations to the investment manager or investment committee with respect to such securities or other property.
 - (3) Notwithstanding the foregoing, the Trustee, without obtaining prior approval or direction from an investment manager or investment committee, shall invest cash balances held by it from time to time in short term cash equivalents including, but not limited to, through the medium of any short term common, collective or commingled trust fund established and maintained by the Trustee subject to the instrument establishing such trust fund, U.S. Treasury Bills, commercial paper (including such forms of commercial paper as may be available through the Trustee's Trust Department), certificates of deposit (including certificates issued by the Trustee in its separate corporate capacity), and similar type securities, with a maturity not to exceed one year; and, furthermore, sell such short term investments as may be necessary to carry out the instructions of an investment manager or investment committee regarding more permanent type investment and directed distributions.
 - (4) The Trustee shall neither be liable nor responsible for any loss resulting to the Fund by reason of any sale or purchase of an investment directed by an investment manager or investment committee nor by reason of the failure to take any action with respect to any investment which was acquired pursuant to any such direction in the absence of further directions of such investment manager or investment committee.
 - (5) Notwithstanding anything in this Agreement to the contrary, the Trustee shall be indemnified and saved harmless by the Company from and against any and all

personal liability to which the Trustee may be subjected by carrying out any directions of an investment manager or investment committee issued pursuant hereto or for failure to act in the absence of directions of the investment manager or investment committee including all expenses reasonably incurred in its defense in the event the Company fails to provide such defense; provided, however, the Trustee shall not be so indemnified if it participates knowingly in, or knowingly undertakes to conceal, an act or omission of an investment manager or investment committee, having actual knowledge that such act or omission is a breach of a fiduciary duty; provided further, however, that the Trustee shall not be deemed to have knowingly participated in or knowingly undertaken to conceal an act or omission of an investment manager or investment committee with knowledge that such act or omission was a breach of fiduciary duty by merely complying with directions of an investment manager or investment committee or for failure to act in the absence of directions of an investment manager or investment committee. The Trustee may rely upon any order, certificate, notice, direction or other documentary confirmation purporting to have been issued by the investment manager or investment committee which the Trustee believes to be genuine and to have been issued by the investment manager or investment committee. The Trustee shall not be charged with knowledge of the termination of the appointment of any investment manager or investment committee until it receives written notice thereof from the Company.

- (d) Following a Change of Control, the Trustee shall have the sole and absolute discretion in the management of the Trust assets and shall have all the powers set forth under Section 6(b). In investing the Trust assets, the Trustee shall consider:
 - (1) the needs of the Plan;
 - (2) the need for matching of the Trust assets with the liabilities of the Plan; and
 - (3) the duty of the Trustee to act solely in the best interests of the Participants and their Beneficiaries.
- (e) The Trustee shall have the right, in its sole discretion, to delegate its investment responsibility to an investment manager who may be an affiliate of the Trustee. In the event the Trustee shall exercise this right, the Trustee shall remain, at all times responsible for the acts of an investment manager. The Trustee shall have the right to purchase an insurance policy or an annuity to fund the benefits of the Plan.
- (f) Prior to a Change of Control, the Company shall have the right at any time, and from time to time in its sole discretion, to substitute assets of equal fair market value for any asset held by the Trust. This right is exercisable by the Company in a nonfiduciary capacity without the approval or consent of any person in a fiduciary capacity.

Section 7. Insurance Contracts

- (a) To the extent that the Trustee is directed by the Company prior to a Change of Control to make payments from part or all of the Trust Fund in insurance contracts, the type and amount thereof shall be specified by the Company. The Trustee shall be under no duty to make inquiry as to the propriety of the type or amount so specified.
- (b) Each insurance contract issued shall provide that the owner thereof shall have the power to exercise all rights, privileges, options and elections granted by or permitted under such contract or under the rules of the insurer.
- (c) The Trustee shall have no power to name a beneficiary of the policy to assign the policy (as distinct from conversion of the policy to a different form), or to loan to any person the proceeds of any borrowing against such an insurance policy.
- (d) No insurer shall be deemed to be a party to the Trust and an insurer's obligations shall be measured and determined solely by the terms of contracts and other agreements executed by the insurer.

Section 8. Disposition Of Income

- (a) Prior to a Change of Control, all income received by the Trust, net of expenses and taxes, may be returned to the Company or accumulated and reinvested within the Trust at the direction of the Company.
- (b) Following a Change of Control, all income received by the Trust, net of expenses and taxes, shall be accumulated and reinvested within the Trust.

Section 9. Accounting By The Trustee

The Trustee shall keep accurate and detailed records of all investments. receipts, disbursements, and all other transactions required to be made, including such specific records as shall be agreed upon in writing between the Company and the Trustee within forty-five (45) days following the close of each calendar year and within forty-five (45) days after the removal or resignation of the Trustee. The Trustee shall deliver to the Company a written account of its administration of the Trust during such year or during the period from the close of the last preceding year to the date of such removal or resignation setting forth all investments, receipts, disbursements and other transactions effected by it, including a description of all securities and investments purchased and sold with the cost or net proceeds of such purchases or sales (accrued interest paid or receivable being shown separately), and showing all cash, securities and other property held in the Trust at the end of such year or as of the date of such removal or resignation, as the case may be. The Company may approve such written account by an instrument in writing delivered to the Trustee. In the absence of the Company's filing with the Trustee objections to any such written account within ninety (90) days after its receipt, the Company shall be deemed to have so approved such written account. In such case, or upon the written approval by the Company of any such written account, the Trustee shall, to the extent permitted by law, be discharged from all liability to the Company for its acts or failures to act described by such written account. The

foregoing, however, shall not preclude the Trustee from having its accounting settled by a court of competent jurisdiction.

Section 10. Responsibility Of The Trustee

- (a) The Trustee shall act with the care, skill, prudence and diligence under the circumstances then prevailing that a prudent person acting in like capacity and familiar with such matters would use in the conduct of an enterprise of a like character and with like aims, provided, however, that the Trustee shall incur no liability to any person for any action taken pursuant to a direction, request or approval given by the Company which is contemplated by, and in conformity with, the terms of the Plan or this Trust Agreement and is given in writing by the Company. In the event of a dispute between the Company and a party, the Trustee may apply to a court of competent jurisdiction to resolve the dispute, subject, however to Section 2(d) hereof.
- (b) The Company hereby indemnifies the Trustee against losses, liabilities, claims, costs and expenses in connection with the administration of the Trust, unless resulting from the gross negligence or misconduct of Trustee. To the extent the Company fails to make any payment on account of an indemnity provided in this Section 10(b), in a reasonably timely manner, the Trustee may obtain payment from the Trust. If the Trustee undertakes or defends any litigation arising in connection with this Trust or to protect a Participant's or Beneficiary's rights under the Plan, the Company agrees to indemnify the Trustee against the Trustee's costs, reasonable expenses and liabilities (including, without limitation, attorneys' fees and expenses) relating thereto and to be primarily liable for such payments. If the Company does not pay such costs, expenses and liabilities in a reasonably timely manner, the Trustee may obtain payment from the Trust.
- (c) Prior to a Change of Control, the Trustee may consult with legal counsel (who may also be counsel for the Company generally) with respect to any of its duties or obligations hereunder. Following a Change of Control the Trustee shall select legal counsel independent from the Company's counsel and may consult with counsel or other experts with respect to its duties and with respect to the rights of Participants or their Beneficiaries under the Plan.
- (d) The Trustee may hire agents, accountants, actuaries, investment advisors, financial consultants or other professionals to assist it in performing any of its duties or obligations hereunder and may rely on any determinations made by such agents and information provided to it by the Company.
- (e) The Trustee shall have, without exclusion, all powers conferred on the Trustee by applicable law, unless expressly provided otherwise herein.
- (f) Notwithstanding any powers granted to the Trustee pursuant to this Trust Agreement or to applicable law, the Trustee shall not have any power that could give this Trust the objective of carrying on a business and dividing the gains therefrom.

The Trustee's compensation shall be as agreed in writing from time to time by the Company and the Trustee. The Company shall pay all administrative expenses and the Trustee's fees and shall promptly reimburse the Trustee for any fees and expenses of its agents or such other costs as the Trustee is entitled to incur hereunder. If not so paid, the fees and expenses shall be paid from the Trust.

Section 12. Resignation And Removal Of The Trustee

- (a) Prior to a Change of Control, the Trustee may resign at any time by written notice to the Company, which shall be effective sixty (60) days after receipt of such notice unless the Company and the Trustee agree otherwise. Following a Change of Control, if the Trustee resigns, the resignation shall only be effective after the appointment of a successor Trustee.
- (b) The Trustee may be removed by the Company on sixty (60) days notice or upon shorter notice accepted by the Trustee prior to a Change of Control. Subsequent to a Change of Control, the Trustee may only be removed by the Company with the consent of a majority of the Participants, after they have been informed of the identity of a successor trustee.
- (c) If the Trustee resigns within two years after a Change of Control, and if the Company fails to act under Section 10(e) below within a reasonable period of time following such resignation, the Trustee shall apply to a court of competent jurisdiction for the appointment of a successor Trustee or instructions.
- (d) Upon resignation or removal of the Trustee and appointment of a successor Trustee, all assets shall subsequently be transferred to the successor Trustee. The transfer shall be completed within sixty (60) days after receipt of notice of resignation, removal or transfer, unless the Company extends the time limit.
- (e) If the Trustee resigns or is removed, a successor shall be appointed by the Company, in accordance with Section 13 hereof, by the effective date of resignation or removal under Sections 10(a) or 10(b) above. If no such appointment has been made, the Trustee may apply to a court of competent jurisdiction for appointment of a successor or for instructions. All expenses of the Trustee in connection with the proceeding shall be allowed as administrative expenses of the Trust.

Section 13. Appointment Of Successor

(a) If the Trustee resigns or is removed in accordance with Section 12 hereof, the Company may appoint, subject to Section 12, another bank, not an affiliate of the Company or any other grantor, any third party national banking association with a market capitalization exceeding \$100,000,000 to replace the Trustee upon resignation or removal. The successor Trustee shall have all of the rights and powers of the former Trustee, including

ownership rights in the Trust. The former Trustee shall execute any instrument necessary or reasonably requested by the Company or the successor Trustee to evidence the transfer.

(b) The successor Trustee need not examine the records and acts of any prior Trustee and may retain or dispose of existing Trust assets, subject to Sections 8 and 9 hereof. The successor Trustee shall not be responsible for and the Company shall indemnify and defend the successor Trustee from any claim or liability resulting from any action or inaction of any prior Trustee or from any other past event, or any condition existing at the time it becomes successor Trustee.

Section 14. Amendment Or Termination

- (a) Prior to a Change of Control, this Trust Agreement may be amended by a written instrument executed by the Trustee and the Company. Notwithstanding the foregoing, no such amendment shall conflict with the terms of the Plan or shall make the Trust revocable after it has become irrevocable in accordance with Section 1 hereof.
- (b) The Trust shall not terminate until the date on which all insurance premiums listed on the schedule referred to in Section 2(c)(1) have been paid or otherwise satisfied, and any payments required under Section 2(c)(3) are completed, or until the Company terminates the Trust (if prior to a Change of Control).
- (c) Prior to a Change of Control, the Company may terminate this Trust at any time, including prior to the time all benefit payments under the Plan have been made. All assets in the Trust at termination shall be returned to the Company.
- (d) This Trust Agreement may not be amended or terminated by the Company for seven (7) years following a Change of Control without the written consent of a majority of the Participants except, if in the opinion of counsel satisfactory to the Trustee, such amendment is necessary to maintain the tax status of this Trust or the inapplicability of ERISA to this Trust.

Section 15. Change Of Control

- - (1) "Potential Change of Control" shall mean the occurrence of any one of the following events:
 - (i) the purchase or other acquisition by any Person of Beneficial Ownership of five percent (5%) or more of either the outstanding shares of common stock or the combined voting power of the Company's then outstanding voting securities entitled to vote generally; provided, however, the purchase or other acquisition by any employee benefit plan (or related trust) sponsored or maintained by (I) Equifax Inc. (to the extent the

acquisition occurs as part of the initial distribution of Company shares on Equifax Inc. shares held by the plan), (II) the Company, or (III) any Subsidiary of the Company, shall be disregarded for purposes of this Section 15(a)(1)(i);

- (ii) the announcement by any person of an intention to take actions which might reasonably result in a Change of Control of the Company;
- (iii) the issuance of a proxy statement by the Company with respect to an election of directors for which there is proposed one or more directors who are not recommended by the Board of Directors of the Company or its nominating committee, where the election of such proposed director or directors would result in a Change of Control; or
- (iv) submission to the Incumbent Board (as defined below) of nominations which, if approved, would change the Executive Officer configuration of the Company (at the Executive Vice President level and above) by fifty percent (50%) or more.
- (2) "Change of Control" shall mean the occurrence of any one of the following events:
 - (i) Voting Stock Accumulations. The accumulation by any Person of

 Beneficial Ownership of twenty percent (20%) or more of the
 combined voting power of the Company's Voting Stock; provided
 that for purposes of this Section15(a)(2)(i), a Change of
 Control will not be deemed to have occurred if the accumulation
 of twenty percent (20%) or more of the voting power of the
 Company's Voting Stock results from any acquisition of Voting
 Stock (I) directly from the Company that is approved by the
 Incumbent Board, (II) by the Company, (III) by any employee
 benefit plan (or related trust) sponsored or maintained by the
 Company or any Subsidiary, or (IV) by any Person pursuant to a
 Business Combination that complies with all of the provisions of
 clauses (I), (II) and (III) of Section 15(a)(2)(ii);
 - Combination, unless, immediately following that Business Combination, (I) all or substantially all of the Persons who were the beneficial owners of Voting Stock of the Company immediately prior to that Business Combination beneficially own, directly or indirectly, more than sixty-six and two-thirds percent (66-2/3%) of the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors of the entity resulting from that Business Combination (including, without limitation, an entity that as a result of that transaction owns the Company or all or substantially all of the Company's assets either directly or through one or

more subsidiaries) in substantially the same proportions relative to each other as their ownership, immediately prior to that Business Combination, of the Voting Stock of the Company, (II) no Person (other than the Company, that entity resulting from that Business Combination, or any employee benefit plan (or related trust) sponsored or maintained by the Company, any Eighty Percent (80%) Subsidiary or that entity resulting from that Business Combination) beneficially owns, directly or indirectly, twenty percent (20%) or more of the then outstanding shares of common stock of the entity resulting from that Business Combination or the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors of that entity, and (III) at least a majority of the members of the Board of Directors of the entity resulting from that Business Combination were members of the Incumbent Board at the time of the execution of the initial agreement or of the action of the board providing for that Business Combination;

- (iii) Sale of Assets. A sale or other disposition of all or substantially all of the assets of the Company; or
- (iv) Liquidation or Dissolutions. Approval by the shareholders of the Company of a complete liquidation or dissolution of the Company, except pursuant to a Business Combination that complies with all of the provisions of clauses (I), (II) and (III) of Section 15(a)(2)(ii).

For purposes of this Section 15(a), the following definitions will apply:

"Beneficial Ownership" means beneficial ownership as that term is used in Rule 13d-3 promulgated under the Exchange Act.

"Business Combination" means a reorganization, merger or consolidation of the Company.

"Eighty Percent (80%) Subsidiary" means an entity in which the Company directly or indirectly beneficially owns eighty percent (80%) or more of the outstanding Voting Stock.

"Exchange Act" means the Securities Exchange Act of 1934, including amendments, or successor statutes of similar intent.

"Incumbent Board" means a Board of Directors at least a majority of whom consist of individuals who either are (a) members of the Company's Board of Directors as of the day after the spinoff of the Company from Equifax Inc. becomes effective or (b) members who become members of the Company's Board of Directors subsequent to said date whose election, or nomination for election by the Company's shareholders, was approved by a vote of at least two-thirds (2/3) of the directors then comprising the Incumbent Board (either by a specific vote or by approval of the proxy statement of the

Company in which that person is named as a nominee for director, without objection to that nomination), but excluding, for that purpose, any individual whose initial assumption of office occurs as a result of an actual or threatened election contest (within the meaning of Rule 14a-11 of the Exchange Act) with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board of Directors.

"Person" means any individual, entity or group (within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act).

"Subsidiary" means an entity more than fifty percent (50%) of whose equity interests are owned directly or indirectly by the Company.

"Voting Stock" means the then outstanding securities of an entity entitled to vote generally in the election of members of that entity's Board of Directors.

For purposes of this Section 15(a), the Incumbent Board, by a majority vote, shall have the power to determine on the basis of information known to them (a) the number of shares beneficially owned by any person, entity or group; (b) whether there exists an agreement, arrangement or understanding with another as to matters referred to in this Section 15(a); and (c) such other matters with respect to which a determination is necessary under this Section 15(a).

(b) The General Counsel of the Company shall have the specific authority to determine whether a Potential Change of Control or Change of Control has transpired under the guidance of this Section 15(a) and shall be required to give the Trustee notice of a Change of Control or Potential Change of Control. The Trustee shall be entitled to rely upon such notice, but if the Trustee receives notice of a Potential Change of Control or Change of Control from another source, the Trustee shall be required to make its own independent determination.

Section 16. Miscellaneous

- (a) Any provision of this Trust Agreement prohibited by law shall be ineffective to the extent of any such prohibition, without invalidating the remaining provisions hereof.
- (b) The Company hereby represents and warrants that the Plan has been established, maintained and administered in accordance with all applicable law, including without limitation, ERISA. The Company hereby indemnifies and agrees to hold the Trustee harmless from all liabilities, including attorney's fees, relating to or arising out of the establishment, maintenance and administration of the Plan. To the extent the Company does not pay any of such liabilities in a reasonably timely manner, the Trustee may obtain payment from the Trust.
- (c) Benefits payable to Participants and their Beneficiaries under this Trust Agreement may not be anticipated, assigned (either at law or in equity), alienated, pledged, encumbered

or subjected to attachment, garnishment, levy, execution or other legal or equitable process.

- (d) This Agreement is binding upon the successors and assigns of the Company and the Trustee.
- (e) This Trust Agreement shall be governed by and construed in accordance with the laws of North Carolina.

IN WITNESS WHEREOF, this Grantor Trust Agreement has been executed on behalf of the parties hereto on the day and year first above written.

CERTEGY INC.

By:
Name: Title:
WACHOVIA BANK, N.A.
By:
Name: Title:

AGREEMENT REGARDING LEASES

THIS AGREEMENT REGARDING LEASES (this "Agreement") is made and entered into as of the _____ day of _____, 2001, by and between EQUIFAX INC., a Georgia corporation ("Equifax"), and EQUIFAX PAYMENT SERVICES, INC., a Delaware corporation ("the name of which entity will be changed to "Certegy Payment Services, Inc.")("Certegy Payment Services").

WITNESSETH:

WHEREAS, Certegy Payment Services is currently a wholly-owned subsidiary of Equifax;

WHEREAS, Equifax presently intends to transfer and assign to Certegy Inc., a Georgia corporation ("Certegy"), as a contribution to the capital of Certegy, the capital stock of Certegy Payment Services and certain related assets (the "Spin Transaction"); and

WHEREAS, each of Certegy Payment Services and its subsidiaries (collectively, the "Certegy Payment Services Group"; the persons and entities composing the Certegy Payment Services Group are herein referred to as "Certegy Payment Services Group Companies") and Equifax and its subsidiaries other than those that compose the Certegy Payment Services Group Companies (collectively, the "Equifax Group"; the persons and entities composing the Equifax Group are herein referred to as "Equifax Group Companies") have entered into certain Lease Agreements and Lease Guarantees, and the parties hereto desire to agree upon certain matters with respect to such Lease Agreements and Lease Guarantees in connection with the Spin Transaction, as set forth in this Agreement.

NOW, THEREFORE, for and in consideration of the sum of Ten and No/100 Dollars (\$10.00), the foregoing premises and the respective undertakings of the parties, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Equifax and Certegy Payment Services hereby agree as follows:

- 1. Certegy Payment Services Group Leases; Equifax Guarantees.
 - (a) Certegy Payment Services Group Leases. Certain Certegy

Payment Services Group Companies are currently the "tenants" or "lessees" under those certain Lease Agreements identified on Exhibit "A" attached hereto and

incorporated herein (the "Certegy Payment Services Group Leases"), and such Certegy Payment Services Group Companies will continue to be the "tenants" or "lessees" under their Lease Agreements immediately after the consummation of the Spin Transaction. The parties intend that the Spin Transaction shall not constitute an assignment or other transfer under any of the Certegy Payment Services Group Leases that would require the consent of the "landlord" or "lessor" thereunder, but, in the event the Spin Transaction is deemed to be such an assignment or other transfer under any of the Certegy Payment Services Group Leases, then Equifax and Certegy Payment Services hereby agree to reasonably cooperate with each other in obtaining any such required consent of the "landlord" or "lessor" thereunder. Certegy Payment Services shall, and hereby agrees to, indemnify, defend and hold Equifax harmless from, against and in respect of any

actions, causes of action, suits, claims, demands, judgments, or proceedings asserted against, imposed upon or suffered or incurred by Equifax, and from, against and in respect of any liabilities, damages, losses, costs, expenses (including counsel fees and expenses and disbursements of counsel), amounts of judgment, assessments, fines or penalties, and amounts paid in compromise or settlement, asserted against, imposed upon or suffered or incurred by Equifax, in connection with or arising out of or by reason of any one or more of the Certegy Payment Services Group Leases in connection with the Spin Transaction or otherwise.

(b) Equifax Guarantees. Equifax has entered into lease guarantees

(the "Equifax Guarantees") with respect to the duties, obligations, liabilities, and responsibilities of the "tenant" or "lessee" under certain of the Certegy Payment Services Group Leases, as identified on Exhibit "B" attached hereto and

incorporated herein. Certegy Payment Services shall, and hereby agrees to, indemnify, defend and hold Equifax harmless from, against and in respect of any actions, causes of action, suits, claims, demands, judgments, or proceedings asserted against, imposed upon or suffered or incurred by Equifax, and from, against and in respect of any liabilities, damages, losses, costs, expenses (including counsel fees and expenses and disbursements of counsel), amounts of judgment, assessments, fines or penalties, and amounts paid in compromise or settlement, asserted against, imposed upon or suffered or incurred by Equifax in connection with or arising out of or by reason of any one or more of the Equifax Guarantees. Without limiting any of the rights or remedies of Equifax under the bistribution Agreement (as hereinafter defined) or otherwise, Equifax shall be subrogated to the right, title, and interest of the "landlord" or "lessor" under each of the Equifax Guarantees and all rights or remedies of the "landlord" or "lessor" thereunder to the extent Equifax is required to honor any such Equifax Guarantees and to the extent permitted under the terms of the applicable Certegy Payment Services Group Lease.

- 2. Equifax Group Leases; Subletting; Assignment.
- (a) Equifax Group Leases. Certain Equifax Group Companies are

currently the "tenants" or "lessees" under those certain Lease Agreements identified on Exhibit "C" attached hereto and incorporated herein (the "Equifax

Group Leases"), and all or a portion of the premises under each of such Equifax Group Leases are currently occupied by one or more Certegy Payment Services Group Companies. The parties desire that portions of the premises under certain Equifax Group Leases be sublet, and that certain of the Equifax Group Leases be assigned, to the Certegy Payment Services Group Company currently occupying all or a portion of the premises thereunder, as set forth in subparagraphs (b) and (c), below.

(b) Subletting. Certegy Payment Services Group Companies currently

occupy all or a portion of the premises under each of the Equifax Group Leases identified as "to be sublet" on Exhibit "C" hereto, which occupied portion of

each such premises is more particularly described on said Exhibit "C". Each of

the Equifax Group Companies that is the "tenant" or "lessee" under each such Equifax Group Lease (each as a "Sublandlord" hereunder) does hereby sublease to the Certegy Payment Services Company identified on said Exhibit "C" as the

"Subtenant" with respect to such Equifax Group Lease, and each such Subtenant does hereby sublease from its respective Sublandlord, for the term and on the terms and conditions hereinafter provided in Exhibit "D" attached hereto and

incorporated herein, such portion of the premises under such

Equifax Group Lease as is more particularly described in said Exhibit "C", and

herein referred to, as the "Sublet Premises". For purposes of Exhibit "D" and

this Agreement, and with respect to each such Equifax Group Lease, the term "Total Premises" means the entire "premises" under such Equifax Group Lease. In the event any such subletting requires the consent of the "landlord" or "lessor" under the applicable Equifax Group Lease, then the respective Sublandlord and Subtenant hereby agree to reasonably cooperate with each other in obtaining any such required consent. All statements, covenants, agreements, representations and warranties, if any, made herein (including, without limitation, in Exhibit

"D" hereto) by each Sublandlord shall be deemed to be made by such party only

with respect to itself or to the Equifax Group Lease or Sublet Premises of such party, as the case may be, and shall not be deemed to be made by such party with respect to any other Sublandlord or to any other Equifax Group Lease or Sublet Premises, and each Sublandlord shall have liability and responsibility under this Agreement with respect to such subletting only in respect of the Sublet Premises of that Sublandlord and shall have no liability or responsibility in respect of any other Sublet Premises.

(c) Assignment. Certegy Payment Services Group Companies currently

occupy all or a portion of the premises under each of the Equifax Group Leases identified as "to be assigned" on Exhibit "C" hereto. Each of the Equifax Group

Companies that is the "tenant" or "lessee" under each such Equifax Group Lease (each as an "Assignor" hereunder) does hereby assign all of its right, title, and interest in and to its respective Equifax Group Lease to the Certegy Payment Services Company identified on said Exhibit "C" as the "Assignee" with respect

to such Equifax Group Lease, and each such Assignee does hereby assume all of the obligations, duties, responsibilities, and liabilities of the "tenant" or "lessee" under such Equifax Group Lease with respect to periods of time from and after the date of this Agreement, pursuant to and in accordance with the terms and conditions hereinafter provided in Exhibit "E" attached hereto and

incorporated herein. In the event any such assignment requires the consent of the "landlord" or "lessor" under the applicable Equifax Group Lease, then the respective Assignor and Assignee hereby agree to reasonably cooperate with each other in obtaining any such required consent. All statements, covenants, agreements, representations and warranties, if any, made herein (including, without limitation, in Exhibit "E" hereto) by each Assignor shall be deemed to

be made by such party only with respect to itself or to the Equifax Group Lease of such party, as the case may be, and shall not be deemed to be made by such party with respect to any other Assignor or to any other Equifax Group Lease, and each Assignor shall have liability and responsibility under this Agreement with respect to such assignment only in respect of the Equifax Group Lease of that Assignor and shall have no liability or responsibility in respect of any other Equifax Group Lease.

- 3. Distribution Agreement.
- (a) Compliance. Certegy Payment Services shall comply with, abide

by, and perform all of the terms, covenants, conditions, agreements, requirements, restrictions and provisions of Section 2.08 of the Distribution Agreement (as hereinafter defined) as to the Certegy Payment Services Group Leases and the Equifax Guarantees and with respect to each Sublet Premises and the subleasing and assignments set forth in this Agreement. The term "Distribution Agreement'

means that certain Distribution Agreement dated as of the date of this Agreement executed by and between Equifax and Certegy in connection with the Spin Transaction.

- (b) Dispute Resolution. Any disputes arising under this Agreement, $% \left(\frac{1}{2}\right) =\frac{1}{2}\left(\frac{1}{2}\right) +\frac{1}{2}\left(\frac{1}{2}\right) +\frac{1}{2}\left$
- and any liability of either party with respect to the attorneys' fees or costs incurred by the other party with respect to such dispute, shall be resolved in accordance with Section 15.10 of the Distribution Agreement in the same manner and with the same effect as if said Section were set forth in full and at length herein and as if Certegy Payment Services and each Sublandlord and Assignee hereunder were the parties thereto other than Equifax, and as if Equifax and each Sublandlord and Assignor hereunder were the parties thereto other than Certegy, and said Section is hereby incorporated herein.
 - 4. Further Assurances. From time to time after the date hereof, each

of Equifax and Certegy Payment Services, and any Sublandlord, Subtenant, Assignor, and Assignee, shall, upon written request, do all such additional and further acts, and shall execute and deliver all such additional and further assignments, subleases, and other instruments and documents, as any other party hereto may reasonably require to effectuate the terms and conditions of this Agreement.

- 5. General Provisions.
- (a) Notices. Whenever any notice, demand or request is required or

permitted to be given by one party hereto to the other party under this Agreement, such notice, demand or request shall be in writing and shall be delivered by hand, be sent by registered or certified mail, postage prepaid, return receipt requested, or be sent by nationally recognized commercial courier for next business day delivery, to the addresses set forth below such party's respective execution hereof, or to such other addresses as are specified by written notice given in accordance herewith. All notices, demands or requests delivered by hand shall be deemed given upon the date so delivered; those given by mailing as hereinabove provided shall be deemed given on the date of deposit in the United States Mail; and those given by commercial courier as hereinabove provided shall be deemed given on the date of deposit with the commercial courier. Nonetheless, the time period, if any, in which a response to any notice, demand or request must be given shall commence to run from the date of receipt of the notice, demand or request by the addressee thereof. Any notice, demand or request not received because of changed address of which no notice was given as hereinabove provided or because of refusal to accept delivery shall be deemed received by the party to whom addressed on the date of hand delivery, on the first business day after deposit with commercial courier, or on the third business day following deposit in the United States Mail, as the case may be.

- (b) Headings. The use of headings, captions and numbers in this

 Agreement is solely for the convenience of identifying and indexing the various provisions in this Agreement and shall in no event be considered otherwise in
- provisions in this Agreement and shall in no event be considered otherwise in construing or interpreting any provision in this Agreement.

(c) Exhibits. Each and every exhibit referred to or otherwise

mentioned in this Agreement is attached to this Agreement and is and shall be construed to be made a part of

this Agreement by such reference or other mention at each point at which such reference or other mention occurs, in the same manner and with the same effect as if each exhibit were set forth in full and at length every time it is referred to or otherwise mentioned.

(d) Defined Terms. Capitalized terms used in this Agreement shall

have the meanings ascribed to them at the point where first defined, irrespective of where their use occurs, with the same effect as if the definitions of such terms were set forth in full and at length every time such terms are used.

- (e) Pronouns. Wherever appropriate in this Agreement, personal pronouns shall be deemed to include the other genders and the singular to include the plural.
- (f) Severability. If any term, covenant, condition or provision of this Agreement, or the application thereof to any person or circumstance, shall ever be held to be invalid or unenforceable, then in each such event the remainder of this Agreement or the application of such term, covenant, condition or provision to any other person or any other circumstance (other than those as to which it shall be invalid or unenforceable) shall not be thereby affected, and each term, covenant, condition and provision hereof shall remain valid and enforceable to the fullest extent permitted by law.
- (g) Non-Waiver. Failure by any party to complain of any action, non-action or breach of any other party shall not constitute a waiver of any aggrieved party's rights hereunder. Waiver by any party of any right arising from any breach of any other party shall not constitute a waiver of any other right arising from a subsequent breach of the same obligation or for any other default, past, present or future.
- (h) Rights Cumulative. All rights, remedies, powers and privileges conferred under this Agreement on the parties shall be cumulative of and in addition to, but not restrictive of or in lieu of, those conferred by law or at equity.
 - (i) Time of Essence. Time is of the essence of this Agreement.
- (j) Applicable Law. This Agreement shall be governed by, construed under and interpreted and enforced in accordance with the laws of the State of Georgia.
- (k) Entire Agreement; Modification. This Agreement supersedes all prior discussions and agreements between Equifax and Certegy Payment Services, and any Sublandlord. Subtenant, Assignor, or Assignee, with respect to the Certegy Payment Services Group Leases, the Equifax Group Leases, the Equifax Guarantees, the subletting and assigning hereunder and other matters expressly set forth herein, and this Agreement contains the sole and entire understanding between Equifax and Certegy Payment Services, and the Sublandlords, Subtenants, Assignors, and Assignees, with respect thereto. This Agreement shall not be modified or amended except by an instrument in writing executed by or on behalf of Equifax and Certegy Payment Services; provided, however, that if any such

modification or amendment affects any subletting described in Section 2(b), above, or any assignment described in Section 2(c), then the Sublandlord and

Subtenant or the Assignor and Assignee, as the case may be, shall also be a party to such modification or amendment instrument.

- (1) Counterparts. This Agreement may be executed in several counterparts, each of which shall be deemed an original, and all of such counterparts together shall constitute one and the same instrument.
- (m) Authority. Each party hereto warrants and represents that such party has full and complete authority to enter into this Agreement and each person executing this Agreement on behalf of a party warrants and represents that he has been fully authorized to execute this Agreement on behalf of such party and that such party is bound by the signature of such representative.
- (n) No Construction Against Preparer. No provision of this Agreement shall be construed against or interpreted to the disadvantage of any party by any court or other governmental or judicial authority by reason of such party's having or being deemed to have prepared or imposed such provision.
- (o) Successors and Assigns. This Agreement shall bind and benefit the successors and assigns of the parties with the same effect as if mentioned in each instance where a party hereto is named or referred to.

IN WITNESS WHEREOF, Equifax and Certegy Payment Services have caused their duly authorized representatives to execute, seal and deliver this Agreement Regarding Leases, all as of the day and year first written above.

	EQUIFAX:
	Equifax Inc., a Georgia corporation
Date executed:	By:
	Name:
	Title:
	[CORPORATE SEAL]
	Initial Address for Notices:
	Equifax Inc.

Attn: Phillip J. Mazzilli, CFO with a copy to:

1550 Peachtree Street Atlanta, Georgia 30309

Equifax Inc. 1550 Peachtree Street Atlanta, Georgia 30309 Attn: Kent E. Mast, General Counsel

7

	Equifax Payment Services, Inc., a Delaware corporation
Date executed:	By:
	Name:
	Title:
	[CORPORATE SEAL]
	Initial Address for Notices:
	Equifax Payment Services, Inc. c/o Certegy Inc.
	Attn: Michael T. Volkommer, CFO
	with a copy to:
	Equifax Payment Services, Inc. c/o Certegy Inc.
	Attn: Bruce S. Richards, General Counsel

PAYMENT SERVICES:

$\label{thm:constraint} \mbox{ Agreement and Acknowledgment of Sublandlords and Subtenants:} \\$

Each of the following parties hereby agree to the terms and provisions of the foregoing Agreement Regarding Leases, and is hereby made a party thereto, solely with respect to the subleasing of the Sublet Premises with respect to which such party is Sublandlord or Subtenant pursuant to Section 2(b) of the foregoing Agreement Regarding Leases.

SUBLANDLORDS:	SUBTENANTS:
Equifax Information Services LLC (f/k/a Equifax Credit Information Services, Inc.)	
ву:	ву:
Name:	Name:
Title:	Title:
[CORPORATE SEAL]	[CORPORATE SEAL]
Equifax Knowledge Engineering, Inc. (f/k/a Market Knowledge, Inc.)	
ву:	ву:
Name:	Name:
Title:	Title:
[CORPORATE SEAL]	[CORPORATE SEAL]
Initial Address for Notices for each Sublandlord:	Initial Address for Notices for each Subtenant:
[Name of Sublandlord] c/o Equifax Inc. 1550 Peachtree Street Atlanta, Georgia 30309 Attn: Phillip J. Mazzilli, CFO with a copy to:	[Name of Subtenant] c/o Certegy Inc. Attn: Michael T. Volkommer, CFO with a copy to:
[Name of Sublandlord] c/o Equifax Inc. 1550 Peachtree Street Atlanta, Georgia 30309 Attn: Kent E. Mast, General Counsel	[Name of Subtenant] c/o Certegy Inc
	General Counsel

$\label{thm:constraints} \mbox{ Agreement and Acknowledgment of Assignors and Assignees:} \\$

Each of the following parties hereby agree to the terms and provisions of the foregoing Agreement Regarding Leases, and is hereby made a party thereto, solely with respect to the assignment of the Equifax Group Lease with respect to which such party is Assignor or Assignee pursuant to Section 2(c) of the foregoing Agreement Regarding Leases.

ASSIGNORS:	ASSIGNEES:	
Equifax Inc.		
By:	ву:	
Name:	Name:	
Title: [CORPORATE SEAL]	Title: [CORPORATE SEAL]	
Equifax Information Services LLC (f/k/a Equifax Credit Information Services, Inc.)		
By: Name: Title:	By: Name: Title:	
[CORPORATE SEAL]	[CORPORATE SEAL]	
Initial Address for Notices for each	Initial Address for Notices for each	
Assignor:	Assignee:	
[Name of Assignor] c/o Equifax Inc. 1550 Peachtree Street Atlanta, Georgia 30309 Attn: Phillip J. Mazzilli, CFO with a copy to:	[Name of Assignee] c/o Certegy Inc. Attn: Michael T. Volkommer, CFO with a copy to:	
[Name of Assignor] c/o Equifax Inc. 1550 Peachtree Street Atlanta, Georgia 30309 Attn: Kent E. Mast, General Counsel	[Name of Assignee] c/o Certegy Inc	

General Counsel

11

SUBSIDIARIES

After the distribution, Certegy Inc. expects to have the following subsidiaries, each of which will be wholly owned by the Registrant, except as noted below:

Name of Subsidiary	State or Country of Incorporation
AGES Participacoes S.A.	Brazil
Aircrown Ltd.	England
Card Brazil Holdings, Inc.	Georgia
Card Brazil Holdings Ltda.	Brazil
Card Brazil LLC	Georgia
Central Credit Services Ltd.	Scotland
Equifax Asia Pacific Holdings, Inc.	Georgia
Equifax Australia Plc	England
Equifax Card Services, Inc.	Florida
Equifax Card Solutions Australia Pty Ltd.	Australia
Equifax Card Solutions Ltd.	England
Equifax Card Solutions S.A.	France
Equifax (Cayman Islands) Ltd.	Cayman Islands
Equifax Check Services, Inc.	Delaware
Equifax E-Banking Solutions, Inc.	Georgia
Equifax Ltd.	New Zealand
Equifax Payment Recovery Services, Inc.	Georgia
Equifax Payment Services, Inc.	Delaware
Equifax Pty Ltd.	Australia
Equifax SNC	France
Financial Insurance Marketing Group, Inc.	District of Columbia
First Bankcard Systems, Inc.	Georgia
Payment Brasil Holdings Ltda.	Brazil
Payment Chile, S.A.	Chile
Payment Europe LLC	Georgia
Payment South America Holdings, Inc.	Georgia

Payment South America LLC Georgia Payment U.K. Ltd. England Procard S.A. Chile Retail Credit Management Ltd. England Telecredit Canada Inc. Canada Transax France Plc England Transax Ireland Ltd. Ireland Transax Plc England Unnisa - Solucoes em Meios de Pagamento Ltda Brazil VIV Plc England

Equifax Card Services, Inc. will own a 51% interest in Circle of Value, Ltd. Payment Brazil Holdings Ltda. will own a 51% interest in Partech Ltda. (Brazil).

INFORMATION STATEMENT

[Equifax logo]

EQUIFAX INC. 1550 Peachtree Street, N.W. Atlanta, Georgia 30309

June 12, 2001

Dear Fellow Shareholder:

I am pleased to report that the previously announced spin-off of our payment services businesses is expected to take place on July 7, 2001. Certegy Inc., a recently formed Georgia corporation that will own all of our payment services businesses, will commence operation on that day as an independent public company. Certegy's shares of common stock will be listed on the New York Stock Exchange under the symbol "CEY."

Holders of record of Equifax common stock as of 5:00 p.m., Atlanta, Georgia, time on June 27, 2001, which will be the record date, will receive one share of Certegy common stock for every two shares of Equifax common stock held. No action is required on your part to receive your Certegy shares. You will not be required either to pay anything for the new shares or to surrender any shares of Equifax common stock.

No fractional shares of Certegy common stock will be issued. If you otherwise would be entitled to a fractional share, you will receive a check for the cash value of that share, which may be taxable to you. The distribution will otherwise be tax-free to Equifax and to you to the extent you receive Certegy common stock. In due course, we will provide you with information to enable you to compute your tax bases in both Equifax and Certegy common stock.

The enclosed information statement explains the distribution of shares of Certegy common stock in detail and contains important information about Certegy, including financial statements. We urge you to read it carefully.

Very truly yours,

/s/ Thomas F. Chapman
----Thomas F. Chapman
Chairman and Chief Executive Officer

CERTEGY INC.

P.O. Box 349

Alpharetta, Georgia 30009

(678) 867-8000

June 12, 2001

Dear Equifax Shareholder:

We are very pleased that you will soon be a shareholder of Certegy Inc.

We provide credit and debit card processing and check risk management services to financial institutions and merchants throughout the world. We believe that the separation of our businesses from the businesses of our corporate parent, Equifax Inc., will enhance our ability to increase penetration of the markets we currently serve, attract new customers, expand our global operations and pursue new business opportunities. As an independent company, we can more effectively focus on our objectives and support the capital needs of our company, bringing value to you as a shareholder.

The Certegy shares you are receiving have been approved for listing on the New York Stock Exchange, subject to issuance, under the symbol "CEY."

This is a very exciting time and we are enthusiastic about our future as a new, independent public company. We look forward to your support and participation in our success.

Congratulations on becoming one of our "founding" shareholders!

Very truly yours,

/s/ Lee A. Kennedy
Lee A. Kennedy
President and Chief Executive
Officer

INFORMATION STATEMENT

CERTEGY INC.

Distribution of Approximately 68,121,712 Shares of Common Stock

This information statement is being furnished in connection with the distribution by Equifax Inc. to holders of its common stock of all the outstanding shares of common stock of Certegy Inc. Prior to the distribution, Equifax will transfer to us its payment services businesses described in this information statement.

Shares of our common stock will be distributed to holders of Equifax common stock of record as of 5 p.m., Atlanta, Georgia, time on June 27, 2001, which will be the record date. Equifax shareholders will receive one share of our common stock for every two shares of Equifax held on the record date. Certificates representing shares of Certegy common stock will be mailed on July 7, 2001 or as soon as possible thereafter. Shareholders will receive cash in lieu of fractional shares, which may be taxable.

No shareholder approval of the distribution is required or sought. Please do not send us a proxy. You will not be required to pay for the shares of our common stock that you receive in the distribution, or surrender or exchange shares of Equifax common stock in order to receive our common stock, or take any other action in connection with the distribution. Each share of our common stock distributed will be accompanied by one common stock purchase right. There is no current trading market for our common stock. We have been approved, subject to issuance, to list our common stock on the New York Stock Exchange under the symbol "CEY."

As you review this information statement, you should carefully consider the matters described under the caption "Risk Factors" beginning on page 13.

Neither the Securities and Exchange Commission nor any state securities commission has approved or disapproved these securities or determined if this information statement is truthful or complete. Any representation to the contrary is a criminal offense.

This information statement does not constitute an offer to sell or the solicitation of an offer to buy any securities.

Shareholders of Equifax with inquires related to the distribution should contact Equifax's transfer agent, SunTrust Bank, Stock Transfer Department, P.O. Box 4625, Atlanta, Georgia 30302, or by telephone at (800) 568-3476, or Equifax's Investor Relations Department at Equifax, P.O. Box 4081, Atlanta, Georgia 30302 or by telephone at (404) 885-8304.

The date of this information statement is June 12, 2001.

INFORMATION STATEMENT

TABLE OF CONTENTS

	Page
Summary	1
The Distribution	8
Risk Factors	
Forward-Looking Statements	18
Dividend Policy	
Capitalization	
Selected Financial Data	21
Management's Discussion and Analysis of Financial Condition and Results	
of Operations	
Business	
Relationship Between Equifax and Our Company After the Distribution	
Management	
Ownership of Our Common Stock	
Description of Capital Stock	
Indemnification of Directors and Officers	75
Independent Public Accountants	
Where You Can Obtain Additional Information	
Index to Financial Statements	F-1

SUMMARY

The following summary highlights selected information contained in this information statement relating to the distribution and the businesses to be contributed to us by Equifax on or prior to the distribution. Because this is a summary, it does not contain all the details concerning the distribution and our business, including information that may be important to you. We urge you to read the entire information statement carefully, especially the risks relating to the distribution, and our business and industry discussed under "Risk Factors" and our financial statements.

We describe in this information statement the businesses to be contributed to us by Equifax Inc. prior to the distribution, described under "--The Distribution," as if they were our businesses for all historical periods described. After the distribution, we will be an independent public company, and Equifax will have no continuing stock ownership in us. Accordingly, our historical financial results as part of Equifax contained in this information statement may not reflect our financial results in the future as an independent company or what our financial results would have been had we been a stand-alone company during the periods presented.

OUR BUSINESS

Overview

We provide credit and debit card processing and check risk management services to financial institutions and merchants throughout the world. Last year, we processed over 1.7 billion payment transactions, serviced over 27 million card accounts, and authorized over \$28 billion of check transactions worldwide. Our business is comprised of two segments, Card Services and Check Services.

Card Services provides credit and debit card processing services, and e-banking services. Our card issuer services enable banks, credit unions, retailers and others to issue VISA and MasterCard credit and debit cards, private label cards, and other electronic payment cards. The majority of our card issuer customers subscribe to our full service programs, where we provide essentially all of the operations and support necessary to support a card issuer's credit and debit card issuing program. Our merchant processing services enable retailers and other businesses to accept credit, debit, and other electronic payment cards from purchasers of their goods and services. We provide these services both directly to merchants that accept cards, and through contracts with financial institutions and others where our products and services enable them to service the card processing needs of their merchant customers. Our e-banking business enables banks to provide electronic banking services to their business and individual banking customers.

Card Services' target market in the U.S. consists of banks, credit unions, other card issuers, and retailers. Within these markets, we believe we have established ourselves as the leader, based on market share, in providing comprehensive card processing services to credit unions and to independent community bank card issuers. We process for approximately 4,000 U.S. credit unions and approximately 2,200 independent community banks, principally through our strategic alliances with the two premier associations representing independent financial institutions. These associations are the Independent Community Bankers Association, or ICBA, and Card Services for Credit Unions, or CSCU. Our revenue from international card processing operations has grown rapidly since we entered the international market in 1998, and now accounts for 23% of Card Services' revenue. The international card processing market has grown rapidly over the last several years. We believe that strong growth will continue well into the future, fueled by the rapid development of credit economies in Asian-Pacific and South American countries, international expansion of U.S. card issuers, and outsourcing trends among international financial institutions.

Check Services provides check risk management products and services which enable retailers, hotels, automotive dealers, telecommunications companies, supermarkets, casinos, mail order houses and other businesses to minimize losses from dishonored checks, maximize check acceptance, and improve customer service. Our diverse and flexible portfolio of check risk management products and services, including check guarantee, check verification, bad check collections, and risk management consulting, allows us to tailor solutions to meet the specific needs of the customer.

We believe that strong demand for cost-effective, accurate, and innovative check risk management creates significant growth opportunities for our check risk management business in the U.S., which has benefited from our development and launch of a series of enhanced risk modeling capabilities and check risk management products, and our increased focus on sales and marketing to regional and local retailers to supplement our traditional leadership in the national retailer market. We provide our check risk management products and services internationally in Canada, the U.K., Ireland, France, Australia, and New Zealand. In terms of aggregate market share, we believe we are the leading provider of check risk management products and services in those combined geographies. The low penetration of check risk management products and services in international markets creates a substantial growth opportunity for our Check Services business.

Since 1990, our businesses have been operated as part of Equifax. As a whole, Equifax enables and secures global commerce through its information management, transaction processing, direct marketing, and customer relationship management businesses. Equifax serves the financial services, retail, credit card, transportation, telecommunications/utilities, information technology and healthcare industries and government. Through this distribution, Equifax is spinning off its transaction processing businesses, which are the businesses described in this information statement as our businesses. In fiscal year 2000, our businesses accounted for approximately 24% of Equifax's assets, 40% of Equifax's net income, and 40% of Equifax's total revenue. Equifax believes that separating its two major lines of businesses in the distribution will provide Equifax and our company with greater managerial and operational flexibility to respond to changing market conditions in our differing business environments and to achieve our respective strategic and financial objectives. Equifax also believes that the separation will provide Equifax and Certegy with additional financial flexibility to pursue growth opportunities and will lead to better investor understanding of the different businesses.

Our Strategy

Our strategic objective is to strengthen our position as a leading global provider of payment processing and check risk management services. We intend to concentrate on the following strategies to accomplish our objective:

- . Leverage the competitive strengths of our Card Services business in the U.S., including our CSCU and ICBA alliances, to further increase our share of revenue in the U.S. card and merchant processing markets.
- . Grow the international customer base and processing volumes of our Card Services business.
- . Increase Card Services' revenues from new and existing products and services.
- . Leverage the competitive strengths of our Check Services business, which include what we believe are the industry's most advanced check risk management algorithms and systems, to increase its share of the markets it traditionally serves.
- . Continue our development and utilization of increasingly sophisticated risk modeling tools in our Check Services segment to differentiate our capabilities from the competition.
- . Enter new markets such as check cashing, gaming, grocery, government and Internet commerce by combining our current risk management and identity authentication products and services.

THE DISTRIBUTION

Please see "The Distribution" for a more detailed description of the matters described below.

Distributing Company	Equifax Inc., a worldwide leader in enabling and securing global commerce through its information management, transaction processing, direct marketing, and customer relationship management businesses. Equifax currently operates two major lines of business commonly referred to as information services and payment services.
Distributed Company	Certegy Inc., which will own the payment services businesses conducted by Equifax and described in this information statement. Please see "Management's Discussion and Analysis of Financial Condition and Results of Operations" and "Business" for a description of these businesses.
Distribution Ratio	Each holder of Equifax common stock other than three grantor trusts created by Equifax for purposes of holding shares to fund obligations arising under certain employee benefits plans or programs, will receive a dividend of one share of our common stock, and the associated common stock purchase right, for every two shares of Equifax common stock held on the record date.
Securities to be Distributed	Based on Equifax having 136,243,425 shares of common stock entitled to receive Certegy shares in the distribution outstanding on March 31, 2001, and assuming no exercise of outstanding options, approximately 68,121,712 shares of our common stock, together with the associated common stock purchase rights, will be distributed. The shares of our common stock to be distributed will constitute all of the outstanding shares of our common stock immediately after the distribution. Equifax shareholders will not be required to pay for the shares of our common stock to be received by them in the distribution, or to surrender or exchange shares of Equifax common stock in order to receive our common stock, or to take any other action in connection with the distribution.
Fractional Shares	Fractional shares of our common stock will not be distributed. Fractional shares held by owners of record will be aggregated and sold in the public market by the distribution agent. The aggregate net cash proceeds of these sales will be distributed ratably to those shareholders who would otherwise have received fractional interests. These proceeds may be taxable to those shareholders.
Distribution Agent, Transfer Agent, and Registrar for Shares	SunTrust Bank will be the distribution agent, transfer agent and registrar for the shares of our common stock.
Record Date	The record date is as of 5 p.m. Atlanta, Georgia, time on June 27, 2001.
Effective Time	Our separation from Equifax will be effective as of 11:59 p.m., Atlanta, Georgia, time on June 30, 2001. At the effective time, we

will become independent for accounting purposes. The distribution will actually be complete, however, on the distribution date, when Equifax distributes all of the shares of Certegy common stock to its shareholders of record at the close of business on the record date. Regardless of when shares of Certegy common stock are distributed, Certegy and its new public shareholders will be the sole beneficiaries of the operation of the payment services businesses after the effective time.

Distribution Date..... July 7, 2001.

Federal Income Tax Consequences of the Distribution.....

Equifax has received a ruling from the Internal Revenue Service to the effect that the distribution will qualify as a tax-free transaction under Sections 355 and 368(a)(1)(D) of the Internal Revenue Code of 1986, as amended.

Stock Exchange Listing.....

There is not currently a public market for our common stock. We have been approved, subject to issuance, to list our common stock on the New York Stock Exchange under the symbol "CEY." It is anticipated that trading will commence on a when issued basis prior to the distribution. When issued trading refers to a transaction made conditionally because the security has been authorized but not yet issued. On the first trading day following the distribution date, when issued trading in respect of our common stock will end and regular way trading will begin. Regular way trading refers to trading after a security has been issued and typically involves a transaction that settles on the third full business day following the date of the transaction. We cannot predict the trading prices for our common stock before or after the distribution date, however, we believe the presence of a when issued trading market prior to the distribution may have a stabilizing effect on the price of our common stock following the distribution.

Credit Facilities.....

We have obtained commitments for a \$300 million threeyear unsecured revolving credit facility and a \$100 million 364 day unsecured revolving credit facility. Proceeds from these facilities will fund a cash payment to Equifax of \$275 million to reflect our share of Equifax's pre-distribution debt. These credit facilities will also be used to fund our working capital and acquisition needs after the distribution. These credit facilities will have a variable interest rate based on market rates. The credit agreements will contain financial and non-financial covenants customary for financings of this nature.

Relationship between Equifax and Us After the Distribution......

Following the distribution, we will be an independent public company, and Equifax will have no continuing stock ownership interest in us. Prior to the distribution, we will enter into a distribution agreement and several ancillary agreements with Equifax for the purpose of accomplishing the contribution of Equifax's payment services businesses to us and the distribution of

our common stock to Equifax's shareholders. These agreements also will govern our relationship with Equifax subsequent to the distribution and provide for the allocation of employee benefits, tax and some other liabilities and obligations attributable to periods prior to the distribution. These agreements also include arrangements with respect to intellectual property and various interim services. The distribution agreement will provide that we generally will indemnify Equifax against liabilities arising out of the payment services businesses being transferred to us and that Equifax generally will indemnify us against liabilities arising out of the businesses Equifax is retaining. Please see "Relationship Between Equifax and Our Company After the Distribution" for a more detailed description of these agreements.

Post-Distribution Dividend Policy.. We do not anticipate paying any dividends on our common stock in the foreseeable future. The declaration and payment of dividends after the distribution, however, will be at the discretion of our board of directors.

Certain Anti-Takeover Effects..... Certain provisions of our articles of incorporation and bylaws may have the effect of making more difficult the acquisition of control of Certegy by a third-party in a transaction not approved by our board of directors. Our rights agreement also would make such a transaction more difficult. Moreover, certain provisions of the tax sharing and indemnification agreement we will enter into with Equifax could discourage or delay potential acquisition proposals.

Risk Factors...... Shareholders should carefully consider the matters

discussed under "Risk Factors."

Our Principal Executive Offices.... 1550 Peachtree Street, N.W. Atlanta, Georgia 30309

Our Mailing Address..... P.O. Box 349

Alpharetta, Georgia 30009

(678) 867-8000

SUMMARY HISTORICAL AND PRO FORMA COMBINED FINANCIAL DATA

Our summary historical and pro forma combined financial data set forth below should be read in conjunction with the "Combined Financial Statements of the Equifax Payment Services division," including the notes to those statements, "Management's Discussion and Analysis of Financial Condition and Results of Operations," and the "Pro Forma Combined Financial Statements of the Equifax Payment Services division," including the notes to those statements, included elsewhere in this information statement.

The historical income statement data for the years ended December 31, 2000, 1999, and 1998 and the historical balance sheet data as of December 31, 2000 and 1999 are derived from the combined financial statements included elsewhere in this information statement that have been audited by Arthur Andersen LLP, Equifax's independent public accountants. The historical income statement data for the three months ended March 31, 2001 and 2000 and for the years ended December 31, 1997 and 1996, and the historical balance sheet data as of March 31, 2001 and 2000 and December 31, 1998, 1997, and 1996 are derived from unaudited combined financial statements that have been prepared by management and are not included in this information statement. Operating expenses in the historical income statements reflect direct charges of the payment services business together with allocations of certain Equifax corporate expenses that have been charged to us based on usage or other methodologies appropriate for such expenses. In the opinion of management, these allocations have been made on a reasonable basis.

The summary pro forma combined financial data reflects adjustments to our historical combined balance sheet as of March 31, 2001 as if the distribution to shareholders had occurred on March 31, 2001 and to our historical combined income statements for the three months ended March 31, 2001 and for the year ended December 31, 2000 as if the distribution had occurred on January 1, 2000. The summary pro forma combined financial data we have included in this information statement reflects our historical financial position and results of operations, with adjustments made for the new credit facilities that will be used to fund the \$275 million payment to Equifax in conjunction with the distribution and the related interest expense; additional operating expenses related to incremental pension expense, insurance costs, corporate headquarters rent, and stand-alone public company costs for audit, director, and stock exchange fees; and the related income tax benefit resulting from the pro forma adjustments.

Neither the historical financial information nor the pro forma data presented below is necessarily indicative of what our results of operations or financial position would have been had we operated as an independent company during the periods presented, nor is it necessarily indicative of our future performance as an independent company.

Summary Historical Combined Financial Data

Three Months Ended

	March	31,		Year Ende			
	2001	2000	2000	1999	1998	1997	1996
	(1	Dollars in	thousands,	except pe	r share da	ta)	
Revenues Operating expenses	•	•	631,403	553,687	462,829	407,462	320,286
Operating income	26,490						
Other income (expense), net	(495)	(317)	8	1,410	(916)	(4,330)	(631)
Income before income taxes and minority interests	25, 995	23, 593	147,167	128,895	102,375	74,811	63,594
Provision for income taxes	(10,138)	(9,235)	(57,609)	(54, 272)	(40,505)	(29,240)	(24,476)
earnings, net of tax	(618)	514	(1,096)	6	(780)		(240)
Net income	\$ 15,239 =======	\$ 14,872 =======	\$ 88,462 ======	\$ 74,629 ======	\$ 61,090 ======	\$ 45,571 ======	•
Basic earnings per share(1)	\$ 496,497 \$	\$ 504,800 \$	\$502,445 \$	\$495,255 \$	\$492,704 \$	\$273,966 \$	\$229,655 \$
-cqu⊥ιy	φ 3∠1,∠98	φ ∠ 19,312	Φ3 ∠3,018	φ ∠/ 1,490	φ340,193	φ±5∠,∠∠3	φ±31,990

	Three Months Ended March 31, 2001	Ended				
	(Dollars in thousands, except per share data)					
Revenues	\$194,976 170,111 (2)	\$778,562 637,903 (2)				
Operating income	24,865					
Income before income taxes and minority interests	(7,707)(4)	118,932 (46,569)(4) (1,096)				
Net income	\$ 11,436 ======	•				
Basic earnings per share(1)	\$ 0.17 \$496,497 \$275,000 (5) \$ 52,298 (5)	\$502,445 \$275,000 (5)				

- ------

- (1) Using a distribution ratio of one share of Certegy common stock for every two shares of Equifax common stock held. Weighted average shares outstanding is computed by applying the distribution ratio to the historical Equifax weighted average shares outstanding for all periods presented.
- (2) Includes adjustment to reflect additional annual expenses of approximately \$6.5 million, which specifically relate to incremental pension expense, insurance costs, corporate headquarters rent, and stand-alone public company costs for audit, director, and stock exchange fees. The increase in pension expense is a result of us no longer benefitting from the overfunded status of the consolidated Equifax pension plan.
- (3) Includes adjustment to reflect interest expense on the \$275 million of debt to be used to fund the cash payment to Equifax in conjunction with the distribution, at an annual rate of LIBOR plus 100 basis points (7.54% for the year ended December 31, 2000 and 6.34% for the three months ended March 31, 2001), and includes amortization of financing costs over the three-year term of the debt. No Equifax corporate debt or related interest expense is included in the historical financial statements, as these amounts have not historically been allocated to the operating divisions by Equifax.
- (4) Includes adjustment to reflect the income tax benefit resulting from the pro forma adjustments using our effective tax rate for the period.
- (5) Includes adjustment to reflect the capitalization of the Equifax equity investment and the \$275 million of debt to be used to fund the cash payment to Equifax in conjunction with the distribution.

General

On October 2, 2000, the board of directors of Equifax announced a preliminary decision to spin-off Equifax's payment services businesses in a tax-free distribution of shares to Equifax's shareholders. On June 11, 2001, the board of directors of Equifax formally approved the distribution of all of the outstanding shares of our common stock and declared a dividend payable to holders of record of Equifax's common stock at 5 p.m., Atlanta, Georgia, time on June 27, 2001, which will be the record date, of one share of our common stock and an associated common stock purchase right for every two shares of Equifax common stock held by such holder on the record date. Please see "Description of Capital Stock--Rights Agreement" for a description of the common stock purchase rights.

Manner of Effecting the Distribution

The general terms and conditions relating to the distribution are set forth in the distribution agreement between us and Equifax. The distribution agreement provides generally that the transfer by Equifax of the stock, assets and liabilities of its payment services businesses to Certegy, and our separation from Equifax for accounting purposes, will be effective as of 11:59 p.m., Atlanta, Georgia, time on June 30, 2001. Prior to the distribution date, Equifax will deliver all outstanding shares of our common stock to the distribution agent for distribution on July 7, 2001 or as promptly as practicable thereafter.

For most Equifax shareholders who own Equifax common stock in registered form on the record date, our distribution agent will automatically mail to those shareholders a Certegy common stock certificate. For shareholders who own Equifax common stock through a broker or other nominee, their receipt of Certegy common stock certificates will depend on the arrangements they have with the nominee holding their Equifax shares. As further discussed below, fractional shares will not be distributed. A delivery of a share of our common stock in connection with the distribution also will constitute the delivery of the common stock purchase right associated with such share. These rights are intended to have anti-takeover effects. The existence of the rights may deter a potential acquiror from making a takeover proposal or a tender offer. For a more detailed discussion of these rights, please see "Description of Capital Stock--Rights Agreement." Unless the context otherwise requires, reference in this information statement to our common stock will include the related common stock purchase rights.

Equifax shareholders will not be required to pay for shares of our common stock received in the distribution, or to surrender or exchange shares of Equifax common stock in order to receive our common stock, or to take any other action in connection with the distribution. No vote of Equifax shareholders is required or sought in connection with the distribution, and Equifax shareholders have no dissenters' rights in connection with the distribution.

Fractional shares of our common stock will not be issued to Equifax shareholders as part of the distribution. In lieu of receiving fractional shares, each holder of Equifax common stock who would otherwise be entitled to receive a fractional share of our common stock will receive cash for the fractional interest, which may be taxable to such holder. For an explanation of the tax consequences of the distribution, please see "The Distribution--Federal Income Tax Consequences of the Distribution." The distribution agent will, as soon as practicable after the distribution date, aggregate fractional shares into whole shares, sell them in the open market at the prevailing market prices and distribute the aggregate proceeds ratably to Equifax shareholders otherwise entitled to fractional interests. The amount of this payment will depend on the prices at which the aggregated fractional shares are sold by the distribution agent in the open market shortly after the distribution date. Equifax will be responsible for any payment of brokerage fees. The amount of these brokerage fees is not expected to be material to Equifax.

In order to be entitled to receive shares of our common stock in the distribution, Equifax shareholders must be shareholders as of 5 p.m., Atlanta, Georgia, time on the record date, June 27, 2001.

The chart set forth below illustrates the distribution ratio by showing the number of our shares and/or the amount of cash that an Equifax shareholder would receive pursuant to the distribution for varying amounts of Equifax shares held as of the record date for the distribution, assuming a distribution ratio of one share of our common stock for every two shares of Equifax common stock held.

common stock held 1 Share		2 Shares	200 Shares	201 Shares		
Shares of our common stock to be received upon distribution	0 shares of our common stock and cash for .5 fractional share	1 share of our common stock and no cash	100 shares of our common stock and no cash	100 shares of our common stock and cash for .5 fractional share		

Because the distribution will be effected in the manner described above, many of our shareholders may hold odd lots, or blocks of less than 100, of our shares. An investor selling an odd lot may be required to pay a higher commission rate than an investor selling round lots, or blocks of 100 shares.

Reasons for the Distribution

Sharos of Equifax

Equifax's board of directors believes that the distribution is in the best interests of Equifax and its shareholders and that the separation of our businesses from Equifax's other businesses will provide Equifax and our company with greater managerial and operational flexibility to respond to changing market conditions in our differing business environments, as well as provide both companies with additional financial flexibility to pursue growth opportunities. In deciding how to achieve Equifax's business purpose for separating our businesses from the other businesses conducted by Equifax, management determined that a spin-off of our business was the only transaction that would be nontaxable, practical and not unduly expensive. In determining that the separation is in the best interests of Equifax and its shareholders, Equifax's board of directors considered the potential negative effect of restrictions, due to tax considerations caused by the spin-off, on significant stock transactions for both Equifax and us following the distribution. These tax considerations are more fully discussed under "--Federal Income Tax Consequences of the Distribution."

The following discussion of the reasons for the distribution includes forward-looking statements that are based upon numerous assumptions with respect to the trading characteristics of our common stock, the ability of our management to succeed in taking advantage of growth opportunities and our ability to succeed as a stand-alone company. Many of these factors are discussed below under the captions "Risk Factors" and "Forward-Looking Statements."

Management Considerations. At present, our businesses and the other businesses conducted by Equifax are operated as separate divisions under a single corporate parent entity. The distribution is expected to benefit each of the two divisions, allowing the management of each line of business to design and implement corporate policies and strategies that are based primarily on its business characteristics and most appropriate to each of the respective businesses. Furthermore, management believes that after the separation, there are opportunities to improve our profitability and our business performance as a stand-alone company.

Enhanced Ability to Expand the Scope and Level of Services Provided to Customers and to Pursue New Business Opportunities. We also expect that the separation will improve each company's ability to expand the scope and level of services they provide to their respective customer bases, and to pursue new business opportunities. For instance, the largest customers of Equifax's information services group are large, national financial institutions, which compete with our Card Services segment's principal customer base in the U.S., namely the credit unions and independent community banks. Additionally, several significant customers of Equifax's information services business compete with our merchant transaction processing business. We believe that eliminating these existing inherent conflicts through the separation will enable us to offer more

aggressively a wider array of services to the credit unions and community banks to enable them to compete more effectively in the payment card market, as well as allow us to compete more aggressively for merchant processing business.

Enhanced Ability to Execute Growth Strategies. Each company will also be able to pursue growth strategies that fit the different prospects of each business. We expect that the ability of each company to pursue strategic alliances, acquisitions and other investment opportunities will be enhanced by the provision of differentiated access to the capital markets. Additionally, each line of business will be financially independent and able to concentrate its financial resources wholly on its own operations.

Improved Ability to Hire, Retain and Motivate Key Personnel. The distribution will also permit each company to design incentive compensation programs that relate more directly to its own business characteristics and performance and will provide each company with a "pure-play" publicly traded equity for use in its incentive compensation programs.

Investor Understanding; Public Relations. The distribution should facilitate a more focused evaluation of the unique investment opportunities of each company given the different market opportunities attributable to each business line. In addition, each company will be able to focus its public relations efforts on cultivating its own separate identity.

Interests of Persons in the Distribution

Two of our directors, Mr. Thomas F. Chapman and Mr. Lee A. Kennedy, are members of the Equifax board of directors and voted to approve the distribution. Mr. Chapman, the current Chairman of the board of directors and Chief Executive Officer of Equifax, will serve on our board of directors, and as Chairman of our board, for a transitional period of one year after the distribution. For his service as Chairman of our board of directors, Mr. Chapman will receive an annual retainer of \$250,000 payable in cash, shares of our common stock, or a combination of both. He will also receive \$1,000 for attendance at each meeting of the board or a board committee, and an option to purchase 2,000 shares of our common stock upon the effective date of our 2001 Non-Employee Director Stock Option Plan. Mr. Chapman will remain Chairman of the Equifax board of directors at the time of the distribution. Mr. Kennedy, our President and Chief Executive Officer, will resign from the Equifax board of directors at the time of the distribution. Please see "Management--Executive Compensation."

Results of the Distribution

After the distribution, we will be an independent public company owning and operating what has previously been Equifax's payment services businesses. Immediately after the distribution, we expect to have approximately 10,082 holders of record of shares and approximately 68,121,712 shares outstanding, based on the number of record shareholders and outstanding shares of Equifax common stock entitled to receive Certegy shares in the distribution on March 31, 2001 and assuming no exercise of outstanding options, and after giving effect to the delivery to shareholders of cash in lieu of fractional shares of our common stock. The actual total number of our shares of common stock to be distributed will depend on the number of shares of Equifax common stock outstanding on the record date.

For information regarding options to purchase our common stock that will be outstanding after the distribution, see "Relationship Between Equifax and Our Company After the Distribution--Employee Benefits Agreement" and "Management." Prior to the distribution, we will enter into several agreements with Equifax in connection with, among other things, intellectual property and various services. For a more detailed description of these agreements, please see "Relationship Between Equifax and Our Company After the Distribution."

The distribution will not affect the number of outstanding shares of Equifax common stock or any rights of Equifax shareholders in their Equifax shares.

Equifax has received a ruling from the Internal Revenue Service, or the IRS, to the effect that the contributions constitute a reorganization pursuant to Section 368(a)(1)(D) of the Internal Revenue Code, or the Code, and that the distribution will qualify as a tax-free distribution for federal income tax purposes under Section 355 of the Code. The ruling provides that for U.S. Federal income tax purposes:

- . Equifax will not recognize any gain or loss upon the distribution;
- . no gain or loss will be recognized by, or be includible in the income of, a shareholder of Equifax common stock solely as the result of the receipt of our common stock in the distribution, except, as described below, in connection with cash received in lieu of fractional shares of our common stock;
- . the basis of the Equifax common stock and our common stock in the hands of Equifax's shareholders immediately after the distribution will be the same as the basis of the Equifax common stock immediately before the distribution, allocated between the common stock of Equifax and our common stock in proportion to their relative fair market values on the date of the distribution;
- . the holding period of our common stock received by Equifax shareholders will include the holding period of their Equifax common stock, provided that such Equifax common stock is held as a capital asset on the date of the distribution; and
- . shareholders of Equifax who receive cash from the distribution agent in respect of fractional shares will recognize gain or loss on the sale of the fractional share interest equal to the difference between the cash received and the holder's basis in such fractional share interest. Such gain or loss will be capital gain or loss to such holder, provided the fractional share interest is a capital asset in the hands of such holder.

Although the ruling relating to the qualification of the distribution as a tax-free transaction is generally binding on the IRS, the continuing validity of the ruling is subject to factual representations and assumptions. Neither Equifax nor we are aware of any facts or circumstances that would cause such representations and assumptions to be untrue.

If the distribution were not to qualify as a tax-free transaction, Equifax would recognize taxable gain equal to the excess of the fair market value of our common stock distributed to Equifax shareholders over Equifax's tax basis in our common stock. In addition, each shareholder who receives our common stock in the distribution would generally be treated as receiving a taxable distribution in an amount equal to the fair market value of our common stock received.

Even if the distribution otherwise qualifies for tax-free treatment under Sections 355 and 368(a)(1)(D) of the Code, it may be disqualified as tax-free to Equifax under Section 355(e) of the Code if 50% or more of the stock of Equifax or us is acquired as part of a plan or series of related transactions that includes the distribution. For this purpose, any acquisitions of our stock or Equifax's stock within two years before or after the distribution are presumed to be part of such a plan, although Equifax or we may be able to rebut that presumption. If such an acquisition of our stock or Equifax's stock triggers the application of Section 355(e), Equifax would recognize taxable gain as described above, but the distribution would generally be tax-free to each Equifax shareholder. Under the tax sharing and indemnification agreement between Equifax and us, we would be required to indemnify Equifax against that taxable gain if it were triggered by an acquisition of our stock. Please see "Relationship Between Equifax and Our Company After the Distribution--Tax Sharing and Indemnification Agreement" for a more detailed discussion of the tax sharing and indemnification agreement between Equifax and us.

U.S. Treasury regulations require each Equifax shareholder that receives shares of our stock in the distribution to attach to the shareholder's U.S. Federal income tax return for the year in which such stock is received a detailed statement setting forth such data as may be appropriate to show the applicability of Section

355 of the Code to the distribution. Within a reasonable period of time after the distribution, Equifax will provide its shareholders who receive our common stock pursuant to the distribution with the information necessary to comply with such requirement.

Each Equifax shareholder should consult his or her tax adviser about the particular consequences of the distribution to him or her, including the application of state, local and foreign tax laws, and possible changes in tax law that may affect the tax consequences described above.

Listing and Trading of Our Common Stock

There is not currently a public market for our common stock. We have been approved, subject to issuance, to list our common stock on the New York Stock Exchange under the symbol "CEY." It is anticipated that trading will commence on a "when issued" basis prior to the distribution. A when issued listing may be identified by the "wi" letters next to the Certegy common stock listing on the New York Stock Exchange Composite Tape. If the conditions to the distribution are not satisfied and our shares of common stock are not distributed, all when issued trading will become null and void. If the distribution occurs as planned, on the first trading day following the distribution date, when issued trading in our common stock will end and regular way trading will begin. Regular way trading refers to trading after a security has been issued and typically involves a transaction that settles on the third full business day following the date of a transaction.

We cannot assure you as to the price at which our common stock will trade before, on, or after the distribution date. Although the price at which our stock trades may fluctuate significantly until our common stock is fully distributed and an orderly market develops, we believe the presence of a when issued trading market may have a stabilizing effect on the price of our common stock following the distribution. In addition, the combined trading prices of our common stock and Equifax common stock held by shareholders after the distribution may be less than, equal to, or greater than the trading price of Equifax common stock prior to the distribution.

The shares distributed to Equifax shareholders will be freely transferable, except for shares received by people who may have a special relationship or affiliation with us. People who may be considered our affiliates after the distribution generally include individuals or entities that control, are controlled by, or are under common control with us. This will include our directors and certain of our officers. Persons who are our affiliates will be permitted to sell their shares only pursuant to an effective registration statement under the Securities Act of 1933, as amended, or an exemption from the registration requirements of the Securities Act, such as exemptions afforded by Section 4(2) of the Securities Act or Rule 144 thereunder.

Reasons for Furnishing This Information Statement

This information statement is being furnished by Equifax solely to provide information to shareholders of Equifax who will receive shares of our common stock in the distribution. It is not, and is not to be construed as, an inducement or encouragement to buy or sell any of our securities. The information contained in this information statement is believed by us to be accurate as of the date set forth on its cover. Changes may occur after that date, and we will not update the information except in the normal course of our respective public disclosure obligations and practices.

RISK FACTORS

You should carefully consider each of the following risks and all the other information contained in this information statement. Some of the following risks relate principally to the distribution while other risks relate to our business in general and the industry in which we operate.

Risks Related to Our Separation From Equifax

We have no history operating as an independent company, and we may be unable to make the changes necessary to operate as a stand-alone business, or we may incur greater costs as a stand-alone company that may cause profits to decline. Prior to the distribution, our businesses were operated by Equifax as a separate division within Equifax's broader corporate organization rather than as a stand-alone company. Equifax assisted us by providing financing and a number of corporate support functions such as accounting, financial management, tax, human resources administration, legal, and other general support. Following the distribution, Equifax will have no obligation to assist us other than providing the interim services described in "Relationship Between Equifax and Our Company After the Distribution." We are in the process of creating our own, or engaging third parties to provide, systems and business functions to replace many of the systems and business functions Equifax provides us. We may not be successful in implementing these systems and business functions or in transitioning data from Equifax's systems to ours in a timely and efficient manner. If we do not have in place our own systems and business functions or if we do not have agreements with other service providers once our interim services agreement with Equifax expires, we may not be able to operate our business efficiently. Because our businesses have not been operated as an independent company, operating independently may result in additional operating costs and lower profits.

Our historical financial information may not be representative of our results as a separate company and, therefore, may not be reliable as an indicator of our historical or future results. The historical financial information we have included in this information statement may not reflect what our results of operations, financial position and cash flows would have been had we been a stand-alone company during the periods presented or what our results of operations, financial position and cash flows will be in the future. This is because:

- our combined financial statements reflect allocations, primarily with respect to corporate overhead, for services provided to us by Equifax, which allocations may not reflect the costs we will incur for similar services as a stand-alone company;
- . no allocations have been made in our combined financial statements for debt or related interest expense of Equifax; and
- . the financial information does not reflect changes that we expect to occur in the future as a result of our separation from Equifax, including changes in how we fund our operations and how we allocate income tax expenses.

Therefore, our combined financial statements may not be indicative of our future performance as an independent company. For additional information about our past financial performance and the basis of presentation of our combined financial statements, please see "Selected Financial Data," "Management's Discussion and Analysis of Financial Condition and Results of Operations" and our combined financial statements and the notes thereto included elsewhere in this information statement.

Because there has not been a public market for our common stock, its market price cannot be predicted, and you may not be able to resell your shares at or above the initial market price of our stock following the distribution. There has been no trading market for our common stock. Accordingly, we cannot predict the extent to which investors' interest will lead to a liquid trading market or whether the market price of our common stock will be volatile. Some of the Equifax shareholders who receive our shares may sell them immediately following the distribution, which could delay the development of an orderly trading market in our

shares. Until an orderly market develops, the prices at which our shares trade may fluctuate significantly. In addition, the price of our shares may be depressed until investors have an opportunity to fully familiarize themselves with our business and how it relates to and competes within our industry.

Risks Relating To Our Business and Industry

Our market is highly competitive, and some of our competitors may have substantially greater resources than we do. We face direct competition from third party payment processors and companies that market software for the electronic payment industry. We also compete against software and transaction processing systems developed and used in-house by our potential customers. We cannot assure that competitors will not develop products and services that are superior to or that achieve greater market acceptance than our products and services. We expect the market for our products and services to remain highly competitive. Our failure to remain competitive may have a material adverse effect on our business, financial condition and results of operation. The sizes of competitors vary across market segments, as do the resources we have allocated to the segments we target. Therefore, certain of our competitors may have significantly greater financial, technical, marketing or other resources than we do in each of our market segments or overall. As a result, our competitors may be in a position to respond more quickly than we can to new or emerging technologies and changes in customer requirements, or may devote greater resources than we can to the development, promotion, sale and support of their products and services. Moreover, new competitors or alliances among our competitors may emerge and rapidly decrease our market share. We cannot assure you that we can maintain our competitive position against current and potential competitors, especially those with significantly greater resources than we have, or that competitive pressures will not have a material adverse effect on our business, financial condition and results of operations.

A significant portion of our business is derived from two key strategic relationships, and a loss of either of those relationships could adversely affect our profits. We have long-term contractual alliances with two associations representing independent financial institutions in the U.S., the Independent Community Bankers Association, or ICBA, and Card Services for Credit Unions, or CSCU. Under each of these alliances, which expire in 2004, we are the association's exclusive partner for offering card processing services to that association's members. As a result, approximately 24% of our revenues are derived from ICBA and CSCU member institutions. An early termination of, or significant adverse change in, our relationships with either or both of these associations could harm our ability to retain a substantial portion of our customers and to attract new customers, and have a material adverse effect on our business.

Demand for our products and services is sensitive to the level of consumer transactions effected by our customers, and accordingly, our revenues could be affected negatively by a general economic slowdown or any other event causing a material slowing of consumer spending. A significant portion of our revenue is derived from transaction processing fees. Any changes in economic factors that adversely affect consumer spending and related consumer debt, or a reduction in credit and debit card use, would reduce the volume of transactions that we process, and have an adverse effect on our business, financial condition and results of operations.

We rely heavily on a small number of specific business segments for the majority of our revenues. Revenues of our card processing business are primarily derived from independent community banks and credit unions, while a significant portion of our check risk management services revenues are generated by several large national or regional retail chains. The financial condition of these customers and their willingness to pay for our products and services are affected by general market conditions, competitive pressures, and operating margins within their industry. The retailing industry is currently facing downturns in demand and structural changes in the industry. We could experience declines in revenue as a result of factors adversely affecting independent banks, credit unions, and retailers.

Continued consolidation in the financial services and retail industries could reduce our customer base and cause our sales to fall. Consolidation of our customers could result in a smaller market for our products

and services. After consolidation, banks and other financial institutions and retailers may realign management responsibilities and reexamine strategic and purchasing decisions with potential adverse effects on demand for our products or services. We may lose relationships with key constituencies within our customer's organization due to budget cuts, layoffs, or other disruptions following a consolidation. In addition, consolidation may result in a change in the technological infrastructure of the combined entity. Our products and services may not integrate with this new technological infrastructure. In addition, the acquiring institution may have its own in-house system or outsource to competitors.

To continue to grow profitably, we must expand our share of the credit and debit card transaction processing market and enter new markets, and the failure to do this will adversely affect the value of our shares. Our card processing business is concentrated in the independent community bank and credit union segments of the market and we have achieved a significant degree of penetration of these market segments. While we intend to continue our vigorous pursuit of expansion within these segments, our future growth and profitability will significantly depend upon our ability to penetrate other markets, including emerging markets for electronic transaction processing, such as international transaction processing and Internet payment systems. As part of our strategy to achieve this expansion, we will continue to seek acquisition opportunities, investments and alliance relationships that will facilitate our expansion. We may not be able to complete suitable acquisitions, investments or alliances, and if we do, they may not provide us with the benefits we anticipated. Also, we may not have adequate financial and technological resources to develop products and distribution channels that will satisfy the demands of new markets.

To remain competitive and grow our revenues, we must continually update our products and services, a process which could result in increased research and development costs and the loss of revenues and customers if the new products and services do not perform as intended or are not accepted in the marketplace. The credit and debit card transaction processing and check services markets in which we compete include a wide range of products and services including electronic transaction processing, check authorization and other customer support services. The market is characterized by technological change, new product introductions, evolving industry standards and changing customer needs. In order to remain competitive, we are continually involved in research and development projects. These projects carry the risks associated with any research and development effort, including cost overruns, delays in delivery and performance problems. Our market is constantly experiencing technological changes. A delay in the delivery of new products or services could render them less desirable by our customers, or possibly even obsolete. In addition, the products and services we deliver to the electronic payments market are designed to process transactions and deliver reports and other information on those transactions at very high volumes and processing speeds. Any performance issue that arises with a new product or service could result in significant processing or reporting errors. As a result of these factors, our research and development efforts could result in increased costs that could reduce our revenues and operating profit if promised new products are not timely delivered to our customers, or a loss of revenue or possible claims for damages if new products and services do not perform as anticipated.

Our revenues from the sale of services to VISA and MasterCard organizations are dependent upon our continued VISA and MasterCard certification and financial institution sponsorship, and loss or suspension of this certification and sponsorship could adversely affect our business. In order to provide our transaction processing services, we must be designated a certified processor by, and be a member service provider of, MasterCard and be designated as an independent sales organization of VISA. This designation is dependent upon our being sponsored by member clearing banks of both organizations and our continuing adherence to the standards of the VISA and MasterCard associations. The member financial institutions of VISA and MasterCard, some of which are our competitors, set the standards with which we must comply. If we fail to comply with these standards, our designation as a certified processor, a member service provider or as an independent sales organization could be suspended or terminated. The termination of our member service provider status or our status as a certified processor, or any changes in the VISA and MasterCard rules that prevent our registration or otherwise limit our ability to provide transaction processing and marketing services

for the VISA or MasterCard organizations would result in the loss of business from VISA or Mastercard issuing customers, and lead to a reduction in our revenues.

Our customers are subject to a regulatory environment and to industry standards that may change in a manner that reduces the number of transactions in which our customers engage and therefore reduces our revenues. Our customers are subject to a number of government regulations and industry standards with which our products and services must comply. For example, our products are affected by VISA and MasterCard electronic payment standards that generally are changed twice a year. In addition, action by regulatory authorities relating to credit availability, data usage, privacy or other related regulatory developments could have an adverse effect on our customers and therefore could have a material adverse effect on our business, financial condition and results of operations.

If the security of our databases is compromised, our reputation could suffer and we could lose customers. If the security of our databases is compromised, our business could be adversely affected. We collect personal consumer data, such as names and addresses, social security numbers, drivers license numbers, checking and savings account numbers and payment history records. Unauthorized access to our databases could result in the theft or publication of personal confidential information and the deletion or modification of personal records or otherwise cause interruptions in our operations. These concerns about security are increased when we transmit information over the Internet.

A security or privacy breach may have any one or more of the following effects:

- . deter customers from using our products and services;
- . harm our reputation;
- . expose us to liability; or
- increase our operating expenses to correct problems caused by the breach.

If we experience system failures, the products and services we provide to our customers could be delayed or interrupted, which could harm our business and reputation and result in the loss of customers. Our ability to provide reliable service largely depends on the efficient and uninterrupted operations of our computer network systems and data centers. Any significant interruptions could harm our business and reputation and result in a loss of customers. Our systems and operations could be exposed to damage or interruption from fire, natural disaster, power loss, telecommunications failure, unauthorized entry and computer viruses. Although we have taken steps to prevent a system failure, we cannot be certain that our measures will be successful and that we will not experience system failures. Further, our property and business interruption insurance may not be adequate to compensate us for all losses or failures that may occur.

We plan to continue expansion of our international operations, which will subject us to additional business risks and may cause our profitability to decline due to increased costs. We believe that the international market for our products is growing rapidly, and we expect to commit significant resources to expand our international sales and marketing activities. As we expand internationally, we will be increasingly subject to a number of risks and potential costs, including:

- . political and economic instability;
- . unexpected changes in regulatory requirements and policy, the adoption of laws detrimental to our operations such as legislation relating to the collection of personal data over the Internet or laws, regulations or treaties governing the export of encryption related software;
- . burdens of complying with a wide variety of other laws and regulations;
- . failure to adequately manage currency exchange rate fluctuations;
- potentially adverse tax consequences, including restrictions on the repatriation of earnings;

- . potential difficulty of enforcing agreements and collecting receivables in some foreign legal systems; and
- . general economic conditions in international markets.

We may not be able to overcome these barriers or avoid significant expenditures in addressing these potential risks.

The reliability of market data included in this information statement is uncertain. We have not independently verified this data, and if it is inaccurate the growth potential of our business and markets may be less than anticipated. We have included market data from industry publications in this information statement. The reliability of the data cannot be assured. Market data used throughout this information statement was obtained from internal company surveys and industry publications. Industry publications generally state that the information contained in these publications has been obtained from sources believed to be reliable, but that its accuracy and completeness is not guaranteed. Although we believe market data used in this information statement to be reliable, it has not been independently verified. Similarly, internal company surveys, which we believe to be reliable, have not been verified by independent sources.

Anti-takeover provisions of our articles of incorporation and bylaws, our rights agreement and provisions of Georgia law could delay or prevent a change of control that you may favor. Provisions of our articles of incorporation and bylaws, our rights agreement and provisions of applicable Georgia law, which will be in effect after the distribution, may discourage, delay or prevent a merger or other change of control that shareholders may consider favorable. The provisions of our articles and bylaws, among other things, will:

- . divide our board of directors into three classes, with members of each class to be elected in staggered three-year terms;
- . limit the right of shareholders to remove directors;
- regulate how shareholders may present proposals or nominate directors for election at annual meetings of shareholders; and
- authorize our board of directors to issue preferred shares in one or more series, without shareholder approval.

Our rights agreement is designed to protect our shareholders in the event of an unsolicited offer and other takeover tactics which in the opinion of our board, could impair our board's ability to represent shareholder interests. The provisions of the rights agreement may render an unsolicited takeover of our company more difficult or less likely to occur, or might delay, deter or prevent such a takeover. We are also subject to provisions of Georgia law which may restrict certain business combinations. Please see "Description of Capital Stock" for a more detailed description of these provisions, the rights agreement, and these provisions of Georgia law.

Also, under Section 355(e) of the Internal Revenue Code, the distribution would be treated as a taxable transaction if one or more persons acquire directly or indirectly 50% or more of our or Equifax's stock, measured by vote or value, as part of a plan that is, or series of related transactions that are, linked to the distribution under the rules of Section 355(e). For this purpose, any acquisitions of our stock within two years before or after the distribution are presumed to be part of such a plan, although we may be able to rebut that presumption. If such an acquisition of our stock triggers the application of Section 355(e), under the tax sharing and indemnification agreement, we would be required to indemnify Equifax for the resulting tax. This indemnity obligation might discourage, delay or prevent a change of control that shareholders may consider favorable. Please see "The Distribution--Certain Federal Income Tax Consequences" and "Relationship Between Equifax and Our Company After the Distribution--Tax Sharing and Indemnification Agreement" for a more detailed discussion of Section 355(e) of the Code and the tax sharing and indemnification agreement.

FORWARD-LOOKING STATEMENTS

This information statement contains forward-looking statements that are based on current expectations, estimates, forecasts, and projections about the industry in which we operate, and management's beliefs and assumptions. Such statements include, in particular, statements about our plans, strategies, and prospects under the headings "Summary," "Risk Factors," "Management's Discussion and Analysis of Financial Condition and Results of Operations" and "Business." Words such as "expects," "anticipates," "intends," "plans," "believes," "seeks," "estimates," variations of such words, and similar expressions are intended to identify such forward-looking statements.

These forward-looking statements are not guarantees of future performance and are based on a number of assumptions and estimates that are inherently subject to significant risks and uncertainties, many of which are beyond our control, cannot be foreseen, and reflect future business decisions that are subject to change. Therefore, actual outcomes and results may differ materially from what is expressed or forecasted in such forward-looking statements. The factors that could affect our results include the matters discussed under the heading "Risk Factors"; a change in the growth rate of the overall U.S. economy, or the international economies where we do business, such that consumer spending and related consumer debt are impacted; a decline or change in the marketing techniques of credit card issuers; a reversal of the trend toward credit card use increasing as a percentage of total consumer expenditures; unanticipated cancellation or termination of customer contracts; risks associated with investments and operations in foreign countries, including regulatory environments, exchange rate fluctuations, and local political, social, and economic factors; and the extent to which we can continue successful development and marketing of new products and services.

We caution that such factors are not exclusive. All of the forward-looking statements made in this information statement are qualified by these cautionary statements and readers are cautioned not to place undue reliance on these forward-looking statements, which speak only as of the date of this information statement. Except as required under the federal securities laws and the rules and regulations of the Securities and Exchange Commission, we do not have any intention or obligation to update publicly any forward-looking statements after we distribute this information statement, whether as a result of new information, future events or otherwise.

DIVIDEND POLICY

We do not anticipate paying any dividends on our common stock in the foreseeable future because we expect to retain our future earnings for use in the operation and expansion of our business. The declaration and payment of dividends after the distribution, however, will be at the discretion of our board, and will depend, among other things, upon our investment policy and opportunities, results of operations, financial condition, cash requirements, future prospects, and other factors that may be considered relevant by our board of directors, including restrictions on our ability to declare and pay dividends and distributions on our shares of common stock.

Georgia law prohibits us from paying dividends or otherwise distributing funds to our shareholders, except out of legally available funds. Accordingly, we cannot pay dividends if as a result, we would be unable to pay our debts as they become due in the usual course of business, or our total assets would be less than the sum of our total liabilities plus the amount that would be needed, if we were to be dissolved at the time of the distribution, to satisfy the preferential rights upon dissolution of shareholders whose preferential rights are superior to those receiving the distribution.

CAPITALIZATION

The following table sets forth our combined capitalization as of March 31, 2001, on a historical and pro forma basis, to give effect to the distribution as if it had occurred on March 31, 2001. This table should be read in conjunction with "Management's Discussion and Analysis of Financial Condition and Results of Operations" and our historical and pro forma combined balance sheets, including the notes to those statements, included elsewhere in this information statement. The pro forma information may not necessarily reflect the debt and capitalization of our business in the future or as it would have been had we been a separate, independent company at March 31, 2001, or had the distribution actually been effected on that date.

Based upon the relative financial conditions, results of operations, and prospects of our company and Equifax, Equifax determined that \$275 million of existing debt would create an appropriate starting capitalization for us. Accordingly, we will make a cash payment to Equifax at the time of the distribution of \$275 million. We will use the unsecured revolving credit facilities that we intend to enter into prior to the distribution to fund this cash payment to Equifax.

Historical Pro Forma(1)
(Dollars in thousands)
Debt
Shareholder's Equity Equifax equity investment
Preferred stock, \$.01 par value, 100,000,000 shares authorized, none issued and outstanding (actual) and none issued and outstanding
(pro forma)
(pro forma) 681
Paid-in capital
Total Shareholder's Equity
Total Capitalization

⁽¹⁾ Pro forma combined debt and shareholder's equity at March 31, 2001 presents the financial condition of Certegy as if the distribution had occurred on March 31, 2001.

SELECTED FINANCIAL DATA

Our selected historical combined financial data set forth below should be read in conjunction with the "Combined Financial Statements of the Equifax Payment Services division," including the notes to those statements, and "Management's Discussion and Analysis of Financial Condition and Results of Operations," included elsewhere in this information statement.

The historical income statement data for the years ended December 31, 2000, 1999, and 1998 and the historical balance sheet data as of December 31, 2000 and 1999 are derived from the combined financial statements included elsewhere in this information statement that have been audited by Arthur Andersen LLP, Equifax's independent public accountants. The historical income statement data for the three months ended March 31, 2001 and 2000 and for the years ended December 31, 1997 and 1996 and the historical balance sheet data as of March 31, 2001 and 2000 and December 31, 1998, 1997, and 1996 are derived from unaudited combined financial statements that have been prepared by management and are not included in this information statement. Operating expenses in the historical income statements reflect direct charges of the payment services business together with allocations of certain Equifax corporate expenses that have been charged to us based on usage or other methodologies appropriate for such expenses. In the opinion of management, these allocations have been made on a reasonable basis.

The historical financial information presented below is not necessarily indicative of what our results of operations or financial position would have been had we operated as an independent company during the periods presented, nor is it necessarily indicative of our future performance as an independent company.

Three Months

Selected Financial Data

	Ended Mai	ch 31,		Year Ende				
	2001 2000		2000	1999	1998	1997	1996	
	(1	(Dollars in thousands, except per share data						
Revenues Operating expenses	•	\$177,561 153,651	\$778,562 631,403	\$681,172 553,687	\$566,120 462,829	\$486,603 407,462	\$384,511 320,286	
Operating income Other income (expense),	26,490	23,910	147,159	127,485	103,291	79,141	64,225	
net	(495)	(317)	8	1,410	(916)	(4,330)	(631)	
Income before income taxes and minority interests	25,995	23,593	147,167	128,895	102,375	74,811	63,594	
taxes Minority interests in	(10,138)	(9,235)	(57,609)	(54,272)	(40,505)	(29,240)	(24,476)	
earnings, net of tax	(618)	514	(1,096)	6	(780)		(240)	
Net income	\$ 15,239 ======	\$ 14,872 ======	\$ 88,462 ======	•	•	•	\$ 38,878 ======	
Basic earnings per share(1)	\$ 0.22 \$496,497	\$ 0.22 \$504,800	\$ 1.32 \$502,445	\$ 1.09 \$495,255	\$ 0.86 \$492,704	\$ 0.63 \$273,966	\$ 0.53 \$229,655	

⁽¹⁾ Using a distribution ratio of one share of Certegy common stock for every two shares of Equifax common stock held. Weighted average shares outstanding is computed by applying the distribution ratio to the historical Equifax weighted average shares outstanding for all periods presented.

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

You should read the following discussion in conjunction with "Selected Financial Data," the "Combined Financial Statements of the Equifax Payment Services division," including the notes to those statements, the "Pro Forma Combined Financial Statements of the Equifax Payment Services division," including the notes to those statements, and the other financial information included elsewhere in this information statement. This discussion contains forward-looking statements. Please see "Forward-Looking Statements" and "Risk Factors" for a discussion of the uncertainties, risks and assumptions associated with these statements.

Overview

We provide credit and debit card processing and check risk management services to financial institutions and merchants throughout the world. Our business is comprised of two segments, Card Services and Check Services. Card Services provides credit and debit card issuer services, merchant processing services, and e-banking services in the U.S., the U.K., Brazil, and Chile. Check Services provides check risk management services and related processing services in the U.S., the U.K., Canada, France, Ireland, Australia, and New Zealand.

Our card issuer services enable banks, credit unions, retailers, and others to issue VISA and MasterCard credit and debit cards, private label cards, and other electronic payment cards. Our merchant processing services enable retailers and other businesses to accept credit, debit, and other electronic payment cards from purchasers of their goods and services. Our e-banking services enable banks to provide electronic banking services to their customers, allowing them to compete for and retain customers more effectively and to generate non-interest fee income.

Our check risk management services, which utilize our proprietary check authorization systems and risk assessment decision platforms, enable retailers, hotels, automotive dealers, telecommunications companies, supermarkets, casinos, mail order houses, and other businesses to minimize losses from dishonored checks, maximize check acceptance, and improve customer service. Our services include check guarantee, where we accept the risk of bad checks presented to our customers, verification services, where we determine the likelihood that a check will clear and the customer retains the risk, and certain combinations of guarantee and verification services. We also provide related service offerings, including risk management consulting and marketing services, which enable retailers to cross-sell and increase their customer retention.

Separation from Equifax

We were incorporated under the laws of the State of Georgia on March 2, 2001, as a wholly-owned subsidiary of Equifax. We will have no material assets or activities until the contribution to us by Equifax of the payment services businesses described in this information statement, which is expected to occur prior to the distribution and will be effective as of 11:59 p.m., Atlanta, Georgia, time June 30, 2001. Equifax conducts these businesses through various operating groups and subsidiaries. After the distribution, we will be an independent public company, and Equifax will have no continuing stock ownership interest in us.

Our combined financial statements have been prepared on the historical cost basis in accordance with accounting principles generally accepted in the U.S., and reflect the historical financial position, results of operations, and cash flows of the businesses to be transferred to us from Equifax in connection with the distribution. The financial information included in this information statement, however, is not necessarily indicative of what our results of operations or financial position would have been had we operated as an independent company during the periods presented, nor is it necessarily indicative of our future performance as an independent company.

Intercompany transactions between entities included in the combined financial statements have been eliminated. We have been allocated certain Equifax corporate expenses based on an estimate of the proportion of corporate expenses allocable to us, utilizing such factors as revenues, number of employees, and other relevant factors. We believe that these allocations have been made on a reasonable basis. We believe that, had we been operating on a stand-alone basis, we would have incurred additional annual expenses of approximately \$6.5 million, which specifically relate to incremental pension expense, insurance costs, corporate headquarters rent, and stand-alone public company costs for audit, director and stock exchange fees. The increase in pension expense is a result of us no longer benefitting from the over-funded status of the consolidated Equifax pension plan. We believe that all other costs allocated to us are a reasonable representation of the costs that we would have incurred if we had performed these functions as a stand-alone company.

In conjunction with the separation of our businesses from the other businesses of Equifax, we will enter into various agreements with Equifax that address the allocation of assets and liabilities and that define our relationship with Equifax after the separation, including a distribution agreement, tax sharing and indemnification agreement, employee benefits agreement, intercompany data purchase agreement, intellectual property agreement, transition support agreement, and agreement regarding leases.

Components of Income Statement

We generate revenue from (i) charges based on transaction volume, accounts, or cards processed and fees for various services and products within Card Services, and (ii) charges based upon transaction volume and fees for various services and products within Check Services. Revenues depend upon a number of factors, such as demand for and price of our services, the technological competitiveness of our product line, our reputation for providing timely and reliable service, competition within our industry, and general economic conditions.

Costs of services consist primarily of the costs of transaction processing systems, personnel to develop and maintain applications and operate computer networks and to provide customer support, losses on check guarantee services, interchange and other fees on merchant processing, and depreciation and occupancy costs associated with the facilities performing these functions. Our selling, general and administrative expenses consist primarily of salaries, wages and related expenses paid to sales, non-revenue customer support functions and administrative employees and management, and certain Equifax corporate costs that have been allocated to us.

Results of Operations

The following table summarizes our combined results for the three months ended March 31, 2001 and 2000 and for each year in the three-year period ended December 31, 2000.

	Three Months Ended March 31,				Year Ended Dece				mber 31,	
		2001		2000	- :	2000		1999	:	1998
		(In mill:	ior	is, except	t į	per sha	re	amount	s)	
Revenue Operating income			-	177.6 23.9	-		-		-	
Other income (expense), net	\$	(0.3) (0.2) 15.2	\$	(0.1)	\$	1.3 (1.3) 88.5	\$	(0.9)	\$	(0.5)
Basic earnings per share	\$	0.22	\$	0.22	\$	1.32	\$	1.09	\$	0.86

Revenue

First Quarter 2001 compared with First Quarter 2000

Our revenue in the first quarter of 2001 of \$195.0 million increased by \$17.4 million, or 9.8%, over first quarter of 2000. Card Services revenue grew 10.1% and Check Services experienced revenue growth of 9.3%.

The growth in revenue was driven by increased volumes and was partially offset by changes in foreign exchange rates. The strengthening of the U.S. dollar against foreign currencies, particularly the British pound and the Brazilian real, reduced revenue growth by \$4.3 million.

Year 2000 compared with Year 1999

Our revenue in 2000 of \$778.6 million increased by \$97.4 million, or 14.3%, over 1999. Card Services revenue grew 16.8% in 2000, while Check Services experienced revenue growth of 9.6% in 2000.

The growth in revenue was driven by increased volumes, in part from acquisitions, and was partially offset by changes in foreign exchange rates. The acquisition of a Chilean card processing operation in January 2000 accounted for \$5.1 million of revenue growth in 2000. The start-up of a card processing operation in the U.K. during June 1999 contributed \$30.0 million and \$13.7 million of total revenue in 2000 and 1999, respectively. The strengthening of the U.S. dollar against foreign currencies, particularly the British pound, reduced revenue growth by \$6.2 million.

Year 1999 compared with Year 1998

Our revenue in 1999 of \$681.2 million increased by \$115.1 million, or 20.3%, over 1998. Our Card Services revenue grew 24.3% and Check Services experienced revenue growth of 13.5%.

The growth in revenues was driven by increased volumes, in part from acquisitions, and was partially offset by changes in foreign exchange rates. The acquisition of a Brazilian card processing operation in September 1998 accounted for \$35.7 million of the 1999 revenue growth and the start-up of our card processing operation in the U.K. during June 1999 contributed \$13.7 million of revenue growth. The strengthening of the U.S. dollar against foreign currencies, particularly the Brazilian real, reduced revenue growth by \$12.6 million.

Operating Expenses

First Quarter 2001 compared with First Quarter 2000

Our total operating expenses in first quarter 2001 of \$168.5 million increased \$14.8 million, or 9.7%, over first quarter 2000. Card Services operating expenses grew \$6.6 million, or 6.5%, Check Services experienced operating expense growth of \$8.1 million, or 16.3%, and Corporate expense of \$2.2 million increased \$.1 million, or 8.0%.

Cost of services increased by \$14.3 million, or 11.1%, over first quarter 2000, principally driven by increased volumes in both business segments, including higher guarantee loss rates in Check Services. An increase in card merchant and issuing volume added \$6.5 million of cost and higher check volume and loss rates added \$2.3 million and \$5.5 million of cost, respectively.

Selling, general and administrative expense increased \$.5 million, or 2.0%, over first quarter 2000.

Year 2000 compared with Year 1999

Our total operating expenses in 2000 of \$631.4 million increased \$77.7 million, or 14.0%, over 1999. Card Services operating expenses grew \$60.2 million, or 17.3%, Check Services experienced operating expense growth of \$17.0 million, or 8.5%, and Corporate expense of \$7.8 million increased \$.5 million, or 6.5%.

Cost of services increased by \$69.4 million, or 14.9%, over 1999, principally driven by increased volumes in both business segments. An increase in card merchant and issuing volume added \$65.6 million of cost and higher check volume added \$3.8 million of cost.

Selling, general and administrative expense increased \$8.3 million, or 9.6%, largely attributable to increased selling expenses within Check Services.

Our total operating expenses in 1999 of \$553.7 million increased \$90.9 million, or 19.6%, over 1998. Card Services operating expenses grew \$69.7 million, or 25.1%, Check Services experienced operating expense growth of \$20.7 million, or 11.6%, and Corporate expense of \$7.3 million increased \$.5 million, or 7.4%.

Cost of services increased by \$81.4 million, or 21.2%, over 1998, principally driven by increased volumes in both business segments. An increase in card merchant and issuing volume added \$67.3 million of cost and higher check volume added \$14.1 million of cost.

Selling, general and administrative expense increased \$9.4 million, or 12.1%, largely attributable to increased selling expenses within Check Services and higher administrative expenses within Card Services.

Operating Income

First Quarter 2001 compared with First Quarter 2000

Operating income of \$26.5 million in the first quarter of 2001 increased by \$2.6 million, or 10.8%, over the first quarter of 2000. Combined operating margins were 13.6% in the 2001 quarter and 13.5% in the 2000 quarter. Our increased operating income has been principally driven by revenue growth and improved profits of Card Services, but has been negatively impacted by decreased profits of Check Services caused by higher comparative check guarantee loss rates.

Year 2000 compared with Year 1999

Operating income of \$147.2 million in 2000 increased \$19.7 million, or 15.4%, over 1999. Combined operating margins were 18.9% in 2000 and 18.7% in 1999. Our increased operating income has been principally driven by revenue growth.

Year 1999 compared with Year 1998

Operating income of \$127.5 million in 1999 increased \$24.2 million, or 23.4%, over 1998. Combined operating margins were 18.7% in 1999 and 18.2% in 1998. Our increased operating income has been principally driven by revenue growth.

Other Income (Expense), Net

Other income (expense) principally consists of gains on sales of businesses and net foreign exchange losses. In year 2000, other income includes a \$2.2 million pre-tax gain on the sale of our investment in a card processing operation in India. In year 1999, other income includes a \$3.0 million pre-tax gain on the sale of a minority owned business in Brazil.

Interest Expense

Interest expense principally consists of interest paid on a line of credit held by Unnisa, our card processing business in Brazil, and interest charged by Equifax on overnight funds borrowed on our behalf. We have not been allocated any Equifax corporate debt or related interest expense as these amounts have not historically been allocated to the operating divisions by Equifax. Based on current rates, interest expense is expected to increase by approximately \$16.0 million, representing the annual interest at a rate of LIBOR plus 100 basis points on the \$275 million of debt.

Effective Tax Rate

We are included in the consolidated federal income tax return of Equifax. Federal and certain state tax provisions are settled through intercompany accounts, and Equifax makes income tax payments on our behalf. The provision for income taxes in the accompanying combined statements of income reflects federal, state, and foreign taxes calculated using the separate return basis. Our effective tax rates were 39.1%, 42.1%, and 39.6%

in 2000, 1999, and 1998, respectively. Our 1999 effective tax rate was negatively impacted by basis difference of a minority owned business in Brazil that was sold during the year. Our effective tax rate in the first quarter of 2001 was 39.0%, which is the expected rate for the entire year.

Net Income and Basic Earnings per Share

Net income increased 2.5% in the first quarter of 2001, 18.5% in year 2000 and 22.2% in year 1999, driven primarily by revenue growth. The first quarter growth rate was negatively impacted by higher comparative check loss rates and a \$1.1 million change in minority interest in earnings, net of tax, of international card operations. The 2001 quarter includes \$.6 million of earnings reduction related to minority ownership in the profits of our Brazilian card operation. The 2000 quarter includes \$.5 million of earnings increase related to minority ownerships in the net loss of our Brazilian and U.K. card operations. We acquired full ownership of our U.K. card operation in September 2000 and our Brazilian card operation in May 2001.

Basic earnings per share is calculated by dividing net income by the weighted average number of common shares. Weighted average shares outstanding is computed by applying the distribution ratio of .5 shares of our common stock to the historical Equifax weighted average shares outstanding for the same periods presented.

Diluted earnings per share is calculated to reflect the potential dilution that would occur if stock options or other contracts to issue our common stock were exercised and resulted in additional common shares outstanding. Diluted earnings per share is not presented in the combined financial statements, because we have no historical market share prices for our common stock, as regular way public trading will not commence until the first trading day after the distribution occurs. Accordingly, the dilutive effect of stock options cannot be determined. At December 31, 2000, the number of shares of Equifax common stock subject to options held by option holders expected to become Certegy employees was 2,369,000. The exercise prices of those options range from \$7.09 to \$38.60.

Segment Results

The following table summarizes the segment results for the three months ended March 31, 2001 and 2000 and for each year in the three-year period ended December 31, 2000.

	Three Months Ended March 31,				Year Ended December 31,					
	Revenue		Operating Income		Revenue			Operating Income		
	2001	2000	2001	2000	2000	1999	1998	2000	1999	1998
	(In millions)									
Card Services						-	-	•		\$ 79.2 30.9
General Corporate	195.0	177.6	28.7	26.0	778.6	681.2	566.1	155.0	134.8	110.1
Expense			(2.2)	(2.1)				(7.8)	(7.3)	(6.8)
	\$195.0	\$177.6 	\$26.5	\$23.9	\$778.6	\$681.2	\$566.1	\$147.2 	\$127.5 	\$103.3

Based upon current forecasts, management expects that the future relative contribution to revenue and profits, and capital requirements of these segments, will remain consistent with historical percentages.

Card Services

Our Card Services business includes credit and debit card processing operations in the U.S., the U.K., Brazil, and Chile, and our card software operation that principally supports our international card processing platforms. Over the past three years, Card Services has pursued growth in international markets. In September 1998, Card Services expanded its operations into Latin America by acquiring a 59.3% interest in Unnisa, a card processing business in Brazil. We acquired full ownership of Unnisa in May 2001. In June 1999, we started a U.K. card processing operation, owning a 51% interest. We acquired full ownership of our U.K. operation in September 2000. In January 2000, we acquired Procard, Chile's second largest credit card processor. Also in 2000, we entered into a five-year agreement with the National Australia Bank to process cards issued in Australia, New Zealand, the U.K., and Ireland, starting in the second quarter of 2001. National Australia Bank

will be serviced from a new card processing operation that we are establishing in Australia, as well as from our U.K. card processing operation. Card Services plans to utilize the Australian operation to pursue further card processing opportunities in the Asian and Pacific Rim markets.

First Quarter 2001 compared with First Quarter 2000

Card Services' revenue increased \$12.0 million in first quarter 2001, a 10.1% growth over first quarter 2000. Revenue in the U.S. of \$104.7 million in the 2001 quarter grew from \$93.5 million in the prior year quarter, driven by increases in card issuing transactions and merchant volumes. Merchant processing contributed approximately \$8.5 million of the U.S. revenue growth in 2001.

Our international revenue of \$26.5 million in the 2001 quarter grew 2.7% over the 2000 quarter. International revenue includes software licensing revenue, which has declined from \$2.8 million in the 2000 quarter to \$1.7 million in the 2001 quarter. Card Services has de-emphasized software licensing as we grow our global processing operations, which will utilize this proprietary software to run our international card processing operations. Partially offsetting international revenue growth was the strengthening of the U.S. dollar. Exchange rate changes of the Brazilian real and the British pound reduced revenue growth by approximately \$3.1 million. On a local currency basis, international revenue increased approximately 20.9%.

Card Services' operating income increased 30.8% in the 2001 quarter. Operating margins were 17.2% in the 2001 quarter and 14.5% in the 2000 quarter. Operating margins have benefited from revenue increasing at a higher rate than costs in both the U.S. and international operations.

Year 2000 compared with Year 1999

Card Services' revenue increased \$74.6 million in 2000, a 16.8% growth over 1999. Revenue in the U.S. of \$403.8 million in 2000 was an increase of 14.7%, driven by increases in card issuing transactions and merchant volumes. Merchant processing contributed approximately \$37.4 million of the U.S. revenue growth in 2000.

Our international revenue of \$114.7 million in 2000 grew 25.1% over 1999, as the number of cards processed by our international operations increased from 7.8 million at the end of 1999 to approximately 13.3 million at the end of 2000. Our acquisition of a Chilean card processing operation in January 2000 accounted for \$5.1 million of our revenue growth. The start-up of a card processing operation in the U.K. during June 1999 contributed \$30.0 million and \$13.7 million of revenue in 2000 and 1999, respectively. We expect total international cards processed by us to increase to approximately 16.4 million with the commencement of our Australian operation in 2001. International revenue includes software licensing revenue, which has declined from \$23.2 million in 1999 to \$13.0 million in 2000. Card Services has de-emphasized software licensing as it grows its global processing operations, which will utilize this proprietary software to run our international card processing operations. Also partially offsetting international revenue growth was the strengthening of the U.S. dollar in 2000. Exchange rate changes of the Brazilian real and the British pound reduced revenue growth by approximately \$3.0 million in 2000.

Card Services' operating income increased 14.9% in 2000, principally driven by U.S. operations. Operating margins were 21.4% in 2000 and 21.7% in 1999. Reduction of card software licensing, start-up losses of certain international operations, and higher levels of low-margin merchant processing revenues have had negative effects on overall operating margins in 2000.

Year 1999 compared with Year 1998

Card Services' revenue increased by \$86.9 million in 1999, a 24.3% growth over 1998. Revenue in the U.S. of \$352.2 million in 1999 was an increase of 15.6%, driven by increases in card issuing transactions and merchant volumes. Merchant processing contributed approximately \$31.0 million of the U.S. revenue growth in 1999.

Our international revenue of \$91.7 million in 1999 grew 75.7% over 1998, as the number of cards processed by our international operations increased from 2.6 million at the end of 1998 to approximately 7.8 million at the end of 1999. Acquisition of our Brazilian card processing operation in September 1998 accounted for \$54.8 million and \$25.1 million of revenue in 1999 and 1998, respectively. The start-up of our card processing operation in the U.K. during June 1999 contributed \$13.7 million of revenue in 1999. Software licensing revenue declined from \$27.1 million in 1998 to \$23.2 million in 1999. Also partially offsetting international revenue growth was the strengthening of the U.S. dollar in 1999. Exchange rate changes of the Brazilian real and the British pound reduced revenue growth by approximately \$11.5 million in 1999.

Card Services' operating income increased 21.7% in 1999, principally driven by U.S. operations. Operating margins were 21.7% in 1999 and 22.2% in 1998. Reduction of card software licensing, start-up losses of certain international operations, and higher levels of low-margin merchant processing revenues have had negative effects on overall operating margins in 1999.

Check Services

Our Check Services business includes check risk management operations in the U.S., the U.K., Ireland, France, Australia, New Zealand and Canada. We believe check-writing has begun to decline as a total percentage of point-of-sale payments due, in part, to the growing use of debit cards. At the same time, however, demand for our services is strong due to factors that include increasing sophistication of check fraud and higher concentration of bad checks written at the point-of-sale due to a trend of higher credit quality consumers paying more with credit and debit cards and writing fewer checks. These factors are contributing to a growing reliance of retailers and other businesses on outside vendors, such as ourselves, to provide check risk management services.

First Quarter 2001 compared with First Quarter 2000

The revenues of Check Services increased by \$5.5 million in the first quarter of 2001, a 9.4% growth over the first quarter of 2000. Revenue in the U.S. of \$51.7 million in the 2001 quarter was an increase of 12.2%, driven by increased volume largely resulting from the addition of new customers. The face amount of checks authorized in the U.S. totaled \$6.0 billion in the 2001 quarter and \$5.6 billion in the 2000 quarter.

Check Services' international revenue of \$12.1 million in 2001 approximated 2000 revenue of \$12.3 million. On a local currency basis, international revenue increased approximately 7.7%. The strengthening of the U.S. dollar against the British pound reduced international check revenue growth by \$1.2 million. The face amount of checks authorized in the international operations totaled \$.7 billion in both the first quarter of 2001 and 2000.

Check Services' operating income decreased 29.7% in the 2001 quarter. Operating margins were 9.6% in the 2001 quarter and 14.9% in the 2000 quarter. The decline in profitability is attributed to higher check guarantee loss rates in the 2001 quarter.

Year 2000 compared with Year 1999

The revenues of Check Services increased by \$22.8 million in 2000, a 9.6% growth over 1999. Revenue in the U.S. of \$209.2 million in 2000 was an increase driven by increased volume largely resulting from the addition of new customers. The face amount of checks authorized in the U.S. totaled \$25.7 billion in 2000 and \$23.5 billion in 1999.

Our international revenue of \$50.9 million in 2000 grew 1.4% over 1999. On a local currency basis, revenue increased 6.6%, driven by increased volumes. The face amount of checks authorized in the international operations totaled \$3.2 billion in 2000 and \$2.9 billion in 1999. The strengthening of the U.S. dollar against the British pound reduced international check revenue growth by \$3.2 million in 2000.

Check Services' operating income increased 15.1% in 2000. Operating margins were 17.0% in 2000 and 16.2% in 1999. Increased operating income and margin improvements have been driven by higher volumes, operating efficiencies achieved through process automation, and improved risk management in the international operations.

Year 1999 compared with Year 1998

The revenues of Check Services increased by \$28.2 million in 1999, a 13.5% growth over 1998. Revenue in the U.S. of \$187.1 million in 1999 was an increase of 16.1%, driven by increased volume largely resulting from the addition of new customers. The face amount of checks authorized in the U.S. totaled \$23.5 billion in 1999 and \$19.8 billion in 1998.

Our international revenue of \$50.2 million in 1999 grew 4.6% over 1998. On a local currency basis, revenue increased 6.9%, driven by improved pricing. The face amount of checks authorized in the international operations totaled \$2.9 billion in 1999 and \$3.0 billion in 1998. The strengthening of the U.S. dollar against the British pound reduced international check revenue growth by \$1.1 million in 1999.

Check Services' operating income increased 24.3% in 1999. Operating margins were 16.2% in 1999 and 14.8% in 1998. Increased operating income and margin improvements have been driven by higher volumes, operating efficiencies achieved through process automation, and improved risk management and pricing in the international operations.

General Corporate

General corporate expense of \$2.2 million and \$2.1 million in the first quarters of 2001 and 2000, respectively, and \$7.8 million in 2000, \$7.3 million in 1999, and \$6.8 million in 1998 represent certain Equifax corporate expenses that have been allocated to us based on our proportionate amount of revenue, number of employees, and other relevant factors as compared to related totals for Equifax. We believe that these allocations have been made on a reasonable basis. We believe that, had we been operating on a stand-alone basis, we would have incurred additional annual expenses of approximately \$6.5 million, which specifically relate to incremental pension expense, insurance costs, corporate headquarters rent, and stand-alone public company costs for audit, director and stock exchange fees. The increase in pension expense is a result of us no longer benefitting from the over-funded status of the consolidated Equifax pension plan. We believe that all other costs allocated to us are a reasonable representation of the costs that we would have incurred if we had performed these functions as a stand-alone company.

Liquidity and Capital Resources

First Quarter 2001 compared with First Quarter of 2000

Net cash provided by operating activities amounted to \$7.2 million in the first quarter of 2001 as compared with \$35.7 million in the first quarter of 2000. The 2001 amount was reduced by \$31.9 million related to the timing of settlements in the card and merchant processing clearing system. Operating activities provided cash of \$29.6 million in the 2001 quarter and \$26.2 million in the 2000 quarter before the effect of this settlement activity. Cash held by Equifax associated with this settlement process was \$6.7 million and \$59.9 million at March 31, 2001 and 2000, respectively. These amounts are included in the intercompany receivables from Equifax, which are a component of the Equifax equity investment. Operating cash flow has been sufficient to fund capital expenditures.

Net cash used in investing activities amounted to \$12.3 million in 2001 and \$21.0 million in 2000. Capital expenditures, exclusive of acquisitions and investments, amounted to \$12.3 million in 2001 and \$8.5 million in 2000. We expect that total capital expenditures will be approximately \$40 million in 2001. Acquisitions, net of cash acquired, and other investments totaled \$12.5 million in 2000.

Net cash provided by (used in) financing activities amounted to \$2.2 million in 2001 and \$(10.9) million in 2000. Net borrowings from and (repayments to) Equifax amounted to \$2.3 million and \$(10.7) million in 2001 and 2000, respectively.

Year 2000 compared with Year 1999

Net cash provided by operating activities amounted to \$103.8 million in 2000 as compared with \$146.2 million in 1999. The \$42.4 million decline in 2000 was due primarily to the timing of settlements in the card and merchant processing clearing system, which accounted for \$46.4 million of the change in operating cash flow in 2000 versus 1999. Operating activities provided cash of \$125.1 million in 2000 and \$121.2 million in 1999 before the effect of this settlement activity. Cash held by Equifax associated with this settlement process was \$29.0 million and \$50.4 million at December 31, 2000 and 1999, respectively. These amounts are included in the intercompany receivables from Equifax, which are a component of the Equifax equity investment. Operating cash flow has been sufficient to fund capital expenditures, exclusive of acquisitions, in 2000 and

Net cash used in investing activities amounted to \$78.2 million in 2000 and \$30.9 million in 1999. Capital expenditures, exclusive of acquisitions and investments, amounted to \$38.8 million in 2000 and \$50.1 million in 1999. Cash used for acquisitions and other investments, net of cash acquired, totaled \$46.3 million in 2000. In 1999, we received \$2.0 million of cash related to the final purchase price determination of a 1998 acquisition. Cash proceeds from the sale of businesses and other assets amounted to \$6.9 million in 2000 and \$17.9 million in 1999.

Net cash used in financing activities amounted to \$26.8 million in 2000 and \$103.7 million in 1999. Net repayments to Equifax amounted to \$26.4 million and \$106.1 million in 2000 and 1999, respectively.

Year 1999 compared with Year 1998

Net cash provided by operating activities amounted to \$146.2 million in 1999 as compared with \$72.7 million in 1998. Approximately \$43.6 million of the change in operating cash flow in 1999 versus 1998 is due to the timing of settlements in the card and merchant processing clearing system. Operating activities provided cash of \$121.2 million in 1999 and \$91.2 million in 1998 before the effect of this settlement activity. Cash held by Equifax associated with this settlement process was \$50.4 million and \$25.4 million at December 31, 1999 and 1998, respectively. These amounts are included in the intercompany receivables from Equifax, which are a component of the Equifax equity investment. Operating cash flow has been sufficient to fund capital expenditures, exclusive of acquisitions, in 1999 and 1998.

Net cash used in investing activities amounted to \$30.9 million in 1999 and \$205.1 million in 1998. Capital expenditures, exclusive of acquisitions and investments, amounted to \$50.1 million in 1999 and \$47.9 million in 1998. Cash used for acquisitions and other investments, net of cash acquired, totaled \$157.2 million in 1998. In 1999, we received \$2.0 million of cash related to the final purchase price determination of a 1998 acquisition. Cash proceeds from the sale of businesses and other assets amounted to \$17.9 million in 1999.

Net cash (used in) provided by financing activities amounted to \$(103.7) million in 1999 and \$136.5 million in 1998. Net (repayments to) and borrowings from Equifax amounted to \$(106.1) million and \$136.7 million in 1999 and 1998, respectively. The borrowings in 1998 were required to fund acquisition activity.

As described in Note 9 to the combined financial statements, we had \$37.9 million in lease commitments as of December 31, 2000. In addition, under the terms of an agreement, the minority shareholders of Unnisa Ltda., our credit card processing business in Brazil, exercised an option to sell their 40.7% interest to us. We purchased this interest in May 2001 for approximately \$55 million.

We have obtained commitments for unsecured revolving credit facilities, a portion of which we expect will fund a cash payment to Equifax of \$275 million. We believe that our current level of cash and cash

equivalents, \$24.8 million as of March 31, 2001, future cash flows from operations, and the amount of the unsecured revolving credit facilities in excess of the \$275 million payment described above, will be sufficient to meet the needs of our existing operations and planned requirements for the foreseeable future. The increased interest expense related to this borrowing, as previously described, will generate an after-tax use of cash of approximately \$9.8 million.

We regularly evaluate cash requirements for current operations, development activities, and acquisitions. We may elect to raise additional funds for these purposes, either through further bank financing or the public capital markets, as appropriate. Based on our recent financial results and current financial position, we believe that additional funding will be available if required to meet our capital requirements.

Seasonality, Inflation, and Economic Downturns

We are subject to the impact of general economic conditions; however, this has historically been mitigated by the continued demand for payment transaction processing. We are also subject to certain seasonal fluctuations such as peak activity during the holiday buying season.

We do not believe that the rate of inflation has had a material effect on our operating results. However, inflation could adversely affect our financial results were inflation to result in a substantial weakening in economic conditions that adversely affect the level of consumer spending.

Market Risk

Approximately 21% of our revenue for the year ended December 31, 2000 and 41% of our assets at the end of 2000 are associated with operations outside the U.S. The U.S. dollar balance sheets and statements of income for these businesses are subject to currency fluctuations. We are most vulnerable to fluctuations in the Brazilian real and the British pound against the U.S. dollar. Historically, we have not entered into derivative financial instruments to mitigate this risk, as it has not been cost-effective. The impact of currency fluctuations upon profitability has not been significant since both revenue and costs of these businesses are denominated in local currency. If the U.S. dollar had a 10% higher appreciation against our non-U.S. dollar denominated businesses in 2000, revenues and operating income would have been negatively impacted by \$15.2 million and \$.8 million, respectively. We may use derivative financial instruments in the future if we deem it useful in mitigating an exposure to foreign currency exchange rates. The cumulative translation adjustment, largely related to our investment in the Brazilian card processing operation, was a \$57.3 million and \$48.5 million reduction of shareholder's equity at December 31, 2000 and 1999, respectively, and \$69.3 million at March 31, 2001.

Overview

We provide credit and debit card processing and check risk management services to financial institutions and merchants throughout the world. Last year, we processed over 1.7 billion payment transactions, serviced over 27 million card accounts, and authorized over \$28 billion of check transactions worldwide. Our business is comprised of two segments, Card Services and Check Services. Card Services provides credit and debit card processing services, and e-banking services. Check Services provides check risk management services.

Card Services. We provide card issuer services that enable banks, credit unions, retailers and others to issue VISA and MasterCard credit and debit cards, private label cards, and other electronic payment cards. We provide merchant processing services that enable retailers and other businesses to accept credit, debit and other electronic payment cards from purchasers of their goods and services.

We offer a broad range of processing services for card issuers, ranging from full service card programs, to more limited transaction processing services. The majority of our card issuer customers subscribe to our full service programs. We refer to them as full service because we provide essentially all of the operations and support necessary to support a card issuer's credit and debit card issuing program, including cardholder transaction processing, authorization, and back-office support functions. These back-office functions include, among others, invoicing the credit cardholders, receiving and posting cardholder payments, and providing customer service. Our services are menu driven, and offer flexibility for those of our customers that require less than our full service program. Such customers include large card issuing banks that contract with us to provide transaction processing, but choose to invest the capital and human resources necessary to provide their own back-office program support.

Card Services' merchant processing services include front-end authorization and data capture services, and back-end accounting and settlement services. We provide these services both directly to retailers and other merchants that accept electronic payment cards, and through contracts with financial institutions and others where our products and services enable them to service the card processing needs of their merchant customers.

In addition to our card processing programs, Card Services e-banking business enables banks to provide electronic banking services to their customers, allowing them to compete for and retain customers more effectively and to generate non-interest fee income.

Check Services. We are a leading provider of check risk management and related processing products and services. Our check risk management products and services, which utilize our proprietary check authorization systems and risk assessment decision platforms, enable retailers, hotels, automotive dealers, telecommunications companies, supermarkets, casinos, mail order houses, and other businesses to minimize losses from dishonored checks, maximize check acceptance, and improve customer service.

Our diverse and flexible portfolio of check risk management services allows us to tailor solutions to meet the specific needs of the customer. The services include check guarantee, where we accept the risk of bad checks presented to our customers, verification services, where we determine the likelihood that a check will clear and the customer retains the risk, and certain combinations of guarantee and verification services. We also provide related service offerings, including risk management consulting and marketing services, which enable retailers to cross-sell and increase their customer retention.

Our History. We and our predecessors have been providing transaction processing solutions for nearly four decades. Our check services business, originally founded as Telecredit in 1961, pioneered the check risk management industry in the U.S. and Canada by establishing the first centralized electronic database of consumer check-writing histories from which real-time check authorization decisions were delivered to merchants at the point-of-sale. In 1974, Telecredit acquired its card services business, which at the time provided credit card processing services principally for Florida banks and merchants. During the ensuing years

and prior to Equifax's acquisition of Telecredit in 1990, Telecredit expanded its card services business nationally by establishing contractual alliances with two premier associations representing independent financial institutions. These associations are the Independent Community Bankers Association, or ICBA, then known as the Independent Bankers Association of America, and Card Services for Credit Unions, or CSCU. Under each alliance, Telecredit became the association's exclusive partner for offering card processing services to that association's members. Since Equifax's acquisition of Telecredit in 1990, our core businesses have continued to develop through organic growth and strategic acquisitions.

In 1992, Equifax expanded our international check services business into Europe, Australia, and New Zealand through a joint venture with Transax, Europe's largest check risk management company. In 1996, we acquired full ownership of the Transax business, giving our Check Services business a large share of the European market and solidifying our position as one of the leading check risk management companies worldwide.

In 1994, Equifax acquired First Bankcard Systems, an Atlanta-based company that had developed one of the world's most advanced credit card processing software systems. First Bankcard Systems was a leading provider of card processing software serving several major financial institutions and processors throughout the world. We have installed the FBS software in our U.S., U.K., and Brazil processing centers, allowing us to offer our customers the card program features and functions necessary for them to maintain competitive card programs.

In 1998, Equifax expanded our operations into Latin America by acquiring a controlling interest in Unnisa, one of the larger card processing businesses in Brazil. We commenced our U.K. card processing operations in June 1999 through a partnership with Grupe Cofinaoga an Banque Nationale de Paris. In January 2000, we entered the Chilean market with Equifax's acquisition of Procard, the second largest credit card processor in Chile. In the second quarter of 2000, we signed a long-term contract with the National Australia Bank, one of the largest card issuing financial institutions in Australia, to process approximately 4.5 million card accounts issued in Australia, New Zealand, Ireland, and the U.K.

Industry Opportunity

Card Services

We believe that the escalating use of credit, debit, and other electronic payment cards around the globe will continue to present the card processing industry with significant growth opportunities. Increasing demand for e-banking services should create significant opportunities as well.

The U.S. Market. Card Services' target market in the U.S. consists of banks, credit unions, other card issuers, and retailers. Within these markets, we believe we have established ourselves as the leader, based on market share, in providing comprehensive card processing services to credit unions and to independent community bank card issuers. We process for approximately 4,000 U.S. credit unions and approximately 2,200 independent community banks. Card Services provides its services to these two customer groups principally through its long-term contractual alliances with CSCU and ICBA. Each has made us their exclusive partner for offering card issuer and merchant processing services to their credit union and community bank members.

The credit union and community bank markets served by Card Services are large and growing. There are approximately 10,700 credit unions in the U.S., serving an aggregate membership base of approximately 75 million customers. We estimate that approximately 6,000 of those credit unions have their own card programs, and we provide card issuer processing services for approximately 4,000 of those, which in the aggregate, have approximately 14 million cardholders. We currently estimate that the number of people who belong to a credit union in the U.S. is growing at a rate of 3 to 4 percent annually.

There are approximately 10,300 independent community banks in the U.S., approximately 5,300 of which are ICBA members. We provide full service card issuer processing for approximately 2,200 community banks, most of which are ICBA members, including banks that issue cards directly and banks that issue cards through an agent relationship with a lead bank that funds the agent bank's credit card loans.

We believe that overall growth in the number of credit cards issued in the U.S. market will slow over the next few years, predominantly due to increasing market saturation. However, the growth in the number and dollar volume of credit card transactions should continue to outpace the growth of the credit card base as a result of expanding acceptance of credit cards by non-traditional merchants and increasing card usage by cardholders. We expect continued rapid U.S. debit card growth as more financial institutions issue debit cards, convert ATM cards to debit cards, and migrate to on-line, real-time processing.

In light of the above dynamics, we believe competitors in the U.S. card processing market are reformulating their strategies to focus on the following: protecting their current customer bases; taking away market share from the competition through increased emphasis on sales and marketing; developing more cost-effective technologies in order to compete more effectively on price; and adapting their systems and business models to support growing Internet opportunities. We believe that processing businesses that accomplish the above will have the opportunity to achieve strong growth over the next several years.

We anticipate the market for our e-banking services will grow substantially for the foreseeable future, as the number of consumers and businesses using Internet banking will grow substantially. The banking industry continues to utilize third party organizations to provide e-banking technology and service. We estimate that less than 20% of U.S. financial institutions have selected an e-banking service provider and the remainder will need to do so to remain competitive and retain customers. We believe this creates an attractive market opportunity for us.

The International Market. Our revenue from international card processing operations has grown rapidly since we entered the international market in 1998, and now accounts for 23% of Card Services' revenue. This year, we expect to add 4.5 million cards in Australia, New Zealand, Ireland, and the U.K. under a 5-year contract we signed last year with National Australia Bank, one of the largest card issuing financial institutions in Australia.

The international card processing market has grown rapidly over the last several years. We believe that this strong growth will continue well into the future, fueled by the rapid development of credit economies in Asian-Pacific and South American countries, international expansion of U.S. card issuers, and outsourcing trends among international financial institutions. Industry sources indicate that the aggregate dollar volume of international credit card transactions is expected to grow at a compound annual growth rate of 15% to 25% through 2005, although this projected rate of growth may not be representative of our growth or the actual future growth rate of the international card processing market.

Check Services

The U.S. Market. While the total volume of checks written at the point-of-sale in the U.S. remains substantial--over 29 billion checks in 1999, according to our estimates--we believe the total volume of checks presented to retailers has begun to decline. This is due in part to the growing use of debit and private label cards, as well as the recent weak performance of many major national retailers. At the same time, however, demand for cost-effective, accurate, and innovative check risk management products is strong, due to the following factors:

- escalating check write-offs incurred by retailers due to increasing sophistication of check fraud;
- . increasing concentration of bad checks written at the point-of-sale due to the trend that higher credit quality consumers are paying more with credit and debit cards and writing fewer checks;
- retailers' growing reliance on outside vendors to provide efficient point-of-sale check risk management solutions that not only minimize check losses, but also maximize sales through accurate identification of good check-writing customers; and
- . the emergence of new market segments demanding check risk management services such as the grocery, gaming, and check-cashing industries, the government and Internet.

We believe this ongoing strong demand creates significant growth opportunities for our check risk management business in the U.S., which has benefited from our development and launch of a series of enhanced risk modeling capabilities and check risk management products, and our increased focus on sales and marketing to regional and local retailers to supplement our traditional leadership in the national retailer market.

The International Market. Since our acquisition of Transax in Europe, Check Services' international operations have focused on the geographies served by Transax, including the U.K., Ireland, France, Australia, and New Zealand, and the traditional Canadian market. In terms of aggregate market share, we believe we are the leading provider of check risk management services in those combined geographies. While check-writing may have begun to decline as a total percentage of point-of-sale payments, the low penetration of check risk management solutions in international markets creates a substantial growth opportunity for our Check Services business internationally. Some of that opportunity relates to:

- . strong demand in the U.K. and France for flexible check risk management services as large mass retailers and wholesalers, which have traditionally not acquired check risk management services from third party service providers, look to outsourced solutions to reduce expenses and improve customer service;
- . low penetration of check risk management services in the overall retail market in the U.K. and France; i.e., less than 20% of retail merchants in the U.K. and less than 5% in France; and
- . a large potential market for check risk management services in South America.

Our Strategy

Our strategic objective is to strengthen our position as a leading global provider of payment processing and check risk management services. We intend to concentrate on the following strategies to accomplish our objective.

Card Services

Leverage our competitive strengths in the U.S. to further increase our share of revenue in the U.S. card and merchant processing markets, and in e-banking. Those strengths include:

- our long-term contractual alliances with CSCU and ICBA, through which we maintain proven distribution channels and enjoy strong name recognition and quality-of-service ratings;
- . our "full service" processing capabilities, which enable us to provide among the most comprehensive card processing solutions available; and
- . our highly competitive prices.

Grow our customer base and processing volumes substantially outside the U.S. In international markets, we will continue to focus our marketing efforts on leading card processing prospects, develop flexible processing services tailored to the diverse credit cultures in Europe, Latin America and Asia-Pacific, and leverage our competitive advantages. These advantages include our strength in providing full service processing services, our extensive experience in managing international operations, and our proprietary card processing systems. Our proprietary systems are highly scalable and portable, and have been customized to process in numerous country-specific environments in over 27 countries around the world. This customization enables us to enter new geographic markets quickly and less expensively, and positions us to be a preferred vendor for outsourced card processing as this concept starts to take effect outside the U.S.

Increase our revenues from new and existing products and services. We intend to aggressively market our expanded debit card processing products and services and capture a larger share of the rapidly growing debit card markets in the U.S. and abroad. We intend to aggressively market our card marketing services that

assist our customers in growing their cardholder portfolios and e-banking customer bases. We will develop and market new Internet service capabilities that will allow cardholders to manage their credit card accounts and conduct electronic commerce more efficiently and effectively.

Check Services

Leverage our competitive strengths to increase our market share in our traditional markets, both in the U.S. and internationally. Those strengths include what we believe are the industry's most advanced check risk management algorithms and systems, our proven ability to introduce successful new check risk management products, our position as one of the world's leading transaction risk management services providers, and our company's existing operations and customer relationships in Europe, Latin America, and Asia-Pacific

Continue our development and utilization of increasingly sophisticated risk modeling tools to differentiate our capabilities from the competition. These tools include proprietary algorithms and systems that we have developed independently, and others that we have developed with our alliance partners.

Enter new markets such as check cashing, gaming, grocery, government, and Internet commerce by combining our current risk management and identity authentication services. This combined solution provides us with the ability to effectively manage risk in environments where the consumer is not present as well as at the traditional point-of-sale.

Further, for both Card Services and Check Services, we intend to continue to aggressively pursue strategic alliances with, investments in, and acquisitions of, domestic and international companies that would enable us to increase our penetration in our current markets, enter new markets, expand our technology expertise to help us further enhance our processing, risk management and e-banking services, or to increase operating efficiencies.

Products and Services

Card Services

Card Issuer Services. We provide card issuer services to financial institutions that issue VISA and MasterCard credit and debit cards, and to retailers and other issuers of private label payment cards, for use by both consumers and business accounts. Our debit card services support both off-line debit cards, which are processed similarly to credit cards, and on-line debit cards, through which cardholders obtain immediate access to funds in their bank account through ATMs, or merchant point-of-sale terminals. The majority of the card issuer programs that we provide are full service, including essentially all of the operations and support necessary for an issuer to operate a credit and debit card program. More specifically, we process all the debit and credit card transactions on the credit and debit cards issued by our customers, including electronically authorizing the transactions, capturing the transaction data and settling the transactions, and we provide full service back-office support functions for their programs. These support functions include: embossing and mailing their credit and debit cards to their customers; customer service on behalf of the card issuer to their customers; card portfolio management and analysis; invoicing their cardholders; receiving and processing cardholder payments; and pursuing delinquent or fraudulent accounts. We do not make credit decisions for our customers, nor do we fund their card receivables. Our services are menu driven, and offer flexibility for those of our customers that require less than our full service programs. Such customers include large card issuing banks that contract with us to provide transaction processing, but who choose to invest the capital and human resources necessary to provide their own back-office program support.

In the U.S., we have been highly successful in marketing our card issuer services to credit unions and independent community banks. These two customer segments consist predominantly of small and mid-sized

card issuers that cannot independently achieve the economies of scale that would justify setting up their own credit and debit card operations. We provide our card issuer services to these customers primarily through the contracts we have with CSCU and ICBA. We have a standard product offering in place with each of these organizations, which offer these products to their respective members with our company as the service provider. These alliances allow us to leverage the marketing channels of those two organizations, and eliminate the need for us to negotiate price, terms, and service offerings with individual credit unions or community banks. As a result, we believe we are the leading provider, in terms of market share, of comprehensive card processing services to U.S. credit unions and to independent community bank card issuers in the U.S.

We provide card issuer services internationally through our operations in Brazil, Chile, and the U.K. We expect to commence providing card issuer processing services later this year in Australia, New Zealand, and Ireland, pursuant to the major contract we signed with the National Australia Bank in 2000.

Merchant Processing Services. We provide merchant processing services that enable retailers and other merchants to accept electronic payment cards in payment for goods and services. We provide our merchant processing services both directly to merchants who accept credit and debit cards, and through contracts with financial institutions and others where our solutions enable them to service the card processing needs of their merchant customers. Our services comprise card transaction authorization, accounting and data capture, transaction settlement, and dispute resolution.

E-Banking Services. Our principal e-banking products include Internet banking solutions for financial institutions that offer retail Internet banking services to consumers and corporate Internet banking services to businesses. We provide these services either by licensing our products to our customers for their operation in-house, or as an application services provider, or ASP, where the customers are linked to our central service bureau.

Our retail Internet banking services enable our bank customers to offer a wide array of PC-based banking services to consumers, such as on-line account information access and electronic bill payment. Our corporate Internet banking services enable our bank customers to offer the business community various electronic commercial banking services, including transmission of account and other business information between the bank and the business customer, bill payment, funds transfers, loan and account applications, and other electronic services.

We also provide ACH origination products, which enable financial institution customers to comply with various National Automated Clearing House Association rules and regulations, and to manage risk associated with electronic funds transfer. We also provide telephone banking capabilities to our customers.

Check Services

Check Services principally provides check risk management and related processing services to businesses accepting or cashing checks at the point-of-sale. We serve national and regional merchants, including national retail chains such as Sears, Best Buy, Walgreen's, Federated, and Target, hotels, automotive dealers, telecommunications companies, supermarkets, casinos, mail order houses, and other businesses. Our services allow our clients to run their customers' personal and business checks through an authorization process that reveals whether a customer has previously written bad checks and the likelihood that a check will clear.

Our check risk management services include diverse solutions tailored to the specific needs of the customer. They include Welcome Check(R) guarantee services, where we accept the bad check risk associated with checks authorized by our system, and Welcome Check(R) verification services, where our customers retain the risk. We also provide blends of guarantee and verification services to meet specific customer needs. All of our products leverage our proprietary system, PathWays(TM). PathWays provides the flexibility, utilizing our risk management data and proprietary models, to manage check acceptance risk by controlling the risk management parameters on a store by store, or even a cash register by cash register, basis.

In recent years, we believe we have led our industry in the introduction of successful, innovative new solutions for existing and new markets. In addition to PathWays referred to above, we have introduced: 4/Detection(TM), a software product that helps employers combat employee fraud; PayCheck Accept(TM), which enables supermarkets and gaming establishments to reduce the risk of check losses and fraud in connection with their payroll check cashing services; third party check collections for retailers utilizing our verification services; and electronic check risk management solutions enabled for electronic commerce, which enable retailers to safely and securely accept payments over the

We provide our check risk management products and services internationally in Canada, the U.K., Ireland, France, Australia, and New Zealand. Our principal product in all those countries is check guarantee, although mass retailers are beginning to utilize our check verification and collection services. We have also introduced new innovative products and services in the international markets that we serve. These include accounts receivable warranty, which enables a retailer to extend short term credit to its small business customers, and deferred debit solutions, which allow retailers to accept consumer payments electronically.

Sales and Marketing

We market our products and services through a direct sales force and indirect sales channels, such as CSCU and ICBA, independent sales organizations, marketing alliances, and financial institutions. Additionally, we market directly to customers through print advertising and direct mail efforts. We participate in major industry tradeshow and publicity events and actively employ various public relations campaigns. Our strategy is to use the most efficient delivery system available to successfully acquire customers and build awareness of our products and services.

We organize our direct sales force by customer market segment or distribution channel, such as financial institutions, retailers, e-commerce providers, and network and gateway providers. Our direct sales force is capable of selling all or a portion of our products and services to offer the most effective solution for each customer's particular needs. We work collaboratively with our customers and prospective customers to help them identify key needs and create new solutions using our products and services. Because many of our customers use a single product or service, or a combination of products or services, our direct sales force has begun to target existing customers to leverage cross-selling opportunities. Our international sales force is focused on the sale of our card issuer services, merchant processing services, check risk management services and our proprietary card software solutions.

Competition

The markets for card transaction processing, check risk management services, and e-banking services are highly competitive, and we expect that competition will intensify in the future as the e-commerce and Internet markets continue to develop and expand. Our principal competitors include:

- . third-party credit and debit card processors, including First Data, Total System Services, EDS, and Payment Systems for Credit Unions, Inc.;
- third-party software providers, such as PaySys, recently acquired by First Data, who license their card processing systems to financial institutions and third party processors; and
- check authorization, guaranty and risk management service providers, including First Data's TeleCheck Services division, eFunds, and International Check Services.

Some of our competitors are privately-held, and the majority of those that are publicly-held do not release the information necessary to precisely quantify our relative competitive position, which varies depending on the segment of our markets. Based on information appearing in a widely-cited industry publication, The Nilson Report, however, we believe that we are among the largest third party payment transaction processors in the world based on annual revenues. According to that report, First Data is the largest payment systems industry vendor based on annual revenues.

In general, we believe that our ability to compete successfully depends upon a number of factors, including:

- the reliability, security, speed, and capacity of our systems and technical infrastructure;
- the comprehensiveness, scalability, ease of use, and service levels of our products and services;
- our strong relationships with CSCU and ICBA, and the related scale advantages achieved through them;
- our ability to interface with vendors of data processing software and services;
- . our pricing policies and the pricing policies of our competitors and suppliers;
- . our risk assessment and fraud detection expertise;
- the timing of introductions of new products and services by us and our competitors; and
- . our ability to support unique customer requirements.

Research and Development

Our research and development activities have related primarily to the design and development of our payment processing systems and related software applications and risk management platforms. We expect to continue our practice of investing substantial resources to extend the functionality of our proprietary processing systems, and to develop new and innovative solutions in response to the needs of our customers. In addition, we intend to offer products and services that are compatible with new and emerging delivery channels such as the Internet.

Additional Financial Information About Segments, Geographic Areas, and Products and Services

Note 11, entitled "Segment Information," in the "Combined Financial Statements of the Equifax Payment Services division," included elsewhere in this information statement, contains additional information about our business segments, operations within and outside of the U.S., and products and services. This information is incorporated by reference into this section of this information statement.

Intellectual Property Rights

We rely on a combination of contractual restrictions and trademark, copyright, and trade secret law to establish and protect our trademarks, software, and know-how. These legal protections afford only limited protection of our proprietary rights. It may be possible for a third party to copy our products and services or otherwise obtain and use our proprietary information without our permission. There is no assurance that our competitors will not independently develop services and products that are substantially equivalent or superior to ours. In addition, the laws of some foreign countries do not protect our proprietary rights to the same extent as do the laws of the U.S.

Government Regulation

Various aspects of our businesses are subject to federal, state, and foreign regulation. Our failure to comply with any applicable laws and regulations could result in restrictions on our ability to provide our products and services, as well as the imposition of civil fines and criminal penalties.

As a provider of electronic data processing and back-office services to financial institutions, we are subject to regulatory oversight and examination by the Federal Financial Institutions Examination Council, an interagency body comprised of the various federal bank and thrift regulators and the National Credit Union Association. In addition, we may be subject to possible review by state agencies that regulate banks in each state where we conduct our electronic processing activities.

Because we maintain a data base in the U.S. containing the check-writing histories of consumers, and use that information to provide our check risk management services, our check risk management business is subject to the federal Fair Credit Reporting Act and various similar state laws. Among other things, the Fair Credit Reporting Act imposes requirements on us concerning data accuracy, and provides that consumers have the right to know the contents of their check-writing histories, to dispute their accuracy, and to require verification or removal of disputed information. In furtherance of our objectives of data accuracy, fair treatment of consumers, and protection of consumers' personal information, in addition to best ensuring we comply with these laws, we maintain a high level of security for our computer systems in which consumer data resides, and we maintain consumer relations call centers to facilitate efficient handling of consumer requests for information and handling of disputes.

Our check collection services are subject to the Fair Debt Collection Practices Act and various state collection laws and licensing requirements. The Federal Trade Commission, as well as state attorneys general and other agencies, have enforcement responsibility over the collection laws, as well as the various credit reporting laws.

Because we do business in international markets as well as in the U.S., we are subject to laws and regulations in jurisdictions outside the U.S. that regulate many of the same activities that are described above, including electronic data processing and back-office services for financial institutions and use of consumer information.

Congress and various state legislatures and regulators, as well as legislative bodies and regulators outside the U.S., are increasingly focused on the passage of laws and regulations that restrict businesses' purchase and sale, use, or sharing, of nonpublic personal information about consumers. For example, in 1999, Congress passed The Gramm-Leach-Bliley Act, which, among other things, imposes significant obligations on financial institutions designed to ensure that nonpublic personal information pertaining to their account holders is protected from unauthorized disclosure to third parties. While Gramm-Leach-Bliley imposes significant restrictions and requirements on our financial institution customer base, and has an indirect impact on our business in that we process and store account holder data on behalf of those customers, we believe neither that law nor any other recently enacted privacy laws or regulations have a significant negative impact on our businesses. However, U.S. and foreign legislators and regulators continue to consider new laws and regulations in the consumer data protection area, including, among other things, laws and regulations that would further restrict the purchase and sale, use, or sharing of consumer information by financial institutions and other businesses, restrict the use of social security numbers for commercial purposes, and require disclosure by businesses of scoring algorithms used in making credit or similar decisions regarding consumers.

As a provider of Internet based products and services, we may also become subject to laws and regulations that may be adopted in the future with respect to the Internet or e-commerce covering areas such as data protection and other consumer protections, pricing, content, copyrights, taxation, and characteristics and quality of products and services. Moreover, the applicability to the Internet of existing laws governing data protection, property ownership, taxation, defamation, and other areas is uncertain and may remain uncertain for a considerable length of time. Both new regulation, and the applicability and uncertainty of existing regulation, may impede the Internet's growth and slow the demand for our Internet based products and services or increase our cost of providing them.

Although we do not believe that compliance with future laws and regulations related to our businesses, including future consumer protection laws and regulations, will have a material adverse effect on our company, enactment of new laws and regulations may increasingly affect the operations of our business, which could result in substantial regulatory compliance costs, litigation expense, adverse publicity, or loss of revenue.

Employees

We have approximately 5,700 employees, most of whom are full time, including approximately 2,700 in the U.S., 2,100 in Brazil, and 600 in the U.K. Of our total number of employees, approximately 5,100 persons

are engaged in operations, 300 in sales and marketing, and 300 in administrative or other functions. None of our U.S. work force is currently unionized. We have not experienced any work stoppages and consider our relations with our employees to be good.

Properties

Our corporate headquarters will be located in metropolitan Atlanta, Georgia pursuant to a lease that we expect to enter into with an unaffiliated third party prior to the distribution. Our principal operations center and administration, sales, marketing, and development facilities are located in St. Petersburg, Florida, in a 305,000 square foot building. We have other smaller leased operations facilities in Wisconsin, Utah, and Georgia. In support of our international operations, we have leased operations centers in Brazil, Chile, the U.K., France, and Australia. We also have a number of small sales or support offices in the other places where we do business.

We own or lease a variety of computers and other related equipment for our operational needs. We continue to upgrade and expand our computers and related equipment in order to increase efficiency, enhance reliability, and provide the necessary base for business expansion.

We believe that all of our facilities and equipment are in good condition and are well maintained and adequate for our business as presently conducted.

Legal Proceedings

We are party to a number of routine claims and lawsuits incidental to our business. In addition, a class action lawsuit is pending against Equifax Check Services, Inc., a subsidiary of Equifax that will be contributed to us immediately prior to the distribution, in the U.S. District Court for the Eastern District of California. This action, instituted in August 1996 by plaintiffs Gary and Nancy Ballard, is based on a claim that our practice of assessing a service charge on unpaid checks allegedly violated provisions of the federal Fair Debt Collection Practices Act and California's Unfair Business Practices Act during the period from August 1992 through December 1996. The plaintiffs seek, among other remedies, a refund of all service charges collected from California consumers during this period, prejudgment interest, statutory damages under the Fair Debt Collection Practices Act, and attorneys' fees, which amounts in the aggregate, could exceed \$15 million. Equifax has defended, and will continue to defend this action vigorously, however, litigation is inherently uncertain and we may not prevail.

Other than as described above, we believe there is no litigation pending against us that could, individually or in the aggregate, have a material adverse effect on our business.

RELATIONSHIP BETWEEN EQUIFAX AND OUR COMPANY AFTER THE DISTRIBUTION

In connection with the distribution, we will enter into a distribution agreement and a number of ancillary agreements with Equifax for the purpose of accomplishing the contribution to us of the businesses described in this information statement and the distribution. These agreements, which will govern the relationship between Equifax and us after the distribution and provide, among other things, for the allocation of employee benefits, tax and other liabilities and obligations attributable to periods prior to the distribution, will have been prepared before the distribution, and will reflect agreements between affiliated parties established without arms-length negotiation. We anticipate, however, that the terms of these agreements will equitably reflect the benefits and costs of our ongoing relationship with Equifax. The ancillary agreements include:

- . Employee Benefits Agreement;
- . Transition Support Agreement;
- . Intercompany Data Purchase Agreement;
- . Tax Sharing and Indemnification Agreement;
- . Intellectual Property Agreement; and
- . Agreement Regarding Leases.

Of the agreements summarized below, the material agreements have been filed as exhibits to the registration statement of which this information statement forms a part and the summaries of the material agreements are qualified in their entireties by reference to the full text of the agreements.

Distribution Agreement

The distribution agreement will set forth the agreements between Equifax and us with respect to the principal corporate transactions required to effect the contribution and the distribution, and various other agreements governing the continuing relationship between us and Equifax after the distribution.

The Contribution; Allocation of Assets and Liabilities; No Representations and Warranties

In connection with the distribution, Equifax will contribute to us all of the payment services businesses described in this information statement. It will effect this contribution by transferring, or causing its subsidiaries to transfer, all of the issued and outstanding capital stock of its direct or indirect subsidiaries conducting our business, and all of its other material assets used in the conduct of the payment services businesses described in this information statement.

The distribution agreement will also provide generally that all material assets and liabilities of the contributed businesses conducted by Equifax prior to the distribution will be vested solely in us or our subsidiaries as of the effective time of our separation from Equifax. Equifax generally will have no interest in our assets and businesses and no obligation with respect to our liabilities as of effective time of our separation from Equifax. Similarly, we generally will have no interest in the assets of Equifax's other businesses and no obligation with respect to the liabilities of Equifax's retained businesses as of the effective time of our separation from Equifax. Except as otherwise agreed, to the extent Equifax or any of its subsidiaries remains liable on any of our liabilities after the effective time of our separation from Equifax, we have agreed to use our reasonable efforts to obtain the release of Equifax or its subsidiary, and have further agreed not to extend or modify any of those liabilities without obtaining that release.

Except as expressly set forth in the distribution agreement or in any ancillary agreement, neither we nor Equifax will make any representation or warranty as to the assets, businesses, or liabilities transferred or assumed as part of the contribution, as to any consents or approvals required in connection with the transfers, as to the value or freedom from any security interests of any of the assets transferred, as to the absence of any defenses or freedom from counterclaim with respect to any claim of either us or Equifax, or as to the legal sufficiency of any assignment, document, or instrument delivered to convey title to any asset transferred. Except as expressly set forth in any ancillary agreement, all assets will be transferred on an "as is," "where is" basis, and the respective transferees will agree to bear the economic and legal risks that any conveyance is

insufficient to vest in the transferee good and marketable title, free and clear of any security interest and that any necessary consents or approvals are not obtained or that requirements of laws or judgments are not complied with.

The Distribution

The distribution agreement will provide that, subject to the terms and conditions contained in the agreement, we and Equifax will take all reasonable steps necessary and appropriate to cause all conditions to the distribution to be satisfied. Equifax's agreement to consummate the distribution will be subject to the satisfaction or waiver by the Equifax board, in its sole discretion, of the following conditions:

- . Equifax's board or its duly appointed committee has established the record date and the distribution date and any appropriate procedures in connection with the distribution;
- . all necessary regulatory approvals have been received;
- . this information statement has been mailed to the Equifax shareholders;
- the registration statement on Form 10, of which this information statement is a part, has been declared effective under the federal Exchange Act;
- our board of directors named in this information statement has been elected and our amended and restated articles of incorporation and bylaws have been adopted and are in effect;
- . our common stock has been approved for listing on the New York Stock Exchange, subject to official notice of issuance;
- . a private letter ruling from the IRS, which Equifax has obtained, shall continue in effect, and provide that, among other things, the contributions constitute a reorganization pursuant to Section 368(a)(1)(D) of the Internal Revenue Code and that distribution will qualify as a tax-free distribution for federal income tax purposes under Section 355 of the Internal Revenue Code;
- . we have entered into an agreement establishing our new credit facility;
- . we have performed our obligations under the distribution agreement and the ancillary agreements;
- we and Equifax have taken all action necessary or appropriate under the securities or blue sky laws of states or other political subdivisions of the U.S. in connection with the distribution; and
- . no order, injunction, or decree by any court of competent jurisdiction, or other legal restraint or prohibition, preventing consummation of the distribution will have been issued or be in effect.

Following the satisfaction or waiver of the conditions enumerated above, the distribution agreement will provide that Equifax will deliver to the distribution agent a certificate or certificates representing all of the outstanding shares of our common stock. Equifax will instruct the distribution agent to distribute those shares on July 7, 2001 or as soon thereafter as practicable in a proportion equal to one of our shares of common stock for every two shares of Equifax common stock outstanding as of 5 p.m., Atlanta, Georgia, time on June 27, 2001. The distribution agreement will provide that Equifax may terminate the agreement at any time prior to the distribution.

Indemnification

We will agree to indemnify, hold harmless, and defend Equifax, each of its affiliates and each of their respective directors, officers, employees and agents, from and against any and all damage, loss, liability, and expense they may incur or suffer, arising out of or due to:

- . the failure of us or any of our affiliates or any other person to pay, perform, or otherwise discharge any liabilities of the contributed businesses, including, without limitation:
 - . our liabilities under the distribution agreement or any ancillary agreement;

- liabilities incurred in connection with the conduct or operation of the contributed businesses, including any acquired businesses, or our ownership or use of our assets, whether arising before, at, or after the effective time of our separation from Equifax;
- liabilities set forth on the balance sheet of the payment services group of Equifax as of the effective time of our separation from Equifax;
- . liabilities of Equifax or our company relating to any business formerly owned or operated by us or the contributed businesses or arising out of the sale of any of those businesses; or
- . liabilities relating to the acquisition by any of the contributed businesses of any business prior to the effective time of our separation from Equifax, except to the extent such liabilities arise out of the issuance of securities of Equifax in any such acquisition; and
- . any untrue statement or alleged untrue statement of a material fact or omission or alleged omission to state a material fact required to be stated in the registration statement, the preliminary or final information statement, or any amendment or supplement thereto or the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements in the registration statement or this information statement not misleading, except to the extent such liabilities arise out of or are based upon information included in certain sections of the registration statement or this information statement about Equifax.

Equifax will agree to indemnify, hold harmless, and defend us, each of our affiliates and each of our and their respective directors, officers, and employees, from and against any and all damage, loss, liability, and expense we or they may incur or suffer, arising out of or due to:

- . the failure of Equifax or any affiliate of Equifax or any other person to pay, perform, or otherwise discharge any liabilities of Equifax or its affiliates other than liabilities associated with the contributed businesses that we have assumed and liabilities under the distribution agreement or any ancillary agreement; and
- . any untrue statement or alleged untrue statement of a material fact or omission or alleged omission to state a material fact required to be stated in the registration statement, this information statement, or any amendment or supplement thereto or the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements in the registration statement or this information statement not misleading, except to the extent such liabilities arise out of or are based upon information included in any section of the registration statement or this information statement about us.

In circumstances in which the indemnity is unavailable or insufficient, for any reason, to hold harmless a party entitled to indemnification in respect of any indemnifiable losses in connection with any statements or omissions or alleged statements or omissions in the registration statement, this information statement, or any amendment or supplement thereto, each indemnifying party, in order to provide for just and equitable contribution, will contribute to the amount paid or payable by such indemnified party as a result of these indemnifiable losses, in a proportion appropriate to reflect the relative fault of the indemnifying party or parties on the one hand and the indemnified party on the other in connection with the statements or omissions or alleged statements or omissions that resulted in these indemnifiable losses, as well as any other relevant equitable considerations.

The distribution agreement also will specify procedures with respect to claims subject to indemnification and related matters.

Insurance

The distribution agreement will obligate us to use our best efforts to procure and maintain for at least five years directors' and officers' liability insurance coverage at least equal to the amount of Equifax's current

directors' and officers' insurance coverage with respect to directors and officers of Equifax who are or will become our directors and officers as of the date of the distribution for acts as our directors and officers for periods from and after the date of the distribution. The distribution agreement will obligate Equifax to use its best efforts to maintain directors' and officers' liability insurance coverage at least equal to the amount of Equifax's current directors' and officers' liability insurance coverage for at least five years with respect to the directors and officers of Equifax who are or will become directors and officers of us as of the date of the distribution for acts as directors and officers of Equifax or one of its affiliates during periods prior to the date of the distribution.

Dispute Resolution

The distribution agreement will contain provisions that govern, except as otherwise provided in any ancillary agreement, the resolution of disputes, controversies, or claims that may arise between us and Equifax relating to the distribution agreement or any ancillary agreement. These provisions contemplate that efforts will be made to resolve disputes, controversies, and claims by escalation of the matter to senior management and other representatives of us and Equifax. If those efforts are not successful, either we or Equifax may submit the dispute, controversy, or claim to binding arbitration in accordance with the Rules of Arbitration of the International Chamber of Commerce, subject to the provisions of the agreement.

Expenses

The distribution agreement will provide that, except as otherwise provided in the agreement or any ancillary agreement, Equifax will pay all costs and expenses incurred in connection with the consummation of the distribution and the transactions contemplated by the agreement and all costs and expenses incurred in connection with the preparation, execution, delivery, and implementation of the distribution agreement and the ancillary agreements. None of the expenses that Equifax will pay are direct or indirect expenses of Equifax's shareholders relating to the transactions. Equifax will also pay its other expenses of the transaction, including the legal, filing, accounting, printing, and other expenses incurred in connection with the preparation, printing, and filing of the registration statement on Form 10 of which this information statement is a part and this information statement.

Amendments and Waivers; Further Assurances

The distribution agreement will provide that it may only be amended or rights under it waived by an instrument signed by the party to be charged with the amendment or waiver. We and Equifax will agree to use our respective reasonable efforts to:

- . execute and deliver any additional instruments and documents and take any other actions the other party may reasonably request to effectuate the purposes of the distribution agreement and its terms; and
- . to take all actions and do all things reasonably necessary under applicable laws and agreements or otherwise to consummate and make effective the transactions contemplated by the distribution agreement.

Employee Benefits Agreement

We will enter into an employee benefits agreement with Equifax that will provide for the transition of employee plans and programs, including stock option and long-term incentive compensation programs, sponsored by Equifax for its employees who will become our employees. This agreement will allocate responsibility for certain employee benefits matters and liabilities after the distribution. Under the employee benefits agreement we will become liable for providing specified welfare and retirement benefits to our employees after the distribution, which will generally be similar to the benefits Equifax provided to such employees. These benefits include medical, dental, and vision benefits, flexible spending accounts covering health care and dependent care expenses, life and accident insurance plans, short and long term disability plans, a severance plan, a 401(k) plan, a defined benefit pension plan, policies covering vacations, holidays, and sick

leave, an annual incentive plan, a long-term incentive plan, a split-dollar life insurance plan, a stock incentive plan, a deferred compensation plan, and a director stock incentive plan. As of our separation from Equifax, our active employees will generally cease current participation in Equifax employee benefit plans and begin participation in our plans, and we will generally recognize, among other things, our employees' past service with Equifax for purposes of our employee benefit plans. Equifax will retain the responsibility for providing welfare benefits with the exception of severance pay, to our former employees, determined as of the distribution date. For those former employees who are also receiving certain retirement benefits as of the distribution date, Equifax will retain the responsibility of providing pension and 401(k) benefits. Except as specifically provided in the employee benefits agreement, nothing in that agreement will restrict our or Equifax's ability to amend or terminate any of our or Equifax's respective employee benefit plans.

Pension Plan

In connection with our separation from Equifax, we will establish our own qualified defined benefit pension plan, which will be similar to Equifax's defined benefit pension plan as in effect at such time. Generally, our pension plan will cover all current employees of the payment services businesses and all former employees of our businesses who had a vested accrued benefit in the Equifax pension plan immediately prior to the distribution. However, former payment services employees who have terminated employment by the distribution date and are currently receiving benefits from the Equifax pension plan or are currently receiving retiree health benefits will not become participants in our plan. Participants will be given full credit under our pension plan for service and compensation accrued under the Equifax pension plan, subject to the transition provisions described below.

It is contemplated that the Equifax pension plan will transfer assets and liabilities to our plan for benefits earned through the distribution date. The transfer is scheduled to occur immediately after the distribution date, provided that the assets of the Equifax pension plan at the time are at least 106% of total plan liabilities. Pursuant to the transfer, assets will be allocated between the trust for the Equifax pension plan and the trust for our pension plan based on a methodology set forth in the employee benefits agreement. This methodology has been designed to put our plan on a sound financial footing by at least fully funding the benefits to be provided by our plan, while at the same time not creating financial risk to the Equifax pension plan due to the decrease in the assets remaining with that plan. Any excess assets will be shared between the Equifax plan and our plan in accordance with the employee benefits agreement. If the assets of the Equifax pension plan as of the scheduled transfer date are not at least 106% of total liabilities, either we or Equifax may elect not to proceed with the transfer as of the scheduled date. In that case, there will be a transition period after the distribution date during which our employees will be participants in both the Equifax pension plan and our pension plan. During this period, our plan will be structured to provide benefits with an offset for benefits paid from the Equifax pension plan. If the transfer of assets and liabilities does not occur on the scheduled date, we and Equifax intend to arrange for a delayed transfer date, which will occur not later than December 31, 2001, provided that the Equifax pension plan then meets a minimum funding level to be newly specified by the parties. If a later transfer does not occur, then we have agreed to negotiate with Equifax the future structure of the two pension plans.

401(k) Plan

Not later than the distribution date, we will establish our own qualified defined contribution plan under Section 401(k) of the Internal Revenue Code, which will be similar to Equifax's 401(k) Retirement and Savings Plan as in effect at such time. Generally, our 401(k) plan will cover all current employees of the payment services businesses and all former employees who have a vested account balance in the Equifax 401(k) plan immediately prior to the distribution. However, former payment services employees who have terminated employment by the distribution date and are currently receiving benefits from the Equifax pension plan by the distribution date or are currently receiving retiree health benefits will remain in the Equifax 401(k) plan and will not be transferred to our plan. Except for these former employees, the Equifax 401(k) plan will transfer to our 401(k) plan the account balances of each of the payment services participants, and each will be credited under our 401(k) plan with the account balance credited to him or her as of the transfer date under the terms of the Equifax 401(k) plan. During a transition period, an Equifax stock account and a Certegy stock

account will be maintained under the 401(k) plans of both companies to hold the shares of Equifax common stock and our common stock distributed with respect to such Equifax common stock. Participants will not be allowed to add to their funds in the stock accounts of the other company through new contributions or balance transfers.

Health and Welfare Plans

We will assume all liabilities and responsibilities for providing health and welfare benefits to our active employees. As of the date of the distribution, we intend to establish health and welfare plans that are substantially similar to the Equifax plans. During a transition period after the distribution, we will administer some of our plans in conjunction with the respective Equifax plans and provide reimbursement to Equifax for any costs or expenses it incurs in connection with such administration. For those benefits that are provided through insurance, Equifax will take steps to have each insurance carrier agree to allow our employees to continue to be covered by Equifax's policies or through separate contracts on substantially the same basis during the transition period. For those benefits that are funded through the Equifax Voluntary Employees Beneficiary Association trust, we will establish our own Voluntary Employees Beneficiary Association trust and Equifax will transfer assets from its trust to our trust in an amount at least equal to the incurred but unpaid claims for which one trust is liable. If the Equifax trust contains assets in excess of the amount necessary both to make this transfer and to establish a reserve for the incurred but unpaid claims for which the Equifax trust is liable, Equifax will transfer to our trust a portion of this excess, determined in its discretion. For these purposes, claims will be estimated based upon a methodology that is consistent with the prior calculation of trust liabilities by Equifax. We expect the transition period to conclude on or before December 31, 2002. In addition, if any of our employees are eligible to retire from Equifax on the distribution date, or if they become eligible to retire from us by December 31, 2002, applying the requirements in the pension plan for retirement in effect on the distribution date, these employees will be eligible for retiree health benefits from Equifax.

Stock and Incentive Compensation Plans

Stock Options and Restricted Stock. Prior to the date of the distribution, we intend to establish a stock incentive plan that is substantially similar to the Equifax stock incentive plan. With respect to outstanding options, the employee benefits agreement provides that Equifax stock options held by our employees will generally be converted to our stock options at the time of the distribution in accordance with a conversion ratio, through options granted under our stock incentive plan described below. As part of the conversion, we will multiply the number of shares purchasable under each converted stock option by a conversion ratio determined at the time of the distribution and divide the exercise price per share of each option by the same conversion ratio. Fractional shares will be rounded down to the nearest whole number of shares. All other terms of the converted stock options will generally remain the same as those in effect immediately prior to the distribution. With respect to options held by our employees in some foreign countries, if the above conversion method is not permitted or desirable under the foreign tax, securities or other laws, a different approach may be used. Our employees who hold Equifax restricted stock will receive the dividend of our shares and such shares will be subject to the same restrictions as the existing Equifax restricted stock. Also, each of our employees holding Equifax restricted stock will have the right to elect to convert his or her Equifax restricted stock into additional shares of our common stock subject to the same restrictions by multiplying the number of shares of Equifax restricted stock by the conversion

Long-Term Incentive Plan. Under the Equifax long-term incentive plan, with respect to awards for the long-term incentive measurement period that began January 1, 1999 and will end December 31, 2001, Equifax will be responsible for determining the extent to which the established performance criteria have been met and for making any required payments under the plan at the time required under the plan to all participants including our employees, counting employment with us after the distribution as employment with Equifax for purposes of the plan. Prior to the distribution, we intend to establish a new long-term incentive program which will be similar to the current Equifax plan. Our executives' awards issued under the Equifax long-term incentive plan for the long-term incentive measurement period that began January 1, 2000 and was expected to

end December 31, 2002 will be replaced with new awards under our long-term incentive plan with a performance period from the date of the distribution through December 31, 2002. With respect to the awards for the 2001 annual incentive period, such awards, if any, will be payable by us in accordance with the usual terms of the award or will be payable by Equifax based on performance for the portion of the year prior to the distribution and by us for the performance of the rest of the year, all as more fully described in the employee benefits agreement.

Split-Dollar Life Insurance Program. The employee benefits agreement provides that we will continue to maintain for our participating executives the split-dollar life insurance program currently sponsored by Equifax. The split-dollar life insurance policies and related collateral assignments will remain in place and the rights and obligations with respect to our employees will be transferred to us.

Deferred Compensation Plan. The employee benefits agreement provides that we will establish a deferred compensation plan and that the accounts of our executives who are participants under the Equifax deferred compensation plan will be transferred to this new plan.

Transition Support Agreement

We will enter into a transition support agreement with Equifax prior to the distribution under which, in exchange for the fees specified in that agreement, Equifax will agree to continue to provide various administrative and other transition services to us, and we will agree to provide certain transition services to Equifax.

The period during which each party is required to provide services to the other will vary depending upon the particular category of service, but in no event will any service be provided beyond 24 months from the effective time of our separation from Equifax.

The transition support agreement will provide that it may be terminated by mutual consent of the parties, by either party for a material uncured breach, the insolvency of either party, or the acquisition of either party by a competitor of the non-acquired party, but such termination will be applicable only with respect to services provided by the non-acquired party to the portion of the acquired party's business that has been affected by the change in control. The party receiving the services under this agreement will indemnify the party furnishing the services, subject to certain limitations, for losses resulting from the provider's furnishing or failure to furnish the services unless the party receiving the services commits willful misconduct or gross negligence. The party furnishing the services will indemnify the party receiving the services, subject to certain limitations, for losses resulting from its willful misconduct or gross negligence in providing services.

Intercompany Data Purchase Agreement

Under the intercompany data purchase agreement that we will enter into with Equifax, each party will have the right to purchase from the other certain data for use in its operations, on a non-exclusive, arm's-length basis. The term of the agreement will be for two years following the effective time of our separation from Equifax, and may be renegotiated after that period.

Tax Sharing and Indemnification Agreement

We will enter into a tax sharing and indemnification agreement with Equifax that will govern the allocation between the companies of federal, state, local, and foreign tax liabilities and related tax matters, such as the preparation and filing of tax returns and tax contests, for the taxable periods before and after the distribution.

The tax sharing and indemnification agreement will have the following provisions that concern events which might occur after the distribution that could have an adverse affect on the tax treatment of the distribution:

. Each company will be responsible for, and will indemnify the other company from and against, any tax liability resulting from any action by such company that may cause the tax treatment of the

distribution, the preceding contributions of capital and related transactions to be different than as contemplated in the IRS ruling request, including:

- . a determination that the contribution and distribution do not qualify for tax-free treatment under Sections 368(a)(1)(D) and 355 of the Code; or
- . a determination that the distribution is taxable to Equifax pursuant to Sections 355(e) of the Code.
- . Each company will be responsible for, and will indemnify the other company from and against, any tax liability resulting from any breach of a factual statement or representation with respect to the indemnifying company made by Equifax to the IRS in connection with the IRS ruling request.
- . To maintain the qualification of the distribution as tax-free under Sections 368(a)(1)(D) and 355 of the Code, there are material limitations on transactions in which either company may be involved during the two-year period following the distribution date. Specifically, during this two-year period, each company will agree to refrain from engaging in any of the following transactions unless it obtains (i) a private letter ruling from the IRS, (ii) or an opinion from tax counsel reasonably acceptable to the other party providing that the transaction should not affect the tax-free treatment of the distribution and the preceding contributions of capital, or (iii) the consent of the other party to the tax sharing and indemnification agreement:
 - . the company's liquidation or merger with another corporation;
 - the redemption, purchase, or reacquisition of more than a de minimis amount of the company's own capital stock;
 - . the sale, exchange, distribution or other disposition of assets used in the active conduct of the company's businesses relied upon in the IRS ruling request to satisfy Section 355(b) of the Code, unless after giving effect to the sale, exchange, distribution or other disposition of these assets, the company would continue to satisfy Code Section 355(b);
 - . the discontinuance of the active conduct of the company's current trades or businesses being relied upon in the IRS ruling request; or
 - . any other transaction resulting in the direct or indirect acquisition of the indemnifying company's stock representing a 50% or greater interest in that company within the meaning of Section 355(e) of the Code.

If a company enters into any of these transactions, that company will be responsible for, and will indemnify the other company from and against, any tax liability resulting from any such transaction.

The tax sharing and indemnification agreement also will contain the following technical provisions:

- . We will be responsible for the respective federal, state, and foreign income tax liabilities attributable to any of our subsidiaries relating to all taxable periods. Accordingly, we will indemnify Equifax and its subsidiaries against any income tax liabilities attributable to any of our subsidiaries.
- . Similarly, Equifax will be responsible for the respective federal, state, and foreign income tax liabilities attributable to Equifax or its subsidiaries, other than us and our subsidiaries, relating to all taxable periods. Accordingly, Equifax will indemnify us and our subsidiaries against any such tax liabilities attributable to any of Equifax's remaining subsidiaries.
- . Any tax refund or tax benefit received by either company that is on account of or otherwise attributable to the other company will be paid by the receiving company to the other company.

- . Following the distribution, the company to which a tax return relates generally will be responsible for preparing and filing such return, with the other company providing the requisite information, assistance, and cooperation.
- . Each company generally will be responsible for handling, settling, and contesting any tax liability for which it is liable under the terms of the tax sharing and indemnification agreement.

Intellectual Property Agreement

We will enter into an intellectual property agreement with Equifax that will address the allocation between us and Equifax of copyrights, trademarks, software, and other intellectual property. The agreement will provide that Equifax will transfer to us, without charge, title to any of the intellectual property used solely or primarily in our businesses, effective at the effective time of our separation from Equifax. Equifax will retain title to any of the intellectual property it uses solely or primarily in its businesses. As of the effective time, each party will license any intellectual property that it will own after giving effect to this allocation of ownership to the other party, if the other party uses that intellectual property. These licenses will be granted on a royalty free, non-exclusive basis, subject to agreed limitations. Substantially all of these licenses will be perpetual. Each party will agree to cooperate with the other after the effective time to carry out the purposes of the intellectual property agreement and the related licenses granted by each party to the other.

This agreement will also state that Equifax will provide us with certain rights to specified third party software and other intellectual property necessary for the continued operation of our businesses. After the effective time, we will be responsible for on-going maintenance, support, and upgrades associated with the continued use of these third party rights at our expense. Each party will agree to cooperate after the distribution with the other party in the identification, negotiation, assignment, and acquisition of these third party rights as may be reasonably necessary to carry out the purpose of the intellectual property agreement.

Agreement Regarding Leases

Certain of our subsidiaries will enter into an agreement regarding leases with Equifax and certain of its subsidiaries. The agreement regarding leases provides that after the effective time of the separation, we will sublease portions of various office facilities leased by Equifax entities, and the sublandlord and the subtenant will share the use of those office facilities. These subleases will generally provide for the pro rata allocation of rental and other occupancy costs based upon the square footage of the subleased premises and will incorporate and be subject to the provisions of the underlying leases. The agreement also provides that Equifax will assign us its rights under a number of leases for space occupied exclusively by payment services businesses. These assignments will generally require us to assume all duties and obligations of the "tenant" under the assigned leases after the assignment date. Under the agreement regarding leases we will be obligated to indemnify and hold Equifax harmless from any losses, costs, claims, or damages arising out of certain lease guarantees with respect to certain of our leases entered into by Equifax prior to the effective time of our separation from Equifax.

Our Directors and Executive Officers

Following the distribution, our board of directors will be comprised of seven directors, only one of whom will be an executive officer. Thomas F. Chapman and Lee A. Kennedy, who are directors and employees of Equifax, and Kent E. Mast, who is an employee of Equifax, are currently serving as our directors and have been our directors since our inception. In connection with the distribution, Mr. Mast will resign and David K. Hunt, Kathy Brittain White, Richard Child, Charles T. Doyle, and Robert H. Bohannon will be elected to the board prior to the distribution date.

Our board of directors will be divided into three classes. It is expected that Messrs. Chapman and Doyle will be Class I directors, with terms expiring at the annual meeting of shareholders to be held in fiscal year 2002, Messrs. Bohannon, Child and Kennedy will be Class II directors, with terms expiring at the annual meeting of shareholders to be held in fiscal year 2003, and Mr. Hunt and Ms. White will be Class III directors, with terms expiring at the annual meeting of shareholders to be held in fiscal year 2004. Commencing with the annual meeting of shareholders to be held in fiscal 2002, directors for each class will be elected at the annual meeting of shareholders held in the year in which the term for such class expires and thereafter will serve for a term of three years.

The following table sets forth information as to persons who serve or who we expect to serve as our directors and executive officers immediately following the distribution. Our board of directors may appoint additional executive officers from time to time.

Name 	Age	Position(s)
Thomas F. Chapman	57	Chairman of the Board
Lee A. Kennedy	50	President, Chief Executive Officer and Director
Larry J. Towe	54	Executive Vice President and Chief Operating Officer
Michael T. Vollkommer	42	Corporate Vice President and Chief Financial Officer
Bruce S. Richards	46	Corporate Vice President, General Counsel and Secretary
Gerard Ballard	44	Corporate Vice President and Chief Technology Officer
Gerald A. Hines	53	Senior Vice President and Group ExecutiveCard Solutions
Jeffrey S. Carbiener	39	Senior Vice President and Group ExecutiveCheck Solutions
David K. Hunt	55	Director
Kathy Brittain White	51	Director
Richard Child	44	Director
Charles T. Doyle	66	Director
Robert H. Bohannon	56	Director

Thomas F. Chapman is currently a director, and will remain a director and serve as our Chairman of the Board following the distribution. Mr. Chapman has been Chairman of the Board and Chief Executive Officer of Equifax since May 1999, and will continue to serve as Chairman of the Board and Chief Executive Officer of Equifax following the distribution. Mr. Chapman served as President and Chief Executive Officer of Equifax from January 1998 to April 1999, and as President and Chief Operating Officer of Equifax from August 1997 to January 1998. Before that, he served as Executive Vice President and Group Executive of Equifax since April 1993. Mr. Chapman has been a director of Equifax since January 1994 and will remain a director of Equifax following the distribution. Mr. Chapman currently serves as a director of The Southern Company, an energy company.

Lee A. Kennedy is currently a director, and will remain a director and serve as our President and Chief Executive Officer following the distribution. Mr. Kennedy currently serves as President, Chief Operating Officer, and a director of Equifax, positions he has held since June 1999, and from which he will resign

immediately prior to the distribution. From June 1997 to June 1999, Mr. Kennedy served as Executive Vice President and Group Executive of Equifax. From July 1995 to July 1997 he served as President of Equifax's Payment Services group. From 1992 until July 1995, Mr. Kennedy was Senior Vice President and General Manager of Equifax Card Services.

Larry J. Towe will be our Executive Vice President and Chief Operating Officer following the distribution. Mr. Towe currently serves as Executive Vice President and Group Executive--Payment Services, positions he has held since July 1999. From May 1997 to July 1999, Mr. Towe served as Senior Vice President and General Manager, Equifax Card Solutions, International. Before that, since May 1996, he served as President, FBS Software, a provider of software solutions for payment cards, collections and merchant processing, which Equifax acquired in July 1994.

Michael T. Vollkommer will be our Corporate Vice President and Chief Financial Officer following the distribution. Mr. Vollkommer currently serves as Corporate Vice President and Controller of Equifax, a position he has held since November 1999. From December 1998 to August 1999, Mr. Vollkommer was Vice President--Finance of Superior TeleCom, Inc., a manufacturer of copper wire and cable products. Before that, he held executive officer positions with Alumax Inc., a producer of primary aluminum and fabricated aluminum products, including Vice President and Chief Financial Officer from December 1997 to August 1998, Vice President-Strategic Planning and Corporate Development from June 1997 to December 1997, and Vice President and Controller from January 1994 to June 1997.

Bruce S. Richards will be our Corporate Vice President, General Counsel, and Secretary following the distribution. Mr. Richards has been Corporate Vice President of Equifax since November 2000. From November 1996 to November 2000, Mr. Richards served as Corporate Vice President and General Counsel of Equifax. From January 1993 to November 1996, he served as Senior Vice President and Group Counsel of Equifax's Financial Services Group.

Gerard Ballard will be our Corporate Vice President and Chief Technology Officer following the distribution. Mr. Ballard currently serves as Chief Technology Officer of Equifax Payment Services, a position he has held since February 2001. From June 1997 to December 2000, Mr. Ballard served as Executive Vice President and Chief Information Officer of Vital Processing Services, LLC, a provider of technology-based commerce enabling services. From September 1995 to June 1997, he served as Vice President of Equifax Payment Services.

Gerald A. Hines will be our Senior Vice President and Group Executive--Card Solutions following the distribution. Mr. Hines currently serves as Senior Vice President Card Solutions--Americas for Equifax Payment Services, a position he has held since 1997. From April 1993 to August 1997, he was employed with AT&T Universal Card Services, a credit card issuer, where he served as Executive Vice President and Chief Operating Officer.

Jeffrey S. Carbiener will be our Senior Vice President and Group Executive-Check Solutions following the distribution. Mr. Carbiener currently serves as Senior Vice President, Equifax Check Solutions, a position he has held since 1998. He has held various management positions with both Equifax Card Services and Check Services since 1991.

David K. Hunt will be a director following the distribution. Mr. Hunt currently serves as Chief Executive Officer of PlanSoft Corporation, an Internet-based business-to-business solutions provider in the meeting and convention industry, a position he has held since May 1999. From January 1997 to April 1999, he served as President, Chief Executive Officer, and a director of Global Payment Systems, a transaction processing service provider. Mr. Hunt served as President and Chief Executive Officer of AT&T Universal Card Services Corporation, a credit card issuer, from April 1993 to November 1996.

Kathy Brittain White will be a director following the distribution. Ms. White currently serves as Executive Vice President--E-Business and Chief Information Officer of Cardinal Health, a health care products and services company, positions she has held since March 1999. From October 1996 to March 1999, she served as Senior Vice President and Chief Information Officer of Allegiance Healthcare, a provider of health care products and cost management services to the health care industry. Ms. White served as Corporate Vice President and Chief Information Officer of Baxter International, a global medical products and services company, from April 1995 to October 1996 and moved to Allegiance Healthcare following its spin-off from Baxter International in 1996. Ms. White currently serves as a director of National Service Industries, a company in the lighting equipment, chemicals, textile rental, and envelope businesses.

Richard Child will be a director following the distribution. Mr. Child has served as a business consultant to the financial and payments industry since June 2000. From November 1999 to May 2000, he served as Executive Vice President of ZonaFinanciera.com., a financial services portal. Mr. Child served as Executive Vice President and President for MasterCard International, a global bank card association, and as President of its Latin American division, from January 1996 to April 1999, and held various executive positions with MasterCard International since 1983.

Charles T. Doyle will be a director following the distribution. Mr. Doyle has served as Chairman of the Board of Texas First Bank, an independent community bank in Galveston County, Texas, since October 1972, and as Chairman and Chief Executive Officer of Texas Independent Bancshares, Inc., of Texas City, a provider of correspondent banking services to community banks since July 1979. He has also served as Chairman of the Board of Rust, Watt & Haney, Inc., an independent general insurance agency and subsidiary of Texas Independent Bancshares, since September 2000. From January 1996 until December 1998, Mr. Doyle was on the Federal Advisory Council to the Board of Governors of the Federal Reserve, and from January 1985 to December 1991, he served as Director of the Federal Reserve Bank in Dallas, Texas. Mr. Doyle currently serves as a director of bank card associations VISA USA, Inc., VISA International, Inc. and Inovant, a transaction processor for merchants and issuers of credit cards.

Robert H. Bohannon will be a director following the distribution. Mr. Bohannon currently serves as the Chairman of the Board, President and Chief Executive Officer of Viad Corp., a payments and conventions and events services provider, positions he has held since January 1997. He jointed Viad in 1993, serving as President and Chief Executive Officer of Travelers Express Company, the payment services and money order subsidiary of Viad, until August 1996, when he was appointed President and Chief Operating Officer of Viad after its spin-off of The Dial Corporation, a consumer products company. Mr. Bohannon currently serves as a director of Viad, and has been a member of Viad's board since 1996.

Annual Meeting

Our first annual meeting of shareholders after the distribution is expected to be held in April 2002. The annual meeting will be held at our principal office or at such other place and on such date as may be fixed from time to time by resolution of our board of directors.

Committees of the Board of Directors

Our board of directors will establish committees, described below, to assist in the discharge of its responsibilities.

Audit Committee

The audit committee will conduct its duties consistent with its written charter, which will include a review of: our financial reports and other financial information; systems of internal controls regarding finance, accounting, legal compliance, and ethics; and auditing, accounting, and financial reporting processes. The audit committee will annually recommend to the board of directors the firm of independent public accountants to be

selected as our auditors. The audit committee will also consult with our independent accountants, approve the scope of their audit and other work, and meet with members of our management including our director of internal audit. We expect that the following directors will serve on our audit committee: David K. Hunt, Chair; Charles T. Doyle; and Richard Child.

Compensation and Human Resources Committee

The compensation and human resources committee will conduct its duties consistent with its written charter, which will include responsibility for approving and monitoring executive compensation plans, policies, and programs. The committee will review and establish the compensation of and grants made to our officers, except for the chief executive officer. The committee will recommend the salary and incentive compensation for our chief executive officer, subject to the approval of our board. The committee will review and advise our management, as necessary, on succession planning and other significant human resource matters. In addition, the committee will monitor the effectiveness and funded status of any retirement plans we establish and our 401(k) plan, and approve or review significant employee benefit plan actions. We expect that the following directors will serve on our compensation and human resources committee: Kathy Brittain White, Chair; David K. Hunt; and Robert H. Bohannon.

Special Committees

The board of directors may from time to time establish special committees to act on behalf of the board of directors on matters delegated to it by the full board. This may include matters such as approval of final terms of acquisitions and divestitures, alliances, and capital expenditures.

Compensation Committee Interlocks and Insider Participation

None of the members of our compensation and human resources committee will have served as an officer or an employee of our businesses during the previous fiscal year, nor is any member expected to serve as one of our officers or employees following the distribution.

Director Compensation

Directors who are our salaried employees will receive no additional compensation for services as a director or as a member of a committee of our board. Following the distribution date, all non-employee directors other than our chairman of the board will receive an annual retainer of \$20,000, plus \$1,000 for each board or committee meeting he or she attends. Our non-employee chairman of the board will receive an annual retainer of \$250,000 payable in cash, shares of our common stock, or a combination of both, plus \$1,000 for each board or committee meeting he attends. The chair of each standing committee of our board will receive an additional annual fee of \$4,000. We will also reimburse each non-employee director for out-of-pocket expenses incurred in connection with attendance at board and committee meetings.

We intend to adopt a deferred compensation plan for the benefit of our non-employee directors. Under this plan, a non-employee director may defer and be deemed to invest up to 100% of his or her director's fees in either a stock fund representing our common stock or in an interest bearing account. Interest on deferred amounts deemed to be invested in the interest bearing account will be credited monthly to our directors' accounts at the prime lending rate on the first day of each month as reported in the Wall Street Journal. All deferred fees will be held in our general funds and will be paid in cash. In general, deferred amounts will not be paid until after the director terminates service from our board, at which time, they will be paid either in a lump sum or in annual payments of not more than ten years, as determined by the director.

Each non-employee director will also be granted options to purchase shares of our common stock pursuant to the Certegy Inc. 2001 Non-Employee Director Stock Option Plan described below.

Certegy Inc. 2001 Non-Employee Director Stock Option Plan

We intend to adopt, with the approval of Equifax in its capacity as our sole shareholder, the Certegy Inc. 2001 Non-Employee Director Stock Option Plan. The full text of the director plan has been filed as an exhibit to the registration statement on Form 10 that we have filed with the SEC.

We will establish the director plan to encourage ownership of our common stock by our directors, which gives directors an increased incentive to devote their efforts to our success on behalf of shareholders. The director plan will also help us to attract qualified directors. Each director who is not employed by us or any of our affiliates will be eligible to participate in the director plan.

Grants of awards under the director plan will be automatic. We intend the director plan to be a "formula plan" for purposes of Section 16(b) of the Exchange Act. Our board of directors will administer and interpret the director plan.

Shares subject to the director plan may be authorized but unissued shares or shares that were once issued and subsequently reacquired by us. The total number of shares of common stock for which options may be granted under the director plan will be 200,000 shares, subject to adjustment. Awards granted pursuant to the director plan will be subject to the following terms and conditions:

- . On the day following the distribution date, each non-employee director will be granted an option to purchase 2,000 shares of our common stock. Each person who later becomes a non-employee director will also be granted an option to purchase 2,000 shares of our common stock on the date that he or she becomes a non-employee director. In addition, as of the day following the annual meeting of our shareholders in 2002, and on the day following each subsequent annual meeting of our shareholders, each non-employee director serving on that date will be granted an option to purchase 2,000 shares of our common stock.
- . Each option granted under the director plan will become fully exercisable one year after date of grant. Options will also become fully exercisable upon the death, disability, or retirement of the director, as such terms are defined in the director plan, or if we experience a "change of control" as defined in the director's plan.
- . The exercise price for each option granted under the director plan will be the fair market value of the shares of common stock on the date of grant. Each option granted under the director plan will expire five years after the date of grant, unless the director plan and the option agreement provides for earlier termination or an extended exercise period.
- Options granted under the director plan will be assignable by will, by the laws of descent and distribution, or pursuant to an applicable qualified domestic relations order.

Termination and Amendment

The director plan will terminate automatically on the second day following our 2010 annual meeting of shareholders, but our board of directors may terminate the director plan at any time before that date. Our board of directors may amend the director plan at any time without shareholder approval; but it may condition any amendment on the approval of our shareholders if such approval is necessary or deemed advisable with respect to tax, securities or other applicable laws, policies, or regulations. No amendment, modification or termination of the director plan will adversely affect the rights of the participants who have outstanding options without the consent of those participants.

Certain Federal Income Tax Effects

The options granted under the director plan will be non-qualified stock options. Present federal income tax regulations impose no federal income tax consequences to us or a participant upon the grant of a non-qualified stock option. When the participant exercises a non-qualified option, however, he or she will realize ordinary

income in an amount equal to the excess of (1) the fair market value of the stock at the time of exercise, over (2) the exercise price, and we will be allowed a corresponding federal income tax deduction. Any gain that a participant realizes when the participant later sells or disposes of the shares received under the option will be short-term or long-term capital gain, depending on how long he or she held the shares.

Executive Compensation

The compensation of our executive officers will be approved by the compensation and human resources committee of our board of directors. Our compensation and human resources committee will consist entirely of non-employee directors. Our compensation and human resources committee has not yet established the compensation of our executive officers. We expect, however, that the compensation of the executive and other officers will consist principally of base salary, annual cash incentive, and long-term incentive compensation consisting of cash awards and equity-based awards.

Salaries of the executive officers will be based, among other factors, on our compensation and human resources committee's assessment of the executive's responsibilities, experience and performance, compensation data of other companies, and the competitive environment for attracting and retaining executives.

Annual cash incentives for executive officers will be determined each year at or following the end of our fiscal year, in accordance with targets established at or near the beginning of the fiscal year under the Certegy Inc. Key Management Long-Term Incentive Plan. Long-term incentive grants will also be made periodically by the committee under this plan and will be based on the achievement of one or more financial and other objectives over the measurement period established at the time of grant. We anticipate that the factors our compensation and human resources committee will consider in determining the amount of annual and long-term incentives to officers under this plan may include, among others, the executive officer's individual performance, including the quality of strategic plans, organizational and management development, special project leadership and similar indicators of individual performance, and our financial performance, which may be measured by earnings per share, revenues, return on equity, total return to the shareholders in the form of stock price appreciation and dividends, if paid, or other factors. Our executive officers may be granted the option to exchange the right to receive all or a part of the cash incentives that he earns for stock options for shares of our common stock.

Our equity-based awards will consist principally of stock options, which will be granted from time to time under the Certegy Inc. 2001 Stock Incentive Plan described below. We anticipate that our compensation and human resources committee will base grants of stock-based awards on various factors, including the number of shares of common stock outstanding, the number of shares of common stock authorized under the Certegy Inc. 2001 Stock Incentive Plan, the executive officer's ability to contribute to our future success, and the other elements of the executive's compensation.

We are a recently formed subsidiary of Equifax, incorporated on March 2, 2001. We were not in existence on December 31, 2000, the end of Equifax's last fiscal year. Although certain of the individuals who will be serving as our executive officers were performing services in connection with our businesses during the last fiscal year, those individuals were employed by Equifax during such period, and except for Larry J. Towe, Gerald A. Hines, and Jeffrey S. Carbiener, were not dedicated exclusively to our businesses, and, in fact, devoted substantial time and effort to other Equifax businesses or to the Equifax organization in general. Our company had no formalized executive management structure prior to our formation, and certain of the individuals who would have constituted the most highly compensated individuals providing services to our businesses as of the end of the last fiscal year are not, in fact, our executive officers. Accordingly, no historical information on the compensation of executive officers is reported. Our proxy statement for the 2002 annual meeting of shareholders will contain information on compensation paid to our executive officers in fiscal year 2001.

We describe below the current compensation arrangements for our chief executive officer and our four other most highly compensated executive officers, as measured by their base salaries, calculated on an annualized basis, and target bonus amounts. Although we expect that the compensation of our executive officers will consist principally of base salary, annual cash bonus and long-term incentive compensation, the compensation arrangements established by our compensation and human resources committee may differ from the compensation arrangements described below. In connection with the current compensation arrangements for our executive officers described below:

- . Base salary represents the executive officer's current base salary on an annualized basis.
- . Target bonus percentage represents the targeted percentage of annual base salary payable in the form of a bonus for the fiscal year ending December 31, 2001. The actual percentage paid may be greater or less than the target percentage based on a combination of the financial performance of Equifax and the organizations in which each individual was employed for the fiscal year ending December 31, 2001 and the indicators of individual performance described above for the same period.
- . Option grants were made under Equifax's stock option plans in 2000 and 2001. These options were granted as a conversion of all or a portion of annual incentive awards or as a management incentive award. Such option grants and awards may not be indicative of future grants and awards to our executive officers. All options units will be converted to options for our common stock and as described in "Relationship Between Equifax and Our Company After the Distribution-- Employee Benefits Agreement-- Stock and Incentive Compensation Plans."

The current compensation for our chief executive officer and our four other most highly compensated officers is as follows:

- . Lee A. Kennedy, who will serve as our President and Chief Executive Officer following the distribution, currently receives a base salary equal to \$460,000 on an annualized basis. Mr. Kennedy's targeted bonus percentage is 70%. Since December 31, 2000, Mr. Kennedy has received options to purchase up to 4,793 shares of common stock of Equifax and no shares of Equifax restricted stock.
- . Larry J. Towe, who will serve as our Executive Vice President and Chief Operating Officer following the distribution, currently receives a base salary equal to \$330,000 on an annualized basis. Mr. Towe's targeted bonus percentage is 60%. Since December 31, 2000, Mr. Towe has received options to purchase up to 19,496 shares of common stock of Equifax and no shares of Equifax restricted stock.
- Michael T. Vollkommer, who will serve as our Corporate Vice President and Chief Financial Officer following the distribution, currently receives a base salary equal to \$230,000 on an annualized basis. Mr. Vollkommer's targeted bonus percentage is 40%. Since December 31, 2000, Mr. Vollkommer has received options to purchase up to 47,717 shares of common stock of Equifax and no shares of Equifax restricted stock.
- . Jeffrey S. Carbiener, who will serve as our Senior Vice President and Group Executive, Check Solutions following the distribution, currently receives a base salary of \$225,000 on an annualized basis. Mr. Carbiener's targeted bonus percentage is 40%. Since December 31, 2000, Mr. Carbiener has received options to purchase up to 7,000 shares of common stock of Equifax and no shares of Equifax restricted stock.
- . Gerald A. Hines, who will serve as our Senior Vice President and Group Executive, Card Solutions following the distribution, currently receives a base salary equal to \$256,000 on an annualized basis. Mr. Hines' targeted bonus percentage is 40%. Since December 31, 2000, Mr. Hines has received options to purchase up to 7,000 shares of common stock of Equifax and no shares of Equifax restricted stock.

Pension Plans

Prior to the distribution, most of our United States employees, including management employees and executive officers, are participants in the Equifax U.S. Retirement Income Plan, a non-contributory, qualified defined benefit pension plan. Effective at the time of the distribution, we will adopt a qualified defined benefit pension plan that will be substantially similar, in all material respects, to the Equifax pension plan. As explained above, our pension plan may integrate benefits for a period of time with the Equifax pension plan. The following is a summary description of the expected benefits of our pension plan.

In general, under our pension plan, annual retirement benefits will be computed based on length of service and a participant's average total earnings up to a maximum of either 125% of base salary or base salary plus 75% of other earnings, whichever is greater. Benefits under this plan will be computed by averaging the participant's total earnings, including earnings from Equifax, for the highest paid thirty-six consecutive months of employment. However, federal laws place limitations on earnings amounts that may be included in calculating benefits under this plan. In 2001, only the first \$170,000 in eligible earnings could be included in the calculation.

The normal retirement age under our pension plan will be 65. However, employees who attain age 55 with at least 5 years of service, or employees who attain age 50 and the sum of his or her age and years of service is at least equal to 75, can retire early with reduced benefits. Prior service with Equifax will be recognized for all purposes under this plan. For employees who retire prior to age 65, their retirement benefits are reduced by 3% per year for the first five years and by 5% per year thereafter.

Executive Life and Supplemental Retirement Benefit Plan

We will adopt for our participating management employees and executive officers the Executive Life and Supplemental Retirement Benefit Plan in a form that is substantially similar to the one maintained by Equifax. The plan is a split-dollar life insurance program, under which we pay premiums on life insurance policies that are owned by the plan participants. The policies will be collaterally assigned to us by the participants to the extent of our premium payments towards the policies. With respect to the existing policies of our employees, Equifax will assign to us the collateral assignments. The plan is structured to provide participants with current life insurance protection and the potential for future capital accumulation. The amount of life insurance protection may vary depending upon a number of factors, including the employment level of the participant.

If the participant is actively employed, or after the distribution date retires, becomes permanently disabled, or has a qualifying termination of employment following a change of control, we will continue to pay the premiums on their policy. Premium payments have been structured so that no further premium payments are required from us after approximately seven years. After an initial three-year vesting period, the participant obtains a nonforfeitable right to the cash value of the policy minus the aggregate premium payments made by us. Thereafter, participants may direct the investment of all funds accumulated in the policy, including the amount equal to our premium payments. If a participant retires, becomes permanently disabled, or has a qualifying termination of employment following a change of control, the arrangement continues until the maturity of the policy, at the later of age 60 or 15 years in force, when we will recover the aggregate employer premium payments from the policy's cash value and release our interest in the policy. If the participant's job is eliminated by us, our contributions end but the policy will remain in place until maturity, when we will recover the aggregate employer premium payments and release our interest, or it may end earlier at our discretion. If the participant terminates employment for other reasons, we will recover the aggregate employer premiums and release our interest at that point. If a participant is terminated for cause, the participant forfeits all rights to the cash value under his or her policy.

We have also established a rabbi trust in connection with the plan. In the event of a change in control, we are required to fully fund the trust and thereafter the trustee will make premium payments on the participants' policies if for any reason we fail to do so.

We intend to adopt, with the approval of Equifax in its capacity as our sole shareholder, the Certegy Inc. 2001 Key Management Long-Term Incentive Plan. The full text of the incentive plan has been filed as an exhibit to the registration statement on Form 10 that we have filed with the SEC. We will establish the incentive plan to provide short-term and long-term incentive compensation to our officers and other key management personnel who make substantial contributions to our success, and to assist in attracting and retaining the highest quality individuals in key executive positions.

Plan Administration

Our compensation and human resources committee will administer this plan and has the authority to amend, suspend, or terminate this plan as long as such action does not adversely affect any award that is already outstanding under this plan. No awards may be granted under this plan after the tenth anniversary of the date our board approved the plan.

Eliqibility

Our compensation and human resources committee will be authorized in its discretion to make grants under this plan to any of our elected officers and other key management personnel and any elected officers and other key management personnel of any of our subsidiaries. Because the number of elected officers may change, and because the selection of additional participants will be discretionary, it is not possible to determine the exact number of individuals who will be eligible for grants under this plan.

Determination of Benefits

Our compensation and human resources committee will determine the amount of each grant and will establish performance goals that will form the basis for each grant. These performance goals will be based on the achievement of one or more management objectives that are specified at the commencement of the grant measurement period. These management objectives will consist of specified levels of, or growth in, one or more of the following areas, and may be an average over the measurement period: earnings, earnings per share, economic value added, revenue, sales, operating profit, net income, total return to shareholders, market share, profit margin, cash flow/net assets ratio, debt/capital ratio, return on total capital, return on equity, return on assets and common stock price. Management objectives may be company-wide or objectives related to a specific division, subsidiary, affiliate, department, region, or function in which the participant is employed. Management objectives may also be made relative to the performance of other companies. The relevant measurement period may consist of a portion of a year, a single year, a number of years or such other period as the compensation and human resources committee may determine. The long-term incentives will pay various cash amounts depending on the level to which the stated performance goals are achieved. If minimum levels are not achieved, the incentives will pay zero. Further, the compensation and human resources committee will have the discretion to reduce the amount of any incentive that would otherwise be payable, or to determine that no portion of the award will be paid, but the committee will not be able to increase the amount of any award. In addition, the actual performance results may be adjusted to take into account significant or unusual items, for example, acquisition, dispositions or equity restructurings, or other significant items such as accounting or tax charges.

Payment

If the specified performance goals are not met, the employment of the participant is terminated or the status of the employee changes such that the employee is no longer eligible to participate in this plan, then the incentive will be forfeited and cancelled, either in whole or in part. Our compensation and human resources committee will have the authority to provide the participant with the right to elect to convert his or her right to receive cash payment of an incentive for stock options to purchase our common stock, restricted shares of our common stock, or any other equity interest determined by the committee. Any such equity interest would be issued by the committee pursuant to the Certegy Inc. 2001 Stock Incentive Plan. If the participant terminates

employment with us prior to the end of the measurement period as the result of the participant's death, disability, or retirement, the committee will have the authority to waive the requirement of continued employment described above. In that case, the committee will have the discretion to pay a portion of the incentive pro rata in proportion to the length of service during the measurement period and actual performance against goals.

Change in Control

If there is a "change in control", as defined in the plan, of our company during any measurement period, then the participant will be entitled to receive an amount in cash equal to:

- . the target payment if the change in control occurs during the first measurement year, or
- . 150% of the target payment under the incentive if the change in control occurs after the first year of a multiyear measurement period, but no less than the projected payout determined on the effective date of the change in control if the change in control occurs during the last three months of the measurement period.

Limitation of Benefits

Under the incentive plan, no individual will be entitled to receive payments exceeding \$5 million in any calendar year, pursuant to incentives granted in any calendar year. For this purpose, payments include cash and any equity opportunity elected by the individual in lieu of cash, valued as of the date we grant the incentive. An individual will not be able to transfer his or her interest in an incentive.

Federal Income Tax Consequences to Us and the Participants

Under present federal income tax law, long-term incentives paid in cash will be ordinary income to participants in the year paid. For a limited transition period, the compensation we pay pursuant to the incentive plan qualifies as "performance-based compensation" not subject to the limitation on deductibility of executive compensation in excess of \$1,000,000 under Section 162(m) of the Internal Revenue Code. We anticipate seeking shareholder approval of this plan or a new incentive plan before the end of the transition period to continue to qualify such long-term incentives as "performance-based compensation" under Section 162(m).

Certegy Inc. 2001 Stock Incentive Plan

We intend to adopt, with the approval of Equifax in its capacity as our sole shareholder, the Certegy Inc. 2001 Stock Incentive Plan. We will reserve 6,600,000 shares of the authorized but unissued shares of our common stock for issuance under the option plan subject to increase as provided below. The full text of the option plan has been filed as an exhibit to the registration statement on Form 10 that we have filed with the SEC. We will establish the option plan to attract and retain officers and other key employees and to provide these individuals with incentives for superior performance.

Plan Administration

The option plan will be administered by the compensation and human resources committee of our board of directors. This committee will have the discretion to delegate to one or more of our officers its authority and duties under this plan with respect to participants who are not subject to the reporting and other requirements of Section 16 of the Exchange Act. The committee will have the right to terminate this plan at any time, or amend this plan so long as the termination or amendment does not adversely affect any rights of any participant with respect to outstanding awards without that participant's consent. Shareholder approval will be required to increase the number of shares of our common stock that may be granted under this plan, change the class of individuals eligible to become participants, or extend the duration of this plan.

Types and Number of Awards Under the Plan

Our compensation and human resources committee or its delegate will have the authority to grant:

- . stock options, including both incentive and non-qualified stock options;
- . stock appreciation rights or SARs;
- . performance shares or units;
- restricted stock; and
- . deferred stock.

The total number of shares that may be issued pursuant to grants made under this plan is 6,600,000, plus, commencing on January 1, 2002, and on each subsequent January 1, ending on January 1, 2008, an additional number of shares shall be added to the total available, equal to 1 1/2% of the number of shares of our common stock issued and outstanding on that date. The number of shares available will be adjusted to account for shares relating to awards that expire or are transferred, surrendered, or relinquished upon payment of any option price by transfer of shares or upon satisfaction of any withholding amount. The total number of shares issued upon exercise of all incentive stock options under the plan will not exceed 10,000,000 shares. These totals, and the individual limits described below, will be adjusted by our compensation and human resources committee in its discretion to reflect any change in the number of shares of common stock due to any stock dividend, stock split, combination, recapitalization, merger, spin-off, or similar corporate transaction. No individual participant will be awarded option rights or SARs for more than 750,000 shares during any calendar year, and no more than 500,000 shares of restricted stock may be awarded to any individual participant during any calendar year.

Eligibility

Our compensation and human resources committee or its delegate will be authorized to grant options, SARs, restricted stock, or deferred stock under this plan to any of our officers or other key employees, or others performing services for us or any officers, other key employees, or service providers of our subsidiaries and to grant awards of options or restricted stock to our non-employee directors.

Duration of Options and Other Awards

No stock option, SAR, or other award under this plan will be made more than 10 years after the date that this plan is approved by Equifax, as our sole shareholder. No stock option or SAR may be exercised more than 10 years from the date of grant.

Vesting and Exercise of Options

Options become exercisable when they have vested. Vesting schedules, and other specific terms of an option award, will be fixed by our compensation and human resources committee and set forth in an agreement, which will generally provide for vesting in equal increments over a period of years. If specified in the grant, options may become fully vested and exercisable if we experience a "change in control." For purposes of this plan, a "change in control" will have a meaning that is substantially identical to the definition that will be contained in several of our compensation plans and programs, including our 2001 Key Management Long-Term Incentive Plan, and the change in control agreements described below on page 63. The terms of an option grant may also provide for additional options to be granted at then current market value to an optionee upon exercise. Further, any option grant may specify management objectives that must be achieved as a condition to exercise. For this purpose, "management objectives" mean measurable performance objectives, either company-wide or related to a particular subsidiary, division, department, region, or function in which a participant is employed, and may relate to periods of one or more years. Although subject to the discretion of our compensation and human resources committee, options generally will not be exercisable after a participant

terminates employment with us, unless the termination was the result of the participant's death, disability, retirement, or job elimination.

Payment for Options

The exercise price of any stock option granted under this plan generally will be not less than 100% of the market value of our stock on the date of grant. The compensation and human resources committee will not, without the further approval of the shareholders, except for certain capital adjustments, restructurings, or reorganizations, have the authority to re-price any outstanding option rights to reduce the exercise price. Participants will have the right to exercise an option by making payment in any one or more of the following ways, as specified at the time of grant:

- . by cash or check;
- . by transfer of shares of our stock that have been owned by the participant for at least six months, or with respect to options that do not qualify as incentive stock options, by transfer of restricted or deferred shares or other option rights; or
- . by "cashless exercise," where a bank or broker-dealer we have approved sells some of the shares acquired and delivers the proceeds to us.

Stock Appreciation Rights

Our compensation and human resources committee will have the authority to grant SARs separately or in tandem with stock options. SARs granted in tandem with an incentive stock option must be granted concurrently with the option. SARs may be paid in cash, shares, or a combination as specified at the time of grant, and the committee will have the authority to grant to the participant the right to elect among those alternatives. SARs may contain vesting provisions, management objectives, and change of control provisions similar to those described above regarding options.

Restricted Stock

Our compensation and human resources committee may authorize grants of restricted stock which may or may not require additional payment. Each grant of restricted stock will be subject to certain conditions that will constitute a risk of forfeiture for a period determined by the compensation and human resources committee at the time of grant. These conditions, which will be established by the committee, may include management objectives as described above or may require continued employment. Any restricted stock award may require that all dividends or other distributions paid during the period of restriction be subject to these conditions.

Deferred Stock

Our compensation and human resources committee will have the authority to grant rights to receive shares of common stock at the end of a specified deferral period. Awards of deferred stock may be made for no consideration or for an amount that is less than the fair market value on the date of grant. The deferral period must be at least one year, unless subject to earlier lapse or modification in the event of a change of control.

General

Unless otherwise determined by the compensation and human resources committee, no grant under this plan will be transferable other than by will or the laws of descent and distribution. The committee will have the authority to provide that grants will be transferable to members of a participant's immediate family, or trusts for, or partnerships consisting of, those persons.

This plan will specifically authorize our compensation and human resources committee to provide for special terms for grants to persons who are foreign nationals or employed outside the U.S., as our compensation and human resources committee considers necessary to accommodate differences in local law, tax policy, or custom. Our compensation and human resources committee will have the authority to approve

supplements or amendments, restatements, or alternative versions of this plan as it considers necessary or appropriate for these purposes.

Federal Income Tax Consequences to the Company and the Participants

Some of the options granted under this plan may be incentive stock options, also referred to as ISOs, within the meaning of Section 422 of the Internal Revenue Code. Under present federal tax laws, there are no federal income tax consequences to either us or the participant upon the grant or exercise of an ISO. If the participant does not dispose of the stock acquired through the ISO within two years of the date of grant or one year of the date of exercise, any gain realized from a subsequent disposition would constitute long-term capital gain to the participant. If the participant does dispose of the stock prior to the expiration of either of those holding periods, any gain equal to the excess of the fair market value of the stock on the date of exercise or, if less, the amount realized on the disposition of the stock if a sale or exchange, over the option price would constitute ordinary income to the participant. Any additional gain realized upon the disposition would be taxable either as a short-term capital gain or long-term capital gain, depending upon how long the participant held the stock. We would receive a deduction in the amount of any ordinary income recognized by the participant.

Stock options that do not constitute ISOs, which are also known as nonqualified options, may also be granted under this plan. Under present federal tax laws, there are no federal income tax consequences to either us or the participant upon the grant of a nonqualified option. However, the participant will realize ordinary income upon the exercise of a nonqualified option in an amount equal to the excess of the fair market value of the stock acquired upon exercise over the option price, and we will receive a corresponding deduction. Any gain realized upon a subsequent disposition of the stock will constitute either a short-term or long-term capital gain to the participant, depending on how long it is held.

Unless the participant makes a special tax election, restricted stock awards are not taxable to the participant as long as the shares remain nontransferable and subject to a substantial risk of forfeiture. When these transferability restrictions and/or forfeiture risks are removed, the participant generally will recognize as ordinary income the fair market value of the stock, less any amounts that were paid to acquire the stock. We will receive a federal income tax deduction equal to the amount of ordinary income realized by the participant.

No taxable income is recognized by a participant upon the grant of a SAR. Upon the exercise of a SAR, the participant will recognize as ordinary income the cash received, plus the fair market value of any stock acquired in settlement of the SAR, less any amount required to be paid for the SAR. We will receive a federal income tax deduction equal to the amount of ordinary income realized by the participant.

Participants are required to pay tax due upon exercise of a non-qualified stock option, a lapse of restrictions on restricted stock or other recognition event. In the discretion of the committee, tax obligations may be satisfied by selling or forfeiting a portion of the shares granted that would be realized from the grant.

Change of Control Agreements

We expect to enter into change in control agreements with some or all of our executive officers. These agreements will have five-year terms with automatic renewal provisions and become operative only upon a change in control of our company. A "change in control" is generally defined by the agreements to mean:

- . an accumulation by any person, entity, or group of 20% or more of the combined voting power of our voting stock;
- . a business combination resulting in our shareholders immediately prior to the combination owning less than two-thirds of the common stock and combined voting power of the new company;
- . a sale or disposition of all or substantially all of our assets; or
- . approval by the shareholders of our complete liquidation or dissolution.

If any of these events happen, and the executive's employment terminates within six months prior to, or in the case of our agreements with (a) Messrs. Kennedy, Towe and Vollkommer, three years after, and (b) Messrs. Carbiener and Hines, two years after, the date of the change in control, other than from death, disability, or termination for cause or voluntary termination other than for "good reason," he will be entitled to a severance payment and other benefits described in the agreements. The severance payment will equal up to, in the case of our agreements with (a) Messrs. Kennedy, Towe and Vollkommer, three times, and (b) Messrs. Carbiener and Hines, two times, the sum of (i) that executive's highest annual salary for the twelve months prior to the termination, and (ii) the executive's highest bonus for the three years prior to termination.

Benefits payable under this agreement and other compensation or benefit plans of ours are not reduced because of Section 280G of the Internal Revenue Code. Any payments the executive receives will be increased, if necessary, so that after taking into account all taxes, including any excise taxes under Section 4999 of the Code, he would incur as a result of those payments, the executive would receive the same after-tax amount he would have received had no such excise taxes been imposed.

OWNERSHIP OF OUR COMMON STOCK

All of the outstanding shares of our common stock are, and will be, prior to the distribution, held beneficially and of record by Equifax. The following table sets forth information concerning shares of our common stock projected to be beneficially owned immediately after the distribution date by:

- . each person or entity known by us to own more than 5% of the outstanding shares of Equifax's common stock;
- . each person who we currently know will be one of our directors at the time of the distribution;
- . each person who we currently know will be one of our named executive officers at the time of the distribution; and
- . all persons who we currently know will be our directors and executive officers at the time of the distribution as a group.

The share amounts in the table below reflect a distribution ratio of one share of our common stock for every two shares of Equifax common stock currently held by each of the listed persons or entities. To our knowledge, except under applicable community property laws or as otherwise indicated in the footnotes below, each person or entity has sole voting and investment power with respect to the shares of common stock set forth opposite such person's or entity's name. Beneficial ownership is determined in accordance with the rules of the SEC and generally includes voting or investment power with respect to securities. Shares of common stock and options, warrants, and convertible securities that are currently exercisable or convertible within 60 days of June 12, 2001 into shares of our common stock are deemed to be outstanding and to be beneficially owned by the person holding the options or warrants for the purpose of computing the percentage ownership of the person, but are not treated as outstanding for the purpose of computing the percentage ownership of any other person.

Directors and Executive Officers and Five Percent Holders

	Our Common Stock	Percent of Shares Beneficially Owned
Name	Beneficially Owned(1)	
Trustees of General Electric Pension Trust and GE Asset Management		
Incorporated, 3003 Summer Street, Stamford, Connecticut 06904 (3)	4,751,385	6.97%
Thomas F. Chapman	416,540(4)	*
Lee A. Kennedy	141,239(5)	*
Larry J. Towe	79,828(6)	*
Michael T. Vollkommer	27,339(7)	*
Jeffrey S. Carbiener	28,835(8)	*
Gerald A. Hines	69,733(9)	*
David K. Hunt	500(10)	*
Kathy Brittain White		
Richard Child		
Charles T. Doyle	46(11)	*
Robert H. Bohannon		
All directors and executive officers as a group, including those		
named above (13 persons)	839,312	1.23%

^{*} Indicates less than 1%

- (1) The amounts included in this column represent the shares of our common stock that will be beneficially held by such person or entity based on the distribution ratio of one share of our common stock to be received for every two shares of Equifax common stock held by such person or entity. The stock options included in the numbers have not been adjusted to give effect to the distribution.
- (2) Assumes that we will have 68,121,712 shares of common stock outstanding, based on Equifax having 136,243,425 shares of common stock entitled to receive our shares in the distribution outstanding on March 31, 2001.

- (3) As reported in Schedule 13G filed with the SEC on February 14, 2001, the Trustees of General Electric Pension Trust have shared voting and dispositive power for 2,942,546 Equifax shares, and GE Asset Management Incorporated has sole voting and dispositive power for 6,560,024 Equifax shares and shared voting and dispositive power for 2,942,546 shares. General Electric Company disclaims beneficial ownership of all shares reported.
- (4) Includes 2,963 shares of Certegy Common Stock currently owned through the Equifax Inc. Employees 401(k) Retirement and Savings Plan.
- (5) Includes 15,153 shares of Certegy Common Stock currently owned through the Equifax Inc. Employees 401(k) Retirement and Savings Plan.
- (6) Includes 1,228 shares of Certegy Common Stock currently owned through the Equifax Inc. Employees 401(k) Retirement and Savings Plan.
- (7) Includes 101 shares of Certegy Common Stock currently owned through the Equifax Inc. Employees 401(k) Retirement and Savings Plan.
- (8) Includes 892 shares of Certegy Common Stock currently owned through the Equifax Inc. Employees 401(k) Retirement and Savings Plan.
- (9) Includes 86 shares of Certegy Common Stock currently owned through the Equifax Inc. Employees 401(k) Retirement and Savings Plan.
- (10) Includes 500 shares held by Mr. Hunt's wife as to which he disclaims all beneficial ownership.
- (11) Held jointly with spouse.

DESCRIPTION OF CAPITAL STOCK

The following information reflects our articles of incorporation and bylaws as these document will be in effect at the time of the distribution.

Our Authorized Capital Stock

Immediately after the distribution, our authorized capital stock will consist of 300,000,000 shares of common stock par value \$.01 per share, and 100,000,000 shares of preferred stock, par value \$.01 per share. Immediately after the distribution, approximately 68,121,712 shares of our common stock will be outstanding based on Equifax having 136,243,425 shares of common stock entitled to receive shares in the distribution outstanding on March 31, 2001 and assuming no exercise of outstanding options. We have reserved 6,600,000 shares for issuance under the Certegy Inc. 2001 Stock Incentive Plan, and 200,000 shares for issuance under the Certegy Inc. 2001 Non-Employee Directors Stock Incentive Plan. No shares of preferred stock have been issued.

The following summary describes material provisions of the amended and restated articles of incorporation and amended and restated bylaws that will become effective immediately prior to the distribution. You should read copies of these documents, which have been filed as exhibits to the registration statement on Form 10 that we have filed with the SEC.

Our Common Stock

The holders of our common stock will be entitled to one vote for each share on all matters voted on by shareholders, including elections of directors, and will possess all voting power, except as otherwise required by law or provided in any amendment to our articles adopted by our board of directors with respect to any series of our preferred stock. Our articles of incorporation will not provide for cumulative voting in the election of directors. Accordingly, the holders of a majority of our shares voting for the election of directors will be able to elect all of the directors, if they choose to do so, subject to any rights of the holders of any preferred stock to elect directors. Subject to any preferential or other rights of any outstanding series of our preferred stock that may be established by our board of directors, the holders of our common stock will be entitled to such dividends as our board of directors may declare from time to time from legally available funds and, upon our dissolution, will be entitled to receive pro rata all of our assets available for distribution to shareholders.

Our Preferred Stock

Our articles of incorporation will authorize our board of directors to provide for the issuance of shares of our preferred stock in one or more series, and to fix and determine, with respect to any series of our preferred stock, the number of shares of such series and the voting powers, designations, preferences, limitations, and restrictions of those shares. The authority of our board with respect to any series of our preferred stock will include the establishment of all voting powers, preferences, designations, rights, qualifications, limitations, and restrictions described in Section 14-2-601(d) of the Georgia Business Corporation Code.

We believe that the ability of our board of directors to issue one or more series of our preferred stock will provide us with flexibility in structuring possible future financings and acquisitions, and in meeting other corporate needs that might arise. The authorized shares of our preferred stock, as well as shares of our common stock, will be available for issuance without further action by our shareholders, unless action is required by applicable law or the rules of any stock exchange or automated quotation system on which our securities may be listed or traded. The New York Stock Exchange currently requires shareholder approval as a prerequisite to listing shares in several instances, including where the present or potential issuance of shares could result in an increase in the number of outstanding shares of common stock, or in the amount of outstanding voting securities, of at least 20%. If the approval of our shareholders is not required for the issuance of shares of our preferred stock or our common stock, our board may determine not to seek shareholder approval.

Although our board of directors has no intention at the present time of doing so, it could issue a series of our preferred stock that could, depending on the terms of such series, impede the completion of a merger, tender offer, or other takeover attempt. Our board of directors will make any determination to issue such shares based on its judgment as to the best interests of us and our shareholders. Our board of directors, in so acting, could issue our preferred stock with terms that could discourage an acquisition attempt through which an acquiror may be able to change the composition of our board of directors, including a tender offer or other transaction that some, or a majority, of our shareholders might believe to be in their best interests or in which shareholders might receive a premium for their stock over the then current market price of such stock.

Anti-Takeover Effects of Provisions of Our Articles of Incorporation and Bylaws

Board of Directors

Before the distribution, our articles of incorporation and bylaws will divide our board of directors into three classes of directors serving staggered three-year terms. As a result, approximately one-third of our board of directors will be elected each year. One class of our directors will initially serve a one-year term, a second class of our directors will initially serve a two-year term, and the third class will initially serve a three-year term. Beginning in 2002, one class of directors will be elected each year for a three-year term. See "Management--Directors." Under the Georgia Business Corporation Code, our staggered terms for directors may only be adopted, amended, or repealed by the shareholders.

Our staggered board of directors could prevent a party who acquires control of a majority of our outstanding voting stock from obtaining control of our board of directors until the second annual shareholders meeting following the date on which the acquiror obtains the controlling stock interest. This result could have the effect of discouraging a potential acquiror from making a tender offer or otherwise attempting to obtain control of our company.

Number of Directors; Removal; Filling Vacancies

Our articles of incorporation and bylaws will provide that the number of directors will be fixed by resolution of our board of directors, from time to time. If the number of directors is increased or decreased, the increase or decrease will be apportioned among the classes so as to maintain, as nearly as possible, an equal number of directors in each class, provided, however, that no decrease in the number of directors in a class will shorten the term of an incumbent director. Any additional director elected to fill a vacancy resulting from an increase in the size of our board will hold office for a term that coincides with the remaining term of the class to which such director is elected, with certain exceptions.

Our articles of incorporation and bylaws will provide that any vacancy on our board that results from an increase in the number of directors, or from the death, resignation, retirement, disqualification, or removal from office of any director, will be filled by a majority of the remaining members of our board, though less than a quorum, or by the sole remaining director. Any director elected to fill a vacancy resulting from the death, resignation, retirement, disqualification, or removal from office of a director will have the same remaining term as his or her predecessor. Accordingly, our board could temporarily prevent any shareholder from enlarging our board and filling the new directorships with that shareholder's own nominees.

Advance Notice for Shareholder Proposals or Nominations at Meetings

Our bylaws will provide that in order to bring certain matters before the annual meeting of shareholders, including nominations of directors, shareholders must give us notice containing certain information within the time period specified in SEC Rule 14a-8(a)(3)(i). Shareholders making proposals, other than those that appear in a proxy statement after compliance with SEC Rule 14a-8, must file written notice with our board setting forth certain information called for by our bylaws.

Special Meetings of Shareholders

Our bylaws will provide that a special meeting of shareholders may be called by:

- . our chairman or a vice chairman, if any, of our board;
- . our chief executive officer;
- . our president;
- . our board by vote at a meeting;
- . a majority of our board in writing without a meeting; or
- . the unanimous call of our shareholders.

Our bylaws provide that in order to bring certain matters before a special meeting of shareholders, including the nomination of directors, shareholders must give us notice containing certain information no later than the close of business on the earlier of:

- . the 30th day following the public announcement that a matter will be submitted to a vote of the shareholders at a special meeting; or
- . the 10th day following the day on which notice of the special meeting was given.

Other Constituencies

In discharging the duties of their respective positions and in determining what is believed to be in the best interests of our company, our board, committees of our board, and individual directors, in addition to considering the effects of any action on our company or its shareholders, will be authorized under a provision of our articles of incorporation to consider the interests of our employees, customers, suppliers, and creditors and the employees, customers, suppliers, and creditors of our subsidiaries, the communities in which our offices or other establishments are located, and all other factors the directors consider pertinent. This provision will permit our board to consider numerous judgmental or subjective factors affecting a proposal for a business combination, including some non-financial matters, and on the basis of these considerations, our board will be permitted to oppose a business combination or other transaction which, viewed exclusively from a financial perspective, might be attractive to some, or even a majority, of our shareholders.

Amendment of Our Articles

Under the Georgia Business Corporation Code in general, and except as otherwise provided by our articles of incorporation, amendments to our articles of incorporation must be recommended to the shareholders by our board and approved at a properly called shareholder meeting by a majority of the votes entitled to be cast on the amendment by each voting group entitled to vote on the amendment. Our articles of incorporation will require the affirmative vote of the holders of not less than two-thirds of the votes entitled to be cast by the holders of all then outstanding shares of voting stock, voting together as a single class, to make, alter, amend, change, add to, or repeal any provision of, our articles of incorporation or bylaws where such creation, alteration, amendment, change, addition, or repeal would be inconsistent with the provisions of our articles relating to:

- . the number of members of our board;
- . the classification of our board into classes of directors with staggered terms; or
- . the filling of vacancies on our board.

Notwithstanding the foregoing, this two-thirds vote will not be required for any alteration, amendment, change, addition, or repeal recommended by a majority of our board.

Amendment of Our Bylaws

Under the Georgia Business Corporation Code in general, and subject to our articles of incorporation and the requirements of the business combination and fair price provisions described below, our bylaws may be altered, amended, or repealed by our board or by the affirmative vote of the holders of a majority of the shares of our common stock entitled to vote and actually voted on such matter.

Rights Agreement

Our board of directors currently expects to adopt a rights agreement with SunTrust Bank, as rights agent, on or prior to the distribution date, and to declare a special dividend of one right in respect of each share of our common stock distributed in the distribution or otherwise outstanding. When exercisable, each right will entitle the registered holder to purchase from us one share of our common stock at an initial exercise price of \$125, in accordance with the terms of the rights agreement and subject to customary anti-dilution adjustments described below. The rights agreement has been filed as an exhibit to the registration statement and the following summary is qualified in its entirety by reference to the full text of the rights agreement. For information on how to receive the rights agreement, please see "Where You Can Obtain Additional Information."

Anti-Takeover Effects

The rights are intended to have anti-takeover effects. If the rights become exercisable, the rights will cause substantial dilution to a person or group that attempts to acquire or merge with us on terms not approved by our board. unless the offer is conditioned on a substantial number of rights being acquired. Accordingly, the existence of the rights may deter a potential acquiror from making a takeover proposal or tender offer. The rights should not interfere with any merger or other business combination approved by our board of directors since we may redeem the rights at a nominal price, as described below. Thus, the rights are intended to encourage persons who may seek to acquire control of us to initiate such an acquisition through negotiations with our board of directors. However, the effect of the rights may be to discourage a third party from making a tender offer or otherwise attempting to obtain a substantial equity position in our equity securities or seeking to obtain control of us. To the extent any potential acquirors are deterred by the rights, the rights may have the effect of preserving incumbent management in office.

Separation, Distribution, and Exercisability of Rights

Initially, the rights will attach to all outstanding shares of our common stock and will be represented by the certificates for those outstanding shares. No separate rights certificates will initially be distributed. The rights will separate from the common stock for distribution to the holders of the rights as of the close of business on the tenth calendar day after the earliest of:

- . a public announcement that a person or group has acquired, or obtained the right to acquire, beneficial ownership of 15% or more of our outstanding shares of common stock, which we refer to as an acquiring person, or such later date as may be specified by our board;
- initiation of a tender offer or exchange offer by a person or group, the consummation of which would result in beneficial ownership by that person or group of 15% or more of our outstanding shares of common stock;
- . the first date of public announcement of the first occurrence of any of the following "flip-in" events:
 - any acquiring person becomes the beneficial owner of 20% or more of the outstanding shares of our common stock;
 - . any acquiring person merges into us or combines with us and we are the surviving corporation, or any acquiring person effects certain other transactions with us, as described in the rights agreement; or

- . any reclassification of securities or recapitalization or reorganization of our company which has the effect of increasing by more than 1% the proportionate share of the outstanding shares of any class of our equity securities or any of its subsidiaries beneficially owned by an acquiring person;
- the first date of public announcement of the first occurrence of any of the following "flip-over" events to occur after public announcement that a person or group has become an acquiring person:
 - we merge with or into any person and we are not the surviving corporation;
 - any person merges with or into us and we are the surviving corporation, but shares of our common stock are changed or exchanged;
 - . we are party to a statutory share exchange; or
 - . we sell 50% or more of our assets or earning power.

Until the rights separate and are distributed following these events, the rights will automatically transfer with our shares of common stock, and new common stock certificates issued in the distribution or after the distribution upon transfer or new issuance, will contain a notation incorporating the terms of our rights agreement by reference. Promptly after the rights separate from our shares of common stock we will mail rights certificates to holders of record of our common stock on the date when the separation occurs and, thereafter, the separate rights certificates alone will represent the rights.

Effective as of the date on which the rights separate and are distributed, rights that are or were beneficially owned by an acquiring person or an acquiring person's affiliate or associate, or by any transferee of any of the foregoing, will be void.

The rights will not be exercisable until the date on which the rights separate and are distributed, and will expire on the final expiration date, which will be 10 years from the date that our rights agreement commences, unless we earlier exchange or redeem the rights.

Various additional features that will be contained in our rights agreement are described below.

"Flip-In" Event

If a flip-in event has occurred and if we have not redeemed, or exchanged the rights, each holder of a right, other than an acquiring person, will thereafter have the right to receive, upon exercise at the then current exercise price of the right, shares of our common stock or, under certain circumstances, an economically equivalent security or securities, having a market value of twice the exercise price. For example, if we assume that the initial exercise price of \$125 is in effect on the date that the flip-in event occurs, and that the market price per share of our common stock, as determined pursuant to the provisions of the rights agreement, were \$25, each right not owned beneficially by an acquiring person at or after the time of such an occurrence would entitle its holder after the separation date to purchase ten shares of our common stock having an aggregate market value of \$250 from us for a purchase price of \$125.

"Flip-Over" Event

If a flip-over event has occurred and if we have not redeemed, or exchanged the rights, each holder of a right will thereafter have the right to receive, upon the exercise at the then current exercise price of the right, that number of shares of common stock or, under certain circumstances, an economically equivalent security or securities, of the other person involved in the transaction that constitutes that flip-over event at the time of such transaction would have a market value of twice the exercise price of the right.

Exchange Feature

At any time after the later of the date on which the rights separate and are distributed and the first occurrence of a flip-in or flip-over event and before the acquisition by any person or group of affiliated or associated persons of 50% or more of the outstanding shares of our common stock, our board may exchange all or part of the rights, except for those held by such person or group or which have otherwise become void, in whole or in part, at an exchange ratio of one share of common stock per right, subject to adjustment.

Redemption of Rights

Our board will have the right to redeem all of the rights at a redemption price of \$0.01 per right, subject to adjustment, at any time before the close of business on the later of:

- . the date on which the rights separate and are distributed, and
- . the first date of public announcement that a person has become an acquiring person.

The board may extend the deadline for redeeming the rights if it does so before the deadline has passed. The board's right to redeem the rights also may be reinstated under certain circumstances described in the rights agreement. The right to exercise the rights will terminate upon redemption, and at such time, the holders of the rights will have the right to receive only the redemption price for each right held.

Adjustments

The exercise price payable and the number of shares of common stock or other securities issuable upon exercise of the rights will be subject to adjustment from time to time to prevent dilution:

- if there is a stock dividend on, or a subdivision, combination, or reclassification of, our common stock;
- . upon the grant to holders of our common stock of certain rights or warrants to subscribe for or purchase shares of our common stock at a price, or securities convertible into shares of our common stock with a conversion price, less than the then current market price of our common stock; or
- . upon the distribution to holders of our common stock of evidences of indebtedness, cash other than pursuant to regular periodic cash dividends, assets, stock other than pursuant to dividends payable in shares of our common stock, or of subscription rights or warrants.

Amendment of Rights

Our board will have the authority to amend the rights agreement without the approval of any holders of the rights, including amendments which add other events requiring adjustment to the exercise price payable and the number of shares of our common stock or other securities issuable upon exercise of the rights, or which modify procedures relating to redemption of the rights, provided that our board will not have the right to amend the rights agreement when the rights are not then redeemable, if the amendment were to decrease the stated exercise price or the period of time remaining until the final expiration date or modify a time period relating to when the rights may be redeemed.

Anti-Takeover Legislation--Georgia Law

We will elect to be covered by two provisions of the Georgia Business Corporation Code, or Georgia Code, that restrict business combinations with interested shareholders: the business combination provision and the fair price provision. These provisions do not apply to a Georgia corporation unless its bylaws specifically make the statute applicable, and once adopted, in addition to any other vote required by the corporation's articles of incorporation or bylaws to amend the bylaws, such a bylaw may be repealed only by the affirmative vote of at least two-thirds of the continuing directors and a majority of the votes entitled to be cast by the voting shares of such corporation, other than shares beneficially owned by an interested shareholder and, with respect to the fair price provision, his, her, or its associates and affiliates.

Interested Shareholders Transactions

The business combination provision of the Georgia Code generally prohibits Georgia corporations from entering into certain business combination transactions with any "interested shareholder," generally defined as any person other than the corporation or its subsidiaries beneficially owning at least 10% of the outstanding voting stock of the corporation, for a period of five years from the date that person became an interested shareholder, unless:

 prior to that shareholder becoming an interested shareholder, the board of directors of the corporation approved either the business combination or the transaction by which the shareholder became an interested shareholder;

- . in the transaction in which the shareholder became an interested shareholder, the interested shareholder became the beneficial owner of at least 90% of the voting stock outstanding, excluding, for purposes of determining the number of shares outstanding, "Insider Shares," as defined below, at the time the transaction commenced; or
- . subsequent to becoming an interested shareholder, such shareholder acquired additional shares resulting in the interested shareholder being the beneficial owner of at least 90% of the outstanding voting shares, excluding, for purposes of determining the number of shares outstanding, Insider Shares, and the transaction was approved at an annual or special meeting of shareholders by the holders of a majority of the voting stock entitled to vote thereon, excluding from such vote, Insider Shares and voting stock beneficially owned by the interested shareholder.

For purposes of this provision, Insider Shares refers generally to shares owned by:

- . persons who are directors or officers of the corporation, their affiliates, or associates;
- . subsidiaries of the corporation; and
- . any employee stock plan under which participants do not have the right, as determined exclusively by reference to the terms of such plan and any trust which is part of such plan, to determine confidentially the extent to which shares held under such plan will be tendered in a tender or exchange offer.

A Georgia corporation's bylaws must specify that all requirements of this provision apply to the corporation in order for this provision to apply. Our bylaws will contain a provision stating that all requirements of this provision, and any successor provision, apply to us.

Fair Price Requirements

The fair price provision of the Georgia Code imposes certain requirements on business combinations of a Georgia corporation with any person who is an "interested shareholder" of that corporation. In addition to any vote otherwise required by law or the corporation's articles of incorporation, under the fair price provision, business combinations with an interested shareholder must meet one of the three following criteria designed to protect a corporation's minority shareholders:

- . the transaction must be unanimously approved by the "continuing directors" of the corporation, generally directors who served prior to the time an interested shareholder acquired 10% ownership and who are unaffiliated with such interested shareholder, provided that the continuing directors constitute at least three members of the board of directors at the time of such approval;
- . the transaction must be recommended by at least two-thirds of the continuing directors and approved by a majority of the votes entitled to be cast by holders of voting shares, other than voting shares beneficially owned by the interested shareholder who is, or whose affiliate is, a party to the business combination; or
- . the terms of the transaction must meet specified fair pricing criteria and certain other tests.

A Georgia corporation's bylaws must specify that all requirements of the fair price provision apply to the corporation in order for the fair price provision to apply. Our bylaws will contain a provision stating that all requirements of the fair price provision, and any successor provisions thereto, apply to us.

Removal of Directors

The Georgia Code also contains a provision commonly referred to as the removal provision, which among other things, limits the ability of shareholders of a Georgia corporation to remove its directors if they serve staggered terms. The removal provision generally provides that:

 directors having staggered terms may be removed only for "cause," unless the corporation's articles of incorporation or a bylaw provision adopted by the corporation's shareholders provides otherwise;

- . directors may be removed only by a majority vote of the shares entitled to vote for the removal of directors; and
- . a director may be removed by a corporation's shareholders only at a meeting called for the purpose of removing him or her and the meeting notice must state that purpose, or one of the purposes, of the meeting is removal of the director.

Neither our articles of incorporation nor our bylaws will contain a provision permitting the removal of our directors other than for cause. Accordingly, the removal provision could have the effect of restricting the ability of our shareholders to remove incumbent directors and fill the vacancies created by such removal with their own nominees.

No Preemptive Rights

No holder of any class of stock authorized at the distribution date will have any preemptive right to subscribe to any kind or class of our securities.

Transfer Agent and Registrar

SunTrust Bank will be the transfer agent and registrar for our common stock.

INDEMNIFICATION OF DIRECTORS AND OFFICERS

Our bylaws will require us to indemnify and hold harmless any director or officer who was or is a party or is threatened to be made a party, to any threatened, pending, or completed action, suit or proceeding whether civil, criminal, administrative or investigative, including any action or suit by or in the right of our company, because the person is or was our director or officer against liability incurred in such proceeding. Our bylaws will generally prohibit us from indemnifying any officer or director who is adjudged liable to us or is subjected to injunctive relief in favor of us for:

- any appropriation, in violation of the director's or officer's duties, of any business opportunity;
- acts or omissions that involve intentional misconduct or a knowing violation of law;
- . unlawful corporate distributions as set forth in section 14-2-832 of the Georgia Code; or
- . any transactions from which the director derived an improper personal benefit.

We intend to purchase and maintain insurance on behalf of our officers and directors against liability asserted against or incurred by these persons in their capacity as an officer or director, or arising out of their status as an officer or director, regardless of whether we would have the power to indemnify or advance expenses to these persons against these liabilities under our bylaws or the Georgia Code.

Our bylaws will require us, under certain circumstances, to advance expenses to our officers and directors who are parties to an action, suit, or proceeding for which indemnification may be sought. Our bylaws will permit, but will not require, us to indemnify and advance expenses to our employees or agents who are not officers or directors to the same extent and subject to the same conditions that a corporation could, without shareholder approval under Section 14-2-856 of the Georgia Code.

Our articles of incorporation will eliminate the liability of our directors to us or our shareholders for monetary damages for any action taken, or any failure to take action, as a director to the extent permitted under the Georgia Code. Our directors will remain liable, however, for:

- any appropriation, in violation of the director's duties, of any business opportunity;
- acts or omissions that involve intentional misconduct or a knowing violation of law;
- . unlawful corporate distributions as set forth in section 14-2-832 of the Georgia Code: or
- any transactions from which the director received an improper personal benefit.

If the Georgia Code is amended to authorize corporate action further eliminating or limiting the personal liability of directors, the liability of our directors will be eliminated or limited to the fullest extent permitted by the Georgia Code, as amended, without further action by the shareholders. These provisions in our articles of incorporation may limit the remedies available to a shareholder in the event of breaches of any director's duties.

The distribution agreement will provide for indemnification by us of Equifax and its directors, officers, and employees for some liabilities, including liabilities under the Securities Act.

INDEPENDENT PUBLIC ACCOUNTANTS

The combined financial statements for us as of December 31, 2000 and December 31, 1999, and for each of the three years in the period ended December 31, 2000, appearing in this information statement have been audited by Arthur Andersen LLP, independent public accountants, as set forth in their report thereon appearing elsewhere herein, and are included in reliance upon said report given on the authority of such firm as experts in giving said reports.

WHERE YOU CAN OBTAIN ADDITIONAL INFORMATION

We have filed with the SEC a registration statement under the Exchange Act with respect to the shares of our common stock and the associated rights being issued in the distribution. This information statement does not contain all of the information set forth in the registration statement and the exhibits thereto, to which reference is hereby made. With respect to each contract, agreement, or other document filed as an exhibit to the registration statement, reference is made to such exhibit for a more complete description of the matter involved. The registration statement and the exhibits thereto filed by us with the SEC may be inspected at the public reference facilities of the SEC listed below.

After the distribution, we will be subject to the informational requirements of the Exchange Act, and in accordance therewith will file reports, proxy statements, and other information with the SEC. Such reports, proxy statements, and other information can be inspected and copied at the public reference facilities at its principal offices at Room 1024, 450 Fifth Street, N.W., Washington, D.C. 20549, as well as at the Regional Offices of the SEC maintained by the SEC at Seven World Trade Center, Thirteenth Floor, New York, New York 10048 and Citicorp Center, 500 West Madison Street, Suite 1400, Chicago, Illinois 60661. Copies of such information may be obtained from the Public Reference Section of the SEC at 450 Fifth Street, N.W., Washington, D.C. 20549 at prescribed rates. The Commission also maintains a World Wide Web site (http://www.sec.gov) that contains reports, proxy, and information statements and other information regarding registrants that file electronically with the SEC.

After the distribution, our shares will be listed on the New York Stock Exchange. When our shares commence trading on the New York Stock Exchange, such reports, proxy statements, and other information will be available for inspection at the offices of the New York Stock Exchange, 20 Broad Street, New York, New York 10005.

We intend to furnish holders of our common stock with annual reports containing combined financial statements audited by independent accountants, beginning with the fiscal year ending December 31, 2001.

No person is authorized to give any information or to make any representations other than those contained in this information statement, and, if given or made, such information or representations must not be relied upon as having been authorized. Neither the delivery of this information statement nor any distribution of securities made hereunder will imply that there has been no change in the information set forth herein or in our affairs since the date hereof.

INDEX TO FINANCIAL STATEMENTS

	Page
Equifax Payment Services division (To be reorganized as Certegy Inc.) Historical:	
Report of Independent Public Accountants	F-2
and 1998.`	F-3
31, 2000 and 1999	F-4
1999, and 1998	F-5
December 31, 2000, 1999, and 1998	F-6
Notes to Combined Financial Statements	F-7
Report of Independent Public Accountants as to Schedule	
Combined Schedule IIValuation and Qualifying AccountsPro Forma (Unaudited):	F-21
Introduction to the Pro Forma Combined Financial Statements Pro Forma Combined Statement of Income for the Three Months ended March	F-22
31, 2001Pro Forma Combined Statement of Income for the Year ended December 31,	F-23
2000Pro Forma Combined Balance Sheet as of March 31, 2001	F-25
Notes to Pro Forma Combined Financial Statements	F-26

REPORT OF INDEPENDENT PUBLIC ACCOUNTANTS

To Equifax Inc.:

We have audited the accompanying combined balance sheets of the Equifax Payment Services division (to be reorganized as Certegy Inc., a Georgia corporation--Note 1) as of December 31, 2000 and 1999 and the related combined statements of income, changes in shareholder's equity, and cash flows for each of the three years in the period ended December 31, 2000. These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with auditing standards generally accepted in the United States. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of the Equifax Payment Services division as of December 31, 2000 and 1999 and the results of its operations and its cash flows for each of the three years in the period ended December 31, 2000, in conformity with accounting principles generally accepted in the United States.

/s/ Arthur Andersen LLP

Atlanta, Georgia February 23, 2001

COMBINED STATEMENTS OF INCOME EQUIFAX PAYMENT SERVICES DIVISION

(To be reorganized as Certegy Inc.--Note 1) (In thousands, except per share amounts)

		ed 31,	Year End	ed Decembe	r 31,
		2000		1999	1998
	(Unaud:				
Revenues	\$194,976	\$177,561	•	\$681,172	\$566,120
Operating expenses: Costs of services Selling, general and	143,444	129,099	535,751		384,933
administrative	•	•	•	87,308	•
	168,486	153,651	631,403	553,687	462,829
Operating income Other income (expense),	26,490	23,910			
net	(316) (179)	(171) (146)	1,309 (1,301)	2,311 (901)	(383) (533)
Income before income taxes and minority interests Provision for income taxes Minority interests in	25,995 (10,138)	23,593 (9,235)			102,375 (40,505)
earnings, net of tax	(618)	514	(1,096)	6	(780)
Net income		\$ 14,872 =======		\$ 74,629 ======	
Basic earnings per share	\$ 0.22		\$ 1.32		\$ 0.86
Basic weighted average shares outstanding		66,959	67,200		70,699
Pro forma basic earnings per share (unaudited)	\$ 0.17 ======		\$ 1.06 ======		

COMBINED BALANCE SHEETS EQUIFAX PAYMENT SERVICES DIVISION

(To be reorganized as Certegy Inc.--Note 1) (In thousands)

		Decembe	
	March 31, 2001	2000	1999
	(Unaudited)		
	(Gildan Lead)		
ASSETS Current assets: Cash and cash equivalents Trade accounts receivable, net of allowance for doubtful accounts of \$2,142, \$2,148	\$ 24,771	\$ 29,794	\$ 33,617
and \$2,219, respectively Settlement receivables	83,800 74,948	99,472	102,901 67,963
Other receivables	6,845	48,173 7,706	3,603
Deferred income taxes (Note 6)	4,894	4,827	4,902
Other current assets	9,105	9,235	8,970
Total current assets Property and equipment, net (Note 2) Intangibles, net (Note 2) Other assets, net (Note 2)	204,363 32,038 174,707 85,389	199,207 32,806 184,612 85,820	221,956 36,910 148,823 87,566
Total Assets	\$496,497 ======	\$502,445 ======	\$495,255 ======
LIABILITIES AND SHAREHOLDER'S EQUITY Current liabilities:			
Trade accounts payable Settlement payables Notes payable Accrued salaries and bonuses Income taxes payable Other current liabilities	\$ 13,102 81,614 443 7,514 7,527 40,506		4,484 36,573
Total current liabilities Deferred income taxes (Note 6) Other long-term liabilities	150,706 11,329 1,240	11,390 1,280	204,122 14,305 1,448
Total liabilities			219,875
Commitments and contingencies (Note 9)			
Minority interests	5,924	5,906	3,890
Shareholder's equity: Equifax equity investment Cumulative translation adjustment	396,558 (69,260)	380,906 (57,288)	319,952 (48,462)
Total shareholder's equity	327,298	,	271,490
Total Liabilities and Shareholder's Equity	\$496,497 ======	\$502,445 ======	\$495,255

COMBINED STATEMENTS OF CASH FLOWS EQUIFAX PAYMENT SERVICES DIVISION

(To be reorganized as Certegy Inc.--Note 1) (In thousands)

	Three Mo Ended Ma	rch 31.	ths h 31, Year Ended December		
	2001	2000	2000	1999	1998
	(Unaud:				
Cash flows from operating					
activities:					
Net income	\$ 15,239	\$ 14,872	\$ 88,462	\$ 74,629	\$ 61,090
amortization and Gain from sale of	10,365	10,530	42,698	35,758	27,839
investments			(2,188)	(2,997)	
Minority interests in earnings	618	(514)	1,096	(6)	780
Changes in assets and liabilities, excluding effects of acquisitions: Accounts receivable,		, ,	·	, ,	
net	15,099	12,181	1,568	(13, 214)	(9,923)
payables	(11,556)	(12,448)	(1,240)	21,579	9,040
Settlement receivables and payables, net	(22,374)	9,511	(21,353)	25,020	(18,583)
Other current assets Deferred income taxes	131 (247)	7 	(335) (2,961)	321 3,921	(120) 1,643
Other long-term					
liabilities Other assets			(1,795)	1,324	1,213
Net cash provided by					
operating activities				146,220	
Cash flows from investing activities: Additions to property and					
equipment	(3,112)	(2,284)	(11,149)	(9,371)	(18,095)
Additions to other assets		(6,215)	(27,640)	(40,740)	(29,798)
Acquisitions, net of cash acquired Investments in uncombined		(12,502)	(46,257)	2,020	(137,301)
affiliates				(700)	(19,888)
Proceeds from sale of investments			6,850	17,857	
Net cash used in investing					
activities	(12,321)	(21,001)	(78,196)	(30,934)	(205,082)
Cash flows from financing activities: Net borrowings from					
(repayments to) Equifax Contributions from	2,271	(10,694)	(26,353)	(106,059)	136,676
minority interests Change in notes payable	 (79)	(227)	(466)	3,223 (871)	(130)
Net cash provided by (used in) financing activities	2,192	(10,921)	(26,819)		136,546
Effect of foreign currency exchange rates on cash	(2,142)	53	(2,592)	(752)	456
Net cash (used) provided	(5,023)	3,796	(3,823)	10,827	4,574
Cash and cash equivalents, beginning of period	29,794	33,617	33,617	22,790	18,216
Cash and cash equivalents,					
end of period	\$ 24,771	\$ 37,413	\$ 29,794	\$ 33,617	\$ 22,790

_____ ______

COMBINED STATEMENTS OF CHANGES IN SHAREHOLDER'S EQUITY EQUIFAX PAYMENT SERVICES DIVISION

(To be reorganized as Certegy Inc.--Note 1) (In thousands)

	Equifax Equity Investment	Income (Loss)	Equity	Income
Balance, December 31, 1997	\$ 151,671	\$ 552	\$ 152,223	
Net income Foreign currency translation	61,090		61,090	\$ 61,090
adjustment Net transactions with		(2,756)	(2,756)	(2,756)
Equifax	138,236		138,236	
Balance, December 31, 1998	350,997	(2,204)	348,793	\$ 58,334 ======
1999 changes: Net income Foreign currency	74,629		74,629	\$ 74,629
translation adjustment Net transactions with		(46,258)	(46,258)	(46,258)
Equifax	(105,674)		(105,674)	
Balance, December 31, 1999	319,952	(48,462)	271,490	\$ 28,371 ======
2000 changes: Net income Foreign currency translation	88,462		88,462	\$ 88,462
adjustment Net transactions with		(8,826)	(8,826)	(8,826)
Equifax	(27,508)		(27,508)	
Balance, December 31, 2000	380,906	(57, 288)	323,618	\$ 79,636 ======
2001 changes (unaudited): Net income Foreign currency	15, 239		15,239	\$ 15,239
translation adjustment Net transactions with		(11,972)	(11,972)	(11,972)
Equifax	413		413	
Balance, March 31, 2001 (unaudited)	\$ 396,558 ======	\$(69,260) =====	\$ 327,298 ======	\$ 3,267 ======

NOTES TO COMBINED FINANCIAL STATEMENTS

Note 1--Spin-off and Basis of Presentation

In October 2000, the Board of Directors of Equifax Inc. ("Equifax") announced its intent to spin-off the Payment Services division, subject to certain conditions, into a separate publicly traded company with its own management and Board of Directors (the "Distribution"). This Distribution is expected to occur during the third quarter of 2001 and will be accomplished by transferring the assets, liabilities, and stock of the businesses that comprise the Payment Services division to Certegy Inc. ("Certegy"), a recently formed holding company, and then distributing all of the shares of common stock of Certegy to Equifax's shareholders. Equifax shareholders will receive one share of Certegy common stock for every two shares of Equifax common stock held as of the date of distribution (the "Distribution Date"). After the Distribution, Certegy and Equifax will be two separate public companies. Certegy was incorporated on March 2, 2001, under the name Equifax PS, Inc., as a whollyowned subsidiary of Equifax. Certegy will not have any operations, assets, or liabilities until the contribution to Certegy by Equifax of the Payment Services division prior to the Distribution and effective as of June 30, 2001.

These combined financial statements include the accounts of the Equifax businesses that comprise its Payment Services division (collectively referred to as the "Company"). The Company provides credit and debit card processing and check risk management services to financial institutions and merchants throughout the world, through two segments, Card Services and Check Services. Card Services provides credit and debit card issuer services, merchant processing services, and e-banking services in the U.S., the U.K., Brazil, and Chile. Check Services provides check risk management services and related processing services in the U.S., the U.K., Canada, France, Ireland, Australia, and New Zealand.

Card issuer services enable banks, credit unions, retailers, and others to issue VISA and MasterCard credit and debit cards, private label cards, and other electronic payment cards. Merchant processing services enable retailers and other businesses to accept credit, debit, and other electronic payment cards from purchasers of their goods and services. E-banking services enable banks to provide electronic banking services to their customers, allowing them to compete for and retain customers more effectively and to generate non-interest fee income.

Check risk management services, which utilize the Company's proprietary check authorization systems and risk assessment decision platforms, enable retailers, hotels, automotive dealers, telecommunications companies, supermarkets, casinos, mail order houses, and other businesses to minimize losses from dishonored checks, maximize check acceptance, and improve customer service. The services include check guarantee, where the Company accepts the risk of bad checks presented to customers, verification services, where the Company determines the likelihood that a check will clear and the customer retains the risk, and certain combinations of guarantee and verification services. Check Services also provides related service offerings, including risk management consulting and marketing services, which enable retailers to cross-sell and increase their customer retention.

The combined financial statements have been prepared on the historical cost basis in accordance with accounting principles generally accepted in the U.S., and present the Company's financial position, results of operations, and cash flows as derived from Equifax's historical financial statements. Intercompany transactions between entities included in the combined financial statements have been eliminated. As further described in Note 5, certain Equifax corporate expenses have been allocated to the Company based on an estimate of the proportion of corporate expenses allocable to the Company, utilizing such factors as revenues, number of employees, and other relevant factors. In the opinion of management, these allocations have been made on a reasonable basis. Management believes that, had the Company been operating on a stand-alone basis, the Company would have incurred additional annual expenses of approximately \$6.5 million, which specifically relate to incremental pension expense, insurance costs, corporate headquarters rent, and stand-alone public company costs for audit, director, and stock exchange fees. The increase in pension expense is a result of the

Company no longer benefitting from the over-funded status of the consolidated Equifax pension plan. Management believes that all other costs allocated to the Company are a reasonable representation of the costs that would have been incurred if the Company had performed these functions as a stand-alone company. The combined financial statements do not include any allocation of Equifax corporate debt or related interest expense as these amounts have not been historically allocated to the Company.

In conjunction with the separation of their businesses, the Company and Equifax will enter into various agreements that address the allocation of assets and liabilities between them and that define their relationship after the separation, including the distribution agreement, the tax sharing and indemnification agreement, the employee benefits agreement, the intercompany data purchase agreement, the intellectual property agreement, and the transition support agreement.

Note 2--Significant Accounting Policies

Use of estimates--The preparation of financial statements in conformity with accounting principles generally accepted in the U.S. requires management to make estimates and assumptions. These estimates and assumptions affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements as well as reported amounts of revenues and expenses during the reporting period. Actual results could differ from these estimates.

Revenue recognition--Revenues from credit and debit card processing and related services are recognized based on a specified amount per account, per card, or per transaction when processed or as services are rendered. Merchant processing provides a range of processing services, including authorizing card transactions at the point-of-sale, capturing and transmitting data affecting settlement of payments, and assisting merchants in resolving billing disputes with their customers. Revenues for these services are recognized in the period the transactions are processed or when the services are performed and include interchange fees, which are processing fees paid to credit card associations.

Check guarantee is the process of electronically authorizing a check being presented to the Company's merchant customer, through an extensive database, and guaranteeing the face value of the check to the merchant customer. If a guaranteed check is dishonored, the Company reimburses the merchant for the check's face value, and pursues collection from the delinquent checkwriter. Revenues for check guarantee services are based on a percentage of the face value of each guaranteed check and are recognized when the obligations to the customer are fulfilled. At the time checks are guaranteed, the Company records a valuation allowance for losses on uncollectible checks based on historical loss experience. Check verification services are similar to check guarantee services, except the Company does not guarantee the verified checks, and the risk of loss is retained by the customer. Revenues for these services are based on a fixed amount per check and are recognized when the checks are verified.

The Company licenses software products that allow customers to manage their credit card programs. These products include a complete suite of UNIX and mainframe credit card issuing and acquiring software. Software license revenues are recognized in accordance with Statement of Positions 97-2, "Software Revenue Recognition." In certain software arrangements, the Company provides consulting services, which include implementation and upgrades to the existing base software. For license sales that do not include consulting services, revenue is recognized when delivery has occurred, the license fee is fixed and determinable, collectibility is probable, and evidence of an arrangement exists. For professional services related to software and for licenses that include consulting or processing services, revenue is recognized over the period the services are performed. Software maintenance and support revenues are recognized over the term of the contract or as services are performed. Software licensing and related revenue totaled \$13.0 million, \$23.2 million, and \$27.1 million in 2000, 1999, and 1998, respectively.

Earnings per share--Basic earnings per share ("EPS") is calculated as net income divided by the weighted average number of common shares. Weighted average shares outstanding is computed by applying

the distribution ratio of 0.5 shares of Certegy common stock to the historical Equifax weighted average shares outstanding for the same periods presented.

Diluted EPS is calculated to reflect the potential dilution that would occur if stock options or other contracts to issue common stock were exercised and resulted in additional common shares outstanding. Diluted EPS is not presented in these financial statements, as there are no historical market share prices for Certegy common stock, as public trading will not commence until the Distribution occurs. Accordingly, the dilutive effect of stock options cannot be determined. At December 31, 2000, the number of shares of Equifax common stock subject to options held by option holders expected to become Certegy employees was 2,369,000. The exercise prices of those options range from \$7.09 to \$38.60.

Pro forma earnings per share--Pro forma EPS is computed by dividing pro forma net income by the weighted average number of common shares. Pro forma net income includes adjustments assuming that the Distribution had taken place at the beginning of 2000. The following table provides a reconciliation of net income to pro forma net income:

	Three Months Ended March 31, 2001	Year Ended December 31, 2000
	(In th	nousands)
Net income	\$15,239 (1,625) (4,609) 2,431	\$ 88,462 (6,500) (21,735) 11,040
Pro forma net income (unaudited)	\$11,436 ======	\$ 71,267 ======
Pro forma basic earnings per share (unaudited)	\$ 0.17 ======	\$ 1.06 ======

Settlement receivables and payables--Settlement receivables and payables result from timing differences in the Company's settlement process with merchants, financial institutions, and credit card associations related to merchant and card processing. Cash held by Equifax associated with this settlement process was \$29.0 million, \$50.4 million, and \$25.4 million at December 31, 2000, 1999, and 1998, respectively. These amounts are included in the intercompany receivables due from Equifax, which are a component of the Equifax equity investment.

Property and equipment--The cost of property and equipment is depreciated on a straight-line basis over estimated useful lives as follows: buildings--30 years; leasehold improvements--not to exceed lease terms; data processing equipment--3 to 5 years; and furniture--3 to 10 years.

	2000	1999
	(In thou	sands)
Land, buildings and improvements Data processing equipment and furniture	•	\$ 8,065 76,249
Less accumulated depreciation	,	84,314 (47,404)
	\$ 32,806 ======	\$ 36,910 ======

Intangibles--Intangibles include goodwill and purchased merchant contracts. Goodwill of \$160,945,000 (net) and \$148,823,000 (net) at December 31, 2000 and 1999, respectively, is amortized using the straight-line method over estimated useful lives of 20 to 40 years, with a weighted average life of 25 years. Purchased merchant contracts of \$23,667,000 (net) at December 31, 2000, are amortized using the straight-line method

over an estimated useful life of 11 years. Useful lives are principally determined by management based on the nature and geographic location of the business acquired, and the relative stability and rate of technological change inherent in each business. Goodwill amortization expense was \$7,939,000 in 2000, \$7,216,000 in 1999, and \$3,881,000 in 1998. As of December 31, 2000 and 1999, accumulated goodwill amortization was \$23,845,000 and \$17,043,000, respectively. Purchased merchant contract amortization expense was \$333,000 in 2000. As of December 31, 2000, accumulated purchased merchant contract amortization was \$333,000.

Other assets--Other assets principally consist of systems development and other deferred costs, and purchased software. The costs of internally developed and purchased software used to provide services to customers or internal administrative services are capitalized and amortized on a straight-line basis over five to ten years, as determined by their estimated useful lives. Maintenance and repairs are expensed as incurred. Other assets are amortized using the straight-line method over estimated useful lives of five to ten years. Amortization expense for other assets was \$20,566,000 in 2000, \$15,418,000 in 1999, and \$8,574,000 in 1998. As of December 31, 2000 and 1999, accumulated amortization was \$57,075,000 and \$37,507,000, respectively.

Other assets, net at December 31, 2000 and 1999 consist of the following:

	2000	1999
	(In thou	ısands)
Systems development and other deferred costs Purchased software Investments in uncombined affiliates Other	7,809 350	10,502
	\$85,820	\$87,566
	======	======

Impairment of long-lived assets--The Company regularly evaluates whether events and circumstances have occurred which indicate that the carrying amount of property and equipment, intangibles, or other assets may warrant revision or may not be recoverable. When factors indicate that assets should be evaluated for possible impairment, the Company uses an estimate of the future undiscounted net cash flows of the related business over the remaining life of the asset in measuring whether the carrying amount is recoverable. To the extent these projections indicate that future undiscounted cash flows are not sufficient to recover the carrying amounts of the related assets, the underlying assets are written down by charges to expense so that the carrying amount is equal to fair value, primarily determined based on future discounted cash flows.

Minority interests--Minority interests in earnings of combined subsidiaries represent the minority shareholders' share of the after-tax net income or loss of various combined subsidiaries. The minority interests in the combined balance sheets reflect the original investments by these minority shareholders in the combined subsidiaries, along with their proportional share of the earnings or losses of the subsidiaries, net of dividends.

Foreign currency translation--The Company has foreign subsidiaries whose functional currency is their local currency. Gains and losses on transactions denominated in currencies other than the functional currencies are included in determining net income for the period in which exchange rates change. The assets and liabilities of foreign subsidiaries, including long-term intercompany balances, are translated at the year-end rate of exchange, and income statement items are translated at the average rates prevailing during the year. The resulting translation adjustment is recorded as a component of shareholder's equity. Other foreign currency transaction gains and losses are recorded in other income.

Combined statements of cash flows--The Company considers cash equivalents to be short-term cash investments with original maturities of three months or less.

	2000	1999	1998
	(In	thousan	ds)
Income taxes, net of amounts refunded	. ,	. ,	. ,

Cash paid for income taxes represents payments for foreign and certain state income taxes. Payments for federal and unitary state income taxes are reflected as a component of net transactions with Equifax in the combined financial statements.

Financial instruments--The Company's financial instruments consist primarily of cash and cash equivalents, accounts receivable, and accounts and notes payable. The carrying amounts of these items approximate their fair market values due to their short maturity. During 2000, the Company did not hold any material derivative financial instruments.

Recent Accounting Pronouncements and Accounting Change--In June 1998, the FASB issued Statement of Financial Accounting Standards No. 133 (SFAS 133), "Accounting for Derivative Instruments and Hedging Activities." SFAS 133 establishes accounting and reporting standards for derivative instruments and hedging activities and is effective (as amended by SFAS No. 137) on January 1, 2001 for the Company. Based on its current level of derivative instruments and hedging activities, the adoption of SFAS 133 did not have a significant impact on its financial statements or reported earnings.

Unaudited interim financial information--The accompanying interim combined financial statements are unaudited and have been prepared by the Company in accordance with accounting principles generally accepted in the U.S. In the opinion of management of the Company, these combined financial statements contain all adjustments (consisting only of normal recurring adjustments) which are necessary for a fair presentation of the interim periods. Results of operations for interim periods are not necessarily indicative of results of operations for the entire year.

Note 3--Acquisitions and Investments in Uncombined Affiliates

During 2000 and 1998, the Company acquired, made equity investments, or increased its ownership in the following businesses:

Business	Date Acquired	Industry Segment	Percentage Ownership
Equifax Card Solutions Limited			
(U.K.)	September 2000	Card Solutions	100.0%(1)
Check-A-Cheque Ltd. (U.K.)	March 2000	Check Solutions	100.0%
Rexora (U.K.)	January 2000	Check Solutions	100.0%
Procard S.A. (Chile)	January 2000	Card Solutions	100.0%
Unnisa Ltda. (Brazil)	September 1998	Card Solutions	59.3%
Proceda S.A. (Brazil)	September 1998	Card Solutions	34.0%(2)

- (1) Increased from 51.0% ownership when started in 1999 to 100% in 2000.
- (2) Subsequently sold in April 1999.

In 2000, the Company increased its ownership in Equifax Card Solutions Limited, a U.K. card processing business, from 51% to 100%, and acquired Check-A-Cheque Ltd. and Rexora to compliment its check business in Europe, and Procard S.A., a card processing business in Chile. The Company also acquired a portfolio of credit card merchant processing accounts from Heartland Payment Systems. These acquisitions had an aggregate cash purchase price of \$46.3 million, with \$24.7 million allocated to goodwill, \$22.0 million allocated to merchant contracts, \$1.8 million allocated to assets, and \$2.2 million of assumed liabilities. If these 2000 acquisitions had occurred as of the beginning of 1999, the unaudited pro forma revenues and net income of the Company would have been \$830.1 million and \$88.8 million in 2000 and \$748.3 and \$74.6 million in 1999, respectively.

In 1998, the Company acquired a 59.3% interest in Unnisa Ltda. ("Unnisa"), a card processing business in Brazil (Note 9). The total purchase price was \$138.0 million, with \$134.9 million allocated to goodwill, \$16.7 million allocated to assets (including \$0.7 million of cash), and \$13.6 million of assumed liabilities. Additionally, the Company acquired a minority interest in Proceda S.A. ("Proceda") (Note 4) totaling \$16.8 million and invested an additional \$3.1 million in its card processing operation in India, which were accounted for under the equity method. If these 1998 acquisitions had occurred as of the beginning of 1998, the unaudited pro forma revenues and net income of the Company would have been \$612.6 million and \$58.1 million in 1998, respectively.

The above acquisitions were accounted for as purchases and their results of operations have been included in the combined statements of income from the dates of acquisition.

Note 4--Divestitures

In September 2000, the Company sold its 50% interest in its card processing operation in India for \$6,850,000, which resulted in a pre-tax gain of \$2,188,000. In April 1999, the Company sold its 34% equity interest in Proceda in Brazil for \$17,857,000, which resulted in a pre-tax gain of \$2,997,000. These items were recorded in other income.

Note 5--Related Party Transactions

There are no material intercompany purchase or sales transactions between Equifax and the Company. Under Equifax's centralized cash management system, excess cash sent to Equifax and short-term advances from Equifax are reflected as intercompany receivables or payables and are included in the Equifax equity investment in the accompanying combined balance sheets. Intercompany receivables from Equifax at December 31, 2000 and 1999 were \$117,459,000 and \$89,950,000, respectively (Note 7). The average balance of the intercompany receivables was \$103,705,000, \$37,114,000, and \$53,394,000 in 2000, 1999, and 1998, respectively. The intercompany receivables (payables) include the following activity:

	2000	1999	1998
	(In	thousands)
Beginning balance	(11,148)	(10,086)	(8,852)
Ending balance	\$117,459 ======	\$ 89,950 ======	\$ (15,724) ======

The Company was charged with Equifax corporate costs in the amount of \$11,148,000 in 2000, \$10,086,000 in 1999, and \$8,852,000 in 1998. These allocations were based on an estimate of the proportion of corporate expenses related to the Company, utilizing such factors as revenues, number of employees, number of transactions processed, and other applicable factors. In the opinion of management, these costs have been allocated on a reasonable basis. Approximately \$3,345,000, \$2,762,000, and \$2,031,000 of these amounts were allocated to the Company's two operating segments in 2000, 1999, and 1998, respectively.

Note 6--Income Taxes

Historically, the Company has been included in the consolidated federal income tax return of Equifax. The Company's provision for income taxes in the accompanying combined statements of income reflects federal, state, and foreign income taxes calculated using the separate return basis. The Company records deferred income taxes using enacted tax laws and rates for the years in which the taxes are expected to be paid.

Deferred income tax assets and liabilities are recorded based on the differences between the financial reporting and income tax bases of assets and liabilities. The provision for income taxes consists of the following:

	2000	1999	1998
	(In	thousands	3)
Current: Federal State Foreign	8,095 3,343	\$39,321 6,532 1,243	4,042 2,411
	59,346	47,096	36,121
Deferred: FederalStateForeign	(668) (160) (909)	,	
	(1,737)	7,176	4,384
	\$57,609	\$54,272	\$40,505
	======	======	======

	2000	1999	1998
	(I	n thousand	ds)
United States	•	\$121,683 7,212	•
	\$147,167 ======	\$128,895 ======	\$102,375 ======

The provision for income taxes is reconciled with the federal statutory rate as follows:

	2000	1999	1998
	(In 1	housands)
Federal statutory rate	35.0%	35.0%	35.0%
Provision computed at federal statutory rate State and local taxes, net of federal tax benefit Other	,	\$45,114 4,958 4,200	\$35,831 3,033 1,641
	\$57,609 =====	\$54,272 ======	\$40,505 =====

Components of the Company's deferred income tax assets and liabilities at December 31, 2000 and 1999 are as follows:

	2000	1999	
	(In thou	sands)	
Deferred income tax assets: Reserves and accrued expenses Net operating loss carryforwards Other	4,082	•	
	10,841	9,707	

Deferred income tax liabilities:		
Other assets	(14,418)	(16,390)
Depreciation		
Undistributed earnings of foreign subsidiaries	(1,271)	(1,271)
Other	(766)	(22)
	(17,404)	(19, 110)

	2000	1999
	(In thou	sands)
Net deferred income tax liability	(6,563)	(9,403)
Less: Current deferred tax asset	4,827	4,902
Non-current deferred tax liability	\$(11,390) ======	\$(14,305) ======

Note 7--Shareholder's Equity

Equifax equity investment--Equifax's equity investment includes the original investments in the Company, accumulated income of the Company, and the net intercompany receivable due from Equifax reflecting transactions described in Note 5.

In connection with the Distribution, the net intercompany receivable due from Equifax will be capitalized. In addition, the Company will enter into an unsecured revolving credit facility that will be partially used to fund a cash payment to Equifax on or before the date of the Distribution of \$275 million.

Stock incentive plans--The Company intends to adopt the Certegy Inc. 2001 Stock Incentive Plan (the "Omnibus Plan"). The Omnibus Plan will authorize grants of stock options, stock appreciation rights, restricted stock, deferred stock, and performance shares or units. Additionally, the Company intends to adopt the Certegy Inc. 2001 Non-Employee Director Stock Option Plan, pursuant to which options to purchase Certegy common stock will be available for grant to non-employee directors.

Historically, the Company participated in the Equifax Stock Option Plans (the "Stock Option Plans"), which provide qualified and non-qualified stock options to officers and employees of Equifax at exercise prices not less than market value on the date of grant. Generally, options vest proportionately over a four-year period and are exercisable for ten years from the grant date. Certain of the Stock Option Plans also provide for awards of restricted shares of Equifax's common stock.

Pursuant to the employee benefits agreement, Equifax stock options held by the Company's employees will be converted to Certegy options at the time of the Distribution. In accordance with the provisions of EITF 90-9, the Company will multiply the number of shares purchasable under each converted stock option by a ratio determined at the time of the Distribution and divide the exercise price per share of each option by the same ratio. Fractional shares will be rounded down to the nearest whole number of shares. In accordance with the provisions of EITF 90-9, all other terms of the converted stock options will remain the same as those in effect immediately prior to the Distribution. Accordingly, no compensation expense will result from the replacement of the options. At December 31, 2000, the number of shares of Equifax common stock subject to options held by option holders expected to become Certegy employees was 2,369,000. The exercise prices of such options range from \$7.09 to \$38.60. The ultimate number of stock options to be held by Certegy employees and the number and exercise prices of Certegy stock options to be issued subject to the above calculation, cannot yet be determined.

Rights agreement--Certegy anticipates its Board of Directors will adopt a Rights Agreement (the "Rights Agreement") on or prior to the Distribution. If adopted, the Rights Agreement will contain provisions to protect Certegy shareholders in the event of coercive, unfair or inadequate takeover bids and practices that are not approved by the Certegy Board of Directors. Pursuant to the Rights Agreement, if adopted, Certegy would issue one Common Stock Purchase Right (a "Right") for each share of Certegy common stock providing the right to purchase one share of common stock. The Rights will be represented by, and will trade together with, the Certegy common stock. The Rights will not be immediately exercisable and will not become exercisable unless certain events occur. Among the events is the acquisition of 15% or more of Certegy's common stock by a person or group of affiliated or associated persons ("Acquiror"). In addition, if certain triggering events occur, including acquisition of 20% or more of Certegy's common stock by an Acquiror, and if the Rights have not previously been redeemed, each Right that is not held by Acquiror will entitle its holder to purchase shares of common stock at a discounted price.

Common and preferred stock--Certegy will have 300 million shares of common stock, par value \$0.01 per share, and 100 million shares of preferred stock, par value \$0.01 per share (the "Preferred Stock"), authorized as of the Distribution Date. Approximately 68.1 million shares of common stock and no shares of Preferred Stock are expected to be issued as of the Distribution Date.

Note 8--Employee Benefits

Historically, the Company participated in the Equifax employee benefit plans and was allocated a portion of the plans' costs based on an estimate of the proportion of expense related to the Company. In the opinion of management, the expenses have been allocated on a reasonable basis. Upon the Distribution, the Company will provide benefits to its employees, which will generally be similar to those benefits provided under Equifax's plans.

U.S. Retirement Plan--The Equifax noncontributory qualified retirement plan (the "Retirement Plan") covers most U.S. salaried employees. Benefits are primarily a function of salary and years of service. Retirement Plan provisions and funding meet the requirements of the Employee Retirement Income Security Act of 1974, as amended. Total pension income (expense) allocated to the Company and included in the accompanying combined statements of income was \$3,347,000 in 2000, \$3,397,000 in 1999, and \$(634,000) in 1998. The components of pension income (expense) other than service costs, which are allocated directly, are allocated to the businesses comprising the Payment Services division in proportion to total payroll costs. Information relating to accumulated benefits and plan assets as they may be allocable to the Company's participants at December 31, 2000 and 1999 is not available.

Employee Retirement Savings Plan--The Equifax retirement savings plan provides for annual contributions within specified ranges, determined at the discretion of the Board of Directors, for the benefit of eligible employees in the form of cash or shares of Equifax's common stock. The Company's portion of the plan's expenses was \$1,115,000, \$1,820,000, and \$928,000 in 2000, 1999, and 1998, respectively. Expenses for this plan are a direct function of the contributions made by the employees of the Company.

Postretirement Benefits--The Equifax unfunded healthcare and life insurance benefit plans cover eligible retired employees. Substantially all U.S. employees may become eligible for these benefits if they reach normal retirement age while working for Equifax and satisfy certain years of service requirements. Equifax accrues the cost of providing these benefits over the active service period of the employee. Expenses of \$535,000 in 2000, \$595,000 in 1999, and \$553,000 in 1998 were allocated to the Company in proportion to total payroll costs.

Note 9--Commitments and Contingencies

Leases--The Company's operating leases involve principally office space and office equipment. Rental expense relating to these leases was \$12,601,000 in 2000, \$11,275,000 in 1999, and \$9,900,000 in 1998.

Future minimum payment obligations for noncancelable operating leases exceeding one year are as follows as of December 31, 2000:

	Amount
	(In thousands)
2004	ф O 770
2001	
2002	5,437
2003	4,296
2004	3,875
2005	3,547
Thereafter	12,016
	\$37,943
	======

Data processing services agreements--The Company has separate agreements with EDS and IBM, which expire between 2009 and 2010, for portions of its computer data processing operations and related functions.

The estimated aggregate contractual obligation remaining under these agreements is approximately \$286 million as of December 31, 2000. However, these amounts could be more or less depending on various factors such as the inflation rate, the introduction of significant new technologies, or changes in the Company's data processing needs as a result of acquisitions or divestitures. Under certain circumstances, such as a change in control of the Company or for the Company's convenience, the Company may terminate these agreements. However, the agreements provide that the Company must pay a termination charge in the event of such a termination.

Change in control agreements--Certegy intends to enter into change in control agreements with certain of its officers prior to the Distribution, which provide certain severance pay and benefits in the event of a termination of the officer's employment under certain circumstances following a "change in control" of Certegy. "Change in control" is defined as the accumulation by any person, entity, or group of 20% or more of the combined voting power of Certegy voting stock or the occurrence of certain other specified events.

Litigation--A number of lawsuits seeking damages are brought against the Company each year in the ordinary course of business. The Company provides for estimated legal fees and settlements relating to pending lawsuits. In addition, a class action lawsuit is pending against Equifax Check Services, Inc., a subsidiary of Equifax that will be contributed to the Company immediately prior to the Distribution, in the U.S. District Court for the Eastern District of California. This action is based on a claim that the Company's practice of assessing a service charge on unpaid checks allegedly violated provisions of the federal Fair Debt Collection Practices Act and California's Unfair Business Practices Act during the period from August 1992 through December 1996. The plaintiffs seek, among other remedies, a refund of all service charges collected from California consumers during this period, prejudgment interest, statutory damages under the Fair Debt Collection Practices Act, and attorneys' fees, which amounts in the aggregate, could exceed \$15 million. The Company has defended, and will continue to defend this action vigorously, however, litigation is inherently uncertain and the Company may not prevail. In the opinion of management, the ultimate resolution of these matters will not have a materially adverse effect on the Company's financial position, liquidity, or results of operations.

Unnisa put option--Under the terms of an agreement, the minority shareholders of Unnisa, the Company's credit card processing business in Brazil, exercised an option to sell their 40.7% to the Company. The Company purchased this interest in May 2001 for approximately \$55 million.

Note 10--Quarterly Combined Financial Data (Unaudited)

Quarterly operating revenues and operating income by reportable segment (Note 11) and other summarized quarterly financial data for 2000 and 1999 are as follows (in thousands, except per share amounts):

2000	First	Second	Third	Fourth
Operating revenues: Card Services(2)		\$130,157 62,951	\$131,602 63,074	\$137,507 75,710
	\$177,561	\$193,108 	\$194,676	\$213,217
Operating income: Card Services(2) Check Services		10,667	\$ 32,487 10,765	14,088
General Corporate Expense	•	38,583 (2,034)	,	•
	\$ 23,910	. ,	\$ 41,434 ======	/
Net income		\$ 22,468	\$ 24,931	\$ 26,191
Per common share (basic): Net income(1)				

1999	First	Second	Third	Fourth
Operating revenues: Card Services(3)		\$107,329 56,273	\$115,390 59,695	\$120,555 70,801
	\$151,129	\$163,602	\$175,085	\$191,356
	======	=======	======	======
Operating income: Card Services(3)		8,948	\$ 24,319 10,551	12,979
General Corporate Expense	,	30,767 (1,932)	,	,
	\$ 26,746 ======	\$ 28,835 ======	\$ 33,096 ======	\$ 38,808 ======
Net income	\$ 15,020 ======	\$ 18,157 ======	\$ 18,106 ======	\$ 23,346 ======
Per common share (basic): Net income(1)	\$ 0.22 ======	\$ 0.26 =====	\$ 0.27 ======	\$ 0.34 ======

⁽¹⁾ Weighted average shares outstanding is computed by applying the distribution ratio of one share of Certegy common stock for every two shares of Equifax common stock held to the historical Equifax weighted average shares outstanding for all periods presented.

(2) The second quarter of 2000 includes revenues and operating income of

approximately \$4.9 million from software license sales.

⁽³⁾ The first quarter of 1999 includes revenues and operating income of approximately \$7.0 million from software license sales.

Segment information has been prepared in accordance with Statement of Financial Accounting Standards No. 131 (SFAS 131), "Disclosures About Segments of an Enterprise and Related Information." The Company has two segments, credit and debit card processing (Card Services) and check risk management services (Check Services). Segments were determined based on products and services provided by each segment (Note 1) and represent components of the Company about which separate internal financial information is maintained and evaluated by senior management in deciding how to allocate resources and in assessing performance. The accounting policies of the segments are the same as those described in the Company's summary of significant accounting policies (Note 2). The Company evaluates the segment performance based on its operating income. Intersegment sales and transfers, which are not material, have been eliminated.

	2000 1999		1998			
	Amount	%		t %	Amou	nt %
Operating revenues: Card Services		33	237,2	68 3	5% \$357, 5 209,	106 37
	\$778,562 ======	100%	\$681,1	72 10	0% \$566, = =====	120 100%
Operating income: Card Services	\$110,757	71% 29	\$ 96,3 38,4	68 7 41 2	1% \$ 79,; 9 30,;	209 72% 903 28
General Corporate Expense	154,962 (7,803)	100%	134,8	09 10	0% 110,; (6,	112 100%
	\$147,159 ======		\$127,4 =====	85	\$103,	291
Total assets at December 31: Card Services	\$419,270	83% 17	\$414,7 80,4	86 8 69 1	4% \$416,	450 85% 254 15
	\$502,445 ======	100%	\$495,2	55 10	0% \$492, = =====	704 100%
					1999	
Depreciation and amortization: Card Services				6,660	\$28,493 7,265	6,956
					\$35,758 ======	
Capital expenditures (excluding pro and other assets acquired in acqui Card Services	sitions):			35 478	\$47,502	\$37 <u>0</u> 53
Check Services					2,609	
			\$	38,789	\$50,111 ======	\$47,893

Financial information by geographic area is as follows (dollars in thousands):

	2000		0 1999		2000 1999 19		1998	
	Amount %		Amount	%	Amount	%		
Operating revenues (based on location of customer):								
United States	\$615,221	79%	\$548,192	80%	\$476,225	84%		
United Kingdom	62,512	8	47, 189	7	33,493	6		
Brazil	66,483	9	54,904	8	28,625	5		
Other	34, 346		30,887			5		
	\$778,562	100%	\$681,172	100%	\$566,120	100%		
	======	===	======	===	======	===		
Long-lived assets at December 31:								
United States	\$142,696	47%	\$134,455	49%	\$132,468	42%		
United Kingdom	50,351	17	37,960	14	26,492	8		
Brazil	87,963	29	94,207	35	153,132	48		
Other	22,228		,	2	7,791	2		
	\$303,238	100%	\$273,299	100%	\$319,883	100%		
	======	===	======	===	======	===		

Revenues from external customers by the Company's product and service offerings are as follows (dollars in thousands):

	2000		1999		1998	
	Amount	% 	Amount	% 	Amount	%
Card Issuer Services	,		•		,	
Merchant Processing Services	149, 276	19	111,863	16	80,840 27,012	14
vener						
	\$778,562 ======	100%	\$681,172 ======	100%	\$566,120 ======	100%

REPORT OF INDEPENDENT PUBLIC ACCOUNTANTS AS TO SCHEDULE

We have audited in accordance with auditing standards generally accepted in the United States, the combined financial statements of the Equifax Payment Services division (to be reorganized as Certegy Inc., a Georgia corporation) included in this information statement on Form 10, and have issued our report thereon dated February 23, 2001. Our audit was made for the purpose of forming an opinion on those statements taken as a whole. The schedule listed in the index on page F-1 is the responsibility of the Company's management and is presented for purposes of complying with the Securities and Exchange Commission's rules and is not part of the basic financial statements. This schedule has been subjected to the auditing procedures applied in the audit of the basic financial statements and, in our opinion, fairly states in all material respects, the financial data required to be set forth therein in relation to the basic financial statements taken as a whole.

/s/ Arthur Andersen LLP

Atlanta, Georgia February 23, 2001

(To be reorganized as Certegy Inc.)

COMBINED SCHEDULE II Valuation and Qualifying Accounts (In thousands)

Column A	Column B	Column C	Column D	Column E
Description	Balance at Beginning of Period	Charged to Costs	Accounts	Balance at End of Period
Trade Receivable				
Allowances	Ф БОБ	ф 4 CEO	Ф 650	#4 FOF
December 31, 1998	\$ 595	\$ 1,658	\$ 658	\$1,595
December 31, 1999	1,595	2,397	1,773	2,219
December 31, 2000	2,219	1,310	1,381	2,148
Allowances for Card				
transaction losses and				
Check claim losses (1)				
December 31, 1998	\$4,612	\$80,029	\$76,715	\$7,926
December 31, 1999	7,926	86,104	88,650	5,380
December 31, 2000	5,380	104,758	107,434	2,704
December 31, 2000	5,300	104,756	101,434	2,704

- ------

⁽¹⁾ Included in other current liabilities.

(To be reorganized as Certegy Inc.)
Introduction to the Pro Forma Combined Financial Statements
(Unaudited)

In October 2000, the Board of Directors of Equifax announced its intent to spin-off the Payment Services division, subject to certain conditions, into a separate publicly traded company with its own management and Board of Directors (the "Distribution"). This Distribution is expected to occur during the third quarter of 2001 and will be accomplished by transferring the assets, liabilities, and stock of the businesses that comprise the Payment Services division to Certegy Inc. ("Certegy"), a recently formed holding company, and then distributing all of the shares of common stock of Certegy to Equifax's shareholders. Equifax shareholders will receive one share of Certegy common stock for every two shares of Equifax common stock held as of the date of distribution. After the Distribution, Certegy and Equifax will be two separate public companies. Certegy was incorporated on March 2, 2001 under the name Equifax PS, Inc., as a wholly-owned subsidiary of Equifax. Certegy will not have any operations, assets, or liabilities until the contribution to Certegy by Equifax of the Payment Services division prior to the Distribution and effective as of June 30, 2001.

The following pro forma combined financial statements have been prepared as if the distribution had taken place on March 31, 2001 for the pro forma combined balance sheet and January 1, 2000 for the pro forma combined statements of income.

The unaudited pro forma combined financial statements are not necessarily indicative of the results had the distribution occurred on the dates indicated or the expected financial position or results of operations in the future. The unaudited pro forma combined financial statements should be read in conjunction with the separate historical financial statements of the Company, including the notes to those statements contained elsewhere herein, and in conjunction with the related notes to these unaudited pro forma combined financial statements.

(To be reorganized as Certegy Inc.)

PRO FORMA COMBINED STATEMENT OF INCOME FOR THE THREE MONTHS ENDED MARCH 31, 2001 Unaudited

(In thousands, except per share amounts)

		Pro Forma Adjustments	Pro Forma Combined
Revenues	\$194,976 	\$ 	\$194,976
Operating expenses: Costs of services Selling, general and administrative	143,444 25,042	1,175 (a) 450 (a)	25,492
		1,625	170,111
Operating income	(316)	(1,625) (4,609)(b)	24,865 (316)
Income before income taxes and minority interests	(10, 138)	(6,234) 2,431 (c)	
Net income	\$ 15,239	\$(3,803) ======	\$ 11,436 ======
Basic earnings per share	\$ 0.22		\$ 0.17
Basic weighted average shares outstanding	68,004 =====		68,004 ======

The accompanying notes are an integral part of these Pro Forma Combined Financial Statements.

(To be reorganized as Certegy Inc.)

PRO FORMA COMBINED STATEMENT OF INCOME FOR THE YEAR ENDED DECEMBER 31, 2000 Unaudited (In thousands, except per share amounts)

	Historical	Pro Forma Adjustments	Pro Forma Combined
			Compined
Revenues	\$778,562	\$	\$778,562
Operating expenses:			
Costs of services Selling, general and administrative	535,751 95,652	4,700 (a) 1,800 (a)	
	631,403	6,500	637,903
Operating income	1,309	(6,500) (21,735)(b)	140,659 1,309
Income before income taxes and minority interests	147,167	(28, 235)	118,932
Provision for income taxes	(57,609)	` ' '	•
tax	(1,096)		(1,096)
Net income	\$ 88,462 ======		\$ 71,267 ======
Basic earnings per share			\$ 1.06 =====
Basic weighted average shares			
outstanding	67,200 =====		67,200 =====

The accompanying notes are an integral part of these $\ensuremath{\mathsf{Pro}}$ Forma Combined Financial Statements.

(To be reorganized as Certegy Inc.)

PRO FORMA COMBINED BALANCE SHEET

MARCH 31, 2001 Unaudited (In thousands)

	Historical	Pro Forma Adjustments	Pro Forma Combined
ASSETS Current assets: Cash and cash equivalents Trade accounts receivable, net of	\$ 24,771	\$	\$ 24,771
allowance for doubtful accounts of \$2,142	83,800 74,948 6,845 4,894 9,105	 	83,800 74,948 6,845 4,894 9,105
Total current assets	204,363 32,038 174,707 85,389		204,363 32,038 174,707 85,389
Total Assets	\$496,497 ======	\$ =======	\$496,497 ======
LIABILITIES AND SHAREHOLDER'S EQUITY Current liabilities: Trade accounts payable Settlement payables Notes payable Accrued salaries and bonuses Income taxes payable Other current liabilities	\$ 13,102 81,614 443 7,514 7,527 40,506	\$ 	\$ 13,102 81,614 443 7,514 7,527 40,506
Total current liabilities	150,706 11,329 1,240	 275,000 (d) 	150,706 11,329 275,000 1,240
Total liabilities	163,275	275,000	438,275
Minority interests	5,924		5,924
Shareholder's equity: Common stock and paid-in capital Equifax equity investment Cumulative translation adjustment	396,558 (69,260)	121,558 (d) (396,558)(d)	121,558 (69,260)
Total shareholder's equity		(275,000)	52,298
Total Liabilities and Shareholder's Equity	\$496,497 ======	\$ ======	\$496,497 ======

The accompanying notes are an integral part of these Pro Forma Combined Financial Statements.

(To be reorganized as Certegy Inc.)
Notes to Pro Forma Combined Financial Statements
(Unaudited)

1. Pro Forma Combined Statement of Income Adjustments

The following pro forma adjustments were made to the historical combined statements of income of the Company for the three months ended March 31, 2001 and for the year ended December 31, 2000 to reflect the Distribution as if it had occurred on January 1, 2000.

- a. To reflect additional annual expenses of approximately \$6.5 million, which specifically relate to incremental pension expense, insurance costs, corporate headquarters rent, and stand-alone public company costs for audit, director, and stock exchange fees. The increase in pension expense is a result of the Company no longer benefitting from the overfunded status of the consolidated Equifax pension plan.
- b. To reflect interest expense on the \$275 million of debt to be used to fund the cash payment to Equifax in conjunction with the Distribution, at an annual rate of LIBOR plus 100 basis points (7.54% for the year ended December 31, 2000 and 6.34% for the three months ended March 31, 2001), and includes amortization of financing costs over the three-year term of the debt. No Equifax corporate debt or related interest expense is included in the historical financial statements, as these amounts have not historically been allocated to the operating divisions by Equifax.
- c. To reflect the income tax benefit resulting from the pro forma adjustments using the Company's effective tax rate for the period.

2. Pro Forma Combined Balance Sheet Adjustment

The following pro forma adjustment was made to the historical combined balance sheet of the Company as of March 31, 2001 to reflect the Distribution as if it had occurred on March 31, 2001.

d. To reflect the capitalization of the Equifax equity investment and the \$275 million of debt to be used to fund the cash payment to Equifax in conjunction with the Distribution.